Page 1 of 470 Docket No. DG 20-105

Exhibit 49

STATE OF NEW HAMPSHIRE PUBLIC UTILITIES COMMISSION

Liberty Utilities (EnergyNorth Natural Gas) Corp. d/b/a Liberty

Request for Change in Rates

Docket No. DG 20-105

SETTLEMENT AGREEMENT ON PERMANENT RATES

This Settlement Agreement on Permanent Rates ("Settlement Agreement") is entered into

this 29th day of June, 2021, by and among Liberty Utilities (EnergyNorth Natural Gas) Corp. d/b/a

Liberty ("Liberty" or the "Company"), the Staff of the Public Utilities Commission participating

in this proceeding ("Staff"), and the Office of the Consumer Advocate ("OCA") (together,

"Settling Parties"). This Settlement Agreement resolves all issues among the Settling Parties

regarding the Company's request to establish permanent rates in Docket No. DG 20-105, with the

exception of Liberty's request to recover costs incurred to investigate, evaluate, and assess the

development of the Granite Bridge Liquefied Natural Gas tank and related gas pipeline ("Granite

Bridge Project").

SECTION 1. INTRODUCTION AND PROCEDURAL HISTORY

On July 1, 2020, Liberty filed with the New Hampshire Public Utilities Commission 1.1

("Commission") a Notice of Intent to File Rate Schedules pursuant to N.H. Code Admin. Rules

Puc 1604.05. On July 31, 2020, the Company filed its Petition for Permanent and Temporary

Rates ("Petition"), including proposed tariffs and rate schedules, testimony, attachments and other

information supporting the Petition. Liberty's Petition requested that the Commission grant: (1) a

permanent increase in Liberty's distribution rates effective with service rendered on or after

September 1, 2020, designed to yield an increase of \$13,497,250 in annual revenue above the then

Page 2 of 470 Docket No. DG 20-105

Exhibit 49

total Distribution Revenue of \$86,698,2601; (2) temporary rates effective with service rendered on

or after October 1, 2020, designed to yield an increase of \$6,500,000 in annual revenue pending

the Commission's final determination on the Company's request for a permanent rate increase;

and (3) a step adjustment in rates designed to yield an increase of \$5,680,641 in annual revenue to

recover costs associated with approximately \$38 million of capital expenditures projected to be

placed in service during 2020, to be effective no earlier than August 1, 2021, and two additional

step adjustments for capital expenditures projected to be placed in service in 2021 and 2022. The

Petition requested approval of a 10.51 percent return on equity ("ROE"), and a capital structure

consisting of 50.15 percent equity and 49.85 percent debt.

1.2 On July 8, 2020, the OCA filed a letter of participation in this docket pursuant to RSA

363:28. The Commission received no petitions for intervention in this docket by other parties.

On August 19, 2020, the Commission issued Order No. 26.395 suspending Liberty's 1.3

proposed gas service tariff for temporary and permanent rate increases pending further

investigation.

On September 16, 2020, the Commission held a hearing on temporary rates at which 1.4

Liberty amended its request such that temporary rates would be set at current rates, and to approve

a revenue requirement for temporary rates of \$92,890,325, with the \$4,994,290 increase in allowed

revenues accomplished by adjusting the allowed revenue per customer ("RPC") amounts for each

of Liberty's rate classes.² On September 30, 2020, the Commission issued Order No. 26,412

¹ The Distribution Revenue of \$86,698,260 does not include revenue from Indirect COG Revenue and Other Revenue although both are included in total revenue requirement for the purpose of a distribution

base rate case.

² The approved temporary rates revenue requirement of \$92,890,325 does not include \$1,993,588 from

Indirect Cost of Gas (COG) Revenue.

Page 3 of 470 Docket No. DG 20-105

Exhibit 49

approving Liberty's request for temporary rates effective with service rendered on and after

October 1, 2020.

1.5 On October 5, 2020, the Commission approved an initial procedural schedule for

adjudication of the Company's permanent rate request that included multiple rounds of discovery,

technical sessions, settlement conferences, Staff and intervenor testimony, Company rebuttal

testimony, and hearings. On November 20, 2020, the Company filed a motion to amend the

Petition and supplemental pre-filed direct testimony in support of recovery of Granite Bridge

Project costs.³ The Commission subsequently amended the procedural schedule to accommodate

the motion to amend the petition. On March 2, 2021, the Company filed an update to its proposed

permanent rate request, including a total revenue requirement of \$99,786,9844 and a proposed

permanent rate increase of \$4,933,718⁵. Staff and OCA each filed testimony on March 18, 2021,

and the Company filed rebuttal testimony on April 29, 2021.

1.6 Staff's testimony recommended the following: (1) an ROE of 9.0 percent and a capital

structure of 49.21 percent equity and 50.79 percent long term debt; (2) a permanent decrease to

revenue requirement of \$2,240,114 below the temporary rate level; (3) one step increase for 2020

of \$5,157,187 subject to audit, and no step adjustments for future years; and other proposed

OCA's testimony recommended an ROE of 8.9 percent, did not include a adjustments.

³ The motion to amend the Petition was filed pursuant to the Commission's order in Docket No. DG 17-198 in connection with the Granite Bridge Project, in which the Commission held: "Requests for authority

to recover capital project and supply planning costs are appropriately reviewed in a full rate case." Order

No. 26,409 at 13 (Oct. 6, 2020).

⁴ The \$99,786,984 includes \$4,027,585 of Indirect COG Revenue and \$1,207,376 of Other Revenue.

⁵ Bates II-132R. The requested rate increase of \$4,933,718 plus the revenue increase at temporary rates

of \$4,994,290 equates to a revenue increase of \$9,928,008.

Page 4 of 470 Docket No. DG 20-105

No. DG 20-105 Exhibit 49

recommendation for a capital structure different from the Company's proposal, and included other

proposed adjustments and rate design elements.

1.7 In the weeks prior to and following the Company's submission of rebuttal testimony, the

Company, Staff, and the OCA engaged in settlement discussions. Based upon these discussions,

the Settling Parties agreed to the terms of this Settlement Agreement, subject to Commission

approval. The Settling Parties recommend and request that the Commission approve this

Settlement Agreement without modification.

SECTION 2. REVENUE REQUIREMENT

2.1 The Revenue Requirement consists of three subparts: (1) allowed revenue from

distribution rates ("Distribution Revenue"); (2) allowed revenue from cost of gas rates ("Indirect

COG Revenue"); and (3) revenue outside of rates such as special contract revenue and property

leases ("Other Revenue"), (collectively referred to as the "Revenue Requirement").

In a decoupling framework a key metric is the increase in the Distribution Revenue.

Distribution rates are then set to best achieve this Distribution Revenue and are reconciled through

the revenue decoupling adjustment factor each year.

The following table summarizes the revenue requirement and revenue allocation

agreements of the Settling Parties.

4

Docket No. DG 21-130 Exhibit 26 Page 5 of 470

Docket No. DG 20-105 Exhibit 49

A	В	С	D
Proposed Approved Revenue Calculation			
Distribution Revenue (\$84,591,458+\$2,106,802)	\$ 86,698,260		(Exhibit 5, line 1+3)
Indirect COG Revenue	\$ 1,993,587		(Fixed Indirect)
Other Revenue	\$ 1,197,776		(Exhibit 5, line 5)
Total Approved Revenue from Test Year		\$ 89,889,623	(B1+B2+B3)
Increase to Distribution Revenue via Temp Rates Settlement	\$ 4,994,290		(Exhibit 5, line 2+4)
Total Approved Revenue After Temp Rates Settlement		\$ 94,883,913	(C4+B5)
Order 26,412 did not include COG of \$1,993,587	\$ (1,993,587)		Existing Indirect COG Allocation
Temp Rate Order 26,412 rev requirement for temp rates		\$ 92,890,326	(C6+B7) Temp Rate Order 26,412
Settlement Increase to Approved Rev above Temp Rates	\$ 1,300,000		Perm Rates Settlement Agreement
Total Proposed Approved Revenue Increase from Test Year	\$ 6,294,290		(B5+B9)
		\$ 96,183,913	1
Approved Distrib Revenue (Target Revenue on Rates 5)	\$ 91,082,950		(C11-B13-B14) (tie to Rates 5)
Approved Indirect COG Revenue	\$ 3,893,587		(B17)
Other Revenue	\$ 1,207,376		(B3+\$9,600 from Audit Issue 6)
A	В	С	D
Proposed Approved Revenue Allocation			
Total Increase in Approved Revenue	\$ 6,294,290		(B5+B9)
Amount of increase allocated to Indirect COG 11/1/21*	\$ 1,900,000		Perm Rates Settlement (COSS)
Total Indirect COG Revenue	\$ 3,893,587		(B16-B7)
Remainder of increase allocated to Dist Rates	\$ 4,394,290		(B15-B16)
Adj to Rates (Permanent Increase less Temp Increase)	\$ (600,000)		(B18-B5)
Total Revenue Increase Above Temp per Settlement	\$ 1,300,000		(B16+B19)
Indirect COC increase to be Allocated to EN & Kasas	¢ 1,000,000		(B16)
			1, ,
` ,	, .		Perm Rates Settlement Agreement
Indirect COG Allocation to Energy North*	\$ 1,691,8/1		(B21-B22)
*Reconciled to 10/1/2020			
	Proposed Approved Revenue Calculation Distribution Revenue (\$84,591,458+\$2,106,802) Indirect COG Revenue Other Revenue Total Approved Revenue from Test Year Increase to Distribution Revenue via Temp Rates Settlement Total Approved Revenue After Temp Rates Settlement Order 26,412 did not include COG of \$1,993,587 Temp Rate Order 26,412 rev requirement for temp rates Settlement Increase to Approved Rev above Temp Rates Total Proposed Approved Revenue Increase from Test Year Approved Distrib Revenue (Target Revenue on Rates 5) Approved Indirect COG Revenue Other Revenue	Proposed Approved Revenue Calculation Distribution Revenue (\$84,591,458+\$2,106,802) \$86,698,260 Indirect COG Revenue \$1,993,587 Other Revenue \$1,197,776 Total Approved Revenue from Test Year Increase to Distribution Revenue via Temp Rates Settlement \$4,994,290 Total Approved Revenue After Temp Rates Settlement Order 26,412 did not include COG of \$1,993,587 \$(1,993,587) Temp Rate Order 26,412 rev requirement for temp rates Settlement Increase to Approved Rev above Temp Rates Settlement Increase to Approved Rev above Temp Rates \$1,300,000 Total Proposed Approved Revenue Increase from Test Year Approved Distrib Revenue (Target Revenue on Rates 5) \$91,082,950 Approved Indirect COG Revenue \$3,893,587 Other Revenue \$6,294,290 Anount of increase allocated to Indirect COG \$11/1/21* \$1,900,000 Total Indirect COG Revenue \$3,893,587 Remainder of increase allocated to Dist Rates \$4,394,290 Adj to Rates (Permanent Increase less Temp Increase) \$600,000 Total Revenue Increase Above Temp per Settlement \$1,300,000 Indirect COG Allocation to Keene (Production Cost)* \$1,691,871	Proposed Approved Revenue Calculation Distribution Revenue (\$84,591,458+\$2,106,802) \$86,698,260 Indirect COG Revenue \$1,993,587 Other Revenue \$1,197,776 Total Approved Revenue from Test Year \$89,889,623 Increase to Distribution Revenue via Temp Rates Settlement \$4,994,290 Total Approved Revenue After Temp Rates Settlement \$5,43,935,877 Temp Rate Order 26,412 rev requirement for temp rates \$1,300,000 Total Proposed Approved Revenue Increase from Test Year \$6,294,290 Total Approved Revenue per Settlement for the Test Year \$96,183,913 Approved Distrib Revenue (Target Revenue on Rates 5) \$91,082,950 Approved Indirect COG Revenue \$3,893,587 Other Revenue \$1,207,376 A B C Proposed Approved Revenue Allocation Total Increase in Approved Revenue \$5,294,290 Amount of increase allocated to Indirect COG 11/1/21* \$1,900,000 Total Indirect COG Revenue \$3,893,587 Remainder of increase allocated to Dist Rates \$4,394,290 Adj to Rates (Permanent Increase less Temp Increase) \$6,000,000 Total Revenue Increase Above Temp per Settlement \$1,300,000 Indirect COG increase to be Allocated to EN & Keene \$1,900,000 Indirect COG Allocation to Keene (Production Cost)* \$208,129 Indirect COG Allocation to Keene (Production Cost)* \$208,129 Indirect COG Allocation to Energy North* \$1,691,871

The Company shall be allowed a Revenue Requirement increase of \$6,294,290, which is a \$1.3 million increase above the level provided by temporary rates. Based on the cost of service study the Settling Parties agree that \$1,900,000 of the \$6,265,231 revenue increase shall be applied to the next COG proceeding as described in Section 13.3, effective for service rendered on and after November 1, 2021. In addition, a surcharge shall be added through the LDAC to reconcile this \$1,900,000 addition to the Indirect COG Revenue back to the effective date of temporary rates of October 1, 2020. The remaining \$4,394,290 in revenue requirement shall be collected from distribution rates effective for service rendered on and after August 1, 2021, to be reconciled back to October 1, 2020, the effective date of temporary rates, consistent with Order No. 26,412 (Sept.

Page 6 of 470 Docket No. DG 20-105

Exhibit 49

30, 2020) in this proceeding. Because the temporary rate revenue increase was an increase of

\$4,994,290, the result is that the permanent rate change shall be a decrease of \$600,000 from

temporary rates. In addition, the reconciliation from temporary rates to permanent rates shall be a

reduction to account for the over-collection of distribution rates that occurred from October 1,

2020, to the implementation of permanent rates.

2.2 The agreed Revenue Requirement increase of \$6,265,231 includes the following cost of

service elements: (1) recovery of costs associated with the New Hampshire Department of

Administrative Services special contract of \$1,047,589 over three years (\$349,196/year); (2)

amortization of excess accumulated deferred income tax ("EADIT") for protected property over

28.93 years and for non-protected property over 20 years as further described in Section 3.3; (3)

amortization of remaining 'costs to achieve' from Docket No. DG 06-107 of \$48,197 over three

years (\$16,066/year)⁶; (4) continued amortization of \$1,657,796 per year for the depreciation

reserve imbalance amortization, subject to adjustment based upon updated cost of removal and

average service life information to be determined pursuant to Section 3.2 of this Settlement

Agreement; and (5) the cost of capital provided in Section 4 of this Settlement Agreement.

2.3 The agreed revenue increase reflects adjustments that have been made in order to reach

settlement and shall not establish precedent for future rate proceedings.

⁶ As referenced in Schedule RR-EN-3-6, adjustment 6 from the March 2, 2021, Corrections and Update

filing.

Page 7 of 470 Docket No. DG 20-105

D. DG 20-105 Exhibit 49

SECTION 3. RATE BASE

3.1 The lead/lag days in Cash Working Capital shall be 25.72.

3.2 The Company shall perform a study during calendar year 2021 based on a sampling of

different sized 2021 mains and services capital projects to determine the cost of removal

percentages that should be applied to mains and services. The Company shall also obtain a new

full depreciation study based on 2021 end of year plant balances, which study shall review and

incorporate the results of the cost of removal study. The depreciation study shall be used to assess

and update the depreciation reserve imbalance by making the necessary adjustments to the annual

amortization amount of \$1,657,796. The determination of the depreciation lives and rates

applicable to various plant accounts shall adjust the annual depreciation expense amounts. Liberty

shall file the updated depreciation study along with a report on its findings by May 1, 2022, for

review by Staff and the OCA as part of the second step adjustment review. Any adjustments based

on the updated depreciation study shall be reflected in the second step increase to take effect on

August 1, 2022.

3.3 As a result of tax law changes from the Tax Cuts and Jobs Act of 2017, Liberty has a total

of \$37,855,214 of excess accumulated deferred income tax liabilities that will be amortized as

credits to the benefit of customers as follows: (i) protected items totaling \$33,434,927 will be

amortized as annual credit of \$1,155,718 over 28.93 years; and (ii) unprotected items totaling

\$4,420,287 will be amortized as an annual credit of \$221,014 over 20 years.

7

Page 8 of 470 Docket No. DG 20-105

SECTION 4. COST OF CAPITAL

4.1 The Company shall be allowed a return on equity of 9.3 percent.

4.2 The Settling Parties have agreed that a capital structure of 52.0 percent equity and 48.0 percent debt shall be used for purposes of determining the Company's revenue requirement in this proceeding, which results in the following pre-tax and after-tax weighted average costs of capital based on the current federal and state tax rates and the Company's current cost of long-term debt:

Description	ption Capital Cost of Structure Capital Weighted Cost of Capital		Cost of	Tax Rate	Pre-Tax
Common Stock	52.00%	9.30%	4.84%	27.08%	6.64%
Long-Term Debt	48.00%		2.12%	27.0070	2.12%
Total	100.00%		6.96%		8.76%

SECTION 5. STEP ADJUSTMENTS

- 5.1 The Company shall be allowed two step adjustments as follows:
 - (a) Step 1 shall reflect an increase to account for certain capital projects placed in service during calendar year 2020 and shall be implemented on August 1, 2021. This first step adjustment reflects adjustments that have been made to the revenue requirement in order to reach settlement. The first step shall be subject to the following conditions:
 - i. The revenue requirement for this step shall be capped at a \$4.0 million increase to annual Distribution Revenue.
 - ii. The step shall be based on the projects closed to plant in 2020, and shall exclude new business/growth-related projects.

Page 9 of 470 Docket No. DG 20-105

Exhibit 49

iii. The projects that may be included in the step are identified in the listing

attached as Appendix 1.

iv. Local property taxes shall not be included in the calculation and will be

recovered through the Property Tax Adjustment Mechanism in Section 6 of

the Settlement Agreement. State utility property taxes for all projects listed

in Appendix 1, calculated using the statutory tax rate in RSA 83-F:2, shall

be included in the step adjustment calculation, shall count toward the cap,

and shall be given first priority of recovery.

(b) Step 2 shall reflect an increase to account for certain capital projects placed in

service during calendar year 2021 and shall be effective August 1, 2022. This

second step adjustment reflects adjustments that have been made to the revenue

requirement in order to reach settlement. The second step shall be subject to the

following conditions:

i. The revenue requirement for this step shall be capped at a \$3.2 million

annual increase to Distribution Revenue from the projects referenced in iii.

below.

ii. The step shall be based on the projects closed to plant in 2021, and shall

exclude new business/growth-related projects.

iii. The projects and programs that may be included in this step are identified

in the listing attached as Appendix 2, including Keene CNG Phase 1 costs

as further described in Section 7.2. The Settling Parties agree that the

Company may substitute other similar non-growth projects prior to the

Page 10 of 470

Docket No. DG 20-105 Exhibit 49

commencement of the review period if projects identified in Appendix 2 are

not deployed.

iv. Local property taxes shall not be included in the calculation and will be

recovered through the Property Tax Adjustment Mechanism in Section 6 of

the Settlement Agreement. State property taxes for all projects listed in

Appendix 2 as it may be adjusted pursuant to iii. above, calculated using the

statutory tax rate in RSA 83-F:2, shall be included in the step adjustment

calculation and shall be given first priority of recovery.

v. The step adjustment shall include adjustments resulting from the updated

depreciation study as provided in Section 3.2 of the Settlement Agreement.

5.2 For the second step, the Company shall make a filing on or before April 8, 2022, with rates

effective August 1, 2022. The filing shall include, at least, the following documentation and

process steps:

(a) The Company shall provide the amount of the investments to be included in the

step increase (by project) and detailed project descriptions including the initial

budget, the final cost, the treatment of any related retirements, and the date each

project was booked to plant in-service.

(b) For each project Liberty shall provide all Company project documents including,

but not limited to, Business Cases, Capital Project Expenditure Applications,

Change Order Forms, Project Close Out Reports, and work orders.

(c) Staff and/or the OCA may request additional information after reviewing the initial

filings.

Page 11 of 470 Docket No. DG 20-105

Exhibit 49

(d) The Company shall propose a rate increase effective August 1, 2022, to recover the

revenue requirement associated with the second step adjustment up to the \$3.2

million cap.

5.4 For the second step, if the actual cost of the capital additions is less than the budgeted

amounts, the actual amounts shall be used to calculate the step adjustments. If the actual cost of

the capital additions exceeds the budgeted amounts for a particular project, the Company may seek

recovery of the excess through this step adjustment process, subject to the cap. The Company may

otherwise seek recovery in its next rate case for any above-budget investments not approved in a

step adjustment described here. The revenue requirement for the step adjustments will be

calculated in a manner similar to that used in the Company's filing seeking approval of the first

step adjustment. The step increase shall be subject to Staff audit and reconciliation based on the

results of the audit, as approved by the Commission.

5.5 Nothing in this Settlement Agreement shall preclude Staff or the OCA from contesting the

prudence of individual investments requested for recovery within the step increases.

5.6 The Company shall not request recovery of any capital costs associated with plant placed

in service outside of the above-described step adjustments until the Company's next distribution

rate case filing, which shall be based on a test year ending no sooner than December 31, 2022.

SECTION 6. PROPERTY TAX ADJUSTMENT MECHANISM

6.1 The Company shall be authorized to implement a Property Tax Adjustment Mechanism

("PTAM") to allow the Company to request recovery or refund of local property tax expenses, as

compared to the amount in base rates, beginning with the April 1, 2020, through March 31, 2021,

property tax year. Consistent with RSA 72:8-e, property tax over- or under-recoveries as

Page 12 of 470 Docket No. DG 20-105

Exhibit 49

compared to the amount in base distribution rates shall be adjusted annually through the PTAM.

The amount included in base distribution rates for local property tax expense shall be \$8,924,897,

as shown in Appendix 3.

6.2 The PTAM shall be a fully reconciling property tax adjustment mechanism except for the

exclusion of State of New Hampshire utility property taxes. The State of New Hampshire utility

property taxes levied on the step eligible investments shall be collected as part of the step revenue

requirement as detailed in Section 5 above.

6.3 On an annual basis beginning with the property tax year that commenced April 1, 2020,

actual local property tax amounts from the property tax bills received during a calendar year shall

be compared against the amount in base rates as of the March 31 end of each property tax year,

and any variances will be reconciled through the PTAM. With respect to the initial year of

reconciliation, local property tax bills received during calendar year 2020 will be compared to the

calculated amount in distribution rates through March 31, 2021, which includes the recoupment of

the property tax amount reconciled between temporary and permanent rates. Reconciliation in

subsequent years will be handled in a similar fashion. Annual property tax billed amounts shall

be adjusted for any credits received due to abatement proceeds received for tax years preceding

the test year. The PTAM shall reconcile and provide for recovery or credit for any over- or under-

recoveries beginning with the April 1, 2020, through March 31, 2021, property tax year.

6.4 The PTAM shall be established annually based on a full reconciliation with monthly

compounded interest for any over- or under-recoveries occurring in prior year(s). Interest shall be

calculated at the prime rate, to be fixed on a quarterly basis and to be established as reported in

⁷ Property tax bills received during a calendar year cover the annual property tax year that begins on April

1 and runs through the following March 31.

Page 13 of 470 Docket No. DG 20-105

Exhibit 49

The Wall Street Journal on the first business day of the month preceding the calendar quarter

("Prime Rate").

6.5 The Company shall submit the initial PTAM filing covering the period described in Section

6.3 on or before August 20, 2021, and the PTAM distribution rate adjustment for that period shall

be effective with service rendered on and after November 1, 2021. Filings covering subsequent

property tax years shall be made on or before March 10 using property tax bills received during

the prior calendar year for adjustments to distribution rates effective May 1.

6.6 Billing determinants used for adjusting the Revenue Per Customer (RPC) amounts for the

initial PTAM adjustment, as described in Section 11, for the purpose of the PTAM adjustment,

shall be the same as the calendar year 2020 billing determinants used for the purpose of the first

step adjustment (see Section 5.1). Billing determinants used for subsequent PTAM adjustments

shall be the annual billing determinants for the calendar year that corresponds to the year that the

local property tax bills were received for that particular PTAM reconciliation.

SECTION 7. KEENE CONVERSION TO CNG

7.1 Keene Cost of Gas. The Company shall recover one-half of the incrementally higher CNG

supply costs as compared to the propane supply cost,8 incurred from the commencement of CNG

service through October 31, 2021, to be recovered through inclusion over one year in the next

Keene cost of gas during the winter or summer periods consistent with the season in which the

incremental costs were originally incurred. The Company shall provide the supporting calculations

8 Incremental CNG supply cost/savings shall be calculated by multiplying the CNG therm purchases by the difference between the average per therm CNG supply cost and the propane supply costs for the applicable summer/winter period. Average CNG supply costs shall include all CNG supplier charges properly allocated between summer and winter periods. Average propane supply costs shall include Mont Belvieu propane pricing, transportation costs, and Broker Fee.

Page 14 of 470 Docket No. DG 20-105

DG 20-105 Exhibit 49

in the Winter 2021-2022 Keene Cost of Gas filing. Incremental CNG supply costs through the

2020-2021 winter period are provided in Appendix 4.

(a) Beginning November 1, 2021, if the CNG supply cost is higher than the propane supply

cost as described in footnote 8, the Company shall recover one-half of the incrementally

higher CNG supply cost, as determined through the cost of gas reconciliation process. If

the CNG supply cost is lower than the propane supply cost, the Company shall recover and

retain the full amount of the incrementally lower CNG supply cost up to the amount of

incrementally higher CNG costs accrued since the commencement of CNG service, which

have not then been recovered from customers, at which point the Company shall recover

and retain one-half of the incrementally lower CNG supply costs. Reconciliation of the

incremental CNG supply costs shall occur semi-annually in the Winter and Summer Cost

of Gas filings, as applicable. A sample calculation of incremental CNG supply

costs/savings, using non-confidential, hypothetical pricing, is illustrated in Appendix 5.

(b) CNG demand costs shall be allocated 75% to the winter period and 25% to the

summer period until such time as otherwise determined by the Commission in a future

proceeding. Any change in allocation shall be implemented during a Winter Cost of Gas

filing.

7.2 Keene Expansion. Phase 1 of the Keene conversion to natural gas shall consist of: (1)

installation of the existing temporary CNG facility; (2) conversion of the propane-air customers

premises at the Monadnock Marketplace to natural gas as of the date of this Settlement; and (3)

acquisition of customers at any additional premises not currently physically connected to the gas

utility system in Keene after the date of this settlement who would be served CNG from both the

existing CNG temporary facility and through existing mains. An "existing main" is one that has

Page 15 of 470 Docket No. DG 20-105

Exhibit 49

a Maximum Allowable Operating Pressure ("MAOP") greater than 0.5 psig and that can be safely

used to deliver natural gas, satisfying all applicable safety rules and regulations. Inclusion of

additional customers in Phase 1 as described in this section (3) shall not change the time frame of

the risk sharing mechanism. Revenue from new customers at existing premises converted from

propane-air to CNG shall not be included in the discounted cash flow analysis of the risk sharing

mechanism. The Settling Parties agree that the existence of Phase 1 shall not be used to justify any

further development of CNG in Keene beyond Phase 1.

(a) The Company shall be allowed to seek recovery of the Phase 1 costs as part of the

second step adjustment (excluding the cost of the Production Avenue land), subject to

the risk sharing mechanism established in Order No. 26,122 (April 27, 2018) at 39 as

clarified in Order No. 26,274 (July 26, 2019). The Company retains the right to seek

recovery of the cost of the Production Avenue land at a future time. As part of the

second step adjustment, the Company shall be allowed to update the recovery of the

Phase 1 costs to account for the revenue and costs associated with additional Phase 1

customers who began taking service or committed to take service on or before August

1, 2022, the effective date of the step adjustment, subject to the risk sharing mechanism

established in the above orders. If a customer committed to take service does not

actually take service prior to the next base rate case then the August 1, 2022, risk

sharing adjustment calculation shall be recalculated and any shortfall shall be adjusted

accordingly to reflect the actual service. The adjustment for the risk-sharing

mechanism for the Phase 1 investments included in rate base as of December 31, 2019,

is \$21,736 as outlined in Appendix 6, which amount is already taken into account in

the revenue requirement described in Section 2.

Page 16 of 470 Docket No. DG 20-105

Exhibit 49

(b) Phase 2 shall not be implemented until Liberty satisfies the conditions of Order No.

26,122 (Apr. 27, 2018) as clarified in Order No. 26,274 (July 26, 2019), and the

Company's tariff, as applicable. Incremental revenue from Phase 1 new customers

may not be used to justify Phase 2. If a customer who committed to take service does

not actually take service prior to the effective date of temporary rates in the next base

rate case then the risk sharing adjustment calculation shall be recalculated and any

shortfall shall be adjusted accordingly to reflect the actual service.

(c) The Settling Parties agree that prior to the implementation of any future phase the ten

year net present value of such phase shall not be negative.

(d) The following definitions shall apply to Phase 2 and future phases of the conversion of

Keene customers from propane-air to natural gas (including renewable natural gas):

(i) A "phase" shall be one or more customer additions or conversions which require

the installation of 100 feet or more of new main. The addition or conversion of new

customers requiring less than 100 feet of new main shall not be a "phase," but shall

comply with the Company's line extension tariff.

(ii) Converting existing main from propane-air to natural gas shall be considered a

"new main" unless the existing main has an MAOP of greater than 0.5 psig and can

be safely used to deliver natural gas, satisfying all applicable safety rules and

regulations.

SECTION 8. PELHAM EXPANSION

8.1 The agreed revenue requirement includes a reduction of (\$61,871) pursuant to the risk

sharing mechanism approved by the Commission in Docket No. DG 15-362, in Order 29,987,

16

Page 17 of 470 Docket No. DG 20-105

DG 20-105 Exhibit 49

based on the Company's discounted cash flow ("DCF") analysis that compares the revenue

requirement of the take station on the Concord Lateral with the anticipated annual revenue from

new Pelham customers. The anticipated revenue includes expected revenue from a large industrial

customer who signed an agreement to take service in 2016, has not yet taken service for various

reasons, but is still expected to take service from Liberty in the near future.

8.2 The Settling Parties agree to maintain this adjustment for the August 1, 2021 – July 31,

2022 rate year. If the large industrial customer has not taken service from Liberty on or before

August 1, 2022, the anticipated revenue from this customer shall be removed from the DCF

calculation and the adjustment to remove revenue from that customer shall be made in the second

step adjustment.

SECTION 9. INATGAS COSTS

9.1 Consistent with the Commission's determination in Docket No. DG 17-048, and in light of

the increased capital costs associated with the iNATGAS CNG fueling facility, the Settling Parties

agree that there will be a reduction of the overall revenue requirement of \$301,747 for ratemaking

purposes, such amount already being reflected in the Revenue Requirement increase described in

Section 2.1, until the time of the Company's next distribution rate case.

SECTION 10. GRANITE BRIDGE COST RECOVERY EXCLUDED FROM

SETTLEMENT

10.1 The Company's Petition, as amended, requests Commission approval for recovery of the

costs incurred to investigate, evaluate, and assess the development of the Granite Bridge Project,

which the Company calculates as approximately \$7.5 million (the "Granite Bridge Project Costs").

The Settling Parties agree that the current rate case is the appropriate docket for adjudication of

Page 18 of 470 Docket No. DG 20-105

Exhibit 49

the Granite Bridge Project Costs. The recovery of the Granite Bridge Project Costs are excluded

from this Settlement Agreement and shall be litigated within this proceeding.

SECTION 11. RATES AND RATE DESIGN

Decoupling. As this is the first general rate case since the implementation of decoupling, 11.1

the Settling Parties agree that this is an opportunity to clarify the process surrounding the

decoupling mechanism and the associated tariff language. The Agreement consists of five points

regarding decoupling:

(a) The calculation of the revenue per customer (RPC) for permanent rates shall include:

i. the end of year calendar month bill count adjustment in the denominator of the

calculation for the test year;

ii. the volumetric therms used for the calculation shall reflect the monthly bill counts

adjusted for the end of year calendar month bill counts; and

iii. the RPC for the permanent rate increase shall not change until the next rate case.

(b) The calculation of the incremental revenue per customer for subsequent non-rate case rate

changes such as, but not limited to, step adjustments, property tax reconciliation, and

temporary rates, shall (i) use actual calendar month bill counts for the same time period

being used to determine the calculation of each new RPC, and (ii) add each incremental

RPC to the RPC from the rate case.

(c) Because the MEP Premium⁹ is not subject to decoupling, the RPC calculations that are

used to calculate the allowed revenue and the Revenue Decoupling Adjustment Factor shall

not include the MEP Premium.

⁹ MEP Premium is the premium charge to customers in the Managed Expansion Program. In Docket No. DG 17-048, the Commission approved the decoupling proposal in the settlement (Order 26,122 at 45) which states in part:

"Managed Expansion Program customers are subject to decoupling, but the expansion surcharge dollars (i.e., the

30% distribution premium) are excluded from the decoupling calculation" (DG 17-048 Exhibit 29 at 11)

Page 19 of 470 Docket No. DG 20-105

DG 20-105 Exhibit 49

(d) Each month the Company shall record a Revenue Decoupling Adjustment (RDA) in the

balance sheet RDA Accounts in accordance with generally accepted accounting principles,

including: (i) the Revenue Decoupling Adjustment which is the difference between the

Monthly Allowed Revenue and the Monthly Actual Distribution Revenue; (ii) the

reconciliation amounts collected or distributed through the RDAF recorded in the RDA

Accounts for each Customer Class Group; and (iii) the accrued interest on the RDA

Accounts calculated on the average monthly balance using the prime lending rate.

(e) The RPC calculations, including equivalent bill calculations and associated usage per

customer, shall be submitted with each rate increase filing and the associated tariff

compliance filing.

The tariff has been amended as shown in Appendix 11 to effectuate the above understanding.

11.2 Revenue Calculations.

(a) Indirect Gas Costs. The Settling Parties agree to include \$3,893,588 of Indirect COG

Revenue for recovery through the COG which includes \$1,900,000 of the revenue

requirement increase identified in 2.1. The Indirect COG Revenue shall not change until

the next rate case. The Indirect COG Revenue collected through COG rates shall be

included in the revenue calculation for all future filings. The Settling Parties agree that the

\$3,893,588 of Indirect COG Revenue includes \$206,248 of propane production costs

allocated to Keene and \$1,881 of Phase 1 conversion costs. The Keene COG tariff shall

reflect the combined total of \$208,129 as indirect gas costs.

(b) The Settling Parties agree that for purposes of calculating revenue in filings to the

Commission the Company shall use the following guidelines:

Page 20 of 470 Docket No. DG 20-105

Exhibit 49

i. The revenue per customer for low-income customers shall not be different from

customers not categorized as low-income. The discount provided to low-income

customers shall not be included in the revenue calculation as it is reconciled

separately through the Gas Assistance Program (GAP) portion of the LDAC.

11.3 The Company's customer charges shall be set at the levels identified in Appendix 8.

Thereafter, the Company's residential customer charges shall remain as set until the Company's

next rate case. Specifically, any base rate increase and any surcharges or sur-credits for residential

customers provided for in this Settlement Agreement shall be collected solely through changes in

consumption charges for residential customers.

SECTION 12. TEST YEAR

12.1 The test year for the Company's next general distribution rate case shall be no sooner than

the twelve-month period ending December 31, 2022.

SECTION 13. OTHER ISSUES

On or before November 30, 2021, the Settling Parties shall meet to review a list of 13.1

regulatory reports currently required by the Commission, and discuss areas for potential

elimination, consolidation, decreased frequency, and other measures to streamline reporting

requirements. The Settling Parties shall submit individual or collective recommendations to the

Commission following such meeting.

13.2 Liberty agrees to include in its Form F-1 quarterly rate of return reports adjustments of

iNATGAS and risk sharing disallowances as described above. Reporting to include Indirect COG

Revenue. Until changed in a future rate proceeding, the adjustment amounts shall be as follows:

iNATGAS of \$301,747, Keene risk sharing of \$21,736, and Pelham risk sharing of \$61,871.

Page 21 of 470

Docket No. DG 20-105 Exhibit 49

SECTION 14. RECOUPMENT

14.1 Subject to Staff audit and adjustment for the difference between estimated and actual

expense, the Company shall recover over one year \$856,864.64 in rate case expenses commencing

on November 1, 2021, through the LDAC mechanism, as shown on Appendix 9. The Company

agrees to submit by August 1, 2021, an accounting of its rate case expenses, with appropriate

supporting documentation, for review by Staff and the OCA and subsequent approval by the

Commission. Staff shall provide its recommendation for rate case expense recovery to the parties

as soon as reasonably possible, and the Company shall be authorized to recover the approved rate

case expenses beginning with service rendered as of November 1, 2021. Any necessary

adjustments to rate case expenses, including adjustments for any invoices received subsequent to

the August 1, 2021, filing date, will be reviewed as part of the LDAC proceeding.

14.2 The permanent rate increase agreed to in Section 2.1 shall be effective for all service

rendered on and after August 1, 2021.

14.3 **Distribution Revenue Reconciliation.** The difference between the Distribution Revenues

obtained from the rates prescribed in the temporary rate order, Order No. 26,412, and the

Distribution Revenues that would have been obtained under the rates designed to collect the

Approved Distribution Revenue finally determined after review and approval of this Settlement

Agreement, if applied during the period that the temporary rate order was in effect from October

1, 2020, to July 31, 2021, shall be returned to customers over a period of 12 months beginning

with service rendered as of November 1, 2021. The total estimated amount of recoupment is

(\$570,933), as shown on Appendix 10, and shall be reconciled through the LDAC mechanism.

Any necessary adjustments to the recoupment amount will be reviewed as part of the LDAC

proceeding.

21

Page 22 of 470 Docket No. DG 20-105

. DG 20-105 Exhibit 49

14.4 Indirect COG Revenue Recoupment. The difference between the Indirect COG

Revenues obtained in accordance with Exhibit 5, of \$1,993,587, and the Indirect COG Revenues

that would have been obtained under the Approved Indirect COG Revenues finally determined

after review and approval of this Settlement Agreement, if applied during the period since the

temporary rate order, is \$1,900,000 and shall be recovered over a period of 12 months through the

LDAC mechanism. Any necessary adjustments to the recoupment amount will be reviewed as part

of the LDAC proceeding.

SECTION 15. EFFECTIVE DATE

15.1 This Settlement Agreement is subject to and shall become effective upon Commission

approval, with new permanent distribution rates to become effective as of August 1, 2021. The

Settling Parties shall use best efforts to obtain Commission approval on or before July 30, 2021.

SECTION 16. GENERAL PROVISIONS

16.1 A revised tariff intended to incorporate the provisions of this Settlement Agreement is

included as Appendix 11.

16.2 This Settlement Agreement is expressly conditioned upon the Commission's acceptance of

all its provisions, without change or condition. If the Commission does not accept this Settlement

Agreement in its entirety, without change or condition, or if the Commission makes any findings

that go beyond the scope of this Settlement Agreement, and any of the Settling Parties notify the

Commission within five business days of their disagreement with any such changes, conditions,

or findings, the Agreement shall be deemed to be withdrawn, in which event it shall be deemed to

be null and void and without effect, shall not constitute any part of the record in this proceeding,

and shall not be relied on by Staff or any party to this proceeding or by the Commission for any

other purpose.

Page 23 of 470 Docket No. DG 20-105

Exhibit 49

16.3 Under this Settlement Agreement, the Settling Parties agree to this joint submission to the

Commission as a resolution of the issues specified herein only.

16.4 The Settling Parties agree that the Commission's approval of this Settlement Agreement

shall not constitute continuing approval of, or precedent for, any particular principle or issue, but

such acceptance does constitute a determination that the adjustments and provisions stated in their

totality are just and reasonable and consistent with the public interest and that the rates

contemplated will be just and reasonable under the circumstances.

16.5 This Settlement Agreement shall not be deemed an admission by any of the Settling Parties

that any allegation or contention in this proceeding by any other party, other than those specifically

agreed to herein, is true and valid. This Settlement Agreement shall not be construed to represent

any concession by any Settling Party hereto regarding positions taken with respect to the

Company's proposals in this docket, nor shall this Settlement Agreement be deemed to foreclose

any Settling Party in the future from taking any position in any subsequent proceedings. The

amounts associated with each of the settlement adjustments detailed herein are liquidated amounts

that reflect a compromise of all the issues in this proceeding

16.6 With the exception of pre-filed testimony and supporting documentation related to Granite

Bridge Project Costs, the pre-filed testimony and supporting documentation previously provided

in this proceeding are not expected to be subject to cross-examination by the Settling Parties, which

would normally occur in a fully litigated case. The Settling Parties agree that all such pre-filed

testimony and supporting documentation should be admitted as full exhibits for the purpose of

consideration of this Settlement Agreement, and be given whatever weight the Commission deems

appropriate. Consent by the Settling Parties to admit all such pre-filed testimony without challenge

23

Page 24 of 470 Docket No. DG 20-105

et No. DG 20-105 Exhibit 49

does not constitute agreement by any of the Settling Parties that the content of the pre-filed

testimony is accurate or that the views of the witnesses should be assigned any particular weight

by the Commission. The resolution of any specific issue in this Settlement Agreement does not

indicate the Settling Parties' agreement to such resolution for purposes of any future proceedings,

nor does the reference to any other document bind the Settling Parties to the contents of, or

recommendations in, that document for purposes of any future proceeding. The Commission's

approval of the recommendations in this Settlement Agreement shall not constitute a determination

or precedent with regard to any specific adjustments, but rather shall constitute only a

determination that the rates resulting from, and other specific conditions stated in this Settlement

Agreement are just and reasonable. The Settling Parties agree to forego cross-examining witnesses

regarding their pre-filed testimony and, therefore, the admission into evidence of any witness's

testimony or supporting documentation shall not be deemed in any respect to constitute an

admission by any party to this Agreement that any allegation or contention in this proceeding is

true or false, except that the sworn testimony of any witness shall constitute an admission by such

witness

16.7 The rights conferred and the obligations imposed on the Settling Parties by this Settlement

Agreement shall be binding on or inure to the benefit of any successors in interest or assignees as

if such successor or assignee was itself a signatory party. The Settling Parties agree to cooperate

in advocating that this Settlement Agreement be approved by the Commission in its entirety and

without modification.

16.8 The discussions that produced this Settlement Agreement have been conducted on the

understanding that all offers of settlement and settlement discussions relating to this docket shall

be confidential, shall not be admissible as evidence in this proceeding, shall be without prejudice

Page 25 of 470

Docket No. DG 20-105 Exhibit 49

to the position of any party or participant representing any such offer or participating in any such

discussion, and are not to be used in connection with any future proceeding or otherwise. The

content of these negotiations, including any documents prepared during such negotiations for the

purpose of reaching a settlement, shall be privileged and all offers of settlement shall be without

prejudice to the position of any party presenting such offer.

16.9 This Settlement Agreement may be executed by facsimile and in multiple counterparts,

each of which shall be deemed to be an original, and all of which, taken together, shall constitute

one agreement binding on all Settling Parties.

SECTION 17. CONCLUSION

17.1 The Settling Parties affirm that the proposed Settlement Agreement will result in just and

reasonable rates and should be approved by the Commission.

Dated: June 29, 2021

Liberty Utilities (EnergyNorth Natural Gas) Corp.

MAluellan

d/b/a Liberty

By its Attorney, Michael J. Sheehan

Dated: June 29, 2021 Staff of the New Hampshire Public Utilities

Commission

Is Paul B. Dexter

By its Attorney, Paul B. Dexter

Docket No. DG 21-130 Exhibit 26 Page 26 of 470

Docket No. DG 20-105 Exhibit 49

Dated: June 29, 2021 Office of the Consumer Advocate

By the Consumer Advocate, Donald M. Kreis

amm

Docket No. DG 21-130 Exhibit 26 Page 27 of 470 Docket No. DG 20-105 Exhibit 49

Table of Appendices

		<u>Bates</u>
1.	Step 1 Projects	028
2.	Step 2 Projects and Keene Phase 1	030
3.	Calculation of Local Property Tax Included in Base Distribution Rates	032
4.	Incremental CNG Supply Costs through 2020/2021	033
5.	Illustration of Calculation of incremental CNG supply costs/savings	034
6.	Keene Risk Sharing Adjustment Calculation.	035
7.	RPC Calculation	036
8.	New Rates and Bill Impacts (includes the capital step increase)	037
9.	Rate Case Expenses	047
10.	Recoupment of distribution	048
11.	Complete Revised Tariff	050

Docket No. DG 20-105 Appendix 1 Page 1 of 2

Liberty Utilities (EnergyNorth Natural Gas) d/b/a Liberty Non-Growth Projects Placed in Service During 2020

Project Number	Project Name	Priority A Postulatory Programs	In service \$\$*	In service Date
8840-1911	Main Replacement LPP-Restoration	4. Regulatory Programs	\$5,419,088	various
8840-1912	Install Main Baboosic Lake Rd at FE Everett Turnpike	5. Discretionary	(\$21,278)	carryover from 2019
8840-1921 8840-1933	Upgrade Synergi Software	5. Discretionary	\$71,545	5/31/2020
	Tilton Control panel replacement	1. Safety	\$124,956	12/31/2020
8840-1936	Locusview place holder	5. Discretionary	\$71,267	12/31/2020
8840-1945	Placeholder for Gas Training & Development	5. Discretionary	(\$534)	2019
8840-1953	Relocation of Engineering from Londonderry to Manchester	5. Discretionary	\$4,000	5/3/2019
8840-2002	Meter Protection Program	2. Mandated	\$797,741	12/31/2020
8840-2003	Cathodic Protection Program	2. Mandated	\$565,735	12/31/2020
8840-2004	Replacement Services Random (Non Leaks)	2. Mandated	\$629,257	12/31/2020
8840-2005	Replacement Services Random (Due to Leaks)	2. Mandated	\$606,382	12/31/2020
8840-2008	Corrosion & Miscellaneous Fitting	2. Mandated	\$308,724	12/31/2020
8840-2009	Valve Installation/Replacement	2. Mandated	\$21,910	12/31/2020
8840-2010	Leak Repairs	2. Mandated	\$2,139,714	7/7/2020 & 12/31/2020
8840-2011	Main Replacement LPP	Regulatory Programs	\$7,193,378	various
8840-2013	Main Replacement Fitting LPP	Discretionary	\$736,551	12/31/2020
8840-2014	K Meter Replacement Program	Discretionary	\$275,342	12/31/2020
8840-2015	Aldyl-A Replacement Program	Discretionary	\$80,424	carryover from billing related to city repaving
8840-2016	Main Replacement Reactive	Discretionary	\$545,410	various
8840-2018	Purchase Misc Capital Equipment & Tools	1. Safety	\$423,950	various
8840-2019	Regulator removal Hi line LOU	Discretionary	\$1,956	8/12/2020
8840-2020	SCADA Capital Improvements	Discretionary	\$129	2/3/2020
8840-2023	Main Replacement City/State Construction	2. Mandated	\$7,415,807	various
8840-2025	Service Replacement Fitting City/State Construction	2. Mandated	\$293,531	12/31/2020
8840-2026	LNG/LPG Capital Improvements	2. Mandated	\$105,941	12/11/2020
8840-2028	Gas System Control & Regulation (ENG)	5. Discretionary	\$400,008	various
8840-2029	Pre-Code Steel Pipe Protection Program/Replacement	2. Mandated	\$63,836	12/31/2020
8840-2030	IT - Software, Equipment & Infrastructure	5. Discretionary	\$63,413	5/1/2020
8840-2031	Gas System Planning & Reliability	5. Discretionary	\$1,409,927	various
8840-2038	IT Systems Allocations - Corporate	5. Discretionary	\$195,891	12/31/2020
8840-2039	Dresser Coupling Replacement Program	2. Mandated	\$466,494	12/31/2020
8840-2043	iRestore System Enhancements	5. Discretionary	\$347,138	12/31/2020
8840-2044	Flir Cameras - Security -Manchester	5. Discretionary	\$717,164	12/19/2020
8840-2062	GIS Mapping	5. Discretionary	\$273,898	12/31/2020
8840-2066	RTU Replacement Program	5. Discretionary	\$34,289	12/31/2020
8840-2084	Electric Meter Worker Meter Training/Testing Wall	1.Safety	\$24,926	7/31/2020
8840-2090	Transportation Fleet and Equipment Purchases	5. Discretionary	\$1,739,571	various
8840-2091	Meter Work Project (Meter Purchases)	2. Mandated	\$1,502,257	various
8840-2093	EN Facilities Capital Improvements	5. Discretionary	\$520,763	various
8840-2094	Install Security Equipment - EN Facilities	2. Mandated	\$37,561	various
8840-2096	Liberty @ Centre Vault Door	2. Mandated	\$7,740	9/3/2020
8843-1820	Keene Propane Air Plant Meter Install	5. Discretionary	\$12,233	in service 2018, \$\$ carryover
8843-2002	Replacement Services Random	2. Mandated	\$286	in service 2019, \$\$ carryover
8843-2009	Service Replacement City/State Construction	2. Mandated	\$313	in service 2019, \$\$ carryover
8843-2011	Main Replacement LPP	2. Mandated	\$368,119	various
8843-2012	Capital Tools/Equipment	5. Discretionary	\$2,426	12/31/2020
8843-2014	Gas System Planning & Reliability	5. Discretionary	\$1,353	in service 2019, \$\$ carryover
8843-2090	Transportation Fleet and Equipment Purchases	5. Discretionary	(\$3,435)	8/31/2020, credit for vehicle upfitting
8843-2093	Facility Improvements & Additions - Keene	5. Discretionary		
8843-2044	• •	•	\$64,185 \$128,202	11/30/2020
	Flir Cameras - Security-Keene	5. Discretionary	\$128,292	various
8843-2022	Propane Boiler Replacement	5. Discretionary	\$16,842	10/23/2020
*. • .	ate may be greater than 2020 spand because there was spanding i	Total	\$36,206,417	

^{*}In Service amounts may be greater than 2020 spend because there was spending in prior years for jobs put in service in 2020

Liberty Utilities (EnergyNorth) Step Increase - EnergyNorth For Non-Growth Projects Placed In-Service During 2020

Rate of Return Calculation

Equity Debt

37

38

Docket No. DG 20-105
Appendix 1
Page 2 of 2

ne	Description	Misc.	. Intangible Plant - M 3 yr	lisc. Intangible 5 yr	Plant - 1	LNG Plant	Mains	Station Equipment	Mains		ns. & Reg. ion Equip.	Services	Meters	Office Equipment	Vehicles	Tools	Communication Equipment	Total
	FERC Account		303	303		320	367	369	376		378	380	381	391	392	394	397	
Capital	l Spending	\$	273,898 \$		577,987	\$ 122,782 \$			466,494		127,041 \$	1,529,769 \$	2,921,224					\$ 36,206,417
2 B Deferre	ed Tax Calculation																	
Tax Mo	lethod		MACRS15	MACRS15		MACRS20	MACRS20	MACRS20	MACRS20	M	IACRS20	MACRS20	MACRS20	MACRS7	MACRS5	MACRS7	MACRS7	
Tax De	epreciation Rate		5.00%		5.00%	3.75%	3.75%	3.75%	3.75%	6	3.75%	3.75%	3.75%	14.29%	20.00%	14.29%	14.29%	
Bonus 1	Depreciation @ 0.00%	\$	- \$		- :	\$ - \$	-	\$ - \$	-	\$	- \$	- \$	-	\$ - \$	- \$	-	\$ -	\$ -
Tax Ba	asis	\$	273,898 \$	6	677,987	\$ 122,782 \$	25,939,975	\$ 400,008 \$	466,494	. \$	127,041 \$	1,529,769 \$	2,921,224	\$ 1,479,706 \$	1,736,136 \$	425,842	\$ 105,556	\$ 36,206,41
) MACR	RS Depreciation	\$	13,695 \$		33,899	\$ 4,604 \$	972,749	\$ 15,000 \$	17,494	. \$	4,764 \$	57,366 \$	109,546	\$ 211,387 \$	347,227 \$	60,835	\$ 15,079	\$ 1,863,645
Tax De	epreciation - Federal	\$	13,695 \$		33,899	\$ 4,604 \$	972,749	\$ 15,000 \$	17,494	\$	4,764 \$	57,366 \$	109,546	\$ 211,387 \$	347,227 \$	60,835	\$ 15,079	\$ 1,863,64
	epreciation - State	\$	13,695 \$		33,899	\$ 4,604 \$	972,749	\$ 15,000 \$	17,494	\$	4,764 \$	57,366 \$	109,546	\$ 211,387 \$	347,227 \$	60,835	\$ 15,079	
Book D	Depreciation Rate		33.33%		20.00%	2.86%	1.92%	2.86%	1.92%	6	2.86%	3.55%	3.13%	5.28%	20.00%	5.26%	10.00%	
Book D	Depreciation	\$	91,290 \$	1	135,597	\$ 3,512 \$	498,048	\$ 11,440 \$	8,957	\$	3,633 \$	54,307 \$	91,434	\$ 78,128 \$	347,227 \$	22,399	\$ 10,556	\$ 1,356,52
Tax ov	ver (under) Book - Federal	\$	(77,595) \$	(1	101,698)	\$ 1,093 \$	474,702	\$ 3,560 \$	8,537	\$	1,131 \$	3,060 \$	18,112	\$ 133,258 \$	- \$	38,435	\$ 4,524	\$ 507,11
Tax ov	ver (under) Book - State		(77,595)	(1	101,698)	1,093	474,702	3,560	8,537		1,131	3,060	18,112	133,258	-	38,435	4,524	507,11
) Deferre	ed Taxes - Federal @ 19.38%		(15,040)	((19,712)	212	92,011	690	1,655		219	593	3,511	25,829	-	7,450	877	98,29
1 Deferre	ed Taxes - State @ 7.70%		(5,975)		(7,831)	84	36,552	274	657		87	236	1,395	10,261	-	2,960	348	 39,04
2 <u>Deferre</u>	ed Tax Balance @ 0.00%	\$	(21,015) \$	((27,543)	\$ 296 \$	128,563	\$ 964 \$	2,312	\$	306 \$	829 \$	4,905	\$ 36,090 \$	- \$	10,409	\$ 1,225	\$ 136,11
-	Base Calculation																	
	n Service	\$	273,898 \$		577,987	\$ 122,782 \$	25,939,975				127,041 \$	1,529,769 \$	2,921,224		1,736,136 \$			\$ 36,206,41
	nulated Depreciation		(91,290)		135,597)	(3,512)	(498,048)	(11,440)	(8,957		(3,633)	(54,307)	(91,434)		(347,227)	(22,399)		(1,356,52
7 <u>Deferre</u>	ed Tax Balance		21,015		27,543	(296)	(128,563)	(964)	(2,312		(306)	(829)	(4,905)		0	(10,409)		 (137,34
·	Rate Base	\$	203,623 \$	5	569,932	\$ 118,975 \$	25,313,364	\$ 387,604 \$	455,225	\$	123,101 \$	1,474,634 \$	2,824,885	\$1,365,487 \$	1,388,908 \$	393,033	\$ 93,776	\$ 34,712,54
	ue Requirement Calculation																	
	on Rate Base @ 8.76%	\$	17,837 \$		49,926		2,217,451				10,784 \$	129,178 \$		\$ 119,617 \$				\$ 3,040,81
-	ciation Expense		91,290	1	135,597	3,512	498,048	11,440	8,957		3,633	54,307	91,434		347,227	22,399	10,556	1,356,52
3 Propert	ty Tax @ \$6.60 per \$1000					810	171,204	2,640	3,079		838	10,096	19,280					 207,94
4 Annual	al Revenue Requirement	\$	109,128 \$	1	185,523	\$ 14,744 \$	2,886,702	\$ 48,034 \$	51,913	\$	15,255 \$	193,581 \$	358,174	\$ 197,745 \$	468,896 \$	56,829	\$ 18,770	\$ 4,605,29

Pre-Tax WACC

6.64%

2.12%

8.76%

Tax

27.08%

After-Tax Cost

9.30%

4.420%

Portion

52.0%

48.0%

100.0%

Capped at \$4,000,000

Docket No. DG 21-130 Exhibit 26 Page 30 of 470 Docket No. DG 20-105 Exhibit 49

Docket No. DG 20-105 Appendix 2 Page 1 of 2

Liberty Utilities Capital Spending Plan

Spending 2021, Step Filing 2022

	CY 2021 Updated Budget
<u>Project</u>	Recovery 2022
Leak Repairs	\$1,750,000
LPP-City/State	\$23,050,010
Aldyl-A Replacement	\$200,000
K Meter Replacement Program	\$350,000
Main Replacement Reactive	\$600,000
Dresser Coupling Replacement	\$500,000
Gas System Planning & Reliability	\$2,900,000
Gas Supply System Enhancements	\$0
Customer First	\$0
Total	\$29,350,010

Liberty Utilities (EnergyNorth) Step Increase - EnergyNorth For Non-Growth Projects Placed In-Service During 2021

Docket No. DG 20-105 Appendix 2 Page 2 of 2

Line	Description		Mains		Mains	Meters		Total		
	FERC Account		367		376		381			
1	Capital Spending	\$	28,500,010	\$	500,000	\$	350,000	\$	29,350,010	
2										
3	Deferred Tax Calculation									
4	Tax Method		MACRS20		MACRS20		MACRS20			
5	Tax Depreciation Rate		3.75%		3.75%		3.75%			
6										
7	Bonus Depreciation @ 0.00%	\$	-	\$	-	\$	-	\$	-	
8										
9	Tax Basis	\$	28,500,010	\$	500,000	\$	350,000	\$	29,350,010	
10	MACRS Depreciation	\$	1,068,750	\$	18,750	\$	13,125	\$	1,100,625	
11	Tay Dame sisting Endand	¢	1 069 750	Φ	10.750	\$	12 125	¢	1,100,625	
12 13	Tax Depreciation - Federal Tax Depreciation - State	\$ \$	1,068,750 1,068,750	\$ \$	18,750 18,750	\$ \$	13,125 13,125	\$	1,100,023	
13	Tax Depreciation - State	Ф	1,008,730	Ф	16,730	Ф	13,123			
15	Book Depreciation Rate		1.92%		1.92%		3.13%			
16	Book Depreciation Nate Book Depreciation	\$	547,200	\$	9,600	\$	10,955	\$	567,755	
17	Book Depreciation	Ψ	347,200	Ψ	7,000	Ψ	10,733	Ψ	301,133	
18	Tax over (under) Book - Federal	\$	521,550	\$	9,150	\$	2,170	\$	532,870	
19	Tax over (under) Book - State	·	521,550	·	9,150	·	2,170	·	532,870	
20	Deferred Taxes - Federal @ 19.38%		101,092		1,774		421		103,286	
21	Deferred Taxes - State @ 7.70%		40,159		705		167		41,031	
22	Deferred Tax Balance @ 0.00%	\$	141,251	\$	2,478	\$	588	\$	144,317	
23										
24	Rate Base Calculation									
25	Plant in Service	\$	28,500,010	\$	500,000	\$	350,000	\$	29,350,010	
26	Accumulated Depreciation		(547,200)		(9,600)		(10,955)		(567,755)	
27	Deferred Tax Balance		(141,251)		(2,478)		(588)		(144,317)	
28	Rate Base	\$	27,811,558	\$	487,922	\$	338,457	\$	28,637,938	
29										
30	Revenue Requirement Calculation									
31	Return on Rate Base @ 8.76%	\$	2,436,293	\$	42,742	\$	29,649	\$	2,508,683	
32	Depreciation Expense		547,200		9,600		10,955	\$	567,755	
33	Property Tax @ \$6.60 per \$1000		188,100		3,300		2,310		193,710	
34	Annual Revenue Requirement	\$	3,171,593	\$	55,642	\$	42,914	\$	3,270,149	
35										

Capped at \$3,200,000

37				
38	Rate of Return Calculation	Portion	After-Tax Cost	Pre-Tax WACC
39	Equity	52.0%	9.30%	6.64%
40	Debt	48.0%	4.420%	2.12%
41		100.0%		8.76%

,	Гах
	27.08%

EnergyNorth Municipal Property Taxes Included In Current Rates As Of March 31, 2021

Docket No. DG 20-105 Appendix 3 Page 1 of 1

<u>Line</u>	<u>Amount</u>	<u>Reference</u>	<u>Source</u>
1	\$9,259,401 EnergyNorth Total Prope	rty Tax	Included in Page 6 of 16 Line 24 in Appendix 1 from Order No. 26,122 in Docket No. DG 17-048
2	\$153,854 Keene Total Property Tax	<	Included in Page 6 of 16 Line 24 in Appendix 1 from Order No. 26,122 in Docket No. DG 17-048
3	(\$1,952,162) EnergyNorth State Prope	erty Taxes	Included in Page 6 of 16 Line 24 in Appendix 1 from Order No. 26,122 in Docket No. DG 17-048
4	(\$15,492) Keene State Property Tax	kes	Included in Page 6 of 16 Line 24 in Appendix 1 from Order No. 26,122 in Docket No. DG 17-048
5	\$7,445,601 2016 Municipal Property	Taxes	Sum of Lines 1 through 4
6	\$187,530 2017 CIBS		Annual incremental property taxes multiplied by the 2016 Municipal Property Tax %
7	\$153,572 2018 CIBS		Annual incremental property taxes multiplied by the 2016 Municipal Property Tax %
8	\$332,299 2018 Capital Step Increas	se (2017 Investment)	Appendix 4 from Order No. 26,122 in Docket No. DG 17-048 multiplied by the 2016 Municpal Property Tax %
9	\$90,647 2019 CIBS		Annual incremental property taxes multiplied by the 2016 Municipal Property Tax %
10	\$90,249 2020 CIBS (Jul 2020 - Sep	2020)	Three months of annual incremental property taxes multiplied by the 2016 Municipal Property Tax $\%$
11 _	\$624,999 DG 20-105 Reconciliation	n (Oct 2020 - Mar 2021)	Temp to Perm Reconciliation to Recoupment (Oct 2020 - Mar 2021)
12	\$8,924,897 Calculated Municipal Pro	perty Taxes Included In Rates Through March 31, 2021	
13			
14	\$10,000,000 Hypothetical Municipal F	Property Taxes Billed April 2020 - March 2021	
15	\$0 Hypothetical Abatement	s or Other Adjustments	
16	\$10,000,000 Hypothetical Adjusted M	lunicipal Property Taxes Billed April 2020 - March 2021	Line 14 plus Line 15
17			
18	\$1,075,103 Increase To Base Rates D	ue To Municipal Property Tax Reconciliation	Line 16 minus Line 12

Docket No. DG 21-130 Exhibit 26 Page 33 of 470 Docket No. DG 20-105 Exhibit 49

Docket No. DG 20-105 Appendix 4 Page 1 of 1

Incremental CNG Supply Costs through October 2021 Amount to be Recovered/(Refunded) Through Keene COG Rates

CNG Increment Cost/Saving Risk Sharing - 50% Shareholder/Ratepayers										
Incremental CNG Supply Costs - October 2019 thru October 2021										
COG Period	Year	Amount	Deferred	(Refund)/Charge						
Summer	2019	5,048		(2,524)						
Winter	2019-20	132,469	132,533	66,299						
Summer	2020	16,214	16,214	8,107						
Winter - Note 1	2020-21	136,525		(68,263)						
Summer - Note 2	2021	4,012		(2,006)						
Total Summer		25,274	16,214	3,577						
Total Winter		268,994	132,533	(1,964)						
Combined Total		294,268	148,747	1,613						

Note 1 - Estimated CNG incremental cost based on projected usage and costs filed in DG 20-152 (Keene 2020-2021 Winter COG). Actual CNG incremental costs/savings to be determined in Keene 2021-2022 Winter COG.

Note 2 - Estimated CNG incremental cost based on projected usage and costs filed in DG 21-050 (Keene 2021 Summer COG). Actual CNG incremental costs/savings to be determined in Keene 2022 Summer COG.

Docket No. DG 20-105 Appendix 5 Page 1 of 1

ILLUSTRATION - CALCULATION OF PROJECTED INCREMENTAL CNG COST/SAVING TO BE INCLUDED IN COG RATES TO BE RECONCILED USING ACTUAL COST - SUBSEQUENT COG RATES ADJUSTED FOR OVER/UNDER RECOVERY

See Line No. 22, 28, 42 and 45

LIBERTY UTILITIES - KEENE DIVISION

CALCULATION OF PURCHASED GAS COSTS SUMMER PERIOD 20XX

LINE		SUMMER PER	RIOD 20XX					
NO.		May-XX	Jun-XX	Jul-XX	Aug-XX	Sep-XX	Oct-XX	TOTAL
1	TOTAL SENDOUT (therms)	60,000	40,000	50,000	40,000	50,000	80,000	320,000
2	CHANGE TO ENDING INVENTORY BALANCE (therms)	-	-	-	-	-	-	-
3	TOTAL REQUIRED PURCHASES (therms)	60,000	40,000	50,000	40,000	50,000	80,000	320,000
4	PROPANE PURCHASE STABILIZATION PLAN DELIVERIES							_
5	Therms	-	-	-	-	-	-	-
6	RATES - from Schedule D							
7	Contract Price Broker Fee	\$0.0000	\$0.0000	\$0.0000	\$0.0000	\$0.0000	\$0.0000	
8 9	Pipeline Fee	incl. incl.			incl. incl.		incl. incl.	
10	PERC Fee	incl.		incl.	incl.		incl.	
11	Trucking Fee	incl.	incl.	incl.	incl.	incl.	incl.	
12	COST PER GALLON	\$0.0000	\$0.0000	\$0.0000	\$0.0000	\$0.0000	\$0.0000	
13	TOTAL COST - Propane Purchase Stabilization Plan Deliveries	\$0	\$0	\$0	\$0	\$0	\$0	\$0
14	AMHERST STORAGE PROPANE DELIVERIES							
15 16	Therms RATES - from Schedule F	-	-	-	-	-	-	-
17	WACOG Price	\$0.0000	\$0.0000	\$0.0000	\$0.0000	\$0.0000	\$0.0000	
18	Trucking Fee	\$0.0000	\$0.0000	\$0.0000	\$0.0000	\$0.0000	\$0.0000	
19	COST PER GALLON	\$0.0000	\$0.0000	\$0.0000	\$0.0000	\$0.0000	\$0.0000	
20	TOTAL COST - Amherst Storage Propane Deliveries	\$0	\$0	\$0	\$0	\$0	\$0	\$0
21	CNG DELIVERIES							
22	Therms	9,000	10,000	10,000	10,000	10,000	13,500	62,500
23 24	RATE PRICE	\$ 0.8000	\$ 0.8500	\$ 0.9000	\$ 0.9000	\$ 0.9000	\$ 0.9000	
25 25	COST PER Therm	\$ 0.8000	<u> </u>	•	·	· · · · · · · · · · · · · · · · · · ·	·	
26	COST - CNG	\$7,200	\$8,500	\$9,000	\$9,000	\$9,000	\$12,150	
27	DEMAND FIXED	\$4,000	\$4,000	\$4,000	\$4,000	\$4,000	\$4,000	\$24,000
28	CNG COST PER Therm (line 29/22)	\$ 1.2444	\$ 1.2500	\$ 1.3000	\$ 1.3000	\$ 1.3000	\$ 1.1963	
29	TOTAL CNG (line 26+27)	\$11,200	\$12,500	\$13,000	\$13,000	\$13,000	\$16,150	\$78,850
30	SPOT PURCHASES	50.000		40.000		40.000	05.000	055.000
31	Therms	50,000	30,000	40,000	30,000	40,000	65,000	255,000
32	RATES - from Schedule C	\$0.0500	\$0,000	#0.7500	\$0.7500	\$0,000	¢0.0500	
33 34	Mont Belvieu Broker Fee	\$0.8500 \$0.0100	\$0.8000 \$0.0100	\$0.7500 \$0.0100	\$0.7500 \$0.0100	\$0.8000 \$0.0100	\$0.8500 \$0.0100	
35	Pipeline Fee	\$0.0000	\$0.0000	\$0.0000	\$0.0000	\$0.0000	\$0.0000	
36	PERC Fee	\$0.0050	\$0.0050	\$0.0050	\$0.0050	\$0.0050	\$0.0050	
37	Supplier Charge	\$0.2000	\$0.2000	\$0.2000	\$0.2000	\$0.2000	\$0.2000	
38 39	Trucking Fee COST PER GALLON - Market Quotes	\$0.1000 \$1.1650	\$0.1000 \$1.1150	\$0.1000 \$1.0650	\$0.1000 \$1.0650	\$0.1000 \$1.1150	\$0.1000 \$1.1650	
40	COST PER GALLON - Market Quotes COST PER THERM - Market Quotes	\$1.1650	\$1.1150	\$1.0650	\$1.0650	\$1.1150	\$1.1650	
41	TOTAL COST - Spot Purchases	\$58,250	\$33,450	\$42,600	\$31,950	\$44,600	¢75 705	\$286,575
42	SPOT PURCHASES -COST PER Therm (line 41/31)	\$1.1650	\$1.2150	\$1.1650	\$1,950	\$1.2150	\$1.2650	φ200,373
43	OTHER ITEMS							
44	Storage - Winter Period 20XX-20XX	\$0	\$0	\$0	\$0	\$0	\$0	\$0
45 46	50% Increment CNG (line (28 - 42) * 22*50%) TOTAL OTHER ITEMS	(\$358)			(\$675) (\$675)		\$464 \$464	(\$1,844)
40		(\$358)	(\$175)	(\$075)	(\$675)	(\$425)	\$464	(\$1,844)
47	<u>TOTAL</u>							
48 40	THERMS (line 5+15+22+31) SENDOUT THERMS (line 1)	59,000 60,000	40,000	50,000	40,000	50,000	78,500	317,500
49 50	COST (line 13+20+29+41+46)	60,000 \$69,093	40,000 \$45,775	50,000 \$54,925	40,000 \$44,275	50,000 \$57,175	80,000 \$92,339	320,000 \$363,581
51	COST PER THERM (line 50/48)	\$1.1711	\$1.1444	\$1.0985	\$1.1069	\$1.1435	\$1.1763	\$1.1451
٠.		÷ · · · · · · ·	+	÷	÷	÷ 100	÷ 00	,

^{**} Shaded cells show what information is considered confidential in future Keene COG filings

Docket No. DG 20-105 Appendix 6 Page 1 of 1

Keene Risk Sharing Adjustment **Update DCF Analysis for Keene Phase 1 Conversion**

\$359,889 Capital Cost Direct (12/31/19 Rate Base) Required Return (pre tax) 8.75% Depreciation 8,997 Property tax rate 1.94% Insurance rate 0.10%

NPV (Delta yrs 1-10, discount rate 10.15%)

(\$270,681.69)

Risk Sharing Calculation	
Permanent Rates Take Effect Year 2	
Average revenue (years 2-4)	\$0
Average revenue requirement (years 2-4)	\$44,297
Difference	(\$44,297)
Revenue Requirement Reduction (50%)	(\$22,149)
Adjustment to Distribution (91.51%)	(\$20,268)
Adjustment to COG (8.49%)	(\$1,880)

	IRS	IRS		Delta			Accumulated								Delta
	MACRS	MACRS	Book	Book	Tax	Deferred	Deferred	Rate	Required	Property			Revenue		Rev Req
Year	Rates	Table	Depr	less Tax	Rate	Inc Tax	Inc Tax	Base	Return	Tax	Insurance	O&M	Requirement	Annual Revenues	less Revenue
			(40 yrs/2.5%	6)						1.94%	0.10%	\$ 35			
								359,889							
	1 5.00%	17,994	8,997	(8,997)	27%	(2,436)	(2,436)	348,456	\$31,003	\$6,971	\$360	\$0	\$47,331	\$0.00	(\$47,330.75)
	2 9.50%	34,189	8,997	(25,192)	27%	(6,822)	(9,259)	332,636	\$29,810	\$6,750	\$360	\$0	\$45,916	\$0.00	(\$45,916.48)
	3 8.55%	30,771	8,997	(21,773)	27%	(5,896)	(15,155)	317,743	\$28,466	\$6,443	\$360	\$0	\$44,266	\$0.00	(\$44,265.84)
	4 7.70%	27,711	8,997	(18,714)	27%	(5,068)	(20,223)	303,678	\$27,198	\$6,155	\$360	\$0	\$42,710	\$0.00	(\$42,709.91)
	5 6.93%	24,940	8,997	(15,943)	27%	(4,317)	(24,540)	290,363	\$26,000	\$5,882	\$360	\$0	\$41,239	\$0.00	(\$41,239.13)
	6 6.23%	22,421	8,997	(13,424)	27%	(3,635)	(28,175)	277,731	\$24,864	\$5,624	\$360	\$0	\$39,846	\$0.00	(\$39,845.58)
	7 5.90%	21,233	8,997	(12,236)	27%	(3,314)	(31,489)	265,420	\$23,772	\$5,380	\$360	\$0	\$38,509	\$0.00	(\$38,509.19)
	8 5.90%	21,233	8,997	(12,236)	27%	(3,314)	(34,802)	253,109	\$22,695	\$5,141	\$360	\$0	\$37,193	\$0.00	(\$37,193.10)
	9 5.91%	21,269	8,997	(12,272)	27%	(3,323)	(38,126)	240,789	\$21,617	\$4,903	\$360	\$0	\$35,877	\$0.00	(\$35,876.58)
1	0 5.90%	21,233	8,997	(12,236)	27%	(3,314)	(41,439)	228,478	\$20,539	\$4,664	\$360	\$0	\$34,560	\$0.00	(\$34,559.87)
1	1 5.91%	21,269	8,997	(12,272)	27%	(3,323)	(44,762)	216,157	\$19,461	\$4,426	\$360	\$0	\$33,243	\$0.00	(\$33,243.36)
1	2 5.90%	21,233	8,997	(12,236)	27%	(3,314)	(48,076)	203,846	\$18,383	\$4,187	\$360	\$0	\$31,927	\$0.00	(\$31,926.65)
1	3 5.91%	21,269	8,997	(12,272)	27%	(3,323)	(51,399)	191,526	\$17,305	\$3,949	\$360	\$0	\$30,610	\$0.00	(\$30,610.14)
1	4 5.90%	21,233	8,997	(12,236)	27%	(3,314)	(54,713)	179,215	\$16,226	\$3,710	\$360	\$0	\$29,293	\$0.00	(\$29,293.43)
1	5 5.91%	21,269	8,997	(12,272)	27%	(3,323)	(58,036)	166,895	\$15,148	\$3,471	\$360	\$0	\$27,977	\$0.00	(\$27,976.92)
1	6 2.95%	10,617	8,997	(1,620)	27%	(439)	(58,475)	157,459	\$14,196	\$3,233	\$360	\$0	\$26,786	\$0.00	(\$26,786.04)
1	7		8,997	8,997	27%	2,436	(56,038)	150,898	\$13,496	\$3,050	\$360	\$0	\$25,903	\$0.00	(\$25,903.14)
1	8		8,997	8,997	27%	2,436	(53,602)	144,337	\$12,922	\$2,923	\$360	\$0	\$25,202	\$0.00	(\$25,201.76)
	9		8,997	8,997	27%	2,436	(51,165)	137,776	\$12,347	\$2,796	\$360	\$0	\$24,500	\$0.00	(\$24,500.38)
	20		8,997	8,997	27%	2,436	(48,729)	131,216	\$11,773	\$2,669	\$360	\$0	\$23,799	\$0.00	(\$23,798.99)
	Required R	eturn (Liberty pr	oposal)												
	Equity	52%	9.30%	12.75%	6.63%										
	Debt	48%	4.42%	4.42%	2.12% 8.75%										

Liberty Utilities (EnergyNorth Natural Gas) Corp. Docket DG 20-105 Revenue Decoupling Adjustment Rates Eff. 8/1/2021

Docket N0. DG 20-105 Appendix 7 Page 1 of 1

Permanent Rates Revenue Per Customer

Rate Class	s January		February		March		April		Мау		June		July		August		September		October	November		December	
R-1/5	\$	26.014	\$	25.540	\$	24.307	\$	22.609	\$	20.956	\$	19.755	\$	18.931	\$	19.019	\$	19.435	\$	20.546	\$	22.982	\$ 25.299
R-3/6	\$	97.157	\$	93.255	\$	74.713	\$	50.567	\$	34.034	\$	25.472	\$	22.948	\$	23.085	\$	25.352	\$	37.025	\$	62.207	\$ 83.921
R-4/7	\$	97.157	\$	93.255	\$	74.713	\$	50.567	\$	34.034	\$	25.472	\$	22.948	\$	23.085	\$	25.352	\$	37.025	\$	62.207	\$ 83.921
G-41/44	\$	235.956	\$	226.979	\$	184.606	\$	128.146	\$	88.800	\$	70.623	\$	66.093	\$	66.385	\$	70.916	\$	94.488	\$	154.776	\$ 204.268
G-42/45	\$	1,578.472	\$	1,524.667	\$	1,241.555	\$	855.091	\$	523.642	\$	346.741	\$	294.872	\$	301.796	\$	360.170	\$	572.697	\$	1,034.777	\$ 1,394.253
G-43/46	\$	8,928.306	\$	8,426.278	\$	7,012.866	\$	4,981.917	\$	1,969.310	\$	1,450.046	\$	1,304.759	\$	1,372.855	\$	1,462.191	\$	2,016.955	\$	5,871.987	\$ 7,656.083
G-51/55	\$	133.825	\$	130.979	\$	121.907	\$	111.427	\$	104.493	\$	98.646	\$	94.516	\$	98.006	\$	98.750	\$	101.809	\$	115.084	\$ 126.203
G-52/56	\$	731.471	\$	706.568	\$	650.770	\$	576.938	\$	402.135	\$	377.110	\$	367.473	\$	377.804	\$	384.365	\$	407.882	\$	611.436	\$ 669.830
G-53/57	\$	6,797.367	\$	6,197.111	\$	5,755.166	\$	4,877.206	\$	2,508.532	\$	2,307.268	\$	2,328.947	\$	2,476.034	\$	2,356.654	\$	2,625.619	\$	5,366.438	\$ 6,077.525
G-54/58	\$	3,719.928	\$	3,726.283	\$	3,387.343	\$	3,833.707	\$	2,775.284	\$	2,874.002	\$	2,966.625	\$	3,090.866	\$	2,982.545	\$	2,965.834	\$	4,662.611	\$ 3,822.712

Step Increase **Revenue Per Customer**

Rate Class		January	February		March		April	May		June		July		August	September	October	November	0	December
R-1/5	\$	1.483	\$ 1.402	\$	1.264	\$	0.987	\$ 0.742	\$	0.563	\$	0.464	\$	0.461	\$ 0.537	\$ 0.767	\$ 1.196	\$	1.535
R-3/6	\$	4.968	\$ 4.490	\$	3.576	\$	2.178	\$ 1.178	\$	0.590	\$	0.464	\$	0.462	\$ 0.630	\$ 1.405	\$ 3.017	\$	4.353
R-4/7	\$	4.968	\$ 4.490	\$	3.576	\$	2.178	\$ 1.178	\$	0.590	\$	0.464	\$	0.462	\$ 0.630	\$ 1.405	\$ 3.017	\$	4.353
G-41/44 G-42/45 G-43/46	\$ \$	10.371 71.556 322.176	9.551 65.275 307.458	\$	7.771 52.763 260.216	\$	5.376 33.854 188.058	\$ 3.848 20.781 125.272	\$	2.999 13.163 80.674	\$	2.860 11.663 74.052	\$	2.877 12.053 74.222	\$ 3.115 14.984 90.747	\$ 4.296 26.315 148.398	\$ 6.950 47.308 230.190	\$	9.342 64.023 306.060
	φ			•		•		_	•		•		•						
G-51/55	\$	6.156	6.082	•	5.241	-	4.545	4.182	•	4.099		4.023	•	4.113	_	4.825	5.232		5.946
G-52/56	\$	31.400	\$ 30.740	\$	24.341	\$	20.081	\$ 17.238	\$	17.150	\$	17.025	\$	17.535	\$ 18.199	\$ 21.044	\$ 23.978	\$	27.933
G-53/57	\$	246.248	\$ 243.066	\$	214.654	\$	186.181	\$ 150.341	\$	140.629	\$	138.297	\$	140.255	\$ 144.706	\$ 168.388	\$ 188.258	\$	211.553
G-54/58	\$	138.456	\$ 145.419	\$	124.103	\$	143.307	\$ 136.199	\$	145.470	\$	155.194	\$	160.877	\$ 160.145	\$ 160.192	\$ 161.125	\$	137.154

Total **Revenue Per Customer**

Rate Class	January	February	March	April	May	June	July	August	S	September	October	ı	November	December
R-1/5	\$ 27.498	\$ 26.942	\$ 25.571	\$ 23.596	\$ 21.698	\$ 20.318	\$ 19.395	\$ 19.480	\$	19.972	\$ 21.314	\$	24.178	\$ 26.834
R-3/6	\$ 102.124	\$ 97.745	\$ 78.289	\$ 52.745	\$ 35.212	\$ 26.062	\$ 23.412	\$ 23.547	\$	25.982	\$ 38.431	\$	65.224	\$ 88.274
R-4/7	\$ 102.124	\$ 97.745	\$ 78.289	\$ 52.745	\$ 35.212	\$ 26.062	\$ 23.412	\$ 23.547	\$	25.982	\$ 38.431	\$	65.224	\$ 88.274
G-41/44	\$ 246.326	\$ 236.530	\$ 192.376	\$ 133.522	\$ 92.648	\$ 73.622	\$ 68.954	\$ 69.262	\$	74.031	\$ 98.783	\$	161.726	\$ 213.610
G-42/45	\$ 1,650.029	\$ 1,589.942	\$ 1,294.318	\$ 888.944	\$ 544.422	\$ 359.904	\$ 306.536	\$ 313.849	\$	375.153	\$ 599.012	\$	1,082.085	\$ 1,458.276
G-43/46	\$ 9,250.482	\$ 8,733.736	\$ 7,273.082	\$ 5,169.975	\$ 2,094.582	\$ 1,530.720	\$ 1,378.810	\$ 1,447.077	\$	1,552.938	\$ 2,165.354	\$	6,102.177	\$ 7,962.143
G-51/55	\$ 139.981	\$ 137.061	\$ 127.148	\$ 115.972	\$ 108.676	\$ 102.744	\$ 98.539	\$ 102.119	\$	102.993	\$ 106.634	\$	120.316	\$ 132.149
G-52/56	\$ 762.870	\$ 737.308	\$ 675.111	\$ 597.019	\$ 419.373	\$ 394.261	\$ 384.498	\$ 395.340	\$	402.564	\$ 428.926	\$	635.414	\$ 697.763
G-53/57	\$ 7,043.615	\$ 6,440.177	\$ 5,969.820	\$ 5,063.387	\$ 2,658.873	\$ 2,447.898	\$ 2,467.245	\$ 2,616.288	\$	2,501.361	\$ 2,794.007	\$	5,554.697	\$ 6,289.078
G-54/58	\$ 3,858.384	\$ 3,871.702	\$ 3,511.446	\$ 3,977.013	\$ 2,911.483	\$ 3,019.472	\$ 3,121.818	\$ 3,251.743	\$	3,142.690	\$ 3,126.025	\$	4,823.736	\$ 3,959.866

Docket No. DG 20-105 Appendix 8 Page 1 of 10

Liberty Utilities (EnergyNorth Natural Gas) Corp.

COMPARATIVE MONTHLY BILLING UNDER PRESENT AND PROPOSED RATES RATE R-1 : RESIDENTIAL NON-HEATING

Present Rates	Winter	Summer
Cost of Gas	\$0.4737	\$0.3935
LDAC	\$0.0589	\$0.0589
Customer charge Sales rate	\$15.50	
First Block Size		
Block 1	\$0.3860	\$0.3860
Block 2	\$0.3860	\$0.3860

Proposed Rates	Winter	Summer
Cost of Gas	\$0.5010	\$0.3935
LDAC	\$0.0589	\$0.0589
Customer charge Sales rate	\$15.39	
First Block Size		
Block 1	\$0.4358	\$0.4358
Block 2	\$0.4358	\$0.4358

•				Ļ	DIOCK Z		ψ0.5000	ψ0.0000	I,	DIOCK Z		ψ0.4000	ψυ.+330								
10																					
11	<u>-</u>																				
12			Mont	hly Bills at P	resent Rat	es			Month	nly Bills at P	roposed Ra	ates		Cha	ange in N	∕lonthly E	3ill		Uni	t Costs	
13	Use per		Winter			Summer			Winter			Summer		Win	ter	Sum	mer	Cur	rent	Propo	osed
	Month	Base	COG /		Base	COG/		Base	COG/		Base	COG/									
14	(therms)	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	\$	%	\$	%	Winter	Summer	Winter	Summer
15																					
16	0	\$15.50	\$0.00	\$15.50	\$15.50	\$0.00	\$15.50	\$15.39	\$0.00	\$15.39	\$15.39	\$0.00	\$15.39	-\$0.11	-0.7%	-\$0.11	-0.7%	\$0.0000	\$0.0000	\$0.0000	\$0.0000
17	2	\$16.27	\$1.07	\$17.34	\$16.27	\$0.90	\$17.18	\$16.26	\$1.12	\$17.38	\$16.26	\$0.90	\$17.17	\$0.05	0.3%	-\$0.01	0.0%	\$8.6686	\$8.5884	\$8.6919	\$8.5844
18	4	\$17.04	\$2.13	\$19.17	\$17.04	\$1.81	\$18.85	\$17.14	\$2.24	\$19.38	\$17.14	\$1.81	\$18.95	\$0.20	1.0%	\$0.09	0.5%	\$4.7936	\$4.7134	\$4.8438	\$4.7363
19	6	\$17.82	\$3.20	\$21.01	\$17.82	\$2.71	\$20.53	\$18.01	\$3.36	\$21.37	\$18.01	\$2.71	\$20.72	\$0.36	1.7%	\$0.19	0.9%	\$3.5019	\$3.4217	\$3.5611	\$3.4536
20	8	\$18.59	\$4.26	\$22.85	\$18.59	\$3.62	\$22.21	\$18.88	\$4.48	\$23.36	\$18.88	\$3.62	\$22.50	\$0.51	2.2%	\$0.29	1.3%	\$2.8561	\$2.7759	\$2.9198	\$2.8123
21	10	\$19.36	\$5.33	\$24.69	\$19.36	\$4.52	\$23.88	\$19.75	\$5.60	\$25.35	\$19.75	\$4.52	\$24.27	\$0.66	2.7%	\$0.39	1.6%	\$2.4686	\$2.3884	\$2.5350	\$2.4275
22	15	\$21.29	\$7.99	\$29.28	\$21.29	\$6.79	\$28.08	\$21.93	\$8.40	\$30.33	\$21.93	\$6.79	\$28.72	\$1.05	3.6%	\$0.64	2.3%	\$1.9519	\$1.8717	\$2.0219	\$1.9144
23	20	\$23.22	\$10.65	\$33.87	\$23.22	\$9.05	\$32.27	\$24.11	\$11.20	\$35.31	\$24.11	\$9.05	\$33.16	\$1.44	4.2%	\$0.89	2.8%	\$1.6936	\$1.6134	\$1.7654	\$1.6578
24	25	\$25.15	\$13.32	\$38.47	\$25.15	\$11.31	\$36.46	\$26.29	\$14.00	\$40.29	\$26.29	\$11.31	\$37.60	\$1.82	4.7%	\$1.14	3.1%	\$1.5386	\$1.4584	\$1.6114	\$1.5039
25	30	\$27.08	\$15.98	\$43.06	\$27.08	\$13.57	\$40.65	\$28.47	\$16.80	\$45.26	\$28.47	\$13.57	\$42.04	\$2.21	5.1%	\$1.39	3.4%	\$1.4353	\$1.3551	\$1.5088	\$1.4013
26	35	\$29.01	\$18.64	\$47.65	\$29.01	\$15.83	\$44.84	\$30.65	\$19.60	\$50.24	\$30.65	\$15.83	\$46.48	\$2.59	5.4%	\$1.64	3.6%	\$1.3615	\$1.2813	\$1.4355	\$1.3280
27	40	\$30.94	\$21.30	\$52.24	\$30.94	\$18.10	\$49.04	\$32.83	\$22.40	\$55.22	\$32.83	\$18.10	\$50.92	\$2.98	5.7%	\$1.89	3.8%	\$1.3061	\$1.2259	\$1.3805	\$1.2730
28	45	\$32.87	\$23.97	\$56.84	\$32.87	\$20.36	\$53.23	\$35.00	\$25.20	\$60.20	\$35.00	\$20.36	\$55.36	\$3.36	5.9%	\$2.13	4.0%	\$1.2630	\$1.1828	\$1.3378	\$1.2303
29	50	\$34.80	\$26.63	\$61.43	\$34.80	\$22.62	\$57.42	\$37.18	\$28.00	\$65.18	\$37.18	\$22.62	\$59.80	\$3.75	6.1%	\$2.38	4.2%	\$1.2286	\$1.1484	\$1.3036	\$1.1961
30	60	\$38.66	\$31.96	\$70.62	\$38.66	\$27.14	\$65.80	\$41.54	\$33.59	\$75.14	\$41.54	\$27.14	\$68.69	\$4.52	6.4%	\$2.88	4.4%	\$1.1769	\$1.0967	\$1.2523	\$1.1448
31	70	\$42.52	\$37.28	\$79.80	\$42.52	\$31.67	\$74.19	\$45.90	\$39.19	\$85.09	\$45.90	\$31.67	\$77.57	\$5.29	6.6%	\$3.38	4.6%	\$1.1400	\$1.0598	\$1.2156	\$1.1081
32	80	\$46.38	\$42.61	\$88.99	\$46.38	\$36.19	\$82.57	\$50.26	\$44.79	\$95.05	\$50.26	\$36.19	\$86.45	\$6.06	6.8%	\$3.88	4.7%	\$1.1124	\$1.0322	\$1.1881	\$1.0806
33	90	\$50.24	\$47.93	\$98.17	\$50.24	\$40.72	\$90.96	\$54.62	\$50.39	\$105.01	\$54.62	\$40.72	\$95.33	\$6.83	7.0%	\$4.38	4.8%	\$1.0908	\$1.0106	\$1.1668	\$1.0592
34	100	\$54.10	\$53.26	\$107.36	\$54.10	\$45.24	\$99.34	\$58.97	\$55.99	\$114.97	\$58.97	\$45.24	\$104.21	\$7.61	7.1%	\$4.87	4.9%	\$1.0736	\$0.9934	\$1.1497	\$1.0421
35	200	\$92.70	\$106.52	\$199.22	\$92.70	\$90.48	\$183.18	\$102.56	\$111.98	\$214.54	\$102.56	\$90.48	\$193.04	\$15.32	7.7%	\$9.86	5.4%	\$0.9961	\$0.9159	\$1.0727	\$0.9652
36																					
37 Estimated Bill Pe	rcentiles																				
38 Winter - 25%	8	\$18.59	\$4.26	\$22.85				\$18.88	\$4.48	\$23.36				\$0.51	2.2%			\$2.8561		\$2.9198	
39 Winter - 50%	20	\$23.22	\$10.65	\$33.87				\$24.11	\$11.20	\$35.31				\$1.44	4.2%			\$1.6936		\$1.7654	
40 Winter - 75%	30	\$27.08	\$15.98	\$43.06				\$28.47	\$16.80	\$45.26				\$2.21	5.1%			\$1.4353		\$1.5088	
41 Summer - 25%	5				\$17.43	\$2.26	\$19.69	•			\$17.57	\$2.26	\$19.83	·		\$0.14	0.7%		\$3.9384		\$3.9667
42 Summer - 50%	11				\$19.75	\$4.98	\$24.72				\$20.19	\$4.98	\$25.16			\$0.44	1.8%		\$2.2475		\$2.2875
43 Summer - 75%	20				\$23.22	\$9.05	\$32.27				\$24.11	\$9.05	\$33.16			\$0.89	2.8%		\$1.6134		\$1.6578

Estimated Bill Percentiles per 2010 MCS

5

Docket No. DG 20-105 Appendix 8 Page 2 of 10

Liberty Utilities (EnergyNorth Natural Gas) Corp.

COMPARATIVE MONTHLY BILLING UNDER PRESENT AND PROPOSED RATES RATE R-3 : RESIDENTIAL HEATING

Present Rates	Winter	Summer
Cost of Gas	\$0.4737	\$0.3935
LDAC	\$0.0589	\$0.0589
Customer charge Sales rate	\$15.50	
First Block Size		
Block 1	\$0.5678	\$0.5678
Block 2	\$0.5678	\$0.5678

Proposed Rates	Winter	Summer
Cost of Gas	\$0.5010	\$0.3935
LDAC	\$0.0589	\$0.0589
Customer charge Sales rate	\$15.39	
First Block Size		
Block 1	\$0.5985	\$0.5985
Block 2	\$0.5985	\$0.5985

12				Mor	nthly Bills at	Present Ra	ates			Mont	hly Bills at P	roposed R	ates		Ch	ange in N	/lonthly Bi	ill		Unit (Costs	
13		Use per		Winter			Summer			Winter			Summer		Win	ter	Sumr	mer	Cur	rent	Propo	osed
		Month	Base	COG /		Base	COG /		Base	COG /		Base	COG /									
14		(therms)	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	\$	%	\$	%	Winter	Summer	Winter	Summer
15																						
16		0	\$15.50	\$0.00	\$15.50	\$15.50	\$0.00	\$15.50	\$15.39	\$0.00	\$15.39	\$15.39	\$0.00	\$15.39	-\$0.11	-0.7%	-\$0.11	-0.7%	\$0.0000	\$0.0000	\$0.0000	\$0.0000
17		10	\$21.18	\$5.33	\$26.50	\$21.18	\$4.52	\$25.70	\$21.38	\$5.60	\$26.98	\$21.38	\$4.52	\$25.90	\$0.47	1.8%	\$0.20	0.8%	\$2.6504	\$2.5702	\$2.6976	\$2.5901
18		25	\$29.70	\$13.32	\$43.01	\$29.70	\$11.31	\$41.01	\$30.35	\$14.00	\$44.35	\$30.35	\$11.31	\$41.66	\$1.34	3.1%	\$0.66	1.6%	\$1.7204	\$1.6402	\$1.7741	\$1.6666
19		50	\$43.89	\$26.63	\$70.52	\$43.89	\$22.62	\$66.51	\$45.32	\$28.00	\$73.31	\$45.32	\$22.62	\$67.94	\$2.79	4.0%	\$1.43	2.1%	\$1.4104		\$1.4662	\$1.3587
20		75	\$58.09	\$39.95	\$98.03	\$58.09	\$33.93	\$92.02	\$60.28	\$41.99	\$102.27	\$60.28	\$33.93	\$94.21	\$4.24	4.3%	\$2.19	2.4%	\$1.3071	\$1.2269	\$1.3636	\$1.2561
21		100	\$72.28	\$53.26	\$125.54	\$72.28	\$45.24	\$117.52	\$75.24	\$55.99	\$131.23	\$75.24	\$45.24	\$120.48	\$5.69	4.5%	\$2.96	2.5%	\$1.2554	\$1.1752	\$1.3123	\$1.2048
22		125	\$86.48	\$66.58	\$153.05	\$86.48	\$56.55	\$143.03	\$90.20	\$69.99	\$160.19	\$90.20	\$56.55	\$146.75	\$7.14	4.7%	\$3.72	2.6%	\$1.2244		\$1.2815	\$1.1740
23		150	\$100.67	\$79.89	\$180.56	\$100.67	\$67.86	\$168.53	\$105.16	\$83.99	\$189.15	\$105.16	\$67.86	\$173.02	\$8.59	4.8%	\$4.49	2.7%	\$1.2037	\$1.1235	\$1.2610	\$1.1535
24		175	\$114.87	\$93.21	\$208.07	\$114.87	\$79.17	\$194.04	\$120.12	\$97.98	\$218.11	\$120.12	\$79.17	\$199.29	\$10.04	4.8%	\$5.26	2.7%	\$1.1890		\$1.2463	\$1.1388
25		200	\$129.06	\$106.52	\$235.58	\$129.06	\$90.48	\$219.54	\$135.08	\$111.98	\$247.07	\$135.08	\$90.48	\$225.56	\$11.49	4.9%	\$6.02	2.7%	\$1.1779		\$1.2353	\$1.1278
26		225	\$143.26	\$119.84	\$263.09	\$143.26	\$101.79	\$245.05	\$150.05	\$125.98	\$276.03	\$150.05	\$101.79	\$251.84	\$12.94	4.9%	\$6.79	2.8%	\$1.1693	\$1.0891	\$1.2268	\$1.1193
27		250	\$157.45	\$133.15	\$290.60	\$157.45	\$113.10	\$270.55	\$165.01	\$139.98	\$304.98	\$165.01	\$113.10	\$278.11	\$14.38	5.0%	\$7.56	2.8%	\$1.1624	\$1.0822	\$1.2199	\$1.1124
28		275	\$171.65	\$146.47	\$318.11	\$171.65	\$124.41	\$296.06	\$179.97	\$153.98	\$333.94	\$179.97	\$124.41	\$304.38		5.0%	\$8.32	2.8%	\$1.1568		\$1.2143	\$1.1068
29		300	\$185.84	\$159.78	\$345.62	\$185.84	\$135.72	\$321.56	\$194.93	\$167.97	\$362.90	\$194.93	\$135.72	\$330.65		5.0%	\$9.09	2.8%	\$1.1521	\$1.0719	\$1.2097	\$1.1022
30		350	\$214.23	\$186.41	\$400.64	\$214.23	\$158.34	\$372.57	\$224.85	\$195.97	\$420.82	\$224.85	\$158.34	\$383.19	\$20.18	5.0%	\$10.62	2.9%		\$1.0645	\$1.2023	\$1.0948
31		400	\$242.62	\$213.04	\$455.66	\$242.62	\$180.96	\$423.58	\$254.78	\$223.96	\$478.74	\$254.78	\$180.96	\$435.74	\$23.08	5.1%	\$12.16	2.9%	\$1.1392		\$1.1969	\$1.0893
32		450	\$271.01	\$239.67	\$510.68	\$271.01	\$203.58	\$474.59	\$284.70	\$251.96	\$536.66	\$284.70	\$203.58	\$488.28	\$25.98	5.1%	\$13.69	2.9%	\$1.1348	-	\$1.1926	\$1.0851
33		500	\$299.40	\$266.30	\$565.70	\$299.40	\$226.20	\$525.60	\$314.62	\$279.96	\$594.58	\$314.62	\$226.20	\$540.82	\$28.88	5.1%	\$15.22	2.9%	\$1.1314		\$1.1892	\$1.0816
34		750 1,000	\$441.35 \$583.30	\$399.45	\$840.80 \$1,115.90	\$441.35 \$583.30	\$339.30	\$780.65 \$1,035.70	\$464.24 \$613.85	\$419.93	\$884.17 \$1,173.76	\$464.24 \$613.85	\$339.30 \$452.40	\$803.54 \$1,066.25	\$43.37 \$57.86	5.2% 5.2%	\$22.89 \$30.55	2.9% 2.9%		\$1.0409 \$1.0357	-	\$1.0714 \$1.0662
35		1,000	ψ505.50	φ332.00	\$1,115.90	ψ505.50	\$432.4U	\$1,033.70	ψ013.03	\$339.9T	φ1,173.70	ψ013.03	ֆ 4 32.40	\$1,000.23	φ37.00	3.2%		2.9%	ֆ1.11 39	φ1.033 <i>1</i>	\$1.1738	\$1.0002
37	Estimated Bill Per	contilos																				
38	Winter - 25%	60	\$49.57	\$31.96	\$81.52			1	\$51.30	\$33.59	\$84.89	I	1		\$3.37	4.1%	Ī		\$1.3587	Ī	\$1.4149	
30	Winter - 50%	100	\$72.28	\$53.26	\$125.54				\$75.24	\$55.99	\$131.23				\$5.69	4.1%			\$1.2554		\$1.3123	
4 ∩	Winter - 75%	175	\$114.87	\$93.21	\$208.07				\$120.12	\$97.98	\$218.11				\$10.04	4.8%			\$1.1890		\$1.2463	
<u>4</u> 1	Summer - 25%	173	ψ. ι τ.υ <i>ι</i>	Ψ50.21	Ψ200.01	\$22.31	\$5.43	\$27.74	Ψ120.12	ψ51.50	Ψ2 10.11	\$22.57	\$5.43	\$28.00	ψ10.04	7.070	\$0.26	0.9%	ψ1.1000	\$2.3119	ψ1.2700	\$2.3336
	Summer - 50%	20				\$26.86	\$9.05	\$35.90				\$27.36	\$9.05	\$36.41			\$0.51	1.4%		\$1.7952		\$1.8205
	Summer - 75%	30				\$32.53	\$13.57	\$46.11				\$33.35	\$13.57	\$46.92			\$0.81	1.8%		\$1.5369	.	\$1.5639

Estimated Bill Percentiles per 2010 MCS

Docket No. DG 20-105 Appendix 8 Page 3 of 10

Liberty Utilities (EnergyNorth Natural Gas) Corp.

COMPARATIVE MONTHLY BILLING UNDER PRESENT AND PROPOSED RATES RATE R-4 : LOW INCOME RESIDENTIAL HEATING

Present Rates	Winter	Summer
Cost of Gas	\$0.4737	\$0.3935
LDAC	\$0.0589	\$0.0589
Customer charge Sales rate	\$8.53	\$15.50
First Block Size		
Block 1	\$0.3123	\$0.5678
Block 2	\$0.3123	\$0.5678

Proposed Rates	Winter	Summer
Cost of Gas	\$0.5010	\$0.3935
LDAC	\$0.0589	\$0.0589
Customer charge	\$8.47	\$15.39
Sales rate		
First Block Size		
Block 1	\$0.3292	\$0.5985
Block 2	\$0.3292	\$0.5985

11	11																					
12				Мо	nthly Bills a	t Present R	lates			Mon	thly Bills at	Proposed R	lates		Ch	ange in	Monthly Bi	II		Unit (Costs	
13	U	Jse per		Winter			Summer			Winter			Summer		Wint	ter	Sumn	ner	Curi	ent	Prop	osed
	ľ	Month	Base	COG /		Base	COG/		Base	COG/		Base	COG /									
14	(ti	therms)	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	\$	%	\$	%	Winter	Summer	Winter	Summer
15																						
16		0	\$8.53	\$0.00	\$8.53	\$15.50	\$0.00	\$15.50	\$8.47	\$0.00	\$8.47	\$15.39	\$0.00	\$15.39	-\$0.06	-0.8%	-\$0.10	-0.7%	\$0.0000	\$0.0000	\$0.0000	\$0.0000
17		10	\$11.65	\$5.33	\$16.98	\$21.17	\$4.52	\$25.70	\$11.76	\$5.60	\$17.36	\$21.38	\$4.52	\$25.90	\$0.38	2.2%	\$0.20	0.8%	\$1.6979	\$2.5699	\$1.7356	\$2.5901
18		25	\$16.34	\$13.32	\$29.65	\$29.69	\$11.31	\$41.00	\$16.69	\$14.00	\$30.69	\$30.35	\$11.31	\$41.66	\$1.04	3.5%	\$0.66	1.6%	\$1.1861	\$1.6401	\$1.2277	\$1.6666
19		50	\$24.15	\$26.63	\$50.78	\$43.89	\$22.62	\$66.51	\$24.92	\$28.00	\$52.92	\$45.32	\$22.62	\$67.94	\$2.14	4.2%	\$1.43	2.1%	\$1.0155	\$1.3301	\$1.0584	\$1.3587
20		75	\$31.95	\$39.95	\$71.90	\$58.08	\$33.93	\$92.01	\$33.15	\$41.99	\$75.15	\$60.28	\$33.93	\$94.21	\$3.25	4.5%	\$2.19	2.4%	\$0.9586	\$1.2268	\$1.0019	\$1.2561
21		100	\$39.76	\$53.26	\$93.02	\$72.28	\$45.24	\$117.52	\$41.38	\$55.99	\$97.37	\$75.24	\$45.24	\$120.48	\$4.35	4.7%	\$2.96	2.5%	\$0.9302	\$1.1752	\$0.9737	\$1.2048
22		125	\$47.57	\$66.58	\$114.14	\$86.47	\$56.55	\$143.02	\$49.61	\$69.99	\$119.60	\$90.20	\$56.55	\$146.75	\$5.46	4.8%	\$3.73	2.6%	\$0.9131	\$1.1442	\$0.9568	\$1.1740
23		150	\$55.38	\$79.89	\$135.27	\$100.67	\$67.86	\$168.53	\$57.84	\$83.99	\$141.83	\$105.16	\$67.86	\$173.02	\$6.56	4.9%	\$4.49	2.7%	\$0.9018	\$1.1235	\$0.9455	\$1.1535
24		175	\$63.18	\$93.21	\$156.39	\$114.86	\$79.17	\$194.03	\$66.07	\$97.98	\$164.05	\$120.12	\$79.17	\$199.29	\$7.66	4.9%	\$5.26	2.7%	\$0.8936	\$1.1088	\$0.9374	\$1.1388
25		200	\$70.99	\$106.52	\$177.51	\$129.06	\$90.48	\$219.54	\$74.30	\$111.98	\$186.28	\$135.08	\$90.48	\$225.56	\$8.77	4.9%	\$6.03	2.7%	\$0.8876	\$1.0977	\$0.9314	\$1.1278
26		225	\$78.80	\$119.84	\$198.63	\$143.25	\$101.79	\$245.04	\$82.52	\$125.98	\$208.51	\$150.05	\$101.79	\$251.84	\$9.87	5.0%	\$6.79	2.8%	\$0.8828	\$1.0891	\$0.9267	\$1.1193
27		250	\$86.61	\$133.15	\$219.76	\$157.45	\$113.10	\$270.55	\$90.75	\$139.98	\$230.73	\$165.01	\$113.10	\$278.11	\$10.98	5.0%	\$7.56	2.8%	\$0.8790	\$1.0822	\$0.9229	\$1.1124
28		275	\$94.41	\$146.47	\$240.88	\$171.64	\$124.41	\$296.05	\$98.98	\$153.98	\$252.96	\$179.97	\$124.41	\$304.38	\$12.08	5.0%	\$8.33	2.8%	\$0.8759	\$1.0766	\$0.9198	\$1.1068
29		300	\$102.22	\$159.78	\$262.00	\$185.84	\$135.72	\$321.56	\$107.21	\$167.97	\$275.18	\$194.93	\$135.72	\$330.65	\$13.18	5.0%	\$9.09	2.8%	\$0.8733	\$1.0719	\$0.9173	\$1.1022
30		350	\$117.84	\$186.41	\$304.25	\$214.23	\$158.34	\$372.57	\$123.67	\$195.97	\$319.64	\$224.85	\$158.34	\$383.19	\$15.39	5.1%	\$10.63	2.9%	\$0.8693	\$1.0645	\$0.9133	\$1.0948
31		400	\$133.45	\$213.04	\$346.49	\$242.62	\$180.96	\$423.58	\$140.13	\$223.96	\$364.09	\$254.78	\$180.96	\$435.74	\$17.60	5.1%	\$12.16	2.9%	\$0.8662	\$1.0589	\$0.9102	\$1.0893
32		450	\$149.07	\$239.67	\$388.74	\$271.01	\$203.58	\$474.59	\$156.58	\$251.96	\$408.54	\$284.70	\$203.58	\$488.28	\$19.81	5.1%	\$13.69	2.9%	\$0.8639	\$1.0546	\$0.9079	\$1.0851
33		500	\$164.68	\$266.30	\$430.98	\$299.40	\$226.20	\$525.60	\$173.04	\$279.96	\$453.00	\$314.62	\$226.20	\$540.82	\$22.02	5.1%	\$15.22	2.9%	\$0.8620	\$1.0512	\$0.9060	\$1.0816
34			\$242.76	\$399.45	\$642.21	\$441.35	\$339.30	\$780.65	\$255.33	\$419.93	\$675.26	\$464.24	\$339.30	\$803.54	\$33.06	5.1%	\$22.89	2.9%	\$0.8563	\$1.0409	\$0.9004	\$1.0714
35	•	1,000	\$320.83	\$532.60	\$853.43	\$583.30	\$452.40	\$1,035.70	\$337.62	\$559.91	\$897.53	\$613.85	\$452.40	\$1,066.25	\$44.10	5.2%	\$30.55	2.9%	\$0.8534	\$1.0357	\$0.8975	\$1.0662
36		•																				
37	Estimated Bill Perc	entiles																				
38	Winter - 25%	70	\$30.39	\$37.28	\$67.67				\$31.51	\$39.19	\$70.70				\$3.03	4.5%			\$0.9668		\$1.0100	
39	Winter - 50%	100	\$39.76	\$53.26	\$93.02				\$41.38	\$55.99	\$97.37				\$4.35	4.7%			\$0.9302		\$0.9737	
40	Winter - 75%	150	\$55.38	\$79.89	\$135.27				\$57.84	\$83.99	\$141.83				\$6.56	4.9%			\$0.9018		\$0.9455	
41	Summer - 25%	14				\$23.45	\$6.33	\$29.78				\$23.77	\$6.33	\$30.10		l	\$0.32	1.1%		\$2.1271		\$2.1503
42	Summer - 50%	25				\$29.69	\$11.31	\$41.00				\$30.35	\$11.31	\$41.66		l	\$0.66	1.6%		\$1.6401		\$1.6666
43	Summer - 75%	40				\$38.21	\$18.10	\$56.30				\$39.33	\$18.10	\$57.43			\$1.12	2.0%		\$1.4076		\$1.4357

Estimated Bill Percentiles per 2010 MCS

Docket No. DG 20-105 Appendix 8 Page 4 of 10

Liberty Utilities (EnergyNorth Natural Gas) Corp.

COMPARATIVE MONTHLY BILLING UNDER PRESENT AND PROPOSED RATES RATE G-41 : COMMERCIAL/INDUSTRIAL - LOW ANNUAL USE, HIGH WINTER USE

Present Rates	Winter	Summer
Cost of Gas	\$0.4724	\$0.3886
LDAC	\$0.0555	\$0.0555
Customer charge	\$57.46	
Sales rate		
First Block Size	100	20
Block 1	\$0.4711	\$0.4711
Block 2	\$0.3165	\$0.3165

Proposed Rates	Winter	Summer
Cost of Gas	\$0.4997	\$0.3886
LDAC	\$0.0555	\$0.0555
Customer charge Sales rate	\$59.55	
First Block Size	100	20
Block 1	\$0.4848	\$0.4848
Block 2	\$0.3309	\$0.3309

3					DIOCK Z		ψυ.5105	ψυ.5105		DIOCK Z		Ψ0.5509	Ψ0.5509								
10																					
11																					
12			Moi	nthly Bills a	it Present R	ates		Monthly Bills at Proposed Rates						Change in Monthly Bill				Unit Costs			
13	Use per		Winter			Summer			Winter			Summer		Win	ter	Sumi	mer	Cur	rent	Propo	osed
	Month	Base	COG /		Base	COG /		Base	COG /		Base	COG /									
14	(therms)	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	\$	%	\$	%	Winter	Summer	Winter	Summer
15																					
16	0	\$57	\$0	\$57	\$57	\$0	\$57	\$60	\$0	\$60	\$60	\$0	\$60	\$2.09	3.6%	\$2.09	3.6%	\$0.0000	\$0.0000	\$0.0000	\$0.0000
17	10	\$62	\$5	\$67	\$62	\$4	\$67	\$64	\$6	\$70	\$64	\$4	\$69	\$2.50	3.7%	\$2.22	3.3%	\$6.7450	\$6.6612	\$6.9945	\$6.8834
18	25	\$69	\$13	\$82	\$68	\$11	\$80	\$72	\$14	\$86	\$71	\$11	\$82	\$3.11	3.8%	\$2.43	3.1%	\$3.2974	\$3.1827	\$3.4218	\$3.2799
19	50	\$81	\$26	\$107	\$76	\$22	\$99	\$84	\$28	\$112	\$79	\$22	\$101	\$4.13	3.8%	\$2.79	2.8%	\$2.1482	\$1.9716	\$2.2309	\$2.0275
20	75	\$93	\$40	\$132	\$84	\$33	\$118	\$96	\$42	\$138	\$87	\$33	\$121	\$5.16	3.9%	\$3.15	2.7%	\$1.7651	\$1.5680	\$1.8339	\$1.6100
21	100	\$105	\$53	\$157	\$92	\$44	\$137	\$108	\$56	\$164	\$96	\$44	\$140	\$6.18	3.9%	\$3.51	2.6%	\$1.5736	\$1.3661	\$1.6354	\$1.4012
22	150	\$120	\$79	\$200	\$108	\$67	\$175	\$125	\$83	\$208	\$112	\$67	\$179	\$8.27	4.1%	\$4.23	2.4%	\$1.3305	\$1.1643	\$1.3857	\$1.1925
23	200	\$136	\$106	\$242	\$124	\$89	\$213	\$141	\$111	\$252	\$129	\$89	\$218	\$10.35	4.3%	\$4.95	2.3%	\$1.2090	\$1.0634	\$1.2608	\$1.0881
24	250	\$152	\$132	\$284	\$140	\$111	\$251	\$158	\$139	\$296	\$145	\$111	\$256	\$12.44	4.4%	\$5.67	2.3%	\$1.1361	\$1.0028	\$1.1858	\$1.0255
25	300	\$168	\$158	\$326	\$156	\$133	\$289	\$174	\$167	\$341	\$162	\$133	\$295	\$14.53	4.5%	\$6.39	2.2%	\$1.0875	\$0.9624	\$1.1359	\$0.9838
26	350	\$184	\$185	\$368	\$171	\$155	\$327	\$191	\$194	\$385	\$178	\$155	\$334	\$16.61	4.5%	\$7.12	2.2%	\$1.0527	\$0.9336	\$1.1002	\$0.9539
27	400	\$200	\$211	\$411	\$187	\$178	\$365	\$207	\$222	\$429	\$195	\$178	\$373	\$18.70	4.6%	\$7.84	2.1%	\$1.0267	\$0.9120	\$1.0734	\$0.9316
28	500	\$231	\$264	\$495	\$219	\$222	\$441	\$240	\$278	\$518	\$228	\$222	\$450	\$22.87	4.6%	\$9.28	2.1%	\$0.9902	\$0.8817	\$1.0360	\$0.9003
29	600	\$263	\$317	\$580	\$250	\$266	\$517	\$273	\$333	\$607	\$261	\$266	\$528	\$27.05	4.7%	\$10.72	2.1%	\$0.9659	\$0.8615	\$1.0110	\$0.8794
30	700	\$294	\$370	\$664	\$282	\$311	\$593	\$307	\$389	\$695	\$294	\$311	\$605	\$31.22	4.7%	\$12.16	2.1%	\$0.9486	\$0.8471	\$0.9932	\$0.8645
31	800	\$326	\$422	\$748	\$314	\$355	\$669	\$340	\$444	\$784	\$327	\$355	\$683	\$35.39	4.7%	\$13.60	2.0%	\$0.9356	\$0.8363	\$0.9798	\$0.8533
32	900	\$358	\$475	\$833	\$345	\$400	\$745	\$373	\$500	\$872	\$360	\$400	\$760	\$39.56	4.8%	\$15.04	2.0%	\$0.9254	\$0.8279	\$0.9694	\$0.8446
33	1,000	\$389	\$528	\$917	\$377	\$444	\$821	\$406	\$555	\$961	\$394	\$444	\$838	\$43.74	4.8%	\$16.48	2.0%	\$0.9173	\$0.8212	\$0.9611	\$0.8376
34	1,250	\$469	\$660	\$1,128	\$456	\$555	\$1,011	\$489	\$694	\$1,183	\$476	\$555	\$1,031	\$54.17	4.8%	\$20.09	2.0%	\$0.9027	\$0.8090	\$0.9461	\$0.8251
35	1,500	\$548	\$792	\$1,340	\$535	\$666	\$1,201	\$571	\$833	\$1,404	\$559	\$666	\$1,225	\$64.60	4.8%	\$23.69	2.0%	\$0.8930	\$0.8010	\$0.9361	\$0.8168
36		•			•	•			•	-	•	•		•	-				-	-	
37 Estimated Bill Pe	ercentiles																				
38 Winter - 25%	70	\$90	\$37	\$127				\$93	\$39	\$132				\$4.95	3.9%			\$1.8199		\$1.8906	
39 Winter - 50%	200	\$136	\$106	\$242				\$141	\$111	\$252				\$10.35	4.3%			\$1.2090		\$1.2608	
40 Winter - 75%	500	\$231	\$264	\$495				\$240	\$278	\$518				\$22.87	4.6%			\$0.9902		\$1.0360	
41 Summer - 25%	0		,	, ,	\$57	\$0	\$57	•		, -	\$60	\$0	\$60	•		\$2.09	3.6%		\$0.0000		\$0.0000
42 Summer - 50%	8				\$61	\$4	\$65				\$63	\$4	\$67			\$2.19	3.4%		\$8.0977		\$8.3720
43 Summer - 75%	45				\$75	\$20	\$95				\$78	\$20	\$97			\$2.72	2.9%		\$2.1062		\$2.1666
	-					, ,	7					, -	T - 1								

Estimated Bill Percentiles per 2010 MCS

9

Docket No. DG 20-105 Appendix 8 Page 5 of 10

Liberty Utilities (EnergyNorth Natural Gas) Corp.

COMPARATIVE MONTHLY BILLING UNDER PRESENT AND PROPOSED RATES RATE G-42 : COMMERCIAL/INDUSTRIAL - MEDIUM ANNUAL USE, HIGH WINTER USE

Present Rates	Winter	Summer
Cost of Gas	\$0.4724	\$0.3886
LDAC	\$0.0555	\$0.0555
Customer charge	\$172.39	
Sales rate		
First Block Size	1000	400
Block 1	\$0.4284	\$0.4284
Block 2	\$0.2855	\$0.2855

Proposed Rates	Winter	Summer
Cost of Gas	\$0.4997	\$0.3886
LDAC	\$0.0555	\$0.0555
Customer charge	\$178.61	
Sales rate		
First Block Size	1000	400
Block 1	\$0.4409	\$0.4409
Block 2	\$0.2988	\$0.2988

10																					
11 12	İ		Mor	athly Bills o	t Present R	atos	1		Monthly	Bills at Pro	nocod D	atos		CI	nange in M	onthly Pill			L Init (Costs	
13	Hee per		Winter	itiliy bilis a	i Fieseiii N	Summer			Winter	DIIIS AL FIU		Summer		Wir		Sumr	nor	Cur			osed
13	Use per Month	Base	COG /		Base	COG /		Base	COG /		Base	COG /		VVII	itei	Sullil	iiei	Cui	IEIIL	гюр	oseu
14	(therms)	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	\$	%	\$	%	Winter	Summer	Winter	Summer
15	(triciiii3)	ratos	LD/10	TOTAL	Tutos	LD/10	TOTAL	rates	LD/10	TOTAL	rtates	LD/10	TOTAL	Ψ	70	Ψ	70	VVIIICI	Outilities	VVIIICI	Guillillei
16	0	\$172	\$0	\$172	\$172	\$0	\$172	\$179	\$0	\$179	\$179	\$0	\$179	\$6.22	3.6%	\$6.22	3.6%	\$0.0000	\$0.0000	\$0.0000	\$0.0000
17	10	\$177	\$5	\$182	\$177	\$4	\$181	\$183	\$6	\$189	\$183	\$4	\$187	\$6.62	3.6%	\$6.34		\$18.1953	•		\$18.7460
18	25	\$183	\$13	\$196	\$183	\$11	\$194	\$190	\$14	\$204	\$190	\$11	\$201	\$7.22	3.7%	\$6.53	3.4%		\$7.7681	\$8.1405	\$8.0294
19	50	\$194	\$26	\$220	\$194	\$22	\$216	\$201	\$28	\$228	\$201	\$22	\$223	\$8.21	3.7%	\$6.85	3.2%	\$4.4041	\$4.3203	\$4.5683	\$4.4572
20	75	\$205	\$40	\$244	\$205	\$33	\$238	\$212	\$42	\$253	\$212	\$33	\$245	\$9.21	3.8%	\$7.16	3.0%	\$3.2548	\$3.1710	\$3.3776	\$3.2665
21	100	\$215	\$53	\$268	\$215	\$44	\$260	\$223	\$56	\$278	\$223	\$44	\$267	\$10.20	3.8%	\$7.47	2.9%	\$2.6802	\$2.5964	\$2.7822	\$2.6711
22	150	\$237	\$79	\$316	\$237	\$67	\$303	\$245	\$83	\$328	\$245	\$67	\$311	\$12.19	3.9%	\$8.10	2.7%	\$2.1056	\$2.0218	\$2.1868	\$2.0757
23	200	\$258	\$106	\$364	\$258	\$89	\$347	\$267	\$111	\$378	\$267	\$89	\$356	\$14.18	3.9%	\$8.72	2.5%	\$1.8183	\$1.7345	\$1.8892	\$1.7781
24	250	\$279	\$132	\$411	\$279	\$111	\$391	\$289	\$139	\$428	\$289	\$111	\$400	\$16.17	3.9%	\$9.35	2.4%	\$1.6459	\$1.5621	\$1.7106	\$1.5994
25	300	\$301	\$158	\$459	\$301	\$133	\$434	\$311	\$167	\$477	\$311	\$133	\$444	\$18.17	4.0%	\$9.97	2.3%	\$1.5309	\$1.4471	\$1.5915	\$1.4804
26	350	\$322	\$185	\$507	\$322	\$155	\$478	\$333	\$194	\$527	\$333	\$155	\$488	\$20.16	4.0%	\$10.60	2.2%	\$1.4488	\$1.3650	\$1.5064	\$1.3953
27	400	\$344	\$211	\$555	\$344	\$178	\$521	\$355	\$222	\$577	\$355	\$178	\$533	\$22.15	4.0%	\$11.22	2.2%	\$1.3873	\$1.3035	\$1.4426	\$1.3315
28	500	\$387	\$264	\$651	\$372	\$222	\$594	\$399	\$278	\$677	\$385	\$222	\$607	\$26.13	4.0%	\$12.55	2.1%	\$1.3011	\$1.1887	\$1.3533	\$1.2138
29	750	\$494	\$396	\$890	\$444	\$333	\$777	\$509	\$416	\$926	\$460	\$333	\$793	\$36.08	4.1%	\$15.87	2.0%	\$1.1862	\$1.0357	\$1.2343	\$1.0568
30	1,000	\$601	\$528	\$1,129	\$515	\$444	\$959	\$620	\$555	\$1,175	\$534	\$444	\$978	\$46.04	4.1%	\$19.20	2.0%	-	\$0.9592	\$1.1747	\$0.9783
31	1,500	\$744	\$792	\$1,535	\$658	\$666	\$1,324	\$769	\$833	\$1,602	\$684	\$666	\$1,350	\$66.34	4.3%	\$25.84	2.0%	-	\$0.8826	\$1.0678	
32	2,000	\$886	\$1,056	\$1,942	\$801	\$888	\$1,689	\$918	\$1,110	\$2,029	\$833	\$888	\$1,721	\$86.64	4.5%	\$32.48	1.9%	-	\$0.8444	\$1.0144	
33	3,000	\$1,172	\$1,584	\$2,755	\$1,086	\$1,332	\$2,418	\$1,217	\$1,666	\$2,883		\$1,332	. ,	\$127.24	4.6%	\$45.77	1.9%	-	\$0.8061	\$0.9609	
34	4,000	\$1,457	\$2,112	\$3,569	\$1,372	\$1,776	\$3,148	\$1,516	\$2,221	\$3,737		\$1,776	. ,	\$167.84	4.7%	\$59.06	1.9%	-	\$0.7870	\$0.9342	
35	5,000	\$1,743	\$2,640	\$4,382	\$1,657	\$2,221	\$3,878	\$1,815	\$2,776	\$4,591	\$1,729	\$2,221	\$3,950	\$208.44	4.8%	\$72.35	1.9%	\$0.8765	\$0.7755	\$0.9181	\$0.7900
36																					
37 Estimated Bill P																					
38 Winter - 25%	1,300	\$686	\$686	\$1,373				\$709	\$722	\$1,431				\$58.22	4.2%			\$1.0559		\$1.1007	1 1
39 Winter - 50%	2,000	\$886	\$1,056	\$1,942				\$918	\$1,110	\$2,029				\$86.64	4.5%			\$0.9710		\$1.0144	
40 Winter - 75%	3,500	\$1,315	\$1,848	\$3,162				\$1,366	\$1,943	\$3,310				\$147.54	4.7%			\$0.9035		\$0.9456	
41 Summer - 25%	45				\$192	\$20	\$212				\$198	\$20	\$218			\$6.78	3.2%		\$4.7034		\$4.8541
42 Summer - 50%	350				\$322	\$155	\$478				\$333	\$155	\$488			\$10.60	2.2%		\$1.3650		\$1.3953
43 Summer - 75%	750				\$444	\$333	\$777				\$460	\$333	\$793			\$15.87	2.0%		\$1.0357		\$1.0568

Estimated Bill Percentiles per 2010 MCS

Docket No. DG 20-105 Appendix 8 Page 6 of 10

Liberty Utilities (EnergyNorth Natural Gas) Corp.

COMPARATIVE MONTHLY BILLING UNDER PRESENT AND PROPOSED RATES RATE G-43 : COMMERCIAL/INDUSTRIAL - HIGH ANNUAL USE, HIGH WINTER USE

Present Rates	Winter	Summer
Cost of Gas	\$0.4724	\$0.3886
LDAC	\$0.0555	\$0.0555
Customer charge Sales rate	\$739.83	
First Block Size		
Block 1	\$0.2633	\$0.1204
Block 2	\$0.2633	\$0.1204

\$794

\$1,161 \$1,944 \$200

\$1,554

\$4,441

\$994

\$2,716

\$6,385

Proposed Rates	Winter	Summer
Cost of Gas	\$0.4997	\$0.3886
LDAC	\$0.0555	\$0.0555
Customer charge	\$765.51	
Sales rate		
First Block Size		
Block 1	\$0.2717	\$0.1295
Block 2	\$0.2717	\$0.1295

11																					
12			Moi	nthly Bills a	t Present R	Rates			Month	nly Bills at F	Proposed R	ates			hange in	Monthly Bill			Unit (Costs	
13	Use per		Winter	-		Summer			Winter			Summer		Wint	er	Sumn	ner	Curr	ent	Prop	osed
	Month	Base	COG /		Base	COG /		Base	COG /		Base	COG /									
14	(therms)	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	\$	%	\$	%	Winter	Summer	Winter	Summer
15	i																				
16	200	\$792	\$106	\$898	\$764	\$89	\$853	\$820	\$111	\$931	\$791	\$89	\$880	\$32.82	3.7%	\$27.50	3.2%	\$4.4904	\$4.2637	\$4.6544	\$4.4011
17	500	\$871	\$264	\$1,135	\$800	\$222	\$1,022	\$901	\$278	\$1,179	\$830	\$222	\$1,052	\$43.53	3.8%	\$30.23	3.0%	\$2.2709	\$2.0442	\$2.3579	\$2.1046
18	1,000	\$1,003	\$528	\$1,531	\$860	\$444	\$1,304	\$1,037	\$555	\$1,592	\$895	\$444	\$1,339	\$61.38	4.0%	\$34.79	2.7%	\$1.5310	\$1.3043	\$1.5924	\$1.3391
19	1,250	\$1,069	\$660	\$1,729	\$890	\$555	\$1,445	\$1,105	\$694	\$1,799	\$927	\$555	\$1,483	\$70.31	4.1%	\$37.06	2.6%	\$1.3831	\$1.1564	\$1.4393	\$1.1860
20	1,500	\$1,135	\$792	\$1,927	\$920	\$666	\$1,587	\$1,173	\$833	\$2,006	\$960	\$666	\$1,626	\$79.24	4.1%	\$39.34	2.5%	\$1.2844	\$1.0577	\$1.3372	\$1.0839
21	1,750	\$1,201	\$924	\$2,124	\$951	\$777	\$1,728	\$1,241	\$972	\$2,213	\$992	\$777	\$1,769	\$88.16	4.1%	\$41.62	2.4%	\$1.2140	\$0.9873	\$1.2643	\$1.0110
22	2,000	\$1,266	\$1,056	\$2,322	\$981	\$888	\$1,869	\$1,309	\$1,110	\$2,419	\$1,025	\$888	\$1,913	\$97.09	4.2%	\$43.90	2.3%	\$1.1611	\$0.9344	\$1.2097	\$0.9564
23	2,500	\$1,398	\$1,320	\$2,718	\$1,041	\$1,110	\$2,151	\$1,445	\$1,388	\$2,833	\$1,089	\$1,110	\$2,200	\$114.94	4.2%	\$48.45	2.3%	\$1.0871	\$0.8604	\$1.1331	\$0.8798
24	3,000	\$1,530	\$1,584	\$3,113	\$1,101	\$1,332	\$2,433	\$1,581	\$1,666	\$3,246	\$1,154	\$1,332	\$2,486	\$132.79	4.3%	\$53.01	2.2%	\$1.0378	\$0.8111	\$1.0821	\$0.8288
25	3,500	\$1,661	\$1,848	\$3,509	\$1,161	\$1,554	\$2,716	\$1,716	\$1,943	\$3,660	\$1,219	\$1,554	\$2,773	\$150.65	4.3%	\$57.56	2.1%	\$1.0026	\$0.7759	\$1.0456	\$0.7923
26	4,000	\$1,793	\$2,112	\$3,905	\$1,221	\$1,776	\$2,998	\$1,852	\$2,221	\$4,073	\$1,284	\$1,776	\$3,060	\$168.50	4.3%	\$62.12	2.1%	\$0.9762	\$0.7495	\$1.0183	\$0.7650
27	4,500	\$1,925	\$2,376	\$4,300	\$1,282	\$1,998	\$3,280	\$1,988	\$2,498	\$4,487	\$1,348	\$1,998	\$3,347	\$186.35	4.3%	\$66.67	2.0%	\$0.9556	\$0.7289	\$0.9970	\$0.7437
28	5,000	\$2,056	\$2,640	\$4,696	\$1,342	\$2,221	\$3,562	\$2,124	\$2,776	\$4,900	\$1,413	\$2,221	\$3,634	\$204.20	4.3%	\$71.23	2.0%	\$0.9392	\$0.7125	\$0.9800	\$0.7267
29	6,000	\$2,320	\$3,167	\$5,487	\$1,462	\$2,665	\$4,127	\$2,396	\$3,331	\$5,727	\$1,543	\$2,665	\$4,207	\$239.91	4.4%	\$80.34	1.9%	\$0.9145	\$0.6878	\$0.9545	\$0.7012
30	•	\$2,583	\$3,695	\$6,278	\$1,583	\$3,109	\$4,691	\$2,667	\$3,886	\$6,554	\$1,672	\$3,109	\$4,781	\$275.61	4.4%	\$89.45	1.9%	\$0.8969	\$0.6702	\$0.9363	\$0.6830
31	8,000	\$2,846	\$4,223	\$7,069	\$1,703	\$3,553	\$5,256	\$2,939	\$4,442	\$7,381	\$1,802	\$3,553	\$5,354	\$311.32	4.4%	\$98.55	1.9%	\$0.8837	\$0.6570	\$0.9226	\$0.6693
32	9,000	\$3,110	\$4,751	\$7,861	\$1,823	\$3,997	\$5,820	\$3,211	\$4,997	\$8,208	\$1,931	\$3,997	\$5,928	\$347.02	4.4%	\$107.66	1.8%	\$0.8734	\$0.6467	\$0.9120	\$0.6587
33	10,000	\$3,373	\$5,279	\$8,652	\$1,944	\$4,441	\$6,385	\$3,482	\$5,552	\$9,035	\$2,061	\$4,441	\$6,502	\$382.73	4.4%	\$116.77	1.8%	\$0.8652	\$0.6385	\$0.9035	\$0.6502
34	•	\$4,689	\$7,919	\$12,608	\$2,546	\$6,662	\$9,207	\$4,841	\$8,328	\$13,169	\$2,708	\$6,662	\$9,370	\$561.25	4.5%	\$162.32	1.8%	\$0.8405		\$0.8779	\$0.6246
35	20,000	\$6,006	\$10,558	\$16,564	\$3,148	\$8,882	\$12,030	\$6,199	\$11,104	\$17,304	\$3,356	\$8,882	\$12,238	\$739.78	4.5%	\$207.87	1.7%	\$0.8282	\$0.6015	\$0.8652	\$0.6119
36	i																				
37	Estimated Bill Percentiles																				
38	Winter - 25% 9,000	\$3,110	\$4,751	\$7,861				\$3,211	\$4,997	\$8,208				\$347.02	4.4%			\$0.8734		\$0.9120	7
39	Winter - 50% 15,000	\$4,689	\$7,919	\$12,608				\$4,841	\$8,328	\$13,169				\$561.25	4.5%			\$0.8405		\$0.8779	J
40	Winter - 75% 25,000	\$7,322	\$13,198	\$20,520				\$7,558	\$13,880	\$21,438				\$918.30	4.5%			\$0.8208		\$0.8575	

\$824 \$1,219 \$2,061

\$200

\$1,554

\$4,441

\$1,024

\$2,773

\$6,502

\$29.78

\$57.56

\$116.77

3.0%

2.1%

1.8%

\$2.2086

\$0.7759

\$0.6385

\$2.2747

\$0.7923

\$0.6502

Estimated Bill Percentiles per 2010 MCS

450

3,500

10,000

41 Summer - 25%

42 Summer - 50%

43 Summer - 75%

Docket No. DG 20-105 Appendix 8 Page 7 of 10

Liberty Utilities (EnergyNorth Natural Gas) Corp.

COMPARATIVE MONTHLY BILLING UNDER PRESENT AND PROPOSED RATES RATE G-51: COMMERCIAL/INDUSTRIAL - LOW ANNUAL USE, LOW WINTER USE

Present Rates	Winter	Summer
Cost of Gas	\$0.4792	\$0.3999
LDAC	\$0.0555	\$0.0555
Customer charge Sales rate	\$57.46	
First Block Size	100	100
Block 1	\$0.2839	\$0.2839
Block 2	\$0.1846	\$0.1846

_		
Proposed Rates	Winter	Summer
Cost of Gas	\$0.5065	\$0.3999
LDAC	\$0.0555	\$0.0555
Customer charge	\$59.57	
Sales rate		
First Block Size	100	100
Block 1	\$0.2920	\$0.2920
Block 2	\$0.1934	\$0.1934

11	11																						
12				Mor	nthly Bills at	Present R	lates			Month	nly Bills at f	Proposed R	ates		Cha	nge in I	Monthly B	ill		Unit	Costs		
13	Use	e per		Winter			Summer			Winter			Summer		Wint	er	Sumi	mer	Cu	ırrent	Prop	osed	
	Mo	onth	Base	COG/		Base	COG /		Base	COG /		Base	COG /										
14	(the	erms)	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	\$	%	\$	%	Winter	Summer	Winter	Summer	
15																							
16		0	\$57	\$0	\$57	\$57	\$0	\$57	\$60	\$0	\$60	\$60	\$0	\$60	\$2.11	3.7%	\$2.11		\$0.0000	\$0.0000	\$0.0000	\$0.0000	
17		10	\$60	\$5	\$66	\$60	\$5	\$65	\$62	\$6	\$68	\$62	\$5	\$67	\$2.46	3.8%	\$2.19		\$6.5646		\$6.8109	\$6.7043	
18		25	\$65	\$13	\$78	\$65	\$11	\$76	\$67	\$14	\$81	\$67	\$11	\$78	\$2.99	3.8%	\$2.31		\$3.1170	\$3.0377	\$3.2368	\$3.1302	
19		50	\$72	\$27	\$98	\$72	\$23	\$94	\$74	\$28	\$102	\$74	\$23	\$97	\$3.88	3.9%	\$2.51		\$1.9678		\$2.0454	\$1.9388	
20	-	75	\$79	\$40	\$119	\$79	\$34	\$113	\$81	\$42	\$124	\$81	\$34	\$116	\$4.76	4.0%	\$2.72		\$1.5847	\$1.5054	\$1.6482	\$1.5416	
21		00	\$86	\$53	\$139	\$86	\$46	\$131	\$89	\$56	\$145	\$89	\$46	\$134	\$5.65	4.1%	\$2.92		\$1.3932	\$1.3139	\$1.4497	\$1.3431	
22		50	\$95	\$80	\$175	\$95	\$68	\$163	\$98	\$84	\$183	\$98	\$68	\$167	\$7.45	4.3%	\$3.36		\$1.1686	\$1.0893	\$1.2183	\$1.1116	
23		00	\$104	\$107	\$211	\$104	\$91	\$195	\$108	\$112	\$221	\$108	\$91	\$199	\$9.26	4.4%	\$3.79		\$1.0563	\$0.9770	\$1.1025	\$0.9959	
24		50	\$114	\$134	\$247	\$114	\$114	\$227	\$118	\$141	\$258	\$118	\$114	\$232	\$11.06	4.5%	\$4.23		\$0.9889	\$0.9096	\$1.0331	\$0.9265	
25		00	\$123	\$160	\$283	\$123	\$137	\$259	\$127	\$169	\$296	\$127	\$137	\$264	\$12.87	4.5%	\$4.67		\$0.9439	\$0.8646	\$0.9868	\$0.8802	
26		50	\$132	\$187	\$319	\$132	\$159	\$291	\$137	\$197	\$334	\$137	\$159	\$297	\$14.67	4.6%	\$5.11		\$0.9118	\$0.8325	\$0.9538	\$0.8471	
27		00	\$141	\$214	\$355	\$141	\$182	\$323	\$147	\$225	\$372	\$147	\$182	\$329	\$16.47	4.6%	\$5.55		\$0.8878	\$0.8085	\$0.9290	\$0.8223	
28		00	\$160	\$267	\$427	\$160	\$228	\$387	\$166	\$281	\$447	\$166	\$228	\$394	\$20.08	4.7%	\$6.43		\$0.8541	\$0.7748	\$0.8942	\$0.7876	
29	-	00	\$178	\$321	\$499	\$178	\$273	\$451	\$185	\$337	\$523	\$185	\$273	\$459	\$23.69	4.7%	\$7.30		\$0.8316		\$0.8711	\$0.7645	
30		00	\$197	\$374	\$571	\$197	\$319	\$515	\$205	\$393	\$598	\$205	\$319	\$524	\$27.30	4.8%	\$8.18		\$0.8156	\$0.7363	\$0.8546	\$0.7480	
31		00	\$215	\$428	\$643	\$215	\$364	\$579	\$224	\$450	\$674	\$224	\$364	\$588	\$30.91	4.8%	\$9.06		\$0.8035	\$0.7242	\$0.8422	\$0.7356	
32		00	\$234	\$481	\$715	\$234	\$410	\$643	\$243	\$506	\$749	\$243	\$410	\$653	\$34.52	4.8%	\$9.94		\$0.7942	\$0.7149	\$0.8325	\$0.7259	
33		000	\$252	\$535	\$787	\$252	\$455	\$707	\$263	\$562	\$825	\$263	\$455	\$718	\$38.13	4.8%	\$10.81		\$0.7867	\$0.7074	\$0.8248	\$0.7182	
34		250	\$298	\$668	\$967	\$298	\$569	\$867	\$311	\$703	\$1,014	\$311	\$569	\$880	\$47.15	4.9%	\$13.01		\$0.7732	\$0.6939	\$0.8109	\$0.7043	
35	1,5	500	\$344	\$802	\$1,146	\$344	\$683	\$1,027	\$359	\$843	\$1,203	\$359	\$683	\$1,043	\$56.17	4.9%	\$15.20	1.5%	\$0.7642	\$0.6849	\$0.8017	\$0.6951	
36																							
	stimated Bill Percer	ntiles																	T				
		45	\$70	\$24	\$94				\$73	\$25	\$98				\$3.70	3.9%			\$2.0955		\$2.1778		
• • •		75	\$100	\$94	\$193				\$103	\$98	\$202				\$8.35	4.3%			\$1.1044		\$1.1521		
40 V		50	\$150	\$241	\$391				\$156	\$253	\$409				\$18.28	4.7%			\$0.8691		\$0.9097		
41 S		6				\$59	\$3	\$62				\$61	\$3	\$64			\$2.16			\$10.3160		\$10.6756	
		60				\$74	\$27	\$102				\$77	\$27	\$104			\$2.59	2.5%		\$1.6970		\$1.7402	
43	Summer - 75% 2	250				\$114	\$114	\$227				\$118	\$114	\$232			\$4.23	1.9%		\$0.9096		\$0.9265	

Estimated Bill Percentiles per 2010 MCS

Docket No. DG 20-105 Appendix 8 Page 8 of 10

Liberty Utilities (EnergyNorth Natural Gas) Corp.

COMPARATIVE MONTHLY BILLING UNDER PRESENT AND PROPOSED RATES RATE G-52: COMMERCIAL/INDUSTRIAL - MEDIUM ANNUAL USE, LOW WINTER USE

P	resent Rates	Winter	Summer
С	ost of Gas	\$0.4792	\$0.3999
L	DAC	\$0.0555	\$0.0555
	ustomer charge ales rate	\$172.39	
	irst Block Size	1000	1000
В	lock 1	\$0.2439	\$0.1767
В	lock 2	\$0.1624	\$0.1004

Proposed Rates	Winter	Summer
Cost of Gas	\$0.5065	\$0.3999
LDAC	\$0.0555	\$0.0555
Customer charge	\$178.49	
Sales rate		
First Block Size	1000	1000
Block 1	\$0.2515	\$0.1846
Block 2	\$0.1704	\$0.1087

•				L	DIOUR Z		ψ0.1021	4000 .	L	DIOOK Z		ψ0.1701	ψ0.1001								
10																					
11	,																				
12				nthly Bills a	t Present R	Rates				lly Bills at F	Proposed R	ates		Ch	ange in <u>I</u>	Monthly B	ill		Unit	Costs	
13	Use per		Winter			Summer			Winter			Summer		Win	ter	Sum	mer	Cur	rent	Propo	osed
	Month	Base	COG /		Base	COG /		Base	COG /		Base	COG /									
14	(therms)	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	\$	%	\$	%	Winter	Summer	Winter	Summer
15																					
16	200	\$221	\$107	\$328	\$208	\$91	\$299	\$229	\$112	\$341	\$215	\$91	\$307	\$13.10	4.0%	\$7.69		\$1.6406	-	\$1.7060	\$1.5325
17	300	\$246	\$160	\$406	\$225	\$137	\$362	\$254	\$169	\$423	\$234	\$137	\$371	\$16.59	4.1%	\$8.49		\$1.3532	-	\$1.4085	\$1.2350
18	400	\$270	\$214	\$484	\$243	\$182	\$425	\$279	\$225	\$504	\$252	\$182	\$435	\$20.09	4.2%	\$9.28		\$1.2096	-	\$1.2598	\$1.0863
19	500	\$294	\$267	\$562	\$261	\$228	\$488	\$304	\$281	\$585	\$271	\$228	\$499	\$23.58	4.2%	\$10.07		\$1.1234	-	\$1.1705	\$0.9970
20	600	\$319	\$321	\$640	\$278	\$273	\$552	\$329	\$337	\$667	\$289	\$273	\$563	\$27.08	4.2%	\$10.87	2.0%	\$1.0659	\$0.9194	\$1.1110	\$0.9375
21	700	\$343	\$374	\$717	\$296	\$319	\$615	\$355	\$393	\$748	\$308	\$319	\$627	\$30.57	4.3%	\$11.66		\$1.0249		\$1.0685	\$0.8950
22	800	\$368	\$428	\$795	\$314	\$364	\$678	\$380	\$450	\$829	\$326	\$364	\$691	\$34.07	4.3%	\$12.45		\$0.9941		\$1.0367	\$0.8632
23	900	\$392	\$481	\$873	\$331	\$410	\$741	\$405	\$506	\$911	\$345	\$410	\$755	\$37.56	4.3%	\$13.25	1.8%	\$0.9701	\$0.8236	\$1.0119	\$0.8384
24	1,000	\$416	\$535	\$951	\$349	\$455	\$804	\$430	\$562	\$992	\$363	\$455	\$819	\$41.06	4.3%	\$14.04	1.7%	\$0.9510	\$0.8045	\$0.9921	\$0.8185
25	1,100	\$433	\$588	\$1,021	\$359	\$501	\$860	\$447	\$618	\$1,065	\$374	\$501	\$875	\$44.59	4.4%	\$14.87	1.7%	\$0.9279	\$0.7819	\$0.9684	\$0.7954
26	1,200	\$449	\$642	\$1,090	\$369	\$546	\$916	\$464	\$674	\$1,139	\$385	\$546	\$931	\$48.12	4.4%	\$15.69	1.7%	\$0.9087	\$0.7630	\$0.9488	\$0.7761
27	1,300	\$465	\$695	\$1,160	\$379	\$592	\$971	\$481	\$731	\$1,212	\$396	\$592	\$988	\$51.65	4.5%	\$16.52	1.7%	\$0.8924	\$0.7471	\$0.9321	\$0.7598
28	1,400	\$481	\$749	\$1,230	\$389	\$638	\$1,027	\$498	\$787	\$1,285	\$407	\$638	\$1,044	\$55.18	4.5%	\$17.35	1.7%	\$0.8785	\$0.7334	\$0.9179	\$0.7458
29	1,500	\$497	\$802	\$1,300	\$399	\$683	\$1,082	\$515	\$843	\$1,358	\$417	\$683	\$1,101	\$58.71	4.5%	\$18.17	1.7%	\$0.8664	\$0.7216	\$0.9055	\$0.7337
30	1,750	\$538	\$936	\$1,474	\$424	\$797	\$1,221	\$558	\$984	\$1,541	\$445	\$797	\$1,242	\$67.54	4.6%	\$20.24	1.7%	\$0.8422	\$0.6979	\$0.8808	\$0.7095
31	2,000	\$579	\$1,069	\$1,648	\$449	\$911	\$1,360	\$600	\$1,124	\$1,724	\$472	\$911	\$1,383	\$76.37	4.6%	\$22.31	1.6%	\$0.8240	\$0.6801	\$0.8622	\$0.6913
32	2,500	\$660	\$1,337	\$1,997	\$500	\$1,139	\$1,638	\$686	\$1,405	\$2,091	\$526	\$1,139	\$1,665	\$94.02	4.7%	\$26.44	1.6%	\$0.7987	\$0.6553	\$0.8363	\$0.6659
33	3,000	\$741	\$1,604	\$2,345	\$550	\$1,366	\$1,916	\$771	\$1,686	\$2,457	\$580	\$1,366	\$1,947	\$111.68	4.8%	\$30.57	1.6%	\$0.7817	\$0.6387	\$0.8190	\$0.6489
34	4,000	\$903	\$2,139	\$3,042	\$650	\$1,822	\$2,472	\$941	\$2,248	\$3,189	\$689	\$1,822	\$2,511	\$146.99	4.8%	\$38.84	1.6%	\$0.7606	\$0.6180	\$0.7973	\$0.6277
35	5,000	\$1,066	\$2,674	\$3,739	\$751	\$2,277	\$3,028	\$1,112	\$2,810	\$3,922	\$798	\$2,277	\$3,075	\$182.30	4.9%	\$47.11	1.6%	\$0.7479	\$0.6055	\$0.7843	\$0.6150
36				-		-				-					-	-			-	-	
37 Estimated Bill Pe	ercentiles																				
38 Winter - 25%	1,040	\$423	\$556	\$979				\$437	\$584	\$1,021				\$42.47	4.3%			\$0.9412		\$0.9821	
39 Winter - 50%	2,000	\$579	\$1,069	\$1,648				\$600	\$1,124	\$1,724				\$76.37	4.6%			\$0.8240		\$0.8622	
40 Winter - 75%	3,500	\$822	\$1,871	\$2,694				\$856	\$1,967	\$2,823				\$129.33	4.8%			\$0.7696		\$0.8066	
41 Summer - 25%	700				\$296	\$319	\$615	.		. ,	\$308	\$319	\$627			\$11.66	1.9%		\$0.8784		\$0.8950
42 Summer - 50%	1,040				\$353	\$474	\$827				\$367	\$474	\$841			\$14.37	1.7%		\$0.7949		\$0.8087
43 Summer - 75%	2,000				\$449	\$911	\$1,360				\$472	\$911	\$1,383			\$22.31	1.6%		\$0.6801		\$0.6913
	,					7	. ,					7	. ,								

Estimated Bill Percentiles per 2010 MCS

Docket No. DG 20-105 Appendix 8 Page 9 of 10

Liberty Utilities (EnergyNorth Natural Gas) Corp.

COMPARATIVE MONTHLY BILLING UNDER PRESENT AND PROPOSED RATES RATE G-53: COMMERCIAL/INDUSTRIAL - HIGH ANNUAL USE, LOAD FACTOR LESS THAN 90%

Present Rates	Winter	Summer
Cost of Gas	\$0.4792	\$0.3999
LDAC	\$0.0555	\$0.0555
Customer charge Sales rate	\$761.39	
First Block Size		
Block 1	\$0.1705	\$0.0818
Block 2	\$0.1705	\$0.0818

_		
Proposed Rates	Winter	Summer
Cost of Gas	\$0.5065	\$0.3999
LDAC	\$0.0555	\$0.0555
Customer charge Sales rate	\$788.87	
First Block Size		
Block 1	\$0.1758	\$0.0875
Block 2	\$0.1758	\$0.0875

· ·				L	DIOCK Z		ψ0.1700	ψ0.0010		DIOUN Z		ψ0.1700	ψ0.0070									
10																						
11																						
12			Mor	nthly Bills a	t Present R	ates		Monthly Bills at Proposed Rates					Change in Monthly Bill				Unit Costs					
13	Use per		Winter			Summer			Winter			Summer		Winter		Summer		Current		Propo	osed	
	Month	Base	COG /		Base	COG /		Base	COG /		Base	COG /										
14	(therms)	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	\$	%	\$	%	Winter	Summer	Winter	Summer	
15																						
16	1,000	\$932	\$535	\$1,467	\$843	\$455	\$1,299	\$965	\$562	\$1,527	\$876	\$455	\$1,332	\$60.13	4.1%	\$33.21	2.6%	\$1.4666	\$1.2986	\$1.5267	\$1.3318	
17	2,500	\$1,188	\$1,337	\$2,524	\$966	\$1,139	\$2,104	\$1,228	\$1,405	\$2,633	\$1,008	\$1,139	\$2,146	\$109.09	4.3%	\$41.79	2.0%	\$1.0098	\$0.8418	\$1.0534	\$0.8585	
18	5,000	\$1,614	\$2,674	\$4,287	\$1,170	\$2,277	\$3,447	\$1,668	\$2,810	\$4,478	\$1,226	\$2,277	\$3,503	\$190.70	4.4%	\$56.10	1.6%	\$0.8575	\$0.6895	\$0.8956	\$0.7007	
19	7,500	\$2,040	\$4,010	\$6,050	\$1,375	\$3,416	\$4,790	\$2,108	\$4,215	\$6,323	\$1,445	\$3,416	\$4,861	\$272.31	4.5%	\$70.41	1.5%	\$0.8067	\$0.6387	\$0.8430	\$0.6481	
20	10,000	\$2,466	\$5,347	\$7,813	\$1,579	\$4,554	\$6,133	\$2,547	\$5,620	\$8,167	\$1,664	\$4,554	\$6,218	\$353.91	4.5%	\$84.71	1.4%	\$0.7813	\$0.6133	\$0.8167	\$0.6218	
21	12,500	\$2,893	\$6,684	\$9,576	\$1,784	\$5,693	\$7,476	\$2,987	\$7,025	\$10,012	\$1,883	\$5,693	\$7,575	\$435.52	4.5%	\$99.02	1.3%	\$0.7661	\$0.5981	\$0.8010	\$0.6060	
22	15,000	\$3,319	\$8,021	\$11,339	\$1,988	\$6,831	\$8,819	\$3,426	\$8,430	\$11,857	\$2,102	\$6,831	\$8,933	\$517.13	4.6%	\$113.33	1.3%	\$0.7560	\$0.5880	\$0.7904	\$0.5955	
23	20,000	\$4,171	\$10,694	\$14,865	\$2,397	\$9,108	\$11,505	\$4,305	\$11,240	\$15,546	\$2,539	\$9,108	\$11,647	\$680.35	4.6%	\$141.95	1.2%	\$0.7433	\$0.5753	\$0.7773	\$0.5824	
24	25,000	\$5,024	\$13,368	\$18,391	\$2,806	\$11,385	\$14,191	\$5,185	\$14,050	\$19,235	\$2,977	\$11,385	\$14,362	\$843.56	4.6%	\$170.56	1.2%	\$0.7357	\$0.5677	\$0.7694	\$0.5745	
25	30,000	\$5,876	\$16,041	\$21,917	\$3,215	\$13,662	\$16,877	\$6,064	\$16,860	\$22,924	\$3,415	\$13,662	\$17,077	\$1,006.78	4.6%	\$199.18	1.2%	\$0.7306	\$0.5626	\$0.7641	\$0.5692	
26	35,000	\$6,729	\$18,715	\$25,443	\$3,624	\$15,939	\$19,563	\$6,943	\$19,670	\$26,613	\$3,852	\$15,939	\$19,791	\$1,170.00	4.6%	\$227.79	1.2%	\$0.7270	\$0.5590	\$0.7604	\$0.5655	
27	40,000	\$7,581	\$21,388	\$28,969	\$4,033	\$18,216	\$22,249	\$7,822	\$22,480	\$30,303	\$4,290	\$18,216		\$1,333.21	4.6%	\$256.41	1.2%	\$0.7242	\$0.5562	\$0.7576	\$0.5626	
28	45,000	\$8,434	\$24,062	\$32,495	\$4,442	\$20,493	\$24,935	\$8,701	\$25,291	\$33,992	\$4,727	\$20,493	\$25,220	\$1,496.43	4.6%	\$285.03	1.1%	\$0.7221	\$0.5541	\$0.7554	\$0.5605	
29	50,000	\$9,286	\$26,735	\$36,021	\$4,851	\$22,770	\$27,621	\$9,580	\$28,101	\$37,681	\$5,165	\$22,770	\$27,935	\$1,659.65	4.6%	\$313.64	1.1%	\$0.7204	\$0.5524	\$0.7536	\$0.5587	
30	55,000	\$10,139	\$29,409	\$39,547	\$5,260	\$25,047	\$30,307	\$10,460	\$30,911	\$41,370	\$5,603	\$25,047	-	\$1,822.86	4.6%	\$342.26	1.1%	\$0.7190	\$0.5510	\$0.7522	\$0.5573	
31	60,000	\$10,991	\$32,082	\$43,073	\$5,669	\$27,324	\$32,993	\$11,339	\$33,721	\$45,059	\$6,040	\$27,324	\$33,364	\$1,986.08	4.6%	\$370.88	1.1%	\$0.7179	\$0.5499	\$0.7510	\$0.5561	
32	75,000	\$13,549	\$40,103	\$53,651	\$6,896	\$34,155	\$41,051	\$13,976		\$56,127	\$7,353	\$34,155	-	\$2,475.73	4.6%			\$0.7154		\$0.7484	\$0.5534	
33	100,000	\$17,811	\$53,470	\$71,281	\$8,941	\$45,540	\$54,481	\$18,372	\$56,201	\$74,573	\$9,541	\$45,540	\$55,081	\$3,291.81	4.6%			\$0.7128		\$0.7457	\$0.5508	
34	150,000	\$26,336	\$80,205		\$13,031	\$68,310	\$81,341	\$27,164		\$111,465	\$13,917	\$68,310	\$82,227	\$4,923.97	4.6%	\$885.97	1.1%	\$0.7103	\$0.5423	\$0.7431	\$0.5482	
35	200,000	\$34,861	\$106,940		\$17,121	\$91,080	\$108,201	\$35,955		\$148,358	\$18,294	\$91,080		\$6,556.14	4.6%			\$0.7090	\$0.5410	\$0.7418	\$0.5469	
36	, i	· ·									•											
37 Estimated Bill P	ercentiles																					
38 Winter - 25%	10,000	\$2,466	\$5,347	\$7,813				\$2,547	\$5,620	\$8,167				\$353.91	4.5%			\$0.7813		\$0.8167		
39 Winter - 50%	15,000	\$3,319	\$8,021	\$11,339				\$3,426	\$8,430	\$11,857				\$517.13	4.6%			\$0.7560		\$0.7904		
40 Winter - 75%	30,000	\$5,876	\$16,041	\$21,917				\$6,064	\$16,860	\$22,924				\$1,006.78	4.6%			\$0.7306		\$0.7641		
41 Summer - 25%	5,000	ψο,σ. σ	Ţ.O,O I I	7= 1,017	\$1,170	\$2,277	\$3,447	Ψ0,001	\$ 10,000	¥==,0=	\$1,226	\$2,277	\$3,503	÷ 1,000.70	1.070	\$56.10	16%	50.7 550	\$0.6895	Ţ0., 0 i i	\$0.7007	
42 Summer - 50%	15,000				\$1,988	\$6,831	\$8,819				\$2,102	\$6,831	\$8,933			\$113.33			\$0.5880		\$0.5955	
43 Summer - 75%	20,000				\$2,397	\$9,108	\$11,505				\$2,539	\$9,108	\$11,647			\$141.95			\$0.5753		\$0.5824	
-10 Cannilla 1070	20,000				Ψ=,001	ψυ, 100	ψ11,000				ΨΞ,000	ψυ, 100	ψ11,041			Ψ1-1.33	1.2/0		ψυ.υτ υυ		ψ0.0024	

Estimated Bill Percentiles per 2010 MCS

Line 1 2

3 4 5

Docket No. DG 20-105 Appendix 8 Page 10 of 10

Liberty Utilities (EnergyNorth Natural Gas) Corp.

COMPARATIVE MONTHLY BILLING UNDER PRESENT AND PROPOSED RATES RATE G-54 : COMMERCIAL/INDUSTRIAL - HIGH ANNUAL USE, LOAD FACTOR GREATER THAN 90%

Present Rates	Winter	Summer
Cost of Gas	\$0.4792	\$0.3999
LDAC	\$0.0555	\$0.0555
Customer charge Sales rate	\$761.39	
First Block Size		
Block 1	\$0.0650	\$0.0353
Block 2	\$0.0650	\$0.0353

Proposed Rates	Winter	Summer
Cost of Gas	\$0.5065	\$0.3999
LDAC	\$0.0555	\$0.0555
Customer charge Sales rate	\$788.87	
First Block Size		
Block 1	\$0.0670	\$0.0374
Block 2	\$0.0670	\$0.0374

11	<u> </u>																					
12				Mor	nthly Bills a	t Present R	lates			Montl	nly Bills at I	Proposed R	lates		Char	nge in M	Ionthly Bill			Unit	Costs	
13		Use per		Winter			Summer			Winter Su			Summer	mer Winter			nter Summer			Current		osed
		Month	Base	COG/		Base	COG /		Base	COG/		Base	COG/									
14		(therms)	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	Rates	LDAC	TOTAL	\$	%	\$	%	Winter	Summer	Winter	Summer
15																						
16		1,000	\$826	\$535	\$1,361	\$797	\$455	\$1,252	\$856	\$562	\$1,418	\$826	\$455	\$1,282	\$56.82	4.2%	\$29.62	2.4%	\$1.3611	\$1.2521	\$1.4179	\$1.2817
17		2,500	\$924	\$1,337	\$2,261	\$850	\$1,139	\$1,988	\$956	\$1,405	\$2,361	\$882	\$1,139	\$2,021	\$100.83		\$32.82		-	\$0.7953	\$0.9446	\$0.8084
18		5,000	\$1,086	\$2,674	\$3,760	\$938	\$2,277	\$3,215	\$1,124	\$2,810	\$3,934	\$976	\$2,277	\$3,253	\$174.19		\$38.16		-	\$0.6430	\$0.7868	\$0.6506
19		7,500	\$1,249	\$4,010	\$5,259	\$1,026	\$3,416	\$4,442	\$1,292	\$4,215	\$5,507	\$1,070	\$3,416	\$4,485	\$247.54	4.7%	\$43.50		\$0.7012		\$0.7342	\$0.5980
20		10,000	\$1,411	\$5,347	\$6,758	\$1,114	\$4,554	\$5,668	\$1,459	\$5,620	\$7,079	\$1,163	\$4,554	\$5,717	\$320.89	4.7%	-		\$0.6758	-	\$0.7079	\$0.5717
21		12,500	\$1,574	\$6,684	\$8,258	\$1,203	\$5,693	\$6,895	\$1,627	\$7,025	\$8,652	\$1,257	\$5,693	\$6,949	\$394.24	4.8%	-		\$0.6606		\$0.6922	\$0.5559
22		15,000	\$1,736	\$8,021	\$9,757	\$1,291	\$6,831	\$8,122	\$1,794	\$8,430	\$10,224	\$1,350	\$6,831	\$8,181	\$467.59	4.8%	-		-	\$0.5415	\$0.6816	\$0.5454
23		20,000	\$2,061	\$10,694	\$12,755	\$1,467	\$9,108	\$10,575	\$2,129	\$11,240	\$13,370	\$1,538	\$9,108	\$10,646	\$614.29	4.8%	-		\$0.6378		\$0.6685	\$0.5323
24		25,000	\$2,386	\$13,368	\$15,754	\$1,644	\$11,385	\$13,029	\$2,465	\$14,050	\$16,515	\$1,725	\$11,385	\$13,110	\$761.00		-		\$0.6302	-	\$0.6606	\$0.5244
25		30,000	\$2,711	\$16,041	\$18,752	\$1,820	\$13,662	\$15,482	\$2,800	\$16,860	\$19,660	\$1,912	\$13,662	\$15,574	\$907.70	4.8%	\$91.56	0.6%	\$0.6251	\$0.5161	\$0.6553	\$0.5191
26		35,000	\$3,036	\$18,715	\$21,751	\$1,997	\$15,939	\$17,936	\$3,135	\$19,670	\$22,805	\$2,099	\$15,939	\$18,038	\$1,054.40	4.8%			\$0.6215		\$0.6516	\$0.5154
27		40,000	\$3,361	\$21,388	\$24,749	\$2,173	\$18,216	\$20,389	\$3,470	\$22,480	\$25,950	\$2,286	\$18,216	\$20,502	\$1,201.11	4.9%	\$112.92	0.6%	\$0.6187	\$0.5097	\$0.6488	\$0.5126
28		45,000	\$3,686	\$24,062	\$27,748	\$2,350	\$20,493	\$22,843	\$3,805	\$25,291	\$29,096	\$2,473	\$20,493	\$22,966	. ,	4.9%	\$123.60	0.5%	\$0.6166	\$0.5076	\$0.6466	\$0.5104
29		50,000	\$4,011	\$26,735	\$30,746	\$2,526	\$22,770	\$25,296	\$4,140	\$28,101	\$32,241	\$2,661	\$22,770	\$25,431		4.9%	\$134.28	0.5%	\$0.6149	\$0.5059	\$0.6448	\$0.5086
30		55,000	\$4,336	\$29,409	\$33,745	\$2,703	\$25,047	\$27,750	\$4,475	\$30,911	\$35,386	\$2,848	\$25,047	\$27,895	\$1,641.21	4.9%	\$144.96	0.5%	\$0.6135	\$0.5045	\$0.6434	\$0.5072
31		60,000	\$4,661	\$32,082	\$36,743	\$2,879	\$27,324	\$30,203	\$4,811	\$33,721	\$38,531	\$3,035	\$27,324	\$30,359	\$1,787.92	4.9%	\$155.64	0.5%	\$0.6124	\$0.5034	\$0.6422	\$0.5060
32		75,000	\$5,636	\$40,103	\$45,739	\$3,409	\$34,155	\$37,564	\$5,816	\$42,151	\$47,967	\$3,597	\$34,155	\$37,752	\$2,228.03	4.9%	\$187.68	0.5%	\$0.6099	\$0.5009	\$0.6396	\$0.5034
33		100,000	\$7,261	\$53,470	\$60,731	\$4,291	\$45,540	\$49,831	\$7,492	\$56,201	\$63,693	\$4,532	\$45,540	\$50,072	\$2,961.54	4.9%	\$241.08	0.5%	\$0.6073	\$0.4983	\$0.6369	\$0.5007
34		150,000	\$10,511	\$80,205	\$90,716	\$6,056	\$68,310	\$74,366	\$10,843		\$95,145	\$6,404	\$68,310	. ,	\$4,428.57	4.9%			-	\$0.4958	\$0.6343	\$0.4981
35		200,000	\$13,761	\$106,940	\$120,701	\$7,821	\$91,080	\$98,901	\$14,195	\$112,402	\$126,597	\$8,276	\$91,080	\$99,356	\$5,895.60	4.9%	\$454.67	0.5%	\$0.6035	\$0.4945	\$0.6330	\$0.4968
36		' <u>-</u>																				
37 Estim	nated Bill Pe	rcentiles																				
38 Winter	er - 25%	4,000	\$1,021	\$2,139	\$3,160				\$1,057	\$2,248	\$3,305				\$144.85	4.6%			\$0.7900		\$0.8263	
39 Winter	r - 50%	30,000	\$2,711	\$16,041	\$18,752				\$2,800	\$16,860	\$19,660				\$907.70	4.8%			\$0.6251		\$0.6553	
40 Winter	er - 75%	100,000	\$7,261	\$53,470	\$60,731				\$7,492	\$56,201	\$63,693				\$2,961.54	4.9%			\$0.6073		\$0.6369	
41 Summ	ner - 25%	15,000				\$1,291	\$6,831	\$8,122				\$1,350	\$6,831	\$8,181			\$59.52	0.7%		\$0.5415		\$0.5454
42 Summ	ner - 50%	50,000				\$2,526	\$22,770	\$25,296				\$2,661	\$22,770	\$25,431			\$134.28	0.5%		\$0.5059		\$0.5086
43 Summ	ner - 75%	80,000				\$3,585	\$36,432	\$40,017				\$3,784	\$36,432	\$40,216			\$198.36	0.5%		\$0.5002		\$0.5027

Estimated Bill Percentiles per 2010 MCS

Line

2

5

Docket No. DG 20-105 Appendix 9 Page 1 of 1

Liberty Utilities (EnergyNorth Natural Gas) Corp. Docket No. DG 20-105 Rate Case Expense As of June 28, 2021

	Sarvica Dravidar Evnand			Estimated Additional	To	otal Estimated	
Service Provider	Exp	ense to Date		Expenses		Expense	Description of Service
FTI Consulting	\$	385,965.46	\$	-	\$	385,965.46	Revenue Requirement, Rate Design, Marginal Cost of Service , Functional Cost of Service, Decoupling Effects on EE, Cost of Capital
Keegan Werlin		114,463.50		70,536.50		185,000.00	Legal Services
Management Applications Consulting		33,245.63				33,245.63	Review Status of Depreciation Reserve
Concentric Energy Advisors		48,381.75		-		48,381.75	Review of Decoupling Mechanism
ScottMadden		27,060.00		7,500.00		34,560.00	Testimony Support
Legal Notices		466.50		-		466.50	
Court Reporter		721.00		5,400.00		6,121.00	
Customer Notice		46,241.00		-		46,241.00	
Miscellaneous		159.60		-		159.60	Printing Expenses
Subtotal		656,704.44		83,436.50		740,140.94	
Staff Consultants							
Blue Ridge Consulting		62,402.50		7,597.50			Revenue Requirement
J. Randall Woolridge		-		33,800.00		33,800.00	Cost of Capital
OCA Consultants							
Exeter Associates		12,923.70		-		12,923.70	_Cost of Service/Rate Design
Subtotal PUC/OCA		75,326.20		41,397.50		116,723.70	_
Grand Total	\$	732,030.64	\$	124,834.00	\$	856,864.64	Total Estimated Amount

^(*) Additional costs are expected but not currently estimated

Docket No. DG 20-105 Appendix 10 Page 1 of 2

Liberty Utilities (EnergyNorth Natural Gas) Corp. Temporary/Permanent Rate Recoupment

I. Monthly Distribution Revenue at Temporary Rates

		2020					2021				
Rate Schedule	October	November	December	January	February	March	April	May	June	July	TOTAL
R-1	\$ 76,616 \$	84,711 \$	95,454	\$ 99,926 \$	88,858 \$	91,687 \$	79,080 \$	75,196 \$	68,477 \$	68,326	\$ 828,332
R-3	2,947,945	4,783,958	6,536,546	7,439,623	6,451,758	5,673,831	3,752,249	2,641,484	1,852,612	1,759,630	43,839,636
R-4	195,818	325,805	445,804	508,422	441,683	396,049	277,730	195,259	135,904	126,589	3,049,061
R-5	2,187	2,562	3,100	3,416	3,180	3,054	2,275	2,072	1,810	1,829	25,484
R-6	15,009	22,662	30,903	35,584	33,310	27,963	19,925	14,016	9,367	9,118	217,858
R-7	261	413	546	621	572	482	326	234	161	157	3,771
G-41	932,318	1,478,898	2,024,612	2,282,210	2,016,717	1,786,547	1,187,441	851,412	636,112	624,777	13,821,043
G-42	880,886	1,514,648	2,059,582	2,313,781	2,019,359	1,805,610	1,144,826	713,276	444,888	407,508	13,304,364
G-43	135,714	382,676	518,929	591,563	513,601	458,625	302,913	116,760	77,990	77,286	3,176,058
G-44	1,041	1,975	3,150	3,761	3,872	3,181	1,058	589	422	421	19,470
G-45	7,371	12,584	16,964	20,438	19,365	16,086	10,014	5,253	2,511	2,196	112,781
G-46	8,656	9,618	10,580	-	962	1,924	2,885	3,847	4,809	5,771	49,051
G-51	140,954	145,796	167,241	173,638	155,900	158,844	130,708	125,151	118,757	120,311	1,437,299
G-52	164,844	223,727	263,583	275,953	246,644	240,664	191,729	141,647	137,763	141,639	2,028,194
G-53	93,319	176,732	203,468	229,170	200,041	194,987	172,159	82,322	75,155	76,556	1,503,908
G-54	77,796	118,770	102,434	103,564	84,186	94,188	107,939	67,254	69,095	75,890	901,117
G-55	313	336	373	500	482	506	310	281	267	295	3,663
G-56	-	-	-	-	-	-	-	-	-	-	-
G-57	-	-	-	-	-	-	-	-	-	-	-
G-58	2,038	2,612	2,876	2,738	2,733	2,941	2,979	1,914	1,796	1,710	24,337
TOTAL	\$ 5,683,085 \$	9,288,482 \$	12,486,145	\$ 14,084,908 \$	12,283,222 \$	10,957,170 \$	7,386,544 \$	5,037,968 \$	3,637,896 \$	3,500,008	\$ 84,345,428

II. Monthly Distribution Revenue at Permanent Rates

		2020					2021				
Rate Schedule	October	November	December	January	February	March	April	May	June	July	TOTAL
R-1	\$ 76,152 \$	84,218	94,911	\$ 99,363 \$	88,357 \$	91,160 \$	78,611 \$	74,739 \$	68,053 \$	67,897	\$ 823,461
R-3	2,925,874	4,747,157	6,485,782	7,381,640	6,401,529	5,629,969	3,723,686	2,621,852	1,839,240	1,747,050	43,503,779
R-4	194,354	323,302	442,346	504,464	438,248	392,989	275,616	193,806	134,922	125,683	3,025,730
R-5	2,174	2,547	3,082	3,397	3,163	3,038	2,261	2,060	1,799	1,817	25,338
R-6	14,897	22,489	30,664	35,308	33,051	27,748	19,773	13,913	9,300	9,054	216,197
R-7	259	410	542	616	567	478	323	232	160	156	3,743
G-41	926,631	1,470,566	2,013,590	2,269,946	2,005,850	1,776,668	1,180,511	846,105	631,908	620,598	13,742,374
G-42	875,717	1,506,057	2,048,035	2,300,871	2,008,078	1,795,435	1,138,223	709,012	442,093	404,904	13,228,426
G-43	134,949	380,623	516,168	588,411	510,863	456,165	301,267	116,087	77,518	76,813	3,158,864
G-44	1,034	1,964	3,132	3,741	3,850	3,163	1,052	586	419	419	19,360
G-45	7,329	12,514	16,869	20,324	19,258	15,996	9,957	5,222	2,495	2,182	112,146
G-46	8,596	9,551	10,506	-	955	1,910	2,865	3,820	4,776	5,731	48,710
G-51	139,956	144,763	166,054	172,404	154,792	157,717	129,784	124,268	117,919	119,463	1,427,121
G-52	163,938	222,578	262,253	274,568	245,405	239,434	190,718	140,842	136,982	140,835	2,017,554
G-53	92,864	175,914	202,534	228,127	199,129	194,089	171,361	81,913	74,777	76,171	1,496,880
G-54	77,482	118,202	101,935	103,062	83,776	93,729	107,417	66,971	68,810	75,582	896,967
G-55	311	333	371	496	479	502	308	279	265	292	3,637
G-56	-	-	-	-	-	-	-	-	-	-	-
G-57	-	-	-	-	-	-	-	-	-	-	-
G-58	2,028	2,598	2,861	2,723	2,719	2,926	2,963	1,905	1,787	1,701	24,210
TOTAL	\$ 5,644,544 \$	9,225,786	\$ 12,401,637	\$ 13,989,464 \$	12,200,069 \$	10,883,116 \$	7,336,699 \$	5,003,612 \$	3,613,221 \$	3,476,347	\$ 83,774,495

Liberty Utilities (EnergyNorth Natural Gas) Corp. Temporary/Permanent Rate Recoupment

Docket No. DG 20-105
III. Recoupment
Appendix 10

											Page 2 of 2	
		2020					2021					
Rate Schedule	October	November	December	January	February	March	April	May	June	July	TOTAL	
R-1	\$ (464) \$	(494) \$	(543) \$	(562) \$	(502) \$	(527) \$	(469) \$	(457) \$	(424) \$	(429) \$	(4,871)	
R-3	(22,071)	(36,801)	(50,764)	(57,983)	(50,228)	(43,862)	(28,563)	(19,632)	(13,372)	(12,580)	(335,857)	
R-4	(1,464)	(2,502)	(3,458)	(3,958)	(3,435)	(3,059)	(2,114)	(1,452)	(982)	(906)	(23,331)	
R-5	(13)	(15)	(17)	(18)	(17)	(17)	(13)	(13)	(12)	(12)	(147)	
R-6	(112)	(173)	(239)	(276)	(259)	(215)	(152)	(104)	(67)	(65)	(1,661)	
R-7	(2)	(3)	(4)	(5)	(4)	(4)	(2)	(2)	(1)	(1)	(28)	
G-41	(5,686)	(8,331)	(11,023)	(12,264)	(10,866)	(9,879)	(6,930)	(5,307)	(4,204)	(4,179)	(78,669)	
G-42	(5,169)	(8,590)	(11,547)	(12,910)	(11,281)	(10,175)	(6,603)	(4,264)	(2,795)	(2,603)	(75,937)	
G-43	(765)	(2,053)	(2,761)	(3,152)	(2,738)	(2,460)	(1,646)	(673)	(472)	(472)	(17,194)	
G-44	(7)	(11)	(18)	(21)	(21)	(18)	(6)	(4)	(3)	(3)	(110)	
G-45	(42)	(70)	(94)	(113)	(107)	(90)	(57)	(31)	(16)	(14)	(636)	
G-46	(60)	(67)	(73)	-	(7)	(13)	(20)	(27)	(33)	(40)	(341)	
G-51	(997)	(1,033)	(1,187)	(1,234)	(1,108)	(1,127)	(924)	(883)	(837)	(848)	(10,178)	
G-52	(906)	(1,149)	(1,330)	(1,385)	(1,240)	(1,230)	(1,011)	(805)	(781)	(804)	(10,640)	
G-53	(455)	(818)	(934)	(1,042)	(912)	(898)	(797)	(409)	(378)	(386)	(7,029)	
G-54	(314)	(568)	(499)	(502)	(410)	(459)	(521)	(283)	(286)	(308)	(4,150)	
G-55	(2)	(2)	(3)	(4)	(3)	(4)	(2)	(2)	(2)	(2)	(26)	
G-56	-	-	-	-	-	-	-	-	-	-	-	
G-57	-	-	-	-	-	-	-	-	-	-	-	
G-58	(10)	(14)	(15)	(15)	(14)	(15)	(15)	(10)	(9)	(9)	(127)	
TOTAL	\$ (38,541) \$	(62,696) \$	(84,508) \$	(95,444) \$	(83,152) \$	(74,054) \$	(49,845) \$	(34,356) \$	(24,675) \$	(23,662) \$	(570,933)	

NOTES: Revenues calculated using actual calendar month billing determinants from October 2020 through March 2021 and calculated using estimated monthly billing determinants from April 2021 through July 2021.

Docket No. DG 21-130 Exhibit 26

Page 50 of 470 Docket No. DG 20-105

Exhibit 49

NHPUC NO. 11 - GAS

LIBERTY UTILITIES (ENERGYNORTH NATURAL GAS) CORP. D/B/A

LIBERTY

SUPERSEDING NHPUC No. 10

TARIFF

FOR

GAS SERVICE

Applicable

in

Thirty-five towns in New Hampshire served in whole or in part.

(For detailed description, see Service Area)

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman Neil Proudman

EFFECTIVE: August 1, 2021 TITLE:

President

Original Revised Page i

CHECK SH	HEET	
I. GENERA	AL TERMS AND CONDITIONS	
1	SERVICE AREA	
2	GENERAL TERMS AND CONDITIONS	S
3	CHARACTER OF SERVICE-EXCLUDIN	NG KEENE CUSTOMERS
4	CHARACTER OF SERVICE KEENE CUS	STOMERS
5	CUSTOMER'S INSTALLATION	
6	APPLICATION FOR SERVICE	
7	CREDIT	
8	SERVICE AND MAIN EXTENSIONS	
9	INTRODUCTION OF SERVICE	1
10	COMPANY EQUIPMENT ON CUSTOM	IER'S PREMISES1
11	SERVICE CONTINUITY	1
12	CUSTOMER'S USE OF SERVICE	1
13	INSPECTIONS	1
14	MEASUREMENT	1
15	METER TESTS	1
16	DISCONNECTION BY THE COMPANY	7 1
17	COST OF GAS CLAUSE - EXCLUDING	6 KEENE 1
18	COST OF GAS CLAUSE - KEENE DIVI	ISION 3
19		T CLAUSE AND NORMAL WEATHER
20	SUPPLY & CAPACITY SHORTAGE AL	LLOCATION POLICY4
II. RATE S	SCHEDULES	4
1	RESIDENTIAL NON-HEATING RATE:	CLASSIFICATION NO. R-1 4
2	RESIDENTIAL HEATING RATE: CLAS	SSIFICATION NO. R-34
3	LOW INCOME RESIDENTIAL HEATIN	NG RATE: CLASSIFICATION NO. R-4 5
4	MANAGED EXPANSION PROGRAM R CLASSIFICATION NO. R-5	RESIDENTIAL NON-HEATING RATE:5
5	MANAGED EXPANSION PROGRAM R CLASSIFICATION NO. R-6	RESIDENTIAL HEATING RATE:5
6		OW INCOME RESIDENTIAL HEATING5
7		E: LOW ANNUAL USE, HIGH WINTER US
8		E: MEDIUM ANNUAL USE, HIGH WINTER 26
DATED:	XXX XX, 2021	ISSUED BY: /s/Neil Proudman
EFFECTIVE:	August 1, 2021	Neil Proudman TITLE: President

Original Page ii Table of Contents

9	COMMERCIAL/INDUSTRIAL SERVICE: HIGH ANNUAL USE, HIGH WINTER USE RATE CLASSIFICATION NO. G-43
10	MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: LOW ANNUAL USE, HIGH WINTER USE RATE CLASSIFICATION NO. G-44 65
11	MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: MEDIUM ANNUAL USE, HIGH WINTER USE RATE CLASSIFICATION NO. G-45
12	MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: HIGH ANNUAL USE, HIGH WINTER USE RATE CLASSIFICATION NO. G-46 69
13	COMMERCIAL/INDUSTRIAL SERVICE: LOW ANNUAL USE, LOW WINTER USE RATE CLASSIFICATION NO. G-51
14	COMMERCIAL/INDUSTRIAL SERVICE: MEDIUM ANNUAL USE, LOW WINTER USE RATE CLASSIFICATION NO. G-5273
15	COMMERCIAL/INDUSTRIAL SERVICE: HIGH ANNUAL USE, LOAD FACTOR LESS THAN 90% RATE CLASSIFICATION NO. G-5375
16	COMMERCIAL/INDUSTRIAL SERVICE: HIGH ANNUAL USE, LOAD FACTOR GREATER THAN 90% RATE CLASSIFICATION NO. G-547
17	MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: LOW ANNUAL USE, LOW WINTER USE RATE CLASSIFICATION NO. G-55 79
18	MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: MEDIUM ANNUAL USE, LOW WINTER USE RATE CLASSIFICATION NO. G-56
19	MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: HIGH ANNUAL USE, LOAD FACTOR LESS THAN 90% RATE CLASSIFICATION NO. G-57
20	MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: HIGH ANNUAL USE, LOAD FACTOR GREATER THAN 90% RATE CLASSIFICATION NO. G-5885
21	OUTDOOR GAS LIGHTING80
22	FIRM RATE SCHEDULES - EXCLUDING KEENE CUSTOMERS 8'
23	FIRM RATE SCHEDULES - KEENE CUSTOMERS
24	FIRM RATE SCHEDULES - MANAGED EXPANSION PROGRAM - EXCLUDING KEENE CUSTOMERS89
25	FIRM RATE SCHEDULES - OUTDOOR GAS LIGHTING90
26	ANITCIPATED SUMMER PERIOD COST OF GAS - EXCLUDING KEENE CUSTOMERS9
27	CALCULATION OF FIRM SALES SUMMER PERIOD COST OF GAS RATE - EXCLUDING KEENE CUSTOMERS92
28	CALCULATION OF SUMMER PERIOD COST OF GAS RATE - KEENE CUSTOMERS93

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman Neil Proudman

EFFECTIVE: August 1, 2021 TITLE:

President

29	CALCULATION OF FIXED WINTER PERIOD COST OF GAS RATE EXCLUDING KEENE CUSTOMERS9		
30	CALCULATION OF FIRM SALES WINTER PERIOD COST OF GAS RATE EXCLUDING KEENE CUSTOMERS		
31	ANTICIPATED WINTER PERIOD COST OF GAS - EXCLUDING KEENE CUS OR GAS LIGHTING		
32	CALCULATION OF FIRM SALES AND FIXED WINTER PERIOD COST O RATE KEENE CUSTOMERS		
33	CALCULATION OF FIRM TRANSPORTATION COST OF GAS RATE	98	
34	ENVIRONMENTAL SURCHARGE - MANUFACTURED GAS PLANTS	99	
35	RATE CASE EXPENSE AND RECOUPMENT FACTOR CALCULATION	100	
36	LOCAL DISTRIBUTION ADJUSTMENT CLAUSE CALCULATION	101	
37	RESIDENTIAL ENERGY EFFICIENCY LOAN PROGRAM	102	
38	NON-RESIDENTIAL ENERGY EFFICIENCY LOAN PROGRAM	103	
III. D	DELIVERY TERMS AND CONDITIONS	104	
1	RATES AND CHARGES	104	
2	DEFINITIONS	104	
3	CHARACTER OF SERVICE	108	
4	GAS SERVICE AREAS AND DESIGNATED RECEIPT POINTS	108	
5	CUSTOMER REQUEST FOR SERVICE FROM COMPANY	109	
6 QUALITY AND CONDITION OF GAS		109	
7	POSSESSION OF GAS	110	
8	COMPANY GAS ALLOWANCE	110	
9	DAILY METERED DELIVERY SERVICE	110	
10	NON-DAILY METERED DELIVERY SERVICE	114	
11	CAPACITY ASSIGNMENT	119	
12	BILLING AND SECURITY DEPOSITS	125	
13	SALES SERVICE	126	
14	PEAKING SERVICE	127	
15	DISCONTINUANCE OF SERVICE	129	
16	OPERATIONAL FLOW ORDERS AND CRITICAL DAYS	130	
17	FORCE MAJEURE AND LIMITATION OF LIABILITY	130	
18	CURTAILMENT	131	
19	TAXES	132	
20	SUPPLIER TERMS AND CONDITIONS	132	
21	CUSTOMER DESIGNATED REPRESENTATIVE	138	
DATED:	XXX XX, 2021 ISSUED BY: /s/Neil Proudmar Neil Proudmar		
EFFECTIVE:	August 1, 2021 TITLE: President	1	

Page 54 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page iv Table of Contents

IV.	ATTACHMENTS	139
1	ATTACHMENT A Supplier Service Agreement	139
2	ATTACHMENT B Schedule of Administrative Fees and Charges	153
3	ATTACHMENT C Capacity Allocators	154

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Docket No. DG 21-130 Exhibit 26 Page 55 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 1 Check Sheet

<u>Page</u>	Revision
Title	Original
TOC i	Original
TOC ii	Original
TOC iii	Original
TOC iv	Original
1	Original
2	Original
3	Original
4	Original
5	Original
6	Original
7	Original
8	Original
9	Original
10	Original
11	Original
12	Original
13	Original
14	Original
15	Original
16	Original
17	Original
18	Original
19	Original
20	Original
21	Original
22	Original
23	Original
24	Original
25	Original
26	Original

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 56 of 470 Docket No. DG 20-105 Exhibit 49

Original Page 2 Check Sheet

NHPUC NO. 11 GAS LIBERTY UTILITIES

<u>Page</u>	Revision
27	Original
28	Original
29	Original
30	Original
31	Original
32	Original
33	Original
34	Original
35	Original
36	Original
37	Original
38	Original
39	Original
40	Original
41	Original
42	Original
43	Original
44	Original
45	Original
46	Original
47	Original
48	Original
49	Original
50	Original
51	Original
52	Original
53	Original
54	Original
55	Original
57	Original

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 57 of 470 Docket No. DG 20-105 Exhibit 49

> Original Page 3 Check Sheet

NHPUC NO. 11 GAS LIBERTY UTILITIES

Page Revision 58 Original 59 Original 60 Original 61 Original 62 Original 63 Original 64 Original 65 Original 66 Original 67 Original 68 Original 69 Original 70 Original 71 Original 72 Original 73 Original 74 Original 75 Original 76 Original 77 Original 78 Original 79 Original 80 Original 81 Original 82 Original 83 Original 84 Original 85 Original 86 Original 87 Original 88 Original

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Original

89

Docket No. DG 21-130 Exhibit 26 Page 58 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 4 Check Sheet

<u>Page</u>	Revision
90	Original
91	Original
92	Original
93	Original
94	Original
95	Original
96	Original
97	Original
98	Original
99	Original
100	Original
101	Original
102	Original
103	Original
104	Original
105	Original
106	Original
107	Original
108	Original
109	Original
110	Original
111	Original
112	Original
113	Original
114	Original
115	Original
116	Original
117	Original
118	Original
119	Original
120	Original
121	Original
122	Original
123	Original

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman President

EFFECTIVE: August 1, 2021 TITLE: President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Docket No. DG 21-130 Exhibit 26 Page 59 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 5 Check Sheet

<u>Page</u>	Revision
124	Original
125	Original
126	Original
127	Original
128	Original
129	Original
130	Original
131	Original
132	Original
133	Original
134	Original
135	Original
136	Original
137	Original
138	Original
139	Original
140	Original
141	Original
142	Original
143	Original
144	Original
145	Original
146	Original
147	Original
148	Original
149	Original
150	Original
151	Original
152	Original
153	Original
154	Original

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u> Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 60 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 6

I. GENERAL TERMS AND CONDITIONS

1 SERVICE AREA

A. <u>Service Area</u>. The area authorized to be served by the Company and to which this tariff applies are the following cities and towns: Allenstown, Amherst, Auburn, Bedford, Belmont, Berlin, Boscawen, Bow, Concord, Derry, Franklin, Gilford, Goffstown, Hanover, Hollis, Hooksett, Hudson, Keene, Laconia, Lebanon, Litchfield, Londonderry, Loudon, Manchester, Merrimack, Milford, Nashua, Northfield, Pelham, Pembroke, Sanbornton, Tilton, Windham, and part of Canterbury and Winnisquam.

2 GENERAL TERMS AND CONDITIONS

- A. <u>Filing</u>. A copy of this tariff is on file with the New Hampshire Public Utilities Commission ("NHPUC" or the "Commission") and is open to inspection at the offices of the Company.
- B. <u>Revisions</u>. This tariff may be revised, amended, supplemented, or otherwise changed from time to time in accordance with the rules of the Commission and such changes, when effective, shall have the same force as the original tariff.
- C. <u>Application</u>. The tariff provisions apply to everyone lawfully receiving gas supply service and/or delivery-only service from the Company under the rates herein and receipt of gas service shall constitute the receiver a customer of the Company as the term is used herein whether service is based upon contract, agreement, accepted signed application, or otherwise.
- D. <u>Statement by Agents</u>. No representative has the authority to modify a tariff rule or provision or to bind the Company by a promise or representation contrary thereto.
- E. No Prejudice of Rights. The failure of the Company to enforce any of the terms of this tariff shall not be deemed a waiver of its right to do so.
- F. <u>Gratuities to Employees</u>. The Company's employees are strictly forbidden to demand or accept any personal compensation or gifts for service rendered by them while working for the Company on the Company's time.
- G. <u>Advance Payments</u>. Payments to the Company for charges provided in these rules and regulations to be borne by the customer shall be made in advance.
- H. <u>Assignment</u>. Subject to the rules and regulations, all contracts by the Company shall be binding upon, and oblige, and continue for the benefit of, the successors and assigns, heirs, executors, and administrators of the parties hereto.

3 CHARACTER OF SERVICE – EXCLUDING KEENE CUSTOMERS

- A. <u>Gas Supply</u>. This Tariff applies only to the supply of gas, having a thermal content of nominally 1,000 British thermal units per cubic foot at supply pressures available in the locality in which the premises to be served are situated.
- B. <u>Determination of Therms</u>. The gas for any billing period, expressed in hundreds of cubic feet (ccf), shall be multiplied by the average BTU of the gas send out as determined below and divided by 1,000 in order to

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 61 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 7

C. Determine the number of therms consumed in the billing period. For billing purposes, gas therms shall be determined on a "dry" basis.

The BTU therm factor of the gas sendout shall be calculated for each billing cycle from the daily weighted average BTU of the natural gas delivered to the Company by its suppliers and the gas produced at the Company's peak-shaving plants. The daily average BTU content shall be determined by appropriate gas measurement devices operated by the Company or its supplier.

- D. <u>Delivery of Gas Supply</u>. The rates specified in this tariff are based upon the supply of service to a single customer through one delivery and metering point.
- E. <u>Use of Service at Separate Properties</u>. The use of service at two or more separate properties will not be combined for billing purposes.

4 CHARACTER OF SERVICE – KEENE CUSTOMERS

- A. <u>Gas Supply</u>. This Tariff applies to the supply of either propane-air gas or natural gas at the company's standard heat content value for the gas supply source, adjusted for temperature and pressure, in the locality in which the premises to be served are situated.
 - 1. For Keene customers on the propane air-gas distribution system, the supply of gas sold will be nominally 740 British thermal units per cubic foot.
 - 2. For Keene customers on the natural gas distribution system, the supply of gas sold will be nominally 1,000 British thermal units per cubic foot.

B. Determination of Therms.

1. The propane-air gas for any billing period, expressed in hundreds of cubic feet (CCF), shall be multiplied by the average BTU/cubic foot of the gas send out as determined below and divided by 1,000 in order to determine the number of therms consumed in the billing period.

The average BTU of the gas sendout for billing purposes shall be calculated for each billing period using the daily average BTU of propane-air gas produced by the company's production plant as determined by appropriate gas heat content measurement devices¹ operated by the company. The average BTU so calculated shall be used in determining the therms for monthly bills based on meter readings and the average BTU for the billing period.

2. The natural gas for any billing period, expressed in hundreds of cubic feet (CCF), shall be multiplied by the average BTU/cubic foot of the gas send out as determined below and divided by 1,000 in order to determine the number of therms consumed in the billing period.

The average BTU of the gas sendout for billing purposes shall be calculated for each billing period using the daily average BTU of compressed or liquefied natural gas delivered to the company in New Hampshire by its suppliers. To determine the daily average BTU, each tanker's BTU content will be provided by the supplier or measured by the Company with appropriate gas heat content measurement devices,¹

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

¹ Typically chromatographs or calorimeters are used for measuring the heat content of gas.

Docket No. DG 21-130 Exhibit 26 Page 62 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 8 General Terms and Conditions

including when more than one tanker is connected to the Company's natural gas system and simultaneously providing flow at any given time. The average BTU so calculated shall be used in determining the therms for monthly bills based on meter readings and the average BTU for the billing period.

5 CUSTOMER'S INSTALLATION

- A. <u>Point of Delivery</u>. Upon request, the Company will designate a point at which the customer shall terminate his piping for connection to the meter of the Company, but such information does not constitute an agreement or obligation on the part of the Company to furnish service.
- B. <u>Space for Meter</u>. The customer shall provide, free of expense to the Company, a dry, warm and otherwise suitable place for the regulator or regulators, meter or meters, or other equipment of the Company which may be necessary for the fulfillment of such contracts as may be entered into with the Company.
- C. <u>Location of Meter</u>. The space provided for the Company's meters and equipment shall be convenient access to the Company's employees and, as near as possible, to the point where the service supply pipe enters the customer's building. Its location shall be such that the meter connections are not concealed by plaster or sheathing and shall be otherwise acceptable to the Company.
- D. <u>Reverse Flow</u>. The customer may be required to install check valves or other devices to prevent compressed air or other gases from entering the Company's mains.

6 APPLICATION FOR SERVICE

- A. <u>Service Contract</u>. Every applicant for gas service may be required to sign a contract, agreement, or other form then in use by the Company covering the special circumstances of the applicant's use of gas and must agree to abide by the rules and regulations and standard requirements of the Company.
- B. <u>Right to Reject</u>. The Company may reject any application for service which would involve excessive cost to supply, or which might affect the supply of service to other customers, or for other good and sufficient reasons.
- C. <u>Special Contracts</u>. Standard contracts shall be for terms as specified in the statement of the rate, but where large or special investment is necessary for the supply of service, contracts of longer terms as specified in the rate, or with a special guarantee of revenue, or both, may be required to safeguard such investment.
- D. <u>Unauthorized Use</u>. Unauthorized connection to the Company's gas service supply facilities, and/or the use of service obtained from the Company without authority, or by any false pretense, may be terminated by the Company without notice. The use of service without notifying the Company and without enabling the

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 63 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 9 General Terms and Conditions

company to read its meter will render the user liable for any amount due for service supplied to the premises from the time of the last meter reading of the Company's meter immediately preceding the user's occupancy as shown by the Company's books.

- E. <u>Managed Expansion Program</u>. The Managed Expansion Program targets gas expansion in specific areas that have high potential for demand. Each Managed Expansion Program project includes a Main Extension. Customers under this program avoid a portion or all of a contribution in aid of construction which would otherwise be required absent the Managed Expansion Program.
- F. <u>Managed Expansion Program ("MEP") Premium</u>. The markup for all MEP rates above the rates that would otherwise be applied absent the Managed Expansion Program. The MEP Premium is 30%.

7 CREDIT

- A. <u>Prior Debts</u>. Service will not be furnished to former customers until any indebtedness to the Company for previous service has been satisfied.
- B. <u>Deposits</u>. Before rendering or restoring service, the Company may require a deposit subject to the Commission's Rules and Regulations. (See Puc 1200 rules).

8 SERVICE AND MAIN EXTENSIONS

- A. <u>Definitions</u>. The following are definitions of terms used in these provisions relative to main and service extensions and are applicable only in the main and service extensions provisions.
 - 1. <u>Service and Main Extensions</u>. Extensions that require the construction of a new gas main and a service from that new main in order to provide requested gas service to a customer.
 - 2. <u>Service Extensions</u>. Extensions from an Existing Gas Main to the point of delivery on the customer's premises.
 - 3. Main Extension. An extension of the new gas main portion of a Service and Main Extension.
 - 4. Existing Gas Main. A main that is installed in the street and through which gas is flowing.
 - 5. <u>Abnormal Costs</u>. Abnormal Costs are service and/or main construction costs that are attributable to frost or ledge (including ditching or backfilling necessitated as a result of the presence of frost or ledge),and/or other conditions not typically encountered in service and/or main construction that are peculiar to the particular service and/or main construction concerned. Abnormal Costs are to be paid by the customer.
 - 6. Extra Footage. The charge (contribution in aid of construction) for Extra Footage is \$63.90 per foot. The charge will be updated annually by calculating the historical average cost per foot for Service Extensions, excluding overheads, for the most recent calendar year and the updated charge shall be effective April 1.
 - 7. Estimated Annual Margin. The Estimated Annual Margin is equal to the estimated revenue to be derived from the monthly Customer Charge and delivery charge to be received from the customer for gas service utilizing the Service and Main Extension or Service Extension during the first twelve (12) months after completion of the extension. The Estimated Annual Margin does not include revenue received by the Company for the cost of gas and local distribution adjustment factor.

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 64 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 10 General Terms and Conditions

- 8. Estimated Cost of Construction. For the purpose of determining the cost of Service and Main Extensions, Estimated Cost of Construction of mains and/or services includes the cost of labor and materials for such construction, and incidental or associated miscellaneous costs, but excluding overheads. Miscellaneous costs include, but are not limited to, meter(s), traffic control and city and town road permits and degradation fees. The customer may perform on-site trenching and backfilling in accordance with the Company's specifications, in which case the Estimated Cost of Construction will be reduced to reflect the costs avoided by the Company as a result of the customer's performance of the work.
- B. <u>Costs of Extensions</u>. In areas where the Company is authorized to operate, subject to the Application for Service provisions of this tariff, service is available as follows:
 - 1. Residential Service Extensions. Residential Service Extensions up to 100 feet in length will be installed at no charge to customers served under either a (i) residential heating rate; or (ii) a residential non-heating rate provided that such extension is installed during the installation of a Main Extension or during the performance of work on cast iron/bare steel main replacements; unless there are Abnormal Costs associated with such extensions, in which case the customer shall be charged for the Abnormal Costs. For residential Service Extensions in excess of 100 feet, the customer will be charged for the Extra Footage, plus any Abnormal Costs. This Section 7(B)(1) shall apply only to Service Extensions and shall not apply to Service and Main Extensions as described in Section 7(B)(3).
 - 2. Commercial and Industrial Service Extensions. Commercial and industrial Service Extensions will be installed at no charge to the customer provided that the Estimated Annual Margin is at least one-sixth of the Estimated Cost of Construction of the Service Extension, excluding any Abnormal Costs. If the Estimated Annual Margin is less than one-sixth of the Estimated Cost of Construction, the customer will be required to pay to the Company, in advance, any amount by which the Estimated Cost of Construction of the Service Extension exceeds six times the Estimated Annual Margin. Abnormal Costs are charged separately and are not included in the Estimated Cost of Construction for the purpose of this calculation. This Section 7(B)(2) shall apply only to Service Extensions and shall not apply to Service and Main Extensions as described in Section 7(B)(3).
 - 3. Service and Main Extensions of Less Than \$1,000,000. The Company shall not commence construction on a Service and Main Extension for which the Estimated Cost of Construction is less than \$1,000,000 until the sum of (i) six times the Estimated Annual Margin for all commercial and industrial customers who have committed to take service, plus (ii) eight times the Estimated Annual Margin for all residential customers who have committed to take service equals or exceeds 25% of the Estimated Cost of Construction.
 - a. <u>Residential.</u> Residential Service and Main Extensions will be installed at no charge to the customer provided that the Estimated Annual Margin is at least one-eighth of the Estimated Cost of Construction of the Service and Main Extensions. If the Estimated Annual Margin is less than one-eighth of the Estimated Cost of Construction, the customer will be required to pay to the Company the difference between the Estimated Cost of Construction and eight times the Estimated Annual Margin, plus any Abnormal Costs.

If the Main Extension will serve more than one location, the Company will calculate the sum of the Estimated Annual Margin from all metered services and the sum of the Estimated Cost of Construction for the Main Extension and all Service Extensions to

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

EFFECTIVE: August 1, 2021 Neil Proudman
TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 65 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 11 General Terms and Conditions

determine whether any payment will be required from the customers to be served. The Company will also include the Estimated Annual Margin and the Estimated Cost of Construction for Service Extensions for all existing premises for which the Company reasonably anticipates will take service, using the assumption that 60% of such premises will take service. If any payment is required, it will be allocated equally among all current metered services that exist as of the date that the Main Extension becomes an Existing Gas Main. Abnormal Costs associated with Main Extensions will be allocated equally among all customers, unless such costs can be attributed to specific customers.

- b. <u>Commercial and Industrial.</u> Commercial and industrial Service and Main Extensions will be installed at no charge to the customer provided that the Estimated Annual Margin is at least one-sixth of the Estimated Cost of Construction of the Service and Main Extensions. If the Estimated Annual Margin is less than one-sixth of the cost of construction of the Service and Main Extensions, the customer will be required to pay to the Company the difference between the Estimated Cost of Construction and six times the Estimated Annual Margin, plus any Abnormal Costs.
- c. If the Main Extension will serve more than one location, the Company will calculate the sum of the Estimated Annual Margin from all metered services and the sum of the Estimated Cost of Construction for the Main Extension and all Service Extensions to determine whether any payment will be required from the customers to be served. The Company will also include in such calculations the Estimated Annual Margin and the Estimated Cost of Construction for Service Extensions for all existing premises for which the Company reasonably anticipates will take service, using the assumption that 60% of such premises will take service. If any payment is required, it will be allocated among all current metered services that exist as of the date that the Main Extension becomes an Existing Gas Main based on each customer's proportional share of the Estimated Annual Margin. Abnormal Costs associated with Main Extensions will also be allocated based on each customer's proportional share of the Estimated Annual Margin, unless such costs can be attributed to specific customers, in which case the costs shall be allocated appropriately to specific customers.
- d. Extensions Serving Customers in More Than One Rate Class. If the Main Extension will serve both residential and commercial or industrial customers, the Company will determine whether a contribution will be required by the customers by calculating the difference between the Estimated Cost of Construction of the Main and Service Extensions and (i) six times the Estimated Annual Margin for all commercial and industrial customers to be served, plus (ii) eight times the Estimated Annual Margin for all residential customers to be served. The Company will also include in the above calculations the Estimated Annual Margin and the Estimated Cost of Construction of Service Extensions for all existing premises for which the Company reasonably anticipates will take service. If the difference described above is positive, the customers will be required to pay to the Company such difference. The amount of payment will be allocated among all metered services that exist as of the date that the Main Extension becomes an Existing Gas Main based on each customer's proportional share of the Estimated Annual Margin. Abnormal Costs associated with Main Extensions will also be allocated based on each customer's proportional share of the Estimated Annual Margin, unless such costs can be attributed to specific customers, in which case the costs shall be allocated appropriately to specific customers.

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 66 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 12 General Terms and Conditions

4. <u>Service and Main Extensions Greater Than or Equal to \$1,000,000</u>. If the cost of the Main Extension equals or exceeds \$1,000,000, then in addition to the requirements specified in Section 7(B)(3), the Company will not commence construction unless a discounted cash flow analysis

demonstrates a positive net present value over a 10-year period of the difference between the Estimated Annual Margin and the revenue requirement associated with the Estimated Cost of Construction.

- A. <u>Failure to Use Installed Gas Service</u>. If a customer fails, within nine months after the date a service is installed under this Section 7, either in whole or in part, to make use of the service, the customer will reimburse the Company for all costs of constructing, removing and retiring the service less any contribution in aid of construction made by the customer for the service, which will be forfeited.
- B. <u>Easements, Etc.</u> The Company is not required to construct extensions other than in public ways unless the customer provides, in advance and without expense or cost to the Company, all necessary permits, consents, authorizations and right-of-way easements, satisfactory to the Company, for the construction, maintenance and operation of the pipeline.
- C. <u>Shortest Distance</u>. Services are run the shortest practical safe distance to the meter location. However, a customer may have the Company install a longer alternate service provided that the customer pays for the extra expense in advance of installation.
- D. <u>Winter Construction</u>. Ordinarily, no new service pipes or main extensions are installed during the winter conditions (when frost is in the ground) unless the customer defrays the extra expenses.
- E. <u>Timing and Refunding of Contribution</u>. Except as otherwise agreed by the Company under unusual circumstances, any required contribution in aid of construction will be made prior to installation by the Company of a service. To help cover the Company's expenses, damages and lost business, if substantial construction of the building or buildings for which gas service has been sought is not commenced by the earlier of (1) November 30th next following submission of the application; or (2) the date when the Company commences construction of the main and service concerned prior to withdrawal of the application, ten percent (10%) of the contribution will be forfeited to the Company and will not be returned to the customer. The balance of the contribution will be refunded if and when the application is withdrawn, or will be applied toward the new contribution if the customer submits a new application for service or subsequently commences construction of the building or buildings. A new application may be submitted at any time.
- F. <u>Reasonable Duration and Non-Discrimination</u>. Under none of the foregoing provisions will the Company be required to install service pipes or to contract main extensions where the business to be secured may not be of reasonable duration or will tend, in any way, to constitute unreasonable discrimination.
- G. <u>Title</u>. Title of all extensions constructed in accordance with the above shall be vested in the Company.
- H. Other Requirements. The Company generally will not approve any application or, if it shall have given such approval, will not proceed or continue with main and/or service construction unless the Company is satisfied

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 67 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 13 General Terms and Conditions

- 1. That the final site plans, sub-division plans and plans and specification for building or buildings to be served by the main and/or service concerned, including plans for waste disposal, water and other associated systems and facilities, have been prepared and approved by owner;
- 2. That all permits, exceptions, approvals and authorizations of governmental bodies or agencies required for construction of such building or buildings and associated systems and facilities have been obtained;
- 3. That the customer is proceeding or plans promptly to proceed with such construction; and
- 4. That nothing has occurred or failed to occur which will or is likely to prevent or interfere with such construction.

9. INTRODUCTION OF SERVICE

- A. <u>Service Contract</u>. Every applicant for gas service may be required to sign a contract, agreement, or other form then in use by the Company covering the special circumstances of his use of gas and must agree to abide by the rules and regulations and standard requirements of the Company.
- B. <u>Defective Installation</u>. The Company may refuse to connect if, in its judgment, the customer's installation is defective, or does not comply with such reasonable requirements as may be necessary for safety, or is in violation of the Company's standard requirements.
- C. <u>Unsatisfactory Installation</u>. The Company may refuse to connect if, in its judgment, the customer's equipment or use thereof might injuriously affect the equipment of the Company or the Company's service to other customers.

10. COMPANY EQUIPMENT ON CUSTOMER'S PREMISES

- A. <u>Meters and Regulators</u>. The Company shall furnish and install, maintain and own, any meter or meters, regulator or regulators required in the supply of service. For certain large customers, the Company shall furnish, install and maintain, at the customer's expense, any remote meter reading equipment to record usage for daily balancing. Such equipment shall remain the property of the Company at all times.
- B. <u>Customer's Responsibility</u>. The customer shall be responsible for safekeeping of the Company's property while on the customer's premises. In the event of injury or destruction of any such property, the customer shall pay the costs of repairs and replacements.
- C. Relocation and/or Replacement of Company Equipment. The original service connection, including piping, meters and all other necessary or incidental equipment, which remains the property of the Company, shall be installed by the Company at its expense unless otherwise expressly provided in this tariff. Subsequent relocation and/or replacement of any such equipment on private property, whether it be for one or more service connections, shall be performed by the Company at the customer's expense unless such work is done at the request of the Company and for its convenience, in which case the Company shall bear the expense.
- D. <u>Protection by Customer</u>. The customer shall protect the equipment of the Company on his premises and shall not permit any persons, except a Company employee having a Company photo identification card or other Company identification, to break any seals upon or do any work on any meter, service supply pipe, or other equipment of the Company located on the customer's premises.
- E. <u>Tampering</u>. In the event the Company's meter or other property is being tampered with or interfered with, the customer being supplied through such equipment shall pay the amount which the Company may

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman President

Docket No. DG 21-130 Exhibit 26 Page 68 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 14 General Terms and Conditions

estimate is due for service used but not registered on the Company's meter and for any repairs or replacements required as well as for costs of inspections, investigations, and protective installation.

- F. <u>Right of Access.</u> The Company's identified employees shall have access to the premises of the customer at all reasonable times for the purpose of reading meters, testing, repairing, removing, or exchanging any or all equipment belonging to the Company.
- G. Ownership and Removal. All equipment supplied by the Company shall remain its exclusive property and the Company shall have the right to remove the same from the premises of the customer at any time after the termination of service for whatever cause.

11 SERVICE CONTINUITY

- A. Regularity of Supply. The Company will use reasonable diligence to provide a continuous, regular and uninterrupted supply of service, but should the supply be interrupted by the Company for the purpose of making repairs, changes, or improvements in any part of its system for the general good of the service or the safety of the public, or should the supply of service be interrupted or fail by reason of accident, strike, legal process, state or municipal interference, or any cause whatsoever beyond its control, the Company shall not be liable for damages, direct or inconsequential, resulting from such interruption or failure.
- B. <u>Notice of Trouble.</u> The customer shall notify the office of the Company immediately should the service be unsatisfactory for any reason or should there be any defects, leaks, trouble, or accident affecting the supply of gas.

12 CUSTOMER'S USE OF SERVICE

- A. <u>Resale Forbidden</u>. The customer shall not, directly or indirectly, sell, sublet, assign, or otherwise dispose of to others, gas purchased from the Company, or any part thereof, without the consent of the Company. This rule does not apply to a public utility Company purchasing gas in bulk expressly for the purpose of delivering it to others.
- B. <u>Fluctuations</u>. Gas service must not be used in such a manner as to cause unusual fluctuations or disturbances in the Company's supply system. In the case of violation of this rule, the Company may discontinue service or require the customer to modify its installation and/or equip it with approved controlling devices.
- C. <u>Additional Load</u>. The service supply pipe, regulators, meters, and equipment supplied by the Company for each customer have definite capacities. The customer shall notify the Company of substantial changes in service requirements or location of appliances.

13 INSPECTIONS

A. <u>Company's Right to Inspect</u>. The Company shall have the right, but shall not be obliged, to inspect any installation before service is introduced or at any time later and reserves the right to reject any piping or appliances not in accordance with the Company's standard requirements. However, such inspection, failure to inspect, or failure to reject shall not render the Company liable or responsible for any losses or damage resulting from defects in the installation, piping or appliances, from violation of Company rules, or from accidents which may occur upon the premises of the customer.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 69 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 15 General Terms and Conditions

14 MEASUREMENT

- A. <u>Supply of Meters</u>. The measurement of gas service shall be by meters furnished and installed by the Company. The Company will select the type and make of metering equipment and may, from time to time, change or alter the equipment. The Company's sole obligation is to supply meters that will accurately and adequately furnish records for billing purposes.
- B. <u>Special Measurements</u>. The Company shall have the right, at its option and its own expense, to place demand meters, pressure gauges, special meters, or other instruments on the premises of any customer for the purpose of determining the adequacy of the Company's service or for making tests of all or any part of the customer's load.

15 METER TESTS

- A. Meter Tests. Meters are tested according to NHPUC Rules and Regulations. (See Puc 500 rules).
- B. <u>Request Tests</u>. The fee for a special request test is \$20.00 when scheduled at the mutual convenience of the Company and the customer; otherwise the amount is \$30.00. (See Puc 500 rules).
- C. <u>Customer's Bill Adjustment</u>. Should any meter fail to register correctly, the quantity of gas consumed will be determined by the Company based on information supplied by the customer and known by the Company subject to NHPUC Rules and Regulations. (See Puc 500 rules).

16 DISCONNECTION BY THE COMPANY

- A. <u>Disconnection by the Company</u>. The Company may disconnect its service to a customer for violation of its rules subject to NHPUC Rules and Regulations. (See Puc 1200 rules).
- B. <u>Non-Payment Shut-Off</u>. The Company may disconnect its service on reasonable notice and remove its equipment in case of non-payment of amounts billed for gas usage.
- C. <u>Shut-Off for Cause</u>. The Company may disconnect its service on reasonable notice if entry or access to its meter or meters is refused, obstructed, or hazardous, or for other violation of the Company's standard requirements.
- D. <u>Safety Shut-Off</u>. The Company may disconnect without notice if the customer's installation has become dangerous or defective.
- E. <u>Defective Equipment</u>. The Company may disconnect without notice if the customer's equipment, or use thereof, might injuriously affect the equipment of the Company or the Company's service to other customers.
- F. <u>Shut-Off for Fraud</u>. The Company may disconnect without notice for abuse, fraud or tampering with the connections, meters or other equipment of the Company.
- G. <u>Reconnection Charge</u>. A reconnection charge is made for reconnection of service discontinued by the Company and is payable in advance in addition to all other amounts due. The reconnection charge is made

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 70 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 16 General Terms and Conditions

instead of the meter account charge. The amount of the reconnection charge is the same as the comparable meter account charge except when it has been necessary to dig up the service pipe or connection to effect discontinuance of service. In such cases, the reconnection charge is the price of removal and restoration of service pipe or connection.

17. COST OF GAS CLAUSE EXCLUDING KEENE

- A. <u>Purpose</u>. The purpose of this Cost of Gas Clause is to establish procedures that allow the Company, subject to the jurisdiction of the Commission, to adjust, on a semiannual basis, its rates for firm gas sales in order to recover the costs of gas supplies, along with any taxes applicable to those supplies, pipeline and storage capacity, production capacity and storage, bad debt expense associated with purchased gas costs, and the costs of purchased gas working capital, to reflect the seasonal variation in the cost of gas, and to credit to customers receiving firm service from the Company all supplier refunds and capacity release sales.
- B. <u>Applicability</u>. This Cost of Gas Clause ("COGC") shall be applicable to the Company and all firm gas sales made by the Company, unless otherwise designated. The application to the clause may, for good cause shown, be modified by the NHPUC. See Section 17(N), "Other Rules."
- C. Cost of Firm Gas Allowable for COGC. All costs of firm gas including, but not limited to, commodity costs, taxes on commodity, demand charges, local production and storage costs, hedging related costs, other gas supply expense incurred to procure and transport supplies and commodity related bad debt expense, the gas used in Company operations, transportation fees, costs associated with buyouts of existing contracts, and purchased gas working capital may be included in the COGC. Any costs recovered through application of the COGC shall be identified and explained fully in the semiannual filings outlined in Section 17(M).
- D. <u>Effective Date of Cost of Gas Factor</u>. The seasonal Cost of Gas Factor ("COG") shall become effective upon NHPUC approval on the first day of each season as designated by the Company. Unless otherwise notified by the NHPUC, the Company shall submit COG filings as outlined in Section 17(M) of this clause on or before the first business day in September...
- E. Definitions. The following terms shall be defined in this section, unless the context requires otherwise.
 - 1. <u>Bad Debt Expense</u>: The uncollectible expense attributed to the portion of the Company's revenue associated with the recovery of gas costs under this clause.
 - 2. <u>Capacity Release Revenues</u>: The economic benefit derived from the sale or release of transportation and storage capacity that the Company has under contract.
 - 3. <u>Carrying Charges</u>: Interest expense calculated on the average monthly balance using the monthly prime lending rate, as reported by the Federal Reserve Statistical Release of Selected Interest Rates, and then added to the end of month balance.
 - 4. <u>Correction Factor</u>: Seasonal Adjustment necessary to align the peak day volumes used to calculate the Commercial and Industrial load factor ratios with the seasonal Commercial and Industrial High Winter and Low Winter throughput volumes applied to the cost of gas calculations.
 - 5. <u>Direct Gas Costs</u>: All purchased gas costs including supplier, storage and pipeline demand and commodity costs, as well as the commodity costs for local manufactured gas (Liquid Propane Gas ("LPG") and Liquefied Natural Gas ("LNG")).

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 71 of 470 Docket No. DG 20-105

Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 17 General Terms and Conditions

- 6. Economic Benefit: The difference between the revenues received and the marginal cost determined to serve non-core customers.
- 7. Inventory Finance Charges: As billed in each Winter Season for annual charges. The total shall represent an accumulation of the projected charges as calculated using the monthly average of financed inventory at the existing or anticipated financing rate through a trust or other financing vehicle.
- 8. Local Production and Storage Capacity Costs: The costs of providing storage service from the Company's storage facilities (i.e., LNG and LPG) as determined in the Company's most recent rate proceeding.
- 9. Market Based Allocator ("MBA"): The method used to allocate gas costs among Commercial and Industrial Customer Classifications. These ratios are presented in Section 17(F).
- 10. Non-Core Commodity Costs: The commodity cost of gas assigned to non-core sales to which the COG is not applied.
- 11. Non-Core Sales: Sales made under non-traditional off-system sales.
- 12. Non-Core Sales Margins: The economic benefit derived from non-core transactions to which the COG is not applied, including non-core sales generated from the use of the Company's Gas Supply Resource portfolio.
- 13. Summer Commodity: The gas supplies procured by the Company to serve firm load in the Summer Season.
- 14. Summer Demand: The gas supply demand and transmission capacity procured by the Company to serve firm load in the Summer Season.
- 15. Summer Season: The calendar months May 1 through October 31.
- 16. Off-System Sales Margin: The economic benefit derived from the non-firm sales of natural gas supplies upstream of Company's distribution system.
- 17. Winter Commodity: The gas supplies procured by the Company to serve firm load in the Winter Season.
- 18. Winter Demand: Gas supply demand, peaking demands, storage and transmission capacity procured by the Company to service firm load in the Winter Season.
- 19. Winter Season: The calendar months November 1 through April 30.
- 20. PR Allocator: The percentage of annual capacity charges assigned to the Winter Season calculated using the Proportional Responsibility Method.
- 21. Purchased Gas Working Capital: The allowable working capital derived from Winter Season and Summer Season demand and commodity related costs.

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman Neil Proudman

Original Page 18 General Terms and Conditions

F. <u>Approved Cost</u>. The Cost of Gas calculation utilizes information periodically established by the NHPUC. The table below lists the approved costs factors:

Variable	Description	Approved Figure
MISC	Miscellaneous Overhead	\$13,170
PS	Production and Storage Capacity	\$1,980,428
WCA%	Working Capital Allowance Percentage	3.91%

Bad Debt % Measurement and Reconciliation Period	COG Recovery Period	Actual Bad Debt Rate	Bad Debt allowed Recovery Rate
May 2010 – April 2011	November 2011 – October 2012	Actual	Actual
	November 2012 – October 2013	Greater than 2.9%	Actual less 0.4
May 2011 – April 2012		2.5% to 2.9%	2.5%
		Less than 2.5%	Actual
May 2012 - April 2013 and each	November 2013 - October 2014 and each subsequent November – October period thereafter	Greater than 3.3%	Actual less 0.8
subsequent May – April period		2.5% to 3.3%	2.5%
thereafter		Less than 2.5%	Actual

If the Company's actual bad debt percentage is reduced to 2.5% or less during any 12 month period, which need not be the same 12 months as the measurement periods defined above, then beginning with the reconciliation filing for the period during which this bad debt percentage was achieved the Company shall thereafter recover its actual gas supply related bad debt on a fully reconcilable basis and the percentages in the table above shall no longer apply. The actual bad debt percentage shall be calculated by dividing the Company's actual net write-offs for the relevant measurement period by its revenue for the same period.

G. Cost of Gas (COG) Calculations by Customer Class. The COG Formula shall be computed on a semiannual basis for three (3) groups of customer classes as shown on the following table. The computation will use forecasts of seasonal gas costs, carrying charges, sendout volumes, and sales volumes. Forecasts shall be based on either historical data or Company projections, but must be weather-normalized. Any projections must be documented in full with each filing.

The COG for the Residential rate classes shall represent the total system average unit cost of gas of meeting firm sales load, projected in each COG filing. The Commercial & Industrial (C&I) Low Winter (LW) and High Winter (HW) rates will be calculated in the following way: first, the demand unit cost of gas, the sum of purchased and stored gas demand costs divided by projected prorated sales, will be multiplied by the applicable load factor ratio and then multiplied by the correction factor. This adjusted demand factor will then be added to the commodity factor, adjustment factor and indirect cost of gas rate to determine the total COG rates for the C&I LW and HW rate classes. The two load factor ratios shall be derived once a year, for effect every November 1 through October 31, using the ratio of the unit capacity cost of each C&I group to the total system unit capacity cost that is determined in the Company's submittal of its Capacity

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 19 General Terms and Conditions

Allocators, for Capacity Assignment purposes, filed with its Winter COG, and as presented in Attachment C of the Delivery Service Terms and Conditions. The Correction Factor aligns the peak day volumes used to calculate the load factor ratios with the seasonal throughput volumes applied to the COG calculations.

GROUP	CUSTOMER CLASSES
Residential	Residential Heating and Non-Heating
Commercial and Industrial: Low Winter Use	G-51 through G-58
Commercial and Industrial: High Winter Use	G-41 through G-46

Winter Season Cost of Gas Formula (CGw)

The Winter Season COG shall be comprised of Winter Demand costs, Winter commodity costs, Winter reconciliation costs, Winter working capital reconciliation, Winter bad debt expenses, local production and storage capacity costs, and miscellaneous and A&G costs calculated at the beginning of the Winter Season according to the following formula:

$$CGw = Dw + Cw + Rw + WCRw + BDw + PS + ((MISC + Rbd) \times W:Sales)$$
A:Sales

Winter Demand Cost (Dw) Formula

Dw = DEMw - NCSMw + WCwd - R1d - R2d

and:

NCSMw = CRRw + OSSMw + SBdw

and:

 $WCwd = (DEMw - NCSMw) \times WCA\% \times CC$

where:

CGw = The total cost of gas for the Winter Season for the Company's firm sales customers previously defined.

BDw = Bad Debt expense for the Winter Season.

Cw = Commodity-related direct gas cost for the Winter Season.

Dw = The total Winter Demand costs.

DEMw = Demand Charges allocated to the Winter Season defined in Section 17(E).

NCSMw = The Non-Core Sales Margins equal to the sum of the Winter Season returnable Capacity Release Revenues, and Off-System Sales Margins.

WCwd = Working Capital allowable associated with demand charges allocated to the Winter Season as defined in Section 17(K).

R1d, R2d = Supplier demand-related refunds - The Supplier refunds associated with refund program credits derived from Account 5541-8048, "Undistributed Gas Suppliers' Refunds." See Section 17(I).

CRRw = The returnable Capacity Release Revenues allocated to the Winter Season. See Section 17(E).

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 74 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 20 General Terms and Conditions

- OSSMw = The returnable Off-System Sales Margins allocated to the Winter Season. See Section 17(E).
- SBdw = Demand revenues received from Firm Stand-By Sales Service customers in the Winter Season.
- WCA % = Percentage of gas costs equivalent to Working Capital Allowance associated with gas costs. Refer to Section 17(F) for this percentage.
- CC = Monthly interest rate as reported by the Federal Reserve Statistical Release of Selected Interest Rates.
- Rw = Reconciliation Costs Winter Season deferred gas costs, Account 1920-1740 balance, inclusive of the associated Account 1920-1740 interest, as outlined in Section 17(J).
- WCRw = Working Capital reconciliation adjustment associated with Winter Gas Costs Account 1163-1422 balance as outlined in Section 17(K).
- PS = The total dollar amount of costs associated with the local production and storage capacity gas less any production and storage capacity assignment revenues. Refer to Section 17(F) for this dollar amount.
- MISC = The total dollar amount of gas costs associated with acquisition, dispatching, Administrative and General expenses and overheads as determined in the Company's most recent rate proceeding. Refer to Section 17(F) for this dollar amount.
- Rbd = Annual Bad Debt Expense reconciliation adjustment Account 1920-1743 balance
- W:Sales = Forecasted firm sales volumes associated with the Winter Season.
- A:Sales = Forecasted annual firm sales volumes.

Winter Season Commodity (Cw) Formula

Cw = COMw + FC + WCwc - R1c - R2cand:

COMw = WSC - NCCCw - SBcw and:

 $WCwc = (COMw + FC) \times WCA\% \times CC$ where:

- COMw = Commodity Charges allocated to the Winter Season as defined in Section 17(E).
- FC = Inventory finance charges as defined in Section 17(E).
- WCwc = Working Capital Allowable Associated with commodity charges allocated to the Winter Season as defined in Section 17(K).
- R1c, R2c = Supplier commodity-related refunds The supplier refunds associated with refund program credits derived from Account 5541-8048, "Undistributed Gas Suppliers' Refunds". See Section 17(I).
- WSC = Commodity charges associated with gas supply sent out in Winter Season as defined in Section 17(E)

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman
EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 75 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 21 General Terms and Conditions

NCCCw = Non-Core Commodity Costs incurred in the Winter Season as defined in Section 17(E).

SBcw = Winter Season commodity revenues received from Firm Stand-By Gas Supply Service

sales customers.

WCA % = Percentage of gas costs equivalent to Working Capital Allowance associated with gas costs.

Refer to Section 17(F) for this percentage.

CC = Monthly interest rate as reported by the Federal Reserve Statistical Release of Selected

Interest Rates.

Winter Bad Debt (BDw) Formula

BDw = BD% x (Dw + Cw + Rw + WCRw)where:

BDw = Forecasted gas supply cost related Bad Debt Expense calculated for Winter Season.

BD% = Bad Debt percentage calculated based on a twelve month basis ending April of each year.

Refer to Section 17(F) Bad Debt Allowed Recovery Rate for this percentage.

Dw = Demand related costs in the Winter Season as previously defined.

Cw = Commodity related costs in the Winter Season as previously defined.

Rw = Reconciliation Costs – Winter Season deferred gas costs as previously defined.

WCRw = Winter Season Working Capital Reconciliation adjustment as previously defined.

Residential Winter Season Cost of Gas (COGwr)

All residential firm sales customers will pay the same Cost of Gas for the Winter Season. The factor represents the total forecasted Winter Season average cost of gas rate. This factor is calculated according to the following formula:

 $COGwr = \underbrace{CGw}_{W:Sales}$

where:

CGw = The total cost of gas for the Winter Season for the Company's firm sales customers

previously defined.

W:Sales = Forecasted sales volumes associated with the Winter Season.

R = Designates the Residential Heating and Residential Non-Heating customer classes.

Summer Season Cost of Gas (COG) Formula (CGs)

The Summer Season COG shall be comprised of Summer demand costs and Summer commodity costs, Summer reconciliation costs, Summer working capital reconciliation, plus a Summer bad debt charge, and a miscellaneous and A&G charge calculated at the beginning of the Summer Season according to the following formula:

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 76 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 22 General Terms and Conditions

$$CGs = Ds + Cs + Rs + WCRs + BDs + ((MISC + Rbd) x Sales)$$

A:Sales

Summer Demand Cost (Ds) Formula

Ds = DEMs + WCsd - Rld - R2d

 $WCsd = DEMs \times WCA\% \times CC$ where:

A:Sales = Forecasted annual sales volumes.

BDs = Bad Debt Expense for Summer Season.

Cs = Commodity-related direct gas costs for the Summer Season.

CGs = The total cost of gas for the Summer Season for the Company's firm sales customer previously defined.

DEMs = Demand charges allocated to the Summer Season defined in Section 17(E).

MISC = The total dollar amount of gas costs associated with acquisition, dispatching, Administrative and General expenses and overheads as determined in the Company's most recent rate proceeding. Refer to Section 17(F) for this dollar amount.

R1d, R2d = Supplier refunds from pipeline demand charges - The per unit supplier refunds associated with refund program credits derived from Account 5541-8048, "Undistributed Gas Suppliers' Refunds." See Section 17(I).

Rs = Summer Season Reconciliation Costs - Account 1920-1741 balance, inclusive of the associated Account 1920-1741 interest, as outlined in Section 17(J).

S:Sales = Forecasted sales volumes associated with the Summer Season.

WCA % = Percentage of gas costs equivalent to Working Capital Allowance associated with gas costs. Refer to Section 17(F) for this percentage.

CC = Monthly interest rate as reported by the Federal Reserve Statistical Release of Selected Interest Rates.

Rbd = Annual Bad Debt Expense reconciliation adjustment - Account 1920-1743 balance.

WCRs = Working Capital reconciliation adjustment associated with Summer gas costs – Account 1163-1424 as outlined in Section 17(K).

WCsd = Working Capital allowable costs associated with demand costs allocated to the Summer Season as defined in Section 17(K).

Summer Season Commodity Cost (Cs) Formula

Cs = COMs - NCCCs + WCsc - R1c - R2c and:

 $WCsc = (COMs - NCCCs) \times WCA\% \times CC$ where:

Docket No. DG 21-130 Exhibit 26 Page 77 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 23 General Terms and Conditions

- COMs = Commodity charges associated with gas supply sent out in the Summer Season as defined in Section 17(E).
- WCsc = Working Capital allowable costs associated with commodity charges allocated to the Summer Season as defined in Section 17(K).
- R1c, R2c = Supplier refunds from pipeline commodity charges The supplier refunds associated with refund program credits derived from Account 5541-8048, "Undistributed Gas Suppliers' Refunds."
- NCCCs = Non-core commodity costs incurred in the Summer Season as defined in Section 17(E).
- WCA % = Percentage of gas costs equivalent to Working Capital Allowance associated with gas costs. Refer to Section 17(F) for this percentage.
- CC = Monthly interest rate as reported by the Federal Reserve Statistical Release of Selected Interest Rates.

Summer Bad Debt (BDs) Formula

- BDs = BD% x (Ds + Cs + Rs + WCRs)where:
- BD% = Bad Debt percentage calculated based on a twelve month basis ending April of each year. Refer to Section 17(F) Bad Debt Allowed Recovery Rate for this percentage.
- BDs = Forecasted gas supply related Bad Debt Expense calculated for Summer Season defined in Section 17(E).
- Ds = Demand related costs in the Summer Season as previously defined.
- Cs = Commodity related costs in the Summer Season as previously defined.
- Rs = Reconciliation Costs Summer deferred gas costs as previously defined.
- WCRs = Summer Season Working Capital Reconciliation adjustment as previously defined.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 78 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 24 General Terms and Conditions

Residential Summer Season Cost of Gas (COGsr)

All residential firm sales customers will pay the same cost of gas for the Summer Season. The factor represents the total forecasted Summer Season average cost of gas rate. This factor is calculated according to the following formula:

COGsr = <u>CGs</u> S:Sales where:

CGs = The total cost of gas for the Summer Season for the Company's firm sales customers as previously defined.

S:Sales = Forecasted sales volumes associated with the Summer Season.

R = Designates the Residential Heating and Residential Non-Heating customer classes.

Commercial and Industrial Winter and Summer Season Cost of Gas

The Commercial and Industrial customer classes Winter Season Cost of Gas will be based on the Winter Season average cost of gas components used for the Residential Winter Season Cost of Gas. A separate Winter Season Cost of Gas factor will be computed for the low winter use class, Rates G-51, G-52, G-53, G-54, G-55, G-56, G-57, and G-58 and a separate Winter Season Cost of Gas Factor will be computed for the high winter use class, Rates G-41, G-42, G-43, G-44, G-45, and G-46.

The Commercial and Industrial customer classes Summer Season Cost of Gas will be based on the Summer Season average cost of gas components used for the Residential Summer Season Cost of Gas. A separate Summer Season Cost of Gas factor will be computed for the low winter use class, Rates G-51, G-52, G-53,G-54, G-55, G-56, G-57, and G-58 and a separate Summer Season Cost of Gas factor will be computed for the high winter use class, Rates G-41, G-42, G-43, G-44, G-45, and G-46.

These Cost of Gas Factors will be computed by applying ratios to the average demand portion of the Winter and Summer Season's cost of gas unit rate times the correction factor and then adding the remaining Residential average cost of gas unit rate.

These factors are calculated according to the following formulas:

Low Winter Use (COGwl) Formula Winter Season

 $COGwl = RATIOl \times CFw \times CGwd + CGwo$

Low Winter Use (COGsl) Formula Summer Season

COGsl = RATIOl x CFs x CGsd + CGso and:

RATIOl = DCcl
DDcl
DDc

and:

High Winter Use (COGwh) Formula Winter Season

 $COGwh = RATIOh \times CFw \times CGwd + CGwo.$

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 25 General Terms and Conditions

High Winter Use (COGsh) Formula Summer Season

COGsh = RATIOh x CFs x CGsd + CGso and

 $RATIOh = \underline{DCch} \quad \div \quad \underline{DCc}$ $DDch = \quad DDc$

and:

CFw = (WL:Sales + WH Sales)

(RATIOl x WL:Sales) + (RATIOh x WH:Sales)

CFs = (SL:Sales + SH:Sales)

(RATIOl x SL:Sales) + (RATIOh x SH:Sales)

 $CGwd = \underline{Dw}$

W:Sales

CGwo = CGw - Dw

W:Sales

 $CGsd = \underline{Ds}$

S:Sales

 $CGso = \underline{CGs - Ds}$

S:Sales

DCcl = Bcl * PLrate + (DDcl - Bcl) * REMrate

DCch = Bch * PLrate + (DDch-Bch) * REMrate

PLrate = PL/PLmdcq

 $REMrate = (\underline{DCc - (Bc*PLrate)})$

DDc - Bc

 $DCc = (\underline{DC \times DDc})$

DD

where:

Bc = The daily base load for all the Commercial and Industrial rate classes

Bch = The daily base load for the Commercial and Industrial rate classes G-41, G-42, G-43, G-

44, G-45 and G-46.

Bcl = The daily base load for the Commercial and Industrial rate classes G-51, G-52, G-53, G-

54, G-55, G-56, G-57 and G-58.

CFs = Summer Season Commercial and Industrial gas cost correction factor.

CFw = Winter Season Commercial and Industrial gas cost correction factor.

CGs = The total cost of gas for the Summer Season for the Company's firm sales customers as

previously defined.

CGw = The total cost of gas for the Winter Season for the Company's firm sales customers as

previously defined.

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman

NHPUC NO. 11 GAS LIBERTY UTILITIES

DC =

RATIO1 =

Original Page 26 General Terms and Conditions

	production and storage capacity costs, as stated in Section 17(F).
DCc =	The Commercial and Industrial rate classes pro-rata share of the annual forecasted pipeline, storage, and peaking demand capacity costs.
DCch =	The Commercial and Industrial pro-rata share of the annual forecasted pipeline, storage, and peaking demand capacity costs allocated to Commercial and Industrial High Winter Use rate classes, G-41, G-42, G-43, G-44, G-45, and G-46.
DCcl =	The Commercial and Industrial pro-rata share of the annual forecasted pipeline, storage, and peaking demand capacity costs allocated to the Commercial and Industrial Low Winter Use rate classes, G-51, G-52, G-53, G-54, G-55, G-56, G-57, and G-58.
DD =	Total peak design day determinants.
DDc =	The peak design day determinants allocated for all the Commercial and Industrial rate classes.
DDch =	The peak design day determinants for the Commercial and Industrial rate classes, G-41, G-42, G-43, G-44, G-45, and G-46.
DDcl =	The peak design day determinants for the Commercial and Industrial rate classes, G-51, G-52, G-53, G-54, G-55, G-56, G-57, and G-58.
Ds =	The total Summer Demand charges as defined below.
Dw =	The total Winter Demand charges as previously defined.
PL =	The annual forecasted pipeline only demand charges
PLmdcq =	The maximum daily contract pipeline volume available to the Company.
PLrate =	The pipeline demand rate.
RATIOh =	Ratio of the average high Winter Use class Cost of Gas low load factor demand capacity

The annual forecasted pipeline, storage and peaking demand charges plus the total

REMrate = The weighted average demand rate for storage and peaking supplies.

S: Sales = Forecasted sales volumes associated with the Summer Season.

SH:Sales = Total Winter Season forecasted Commercial and Industrial high winter use sales.

SL: Sales = Total Winter Season forecasted Commercial and Industrial low winter use sales volumes.

costs to the total average Commercial and Industrial demand capacity costs.

costs to the total average Commercial and Industrial demand capacity costs.

Ratio of the average low Winter Use class Cost of Gas high load factor demand capacity

W:Sales = Forecasted sales volumes associated with the Winter Season.

WH:Sales = Total Winter Season forecasted Commercial and Industrial high winter use sales.

WL: Sales = Total Winter Season forecasted Commercial and Industrial low winter use sales volumes.

<u>H.</u> Non-Core Sales Margins ("NCSM"). One hundred percent (100%) of margins from Off-System Sales and all revenues from Capacity Release will be credited to firm sales customers during the winter season through operation of the COG.

Docket No. DG 21-130 Exhibit 26 Page 81 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 27 General Terms and Conditions

I. <u>Gas Suppliers' Refunds</u>. Account 5541-8048: Refunds from suppliers of gas, from upstream capacity suppliers and suppliers of product demand are credited to Account 5541-8048, "Commodity and Demand Refunds." Transfers from these accounts will reflect as a credit in the semiannual calculation of the COG to be calculated as follows:

Refund programs shall be initiated with each semiannual COG filing and shall remain in effect for a period of one year. The total dollars to be placed into a given refund program shall be net of over/under-returns from expired programs plus refunds received from suppliers since the previous program was initiated. Refunds shall be segregated by demand and commodity charges and distributed volumetrically, producing per unit refund that will return the principal amount with interest as calculated using the Company's average short-term cost of borrowing for the month to the average of the beginning and end of month balances of refunds. The Company shall track and report on all Account 5541-8048 activities as specified in Section 17(K).

- J. Reconciliation Adjustments Various Accounts.
 - 1. The following definitions pertain to reconciliation adjustment calculations:
 - a. Capacity Costs Allowable per Winter Season Formula shall be:
 - (1) Charges associated with upstream storage transmission capacity and product demand procured by the Company to serve firm load in the Winter Season, plus a reallocation of a portion of such charges incurred in the Summer Season to serve firm load.
 - (2) Charges associated with peaking, downstream production and storage capacity to serve firm load dispatching costs, and other administrative and general expenses in connection with purchasing gas supplies in the Winter Season from the Company's most recent test year and allocated to firm sales service.
 - (3) Non-Core Sales Margins or economic benefits associated with returnable capacity release and off-system sales.
 - (4) Credits associated with firm Stand-by Gas Supply Service Monthly Reservation Charges, daily imbalance charges and fixed component of penalty charges billed transportation customers in the Winter peak Season.
 - (5) Winter Season Demand Cost carrying charges.
 - b. Gas Costs Allowable Per Winter Season Formula shall be:
 - (1) Charges associated with gas supplies, including any applicable taxes, purchased by the Company to serve firm load in the Winter Season.
 - (2) Credit non-core commodity costs assigned to non-core customers to which the COGC does not apply, as defined in Section 17(H) (NCCCw).
 - (3) Inventory finance charges (FC).
 - (4) Winter Season commodity cost carrying charges.
 - c. Capacity Costs Allowable Per Summer Season Formula shall be:
 - (1) Charges associated with transmission capacity and product demand procured by the Company to serve firm load in the Summer Season

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 82 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 28 General Terms and Conditions

- (2) Credits associated with daily imbalance charges and fixed component of penalty charges billed transportation customers in the Summer Season.
- (3) Summer Season demand cost carrying charges.
- d. Gas Costs Allowable Per Summer Season Formula shall be:
 - (1) Charges associated with gas supplies, including any applicable taxes, procured by the Company to serve firm load in the Summer Season.
 - (2) Non-core commodity costs associated with non-core sales to which the COG is not applied, as defined in Section 17(E).
 - (3) Summer Season commodity cost carrying charges.
- e. Costs Allowable Per Bad Debt Formula shall be:
 - (1) Costs associated with uncollected gas costs, incurred by the Company to serve sales load. Such costs represent the bad debt expense related to the gas supply related write-off of sales customers and will be computed by multiplying actual gas costs by the Bad Debt Allowed Recovery Rate specified in Section 17(F). The reconciliation adjustment each season will be computed as the difference between the actual supply related bad debt revenues and the actual gas costs multiplied by the actual Bad Debt Allowed Recovery Rate as specified in Section 17(F).
 - (2) Account 1920-1743 Annual Bad Debt, carrying charges.
- 2. Calculation of the Reconciliation Adjustments: These accounts contain the accumulated difference between gas cost revenues and the actual monthly gas costs incurred by the Company. The Company shall separate Account 175 into Winter Season Gas Costs (Account 1920-1740) and Summer Season Gas Costs (Account 1920-1741), Account 1920-1740 shall contain the accumulated difference between revenues toward gas costs calculated by multiplying the Winter Season Gas Cost for each Customer Classification, (COGwr, COGwl and COGwh) times monthly firm sales volumes for each Customer Classification, and the total costs allowable per the Winter Season gas cost formula. Account 1920-1741 shall contain the accumulated difference between revenues toward gas costs calculated by multiplying the Summer Season Gas Cost for each Customer Classification, (COGsr, COGsl and COGsh) times monthly firm sales volumes for each Customer Classification, and the total gas costs allowable per the Summer Season demand formula.

Carrying Charges shall be calculated on the average monthly balance of each subaccount. The interest rate is to be adjusted monthly using the monthly prime lending rate, as reported by the Federal Reserve Statistical Release of Selected Interest Rates.

The annual bad debt reconciliation adjustments Rbd shall be determined for use, incorporating the bad debt balances in Account 1920-1743.

The bad debt account balance contains the accumulated difference between the Bad Debt Allowed Recovery Rate for the applicable period multiplied by the actual gas costs and the actual supply related bad debt revenues for the Winter and Summer COG filings.

The Winter Season reconciliation shall be filed with the NHPUC no later than July 29 of each year.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 83 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 29 General Terms and Conditions

The Summer Season reconciliation shall be filed with the NHPUC no later than January 31 of each year.

- K. Working Capital Reconciliation Adjustments Accounts 1163-1422 and 1163-1424.
 - 1. The following definitions pertain to reconciliation adjustment calculations:
 - a. Working Capital Demand Gas Costs Allowable per Winter Season Gas Formula shall be:
 - (1) Charges associated with upstream storage, transmission capacity, and product demand procured by the Company to serve firm load in the Winter period, plus a reallocation of a portion of such charges incurred in the Summer Season to serve firm load.
 - (2) Carrying charges.
 - b. Working Capital
 - (3) Charges associated with gas supplies, including any applicable taxes, purchased by the Company to serve firm load in the Winter season.
 - (4) Non-core commodity costs associated with non-core sales to which the COG is not applied, as defined in Section 17(E).
 - (5) Carrying charges.
 - c. Working Capital Demand Gas Costs Allowable per Summer Season Gas Formula shall be:
 - (6) Charges associated with upstream storage and transmission capacity procured by the Company to serve firm load in the Summer Season.
 - (7) Carrying charges.
 - d. Working Capital Commodity Gas Costs Allowable per Summer Season Gas Formula shall be:
 - (8) Charges associated with gas supplies, including any applicable taxes, procured by the Company to serve firm load in the Summer Season.
 - (9) Non-core commodity costs associated with non-core sales.
 - (10) Carrying charges.
 - e. The Winter and Summer Cost of Gas working capital allowances shall be calculated by applying the Working Capital Allowance Percentage (WCA%) set forth in Section 17(F).
 - 2. Calculation of the Reconciliation Adjustments
 - a. Accounts 1163-1422 and 1163-1424 contain the accumulated difference between working capital allowance revenues and the actual monthly working capital allowance cost. The actual monthly working capital allowance shall be calculated by multiplying the actual gas costs times the Working Capital Allowance Percentage (WCA%) set forth in Section 17(F), to the actual Direct Gas Costs allowable.
 - b. The Winter Season working capital reconciliation adjustment (WCRw) shall be determined for use in the Winter Season Gas Cost calculations incorporating the Winter Season working

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 84 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 30 General Terms and Conditions

capital account 1163-1422. A Summer Season working capital reconciliation adjustment (WCRs) shall be determined for use in the Summer Season Gas Cost calculations incorporating the Summer Season working capital account 1163-1424 balance.

L. <u>Application of COG to Bills</u>: The Company will employ the COGs as follows: The COGs (\$/therm) for each customer group for each season shall be calculated to the nearest hundredth of a cent per unit and will be applied to each customer's monthly sales volume within the corresponding customer classification. The Cost of Gas will be applied to gas consumed on or after the first day of the month in which the cost of gas becomes effective.

M. Information Required to be Filed with the NHPUC.

- Reconciliation Adjustments: The Company shall file with the NHPUC a seasonal reconciliation of gas
 costs and gas cost collections containing information in support of the reconciliation calculation set out
 in Sections 17(J) (2) and 17(K) (2). Such information shall include the complete list of gas costs
 recoverable through the COGC over the previous season. This seasonal reconciliation shall be filed
 with the respective seasonal COG reconciliation filing, along with complete documentation of the
 reconciliation adjustment calculations.
 - Additionally, information pertaining to the Cost of Gas shall be filed with the NHPUC in accordance with the format established by the NHPUC. Reporting requirements include filing the Company's monthly calculation of the projected over or under-collection with the NHPUC, along with notification by the Company to the NHPUC of any revised COG for the subsequent month, not less than five (5) business days prior to the first day of the subsequent month.

The Company shall file with the NHPUC an annual reconciliation of bad debt expense and bad debt collections containing information in support of the reconciliation calculation set out in Sections 17(J) (1) and 17(J) (2). Such information shall detail the revenues collected as an allowance for bad debt, as well as the actual bad debt expense associated with gas cost recoverable through the COGC over the 12-month period ending April 30th. This annual reconciliation of bad debt expenses shall be filed with the Winter COG reconciliation filing, along with documentation.

- 2. Commercial and Industrial COG Ratio: The following factors will be filed annually by the Company for informational purposes. Significant changes in these factors signal the need to evaluate the COG ratios. These variables will assist in predicting significant shifting of the MBA-based escalator of gas costs and resulting changes in the COG ratios:
 - a. The percentage of load migration from sales to transportation service in the Commercial and Industrial High and Low Winter Use classes.
 - b. The ratio of delivered costs of winter supplies to pipeline delivered supplies.
 - c. The July and August consumption for the Commercial and Industrial High and Low Winter classes as a percentage of their annual consumption.

N. Other Rules.

1. The NHPUC may, where appropriate, on petition or on its own motion, grant an exception from the provisions of this tariff, upon such terms that it may determine to be in the public interest.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u> Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 85 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 31 General Terms and Conditions

- 2. The Company may, without further NHPUC action, adjust the approved COG upward or downward monthly based on the Company's calculation of the projected over or under-collection for the period, but the cumulative adjustments upward shall not exceed twenty-five percent (25%) of the approved COG.
- 3. The Company may, at any time, file with the NHPUC an amended COG.
- 4. The operation of the Cost of Gas Clause is subject to all powers of suspension and investigation vested in the NHPUC.
- 5. The Company shall file both seasonal COG filings on or before the first business day in September. The summer portion of the filing will include COG rates effective May 1 of the following year.
- O. Reconciliation Adjustment Accounts.

1163-1422

<u>Winter Season Gas Working Capital Allowance Reconciliation Adjustment for COGC</u>: This account shall be used to record the cumulative difference between Winter Season gas working capital allowance revenues and Winter Season gas working capital allowance. Entries to this account shall be determined as outlined in the Cost of Gas Clause.

1163-1424

<u>Summer Season Gas Working Capital Allowance Reconciliation Adjustment for COGC</u>: This account shall be used to record the cumulative difference between Summer Season gas working capital allowance revenues and Summer Season gas working capital allowance. Entries to this account shall be determined as outlined in the Cost of Gas Clause.

1920-1740

<u>Winter Season Gas Cost Reconciliation Adjustment for COGC</u>: This account shall be used to record the cumulative difference between Winter Season gas revenues and Winter Season gas costs. Entries to this account shall be determined as outlined in the Cost of Gas Clause.

1920-1741

<u>Summer Season Gas Cost Reconciliation Adjustment for COGC</u>: This account shall be used to record the cumulative difference between Summer Season gas revenues and Summer Season gas costs. Entries to this account shall be determined as outlined in the Cost of Gas Clause.

1920-1743

<u>Annual Bad Debt Reconciliation Adjustment for COGC</u>: This account shall be used to record the cumulative difference between Annual bad debt revenues and annual bad debt costs. Entries to this account shall be determined as outlined in the Cost of Gas Clause.

5541-8048

<u>Commodity and Demand Refunds</u>: This account shall be used to record the refunds from upstream commodity supplies and suppliers of product commodity and transfers of credits in the semiannual calculation of the COG as well as to record the refunds from upstream capacity supplies and suppliers of product demand and transfer of credits in the semiannual calculation of the COG. Entries to this account shall be determined as outlined in the Cost of Gas Clause.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 86 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 32 General Terms and Conditions

- P. <u>Firm Transportation Cost of Gas Charge</u>. To permit the Company to charge its firm transportation customers with a portion of the cost of gas produced by the Company between November 1 and April 30 of each year, there is a Firm Transportation Cost of Gas Charge ("FTCG") which applies to all firm transportation billed under this tariff. This volumetric charge is to compensate firm sales customers for the increase in gas costs, through the use of supplemental liquid fuels, attributable to firm transportation customers during the Winter Period.
 - 1. <u>Application</u>. The FTCG will be calculated for the Winter Period, defined as the period from November 1 through April 30. The FTCG will be applied to billings commencing with the first November revenue billing cycle
 - 2. Purpose. The amount of the FTCG is the estimated liquid costs used for pressure support purposes multiplied by the transportation throughput as a percentage of the total system throughput for the Winter Period. The resulting amount shall be adjusted by the prior period over or under collection, if any, and shall be recovered over the estimated total transportation throughput subject to the FTCG to yield a per therm volumetric charge. The FTCG shall be computed to the nearest one hundredth cent per therm and shown separately on customers' bills. At the conclusion of the Winter Period, the Company will calculate the extent that the FTCG revenues are greater or lesser than actual unit cost. The revenue and liquid costs will be reconciled so that all liquids costs shall be collected from either firm sales or firm transportation customers.
 - 3. <u>Changes</u>. The amount of the FTCG may be changed within the period whenever the unit cost materially deviates from the anticipated unit cost
 - 4. Reporting. The Company shall submit to the New Hampshire Public Utilities Commission, on or before the first business day in September, a copy of the FTCG computation. A reconciliation of the prior period under/over collection will be submitted to the New Hampshire Public Utilities Commission no later than July 29.
- Q. <u>Fixed Price Option Program.</u> Fixed Price Option Program. An alternative to the traditional Winter Period cost of gas pricing mechanism may be elected by a residential customer (rates R-1, R-3, R-4, R-5 or R-6) pursuant to the Company's Fixed Price Option Program (the "Program"). The Company may offer up to 50% of its weather normalized firm sales for the prior Winter Period under the Program. The cost of gas rate offered under the Program will remain fixed for all Winter Period deliveries commencing November 1 and ending April 30. The Company shall submit to the New Hampshire Public Utilities Commission on or before September 1 of each year a copy of the fixed price option computation. Once elected, customers must remain on the Program for the duration of the Winter Period, unless service is terminated. There are no maximum or minimum usage levels. No sign up fees apply.

18 COST OF GAS CLAUSE - KEENE DIVISION

- A. <u>Purpose</u>. To permit the Company to charge its customers in the Keene Division with the cost of gas purchased or produced. A cost of gas rate will be applied to all firm gas billed under this tariff as calculated on the appropriate pages.
- B. <u>Application</u>. A cost of gas rate will be calculated for the winter heating period, defined as the period from November 1 through April 30, and a cost of gas rate will be calculated for the summer period, defined as the period from May 1 through October 31.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 87 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 33 Terms and Conditions

The winter cost of gas rate will be applied to billings commencing with the first November revenue billing cycle; the summer cost of gas rate will be applied to billings commencing with the first May revenue billing cycle.

C. Calculation. The amount of the cost of gas rate is the anticipated unit cost of gas sold.

At the conclusion of each winter and summer period the Company will calculate the extent that cost of gas revenues are greater or less than actual unit costs of gas compared with the anticipated unit costs. The calculated difference (actual gas sales volumes multiplied by the difference between actual and anticipated unit costs) will be carried forward into the computation of the cost of gas rate for the corresponding winter or summer period.

Any excess revenue collected, as determined above, will earn interest as specified by the Commission.

- D. <u>Changes</u>. The cost of gas rate may be adjusted without further Commission action based on the projected over-/under-collection of gas costs, the adjusted rate to be effective the first of the month. Any such rate adjustments may not exceed a maximum rate of 25 percent above the approved rate, but there is no limit on the amount of any rate reductions.
- E. <u>Refunds</u>. When refunds are made to the Company by its suppliers that are applicable to increased charges collected under this provision, the Company will make appropriate refunds to its customers and as the Commission may direct.
- F. Reporting. The Company shall submit to the Commission, at least 30 days prior to the effective date, the proposed winter and summer period cost of gas rate computation. Any monthly adjustments to the cost of gas rate must be filed five (5) business days prior to the first day of the subsequent month (the effective date of the new rate).

The cost of gas rate shall be computed to the nearest one hundredth cent per therm and shown on customers' bills.

G. Fixed Price Option Program. An alternative to the traditional winter period cost of gas rate mechanism may be elected by the customer pursuant to the Company's Fixed Price Option (FPO) Program. The Company may offer up to 50% of its expected firm sales for the winter period under the FPO Program. The cost of gas charge offered under the FPO Program will remain fixed for all winter period billings commencing November 1 and ending April 30 of the effective winter period. Once elected, customers must remain on the FPO Program for the duration of the winter period unless service is terminated. There are no maximum or minimum usage levels. Customers may enroll in this Program by contacting the Company between the October 1 and October 19 period immediately preceding the effective winter period.

19 LOCAL DISTRIBUTION ADJUSTMENT CLAUSE AND NORMAL WEATHER ADJUSTMENT

A. Purpose. The purpose of the Local Distribution Adjustment Clause ("LDAC" or this "Clause") is to establish procedures that allow the Company, subject to the jurisdiction of the NHPUC, to adjust, on an annual basis, its delivery charges in order to recover Conservation Charges ("CC"), Revenue Decoupling Adjustment Factor ("RDAF"), Winter Period Surcharges ("WPS"), Environmental Surcharges ("ES") including the Relief Holder Surcharge ("RHS") and the Manufactured Gas Program Surcharge ("MGP"), rate case expenses and recoupment ("RCE"), Gas Assistance Program costs ("GAP")

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 88 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 34 Terms and Conditions

and any other expenses the NHPUC may approve from time to time. The purpose of the Normal Weather Adjustment ("NWA") is to establish procedures that allow the Company, subject to the jurisdiction of NHPUC, to calculate and apply, for each customer on a monthly basis, the Normal Weather Factor ("NWF").

B. <u>Applicability</u>. This Clause shall be applicable in whole or part to all of the Company's firm sales service and firm delivery service customers. The application of this clause may, for good cause shown, be modified by the NHPUC. See Section 19(K) "Other Rules." _____ This Clause is applicable to all rates and charges on a Rate Class basis. The application of the CC, RDAF, ES, RCE and GAP is applicable to all therms and therefore the application to all customers, including Managed Expansion Program Customers, is at the same rate per therm as the corresponding non-Managed Expansion Program Customers.

C. Conservation Charges Allowable for LDAC.

- 1. <u>Purpose</u>: The purpose of this provision is to establish a procedure that allows the Company, subject to the jurisdiction of the NHPUC, to adjust, on an annual basis, the Conservation Charge, if and when applicable, to firm sales service and firm delivery service throughput in order to recover from firm customers costs associated with its energy efficiency management programs.
- 2. Applicability: A conservation charge shall be applied to therms sold or transported by the Company subject to the jurisdiction of the Commission as determined in accordance with the provision of this rate schedule. Such conservation charge shall be determined annually by the Company, separately for the Residential Heating, and Commercial/Industrial rate categories, subject to review and approval by the Commission as provided for in this rate schedule.
- 3. <u>Calculation of Conservation Charge</u>: The Company will properly assign expenses for forecasted conservation expenditures to the applicable rate categories for a future twelve (12) month period commencing November 1 of each year. The total of such conservation expenditures plus any prior period reconciling adjustments shall be divided by therm sales as forecasted by the Company for the same annual period and rounded to the nearest hundredth of a cent. The resulting conservation charge shall be included in the Company's Local Distribution Adjustment Charge and applied to actual therms sold or transported for the following twelve (12) month period starting November 1 and ending October 31.
- 4. Reporting: The Company shall submit annual reports to the Commission reconciling any difference between the actual conservation expenditures and actual revenues collected under this rate schedule. The difference whether positive or negative will be carried forward into the conservation charge for the next recovery period. Upon completion of the conservation program(s), any over or under collection may be credited or charged to the deferred Winter Period cost of gas account, subject to Commission approval.
- 5. <u>Effective Date</u>: On or before the first business day in September of each year, the Company shall file with the NHPUC for its consideration and approval, the Company's request for a change in the CC applicable to each Rate Category during the next subsequent twelve-month period commencing with the calendar month of November.

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 89 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 35 Terms and Conditions

6. Reconciliation Adjustment: Account 1163-1755 shall contain the cumulative difference between the sum of the DSM expenditures incurred by the Company plus the sum of the DSM repayments and the revenues collected from customers. The Company shall file the reconciliation along with the COG filing on or before the first business day in September of each year.

D. Revenue Decoupling Adjustment Factor.

- 1. Purpose: Revenue decoupling eliminates the link between volumetric sales and Company revenue in order to align the interests of the Company and customers with respect to changing customer usage by establishing an allowed revenue per customer ("RPC"). The Company is allowed to collect that RPC for the number of actual customers it has in a given month. The purpose of the Revenue Decoupling Adjustment Factor ("RDAF") is to establish procedures that allow the Company, subject to the jurisdiction of the NHPUC, to adjust, on an annual basis, its rates for firm gas sales and firm transportation in order to reconcile the difference between the Actual Revenue collected and the Allowed Revenue. The purpose of the Normal Weather Adjustment ("NWA") is to adjust each customer's bill for the difference in delivery charges caused by the variation in actual Heating Degree Days ("HDDs") from normal HDDs during the Winter Period.
- 2. <u>Effective Date</u>: The RDAF and NWA shall take effect beginning on November 1, 2018, and replace the Lost Revenue Adjustment Mechanism (LRAM) established in Order No. 25,932 (Docket No. DE 15-137).
- 3. <u>Applicability</u>: The Revenue Decoupling Adjustment Factor and NWA shall apply to all of the Company's firm tariff rate schedules, excluding special contracts, as determined in accordance with the provisions of this RDAF and NWA.
- 4. <u>Definitions</u>: The following definitions shall apply throughout Section 17D:
 - a. <u>Actual Number of Customers</u> is the actual number of Equivalent Bills for the applicable Rate Class for each applicable month of the Decoupling Year.
 - b. Equivalent Bill. Customers are billed on different days of the month. To calculate the number of customers in a month for purposes of calculating the Monthly Actual Revenue it is necessary to use Equivalent Bills as a representation for customers. Equivalent Bills are calculated by dividing the number of days in the billing period of each customer's bill by 30.
 - c. <u>Billing Year is</u> the 12-months commencing November 1 immediately following the completion of the Decoupling Year.
 - d. <u>Decoupling Year</u>. The first Decoupling Year shall be the 10-month period from November 1, 2018 to August 31, 2019. Each subsequent Decoupling Year shall be the twelve months commencing September 1 through August 31.
 - e. <u>Rate Class</u> are customers taking service pursuant to the rate schedules combined as follows: Rates R-1 and R-5, Rates R-3, R-4, R-6, and R-7, Rates G-41 and G-44, Rates G-42 and G-45, Rates G-43 and G-46, Rates G-51 and G-55, Rates G-52 and G-56, Rates G-53 and G-57, and Rates G-54 and G-58.
 - f. Customer Class Group (CG) is the group of rate schedules combined for purposes of calculating the Revenue Decoupling Adjustment and the RDAF. The two Customer Class Groups are as follows:

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 90 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 36 Terms and Conditions

Residential Customer Class Group (CG1): defined as both Residential Non-Heating Rate Class and Residential Heating Rate Class, shall consist of all customers taking service pursuant to the Company's residential rate schedules. CG1 shall include customers taking service under rate schedules R-1, R-3, R-4, R-5, R-6 and R-7.

<u>Commercial and Industrial Customer Class Group (CG2)</u>: shall consist of all customers taking service pursuant to one of the Company's general service rate schedules, G-41, G-42, G-43, G-44, G-45, G-46, G-51, G-52, G-53, G-54, G-55, G-56, G-57 and G-58.

g. Distribution Revenue is the revenue from the Company's firm sales service and firm delivery service customers and does not include other revenue, special contracts, and all revenues recovered from Cost of Gas filings.

5. Calculation of Revenue Per Customer

a. Definitions:

- i. <u>Initial Approved Revenue</u> is the Distribution Revenue for each Rate Class less MEP Premium revenues, all as approved by Order of the Commission in a rate case, to determine the Commission approved Initial RPC per Rate Class.
- ii. <u>Incremental Approved Revenue</u> is the amount of any change to the Initial Approved Revenue that is approved by Order of the Commission between rate cases and used to determine the Commission approved Incremental RPC and new Approved RPC.
- iii. <u>Initial Revenue per Customer ("Initial RPC")</u> is calculated for each Rate Class and approved by Order of the Commission in a rate case.
- iv. Incremental Revenue per Customer ("Incremental RPC") divides the Incremental Approved Revenue for each Rate Class based on the monthly Equivalent Bills for that Rate Class for the test period associated with the Incremental Approved Revenue. The Incremental RPC is approved by Order of the Commission in a proceeding other than a rate case that results in a permanent adjustment to base distribution rates. Such proceedings may include, but are not limited to, step adjustments, property tax reconciliation, and temporary rates.
- v. <u>Approved Revenue per Customer ("Approved RPC")</u> is the sum of the Initial RPC per Rate Class and all Incremental RPC's per Rate Class, if any.
- b. A separate RPC is calculated for each month of the decoupling year for each Rate Class using Approved Revenue and Equivalent Bills for each month.
- c. The Initial RPC is set for each month by calculating the Initial Approved Revenue by Rate Class by month from the approved rate schedule rates by Rate Class for that month. The rate schedule rates adjusted for the MEP Premium are multiplied by the appropriate monthly billing units, Equivalent Bills for customer charges, therms for the blocked therm charges, to produce the Initial Approved Revenue by Rate Class by month. The monthly Initial Approved Revenue are then divided by the Equivalent Bills for that month to produce the Initial RPC's for each Rate Class. For the Initial RPC's, the Equivalent Bills and associated blocked therm billing units shall be adjusted by an End of Year Bill Adjustment that reflects the number of customers that received service at the end of the test year.
- d. The Incremental RPC is set for each month by calculating the Incremental Approved Revenue by Rate Class by month from the approved rate schedule rates by Rate Class for that month. The rate schedule rates adjusted for the MEP Premium are multiplied by the appropriate monthly billing units, Equivalent Bills for customer charges, therms for the blocked therm charges, to produce the Incremental Approved Revenue by Rate Class by month. The monthly

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 91 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 37 Terms and Conditions

Initial Approved Revenue are then divided by the Equivalent Bills for that month to produce the Initial RPC's for each. For the Incremental RPC's, the Equivalent Bills and associated blocked therm billing units shall be those bills and units for the test period associated with the Incremental Approved Revenue.

6. Calculation of Normal Weather Adjustment

- a. Definitions.
 - i. Real-time normal weather adjustment is the difference between actual distribution revenue billed to each customer in each billing cycle for each month or portion thereof during the Winter Period, and what distribution revenue for each customer's bill would have been based on weather normalized therm deliveries for the same period. The resulting charge or credit will be added to or subtracted from each customer's bill at the time the bill is rendered (i.e., "real time").
 - ii. Winter Period. The time period from November 1 of a given year through April 30 of the following year inclusive.
 - iii. Base Load Factor for each customer is the customer's most recent two-year average daily delivered therms for actual bills rendered for those billing periods that are completely within the June 1 through August 31 calendar period excluding such billing periods that are only partially within the June 1 to August 31 period. If a customer has less than two-year's billing history, then the customer's available history for the months of June through August as defined above will be used to calculate the average daily delivered therms; and if a customer has no billing history for the months of June through August as defined above, then the average daily delivered therms for the calendar months of June through August for the rate schedule under which the customer is served will be used.
 - iv. Base Usage for each bill is the current Base Load Factor times the number of days in the billing period.
 - v. Heating Usage for each bill is the difference between the actual delivered therms for that bill less the Base Usage for that bill. If the calculated Heating Usage is less than zero, then the Heating Usage for that bill is set equal to zero.
 - vi. Heating Degree Days (HDD) for each day is sixty-five (65) minus the average temperature in degrees Fahrenheit for that day. If the calculated HDD is less than zero, then the HDD for that day is set equal to zero.
 - vii. Normal Heating Degree Days (Normal HDD) for each day is the thirty-year average HDD for that day.
- viii. Normal Weather Adjustment Slope (NWA Slope) for each bill is the Heating Usage divided by the sum of actual HDD during the billing period.
 - ix. Normal Heating Usage for each bill is the NWA Slope times the sum of the Normal HDD for the billing period.
 - x. Normal Usage for each bill is the sum of the Base Usage and the Normal Heating Usage.
- xi. Normal Weather Factor (NWF) for each bill is

$$NWF = \frac{DeliveryCharge_{Normal}}{DeliveryCharge_{Actual}} - 1$$

ISSUED BY: /s/Neil Proudman DATED: XXX XX, 2021

Neil Proudman

EFFECTIVE: August 1, 2021 TITLE: President

091

Docket No. DG 21-130 Exhibit 26 Page 92 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 38
Terms and Conditions

where <u>Delivery Charge Normal</u> is the calculated delivery charge for Normal Usage for the rate schedule applicable to that bill or portion thereof during the Winter Period and <u>Delivery Charge Actual</u> is the calculated delivery charge for actual delivered therms for the rate schedule applicable to that bill or portion thereof during the Winter Period.

The Normal Weather Adjustment (NWA) for each bill is

$NWA = DeliveryCharge_{Actual} \times NWF$

where Delivery Charge Actual is the calculated delivery charge for actual delivered therms for the rate schedule applicable to that bill or portion thereof during the Winter Period.

7. Application of the NWA to Customer Bills

The NWA charge or credit will be separately stated, and added to or subtracted from each bill as applicable. Each bill will have a separate line titled "Normal Weather Adj.," which line will include the total variable distribution charges, the NWF percentage, and the resulting charge or credit.

8. Calculation of Revenue Decoupling Adjustment and Reconciliation

- a. <u>Definitions</u>
 - i. <u>Monthly Actual Distribution Revenue</u> is the monthly billed Distribution Revenue less the MEP Premium for that month.
 - ii. <u>Monthly Allowed Revenue</u> is the Approved RPC per Rate Class for the applicable month multiplied by the Actual Equivalent Bills for that month.
- iii. <u>Forecasted Throughput Volume</u> is the forecasted firm tariff throughput for a given Customer Group for the Billing Year.
- iv. <u>Revenue Decoupling Accounts</u> ("RDA Accounts") are the accounts established on the balance sheet for the purpose of recording the Revenue Decoupling Adjustment for each Customer Class Group.
- b. Description of Revenue Decoupling Adjustment and Reconciliation

Each month the Company will record a Revenue Decoupling Adjustment in the RDA Accounts in accordance with generally accepted accounting principles. The Revenue Decoupling Adjustment is the difference between the Monthly Allowed Revenue and the Monthly Actual Distribution Revenue. In addition, the reconciliation amounts collected or distributed through the RDAF are also recorded in the RDA Accounts for each Customer Class Group. The RDA Accounts accrue interest on the average monthly balance using the prime lending rate. At the conclusion of each Decoupling Year, the sum of the balance in each of the RDA Accounts for each Customer Class Group shall be used to determine the RDAF for the next Billing Year.

The RDAF to be applied to customers' bills in the Billing Year is the balance in the RDA Accounts at the end of the Decoupling Year for each Customer Class Group divided by the Forecasted Throughput Volume for that Customer Class Group.

9. Application of the RDAF to Customer Bills

The RDAF (\$ per therm) shall be calculated annually for each Customer Group and shall be truncated at four decimal points per therm. The annual calculated Customer Group RDAF will be applied to the monthly firm tariff throughput for each customer in that particular Customer Group, effective November 1 of the given year.

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 93 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 39 Terms and Conditions

10. Information to be Filed with the Commission

Information pertaining to the RDAF will be filed annually with the Commission consistent with the filing requirements of all costs and revenue information included in the LDAC. Such information shall include:

- a. The calculation of the RDA Account Balance at the end of the Decoupling Year.
- b. The calculation of the RDAF for the Decoupling Year by Customer Class Group to be applied to the upcoming Billing Year.
- c. The approved calculation of the RDAF for the previous Billing Year by Customer Class Group.
- d. The calculation for the each Commission Approved RPC and the associated Equivalent Bills for the Initial RPC and each Incremental RPC.

E. Environmental Surcharges ("ES") Allowable for LDAC.

- 1. <u>Purpose</u>: In order to recover expenditures associated with former manufactured gas Programs, there shall be an ES Rate applied to all firm volumes billed under the Company's delivery service charges.
- 2. <u>Applicability</u>: An annual ES Rate shall be calculated effective every November 1 for the annual period of November 1 through October 31. The annual ES Rate shall be filed with the Company's Winter season Cost of Gas Clause ("COG") filing and be subject to review and approval by the Commission. The annual ES Rate shall be applied to firm sales and to firm delivery throughput as a part of the LDAC. Special contract customers are exempt from the ES.
- 3. <u>Costs Allowable</u>: All approved environmental response costs associated with manufactured gas Programs may be included in the ES Rate

The total annual charge to the Company's customers for environmental response costs during any annual ES recovery period shall not exceed five percent (5%) of the Company's total revenues from firm gas sales and delivery throughput during the preceding twelve (12) month period ending June 30. The total annual charge shall represent the ES expenditures reflected in the calculation of the ES Rate to be in effect for the upcoming twelve-month period, November 1 through October 31. If this recovery limitation results in the Company recovering less than the amount that would otherwise be recovered in a particular ES Recovery Year, then the Company would defer this unrecovered amount, with interest, calculated monthly on the average monthly balance, until the next recovery period in which this amount could be recovered without violating the 5% limitation. The interest rate is to be adjusted monthly using the monthly prime lending rate, as reported by the Federal Reserve Statistical Release of Selected Interest Rates.

4. <u>Effective Date</u>: On or before the first business day in September of each year, the Company shall file with the NHPUC for its consideration and approval, the Company's request for a change in the ES applicable to all firm sales and firm delivery service throughput for the subsequent twelvemonth period commencing with the calendar month of November.

5. <u>Definitions</u>:

Environmental Response Costs shall include all costs of investigation, testing, remediation, litigation expenses, and other liabilities relating to manufactured gas Program sites, disposal sites, or other sites onto which material may have migrated, as a result of the operating or decommissioning of New Hampshire gas manufacturing facilities. These cost shall include the costs of the closure of the Relief Holder and pond at Gas Street, Concord, NH. The ES shall also include the expenses incurred by the Company in pursuing insurance and third-party claims and any recoveries or other benefits received by the Company as a result

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 94 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 40 Terms and Conditions

- 6. <u>Reconciliation Adjustments:</u> Prior to the Winter Period COG, the Company shall calculate the difference between (a) the revenues derived by multiplying firm sales and delivery throughput by the ES Rate, and (b) the historical amortized costs approved for recoveries in the prior November's Annual ES Recovery Period. Account 1920-1863 shall contain the cumulative difference and the Company shall file the reconciliation along with its COG filing on or before the first business day in September of each year.
- 7. Calculation of the ES: The ES Rate calculated annually consists of one-seventh of actual response costs incurred by the Company in the twelve-month period ending June 30 of each year until fully amortized (over seven years). Any insurance and third-party recoveries or other benefits for the twelve month period ending June 30 shall be applied to reduce the unamortized balance, shortening the amortization period. The sum of these amounts is then divided by the Company's forecast of total firm sales and delivery throughput for the upcoming twelve months of November 1 through October 31.
- 8. <u>Application of ES to Bills</u>: The annual ES Rate shall be calculated to the nearest one one-hundredth of a cent per therm and shall be applied to the monthly firm gas sales and firm delivery service throughput by being included in the determination of the annual LDAC, and also shall be included in the Distribution Adjustment of the Delivery Charges of each firm customer's bill.
- F. Expenses Related to Rate Cases/Temporary Rate Reconciliation Allowable for LDAC.
 - 1. <u>Purpose</u>: The purpose of this provision is to establish a procedure that allows the Company to adjust its rates for the recovery of NHPUC-approved rate case expenses and the reconciliation of temporary rates.
 - 2. <u>Applicability</u>: The Rate Case Expenses/Temporary Rate Reconciliation ("RCE") shall be applied to all firm tariffed customers. The RCE will be determined by the Company, as defined below.
 - 3. <u>Rate Case Expenses Allowable for LDAC</u>: The total amount of the RCE will be equal to the amount approved by the Commission.
 - 4. <u>Effective Date of Rate Case Expense Charge</u>: The effective date of the RCE will be determined by the NHPUC in an individual rate proceeding.
 - 5. <u>Definition</u>: The RCE includes all rate case-related expenses approved by the NHPUC. This includes legal expenses, costs for bill inserts, costs for legal notices, consulting fees processing expenses, and other approved expenses. The temporary Rate reconciliation will include the variance between the delivery revenues obtained from the rates prescribed in the temporary rate order and the delivery revenues obtained from the final rates approved by the NHPUC.
 - 6. Rate Case Expense/Temporary Rate Reconciliation (RCE) Factor Formulas: The RCE will be calculated according to the Commission Order issued in an individual proceeding to establish details including the number of years over which the RCE shall be amortized and the allocation of recovery among rate classes. In general, the RCE Factor will be derived by dividing the annual portion of the total RCE, plus the RCE Reconciliation Adjustment, by forecast firm annual throughput.
 - 7. <u>Reconciliation Adjustments:</u> Account 1930-1745 shall contain the accumulated difference between revenues toward Rate Case Expenses as calculated by multiplying the Rate Case Expense Factor ("RCEF") times the appropriate monthly volumes and Rate Case Expense allowed, plus carrying charges added to the end-of-month balance. The carrying charges shall be calculated beginning on the first month of the recovery period by applying the monthly prime lending rate, as reported by the Federal Reserve Statistical Release of Selected Interest Rates to the average monthly balance.

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 95 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 41 Terms and Conditions

At the end of the recovery period, any under or over recovery will be included in an active LDAC component, as approved by the Commission.

- 8. <u>Application of RCE to Bills</u>: The RCE (\$ per therm) shall be calculated to the nearest one one-hundredth of a cent per therm and shall be applied to the monthly firm gas sales and firm delivery service throughput by being included in the determination of the annual LDAC, and also shall be included in the Distribution Adjustment of the Delivery Charges of each firm customer's bill.
- 9. <u>Information to be Filed with the NHPUC</u>: Information pertaining to the RCE will be filed with the NHPUC consistent with the filing requirements of all cost and revenue information included in the LDAC. The RCE filing will contain the calculation of the new RCE and will include the updated RCE reconciliation balance.

G. Recoverable Gas Assistance Program Costs.

- 1. <u>Purpose</u>: The purpose of this provision is to establish a procedure that allows the Company, subject to the jurisdiction of the NHPUC, to recover the revenue shortfall (costs) associated with customers participating in the Gas Assistance Program ("GAP"). Such costs, as well as, associated administrative and marketing costs shall be recovered by applying a GAP rate to all firm sales and transportation service throughput.
- 2. GAP Rate shall be applied to all firm sales and transportation tariff customers. The GAP Rate shall be filed with the Company's Winter season Cost of Gas Clause filing and shall be determined annually by the Company and be subject to review and approval by the Commission.
- 3. <u>Effective Date</u>: On or before the first business day in September of each year, the Company shall file with the NHPUC for its consideration and approval, the Company's request for a change in the GAP Rate applicable to all firm sales, delivery and transportation service throughput for the subsequent twelve-month period commencing with the calendar month of November.
- 4. GAP Costs Allowable for LDAC: The costs to be recovered through the GAP Rate shall comprised of the revenue shortfall calculated by forecasting the number of customers enrolled in the GAP and the associated volumetric billing determinants for the upcoming annual recovery period and applying those billing determinants to the difference between the regular and reduced gas assistance program residential base rates, plus the GAP discount applied to the cost of gas, administrative, marketing and startup costs. The GAP Rate shall be calculated by dividing the resulting costs, plus any prior period reconciling adjustment, by the forecast of annual firm sales and transportation service throughput.

5. GAP Factor Formula

 $GAPF = \underline{GAP + A_{GAP}}$

A: TPev

where:

A: Tpev Forecast Annual Throughput Volumes of all firm sales and transportation tariffed customers eligible to receive firm delivery-only service from the Company.

GAP GAP costs comprising of the revenue shortfall associated with customer participation, plus administrative, marketing, IT and start-up costs.

RA_{GAP} GAP Reconciliation Adjustment - Account 1169-1756, inclusive of the associated Account 1169-1756 interest, as outlined in Section 19(G)(6).

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 96 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 42
Terms and Conditions

- 6. Reconciliation Adjustments: Prior to the Company's Winter season Cost of Gas filing, the Company will calculate the difference between (a) the revenue derived by multiplying the actual firm sales and delivery service throughput by the GAP Rate through October 31st, and (b) the actual costs of the program which consists of (1) the revenue shortfall calculated by applying the actual billing determinants of the GAP classes to the difference in the regular and reduced residential base rates and cost of gas rates in effect for the annual reconciliation period and (2) the GAP discount applied to the cost of gas, start-up, administrative and marketing costs associated with the implementation of the program, plus carrying charges calculated on the average monthly balance using the monthly prime lending rate, as reported by the Federal Reserve Statistical Release of Selected Interest Rates. The combined costs will then be recorded in the deferred GAP account 1169-1756. The Company shall file the reconciliation along with its COG filing on or before the first business day in September of each year.
- H. Effective Date of Local Distribution Adjustment Clause. The LDAC shall be filed annually and become effective on November 1 of each year pursuant to NHPUC approval. In order to minimize the magnitude of future reconciliation adjustments, the Company may request interim revisions to the LDAC rates, subject to review and approval of the NHPUC.
- I. <u>Local Distribution Adjustment Clause Formulas</u>. The LDAC shall be calculated on an annual basis, by customer, by summing up the various factors included in the LDAC, where applicable.

LDAC Formula

 $LDAC^{X} = CC^{X} + RDAF^{X} + ES + GREF^{X} + RCE + GAP$ and:

 $ES^{X} = RHS + MGP$

where:

LDAC^X = Annualized class specific LDAC.

 CC^X = Annualized class specific CC or EE Charge.

 $RDAF^{X} = Annualized class specific RDAF.$

ES = Total firm annualized ES.

RHS = Annualized charge to recover the costs of the closure of the Relief Holder at Gas Street, Concord, NH

MGP = Annualized charge to cover the remediation costs related to former manufactured gas plants.

GREF^X = Total firm annualized class specific Gas Restructuring Expense Factor.

RCE = Rate Case Expense Factor.

GAP = Gas Assistance Program Rate

J. <u>Application of LDAC to Bills</u>. The component costs comprising the LDAC (\$ per therm) shall be calculated to the nearest one one-hundredth of a cent per therm and shall be applied to the monthly firm sales and firm delivery service throughput in accordance with the table shown in Section 19(B).

K. Other Rules.

- 1. The NHPUC may, where appropriate, on petition or on its own motion, grant an exception from the provisions of these regulations, upon such terms that it may determine to be in the public interest.
- 2. Such amendments may include the addition or deletion of component cost categories, subject to the

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 97 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 43 Terms and Conditions

review and approval of the NHPUC.

- 3. The Company may implement an amended LDAC with the NHPUC approval at any time.
- 4. The NHPUC may, at any time, require the Company to file an amended LDAC.
- 5. The operation of the LDAC is subject to all powers of suspension and investigation vested in the NHPUC.

L. Amendments to Uniform System of Accounts

- 1163-1755 Energy Efficiency Reconciliation Adjustment: This account shall be used to record the cumulative difference between the sum of DSM and/or EE Expenditures incurred by the Company plus the sum of DSM and/or EE Repayments and the revenues collected from customers pursuant to this clause with respect to a given Rate Category. Entries to this account shall be determined as outlined in the Local Distribution Adjustment Clause, 17(C).
- 1920-1863 Environmental Response Costs Reconciliation Adjustment: This account shall be used to record the cumulative difference between the revenues toward environmental response costs as calculated by multiplying the ES times monthly firm sales volumes and delivery service throughput and environmental response costs allowable per formula. Entries to this account shall be determined as outlined in the Local Distribution Adjustment Clause, 17(E).
- 1930-1745 Rate Case Expense/Temporary Rates Reconciliation Adjustment: This account shall be used to record the cumulative difference between the recovery and actual amounts of third-party incremental expenses associated with the Company's Rate Case initiatives and the difference between the final and temporary distribution rates. Entries to this account shall be determined as outlined in the Local Distribution Adjustment Clause, 17(F).
- 1169-1756 **Gas Assistance Program Reconciliation Adjustment:** This account shall be used to record the cumulative difference between the actual revenue derived from the actual sales and transportation service throughput multiplied by the GAP rate and the actual costs of the program, which consists of the revenue shortfall and all administrative and marketing costs, as outlined in the Local Distribution Adjustment Clause, 17(G).1168-1823 **Revenue Decoupling Adjustment Factor:** The RDA Accounts shall be used to record the difference between Actual Revenue and Allowed Revenue as well as the RDAF Revenue and associated interest as described in 17(D).

20 SUPPLY & CAPACITY SHORTAGE ALLOCATION POLICY

A. <u>DEFINITIONS</u>

The following are definitions of terms used in this subsection and applicable only to this subsection:

- 1. <u>Residential</u>: Service to customers which consists of direct natural gas usage in a residential dwelling for space heating, air conditioning, cooking, water heating and other residential uses
- 2. <u>Commercial</u>: Service to customers engaged primarily in the sale of goods or services including institutions and local, state and federal government agencies for uses other than those involving manufacturing or electric power generation
- 3. <u>Industrial</u>: Service to customers engaged primarily in a process which creates or changes raw or unfinished materials into another form or product including the generation of electric power
- 4. <u>Large Volume</u>: Service to large commercial and industrial customers with an annual gas load greater than 200,000 therms

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 98 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 44 Terms and Conditions

- 5. <u>Seasonal</u>: Service available from April 1 to October 31 to all customers using gas to replace some other fuel or gas for air conditioning purposes
- 6. <u>Firm Sales Service</u>: Service from schedules or contracts under which seller is expressly obligated to supply and deliver specific volumes within a given time period and which anticipates no interruptions, but which may permit unexpected interruption in case the supply to higher priority customers is threatened
- 7. <u>Firm Transportation Service</u>: Service from schedules or contracts under which seller is expressly obligated to deliver specific third-party volumes within a given time period and which anticipates no interruptions, but which may permit unexpected interruption in case the supply to higher priority customers is threatened.
- 8. <u>Plant Protection Gas:</u> Is defined as minimum volumes required to prevent physical harm to the plant facilities or danger to plant personnel, when such protection cannot be afforded through the use of alternate fuel. This includes the protection of such material in process as would otherwise be destroyed, but shall not

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 99 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 45 General Terms and Conditions

include deliveries required to maintain plant production. For the purpose of this definition, propane and other gaseous fuels shall not be considered alternate fuels.

- Feedstock Gas: Is defined as natural gas used as a raw material for its chemical properties in creating an end product
- J. <u>Process Gas</u>: Is defined as gas use for which alternate fuels are not technically feasible such as in applications requiring precise temperature controls and precise flame characteristics. For the purpose of this definition, propane and other gaseous fuels shall not be considered alternate fuels
- K. <u>Boiler Fuel</u>: Is considered to be natural gas used as a fuel for the generation of steam or electricity including the utilization of gas turbines for the generation of electricity
- L. <u>Alternate Fuel Capabilities</u>: Is defined as a situation where an alternate fuel could have been utilized whether or not the facilities for such use have actually been installed, provided however, where the use of natural gas is for plant protection, feedstock or process uses and the only alternate fuel is propane or other gaseous fuel, then the consumer will be treated as if he had no alternate fuel capability.

M. POLICY

In the event that, due to gas supply restrictions or capacity constraints, the Company is unable to deliver the total requirements of its firm, sales or transportation rate customers, the available volumes of gas will be allocated to the Company's firm rate customers in accordance with the provisions of this policy. In the event that the Company, during a curtailment or interruption, requires emergency gas, and takes the gas of the customer, customer shall be compensated for such emergency gas at the customer's alternate cost of fuel as demonstrated to the reasonable satisfaction of the Company.

As curtailment becomes necessary through each succeeding category, the Company will implement full or partial curtailment of a customer, or by groups of customers, taking into consideration customer load characteristics, the Company's delivery system design and Company load characteristics in a manner which is believed to be in the best interests of customers in general.

N. PRIORITIES

Firm rate customers shall be serviced according to the following preference categories with the first and each succeeding category having preference over the succeeding categories:

- 1. Company use for fuel and lost and unaccounted for
- O. Firm sales or transportation service for high priority residential uses including apartment buildings and other multi-unit buildings, small commercial establishments using less than 50 DT on a peak day, schools, hospitals, police protection, fire protection, sanitation facilities and correctional facilities
- P. Firm sales or transportation service for essential agricultural uses, as defined by the Secretary of Agriculture, for full food and natural fiber production, process and feedstock use for fertilizer and agricultural chemicals,

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 100 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 46 General Terms and Conditions

process and feedstock for animal feeds and food, food quality maintenance, food packaging, marketing and distribution for food related products and on farm uses

- Q. Firm sales or transportation service for large commercial requirements (50 DT or more on a peak day), firm industrial requirements for plant protection, feedstock and process needs and firm industrial sales up to 300 DT per day
- R. Firm sales or transportation service for all industrial requirements not specified in (2), (3), (4), (6), or (7)
- S. Firm sales or transportation service including the transportation for industrial requirements for boiler fuel use at less than 1,500 DT per day, but more than 300 DT per day, where alternate fuel capabilities can meet such requirements
- T. Firm sales or transportation service including transportation for industrial requirements for large volume (1,500 DT or more per day) boiler fuel use where alternate fuel capabilities can meet such requirements

U. STORAGE INJECTION

Within each category, storage injection required to meet the needs of higher priorities may be given preference over all other uses within that category.

V. PENALTY

For all unauthorized volumes of gas taken by a customer, the customer shall pay the Company a penalty of five times the daily index for each therm taken. Such penalty shall be added to the regular rates in effect. The Company shall have the right, without obligation, to waive any penalty for unauthorized use of gas, if on the day when the penalty was incurred deliveries to other of the Company's customers were not adversely affected. Continued unauthorized use, at the sole discretion of the Company, may result in termination of service.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 101 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 47 Residential Non-Heating Rate R-1

II. RATE SCHEDULES

RESIDENTIAL NON-HEATING RATE: CLASSIFICATION NO. R-1

Availability

This rate is available to all residential customers who do not have gas space heating equipment, who consume less than 80% of their normal usage in the six winter months of November through April and whose usage does not exceed 100 therms in any winter month. Available for use which is separately metered and billed for each dwelling unit. Availability is limited to use in locations served by the Company's mains and for which the Company's facilities are adequate.

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter: \$0.5130 per day or \$15.39 per 30 day month

Winter Period: All therms per 30 day month at \$0.4358 per therm

Summer Period: All therms per 30 day month at \$0.4358 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Cost of Gas Charge

All gas delivered under this rate is subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charges presented above are exclusive of these charges. Refer to the Firm Rate Schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 102 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 48 Residential Non-Heating Rate R-1

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

Docket No. DG 21-130 Exhibit 26 Page 103 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 49 Residential Heating Rate R-3

2 RESIDENTIAL HEATING RATE: CLASSIFICATION NO. R-3

Availability

This rate is for all residential use for those domestic customers who use gas as the principal household heating fuel. Availability is limited to use in domestic locations which are separately metered and billed and which are served by the Company's mains and for which the Company's facilities are adequate.

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter: \$0.5130 per day or \$15.39 per 30 day month

Winter Period: All therms per 30 day month at \$0.5985 per therm

Summer Period: All therms per 30 day month at \$0.5985 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Cost of Gas Charge

All gas delivered under this rate is subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charges presented above are exclusive of these charges. Refer to the Firm Rate Schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

Eligibility shall be determined based on the reasonable discretion of the Company subject to verification of heating usage.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent ($1\frac{1}{2}$ %) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Docket No. DG 21-130 Exhibit 26 Page 104 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 50 Residential Heating Rate R-3

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

Docket No. DG 21-130 Exhibit 26 Page 105 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11GAS LIBERTY UTILITIES Original Page 51 Residential Gas Assistance Heating Rate R-4

3 GAS ASSISTANCE PROGRAM RESIDENTIAL HEATING RATE: CLASSIFICATION NO. R-4

Availability

This rate is for residential use for those domestic customers who use gas as the principal household heating fuel if any member of the household qualifies for a benefit through one of the programs listed below, subject to the qualification period described under the "Terms and Conditions" of this rate. Availability is limited to use in domestic locations which are separately metered and billed and which are served by the Company's mains and for which the Company facilities are adequate.

Qualified Programs:

- a. Low Income Home Energy Assistance Program (LIHEAP)
- b. Electric Assistance Program (EAP)
- c. Supplemental Security Income Program
- d. Women, Infants and Children Program
- e. Commodity Surplus Foods Program (for women, infants and children)
- f. Elderly Commodity Surplus Foods Program
- g. Temporary Aid to Needy Families Program
- h. Housing Choice Voucher Program (also known as Section 8)
- i. Head Start Program
- j. Aid to the Permanently and Totally Disabled Program
- k. Aid to the Needy Blind Program
- 1. Old Age Assistance Program
- m. Food Stamps Program
- n. Any successor program of a-m

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Winter Customer Charge Per Meter: \$0.2823 per day or \$8.47 per 30 day month

Winter Period: All therms per 30 day month at \$0.3292 per therm

Summer Customer Charge Per Meter: \$0.5130 per day or \$15.39 per 30 day month

Summer Period: All therms per 30 day month at \$0.5985 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive.

Cost of Gas Charge

All gas delivered under this rate is subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Docket No. DG 21-130 Exhibit 26 Page 106 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 52 Residential G Heating Rate R-4

charges presented above are exclusive of these charges. Refer to the Firm Rate Schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

For those customers qualifying for the program this rate R-4 shall apply for a six month winter period. On the date that the six month winter period expires, eligibility for this rate shall expire unless the customer provides the Company with evidence that the customer continues to be eligible for one or more qualifying programs. When the Rate R-4 expires, the rate on each account shall revert back to the non-gas assistance program Residential Heating Rate, R-3. Customers whose eligibility for the program is based on their having qualified for LIHEAP shall be eligible for this rate retroactive to November 1 of the heating season in which they qualified. Eligibility for such customers shall expire the following April 30, subject to their re-qualifying through receipt of LIHEAP or other benefits as set forth above.

Eligibility shall be determined based on the reasonable discretion of the Company subject to verification of heating usage.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent ($1\frac{1}{2}$ %) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 107 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 53 MEP Residential Non-Heating Rate R-5

4 MANAGED EXPANSION PROGRAM RESIDENTIAL NON-HEATING RATE: CLASSIFICATION NO. R-5

Availability

This rate is mandatory for customers taking service in a Managed Expansion Program project area who otherwise would have qualified for Residential Non Heating Rate R-1.

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter: \$0.6670 per day or \$20.01 per 30 day month

Winter Period: All therms per 30 day month at \$0.5666 per therm

Summer Period: All therms per 30 day month at \$0.5666 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Cost of Gas Charge

All gas delivered under this rate is subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charges presented above are exclusive of these charges. Refer to Page 92 of this Tariff for firm rate schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

Service under each Managed Expansion Program project will have a term of ten years. Customers initiating service under this rate must take service hereunder until ten years following the date that the first customer in the particular Managed Expansion Program project takes service. Once the term of service for a particular Managed Expansion Program project expires, customers will thereafter take service under Residential Non Heating Rate R-1.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

Docket No. DG 21-130 Exhibit 26 Page 108 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 54 MEP Residential Non-Heating Rate R-5

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

Docket No. DG 21-130 Exhibit 26 Page 109 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 55 MEP Residential Heating Rate R-6

5 MANAGED EXPANSION PROGRAM RESIDENTIAL HEATING RATE: CLASSIFICATION NO. R-6

Availability

This rate is mandatory for customers taking service in a Managed Expansion Program projects area who otherwise would have qualified for Residential Heating Rate R-3.

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter: \$0.6670 per day or \$20.01 per 30 day month

Winter Period: All therms per 30 day month at \$0.7780 per therm

Summer Period: All therms per 30 day month at \$0.7780 per therm The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Cost of Gas Charge

All gas delivered under this rate is subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to Page 92 of this Tariff for firm rate schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charges presented above are exclusive of these charges. Refer to the Firm Rate Schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

Eligibility shall be determined based on the reasonable discretion of the Company subject to verification of heating usage.

Service under each Managed Expansion Program project will have a term of ten years. Customers initiating service under this rate must take service hereunder until ten years following the date that the first customer in the particular Managed Expansion Program project takes service. Once the term of service for a particular Managed Expansion Program project expires, customers will thereafter take service under Residential Non Heating Rate R-3.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading an estimated bill will be rendered to the customer. Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman Neil Proudman Neil Proudman Neil Proudman TITLE: President</u>

Docket No. DG 21-130 Exhibit 26 Page 110 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 56 MEP Residential Heating Rate R-6

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 111 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 57 MEP Residential Gas Assistance Program Rate R-7

6 MANAGED EXPANSION PROGRAM GAS ASSISTANCE PROGRAM RESIDENTIAL HEATING RATE: CLASSIFICATION NO. R-7

Availability

This rate is mandatory for customers taking service in a Managed Expansion Program project area who otherwise would have qualified for Gas Assistance Program Residential Heating Rate R-4.

Qualified Programs:

- a. Low Income Home Energy Assistance Program (LIHEAP)
- b. Electric Assistance Program (EAP)
- c. Supplemental Security Income Program
- d. Women, Infants and Children Program
- e. Commodity Surplus Foods Program (for women, infants and children)
- f. Elderly Commodity Surplus Foods Program
- g. Temporary Aid to Needy Families Program
- h. Housing Choice Voucher Program (also known as Section 8)
- i. Head Start Program
- j. Aid to the Permanently and Totally Disabled Program
- k. Aid to the Needy Blind Program
- 1. Old Age Assistance Program
- m. Food Stamps Program
- n. Any successor program of a-m

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Winter Customer Charge Per Meter: \$.3670 per day or \$11.01 per 30 day month
Winter Period: All therms per 30 day month at \$0.4279 per therm
Summer Customer Charge Per Meter: \$.6670 per day or \$20.01 per 30 day month
Summer Period: All therms per 30 day month at \$0.7780 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive.

Cost of Gas Charge

All gas delivered under this rate is subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charges presented above are exclusive of these charges. Refer to Page 92 of this Tariff for firm rate schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 112 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 58 MEP Residential Gas Assistance Program Heating Rate R-7

when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

Service under each Managed Expansion Program project will have a term of ten years. Customers initiating service under this rate must take service hereunder until ten years following the date that the first customer in the particular Managed Expansion Program project takes service. Once the term of service for a particular Managed Expansion Program project expires, customers will thereafter take service under Gas Assistance Program Residential Heating Rate R-4.

For those customers qualifying for the program this rate R-7 shall apply for a six month winter period. On the date that the six month winter period expires, eligibility for this rate shall expire unless the customer provides the Company with evidence that the customer continues to be eligible for one or more qualifying programs. When the Rate R-7 expires, the rate on each account shall revert back to the non-gas assistance program Residential Heating Rate, R-6. Customers whose eligibility for the program is based on their having qualified for LIHEAP shall be eligible for this rate retroactive to November 1 of the heating season in which they qualified. Eligibility for such customers shall expire the following April 30, subject to their re-qualifying through receipt of LIHEAP or other benefits as set forth above.

Eligibility shall be determined based on the reasonable discretion of the Company subject to verification of heating usage.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 113 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 59 Commercial/Industrial Rate G-41

7 COMMERCIAL/INDUSTRIAL SERVICE: LOW ANNUAL USE, HIGH WINTER USE RATE CLASSIFICATION NO. G-41

Availability

This rate is available for commercial, industrial and public authority customers in locations served by the Company's mains and for which the Company's facilities are adequate. A customer receiving service under this rate must have annual usage less than or equal to 10,000 therms and a Winter Period usage greater than or equal to 67% of annual usage as determined by the Company's records and procedures.

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter: \$1.9850 per day or \$59.55 per 30 day month

Winter Period: First 100* therms per 30 day month at \$0.4848 per therm

All over 100 therms per 30 day month at \$0.3309 per therm

First 20* therms per 30 day month at \$0.4848 per therm

All over 20 therms per 30 day month at \$0.3309 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to Page 90 or 91 of this Tariff for firm rate schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charge presented above is exclusive of these charges. Refer to the Firm Rate Schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

U.S. Department of Labor Standard Industry Classification Codes will determine eligibility for this tariff.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

^{*}The number of therms billed in the first block will be calculated by multiplying the therms in the first block of the rate by a fraction the numerator of which is the number of days in the billing period and the denominator of which is 30.

Docket No. DG 21-130 Exhibit 26 Page 114 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 60 Commercial/Industrial Rate G-41

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 115 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 61 Commercial/Industrial Rate G-42

8 COMMERCIAL/INDUSTRIAL SERVICE: MEDIUM ANNUAL USE, HIGH WINTER USE RATE CLASSIFICATION NO. G-42

Availability

This rate is for commercial, industrial and public authority customers in locations served by the Company's mains and for which the Company's facilities are adequate. A customer receiving service under this rate must have annual usage greater than 10,000 therms and less than or equal to 100,000 therms and a Winter Period usage greater than or equal to 67% of annual usage as determined by the Company's records and procedures.

Character of Service

Natural gas or equivalent will be supplied at a heat content of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter: \$5.9537 per day or \$178.61 per 30 day month

Winter Period: First 1000* therms per 30 day month at \$0.4409 per therm

All over 1000 therms per 30 day month at \$0.2988 per therm

First 400* therms per 30 day month at \$0.4409 per therm

All over 400 therms per 30 day month at \$0.2988 per therm

*The number of therms billed in the first block will be calculated by multiplying the therms in the first block of the rate by a fraction the numerator of which is the number of days in the billing period and the denominator of which is 30.

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charges presented above are exclusive of these charges. Refer to Page 90 or 91 of this Tariff for firm rate schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 116 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 62 Commercial/Industrial Rate G-42

Terms and Conditions

Dual fuel customers may be required to sign annual contracts with minimum usage requirements in order to qualify for service under this tariff. U.S. Department of Labor Standard Industry Classification Codes will determine eligibility for this tariff.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 117 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 63 Commercial/Industrial Rate G-43

9 COMMERCIAL/INDUSTRIAL SERVICE: HIGH ANNUAL USE, HIGH WINTER USE RATE CLASSIFICATION NO. G-43

Availability

This rate is for commercial, industrial and public authority customers in locations served by the Company's mains and for which the Company's facilities are adequate. A customer receiving service under this rate must have annual usage greater than 100,000 therms and a Winter Period usage greater than or equal to 67% of annual usage as determined by the Company's records and procedures.

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet. Should the customer's consumption fail to meet the availability requirements for this rate, the customer's service will be transferred to the otherwise applicable tariff as described under the terms and conditions of this tariff.

Delivery Charge

Customer Charge Per Meter: \$25.5170 per day or \$765.51 per 30 day month

Winter Period: All therms per 30 day month at \$0.2717 per therm

Summer Period: All therms per 30 day month at \$0.1295 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to Page 90 or 91 of this Tariff for firm rate schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charges presented above are exclusive of these charges. Refer to the Firm Rate Schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

To be eligible for this service, a customer must sign a contract for a one year period, which contract shall include the authority for the Company to monitor the customer's continued qualification for this service. In the event that the customer fails to meet the eligibility criteria set forth in the availability section of this schedule based on a monthly evaluation employing the most recent twelve (12) month period, the Company may require that the customer be billed prospectively under an alternative rate subject to the terms of the customer's Service Agreement. The Service Agreement may contain limitations as to maximum hourly,

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 118 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 64 Commercial/Industrial Rate G-43

daily, or monthly consumption, provisions for charges for excess usage, and other terms and conditions of service.

The customer shall declare maximum seasonal demands and estimated seasonal volumes at the time application for service is made. These declarations shall be updated annually, by August 1.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 119 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 65 Commercial/Industrial Rate G-44

10 MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: LOW ANNUAL USE, HIGH WINTER USE RATE CLASSIFICATION NO. G-44

Availability

This rate is Mandatory for customers taking service in a Managed Expansion Program project area who otherwise would have qualified for Commercial/Industrial Rate G-41.

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter: \$2.5803 per day or \$77.41 per 30 day month

Winter Period: First 100* therms per 30 day month at \$0.6302 per therm

All over 100 therms per 30 day month at \$0.4302 per therm

First 20* therms per 30 day month at \$0.6302 per therm

All over 20 therms per 30 day month at \$0.4302 per therm

*The number of therms billed in the first block will be calculated by multiplying the therms in the first block of the rate by a fraction the numerator of which is the number of days in the billing period and the denominator of which is 30.

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charge presented above is exclusive of these charges. Refer to Page 92 of this Tariff for firm rate schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

U.S. Department of Labor Standard Industry Classification Codes will determine eligibility for this tariff.

Service under each Managed Expansion Program project will have a term of ten years. Customers initiating service under this rate must take service hereunder until ten years following the date that the first

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 120 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 66 Commercial/Industrial Rate G-44

customer in the particular Managed Expansion Program project takes service. Once the term of service for a particular Managed Expansion Program project expires, customers will thereafter take service under Commercial/Industrial Rate G-41.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman
EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 121 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 67 MEP Commercial/Industrial Rate G-45

11 MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: MEDIUM ANNUAL USE, HIGH WINTER USE RATE CLASSIFICATION NO. G-45

Availability

This rate is mandatory for customers taking service in a Managed Expansion Program project area who otherwise would have qualified for Commercial/Industrial Rate G-42.

Character of Service

Natural gas or equivalent will be supplied at a heat content of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter: \$7.7397 per day or \$232.19 per 30 day month

Winter Period: First 1000* therms per 30 day month at \$0.5732 per therm

All over 1000 therms per 30 day month at \$0.3884 per therm

First 400* therms per 30 day month at \$0.5732 per therm

All over 400 therms per 30 day month at \$0.3884 per therm

*The number of therms billed in the first block will be calculated by multiplying the therms in the first block of the rate by a fraction the numerator of which is the number of days in the billing period and the denominator of which is 30.

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charges presented above are exclusive of these charges. Refer to Page 92 of this Tariff for firm rate schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

Dual fuel customers may be required to sign annual contracts with minimum usage requirements in order to qualify for service under this tariff. U.S. Department of Labor Standard Industry Classification Codes will determine eligibility for this tariff.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 122 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 68 MEP Commercial/Industrial Rate G-45

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Service under each Managed Expansion Program project will have a term of ten years. Customers initiating service under this rate must take service hereunder until ten years following the date that the first customer in the particular Managed Expansion Program project takes service. Once the term of service for a particular Managed Expansion Program project expires, customers will thereafter take service under Commercial/Industrial Rate G-42.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 123 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 69 MEP Commercial/Industrial Rate G-46

12 MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: HIGH ANNUAL USE, HIGH WINTER USE RATE CLASSIFICATION NO. G-46

Availability

This rate is mandatory for customers taking service in a Managed Expansion Program project area who otherwise would have qualified for Commercial/Industrial Rate G-43.

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet. Should the customer's consumption fail to meet the availability requirements for this rate, the customer's service will be transferred to the otherwise applicable tariff as described under the terms and conditions of this tariff.

Delivery Charge

Customer Charge Per Meter: \$33.1720 per day or \$995.16 per 30 day month

Winter Period: All therms per 30 day month at \$0.3532 per therm

Summer Period: All therms per 30 day month at \$0.1684 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to Page 92 of this Tariff for firm rate schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charges presented above are exclusive of these charges. Refer to the Firm Rate Schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

To be eligible for this service, a customer must sign a contract for a one year period, which contract shall include the authority for the Company to monitor the customer's continued qualification for this service. In the event that the customer fails to meet the eligibility criteria set forth in the availability section of this schedule based on a monthly evaluation employing the most recent twelve (12) month period, the Company may require that the customer be billed prospectively under an alternative rate subject to the terms of the customer's Service Agreement. The Service Agreement may contain limitations as to maximum hourly, daily, or monthly consumption, provisions for charges for excess usage, and other terms and conditions of service.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 124 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 70 MEP Commercial/Industrial Rate G-46

Service under each Managed Expansion Program project will have a term of ten years. Customers initiating service under this rate must take service hereunder until ten years following the date that the first customer in the particular Managed Expansion Program project takes service. Once the term of service for a particular Managed Expansion Program project expires, customers will thereafter take service under Commercial/Industrial Rate G-43.

The customer shall declare maximum seasonal demands and estimated seasonal volumes at the time application for service is made. These declarations shall be updated annually, by August 1.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 125 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 71 Commercial/Industrial Rate G-51

13 COMMERCIAL/INDUSTRIAL SERVICE: LOW ANNUAL USE, LOW WINTER USE RATE CLASSIFICATION NO. G-51

Availability

This rate is for commercial, industrial and public authority customers in locations served by the Company's mains and for which the Company's facilities are adequate. A customer receiving service under this rate must have annual usage less than or equal to 10,000 therms and a Winter Period usage less than 67% of annual usage as determined by the Company's records and procedures.

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter: \$1.9857 per day or \$59.57 per 30 day month

Winter Period: First 100* therms per 30 day month at \$0.2920 per therm

All over 100 therms per 30 day month at \$0.1934 per therm

Summer Period: First 100* therms per 30 day month at \$0.2920 per therm

All over 100 therms per 30 day month at \$0.1934 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charges presented above are exclusive of these charges. Refer to Page 90 or 91 of this Tariff for firm rate schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is made in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00

Terms and Conditions

Eligibility shall be based on the reasonable discretion of the Company and subject to verification of heating usage. U.S. Department of Labor Standard Industry Classification Code will determine eligibility for this

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

^{*}The number of therms billed in the first block will be calculated by multiplying the therms in the first block of the rate by a fraction the numerator of which is the number of days in the billing period and the denominator of which is 30.

Docket No. DG 21-130 Exhibit 26 Page 126 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 72 Commercial/Industrial Rate G-51

tariff. Dual fuel customers may be required to sign annual contracts with minimum usage requirements in order to qualify for service under this tariff.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 127 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 73 Commercial/Industrial Rate G-52

14 COMMERCIAL/INDUSTRIAL SERVICE: MEDIUM ANNUAL USE, LOW WINTER USE RATE CLASSIFICATION NO. G-52

Availability

This rate is for commercial, industrial and public authority customers in locations served by the Company's mains and for which the Company's facilities are adequate. A customer receiving service under this rate must have annual usage greater than 10,000 therms and less than or equal to 100,000 therms and a Winter Period usage less than 67% of annual usage as determined by the Company's records and procedures.

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet. Should the customer's consumption fail to meet the availability requirements for this rate, the customer's service will be transferred to the otherwise applicable tariff as described under the terms and conditions of this tariff.

Delivery Charge

Customer Charge Per Meter: \$5.9497 per day or \$178.49 per 30 day month

Winter Period: First 1000* therms per 30 day month at \$0.2515 per therm

All over 1000 therms per 30 day month at \$0.1704 per therm

First 1000* therms per 30 day month at \$0.1846 per therm

All over 1000 therms per 30 day month at \$0.1087 per therm

*The number of therms billed in the first block will be calculated by multiplying the therms in the first block of the rate by a fraction the numerator of which is the number of days in the billing period and the denominator of which is 30.

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charge presented above is exclusive of these charges. Refer to Page 90 or 91 of this Tariff for firm rate schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 128 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 74 Commercial/Industrial Rate G-52

Terms and Conditions

To be eligible for this service, a customer must sign a contract for a one year period, which contract shall include the authority for the Company to monitor the customer's continued qualification for this service. In the event that the customer fails to meet the eligibility criteria set forth in the availability section of this schedule based on a monthly evaluation employing the most recent twelve (12) month period, the Company may require that the customer be billed prospectively under an alternative rate subject to the terms of the customer's Service Agreement. The Service Agreement may contain limitations as to maximum hourly, daily, or monthly consumption, provisions for charges for excess usage, and other terms and conditions of service.

The customer shall declare maximum seasonal demands and estimated seasonal volumes at the time application for service is made. These declarations shall be updated annually, by August 1.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 129 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 75 Commercial/Industrial Rate G-53

15 COMMERCIAL/INDUSTRIAL SERVICE: HIGH ANNUAL USE, LOAD FACTOR LESS THAN 90% RATE CLASSIFICATION NO. G-53

Availability

This rate is for commercial, industrial and public authority customers in locations served by the Company's mains and for which the Company's facilities are adequate. A customer receiving service under this rate must have annual usage greater than 100,000 therms, a Winter Period usage less than 67% of annual usage, and a 12 month average usage less than 90% of the average usage of December, January and February as determined by the Company's records and procedures.

Character of Service

Natural gas or equivalent will be supplied at a heat content value of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter: \$26.2957 per day or \$788.87 per 30 day month

Winter Period: All therms per 30 day month at \$0.1758 per therm

Summer Period: All therms per 30 day month at \$0.0875 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charge presented above is exclusive of these charges. Refer to Page 90 or 91 of this Tariff for firm rate schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

To be eligible for this service, a customer must sign a contract for a one year period, which contract shall include the authority for the Company to monitor the customer's continued qualification for this service. In the event that the customer fails to meet the eligibility criteria set forth in the availability section of this schedule based on a monthly evaluation employing the most recent twelve (12) month period, the Company may require that the customer be billed prospectively under an alternative rate subject to the terms of the customer's Service Agreement. The Service Agreement may contain limitations as to maximum hourly,

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 130 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 76 Commercial/Industrial Rate G-53

daily, or monthly consumption, provisions for charges for excess usage, and other terms and conditions of service.

The customer shall declare maximum seasonal demands and estimated seasonal volumes at the time application for service is made. These declarations shall be updated annually, by August 1.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 131 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 77 Commercial/Industrial Rate G-54

16 COMMERCIAL/INDUSTRIAL SERVICE: HIGH ANNUAL USE, LOAD FACTOR GREATER THAN 90% RATE CLASSIFICATION NO. G-54

Availability

This rate is for commercial, industrial and public authority customers in locations served by the Company's mains and for which the Company's facilities are adequate. A customer receiving service under this rate must have annual usage greater than 100,000 therms, a Winter Period usage less than 67% of annual usage, and a 12 month average usage greater than or equal to 90% of the average usage of December, January and February as determined by the Company's records and procedures.

Character of Service

Natural gas or equivalent will be supplied at a heat content value of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter: \$26.2957 per day or \$788.87 per 30 day month

Winter Period: All therms per 30 day month at \$0.0670 per therm

Summer Period: All therms per 30 day month at \$0.0374 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charge presented above is exclusive of these charges. Refer to Page 90 or 91 of this Tariff for firm rate schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

To be eligible for this service, a customer must sign a contract for a one year period, which contract shall include the authority for the Company to monitor the customer's continued qualification for this service. In the event that the customer fails to meet the eligibility criteria set forth in the availability section of this schedule based on a monthly evaluation employing the most recent twelve (12) month period, the Company may require that the customer be billed prospectively under an alternative rate subject to the terms of the customer's Service Agreement. The Service Agreement may contain limitations as to maximum hourly,

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 132 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 78 Commercial/Industrial Rate G-54

daily, or monthly consumption, provisions for charges for excess usage, and other terms and conditions of service.

The customer shall declare maximum seasonal demands and estimated seasonal volumes at the time application for service is made. These declarations shall be updated annually, by August 1.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent $(1\frac{1}{2}\%)$ per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 133 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 79 MEP Commercial/Industrial Rate G-55

17 MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: LOW ANNUAL USE, LOW WINTER USE RATE CLASSIFICATION NO. G-55

Availability

This rate is mandatory for customers taking service in a Managed Expansion Program project area who otherwise would have qualified for Commercial/Industrial Rate G-51.

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter: \$2.5803 per day or \$77.41 per 30 day month

Winter Period: First 100* therms per 30 day month at \$0.3796 per therm

All over 100 therms per 30 day month at \$0.2514 per therm

First 100* therms per 30 day month at \$0.3796 per therm

All over 100 therms per 30 day month at \$0.2514 per therm

*The number of therms billed in the first block will be calculated by multiplying the therms in the first block of the rate by a fraction the numerator of which is the number of days in the billing period and the denominator of which is 30.

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charges presented above are exclusive of these charges. Refer to Page 92 of this Tariff for firm rate schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is made in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00

Terms and Conditions

Eligibility shall be based on the reasonable discretion of the Company and subject to verification of heating usage. U.S. Department of Labor Standard Industry Classification Code will determine eligibility for this tariff. Dual fuel customers may be required to sign annual contracts with minimum usage requirements in order to qualify for service under this tariff.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 134 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 80 MEP Commercial/Industrial Rate G-55

Service under each Managed Expansion Program project will have a term of ten years. Customers initiating service under this rate must take service hereunder until ten years following the date that the first customer in the particular Managed Expansion Program project takes service. Once the term of service for a particular Managed Expansion Program project expires, customers will thereafter take service under Commercial/Industrial Rate G-51.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 135 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 81 MEP Commercial/Industrial Rate G-56

18 MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: MEDIUM ANNUAL USE, LOW WINTER USE RATE CLASSIFICATION NO. G-56

Availability

This rate is mandatory for customers taking service in a Managed Expansion Program project area who otherwise would have qualified for Commercial/Industrial Rate G-52.

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet. Should the customer's consumption fail to meet the availability requirements for this rate, the customer's service will be transferred to the otherwise applicable tariff as described under the terms and conditions of this tariff.

Delivery Charge

Customer Charge Per Meter: \$7.7397 per day or \$232.19 per 30 day month

Winter Period: First 1000* therms per 30 day month at \$0.3270 per therm

All over 1000 therms per 30 day month at \$0.2215 per therm

First 1000* therms per 30 day month at \$0.2400 per therm

All over 1000 therms per 30 day month at \$0.1413 per therm

*The number of therms billed in the first block will be calculated by multiplying the therms in the first block of the rate by a fraction the numerator of which is the number of days in the billing period and the denominator of which is 30.

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charge presented above is exclusive of these charges. Refer to Page 92 of this Tariff for firm rate schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

To be eligible for this service, a customer must sign a contract for a one year period, which contract shall include the authority for the Company to monitor the customer's continued qualification for this service. In

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 136 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10 11 GAS LIBERTY UTILITIES Original Page 82 MEP Commercial/Industrial Rate G-56

the event that the customer fails to meet the eligibility criteria set forth in the availability section of this schedule based on a monthly evaluation employing the most recent twelve (12) month period, the Company may require that the customer be billed prospectively under an alternative rate subject to the terms of the customer's Service Agreement. The Service Agreement may contain limitations as to maximum hourly, daily, or monthly consumption, provisions for charges for excess usage, and other terms and conditions of service.

Service under each Managed Expansion Program project will have a term of ten years. Customers initiating service under this rate must take service hereunder until ten years following the date that the first customer in the particular Managed Expansion Program project takes service. Once the term of service for a particular Managed Expansion Program project expires, customers will thereafter take service under Commercial/Industrial Rate G-52.

The customer shall declare maximum seasonal demands and estimated seasonal volumes at the time application for service is made. These declarations shall be updated annually, by August 1.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent ($1\frac{1}{2}$ %) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 137 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 83 MEP Commercial/Industrial Rate G-57

19 MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: HIGH ANNUAL USE, LOAD FACTOR LESS THAN 90% RATE CLASSIFICATION NO. G-57

Availability

This rate is mandatory for customers taking service in a Managed Expansion Program project area who otherwise would have qualified for Commercial/Industrial Rate G-53.

Character of Service

Natural gas or equivalent will be supplied at a heat content value of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter: \$34.1843 per day or \$1,025.53 per 30 day month

Winter Period: All therms per 30 day month at \$0.2286 per therm

Summer Period: All therms per 30 day month at \$0.1138 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to Page 92 of this Tariff for firm rate schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charge presented above is exclusive of these charges. Refer to the Firm Rate Schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

To be eligible for this service, a customer must sign a contract for a one year period, which contract shall include the authority for the Company to monitor the customer's continued qualification for this service. In the event that the customer fails to meet the eligibility criteria set forth in the availability section of this schedule based on a monthly evaluation employing the most recent twelve (12) month period, the Company may require that the customer be billed prospectively under an alternative rate subject to the terms of the customer's Service Agreement. The Service Agreement may contain limitations as to maximum hourly, daily, or monthly consumption, provisions for charges for excess usage, and other terms and conditions of service.

Service under each Managed Expansion Program project will have a term of ten years. Customers initiating service under this rate must take service hereunder until ten years following the date that the first customer

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 138 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 84 MEP Commercial/Industrial Rate G-57

in the particular Managed Expansion Program project takes service. Once the term of service for a particular Managed Expansion Program project expires, customers will thereafter take service under Commercial/Industrial Rate G-53.

The customer shall declare maximum seasonal demands and estimated seasonal volumes at the time application for service is made. These declarations shall be updated annually, by August 1.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 139 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 85 MEP Commercial/Industrial Rate G-58

20 MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: HIGH ANNUAL USE, LOAD FACTOR GREATER THAN 90% RATE CLASSIFICATION NO. G-58

Availability

This rate is mandatory for customers taking service in a Managed Expansion Program project area who otherwise would have qualified for Commercial/Industrial Rate G-54.

Character of Service

Natural gas or equivalent will be supplied at a heat content value of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter: \$34.1843 per day or \$1,025.53 per 30 day month

Winter Period: All therms per 30 day month at \$0.0871 per therm

Summer Period: All therms per 30 day month at \$0.0487 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charge presented above is exclusive of these charges. Refer to Page 92 of this Tariff for firm rate schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

To be eligible for this service, a customer must sign a contract for a one year period, which contract shall include the authority for the Company to monitor the customer's continued qualification for this service. In the event that the customer fails to meet the eligibility criteria set forth in the availability section of this schedule based on a monthly evaluation employing the most recent twelve (12) month period, the Company may require that the customer be billed prospectively under an alternative rate subject to the terms of the customer's Service Agreement. The Service Agreement may contain limitations as to maximum hourly, daily, or monthly consumption, provisions for charges for excess usage, and other terms and conditions of service.

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 140 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 86 Outdoor Gas Lighting

in the particular Managed Expansion Program project takes service. Once the term of service for a particular Managed Expansion Program project expires, customers will thereafter take service under Commercial/Industrial Rate G-54.

The customer shall declare maximum seasonal demands and estimated seasonal volumes at the time application for service is made. These declarations shall be updated annually, by August 1.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

21 OUTDOOR GAS LIGHTING

Availability

This rate is available for residential outdoor gas lighting where such service is provided from the Company's existing delivery system to a standard gas light fixture or fixtures, located on the customer's premises, and when it is not feasible to meter such service along with other gas used on the premises and bill the same under the rate in effect for all other services. Service under this rate is available at those locations which were receiving service hereunder as of July 1, 2015, and which have continuously received service hereunder since that date.

Rate Per Light Per Month

\$12.81

The above rates shall be adjusted to reflect the recovery of all applicable taxes.

Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a location, an account charge is incurred in addition to all other charges. The account charge is \$20.00 when the visit to the location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

Meters are read and bills are presented monthly.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 87 Firm Rate Schedule

22 FIRM RATE SCHEDULES - EXCLUDING KEENE CUSTOMERS

	Rates effective November 01, 2021 - April 30, 2022 Winter Period				2	Rates Effective August 1, 2021 - October 31, 2021 Summer Period				
	Delivery <u>Charge</u>	Cost of Gas Rate Page 96	LDAC Page 102	2	Total <u>Rate</u>	Delivery <u>Charge</u>	Cost of Gas Rate Page 93	LDAC Page 102	Total <u>Rate</u>	
Residential Non Heating - R-1 Customer Charge per Month per Meter All Therms	\$ 15.39 \$ 0.4358	\$ 0.6050	n \$ 0.05	\$ 89 \$	15.39 1.0997	\$ 15.39 \$ 0.4358	\$ 0.3935	\$ 0.0589	\$ 15.39 \$ 0.8882	
7 til Homo	ψ 0.4000	ψ 0.0000	, ψ 0.000	υυ ψ	1.0007	ψ 0.4000	ψ 0.0000	ψ 0.0000	Ψ 0.0002	
Residential Heating - R-3 Customer Charge per Month per Meter	\$ 15.39			\$	15.39	\$ 15.39			\$ 15.39	
Size of the first block All Therms	All Therms \$ 0.5985	\$ 0.6050	0.05	89 \$	1.2624	All Therms \$ 0.5985	\$ 0.3935	\$ 0.0589	\$ 1.0509	
Residential Heating - R-4 Customer Charge per Month per Meter Size of the first block	\$ 8.47 All Therms			\$	8.47	\$ 15.39 All Therms			\$ 15.39	
All Therms	\$ 0.3292	\$ 0.3328	3 \$ 0.058	89 \$	0.7209	\$ 0.5985	\$ 0.3935	\$ 0.0589	\$ 1.0509	
Operation and a Miller discretization (C. 44)										
Commercial/Industrial - G-41 Customer Charge per Month per Meter	\$ 59.55			\$	59.55	\$ 59.55			\$ 59.55	
Size of the first block Therms in the first block per month at	100 therms \$ 0.4848		1 \$ 0.05	55 \$	1.1434	20 therms \$ 0.4848		\$ 0.0555	\$ 0.9289	
All therms over the first block per month at	\$ 0.3309	\$ 0.603	1 \$ 0.05	55 \$	0.9895	\$ 0.3309	\$ 0.3886	\$ 0.0555	\$ 0.7750	
·										
Commercial/Industrial - G-42	. 470.04			•	470.04	6 470.04			6 470.04	
Customer Charge per Month per Meter Size of the first block	\$ 178.61 1000 therms			\$	178.61	\$ 178.61 400 therms			\$ 178.61	
Therms in the first block per month at	\$ 0.4409	\$ 0.603	1 \$ 0.05	55 \$	1.0995	\$ 0.4409	\$ 0.3886	\$ 0.0555	\$ 0.8850	
All therms over the first block per month at	\$ 0.2988	\$ 0.603	1 \$ 0.05	55 \$	0.9574	\$ 0.2988	\$ 0.3886	\$ 0.0555	\$ 0.7429	
Commercial/Industrial - G-43 Customer Charge per Month per Meter	\$ 765.51			\$	765.51	\$ 765.51			\$ 765.51	
All therms over the first block per month at	\$ 0.2717	\$ 0.603	1 \$ 0.05	55 \$	0.9303	\$ 0.1295	\$ 0.3886	\$ 0.0555		
Commercial/Industrial - G-51 Customer Charge per Month per Meter	\$ 59.57			\$	59.57	\$ 59.57			\$ 59.57	
Size of the first block Therms in the first block per month at	100 therms \$ 0.2920		a & 0.050	55 \$	0.9614	100 therms \$ 0.2920		\$ 0.0555	\$ 0.7474	
•										
All therms over the first block per month at	\$ 0.1934	\$ 0.6139	9 \$ 0.05	55 \$	0.8628	\$ 0.1934	\$ 0.3999	\$ 0.0555	\$ 0.6488	
Commercial/Industrial - G-52										
Customer Charge per Month per Meter	\$ 178.49 1000 therms			\$	178.49	\$ 178.49 1000 therms			\$ 178.49	
Size of the first block Therms in the first block per month at	\$ 0.2515		9 \$ 0.05	55 \$	0.9209	\$ 0.1846		\$ 0.0555	\$ 0.6400	
All therms over the first block per month at	\$ 0.1704	\$ 0.6139	9 \$ 0.05	55 \$	0.8398	\$ 0.1087	\$ 0.3999	\$ 0.0555	\$ 0.5641	
Commercial/Industrial - G-53 Customer Charge per Month per Meter	\$ 788.87			\$	788.87	\$ 788.87			\$ 788.87	
All therms over the first block per month at	\$ 0.1758	\$ 0.6139	9 \$ 0.05	55 \$	0.8452	\$ 0.0875	\$ 0.3999	\$ 0.0555		
Commercial/Industrial - G-54										
Customer Charge per Month per Meter	\$ 788.87			\$	788.87	\$ 788.87			\$ 788.87	
All therms over the first block per month at	\$ 0.0670	\$ 0.6139	9 \$ 0.05	55 \$	0.7364	\$ 0.0374	\$ 0.3999	\$ 0.0555	\$ 0.4928	

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

NHPUC NO. 11 GAS LIBERTY UTILITIES

Customer Charge per Month per Meter All therms over the first block per month at

Commercial/Industrial - G-54
Customer Charge per Month per Meter
All therms over the first block per month at

Original Page 88 Firm Rate Schedule

Rates Effective August 1, 2021 - October 31, 2021

23 FIRM RATE SCHEDULES - KEENE CUSTOMERS

Rates effective November 01, 2021 - April 30, 2022

Winter Period Period Summer Period Cost of Cost of Delivery Gas Rate LDAC Total Delivery Gas Rate LDAC Total Charge Page 98 Page 102 Rate Charge Page 94 Page 102 Rate Residential Non Heating - R-1 Customer Charge per Month per Meter 15.39 15.39 15.39 15.39 All Therms 0.4358 0.9492 0.0589 \$ 1.4439 0.4358 1.2665 \$ 0.0589 1.7612 Residential Heating - R-3 Customer Charge per Month per Meter \$ 15.39 \$ 15.39 \$ 15.39 \$ 15.39 Size of the first block All Therms All Therms 0.5985 \$ 0.9492 \$ 0.0589 \$ 1.2665 \$ 0.0589 \$ 1.6066 0.5985 \$ 1.9239 \$ \$ Residential Heating - R-4 Customer Charge per Month per Meter 8.47 \$ 8.47 15.39 15.39 All Therms Size of the first block All Therms \$ 0.3292 \$ 0.9492 \$ 0.0589 \$ 1 3373 \$ 0.5985 \$ 1 2665 \$ 0.0589 \$ 1 9239 Commercial/Industrial - G-41 59.55 59.55 59.55 Customer Charge per Month per Meter \$ \$ 59.55 \$ \$ Size of the first block 100 therms 20 therms 0.4848 \$ Therms in the first block per month at \$ 0.4848 \$ 0.9492 \$ 0.0555 \$ 1.4895 \$ 1.2665 \$ 0.0555 \$ 1.8068 All therms over the first block per month at 0.3309 \$ 0.9492 \$ 0.0555 \$ 1.2665 \$ 0.0555 \$ 1.3356 0.3309 \$ 1.6529 Commercial/Industrial - G-42 178.61 178.61 178.61 178.61 Customer Charge per Month per Meter Size of the first block 1000 therms 400 therms 0.9492 \$ 0.0555 \$ \$ 0.0555 \$ Therms in the first block per month at \$ 0.4409 \$ 1.4456 0.4409 \$ 1.2665 \$ 1.7629 All therms over the first block per month at \$ 0.2988 \$ 0.9492 \$ 0.0555 \$ 1.3035 0.2988 \$ 1.2665 \$ 0.0555 \$ 1.6208 Commercial/Industrial - G-43 Customer Charge per Month per Meter 765.51 765.51 765.51 765.51 All therms \$ 0.2717 \$ 0.9492 \$ 0.0555 \$ 1.2764 \$ 0.1295 \$ 1.2665 \$ 0.0555 \$ 1.4515 Commercial/Industrial - G-51 Customer Charge per Month per Meter \$ 59.57 \$ 59.57 \$ 59.57 \$ 59.57 Size of the first block 100 therms 100 therms Therms in the first block per month at 0.2920 \$ 0.9492 \$ 0.0555 \$ 1 2967 0.2920 \$ 1.2665 \$ 0.0555 \$ 1.6140 All therms over the first block per month at 0.1934 \$ 0.9492 \$ 0.0555 \$ 0.0555 \$ Commercial/Industrial - G-52 Customer Charge per Month per Meter 178.49 \$ 178.49 \$ 178.49 \$ 178.49 Size of the first block 1000 therms 1000 therms 0.9492 \$ 0.0555 \$ 1.2665 \$ 0.0555 \$ 1.5066 Therms in the first block per month at \$ 0.2515 \$ 1.2562 \$ 0.1846 \$ All therms over the first block per month at \$ 0.1704 \$ 0.9492 \$ 0.0555 \$ 1.1751 \$ 0.1087 \$ 1.2665 \$ 0.0555 \$ 1.4307 Commercial/Industrial - G-53

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

EFFECTIVE: August 1, 2021 TITLE: President

788.87

\$

\$

0.1758 \$

0.0670 \$

0.9492 \$

0.9492 \$

0.0555

0.0555 \$

\$

788.87

1.1805

1.0717

788.87

0.0875 \$

0.0374 \$

788.87

1.4095

1.3594

0.0555 \$

0.0555 \$

1.2665 \$

1.2665 \$

Original Page 89 Firm Rate Schedule

24 FIRM RATE SCHEDULES - MANAGED EXPANSION PROGRAM-EXCLUDING KEENE CUSTOMERS

Potos offostius November 01 2021 April 20 2022

	Rates effective November 01, 2021 - April 30, 2022 Winter Period				2022	Rates Effective August 1, 2021 - October 31, 2021 Summer Period			
	Delivery <u>Charge</u>	Cost of Gas Rate <u>Page 96</u>	LDAC Page 102		Total <u>Rate</u>	Delivery <u>Charge</u>	Cost of Gas Rate Page 93	LDAC Page 102	Total <u>Rate</u>
Residential Non Heating - R-5									
Customer Charge per Month per Meter All Therms	\$ 20.01 \$ 0.5666		0.0589	\$	20.01 1.2305	\$ 20.01 \$ 0.5666	\$ 0.3935	\$ 0.0589	\$ 20.01 \$ 1.0190
Residential Heating - R-6 Customer Charge per Month per Meter	\$ 20.01			\$	20.01	\$ 20.01			\$ 20.01
Size of the first block Therms in the first block per month at	all therms \$ 0.7780	\$ 0.6050	\$ 0.0589	\$	1.4419	all therms \$ 0.7780	\$ 0.3935	\$ 0.0589	\$ 1.2304
Residential Heating - R-7	\$ 11.01			\$	11.01	\$ 20.01			\$ 20.01
Customer Charge per Month per Meter Size of the first block	all therms			Ъ	11.01	all therms			\$ 20.01
Therms in the first block per month at	\$ 0.4279	\$ 0.6050	0.0589	\$	1.0918	\$ 0.7780	\$ 0.3935	\$ 0.0589	\$ 1.2304
Commercial/Industrial - G-44 Customer Charge per Month per Meter	\$ 77.41			\$	77.41	\$ 77.41			\$ 77.41
Size of the first block	100 therms	s				20 therms	3		
Therms in the first block per month at	\$ 0.6302	9 \$ 0.6031	\$ 0.0555	5 \$	1.2888	\$ 0.6302	\$ 0.3886	\$ 0.0555	\$ 1.0743
All therms over the first block per month at	\$ 0.4302	9 \$ 0.6031	\$ 0.0555	\$	1.0888	\$ 0.4302	\$ 0.3886	\$ 0.0555	\$ 0.8743
Commercial/Industrial - G-45 Customer Charge per Month per Meter	\$ 232.19	1		\$	232.19	\$ 232.19			\$ 232.19
Size of the first block	1000 therms			Ψ	202.10	400 therms			Ψ 202.10
Therms in the first block per month at	\$ 0.5732	9 \$ 0.6031	\$ 0.0555	\$	1.2318	\$ 0.5732	\$ 0.3886	\$ 0.0555	\$ 1.0173
All therms over the first block per month at	\$ 0.3884	\$ 0.6031	\$ 0.0555	\$	1.0470	\$ 0.3884	\$ 0.3886	\$ 0.0555	\$ 0.8325
Commercial/Industrial - G-46	\$ 995.16			\$	995.16	\$ 995.16			\$ 995.16
Customer Charge per Month per Meter All therms over the first block per month at	\$ 0.3532		\$ 0.0555		1.0118	\$ 0.1684		\$ 0.0555	
Commercial/Industrial - G-55						•			
Customer Charge per Month per Meter Size of the first block	\$ 77.41 100 therms			\$	77.41	\$ 77.41 100 therms			\$ 77.41
Therms in the first block per month at	\$ 0.3796	\$ 0.6139	\$ 0.0555	\$	1.0490	\$ 0.3796		\$ 0.0555	\$ 0.8350
All therms over the first block per month at	\$ 0.2514	\$ 0.6139	\$ 0.0555	5 \$	0.9208	\$ 0.2514	\$ 0.3999	\$ 0.0555	\$ 0.7068
Commercial/Industrial - G-56					000.40				
Customer Charge per Month per Meter Size of the first block	\$ 232.19 1000 therms			\$	232.19	\$ 232.19 1000 therms			\$ 232.19
Therms in the first block per month at	\$ 0.3270	\$ 0.6139	\$ 0.0555	\$	0.9964	\$ 0.2400	\$ 0.3999	\$ 0.0555	\$ 0.6954
All therms over the first block per month at	\$ 0.2215	\$ 0.6139	\$ 0.0555	5 \$	0.8909	\$ 0.1413	\$ 0.3999	\$ 0.0555	\$ 0.5967
Commercial/Industrial - G-57 Customer Charge per Month per Meter	\$ 1,025.53			\$	1,025.53	\$ 1,025.53			\$ 1,025.53
All therms over the first block per month at	\$ 1,025.53		\$ 0.0555		0.8980	\$ 1,025.53		\$ 0.0555	Ψ 1,020.00
Commercial/Industrial - G-58 Customer Charge per Month per Meter									
All therms over the first block per month at	\$ 1,025.53 \$ 0.0871		\$ 0.0555	\$	1,025.53 0.7565	\$ 1,025.53 \$ 0.0487		\$ 0.0555	\$ 1,025.53 \$ 0.5041

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman President

TITLE:

Docket No. DG 21-130 Exhibit 26 Page 144 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 90 Firm Rate Schedule Firm Rate Schedule

25 FIRM RATE SCHEDULES - OUTDOOR GAS LIGHTING

Outdoor Gas Lighting					
Per Light Per Month	\$12.81				

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman Neil Proudman

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 91 Summer Cost of Gas

26 ANTCIPATED SUMMER PERIOD COST OF GAS EXCLUDING KEENE CUSTOMERS OR GAS LIGHTING

Anticipated Cost of Gas

PERIOD COVERED: SUMMER PERIOD, MAY 1, 2021 THROUGH OCTOBER 31, 2021 (REFER TO TEXT IN SECTION 16 COST OF GAS CLAUSE)

(Col 1)	(Col 2)	(Col 3)
ANTICIPATED DIRECT COST OF GAS		
Purchased Gas:		
Demand Costs:	\$ 2,868,280	
Supply Costs:	4,387,278	
Storage Gas:		
Demand, Capacity:	\$ -	
Commodity Costs:	-	
Produced Gas:	\$ 29,014	
Hedged Contract (Savings)/Loss	<u>\$ -</u>	
Unadjusted Anticipated Cost of Gas		\$ 7,284,571
Adjustments:		
Prior Period (Over)/Under Recovery (as of April 30, 2019)	\$ 105,886	
Interest	(3,492))
Prior Period Adjustments	-	
Broker Revenues	-	
Refunds from Suppliers	-	
Fuel Financing	_	
Transportation CGA Revenues	_	
Interruptible Sales Margin	_	
Capacity Release Margin	_	
Hedging Costs	-	
Fixed Price Option Administrative Costs	-	
Total Adjustments		102,394
Total Anticipated Direct Cost of Gas		\$ 7,386,965
Anticipated Indirect Cost of Gas		
Working Capital:		
Total Unadjusted Anticipated Cost of Gas 05/01/19 - 10/31/19)	\$ 7,284,571	
Working Capital Rate - Lead Lag Days / 365	0.0391	
Prime Rate	3.25%	
Working Capital Percentage	0.127%	
Working Capital	\$ 9,258	
Plus: Working Capital Reconciliation (Acct 1163-1424)	(13,709))
Total Working Capital Allowance		\$ (4,451)
Bad Debt:		
Total Unadjusted Anticipated Cost of Gas 05/01/19 - 10/31/19)	\$ 7,284,571	
Less: Refunds		
Plus: Total Working Capital	(4,451))
Plus: Prior Period (Over)/Under Recovery	105,886	
Subtotal	\$ 7,386,006	
Bad Debt Percentage	1.11%	
Bad Debt Allowance	\$ 81,985	
Plus: Bad Debt Reconciliation (Acct 1163-1754)	(326,326)	
	(020,020)	
Total Bad Debt Allowance		(244,341)
Production and Storage Capacity		-
Miscellaneous Overhead (05/01/19 - 10/31/19)	\$ 13,170	
Times Summer Sales	22,004	
Divided by Total Sales	111,369	
Miscellaneous Overhead		2,602
Total Anticipated Indirect Cost of Gas		\$ (246,190)
Total Cost of Gas		\$ 7,140,775
		Ç 1,140,175

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman

EFFECTIVE: August 1, 2021 TITLE: President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 92 Summer Cost of Gas

27 CALCULATION OF FIRM SALES SUMMER PERIOD COST OF GAS RATE **EXCLUDING KEENE CUSTOMERS**

CALCULATION OF FIRM SALES COST OF GAS RATE
PERIOD COVERED: SUMMER PERIOD, MAY 1, 2021 THROUGH OCTOBER 31, 2021
[Pater to Tay in Section 15 Cost of Gas Clause].

(Refer to Text in S	ection 16 Cost	of Gas Clause)			
(Col 1)		(C	ol 2)	(Col 3)	
Total Anticipated Direct Cost of Gas		\$	7.386.966		
Projected Prorated Sales (05/01/20 - 10/31/20)			2,681,422		
Direct Cost of Gas Rate			:	\$ 0.3257	per therm
Demand Cost of Gas Rate		\$	2.868.280	\$ 0.1265	per therm
Commodity Cost of Gas Rate			4,416,292	\$ 0.1947	per therm
Adjustment Cost of Gas Rate		-			per therm
Total Direct Cost of Gas Rate		\$	7,386,966	\$ 0.3257	per therm
Total Anticipated Indirect Cost of Gas		\$	(246, 190)		
Projected Prorated Sales (05/01/20 - 10/31/20)		2	2,681,422		
Indirect Cost of Gas			:	\$ (0.0109	9) per therm
TOTAL PERIOD AVERAGE COST OF GAS EFFECTIVE 05/01/21			:	\$ 0.3148	per therm
RESIDENTIAL COST OF GAS RATE - 05/01/2021 Approved in Order No 26,419 in DG 20-1-	41	COGsr		\$ 0.3148	3 /therm
Change in rate due to change in under/over recovery				\$ 0.0787	
RESIDENTIAL COST OF GAS RATE - 05/01/2021		COGsr		\$ 0.3935 \$ -	/therm
Change in rate due to change in under/over recovery RESIDENTIAL COST OF GAS RATE - 06/01/2021		COGsr		•	5 /therm
Change in rate due to change in under/over recovery				\$ -	
RESIDENTIAL COST OF GAS RATE - 07/01/2021		COGsr			/therm
Change in rate due to change in under/over recovery RESIDENTIAL COST OF GAS RATE - 08/01/2021		COGsr		\$ - \$ 0.3935	/therm
Change in rate due to change in under/over recovery		COGSF		\$ 0.3938 \$ -	, ,uieiiii
RESIDENTIAL COST OF GAS RATE - 09/01/2021		COGsr		\$ 0.3935	5 /therm
Change in rate due to change in under/over recovery RESIDENTIAL COST OF GAS RATE - 10/01/2021		COGsr		\$ - \$ 0.3935	5 /therm
1202/101 OR OR OR OR OR OR OR O		COGSF		. J.353	, ruiellii
		Maximum (COG +	25%)	\$ 0.3935	5
COM/IND LOW WINTER USE COST OF GAS RATE - 05/01/2021 Approved in Order No 26,	419 in DG 20-1	41 COGsl		\$ 0.3199) /therm
Change in rate due to change in under/over recovery				\$ 0.0800)
COM/IND LOW WINTER USE COST OF GAS RATE - 05/01/2021		COGsl			/therm
Change in rate due to change in under/over recovery COM/IND LOW WINTER USE COST OF GAS RATE - 06/01/2021		COGsl		\$ - \$ 0.3999) /therm
Change in rate due to change in under/over recovery				\$ -	
COM/IND LOW WINTER USE COST OF GAS RATE - 07/01/2021 Change in rate due to change in under/over recovery		COGsl		\$ 0.3999 \$ -	/therm
COM/IND LOW WINTER USE COST OF GAS RATE - 08/01/2021		COGsl			/therm
		0000			
Change in rate due to change in under/over recovery COM/IND LOW WINTER USE COST OF GAS RATE - 09/01/2021		COGsl		\$ -	therm
COM/IND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Change in rate due to change in under/over recovery		COGsl		\$ - \$ 0.3999 \$ -) /therm
COM/IND LOW WINTER USE COST OF GAS RATE - 09/01/2021				\$ - \$ 0.3999 \$ -	
COMIND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Change in rate due to change in under/over recovery COM/IND LOW WINTER USE COST OF GAS RATE - 10/01/2021		COGsl		\$ - \$ 0.3999 \$ -) /therm
COM/IND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Change in rate due to change in under/over recovery COM/IND LOW WINTER USE COST OF GAS RATE - 10/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20	\$ 0.1265	COGsl		\$ - \$ 0.3999 \$ - \$ 0.3999) /therm
COM/IND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Change in rate due to change in under/over recovery COM/IND LOW WINTER USE COST OF GAS RATE - 10/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 "Times: Low Winter Use Ratio (Summer)	1.0620	COGsl		\$ - \$ 0.3999 \$ -) /therm
COM/IND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Change in rate due to change in under/over recovery COM/IND LOW WINTER USE COST OF GAS RATE - 10/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20		COGsl		\$ - \$ 0.3999 \$ - \$ 0.3999) /therm
COM/IND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Change in rate due to change in under/over recovery COM/IND LOW WINTER USE COST OF GAS RATE - 10/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 "Times: Low Winter Use Ratio (Summer) Times: Correction Factor Adjusted Demand Cost of Gas Rate	1.0620 0.9798 \$ 0.1316	COGsl		\$ - \$ 0.3999 \$ - \$ 0.3999) /therm
COM/IND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Change in rate due to change in under/over recovery COM/IND LOW WINTER USE COST OF GAS RATE - 10/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 'Times: Low Winter Use Ratio (Summer) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate	1.0620 0.9798 \$ 0.1316 \$ 0.1947	COGsl		\$ - \$ 0.3999 \$ - \$ 0.3999) /therm
COMIND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Change in rate due to change in under/over recovery COM/IND LOW WINTER USE COST OF GAS RATE - 10/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 "Times: Low Winter Use Ratio (Summer) Times: Correction Factor Adjusted Demand Cost of Gas Rate	1.0620 0.9798 \$ 0.1316	COGsl		\$ - \$ 0.3999 \$ - \$ 0.3999) /therm
COMIND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Change in rate due to change in under/over recovery COM/IND LOW WINTER USE COST OF GAS RATE - 10/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 "Times: Low Winter Use Ratio (Summer) Times: Comerction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjustment Cost of Gas Rate	1.0620 0.9798 \$ 0.1316 \$ 0.1947 0.0045	COGsl		\$ - \$ 0.3999 \$ - \$ 0.3999) /therm
COM/IND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Change in rate due to change in under/over recovery COM/IND LOW WINTER USE COST OF GAS RATE - 10/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 "Times: Low Winter Use Ratio (Summer) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Com/Ind Low Winter Use Cost of Gas Rate	1.0620 0.9798 \$ 0.1316 \$ 0.1947 0.0045 (0.0109)	COGsl COGsl Maximum (COG +	25%)	\$ -0.39999 \$ -0.39999 \$ 0.39999	9 /therm 9 /therm
COM/IND LOW WINTER USE COST OF GAS RATE - 09/01/2021 COM/IND LOW WINTER USE COST OF GAS RATE - 10/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 "Times: Low Winter Use Ratio (Summer) Times: Low Winter Use Ratio (Summer) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjustment Cost of Gas Rate COM/IND HIGH WINTER USE COST OF GAS RATE - 05/01/2021	1.0620 0.9798 \$ 0.1316 \$ 0.1947 0.0045 (0.0109)	COGsl	: 225%)	\$ - 0.3999 \$ - 0.3999 \$ 0.3999 \$ 0.3999	9 /therm 9 /therm
COM/IND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 Average Demand Cost of Gas Rate Effective 05/01/20 Times: Low Winter Use Ratio (Summer) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Demand Cost of Gas Rate COM/IND HIGH WINTER USE COST OF GAS RATE - 05/01/2021 COM/IND HIGH WINTER USE COST OF GAS RATE - 05/01/2021	1.0620 0.9798 \$ 0.1316 \$ 0.1947 0.0045 (0.0109)	COGsl COGsl Maximum (COG +	225%)	\$ -3999 \$ 0.3999 \$ 0.3999 \$ 0.3999 \$ 0.3105 \$ 0.0777	9 /therm 9 /therm
COM/IND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Change in rate due to change in under/over recovery COM/IND LOW WINTER USE COST OF GAS RATE - 10/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 Times: Low Winter Use Ratio (Summer) Times: Low Winter Use Ratio (Summer) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Adjustment Cost of Gas Rate Adjusted Com/Ind Low Winter Use Cost of Gas Rate COM/IND HIGH WINTER USE COST OF GAS RATE - 05/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 05/01/2021	1.0620 0.9798 \$ 0.1316 \$ 0.1947 0.0045 (0.0109)	COGsl COGsl Maximum (COG + :	25%)	\$ 0.3999 \$ 0.3999 \$ 0.3999 \$ 0.3999 \$ 0.3108 \$ 0.0777 \$ 0.0777	i /therm
COMINID LOW WINTER USE COST OF GAS RATE - 09/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 Average Demand Cost of Gas Rate Effective 05/01/20 Times: Low Winter Use Ratio (Summer) Times: Comerction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Low Lost of Gas Rate Adjusted Low Winter Use Cost of Gas Rate Commodity Cost of Gas Rate Adjusted Low Winter Use Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Com/Ind Low Winter Use Cost of Gas Rate COM/IND HIGH WINTER USE COST OF GAS RATE - 05/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 05/01/2021 Change in rate due to change in under/over recovery	1.0620 0.9798 \$ 0.1316 \$ 0.1947 0.0045 (0.0109)	COGsl COGsl Maximum (COG + :	: : : : : : : : : : : : : : : : : : : :	\$ 0.3999 \$ 0.3999 \$ 0.3999 \$ 0.3109 \$ 0.0777 \$ 0.3886 \$ 0.3886	9 /therm 9 /therm 9 /therm 9 /therm
COMIND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Change in rate due to change in under/over recovery COM/IND LOW WINTER USE COST OF GAS RATE - 10/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 'Times: Low Winter Use Ratio (Summer) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate	1.0620 0.9798 \$ 0.1316 \$ 0.1947 0.0045 (0.0109)	COGsl COGsl Maximum (COG + :	225%)	\$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3105 \$ 0.3105 \$ 0.3865 \$ - 3886 \$ - 3886 \$ - 3886	i /therm
COM/IND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 Average Demand Cost of Gas Rate Effective 05/01/20 "Times: Low Winter Use Ratio (Summer) Times: Low Winter Use Ratio (Summer) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commotify Cost of Gas Rate Adjustment Fost of Gas Rate Adjustment Winter Use Cost of Gas Rate COM/IND HIGH WINTER USE COST OF GAS RATE - 05/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 07/01/2021	1.0620 0.9798 \$ 0.1316 \$ 0.1947 0.0045 (0.0109)	COGsl COGsl Maximum (COG + : COGsh COGsh COGsl	25%)	\$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3105 \$ 0.0777 \$ 0.3886 \$ - \$ 0.3885 \$ - \$ 0.3865 \$ -	3 /therm 3 /therm 3 /therm 5 /therm 7 / 5 /therm 5 /therm 5 /therm
COM/IND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 Average Demand Cost of Gas Rate Effective 05/01/20 Times: Low Winter Use Ratio (Summer) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Commodity Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Commodity Cost of Gas Rate Adjusted Commodity Cost of Gas Rate Adjusted Demand Cost of Gas Rate Commind High Winter USE COST Of Gas Rate - 08/01/2021 Change in rate due to change in under/over recovery COMIND HIGH WINTER USE COST Of Gas Rate - 08/01/2021 Change in rate due to change in under/over recovery COMIND HIGH WINTER USE COST Of Gas Rate - 08/01/2021	1.0620 0.9798 \$ 0.1316 \$ 0.1947 0.0045 (0.0109)	COGsl COGsl Maximum (COG + : COGsh COGsh	225%)	\$ 0.3999 \$ 0.3999 \$ 0.3999 \$ 0.3999 \$ 0.3999 \$ 0.3809 \$ 0.3886 \$ 0.3886 \$ 0.3886 \$ 0.3886 \$ 0.3886 \$ 0.3886	3 /therm 3 /therm 9 /therm 9 /therm 6 /therm 6 /therm
COM/IND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 Average Demand Cost of Gas Rate Effective 05/01/20 Times: Low Winter Use Ratio (Summer) Times: Low Winter Use Ratio (Summer) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Low	1.0620 0.9798 \$ 0.1316 \$ 0.1947 0.0045 (0.0109)	COGsl COGsl Maximum (COG + : COGsh COGsh COGsl	: : : : : : : : : : : : : : : : : : : :	\$ 0.3999 \$ 0.3999 \$ 0.3999 \$ 0.3999 \$ 0.3999 \$ 0.3899 \$ 0.3886 \$ - \$ 0.3886 \$ 0.3886	3 /therm 3 /therm 3 /therm 5 /therm 7 / 5 /therm 5 /therm 5 /therm
COMIND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 Average Demand Cost of Gas Rate Effective 05/01/20 Times: Low Winter Use Ratio (Summer) Times: Comerction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Commodity Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Demand Cow Winter Use Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Com/Ind Low Winter Use Cost of Gas Rate COM/IND HIGH WINTER USE COST OF GAS RATE - 05/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 07/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 07/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 07/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021	1.0620 0.9798 \$ 0.1316 \$ 0.1947 0.0045 (0.0109)	COGsl COGsl Maximum (COG+: COGsh COGsh COGsl COGsl	225%)	\$ 0.3999 \$ 0.3999 \$ 0.3999 \$ 0.3999 \$ 0.3999 \$ 0.3100 \$ 0.0777 \$ 0.3886 \$ - \$ 0.3886 \$ - \$ 0.3886 \$ - \$ 0.3886 \$ 5 - \$ 0.3886 \$ 0.3886 \$ 5 - \$ 0.3886 \$ 0	9 /therm 9 /therm 9 /therm 7 / 7 / 7 / 7 / 7 / 7 / 7 / 7 / 7 / 7 /
COMIND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 Average Demand Cost of Gas Rate Effective 05/01/20 Times: Low Winter Use Ratio (Summer) Times: Comerction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Commodity Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Low Winter Use Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Low Winter Use Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Low Winter Use Cost of Gas Rate Adjusted Com/Ind Low Winter Use Cost of Gas Rate COMIND HIGH WINTER USE COST OF GAS RATE - 05/01/2021 Change in rate due to change in under/over recovery COMIND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COMIND HIGH WINTER USE COST OF GAS RATE - 07/01/2021 Change in rate due to change in under/over recovery COMIND HIGH WINTER USE COST OF GAS RATE - 07/01/2021 Change in rate due to change in under/over recovery COMIND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Change in rate due to change in under/over recovery COMIND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Change in rate due to change in under/over recovery	1.0620 0.9798 \$ 0.1316 \$ 0.1947 0.0045 (0.0109)	COGsl COGsl Maximum (COG+ COGsh COGsh COGsl COGsl COGsl	225%)	\$ 0.3999 \$ 0.3999 \$ 0.3999 \$ 0.3999 \$ 0.3999 \$ 0.3100 \$ 0.0777 \$ 0.3886 \$ - \$ 0.3886 \$ - \$ 0.3886 \$ - \$ 0.3886 \$ 5 - \$ 0.3886 \$ 0.3886 \$ 5 - \$ 0.3886 \$ 0	9 /therm 9 /therm 9 /therm 7 / 5 /therm 5 /therm 5 /therm 5 /therm 6 /therm
COM/IND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 Average Demand Cost of Gas Rate Effective 05/01/20 Times: Low Winter Use Ratio (Summer) Times: Comerction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Commodity Cost of Gas Rate Adjusted Cost of Gas Rate Adjusted Cost of Gas Rate Adjusted Confort Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Confort Cost of Gas Rate Adjusted Confort Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Comfort Cost of Gas Rate Adjusted Comfort Cost of Gas Rate Adjusted Cost of Gas Rate COM/IND HIGH WINTER USE COST OF GAS RATE - 05/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021	1.0620 0.9798 \$ 0.1316 \$ 0.1947 0.0045 (0.0109) \$ 0.3199	COGsl COGsl Maximum (COG+ COGsh COGsh COGsl COGsl COGsl	225%)	\$ 0.3999 \$ 0.3999 \$ 0.3999 \$ 0.3999 \$ 0.3999 \$ 0.3100 \$ 0.0777 \$ 0.3886 \$ - \$ 0.3886 \$ - \$ 0.3886 \$ - \$ 0.3886 \$ 5 - \$ 0.3886 \$ 0.3886 \$ 5 - \$ 0.3886 \$ 0	9 /therm 9 /therm 9 /therm 7 / 5 /therm 5 /therm 5 /therm 5 /therm 6 /therm
COM/IND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 Average Demand Cost of Gas Rate Effective 05/01/20 "Times: Low Winter Use Ratio (Summer) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjustment Winter Use Cost of Gas Rate Indirect Cost of Gas Rate Adjustment Winter Use Cost of Gas Rate COM/IND HIGH WINTER USE COST OF GAS RATE - 05/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Common HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20	1.0620 0.9798 \$ 0.1316 \$ 0.1947 0.0045 (0.0109)	COGsl COGsl Maximum (COG + : COGsh COGsh COGsl COGsl COGsl	225%)	\$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3985 \$ 0.3885 \$ 0.3886	9 /therm 9 /therm 9 /therm 7 /therm 7 /therm 6 /therm 6 /therm 6 /therm 6 /therm 6 /therm 6 /therm
COM/IND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 Times: Low Winter Use Ratio (Summer) Times: Correction Factor Adjusted Demand Cost of Gas Rate Effective 05/01/20 Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Com/Ind Low Winter Use Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Com/Ind Low Winter Use Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Com/Ind Low Winter Use Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Com/Ind Low Winter Use Cost of Gas Rate COM/IND HIGH WINTER USE COST OF GAS RATE - 05/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 Times: Correction Factor Times: Correction Factor Times: Correction Factor	1 0620 0.9798 \$ 0.1316 \$ 0.1947 0.0045 (0.0109) \$ 0.3199 \$ 0.1265 0.9890 0.9798	COGsl COGsl Maximum (COG+ COGsh COGsh COGsl COGsl COGsl	225%)	\$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3985 \$ 0.3886 \$ 0.3886	9 /therm 9 /therm 9 /therm 7 /therm 7 /therm 6 /therm 6 /therm 6 /therm 6 /therm 6 /therm 6 /therm
COM/IND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 Average Demand Cost of Gas Rate Effective 05/01/20 Times: Low Winter Use Ratio (Summer) Times: Comection Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Commodity Cost of Gas Rate Adjusted Convolved Winter Use Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Convolved Winter Use Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Convolved Winter Use Cost of Gas Rate COM/IND HIGH WINTER USE COST OF GAS RATE - 05/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 Times: High Winter Use Ratio (Summer)	1.0620 0.9798 \$ 0.1316 \$ 0.1947 0.0045 (0.0109) \$ 0.3199	COGsl COGsl Maximum (COG + : COGsh COGsh COGsl COGsl COGsl	225%)	\$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3985 \$ 0.3885 \$ 0.3886	9 /therm 9 /therm 9 /therm 7 /therm 7 /therm 6 /therm 6 /therm 6 /therm 6 /therm 6 /therm 6 /therm
COM/IND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 Times: Low Winter Use Ratio (Summer) Times: Low Comment of Gas Rate Effective 05/01/20 Times: Low Winter Use Ratio (Summer) Times: Comerction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Demand Low Winter Use Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Demand Low Winter Use Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Demand Low Winter Use Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Com/Ind Low Winter Use Cost of Gas Rate COM/IND HIGH WINTER USE COST OF GAS RATE - 05/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 07/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 07/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 07/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 "Times: Correction Factor Adjusted Demand Cost of Gas Rate Effective 05/01/20 "Times: High Winter Use Ratio (Summer) Times: Correction Factor Adjusted Demand Cost of Gas Rate	1 0620 0.9798 \$ 0.1316 \$ 0.1947 0.0045 (0.0109) \$ 0.3199 \$ 0.1265 0.9899 \$ 0.1226	COGsl COGsl Maximum (COG + : COGsh COGsh COGsl COGsl COGsl	225%)	\$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3985 \$ 0.3885 \$ 0.3886	9 /therm 9 /therm 9 /therm 7 / fill /
COM/IND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 Times: Low Winter Use Ratio (Summer) Times: Low Comment of Gas Rate Effective 05/01/20 Times: Low Winter Use Ratio (Summer) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Commodity Cost of Gas Rate Adjusted Com/Ind Low Winter Use Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Com/Ind Low Winter Use Cost of Gas Rate COM/IND HIGH WINTER USE COST OF GAS RATE - 05/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 Times: Correction Factor Times: Correction Factor Times: Correction Factor	\$ 0.1265 0.9798 \$ 0.1316 \$ 0.1947 0.0045 \$ 0.3199 \$ 0.3199 \$ 0.1265 0.9890 0.9798 \$ 0.1226 \$ 0.1226	COGsl COGsl Maximum (COG + : COGsh COGsh COGsl COGsl COGsl	225%)	\$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3985 \$ 0.3885 \$ 0.3886) /therm) /therm) /therm) /therm / / herm / / /therm / / /therm
COM/IND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 Times: Common Factor Adjusted Demand Cost of Gas Rate Effective 05/01/20 Times: Low Winter Use Ratio (Summer) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Commodity Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Commodity Cost of Gas Rate Adjusted Commodity Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Demand Cost of Gas Rate COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Change in rate due to change in under/over recovery COM/IND HIGH WINTER USE COST OF GAS RATE - 08/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 Times: High Winter Use Ratio (Summer) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Indirect Cost of Gas Rate	\$ 0.1265 0.9798 \$ 0.1316 \$ 0.1947 0.0045 (0.0109) \$ 0.3199 \$ 0.3199 \$ 0.1265 0.9890 0.9792 \$ 0.1947 0.0045 (0.0109)	COGsl COGsl Maximum (COG + : COGsh COGsh COGsl COGsl COGsl	225%)	\$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3985 \$ 0.3885 \$ 0.3886) /therm) /therm) /therm / / / / / / / / / / / / / / / / / / /
COMIND LOW WINTER USE COST OF GAS RATE - 09/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 Times: Low Winter Use Ratio (Summer) Times: Low Comercion Factor Adjusted Demand Cost of Gas Rate Effective 05/01/20 Times: Low Winter Use Ratio (Summer) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjusted Demand Cost of Gas Rate Adjusted Low Winter Use Cost of Gas Rate Adjusted Com/Ind Low Winter Use Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Com/Ind Low Winter Use Cost of Gas Rate Adjusted Com/Ind Low Winter Use Cost of Gas Rate COMIND HIGH WINTER USE COST OF GAS RATE - 05/01/2021 Change in rate due to change in under/over recovery COMIND HIGH WINTER USE COST OF GAS RATE - 06/01/2021 Change in rate due to change in under/over recovery COMIND HIGH WINTER USE COST OF GAS RATE - 07/01/2021 Change in rate due to change in under/over recovery COMIND HIGH WINTER USE COST OF GAS RATE - 07/01/2021 Change in rate due to change in under/over recovery COMIND HIGH WINTER USE COST OF GAS RATE - 07/01/2021 Change in rate due to change in under/over recovery COMIND HIGH WINTER USE COST OF GAS RATE - 07/01/2021 Change in rate due to change in under/over recovery COMIND HIGH WINTER USE COST OF GAS RATE - 07/01/2021 Change in rate due to change in under/over recovery COMIND HIGH WINTER USE COST OF GAS RATE - 07/01/2021 Average Demand Cost of Gas Rate Effective 05/01/20 Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Commodity Cost of Gas Rate Adjusted Demand Cost of Gas Rate	\$ 0.1265 0.9798 \$ 0.1316 \$ 0.1947 0.0045 \$ 0.3199 \$ 0.3199 \$ 0.1265 0.9890 0.9798 \$ 0.1226 \$ 0.1226	COGsl COGsl Maximum (COG + : COGsh COGsh COGsl COGsl COGsl	225%)	\$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3995 \$ 0.3985 \$ 0.3885 \$ 0.3886) /therm) /therm) /therm / / / / / / / / / / / / / / / / / / /

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman TITLE: EFFECTIVE: August 1, 2021 President

Original Page 93 Summer Cost of Gas

28 CALCULATION OF SUMMER PERIOD COST OF GAS RATE - KEENE CUSTOMERS

Period Co	vered: Summer Period May 1, 2021, through October 31, 2021		
Projected	Gas Sales - therms		311,740
Total Anti	cipated Cost of Sendout		\$391,714
Add:	Prior Period Deficiency Uncollected Interest	\$0 \$0	
Deduct:	Prior Period Excess Collected Interest	(\$23,223) \$18	
	Prior Period Adjustments and Interest		(\$23,205)
Total Anti	cipated Cost	_	\$368,509
Cost of G	as Rate - Beginning Period (per therm)	_	\$1.1821
Mid Perio	d Adjustment - June 1, 2021		\$0.0844
Mid Perio	d Adjustment - July 1, 2021		\$0.2015
Mid Perio	d Adjustment - August 1, 2021		
Mid Perio	d Adjustment - September 1, 2021		
Mid Perio	d Adjustment - October 1, 2021		
Revised C	Cost of Gas Rate - Effective July 1, 2021 (per therm)	_	\$1.4680
	to tariff section 17(d), the Company may adjust the approved s rate upward on a monthly basis to the following rate:		
Maximum	Cost of Gas Rate (per therm)	_	\$1.4776

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 94 Winter Cost of Gas

29 CALCULATION OF FIXED WINTER PERIOD COST OF GAS RATE EXCLUDING KEENE CUSTOMERS

II. RATE SCHEDULES CALCULATION OF FIXED WINTER PERIOD COST OF GAS RATE PERIOD COVERED: WINTER PERIOD, NOVEMBER 1, 2020 THROUGH APRIL 30, 2021 (Refer to Text in Section 17(A) Fixed Price Option Program)

(Col 1)	(Col 2)	(C	ol 3)	
1 Total Anticipated Direct Cost of Gas	\$46,922,854			
2 Projected Prorated Sales (11/01/20 - 04/30/21)	88,213,529			
3 Direct Cost of Gas Rate		\$	0.5319	per therm
4 Demand Cost of Gas Rate	\$12,978,688			per therm
5 Commodity Cost of Gas Rate	32,931,719			per therm
6 Adjustment Cost of Gas Rate	1,012,447	\$	0.0115	per therm
7 Total Direct Cost of Gas Rate	\$46,922,854	\$	0.5319	per therm
8 Total Anticipated Indirect Cost of Gas	\$ 2,220,114			
9 Projected Prorated Sales (11/01/20 - 04/30/21)	88,213,529			
10 Indirect Cost of Gas		\$	0.0252	per therm
11 TOTAL PERIOD AVERAGE COST OF GAS EFFECTIVE (11/01/19)		\$	0.5571	
12 Calculation of FPO Excluding Low Income - Rate Code R-3				
13 TOTAL PERIOD AVERAGE COST OF GAS EFFECTIVE (11/01/19)		\$	0.5571	
14 FPO Risk Premium		\$	0.0200	
15 TOTAL PERIOD FIXED PRICE OPTION COST OF GAS RATE EFFECTIVE (11/01/19)		\$	0.5771	
16 RESIDENTIAL COST OF GAS RATE - EXCLUDING LOW INCOME - (R-3) 11/01/20	COGwr	\$	0.5771	/therm
17 Calculation of FPO for Gas Assistance Program - Rate Code R-4				
18 TOTAL PERIOD AVERAGE COST OF GAS EFFECTIVE (11/01/20) FPO (Line 13 * 0.55)		\$	0.3064	
19 FPO Risk Premium (Line 14 * 0.55)		\$	0.0110	
20 TOTAL PERIOD FIXED PRICE OPTION COST OF GAS RATE EFFECTIVE (11/01/19) (Line 15 * 0.55)		\$	0.3174	
21 RESIDENTIAL COST OF GAS RATE - LOW INCOME - (R-4) 11/01/20	COGwr	\$	0.3174	/therm

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman Neil Proudman

TITLE: EFFECTIVE: August 1, 2021 President

Original Page 95 Winter Cost of Gas

30 CALCULATION OF FIRM SALES WINTER PERIOD COST OF GAS RATE - EXCLUDING KEENE CUSTOMERS

CALCULATION OF FIRM SALES COST OF GAS RATE PERIOD COVERED: WINTER PERIOD, NOVEMBER 1, 2020 THROUGH APRIL 30, 2021 (Refer to Text in Section 16 Cost of Gas Clause)

(Refer to Text in	ii oeciii	JII 16 CUSI	Oi Gas Giat	,				
(Col 1)					(Col 2)	((Col 3)	
, ,					. ,		. ,	
Total Anticipated Direct Cost of Gas				\$	46,922,854			
Projected Prorated Sales (11/01/2020 - 04/30/21)					88,213,529			
Direct Cost of Gas Rate						\$	0.5319	per thern
Demand Cost of Gas Rate				\$	12,978,688	\$	0 1474	per thern
Commodity Cost of Gas Rate				٠	32,931,719	\$	0.1471	per them
Adjustment Cost of Gas Rate					1,012,447	\$		per thern
Fotal Direct Cost of Gas Rate				\$	46,922,854	\$		per thern
Total Anticipated Indirect Cost of Gas				\$	2,220,114			
Projected Prorated Sales (11/01/20 - 04/30/21)					88,213,529	\$	0.0050	
ndirect Cost of Gas						ъ	0.0252	per thern
TOTAL PERIOD AVERAGE COST OF GAS EFFECTIVE 11/01/20						\$	0.5571	per ther
						•		
RESIDENTIAL COST OF GAS RATE - EXCLUDING LOW INCOME - 11/01/2	20			COGV	vr	\$	0.5571	/therm
Change in rate due to change in under/over recovery						\$	-	
RESIDENTIAL COST OF GAS RATE - 12/01/2020				COGs	r	\$		/therm
Change in Rate due to change in under/over recovery						\$	(0.0907	
RESIDENTIAL COST OF GAS RATE - 01/01/2021				COG	vr	\$	(0.0388	/therm
Change in Rate due to change in under/over recovery RESIDENTIAL COST OF GAS RATE - 02/01/2021				COGV	45	\$		/therm
Change in Rate due to change in under/over recovery				COGV	vi	\$	0.4276	/tileffiii
RESIDENTIAL COST OF GAS RATE - 03/01/2021				COGv	vr	\$		/therm
Change in Rate due to change in under/over recovery						\$	0.0894	
RESIDENTIAL COST OF GAS RATE - 04/01/2021				COGv	vr	\$	0.6050	/therm
			Maximum	(COG	+ 25%)	\$	0.6964	
TAG ACCIGNANCE DI AN DECIDENTIAL COOT OF CAG DATE D. 4. 14/4/	20			CCC	ur	\$	0 2004	/the ===
GAS ASSISTANCE PLAN RESIDENTIAL COST OF GAS RATE R-4 - 11/01/	4 0			COGv	VI.	\$	u.3064	/therm
Change in rate due to change in under/over recovery RESIDENTIAL COST OF GAS RATE - 12/01/2020				COGs		\$	0.3064	/therm
Change in Rate due to change in under/over recovery				JUGS		\$	(0.0499	
RESIDENTIAL COST OF GAS RATE - 01/01/2021				COGv	vr	š		/therm
Change in Rate due to change in under/over recovery						Ś	(0.0213	
RESIDENTIAL COST OF GAS RATE - 2/1/2021				COGV	vr	\$	0.2352	/therm
Change in Rate due to change in under/over recovery						\$	0.0484	
RESIDENTIAL COST OF GAS RATE - 3/1/2021				COGV	vr	\$	0.2836	/therm
Change in Rate due to change in under/over recovery						\$	0.0492	
RESIDENTIAL COST OF GAS RATE - 4/1/2021				COGV	vr	\$	0.3328	/therm
CAI LOW WINTER USE COST OF GAS RATE - 11/01/20 Change in rate due to change in under/over recovery				COGv	vI	\$	0.5660	/therm
C&I LOW WINTER USE COST OF GAS RATE - 12/01/2020				COGs	:	\$	0.5660	/therm
Change in Rate due to change in under/over recovery						\$	(0.0907	
C&I LOW WINTER USE COST OF GAS RATE - 01/01/2021				COGs	ı	\$		/therm
Change in Rate due to change in under/over recovery						\$	(0.0388	
C&I LOW WINTER USE COST OF GAS RATE - 2/01/2021				COGs	il .	\$		/therm
Change in Rate due to change in under/over recovery						\$	0.0880	
C&I LOW WINTER USE COST OF GAS RATE - 3/01/2021				COGs	il	\$		
Change in Rate due to change in under/over recovery						<u> </u>		/therm
C&I LOW WINTER USE COST OF GAS RATE - 4/01/2021						\$	0.0894	
Average Demand Cost of Gas Rate Effective 11/01/20 Times: Low Winter Use Ratio (Winter) Times: Correction Factor				COGs	1	\$	0.0894	
Adjusted Demand Cost of Gas Rate	\$	0.1471 1.0620 0.9984 0.1560	Maximum			\$	0.0894	
Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate		1.0620 0.9984	Maximum			\$	0.0894 0.6139	
Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate	\$	1.0620 0.9984 0.1560 0.3733 0.0115 0.0252	Maximum		+ 25%)	\$	0.0894 0.6139 0.7075	
Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate	\$	1.0620 0.9984 0.1560 0.3733 0.0115 0.0252	Maximum	(COG	+ 25%)	\$	0.0894 0.6139 0.7075	/therm
Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate C&I HIGH WINTER USE COST OF GAS RATE - 11/01/20 Change in rate due to change in underfover recovery	\$	1.0620 0.9984 0.1560 0.3733 0.0115 0.0252	Maximum	(COG	+ 25%)	\$	0.0894 0.6139 0.7075	/therm
Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate 281 HIGH WINTER USE COST OF GAS RATE - 11/01/20 Change in rate due to change in under/over recovery 281 HIGH WINTER USE COST OF GAS RATE - 12/01/2020	\$	1.0620 0.9984 0.1560 0.3733 0.0115 0.0252	Maximum	(COG	+ 25%)	\$ \$	0.0894 0.6139 0.7075	/therm
Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate C&I HIGH WINTER USE COST OF GAS RATE - 1/10/1/20 Change in rate due to change in under/over recovery Lai HIGH WINTER USE COST OF GAS RATE - 1/20/1/20/20 Change in Rate due to change in under/over recovery	\$	1.0620 0.9984 0.1560 0.3733 0.0115 0.0252	Maximum	(COG	+ 25%)	\$ \$	0.0894 0.6139 0.7075 0.5552 0.5552 (0.0907	/therm
Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate Officer Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate C&I HIGH WINTER USE COST OF GAS RATE - 11/01/20 Change in rate due to change in under/over recovery Call HIGH WINTER USE COST OF GAS RATE - 12/01/2020 Change in Rate due to change in under/over recovery Call HIGH WINTER USE COST OF GAS RATE - 01/01/2021	\$	1.0620 0.9984 0.1560 0.3733 0.0115 0.0252	Maximum	(COGv	+ 25%)	\$ \$ \$ \$ \$ \$	0.0894 0.6139 0.7075 0.5552 0.5552 (0.0907	/therm /therm /therm /therm) /therm
Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate C&I HIGH WINTER USE COST OF GAS RATE - 1/10/1/20 Change in rate due to change in under/over recovery C&I HIGH WINTER USE COST OF GAS RATE - 1/2/0/1/2020 Change in Rate due to change in under/over recovery C&I HIGH WINTER USE COST OF GAS RATE - 0/10/1/2021 Change in Rate due to change in under/over recovery	\$	1.0620 0.9984 0.1560 0.3733 0.0115 0.0252	Maximum	COGv COGs	+ 25%) vh h	\$ \$ \$ \$ \$ \$ \$ \$	0.0894 0.6139 0.7075 0.5552 - 0.5552 (0.0907 0.4645 (0.0388	/therm /therm /therm /therm)
Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate C&I HIGH WINTER USE COST OF GAS RATE - 1/10/1/20 Change in rate due to change in under/over recovery 281 HIGH WINTER USE COST OF GAS RATE - 1/10/1/20/2 Change in Rate due to change in under/over recovery 281 HIGH WINTER USE COST OF GAS RATE - 0/10/1/20/2 Thange in Rate due to change in under/over recovery 281 HIGH WINTER USE COST OF GAS RATE - 2/10/1/20/2 Thange in Rate due to change in under/over recovery 281 HIGH WINTER USE COST OF GAS RATE - 2/10/1/20/2	\$	1.0620 0.9984 0.1560 0.3733 0.0115 0.0252	Maximum	(COGv	+ 25%) vh h	\$ \$ \$ \$ \$ \$ \$ \$ \$	0.0894 0.6139 0.7075 0.5552 - 0.5552 (0.0907 0.4645 (0.0388	/therm /therm /therm /therm /therm /therm /therm
Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate C&I HGH WINTER USE COST OF GAS RATE - 1/10/1/20 Change in rate due to change in under/over recovery CAI HGH WINTER USE COST OF GAS RATE - 1/10/1/20/2 Change in Rate due to change in under/over recovery CAI HGH WINTER USE COST OF GAS RATE - 0/10/1/20/2 Change in Rate due to change in under/over recovery CAI HGH WINTER USE COST OF GAS RATE - 2/10/1/20/2 Change in Rate due to change in under/over recovery CAI HGH WINTER USE COST OF GAS RATE - 2/10/1/20/2 Change in Rate due to change in under/over recovery	\$	1.0620 0.9984 0.1560 0.3733 0.0115 0.0252	Maximum	COGv COGs	+ 25%) vh h	\$ \$ \$ \$ \$ \$ \$ \$	0.0894 0.6139 0.7075 0.5552 - 0.5552 (0.0907 0.4645 (0.0388 0.4257	/therm /therm /therm /therm /therm /therm /therm
Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate 284 HGH WINTER USE COST OF GAS RATE - 1/10/1/20 284 HGH WINTER USE COST OF GAS RATE - 1/20/1/20/20 284 HGH WINTER USE COST OF GAS RATE - 1/20/1/20/20 284 HGH WINTER USE COST OF GAS RATE - 0/10/1/20/21 284 HGH WINTER USE COST OF GAS RATE - 2/0/1/20/21 284 HGH WINTER USE COST OF GAS RATE - 2/0/1/20/21 285 HIGH WINTER USE COST OF GAS RATE - 2/0/1/20/21 286 HIGH WINTER USE COST OF GAS RATE - 2/0/1/20/21	\$	1.0620 0.9984 0.1560 0.3733 0.0115 0.0252	Maximum	COGw COGw COGw	+ 25%) vh h	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$	0.0894 0.6139 0.7075 0.5552 - 0.5552 (0.0907 0.4645 (0.0388 0.4257	/therm /therm /therm) /therm) /therm
Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate Of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate C&I Indirect Cost OF GAS RATE - 11/01/20 Change in rate due to change in underfover recovery C&I HIGH WINTER USE COST OF GAS RATE - 12/01/2020 Change in Rate due to change in underfover recovery C&I HIGH WINTER USE COST OF GAS RATE - 01/01/2021 Change in Rate due to change in underfover recovery C&I HIGH WINTER USE COST OF GAS RATE - 20/11/2021 Change in Rate due to change in underfover recovery C&I HIGH WINTER USE COST OF GAS RATE - 20/11/2021 Change in Rate due to change in underfover recovery C&I HIGH WINTER USE COST OF GAS RATE - 3/01/2021	\$	1.0620 0.9984 0.1560 0.3733 0.0115 0.0252	Maximum	COGW COGW COGW	+ 25%) vh sh	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	0.0894 0.6139 0.7075 0.5552 - 0.5552 (0.0907 0.4645 (0.0388 0.4257 0.6137 0.0894	/therm /therm /therm /therm /therm /therm
Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate Of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate C&I HIGH WINTER USE COST OF GAS RATE - 1/10/1/20 Thange in rate due to change in underfover recovery 24 HIGH WINTER USE COST OF GAS RATE - 1/20/1/20/20 Thange in Rate due to change in underfover recovery 24 HIGH WINTER USE COST OF GAS RATE - 0/10/1/20/21 Thange in Rate due to change in underfover recovery 25 HIGH WINTER USE COST OF GAS RATE - 2/0/1/20/21 Thange in Rate due to change in underfover recovery 26 HIGH WINTER USE COST OF GAS RATE - 3/0/1/20/21 Thange in Rate due to change in underfover recovery 27 HIGH WINTER USE COST OF GAS RATE - 3/0/1/20/21	\$	1.0620 0.9984 0.1560 0.3733 0.0115 0.0252	Maximum	COGw COGw COGw	+ 25%) vh sh	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	0.0894 0.6139 0.7075 0.5552 - 0.5552 (0.0907 0.4645 (0.0388 0.4257 0.6137 0.0894	/therm /therm /therm) /therm) /therm
Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate C&I HIGH WINTER USE COST OF GAS RATE - 1/10/1/20 Change in rate due to change in under/over recovery C&I HIGH WINTER USE COST OF GAS RATE - 1/2/01/2020 Change in Rate due to change in under/over recovery C&I HIGH WINTER USE COST OF GAS RATE - 1/2/01/2021 Change in Rate due to change in under/over recovery C&I HIGH WINTER USE COST OF GAS RATE - 2/01/2021 Change in Rate due to change in under/over recovery C&I HIGH WINTER USE COST OF GAS RATE - 3/01/2021 Change in Rate due to change in under/over recovery C&I HIGH WINTER USE COST OF GAS RATE - 4/01/2021 Change In Rate Sec Cost OF GAS RATE - 4/01/2021 Average Demand Cost of Gas Rate Effective 11/01/20 Times: Corection Factor	\$ \$ \$ \$ \$ \$ \$	1.0620 0.9984 0.1560 0.3733 0.0115 0.0252 0.5660	Maximum	COGy COGy COGy COGy	+ 25%) vh h vh vh vh vh	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	0.0894 0.6139 0.7075 0.5552 - 0.5552 (0.0907 0.4645 (0.0388 0.4257 0.5137	/therm /therm /therm /therm /therm /therm /therm
Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate 281 HIGH WINTER USE COST OF GAS RATE - 11/01/20 281 HIGH WINTER USE COST OF GAS RATE - 12/01/2020 292 HIGH WINTER USE COST OF GAS RATE - 10/01/2021 293 HIGH WINTER USE COST OF GAS RATE - 20/01/2021 294 HIGH WINTER USE COST OF GAS RATE - 20/01/2021 295 HIGH WINTER USE COST OF GAS RATE - 30/01/2021 295 HIGH WINTER USE COST OF GAS RATE - 30/01/2021 295 HIGH WINTER USE COST OF GAS RATE - 30/01/2021 296 HIGH WINTER USE COST OF GAS RATE - 3/01/2021 297 HIGH WINTER USE COST OF GAS RATE - 3/01/2021 298 HIGH WINTER USE COST OF GAS RATE - 3/01/2021 298 HIGH WINTER USE COST OF GAS RATE - 3/01/2021 Average Demand Cost of Gas Rate Effective 11/01/20 Times: High Winter Use Ratio (Winter)	\$ \$ \$ \$ \$ \$ \$	1.0620 0.9984 0.1560 0.3733 0.0115 0.0252 0.5660		COGy COGy COGy COGy	+ 25%) vh h vh vh vh vh	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	0.894 0.6139 0.7075 0.5552 0.5552 0.5552 0.9077 0.4645 (0.0388 0.4257 0.0803	/therm /therm /therm /therm /therm /therm /therm
Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate Datagos In rate due to change in underfover recovery Sal HiGH WINTER USE COST OF GAS RATE - 12/01/2020 Datagos in Rate due to change in underfover recovery Sal HiGH WINTER USE COST OF GAS RATE - 20/01/2021 Datagos in Rate due to change in underfover recovery Sal HIGH WINTER USE COST OF GAS RATE - 20/01/2021 Datagos in Rate due to change in underfover recovery Sal HIGH WINTER USE COST OF GAS RATE - 3/01/2021 Datagos in Rate due to change in underfover recovery Sal HIGH WINTER USE COST OF GAS RATE - 4/01/2021 Datagos in Rate due to change in underfover recovery Sal HIGH WINTER USE COST OF GAS RATE - 4/01/2021 Datagos in Rate due to change in underfover recovery Sal HIGH WINTER USE COST OF GAS RATE - 4/01/2021 Datagos in Rate Gas Rate Effective 11/01/20 Datagos in Rate Rate Rate Rate	\$ \$ \$ \$ \$ \$ \$	1.0620 0.9984 0.1560 0.3733 0.0115 0.0252 0.5660 0.1471 0.9890 0.9984 0.1452		COGy COGy COGy COGy	+ 25%) vh h vh vh vh vh	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	0.894 0.6139 0.7075 0.5552 0.5552 0.5552 0.9077 0.4645 (0.0388 0.4257 0.0803	/therm /therm /therm /therm /therm /therm /therm
Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate 281 HIGH WINTER USE COST OF GAS RATE - 1/20/1/20/20 281 HIGH WINTER USE COST OF GAS RATE - 1/20/1/20/20 281 HIGH WINTER USE COST OF GAS RATE - 2/01/20/21 281 HIGH WINTER USE COST OF GAS RATE - 20/1/20/21 281 HIGH WINTER USE COST OF GAS RATE - 20/1/20/21 281 HIGH WINTER USE COST OF GAS RATE - 3/01/20/21 281 HIGH WINTER USE COST OF GAS RATE - 3/01/20/21 281 HIGH WINTER USE COST OF GAS RATE - 3/01/20/21 281 HIGH WINTER USE COST OF GAS RATE - 3/01/20/21 Average Demand Cost of Gas Rate Effective 11/01/20 Times: High Winter Use Ratio (Winter) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate	\$ \$ \$ \$ \$ \$ \$	1.0620 0.9984 0.1560 0.3733 0.0115 0.0252 0.5660 0.1471 0.9984 0.9989 0.9984 0.1452		COGy COGy COGy COGy	+ 25%) vh h vh vh vh vh	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	0.894 0.6139 0.7075 0.5552 0.5552 0.5552 0.9077 0.4645 (0.0388 0.4257 0.0803	/therm /therm /therm /therm /therm /therm /therm
Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate C&I HIGH WINTER USE COST OF GAS RATE - 1/10/1/20 Change in Rate due to change in underiover recovery C&I HIGH WINTER USE COST OF GAS RATE - 1/2/01/20/20 Change in Rate due to change in underiover recovery C&I HIGH WINTER USE COST OF GAS RATE - 0/10/1/20/21 Change in Rate due to change in underiover recovery C&I HIGH WINTER USE COST OF GAS RATE - 3/01/20/21 Change in Rate due to change in underiover recovery C&I HIGH WINTER USE COST OF GAS RATE - 3/01/20/21 Change in Rate due to change in underiover recovery C&I HIGH WINTER USE COST OF GAS RATE - 4/01/20/21 Average Demand Cost of Gas Rate Effective 11/01/20 Times: Corection Factor Adjusted Demand Cost of Gas Rate	\$ \$ \$ \$ \$ \$ \$	1.0620 0.9984 0.1560 0.3733 0.0115 0.0252 0.5660 0.1471 0.9890 0.9984 0.1452		COGy COGy COGy COGy	+ 25%) vh h vh vh vh vh	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	0.894 0.6139 0.7075 0.5552 0.5552 0.5552 0.9077 0.4645 (0.0388 0.4257 0.0803	/therm /therm /therm /therm /therm /therm /therm
Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate C&I Low Winter Use Cost OF GAS RATE - 1/10/1/20 Change in rate due to change in under/over recovery C&I HIGH WINTER USE COST OF GAS RATE - 1/2/10/1/20/20 Change in Rate due to change in under/over recovery C&I HIGH WINTER USE COST OF GAS RATE - 0/10/1/20/21 Change in Rate due to change in under/over recovery C&I HIGH WINTER USE COST OF GAS RATE - 2/01/20/21 Change in Rate due to change in under/over recovery C&I HIGH WINTER USE COST OF GAS RATE - 3/01/20/21 Change in Rate due to change in under/over recovery C&I HIGH WINTER USE COST OF GAS RATE - 4/01/20/21 Change in Rate Gas Cost Of Gas Rate Effective 11/01/20 Change In Rate Gas Rate Cost of Gas Rate Effective 11/01/20 Change In Rate Gas Rate Cost of Gas Rate Commodify Cost of Gas Rate Adjustment Cost of Gas Rate Adjustment Cost of Gas Rate	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	0.1471 0.9894 0.1560 0.252 0.5660 0.3733 0.0115 0.0252 0.5660 0.1471 0.9890 0.9984 0.1452		COGy COGy COGy COGy	+ 25%) vh h vh vh vh vh	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	0.894 0.6139 0.7075 0.5552 0.5552 0.5552 0.9077 0.4645 (0.0388 0.4257 0.0803	/therm /therm /therm /therm /therm /therm /therm

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman
EFFECTIVE: August 1, 2021
TITLE: President

Original Page 96 Winter Cost of Gas

31. ANTICIPATED COST OF GAS - EXCLUDING KEENE CUSTOMERS

Anticipated Cost of Gas

PERIOD COVERED: WINTER PERIOD, NOVEMBER 1, 2020 THROUGH APRIL 30, 2021 (REFER TO TEXT ON IN SECTION 16 COST OF GAS CLAUSE)

(Col 1)		(Col 2)	(Col 3)	
ANTICIPATED DIRECT COST OF GAS Purchased Gas:				
Demand Costs:	\$	12,022,922		
Supply Costs:	•	28,276,980		
Storage Gas:				
Demand, Capacity:	\$	955,766		
Commodity Costs:		3,064,149		
Produced Gas:		1,590,589		
Hedged Contract (Saving)/Loss Hedge Underground Storage Contract (Saving)/Loss		-		
neuge underground Storage Contract (Saving)/Loss	-			
Unadjusted Anticipated Cost of Gas			\$ 45,9	10,407
Adjustments:	•	0.007.404		
Prior Period (Over)/Under Recovery (as of 05/01/20) Interest	\$	2,227,421 72,812		
Fuel Inventory Revenue Requirement		441,037		
Broker Revenues		(32,725)		
Refunds from Suppliers		- (,)		
Fuel Financing		-		
Transportation CGA Revenues		(4,516)		
Interruptible Sales Margin		-		
Capacity Release and Off System Sales Margins Hedging Costs		(1,736,581)		
Fixed Price Option Administrative Costs		45,000		
Total Adjustments			1,0	12,447
Total Anticipated Direct Cost of Gas			\$ 46,9	22,854
Anticipated Indirect Cost of Gas				
Working Capital:				
Total Unadjusted Anticipated Cost of Gas 11/01/20 - 04/30/21	\$	45,910,407		
Working Capital Rate: Lead Lag Days / 365 Prime Rate		0.0391 3.25%		
Working Capital Percentage		0.127%		
Working Capital	\$	58,347		
Plus: Working Capital Reconciliation (Acct 142.20)		(66,837)		
Total Working Capital Allowance				(8,490)
Bad Debt:				
Total Unadjusted Anticipated Cost of Gas 11/01/20 - 04/30/21	\$	45,910,407		
Less: Refunds Plus: Total Working Capital		(8,490)		
Plus: Prior Period (Over)/Under Recovery		2,227,421		
Subtotal	\$	48,129,338		
Rad Dald Danaston				
Bad Debt Percentage Bad Debt Allowance	\$	<u>1.11%</u> 534,236		
Plus: Bad Debt Reconciliation (Acct 175.52)	φ	(296,628)		
Total Bad Debt Allowance	-	(200,020)	\$ 2	37,608
Production and Storage Capacity			\$ 1,9	80,428
Miscellaneous Overhead 11/01/20 - 04/30/21	\$	13,170		
Times Winter Sales		89,365		
Divided by Total Sales		111,369		
Miscellaneous Overhead				10,568
Total Anticipated Indirect Cost of Gas			\$ 2,2	20,114
Total Cost of Gas			\$ 49,1	42,968

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman Neil Proudman

EFFECTIVE: August 1, 2021 Neil Prouc

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 97 Winter Cost of Gas

32. CALCULATION OF FIRM SALES AND FIXED WINTER PERIOD COST OF GAS RATE - KEENE CUSTOMERS

Calculation of the Cost of Gas Rate

Period Covered: Winter Period November 1, 2020 through April 30, 2021 Projected Gas Sales - therms 1,108,419 Total Calculated Cost of Gas Sendout \$1,310,950 CNG Costs Not Included In Cost of Gas Rate (\$204,754) Total Anticipated Cost of Gas Sendout \$1,106,196 \$30,171 Add: Prior Period Deficiency Uncollected Interest \$136 Deduct: Prior Period Excess Collected \$0 Interest \$0 Prior Period Adjustments and Interest \$30,307 \$1,136,503 Total Anticipated Cost Costs Adjusted per Order No. 26,428 in Docket No. DG 20-152 dated December 02, 2020 Excluding Gas Assistance Gas Assistance Cost of Gas Rate Program Program Non-Fixed Price Option Cost of Gas Rate - Beginning Period (per therm) \$1.0253 \$0.5639 Mid Period Adjustment - December 01, 2020 \$0.0000 \$0.0000 Mid Period Adjustment - January 01, 2021 (\$0.0283) (\$0.0156) Mid Period Adjustment - February 01, 2021 (\$0.0042)(\$0.0023)Mid Period Adjustment - March 01, 2021 \$0.0000 \$0.0000 Mid Period Adjustment - April 01, 2021 \$0.2888 \$0.1588 \$1.2816 \$0.7048 Revised Non-Fixed Price Option Cost of Gas Rate - Effective April 1, 2021 (per therm) Fixed Price Option Cost of Gas Rate (per therm) November 2020 \$1.2300 \$0.6765 Fixed Price Option Cost of Gas Rate (per therm) December 2020-April 2021 \$1.0277 \$0.5652 Pursuant to tariff section 17(d), the Company may adjust the approved cost of gas rate upward on a monthly basis to the following rate:

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

EFFECTIVE: August 1, 2021 TITLE: President

Maximum Cost of Gas Rate - Non-Fixed Price Option (per therm)

\$0.7049

\$1.2816

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 98 Firm Transportation Cost of Gas

33. CALCULATION OF FIRM TRANSPORTATION COST OF GAS RATE

II. RATE SCHEDULES

Calculation of Firm Transportation Cost of Gas Rate
PERIOD COVERED: WINTER PERIOD, NOVEMBER 1, 2020 THROUGH APRIL 30, 2021
(Refer to text in Section 16(Q) Firm Transportation Cost of Gas Clause)

(Col 1)	(Col 2)	(Col 3)	(Col 4)
ANTICIPATED COST OF SUPPLEMENTAL GAS SUPPLIES:			
PROPANE	\$ 568,511		
LNG	1,022,078		
TOTAL ANTICIPATED COST OF SUPPLEMENTAL GAS SUPPLIES ESTIMATED PERCENTAGE USED FOR PRESSURE SUPPORT PURPOSES ESTIMATED COST OF LIQUIDS USED FOR PRESSURE SUPPORT PURPOSES	1,590,589 <u>8.7%</u> \$ 138,381		
PROJECTED FIRM THROUGHPUT (THERMS): FIRM SALES FIRM TRANSPORTATION SUBJECT TO FTCG TOTAL FIRM THROUGHPUT SUBJECT TO COST OF GAS CHARGE	89,364,968 42,456,275 131,821,243	67.8% <u>32.2%</u> 100.0%	
TRANSPORTATION SHARE OF SUPPLEMENTAL GAS SUPPLIES	32.2%	x \$138,381 =	\$ 44,569
PRIOR (OVER) OR UNDER COLLECTION			(40,053)
NET AMOUNT TO COLLECT FROM (RETURNED TO) TRANSPORTATION CUSTOMERS			\$ 4,516
PROJECTED FIRM TRANSPORTATION THROUGHPUT			42,456,275
FIRM TRANSPORTATION COST OF GAS			\$ 0.0001

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 153 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 99 Environmental Surcharge

34. ENVIRONMENTAL SURCHARGE - MANUFACTURED GAS PLANTS

Environmental Surcharge - Manufactured Gas Plants

Manufactured Gas Plants

Total Environmental Surcharge	\$0.0197
Surcharge per therm	<u>\$0.0197</u> per therm
Estimated weather normalized firm therms billed for the twelve months ended 10/31/20 - sales and transportation	179,574,679 therms
Overall Annual Net Increase to Rates	\$3,544,132
Environmental Subtotal	\$3,544,132
July 2019 - June 2020 recovery difference between actual and estimate	\$338,564
First one-third of prior period under recoveries (through June 2019)	\$341,389
Required Annual Environmental Increase	\$2,864,179

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman
Neil Proudman
TITLE: Provident

Docket No. DG 21-130 Exhibit 26 Page 154 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11GAS LIBERTY UTILITIES

Original Page 100 Rate Case Expense

35 RATE CASE EXPENSE AND RECOUPMENT FACTOR CALCULATION

Liberty Utilities (Energy North Natural Gas) Corp. d/b/a Liberty Local Delivery Adjustment Charge (LDAC) increase due to Rate Case Expense and Recoupment For LDAC effective November 1, 2020 - October 31, 2021

Rate Case Expense Remaining from Docket No. DG 17-048 Recoupment Remaining from Docket No. DG 17-048 July 1, 2020 Balance	\$87,069 <u>\$0</u> \$87,069
Plus Estimated Interest from July 2020 through October 2020	\$745
Minus Estimated Recoveries from July 2020 through October 2020	(\$43,733)
Total Estimated Remaining Recovery As of November 1, 2020	\$44,081
Estimated November 2019 - October 2020 Interest	<u>\$538</u>
Total Remaining Recovery	\$44,619
Estimated November 2020 - October 2021 Sales (therms)	179,574,679
RCE & Recoupment rate per therm November 2020 - October 2021	\$0.0002

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman Neil Proudman

TITLE: EFFECTIVE: August 1, 2021 President

Original Page 101 LDAC

36 LOCAL DISTRIBUTION ADJUSTMENT CLAUSE CALCULATION

Local Delivery Adjustment Charge Calculation

Residential Non Heating Rates - R-1, R-5 Energy Efficiency Charge	\$	0.0831	Sales <u>Customers</u>	Transportation Customers
Demand Side Management Charge Conservation Charge (CCx)	_	0.0000	\$ 0.0831	I
Relief Holder and pond at Gas Street, Concord, NH Manufactured Gas Plants		0.0000 0.0197		
Environmental Surcharge (ES)			0.0197	
Revenue Decoupling Adjustment Factor (RDAF) Energy Efficiency Resource Standard Lost Revenue Mechanism			(0.0562 0.0000	
Rate Case Expense Factor (RCEF)			0.0002	
Gas Assistance Program (GAP) LDAC		-	0.0121 \$ 0.0589	
Residential Heating Rates - R-3, R-4, R-6, R-7				
Energy Efficiency Charge	\$	0.0831		
Demand Side Management Charge		0.0000		
Conservation Charge (CCx) Relief Holder and pond at Gas Street, Concord, NH		0.0000	\$ 0.0831	
Manufactured Gas Plants		0.0197		
Environmental Surcharge (ES)			0.0197	
Revenue Decoupling Adjustment Factor (RDAF)			(0.0562	
Energy Efficiency Resource Standard Lost Revenue Mechanism Rate Case Expense Factor (RCEF)			0.0000	
Gas Assistance Program (GAP)			0.0121	
LDAC		_	\$ 0.0589	per therm
Commercial/Industrial Low Annual Use Rates - G-41, G-51, G-44, G-55				
Energy Efficiency Charge		\$0.0441		
Demand Side Management Charge		0.0000		0.0444
Conservation Charge (CCx) Relief Holder and pond at Gas Street, Concord, NH		0.0000	\$ 0.0441	I \$ 0.0441
Manufactured Gas Plants		0.0197		
Environmental Surcharge (ES)			0.0197	
Revenue Decoupling Adjustment Factor (RDAF)			(0.0206	, , ,
Energy Efficiency Resource Standard Lost Revenue Mechanism Rate Case Expense Factor (RCEF)			0.0000	
Gas Assistance Program (GAP)			0.0121	
LDAC		-	\$ 0.0555	5 \$ 0.0555 per therm
Commercial/Industrial Medium Annual Use Rates - G-42, G-52, G-45, G-56				
Energy Efficiency Charge Demand Side Management Charge		\$0.0441 0.0000		
Conservation Charge (CCx)		0.0000	\$0.0441	\$0.0441
Relief Holder and pond at Gas Street, Concord, NH		0.0000		,
Manufactured Gas Plants		0.0197		
Environmental Surcharge (ES) Revenue Decoupling Adjustment Factor (RDAF)			0.0197 (0.0206	
Energy Efficiency Resource Standard Lost Revenue Mechanism			0.0000	
Rate Case Expense Factor (RCEF)			0.0002	
Gas Assistance Program (GAP)		_	0.0121	
LDAC			\$ 0.0555	5 \$ 0.0555 per therm
Commercial/Industrial Large Annual Use Rates - G-43, G-53, G-54, G-46, G-56, G-57,G-58				
Energy Efficiency Charge	\$	0.0441		
Demand Side Management Charge Conservation Charge (CCx)		0.0000	\$0.0441	\$0.0441
Relief Holder and pond at Gas Street, Concord, NH		0.0000	ψυ.υ	ψο.σ
Manufactured Gas Plants		0.0197		
Environmental Surcharge (ES)			0.0197	
Revenue Decoupling Adjustment Factor (RDAF) Energy Efficiency Resource Standard Lost Revenue Mechanism			(0.0206 0.0000	
Rate Case Expense Factor (RCEF)			0.0002	
Gas Assistance Program (GAP)		_	0.0121	0.0121
LDAC			\$ 0.0555	5 \$ 0.0555 per therm

ISSUED BY: <u>/s/Neil Proudman</u> Neil Proudman DATED: XXX XX, 2021

Docket No. DG 21-130 Exhibit 26 Page 156 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 102 LDAC – Residential Energy Efficiency Loan Program

37 Residential Energy Efficiency Loan Program

Availability

Subject to the Terms and Conditions of the Tariff of which it is a part, this program shall allow Customers installing energy-efficiency measures under an energy efficiency program offered by the Company and approved by the Commission ("Participating Customers") to borrow all or a portion of the Customer's share of the installed cost of the energy-efficiency measures ("Customer Loan Amount") from the Company and to repay the Customer Loan Amount through an additional charge on their monthly retail delivery service bill issued by the Company. It is available to Participating Customers who meet the following qualifications:

- A. The Participating Customer must own the property where the energy-efficiency measures are installed; and
- B. A Participating Customer must have an active Delivery Service account with the Company for the property where the energy-efficiency measures are installed and receive Delivery Service under R-1, R-3, R-4, R-5, R-6, R-7; and
- C. The Participating Customer must not have received a disconnect notice from the Company during the twelve months preceding the Participating Customer's request for a loan under this program; and
- D. The Customer Loan Amount has no minimum and must be less than or equal to \$4,000 per customer per year, and must not exceed the Participating Customer's share of the installed cost of the energy efficiency measures installed under the Company's approved energy-efficiency program; and
- E. The Participating Customer must meet the qualifications of the applicable energy-efficiency program through which the energy-efficiency measures are being installed.

At its sole discretion, the Company shall determine eligibility for service under this program subject to the availability of program funds.

Any Participating Customer receiving a loan under this program must remain a Delivery Service customer of the Company at the property where the energy-efficiency measures are installed until the loan has been repaid in full. In the event the Participating Customer ceases to be a Delivery Service Customer of the Company at the property where the energy-efficiency measures are installed, any remaining charges under this program shall immediately become due and payable.

Customer Loan Agreement

Participating Customers shall be required to execute a separate Residential Customer Loan Agreement which will specify the fixed monthly charge and other applicable terms. A Participating Customer can choose to pay the remaining balance owed to the Company at any time. A late payment charge as described in the Terms and Conditions for Delivery Service section of the Company's Tariff is applicable to the monthly charges rendered under this program. Participating Customers are not subject to disconnection of natural gas service for nonpayment of the charges under this program.

The Customer Loan Amount shall be paid to the Company by the Participating Customer through a fixed monthly charge applied over a term of months as established in the Customer Loan Agreement. Participating Customers may specify the repayment term of the Customer Loan Amount subject to the maximum repayment term limit of 48 months.

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 157 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11GAS LIBERTY UTILITIES Original Page 103 LDAC – Non-Residential Energy Efficiency Loan Program

38 Non-Residential Energy Efficiency Loan Program

Availability

Subject to the Terms and Conditions of the Tariff of which it is a part, this program shall allow Customers installing energy-efficiency measures under an energy efficiency program offered by the Company and approved by the Commission ("Participating Customers") to borrow all or a portion of the Customer's share of the installed cost of the energy-efficiency measures ("Customer Loan Amount") from the Company and to repay the Customer Loan Amount through an additional charge on their monthly retail delivery service bill issued by the Company. It is available to Participating Customers who meet the following qualifications:

- A. The Participating Customer must own the property where the energy-efficiency measures are installed; and
- B. A Participating Customer must have an active Delivery Service account with the Company for the property where the energy-efficiency measures are installed and receive Delivery Service under G-41, G-42, G-43, G-44, G-45, G-46, G-51, G-52, G-53, G-54, G-55, G-56, G-57, or G-58; and
- C. The Participating Customer must not have received a disconnect notice from the Company during the twelve months preceding the Participating Customer's request for service under this program; and
- D. The Customer Loan Amount has no minimum and must be less than or equal to \$50,000 for each project, and must not exceed the Participating Customer's share of the installed cost of the energy efficiency measures installed under the Company's approved energy-efficiency program; and
- E. A Participating Customer is limited to \$150,000 per year in loan funds with no limit on the number of projects at the sole discretion of the Company based on program demand. If at any point there are no loan fund recipients or there have been no loan fund recipients in a given year, the Company may petition the Commission to allow a particular customer to receive more than \$150,000 in loan funds in a given year; and
- F. The Participating Customer must meet the qualifications of the applicable energy-efficiency program through which the energy-efficiency measures are being installed.

At its sole discretion, the Company shall determine eligibility for service under this program subject to the availability of program funds.

Any Participating Customer receiving a loan under this program must remain a Delivery Service customer of the Company at the property where the energy-efficiency measures are installed until the loan has been repaid in full. In the event the Participating Customer ceases to be a Delivery Service Customer of the Company at the property where the energy-efficiency measures are installed, any remaining charges under this program shall immediately become due and payable.

Customer Loan Agreement

Participating Customers shall be required to execute a separate Non-Residential Customer Loan Agreement which will specify the fixed monthly charge and other terms of the loan. A Participating Customer can choose to pay the remaining balance owed to the Company at any time. A late payment charge as described in the Terms and Conditions for Delivery Service section of the Company's Tariff is applicable to the monthly charges rendered under this program. Participating Customers are not subject to disconnection of natural gas service for nonpayment of the charges under this program.

The Customer Loan Amount shall be paid to the Company by the Participating Customer through a fixed monthly charge applied over a term of months as established in the Customer Loan Agreement. Participating Customers may specify the repayment term of the Customer Loan Amount subject to the maximum repayment term limit of 60 months.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 104 Delivery Terms and Conditions

III. DELIVERY TERMS AND CONDITIONS

1 RATES AND CHARGES

- 1.1 The Company shall apply this tariff on a non-discriminatory and non-preferential basis to all Customers who obtain service from the Company, except as this tariff is explicitly modified by order of the NHPUC. The provisions of Part III Section 20 of this tariff will specifically apply to all entities designated by the Customer as set forth in Section 20.5 to supply Gas to a Designated Receipt Point for the Customer's account.
- 1.2 The Company reserves the right to impose reasonable fees and charges pursuant to the various provisions of this tariff.
- 1.3 In the event that the Company incurs minimum bill, inventory, transition, take or pay, imbalance, or any other charges associated with the provision of Delivery Service to Customers, the Company may impose an additional charge on the Suppliers serving said Customers as approved by the NHPUC.

2 DEFINITIONS

Adjusted Target Volume

("ATV)

The volume of Gas determined by the Company using a Consumption Algorithm and required to be nominated and delivered each Gas Day by the Supplier on behalf of Customers taking non-daily metered Delivery Service.

Aggregation Pool

One or more Customer accounts whose Gas Usage is served by the same Supplier and aggregated pursuant to Section 20.6 of this tariff for operational purposes, including but not limited to nominating, scheduling, and balancing Gas deliveries to Designated Receipt Point(s) within the associated Gas Service Area.

Annual Reassignment Date

Five (5) Business Days prior to November 1 of each year when the Company reassigns Capacity to Suppliers pursuant to Section 11.6 of this tariff.

Assignment Date

Five (5) Business Days prior to the first Gas Day of each month when the Company assigns Capacity to Suppliers pursuant to Section 11.4 of this tariff.

Authorization Number

A number unique to the Customer generated by the Company and printed on the Customer's bill that the Customer must furnish to the Supplier to enable the Supplier to obtain the Customer's Gas Usage information pursuant to Section 20.4, and to initiate or terminate Supplier Service as set forth in Section 20.5 of this tariff.

Btu

One British thermal unit; i.e., the amount of heat required to raise the temperature of one pound of water one degree Fahrenheit at sixty degrees (60°) Fahrenheit.

Business Day

Monday through Friday excluding holidays recognized by the Company. Where relevant, a Business Day shall consist of the hours during which the Company is open for business with the public. <u>If any performance date referenced in this Tariff is not a Business Day, such performance shall be the next succeeding Business Day.</u>

Capacity

Pipeline Capacity, Storage Withdrawal Capacity, and Peaking Capacity as defined in this tariff.

DATED:

ISSUED BY: <u>/s/Neil Proudman</u> Neil Proudman TITLE: President

EFFECTIVE: August 1, 2021

XXX XX, 2021

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Docket No. DG 21-130 Exhibit 26 Page 159 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 105
Delivery Terms and Conditions

Capacity Allocators The estimated proportions of the Customer's Total Capacity Quantity that

comprise Pipeline Capacity, Storage Withdrawal Capacity and Peaking

Capacity.

Capacity Mitigation Service The service available to Suppliers in accordance with Section 11.10.

City Gate The interconnection between a Delivering Pipeline and the Company's

distribution facilities.

Commodity See Gas.

Company Liberty Utilities (EnergyNorth Natural Gas) Corp. d/b/a Liberty

Company Gas Allowance The difference between the sum of all amounts of Gas received into the

Company's distribution system (including Gas produced by the Company) and the sum of all amounts of Gas delivered from the Company's distribution system divided by said amount of Gas received. Such difference shall include but not be limited to Gas consumed by the Company for its own purposes, line losses, and Gas vented and lost as a result of force majeure, excluding Gas

otherwise accounted for.

Company-Managed Supplies Capacity and Supply contracts held and managed by the Company and made

available to the Supplier pursuant to Section 11.9 of this tariff including

Supply-sharing contracts and load-management contracts.

Consumption Algorithm A mathematical formula used to estimate a Customer's daily consumption.

Critical Day In accordance with Section 16 of this tariff, a day declared at any time by the

Company in its reasonable discretion when unusual operating conditions may

jeopardize operation of the Company's distribution system.

Customer The recipient of Delivery Service whose Gas Usage is recorded by a meter or

group of meters at a specific location and who is a customer of record of the

Company.

Daily Baseload The Customer's average usage per Gas Day that is assumed to be unrelated to

weather.

Daily Index The mid-point_of the range of prices as published by <u>Gas Daily</u> under the

heading "Daily Price Survey, Midpoint, Citygates, Tennessee/Zone 6

(delivered)" for the relevant Gas Day listed under "Flow date(s)".

In the event that the <u>Gas Daily</u> index becomes unavailable, the Company shall apply its daily marginal cost of Gas as the basis for this calculation until such

time that the NHPUC approves a suitable replacement.

Dekatherm Ten Therms.

Delivery Point The interconnection between the Company's facilities and the Customer's

facilities.

Delivery Service The distribution of Gas by the Company on any Gas Day from the Designated

Receipt Point to the Customer's Delivery Point and related Customer services.

Design Peak Season The forecasted Peak Season during which the Company's system experiences

the highest aggregate Gas Usage.

Designated Receipt Point For each Customer, the Company designated interconnection between a

Transporting Pipeline and the Company's distribution facilities at which

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman President

Docket No. DG 21-130 Exhibit 26 Page 160 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 106 Delivery Terms and Conditions

point, or such other point as the Company may designate from time to time for operational purposes, the

Supplier will make deliveries of Gas for the Customer's account.

Designated Representative The designated representative of the Customer, who shall be authorized to act

for, and conclusively bind, the Customer regarding Delivery Service in

accordance with the provisions of Section 21 of this tariff.

Gas Natural Gas that is received by the Company from a Transporting Pipeline at

the Designated Receipt Point and delivered by the Company to the Delivery Point for the Customer's account. In addition, the term shall include amounts of vaporized liquefied natural Gas and/or propane-air vapor that are introduced by the Company into its system and made available to the Customer as the equivalent of natural Gas that the Customer is otherwise

entitled to have delivered by the Company.

Gas Day A period of twenty-four (24) consecutive hours beginning at 10:00 a.m., E.T.,

and ending at 10:00 a.m., E.T., the next calendar day, or other such hours used

by the Transporting Pipeline.

Gas Service Area An area within the Company's distribution system as defined in Section 4 of

this tariff, for the purposes of administering Capacity assignments,

Nominations, balancing, imbalance trading, and Aggregation Pools.

Company's metering equipment at the Delivery Point.

Heating Degree Day A measure used to estimate weather-sensitive Gas consumption calculated by

subtracting the average temperature for each day from the number 65. Each degree day that represents a degree below 65 is considered a Heating Degree

Day.

Heating Factor The Customer's estimated weather-sensitive Gas consumption per Heating

Degree Day.

MMBtu One million Btus.

Maximum Daily Peaking The portion of a Customer's Total Capacity Quantity identified

Quantity ("MDPQ") and allocated as Peaking Capacity, such that the maximum daily amount of

Gas that can be withdrawn from a Supplier's Peaking Service Account pursuant to Section 14 of this tariff shall be equal to the sum of the MDPQs

for all Customers in that Supplier's Aggregation Pool.

Month A calendar month of Gas Days.

Monthly Index The average of the Daily Index numbers for all Gas Days in a Month.

NHPUC The New Hampshire Public Utilities Commission.

Nomination The notice given by the Supplier to the Company that specifies, in accordance

with the Standard Nomination Form attached as Attachment A, an intent to deliver a quantity of Gas to the Designated Receipt Point(s) on behalf of one or more Customers, including the volume to be received, the Designated Receipt Point(s), the Transporting Pipeline, the delivering contract(s), the shipper, and other such non-confidential information as may be reasonably

required by the Company.

Off-Peak Season The consecutive months of May to October, inclusive.

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 161 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Rate Schedule

Original Page 107 Delivery Terms and Conditions

Operational Flow Order The Company's instructions to the Supplier to take such action as conditions

("OFO") require including, but not limited to, diverting Gas to or from the Company's

distribution system pursuant to Section 16 of this tariff.

Peak Day The forecasted Gas Day during which the Company's system experiences the

highest aggregate Gas Usage.

Peak Season The consecutive months of November to April, inclusive.

Peaking Capacity Capacity in addition to upstream pipeline and underground storage Capacity

normally used by the Company to meet daily requirements during a Design

Peak Season and acquired specifically for the Peak Season.

Peaking Service A Company-managed resource consisting of Peaking Capacity and Peaking

Supply.

Peaking Service Account An account whose balance indicates the total volumes of Peaking Service

resources available to a Supplier, where the maximum balance in the account shall equal the Peaking Supply assigned to the Supplier pursuant to this tariff.

Peaking Service Rule Curve A system of operational parameters associated with the use of the Company's

Peaking Capacity including, but not limited to, indicators of the necessary levels of Peaking Supply that must be maintained in Suppliers' Peaking Service Accounts in order for the Company to meet system demands under Design Peak Season conditions. The Company will communicate, by electronic means as determined by the Company or, in the event of failure of such electronic means, by facsimile or other agreeable alternative means, the

Peaking Service Rule Curve as identified in Section 14 of this tariff.

Peaking Supply The aggregate amount of Supply in excess of upstream pipeline and

underground storage Supply required to meet the Company's forecasted Supply needs during a Design Peak Season and acquired specifically for the

Peak Season.

Peaking Supply Allocator An allocation factor that represents the proportion of a Customer's estimated

Gas Usage during the Design Peak Season that is generally served with

Peaking Service supplies.

Pipeline Capacity Transportation capacity on interstate pipeline systems normally used for

deliveries of Gas to the Company's city gates, exclusive of Storage

Withdrawal Capacity.

Pre-Determined Allocation Instructions from the Supplier to the Company for the method allocation of

discrepancies in confirmed Nominations among the Supplier's Aggregation Pools and/or Customers as set forth in the Supplier Service Agreement.

The schedule of rates included in this tariff.

Reference Period A period of at least twelve (12) months for which a Customer's Gas Usage

information is typically available to the Company.

Sales Service Commodity service provided on a firm basis to a Customer who is not

receiving Supplier Service, in accordance with the provisions set forth in this tariff. The provision of Sales Service shall be the responsibility of the Company and shall be provided to the Customer by the Company or its

designated Supplier pursuant to law or regulation.

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 162 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 108
Delivery Terms and Conditions

Seasonal Storage Capacity Contracts for Capacity in off-system storage facilities used to accumulate and

maintain Gas inventories for re-delivery to the Company's city gates normally

during the Peak Season.

Storage Withdrawal Capacity Capacity for the withdrawal of Gas inventories maintained in off-system

storage facilities, as well as the Pipeline Capacity used to deliver such Gas to

the Company's city gates.

Supplier Any entity that has met the Company's requirements set forth in Section 20

of this tariff and that has been designated by a Customer to supply Gas to a Designated Receipt Point for the Customer's account; provided, however, that a Customer may act as its own Supplier in accordance with Section 5.2 of this

tariff.

Supplier Service The sale of Gas to a Customer by a Supplier.

Supplier Service Agreement An agreement, substantially in the form set forth in Attachment A, which must

be executed by the Company and a Supplier in order for the Supplier to serve

Customers on the Company's system.

Supply See Gas.

Therm An amount of Gas having a thermal content of 100,000 Btus.

Total Capacity Quantity The total amount of Capacity assignable to a Supplier on behalf

("TCQ") of a Customer.

Transporting Pipeline The interstate pipeline company that transports and delivers Gas to the

Designated Receipt Point.

3 CHARACTER OF SERVICE

- 3.1 All rates within Part II Rate Schedule are predicated upon service to a Customer at a single Delivery Point and metering installation, except as otherwise specifically provided by a given rate. Where service is supplied to a Customer at more than one Delivery Point or metering installation, each single Delivery Point or metering installation shall be considered to be a separate Customer for purposes of applying the Rate Schedule, except when a Customer is served through multiple points of delivery or metering installations for the Company's own convenience.
- 3.2 The Company may refuse to supply service to loads of unusual characteristics which, in its sole reasonable judgment, might adversely affect the quality of service supplied to other Customers, the public safety or the safety of the Company's personnel. In lieu of such refusal, the Company may require a Customer to install any necessary regulating and protective equipment in accordance with the requirements and specifications of the Company.

4 GAS SERVICE AREAS AND DESIGNATED RECEIPT POINTS

- 4.1 There shall be 1 Gas Service Area defined for purposes of administering Capacity assignments, Nominations, balancing, imbalance trading, and Aggregation Pools pursuant to this tariff. Each such Gas Service Area shall be defined to include the municipalities listed within each such Gas Service Area, as follows:
 - (1) Area 1: Liberty Utilities (EnergyNorth Natural Gas) Corp. d/b/a Liberty.

 The area authorized to be served by the Company and to which this tariff applies are the following cities and towns: Allenstown, Amherst, Auburn, Bedford, Belmont, Berlin,

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman President

Docket No. DG 21-130 Exhibit 26 Page 163 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 109 Delivery Terms and Conditions

Boscawen, Bow, Concord, Derry, Franklin, Gilford, Goffstown, Hollis, Hooksett, Hudson, Laconia, Litchfield, Londonderry, Loudon, Manchester, Merrimack, Milford, Nashua, Northfield, Pelham, Pembroke, Sanbornton, Tilton, Windham, and part of Canterbury.

- 4.2 For each Aggregation Pool as set forth by Section 20.6, the Company will designate at least one specific interconnection between a Transporting Pipeline and the Company's distribution facilities, at which point, or such other point as the Company may designate from time to time, the Supplier will make deliveries for the Aggregation Pool. The interconnections that the Company may assign as the Customer's Designated Receipt Point for the Aggregation Pool are as follows:
 - (1) Name Transporting Pipeline: Tennessee Gas Pipeline Names of City Gates/Meter Numbers:

Nashua/Milford #020132 Manchester #020133 Hooksett #020254 Concord/Laconia #020426 Suncook #020451 Londonderry #020632

(2) Name Transporting Pipeline: Portland Natural Gas Transmission System Names of City Gates/Meter Number

Berlin #020260

5 CUSTOMER REQUEST FOR SERVICE FROM COMPANY

- 5.1 Application for Delivery Service, Sales Service, or any other service offered by the Company to a Customer will be received by any duly authorized representative or agent of the Company.
- 5.2 Before any service from the Company may commence, the Customer must request such service. A Customer applying for Delivery Service only must also arrange for Supplier Service with a Supplier pursuant to Section 20. A Customer may act as its own Supplier provided it meets all of the Supplier requirements delineated in Section 20.

6 QUALITY AND CONDITION OF GAS

- Gas delivered to the Company by or for the Customer shall conform, in all respects, to the Gas quality standards of the Transporting Pipeline. All Gas tendered by a Supplier at a Designated Receipt Point shall be of merchantable quality and shall be interchangeable with Gas purchased by the Company from its Suppliers. The Company reserves the right to refuse non-conforming Gas.
- 6.2 In no event shall the Company be obligated to accept and deliver any Gas that does not meet the quality standards of the Transporting Pipeline.
- 6.3 The Company reserves the right to commingle Gas tendered by a Supplier at a Designated Receipt Point with other Gas, including liquefied natural Gas and propane-air vapor.
- Gas tendered by a Supplier at a Designated Receipt Point will be at a pressure sufficient to enter the Company's distribution system without requiring the Company to adjust its normal operating pressures to receive the Gas. The Company has no obligation to receive Gas at a pressure that exceeds the maximum allowable operating pressure of the Company's distribution system at the Designated Receipt Point.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 164 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 110 Delivery Terms and Conditions

7 POSSESSION OF GAS

- 7.1 Gas shall be deemed to be in the control and possession of the Company after such Gas is delivered to the Designated Receipt Point and until the Gas is delivered to the Customer at the Delivery Point. The Company shall not be responsible for the Gas when the Gas is not in the Company's control and possession.
- 7.2 The Company shall not be liable to the Supplier or the Customer for any loss arising from or out of Delivery Service, including loss of Gas in the possession of the Company or for any other cause, except for the negligence of the Company's own employees or agents.

8 COMPANY GAS ALLOWANCE

8.1 The amount of Gas tendered by the Supplier to the Designated Receipt Point will be reduced, upon delivery to the Customer's Delivery Point, by the Company Gas Allowance. The Company Gas Allowance shall be in effect from November 1 through October 31. Such adjustment shall be recalculated prior to the Company's Peak Season cost of Gas filing with the NHPUC.

9 DAILY METERED DELIVERY SERVICE

9.1 Applicability

Section 9 of this tariff shall be applicable in the following conditions:

- 9.1.1 All Customers whose service may be interrupted at any time during the year shall be required to take daily metered Delivery Service.
- 9.1.2 Any Customer, regardless of annual Gas Usage, may elect daily metered Delivery Service.
- 9.1.3 Customers under Rate Schedules G-43, G-46, G-53, G-54, G-57, and G-58 wishing to take Delivery Service are required to take Daily Metered Delivery Service. In addition, the Company may require a Customer to take daily metered Delivery Service if the Company determines that the daily Gas Usage characteristics of the Customer cannot be accurately modeled using the Company's Consumption Algorithm or if the volumes reasonably anticipated by the Company to be used by the Customer are of a size that may materially affect the integrity of the Company's distribution system.

9.2 Delivery Service Provided

This service provides delivery of Customer purchased Gas from the Designated Receipt Point to the Delivery Point on any Gas Day. For Customers taking Delivery Service under Rate Schedules <u>G-43, G-46, G-53, G-54, G-57, and G-58</u> this service provides firm, year-round delivery of Customer purchased Gas from the Designated Receipt Point to the Delivery Point.

- 9.3 Nominations and Scheduling of Service
 - 9.3.1 The Supplier is responsible for nominating and delivering to the Designated Receipt Point(s) every Gas Day an amount of Gas that equals the aggregated Gas Usage of Customers in the Aggregation Pool plus the Company Gas Allowance in accordance with Section 8 of this tariff.
 - 9.3.2 Nominations shall be communicated to the Company by electronic means as determined by the Company or, in the event of failure of such electronic means, by facsimile or other agreeable alternative means.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 165 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 111 Delivery Terms and Conditions

- 9.3.3 Nominations for the first Gas Day of a Month shall be submitted to the Company no later than two (2) hours prior to the deadline for first of the Month Nominations of the Transporting Pipeline or such lesser period as determined by the Company. The Company will make available, from time to time, a schedule of Nomination due dates. Nominations on weekends, holidays, and non-business hours will be accepted by the Company on a basis consistent with that utilized for its own operations.
- 9.3.4 The Supplier may make daily Nominations including, but not limited to, changes to existing Nominations, within a given Month no later than two (2) hours prior to the deadline for daily Nominations of the Transporting Pipeline for the Gas Day on which the Nomination is to be effective, or such lesser period as determined by the Company. Nominations on weekends, holidays, and non-business hours will be accepted by the Company on a basis consistent with that utilized for its own operations.
- 9.3.5 The Supplier may make intra-Gas Day Nominations, including but not limited to changes to existing Nominations, within a given Gas Day no later than two (2) hours prior to the intra-Gas Day Nomination deadline for the Transporting Pipeline on which the Nomination is to be effective, or such lesser period as determined by the Company. Intra-Gas Day Nominations on weekends, holidays, and non-business hours will be accepted by the Company on a basis consistent with that utilized for its own operations.
- 9.3.6 Nominations will be conditionally accepted by the Company pending confirmation by the Transporting Pipeline. The Company will attempt to confirm the nominated volume with the Transporting Pipeline. In the event of a discrepancy between the volume nominated to the Company by the Supplier and the volume nominated by the Supplier to the Transporting Pipeline, the lower volume will be deemed confirmed. The Company will allocate such discrepancy based on a predetermined allocation method set forth in the Supplier Service Agreement. If no predetermined allocation method has been established prior to the event of such discrepancy, the Company will allocate the discrepancy on a pro rata basis.
- 9.3.7 Nominations may be rejected, at the sole reasonable discretion of the Company, if they do not satisfy the conditions for Delivery Service in effect from time to time.
- 9.4 Determination of Receipts
 - 9.4.1 The quantity of Gas deemed received by the Company for the Supplier's Aggregation Pool at the Designated Receipt Point(s) will equal the volume so scheduled by the Transporting Pipeline(s).
 - 9.4.2 The Company Gas Allowance will be assessed against receipts pursuant to Section 8 of this tariff.
- 9.5 Metering and Determination of Deliveries
 - 9.5.1 The Company shall furnish and install, at the Customer's expense, telemetering equipment and any related equipment for the purpose of measuring Gas Usage at each Customer's Delivery Point. Telemetering equipment shall remain the property of the Company at all times. The Company shall require each Customer to install and maintain, at the Customer's expense, reliable telephone lines and electrical connections that meet the Company's operating requirements. The Company may require the Customer to furnish a dedicated telephone line. If the Customer fails to maintain such telephone lines and electrical connections for fourteen (14) consecutive days after notification by the Company, the Company may discontinue service to the Customer.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 166 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 112 Delivery Terms and Conditions

- 9.5.2 Should a Customer or a Supplier request that additional telemetering equipment or a communication device be attached to the existing telemetering equipment in addition to that provided pursuant to Section 9.5.1, the Company shall install, test, and maintain the requested telemetering equipment or communication device; provided that such telemetering equipment or communication device does not interfere with the operation of the equipment required for the Company's purposes and otherwise meet the Company's requirements. The Customer or Supplier shall provide such telemetering equipment or communication device, unless the Company elects to do so. The Customer or Supplier shall bear the cost of providing and installing the telemetering equipment, communication device, or any other related equipment, and shall have electronic access to the Customer's Gas Usage information. Upon installation, the telemetering equipment or communication device shall become the property of the Company and will be maintained by the Company. The Company shall bill the Customer or Supplier after installation.
- 9.5.3 The Company shall complete installation of telemetering equipment and communication devices, if reasonably possible, within sixty (60) days of receiving a written request from the Customer or Supplier provided that the Customer completes the installation of any required telephone or electrical connections within ten (10) days of such request.
- 9.5.4 The Company may, at its sole discretion, bill the Customer on a calendar month or cycle month basis.

9.6 Balancing

9.6.1 The Supplier must maintain a balance between daily receipts and daily Gas Usage within the following tolerances:

Off-Peak Season: The difference between the Supplier's aggregate actual receipts on the

Transporting Pipeline to each Gas Service Area and the aggregated Gas Usage of Customers in the Aggregation Pool shall be within 15% of said receipts. The Supplier shall be charged 0.1 times the Daily Index

for all differences not within the 15% tolerance.

<u>Peak Season</u>: The difference between the Supplier's aggregate actual receipts on the

Transporting Pipeline to each Gas Service Area and the aggregated Gas Usage of Customers in the Aggregation Pool shall be within 10% of said receipts. The Supplier shall be charged 0.5 times the Daily Index

for all differences not within the 10% tolerance.

<u>Critical Day(s)</u>: The Company will determine if the Critical Day will be aggravated by

an under-delivery or an over-delivery, and so notify the Supplier when

a Critical Day is declared pursuant to Section 16.

Critical Day That Will Be Aggravated by Under-delivery.

<u>Supplier who under-delivers</u>. A Supplier who under-delivers on a Critical Day that will be aggravated by under-delivery shall be charged 5 times the Daily Index for the aggregated Gas Usage of Customers in the Aggregation Pool that exceeds 102% of the Supplier's aggregate actual receipts on the Transporting Pipeline to each Gas Service Area.

<u>Supplier who over-delivers</u>. A Supplier who over-delivers on a Critical Day that will be aggravated by under-delivery shall be charged 0.1 times the Daily Index to the

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 167 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 113 Delivery Terms and Conditions

extent that the difference between the Supplier's aggregate actual receipts on the Transporting Pipeline to each Gas Service Area and the aggregated Gas Usage of Customers in the Aggregation Pool exceeds 20% of said receipts [(Receipts - Usage) > (20% x Receipts)].

Critical Day That Will Be Aggravated by Over-delivery.

<u>Supplier who under-delivers</u>. A Supplier who under-delivers on a Critical Day that will be aggravated by over-delivery shall be charged 0.1 times the Daily Index to the extent that the difference between the Supplier's aggregated Gas Usage of Customers in the Aggregation Pool exceeds 120% of the Supplier's aggregate actual receipts on the Transporting Pipeline to each Gas Service Area.

<u>Supplier who over-delivers</u>. A Supplier who over-delivers on a Critical Day that will be aggravated by over-delivery shall be charged 5 times the Daily Index to the extent that the difference between the Supplier's actual receipts on the Transporting Pipeline to each Gas Service Area and the Supplier's aggregated Gas Usage of Customers in the Aggregation Pool exceeds 2% of said receipts [(Receipts - Usage > (2% x Receipts)].

Point Specific Balancing: In the event that the Transporting Pipeline requires its customers to balance on a point-specific basis, the Supplier must balance pursuant to this Section at each Designated Receipt Point.

- 9.6.2 If the Supplier has an accumulated imbalance within a Month, the Supplier may nominate to reconcile such imbalance, subject to the Company's approval, which approval shall not be unreasonably withheld.
- 9.6.3 In addition to the charges set forth in Section 9.6.1, the Company shall flow through to the Supplier any pipeline imbalance penalty charges attributable to the Supplier.
- 9.6.4 If, as a result of the Company interrupting or curtailing service pursuant to Section 17 of this tariff, the Supplier incurs a daily imbalance penalty due to over delivery, the Company will waive such penalty for the First Day of the interruption or curtailment period. If the Company has issued notice of an interruption or curtailment in service and the Supplier is unable to change its Nomination, or if the Supplier's Gas has been delivered to the Designated Receipt Point, then the Company will credit such Gas against the Supplier's imbalance.
- 9.6.5 The Supplier will maintain a balance between receipts at the Designated Receipt Point(s) and the aggregated Gas Usage of Customers in each Aggregation Pool. If the Transporting Pipeline posts notice on its electronic bulletin board that its customers will be required to adhere to a maximum hourly flow rate, the Supplier will be deemed to have notice that maximum hourly flows will be in effect on the Company's distribution facilities as of the same time and for the same period as maximum hourly flows are in effect on the Transporting Pipeline. The Supplier's maximum hourly flow will be established based on an allocation of even hourly flows of daily receipts of Gas scheduled in the relevant period in accordance with the applicable transportation tariff of the Transporting Pipeline. All Gas Usage in excess of the Supplier's maximum hourly flow rate shall be subject to an additional charge of 5 times the Daily Index for each Dekatherm in excess of the Supplier's maximum hourly flow. The Company will notify the Supplier of the Supplier's maximum hourly flow.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Docket No. DG 21-130 Exhibit 26 Page 168 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 114 Delivery Terms and Conditions

9.6.6 If, during any fifteen (15) consecutive Gas Days, the Supplier delivers an amount less than 70% of the sum of the aggregated Gas Usage of Customers in the Aggregation Pool in said Gas Days, the Company may declare the Supplier ineligible to nominate Gas for the following thirty (30) Gas Days. The Supplier shall have the opportunity to cure the imbalance with the demonstration of verifiable imbalance trades or otherwise within twenty-four (24) hours of notification by the Company. If the Supplier is declared ineligible to nominate Gas for such 30 Gas Days, the Supplier may be reinstated at the end of the 30 Gas Days, provided it posts security equal to the product of: (1) the maximum aggregate daily Gas Usage of Customers in the Aggregation Pool expressed in MMBtu and (2) \$300. If, within twelve (12) months of the first offense, such Supplier is declared ineligible to nominate Gas pursuant to this Section, the Supplier will be disqualified from service under this tariff for one (1) full year from the time of the second disqualification. If the Supplier defaults on its obligations under this tariff, the Company shall have the right to use such security to satisfy the Supplier's obligations. Such security may be used by the Company to secure Gas, transportation, and storage, and to cover other related costs incurred as a result of the Supplier's default. The security may also be used to satisfy any outstanding claims that the Company may have against the Supplier including imbalance charges, cash-out charges, pipeline penalty charges, and other charges.

9.7 Cash Out

For each Aggregation Pool, the Supplier must maintain total Monthly receipts within a reasonable tolerance of total Monthly Gas Usage. Any differences between total Monthly receipts for an Aggregation Pool and the aggregated Gas Usage of Customers in the Aggregation Pool, expressed as a percentage of total Monthly receipts, will be cashed out according to the following schedule:

Imbalance Tier	Over-deliveries	Under-deliveries
0% ≤ 5%	The average of the Daily Indices for the relevant Month.	The highest average of seven consecutive Daily Indices for the relevant Month
$> 5\% \le 10\%$	0.85 times the above stated rate.	1.15 times the above stated rate.
> 10% ≤ 15%	0.60 times the above stated rate.	1.4 times the above stated rate.
> 15%	0.25 times the above stated rate.	1.75 times the above stated rate.

For purposes of determining the tier at which an imbalance will be cashed out, the price will apply only to volumes within a tier. For example, if there is a 7% under-delivery on a Transporting Pipeline, volumes that make up the first 5% of the imbalance are priced at the highest average of the seven (7) consecutive Daily Indices. Volumes making up the remaining 2% of the imbalance are priced at 1.15 times the average of the seven (7) consecutive Daily Indices.

10 NON-DAILY METERED DELIVERY SERVICE

10.1 Applicability

Section 10 of this tariff applies to Customers taking Delivery Service under Rate Schedules G-41, G-42, G-51, G-52 and their Suppliers.

10.2 Delivery Service Provided

This service provides firm, year-round delivery of Customer purchased Gas from the Designated Receipt Point to the Delivery Point on any Gas Day for Customers, without the requirement of

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Docket No. DG 21-130 Exhibit 26 Page 169 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 115 Delivery Terms and Conditions

recording Gas Usage at the Delivery Point on a daily basis. Daily Nominations are calculated by the Company on the basis of a Consumption Algorithm and the Supplier is obligated to deliver to the Designated Receipt Point(s) such quantities.

- 10.3 Nominations and Scheduling of Service
 - 10.3.1 The Supplier is obligated to nominate and deliver the Adjusted Target Volume ("ATV"), as determined in Section 10.3.2, to the Designated Receipt Points on every Gas Day for each Aggregation Pool.
 - 10.3.2 The Company shall determine the ATV for each Aggregation Pool of Customers taking non-daily metered Delivery Service for each Gas Day using a Consumption Algorithm. The ATV shall include the Company Gas Allowance. On each Business Day, the Company will communicate, electronically, by facsimile, or by other agreeable alternative means, the forecasted ATV to the Supplier for at least the subsequent four (4) Gas Days. The ATV in effect for any Gas Day shall be the most recent ATV for that Gas Day communicated to the Supplier by the Company. The ATV for a given Gas Day shall not be effective unless it has been communicated to the Supplier at least two (2) hours prior to the Company's Supplier Nomination deadline for that Gas Day, which shall be at least two (2) hours prior to the deadline for nominations on the Transporting Pipeline, or such lesser period as determined by the Company.
 - 10.3.3 Nominations will be communicated to the Company electronically, by facsimile, or other agreeable alternative means.
 - 10.3.4 Nominations for the first Day of a Month shall be submitted to the Company no later than two (2) hours prior to the deadline for first of the Month nominations of the Delivering Pipeline or such lesser period as determined by the Company. The Company will make available, from time to time, a schedule of nomination due dates. Nominations on weekends, holidays, and non-business hours will be accepted by the Company on a basis consistent with that utilized for its own operations.
 - 10.3.5 The Supplier shall provide an intra-Month nomination no later than two (2) hours prior to the deadline of the Delivering Pipeline for the next Gas Day, or such lesser period as determined by the Company. Nominations on weekends, holidays, and non-business hours will be accepted by the Company on a basis consistent with that utilized for its own operations.
 - 10.3.6 Nominations will be conditionally accepted by the Company pending confirmation by the Transporting Pipeline. The Company will attempt to confirm the nominated volume with the Transporting Pipeline. In the event of a discrepancy between the volume nominated to the Company by the Supplier and the volume nominated by the Supplier to the Transporting Pipeline, the lower volume will be deemed confirmed. The Company will allocate such discrepancy based on a predetermined allocation method set forth in the Supplier Service Agreement. If no predetermined allocation method has been established prior to the event of such discrepancy, the Company will allocate the discrepancy on a pro rata basis. The Company will not confirm any volume nominated by the Supplier in excess of the ATV.
 - 10.3.7 In the event that the Supplier is unable to deliver a confirmed ATV Nomination, the Supplier may make intra-Gas Day Nominations relating to changes to existing Nominations within a given Gas Day no later than two (2) hours prior to the intra-Gas Day Nomination deadline for the Transporting Pipeline on which the Nomination is to be effective, or such lesser period as determined by the Company; provided, however, that the

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 170 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 116 Delivery Terms and Conditions

- 10.3.8 Nomination must be in conformance with the requirements of and must be permitted by the Transporting Pipeline. Intra-Gas Day Nominations on weekends, holidays, and non-business hours will be accepted by the Company on a basis consistent with that utilized by the Company for its own operations. The Company shall not adjust the ATV applied for the Gas Day.
- 10.3.9 Nominations may be rejected if they do not satisfy the conditions for Delivery Service in effect from time to time.
- 10.3.10 All quantities of Gas over-delivered or under-delivered to the Company's system in violation of an Operational Flow Order ("OFO") declared by the Company pursuant to Section 16 will be subject to the Critical Day provisions of Section 10.6.1 of this tariff, and the delivered quantity specified in the OFO will replace the ATV.
- 10.4 Determination of Receipts
 - 10.4.1 The quantity of Gas deemed received by the Company for the Supplier's Aggregation Pool at the Designated Receipt Point(s) will equal the volume so scheduled by the Transporting Pipeline(s).
 - 10.4.2 The Company Gas Allowance will be assessed against receipts pursuant to Section 8 of this tariff.
- 10.5 Metering and the Determination of Deliveries

The Company shall record the Customer's Gas Usage at the Delivery Point by making actual meter reads on a monthly [or bi-monthly] basis. In the event that the Customer's Gas Usage is metered on a bi-monthly basis, the Company shall make available to the Supplier estimates of the Customer's Gas Usage for each of the two billing months.

10.6 Balancing

10.6.1 Any difference between the Supplier's ATV for an Aggregation Pool and the receipts on the Transporting Pipeline to the appropriate Designated Receipt Point(s) will be cashed out by the Company according to the following:

Off-Peak Season: For receipts less than the ATV, the Supplier shall be charged 1.1 times

the Daily Index for the difference. For receipts greater than the ATV, the Supplier shall be charged 0.8 times the Daily Index for the

difference.

Peak Season: For receipts less than the ATV but greater than or equal to 95% of the

ATV, the Supplier shall be charged 1.1 times the Daily Index for the difference. For receipts less than 95% of the ATV, the Supplier shall be charged 1.1 times the Daily Index for the first 5% difference, and the Supplier shall be charged two (2) times the Daily Index for the remaining difference. For receipts greater than the ATV, the Supplier

shall be charged 0.8 times the Daily Index for the difference.

Critical Day(s) The Company will determine if the Critical Day will be aggravated by

an under-delivery or an over-delivery, and so notify the Supplier when

a Critical Day is declared pursuant to Section 16.

Critical Day That Will Be Aggravated by Under-delivery.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 171 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 117 Delivery Terms and Conditions

<u>Supplier who under-delivers</u>. A Supplier who under-delivers on a Critical Day that will be aggravated by under-delivery shall be charged five (5) times the Daily Index for the difference between the ATV and actual receipts.

<u>Supplier who over-delivers</u>. A Supplier who over-delivers on a Critical Day that will be aggravated by under-delivery shall be charged the following amounts for all receipts in excess of the ATV:

- (a) up to 25% in excess of the ATV, the Supplier shall be charged the Daily Index for the difference.
- (b) for receipts in excess of 25% above the ATV, the Supplier shall be charged 0.8 times the Daily Index for the difference.

Critical Day That Will Be Aggravated By Over-delivery.

<u>Supplier who over-delivers</u>. A Supplier who over-delivers on a Critical Day that will be aggravated by over-delivery shall be charged 0.4 times the Daily Index for receipts greater than the ATV.

<u>Supplier who under-delivers</u>. A Supplier who under-delivers on a Critical Day that will be aggravated by over-delivery shall be charged the following amounts--for receipts less than the ATV but greater than or equal to 75% of the ATV, the Supplier shall be charged the Daily Index for the first 25% difference, and the Supplier shall be charged 1.1 times the Daily Index for the remaining difference.

- 10.6.2 In addition to the charges set forth in Section 10.6.1, the Company shall use a daily balancing charge calculation to account for balancing costs it incurs in serving each Aggregation Pool due to differences in forecast versus actual Heating Degree Days. The daily balancing charge shall be based on the sum of the absolute values of the daily differences between the Aggregation Pool's ATV and the recalculated ATV value described in Section 10.7.1 below. Such charge shall be billed to the Supplier monthly and shall reflect the cost of resources used by the Company to balance such differences for each Gas Day of the Month. The Company shall calculate such charge annually in its Winter Season Cost of Gas filing according to a formula as set forth in Attachment B.
- 10.6.3 In addition to the charges set forth in Section 10.6.1, the Company shall use a daily balancing charge calculation to account for balancing costs it incurs in serving each Aggregation Pool due to differences in forecast versus actual Heating Degree Days. The daily balancing charge shall be based on the sum of the absolute values of the daily differences between the Aggregation Pool's ATV and the recalculated ATV value described in Section 10.7.1 below. Such charge shall be billed to the Supplier monthly and shall reflect the cost of resources used by the Company to balance such differences for each Gas Day of the Month. The Company shall calculate such charge annually in its Winter Season Cost of Gas filing according to a formula as set forth in Attachment B.

In the event that the Transporting Pipeline requires its customers to balance on a point-specific basis, the Supplier must balance pursuant to this Section at each Designated Receipt Point.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 172 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 118 Delivery Terms and Conditions

10.6.4 In addition to the charges set forth in Sections 10.6.1 and 10.6.2, the Company shall flow through to the Supplier any pipeline imbalance penalty charges attributable to the Supplier.

10.7 Cash Out

- 10.7.1 The Company shall use a daily cash out calculation to account for imbalances due to differences in forecast versus actual Heating Degree Days. Using a Consumption Algorithm, the Company will recalculate the ATV for each Aggregation Pool for each Gas Day of the Month, substituting actual Heating Degree Days for forecast Heating Degree Days. Daily recalculations shall be compared to the Aggregation Pool's daily ATV, and the difference shall be cashed out at 100% of the Daily Index.
- 10.7.2 During the billing months of both June and December, the Company shall use a six-month cash-out calculation to account for differences in forecast usage versus billed Gas Usage. The Company may cash-out differences in forecast usage versus billed usage at intervals that are less than six months as provided by the Supplier Service Agreement.
 - (1) In the billing month of June, using the recalculated ATV values described in Section 10.7.1, the Company will compare the sum of the recalculated ATV values for each Aggregation Pool for the six-month period of November 1 through April 30 to the sum of billed usage volumes used by each Aggregation Pool for that same period. The differences shall be cashed out at 100% of the average of the Daily Index weighted by actual Heating Degree Days over the same period. The Winter period cash-out shall be calculated and provided to Suppliers within 60 days of the month ending April 30.
 - (2) In the billing month of December, using the recalculated ATV values described in Section 10.7.1, the Company will compare the sum of the recalculated ATV values for each Aggregation Pool for the six-month period of May 1 through October 31 to the sum of the billed usage volumes used by each Aggregation Pool for that same period. The differences shall be cashed out at 100% of the average of the Daily Index over the same period. The Off-Peak period cash-out shall be calculated and provided to Suppliers within 60 days of the month ending October 31.
- 10.7.3 The Company shall allow Suppliers to trade seasonal differences. Prior to the seasonal cash-out, the Company shall make available a list of Suppliers. Aggregation Pools affected by the transaction must be located within the same Gas Service Area as defined in Section 4, unless waived by the Company. All trades must be communicated to the Company within three (3) Business Days following receipt of the list.
- 10.7.4 If, during any fifteen (15) consecutive Gas Days, the Supplier delivers an amount less than 70% of the sum of the ATVs of the Aggregation Pool in said Gas Days, the Company may declare the Supplier ineligible to nominate Gas for the following thirty (30) Gas Days. The Supplier shall have the opportunity to cure the imbalance with the demonstration of verifiable imbalance trades or otherwise within twenty-four (24) hours of notification by the Company. If the Supplier is declared ineligible to nominate Gas for such 30 Gas Days, the Supplier may be reinstated at the end of the 30 Gas Days, provided it posts security equal to the product of: (1) the Supplier's estimated maximum aggregate daily Gas Usage of Customers in the Aggregation Pool expressed in MMBtu and (2) \$300. If, within twelve (12) months of the first offense, such Supplier is declared ineligible to nominate Gas pursuant to this Section, the Supplier will be disqualified from service under this tariff for one (1) full year from the time of the second disqualification. If the Supplier defaults on its obligations under this tariff, the Company shall have the right to use such security to

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 173 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 119 Delivery Terms and Conditions

10.7.5 satisfy the Supplier's obligations. Such security may be used by the Company to secure Gas, transportation, storage, and to cover other related costs incurred as a result of the Supplier's default. The security may also be used to satisfy any outstanding claims that the Company may have against the Supplier including imbalance charges, cash-out charges, pipeline penalty charges, and other charges.

11 CAPACITY ASSIGNMENT

11.1 Applicability

Section 11 of this tariff applies to all Suppliers that have enrolled one or more Customers into one or more Aggregation Pools and shall include Customers acting as their own Supplier. The Company shall assign and the Supplier shall accept each Customer's pro-rata share of Capacity, if any, as established in accordance with this Section.

11.2 Identification of Capacity for Assignment

- 11.2.1 On or before September 15 of each year, the Company shall communicate, by electronic means as determined by the Company or, in the event of failure of such electronic means, by facsimile or other agreeable alternative means, the Capacity to be made available for assignment to Suppliers on each of twelve Assignment Dates beginning in October.
- 11.2.2 The Company shall identify, by Gas Service Area, the specific contracts and resources for assignment to Suppliers based on the Company's Capacity and resource plans. Such identified contracts and resources shall be used to determine the pro-rata shares of Capacity assignable to a Supplier on behalf of the Customers enrolled in its Aggregation Pool.
- 11.2.3 Capacity assigned by the Company may include Company-Managed Supplies that effectuate, at maximum tariff rates, the assignment of certain Capacity contracts including Canadian, Federal Energy Regulatory Commission, 15 U.S.C. § 717(c) or Section 7(c) [Part 157 of the FERC regulations (18 C.F.R. part 157)] and other contracts that are not assignable to third-parties due to governing tariffs.

11.3 Determination of Pro-Rata Shares of Capacity

- 11.3.2 The Company shall establish a Total Capacity Quantity ("TCQ") for each Customer taking Delivery Service. The TCQ represents the total amount of Capacity assignable to a Supplier on behalf of a Customer.
- 11.3.3 For a Customer receiving Sales Service on or after March 14, 2000, the TCQ shall be the Customer's estimated Gas Usage on the Peak Day as determined by the Company each October prior to the Customer's enrollment into Supplier Service. The Company shall derive such estimate using a Daily Baseload and a Heating Factor based upon the Customer's historic Gas Usage during the Reference Period, or the best estimates available to the Company should actual Gas Usage information be partially or wholly unavailable.
- 11.3.4 For a Customer that was either receiving Supplier Service (or the equivalent form of service at the time) on March 14, 2000, or had an executed contract for firm transportation service (i.e., the equivalent of Delivery Service) on file with the Company on or before March 14, 2000, the TCQ shall be zero.
- 11.3.5 A Customer that was either receiving Supplier Service (or the equivalent form of service at the time) on March 14, 2000, or had a written request on file with the Company on or before March 14, 2000 may elect for its Supplier to accept assignment of its pro-rata share of Capacity as determined by the Company in accordance with Section 11.2 and, subject

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 174 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 120 Delivery Terms and Conditions

to availability, as determined by the Company in its sole reasonable discretion. In order to make such election, the Customer must have submitted to the Company, on or before ten (10) days prior to the first Assignment Date prior to the original effective date of this tariff, a completed application for Capacity that is signed by both the Customer and Supplier. All assignments of Capacity made on behalf of such electing Customer shall be executed in accordance with Sections 11 and 14 of this tariff as if the Customer had been receiving Sales Service on or after March 14, 2000

- 11.3.5 For a new Customer taking Supplier Service as its initial service after March 14, 2000, the TCQ shall be zero except in cases where the Customer is a new Customer of record at a meter location where a former Customer of received firm service from the Company any time during the preceding twenty-four (24) months, in which case the TCQ established by the Company for the former Customer shall become the TCQ for the new Customer. The Company may reduce said TCQ value for the new Customer, if, in its sole reasonable discretion, the Company determines that the old Customer's TCQ exceeds the new Customer's estimated future consumption on the Peak Day. In the event that Sales Service is provided at a new meter location for Gas Usage associated with new construction, the TCQ shall be zero, provided that the Customer initiates Supplier Service upon the completion of said new construction in accordance with Section 20.5 of this tariff
- 11.3.6 Once the Company establishes a TCQ for a Customer pursuant to this Section 11.3, it shall remain in effect for the purpose of determining the Customer's pro-rata shares of Capacity until such time that the Customer returns to Sales Service. The Company shall establish a new TCQ value for the Customer pursuant to Section 11.3.2 if the Customer again elects to take Supplier Service after returning to Sales Service, unless otherwise established herein..
- 11.3.7 The Company shall determine the pro-rata shares of Pipeline Capacity, Storage Withdrawal Capacity, and Peaking Capacity assignable to a Supplier on behalf of a Customer as the product of the Customer's TCQ times the applicable Capacity Allocators. The Capacity Allocators for each class of Customers billed under the Company's Rate Schedule shall be set forth annually in Attachment C to this tariff.
- 11.3.8 The Company shall determine the pro-rata share of Seasonal Storage Capacity assignable to a Supplier on behalf of a Customer consistent with the tariffs governing the associated Storage Withdrawal Capacity.
- 11.3.9 The Company shall determine the pro-rata shares of Peaking Supply assignable to a Supplier in accordance with Section 14 of this tariff.

11.4 Capacity Assignments

- 11.4.1 On each Assignment Date, the Company will assign to the Supplier the pro-rata shares of Capacity on behalf of each Customer as determined by the Company in accordance with Sections 11.2, 11.3 and 11.7.
- 11.4.2 The total amount of Pipeline Capacity, Storage Withdrawal Capacity, and Peaking Capacity assigned to the Supplier on behalf of the Customers in an Aggregation Pool shall be at least equal to the cumulative sum of the pro-rata shares of Pipeline Capacity, Storage Withdrawal Capacity, and Peaking Capacity for all Customers enrolled in said Aggregation Pool as of Five (5) Business Days prior to the Assignment Date.
- 11.4.3 Storage Withdrawal Capacity shall be subject to Operational Flow Orders that are issued by the Company pursuant to Section 16 of this tariff, in the event that the Company requires the Supplier to deliver or to store quantities of Gas for the purposes of managing system

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 175 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 121 Delivery Terms and Conditions

- imbalances and maintaining Delivery Service. Whenever the Company assigns incremental Storage Withdrawal Capacity to the Supplier, the Company shall also assign to that Supplier additional Seasonal Storage Capacity pursuant to Section 11.8.
- 11.4.4 The Peaking Capacity assigned to the Supplier shall establish the Maximum Daily Peaking Quantity ("MDPQ") for the Aggregation Pool in the Supplier's Service Agreement. In the event that the Company increases a Supplier's MDPQ, the Company shall also assign to that Supplier additional Peaking Supply pursuant to Section 14.
- 11.4.5 The Company shall execute Capacity assignments in increments of 200 MMBtus. The Supplier shall accept an initial increment of Capacity on the first Assignment Date when the sum of the pro-rata shares of Capacity assigned to the Supplier pursuant to Section 11.4.1 exceeds 150 MMBtus. The Supplier shall accept additional increments of Capacity on the following Assignment Dates commensurate with any cumulative increase in the sum of pro-rata shares of Capacity assigned to the Supplier, as rounded to the nearest 200 MMBtus. Each increment of Capacity accepted by the Supplier shall comprise Pipeline Capacity, Storage Withdrawal Capacity, and Peaking Capacity in proportion to the cumulative increase of the pro-rata shares of assigned Capacity as established in accordance with Section 11.4.1. Section 11.4.2 shall not apply to a Customer that is acting as its own Supplier.
- 11.4.6 If a Customer is acting as its own Supplier, the Company shall assign Capacity to the Customer in an amount equal to the Customer's TCQ, as established pursuant to Section 11.3.

11.5 Release of Contracts

- 11.5.1 With the exception of Company-Managed Supplies and Peaking Capacity, Capacity contracts shall be released by the Company to the Supplier, at the maximum tariff rate or lesser rate paid by the Company and including all surcharges, through pre-arranged Capacity releases, pursuant to applicable laws and regulations and the terms of the governing tariffs.
- 11.5.2 Capacity contracts released to a Supplier on an Assignment Date shall be released for a term beginning on the first Gas Day of the Month following the Assignment Date through the expiration date of the respective capacity contract being assigned. and ending on October 31. For example, contracts assigned to a Supplier on April 25 of a given year shall be released for a term beginning on May 1 of that year and ending on October 31 of that year.
- 11.5.3 The Company reserves the right to adjust releases of Storage Withdrawal Capacity in the event that fifty percent (50%) or more of the total Storage Withdrawal Capacity serving a Gas Service Area has been assigned to Suppliers. Such adjustments may include, but are not limited to, the reassignment of certain Storage Withdrawal Capacity as Company-Managed Supplies in order for the Company to maintain operational control over Capacity resources associated with system balancing, and/or the retention of specific Capacity resources associated with system balancing and the implementation of a balancing charge to offset the associated costs.

11.6 Annual Reassignment of Capacity

11.6.1 On each Annual Reassignment Date, the Company shall adjust the Capacity assignments previously made to a Supplier to conform with the Company's resource and requirements plans. Such previously assigned Capacity shall be replaced by the assignment to the Supplier of the pro-rata shares of the same or similarly situated Capacity on behalf of the

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 176 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 122 Delivery Terms and Conditions

Customers enrolled in the Supplier's Aggregation Pools (as of the first Gas Day of the Month following the Annual Reassignment Date).

- 11.6.2 If the reassignment of Storage Withdrawal Capacity requires adjustments to the Seasonal Storage Capacity previously assigned to a Supplier, the Company shall reassign Seasonal Storage Capacity to such Supplier, and the Company and the Supplier shall address any associated increments and decrements to inventories in place pursuant to Section 11.8 of this tariff.
- 11.6.3 If the reassignment of Peaking Capacity requires adjustments to the MDPQ for the Supplier's Aggregation Pool, the Company shall reassign Peaking Supply to such Supplier, and the Company and the Supplier shall address any associated increments and decrements to supplies pursuant to Section 14 of this tariff.

11.7 Recall of Capacity

- 11.7.1 If the pro-rata shares of Capacity assignable to a Supplier decline because one or more of the Supplier's Customers has returned to Sales Service, the Company shall have the right, but not the obligation, to recall from the Supplier the pro-rata shares of Capacity previously assigned to the Supplier on behalf of such Customers. The decision on whether to exercise its Capacity-recall rights shall be made by the Company in its sole reasonable discretion. If the Company elects to recall Capacity from a Supplier pursuant to this Section, such recall shall be made on the Assignment Date following the effective date of the Customer's return to Sales Service. Notwithstanding the foregoing, in the following circumstances the Company shall be required to recall Capacity associated with Customers returning to Sales Service:
 - (a) The Supplier returning the Customers to Sales Service certifies that it is ceasing all business operations in New Hampshire;
 - (b) The Supplier returning the Customers to Sales Service certifies that it will no longer offer service to a particular market sector (e.g., small commercial and industrial Customers) and, therefore, once such Customers are returned to Sales Service, the Supplier is not eligible to re-enroll Customers of that type; or
 - (c) The Supplier demonstrates that it has provided Supplier Service to the Customer for a 12-month period, and for a period of no less than any 12-month increment, prior to the Customer's return to Sales Service.
- 11.7.2 If the Company elects to recall Storage Withdrawal Capacity from the Supplier pursuant to this Section, the Company shall reduce the Seasonal Storage Capacity associated with the affected Aggregation Pool in accordance with Section 11.8 of this tariff. If the Company elects to reduce the MDPQ in the Supplier Service Agreement, the Company shall reduce the Peaking Supply associated with the affected Aggregation Pool in accordance with Section 14 of this tariff.
- 11.7.3 In the event that a Customer in a Supplier's Aggregation Pool switches to another Supplier, the Company shall recall from the former Supplier said Customer's pro-rata shares of Capacity for reassignment to the new Supplier pursuant to Section 11.4. There shall be no change in the Customer's TCQ used to determine the Customer's pro-rata shares of Capacity for reassignment to the new Supplier. The recall of such Capacity from the Customer's former Supplier and the assignment of Capacity to the new Supplier shall be made on the Assignment Date following the effective date of the Customer's switch in Suppliers.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 177 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 123
Delivery Terms and Conditions

If the Company recalls Storage Withdrawal Capacity from the Customer's former Supplier, the Company shall reduce the Seasonal Storage Capacity associated with the affected Aggregation Pool in accordance with Section 11.8 of this tariff. If the Company reduces the MDPQ in the Customer's former Supplier's Service Agreement, the Company shall also reduce the Peaking Supply associated with the affected Aggregation Pool in accordance with Section 14 of this tariff.

11.7.4 The recall of Capacity by the Company shall entail the recall of released contracts pursuant to governing tariffs and/or the reduction in assigned quantities set forth in the Supplier Service Agreement. The recall of Capacity shall be executed in decrements of 200 MMBtus, commensurate with the cumulative reduction in the pro-rata shares of Capacity assigned to the Supplier, rounded to the nearest 200 MMBtus. Each decrement of Capacity assigned to the Supplier shall comprise Pipeline Capacity, Storage Withdrawal Capacity, and Peaking Capacity in proportion to the cumulative decrease in the pro-rata shares of Capacity recalled from the Supplier.

In the event that a Supplier is declared ineligible to nominate Gas for thirty (30) Gas Days pursuant to Sections 9.6.6 or 10.7.4 of this tariff, the Company shall have the right to recall any or all Capacity assigned to said Supplier. If the Supplier is reinstated at the end of such 30 Gas Days, the Company shall reassign Capacity to the Supplier on the next Assignment Date pursuant to Sections 11.4 and 11.5. There shall be no change in the TCQ values used to determine the Supplier's Customers' pro-rata shares of Capacity for reassignment.

- 11.7.5 In the event that a Supplier is disqualified from service for a one (1) full year pursuant to Sections 9.6.6 or 10.7.4 of this tariff, the Company shall have the right to recall any or all Capacity assigned to said Supplier. If the Supplier is reinstated at the end of such period, the Company shall reassign Capacity to the Supplier on the next Assignment Date pursuant to Sections 11.4 and 11.5. There shall be no change in the TCQ values used to determine the Supplier's Customers' pro rata shares of Capacity reassignments.
- 11.7.6 In the event that the Supplier fails to meet the applicable registration and licensing requirements established by law or regulation, fails to satisfy the requirements and practices as set forth in Section 20.3 of this tariff, fails to be and remain an approved shipper on the upstream pipelines and underground storage facilities on which the Company will assign capacity, fails to make timely payment under the assigned contracts, or fails to comply with or perform any of the obligations on its part established in this tariff or in the Supplier Service Agreement, the Company shall have the right to recall permanently any or all Capacity assigned to said Supplier. This section shall also apply to a Customer acting as its own Supplier.
- 11.7.7 The Supplier shall forfeit its rights to Capacity recalled by the Company pursuant to this Section. Such forfeiture shall be effected in accordance with applicable laws and regulations and the governing tariffs. In the event of Capacity forfeiture pursuant to this Section, the Supplier shall be responsible to compensate the Company for any payments due under the contracts prior to forfeiture, as well as any interest due thereon. The Company will not exercise discretion in the application of the forfeiture provisions of this Section. This section shall also apply to a Customer acting as its own Supplier.

11.8 Seasonal Storage Capacity

11.8.1 On each Assignment Date, the Company shall release Seasonal Storage Capacity to a Supplier that accepts the assignment of Storage Withdrawal Capacity pursuant to Section 11.4. The Company shall assign such Seasonal Storage Capacity consistent with the tariffs governing the release of the associated Storage Withdrawal Capacity.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u> Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 178 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 124 Delivery Terms and Conditions

- 11.8.2 If the Company assigns Seasonal Storage Capacity to a Supplier pursuant to Section 11.8.1 above, the Company shall transfer in-place Gas inventories to the Supplier. The quantity of inventories to be transferred from the Company to the Supplier shall be determined by multiplying the incremental Seasonal Storage Capacity assigned to the Supplier on the Assignment Date times the applicable storage inventory percentage described in Section 11.8.5. The Supplier shall be charged the Company's weighted average cost of inventories in off-system storage facilities for each Dekatherm transferred from the Company to the Supplier. The Company shall communicate, by electronic means as determined by the Company or, in the event of failure of such electronic means, by facsimile or other agreeable alternative means, the Company's weighted average cost of inventories, by Gas Service Area, at least two Business Days prior to each Assignment Date.
- 11.8.3 In the event that the Company recalls Storage Withdrawal Capacity from the Supplier pursuant to Section 11.7, the Company shall also recall Seasonal Storage Capacity from the Supplier. The Company shall determine the total Seasonal Storage Capacity to be recalled from the Supplier in accordance with the tariffs governing the Storage Withdrawal Capacity returned to the Company.
- 11.8.4 If the Company recalls Seasonal Storage Capacity from a Supplier pursuant to Section 11.8.3, the Supplier shall transfer in-place Gas inventories to the Company. The quantity of inventories to be transferred from the Supplier to the Company shall be determined by multiplying the decremental Seasonal Storage Capacity times the applicable storage inventory percentage described in Section 11.8.5. The Supplier shall be reimbursed at the Company's weighted average cost of inventories in off-system storage facilities as of the Assignment Date, for each Dekatherm transferred from the Supplier to the Company. The Company shall communicate, by electronic means as determined by the Company or, in the event of failure of such electronic means, by facsimile or other agreeable alternative means, the Company's weighted average cost of inventories, by Gas Service Area, at least two (2) Business Days prior to each Assignment Date.
- 11.8.5 Seasonal storage inventory percentages shall represent the amount of Seasonal Storage Capacity in each assigned storage resource that is assumed to be filled with inventories as of the first Gas Day of the month following the Assignment Date. Each September, the Company shall communicate, by electronic means as determined by the Company or, in the event of failure of such electronic means, by facsimile or other agreeable alternative means, the storage inventory percentages for each resource that shall be applied to incremental or decremental Seasonal Storage Capacity assignments executed on each of the twelve (12) Assignment Dates beginning in October.

11.9 Company-Managed Supplies

- 11.9.1 The Company shall provide access to and ascribe cost responsibility for the pro-rata shares of certain Capacity contracts including Canadian, Federal Energy Regulatory Commission, 15 U.S.C. § 717(c) or Section 7(c) [Part 157 of the FERC regulations (18 C.F.R. part 157)], and other contracts that are not assignable to third-parties.
- 11.9.2 The Supplier's Service Agreement shall set forth the quantity of each Company-Managed Supply assigned to the Supplier pursuant to Sections 11.4 and 11.8.
- 11.9.3 The Company shall notify the Supplier of the conditions and/or restrictions on the use of Company-Managed Supplies pursuant to the tariffs governing the resources.
- 11.9.4 The Company shall invoice the Supplier for its pro-rata shares of the demand charges for Capacity contracts assigned to the Supplier as Company-Managed Supplies. The Company

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 179 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 125 Delivery Terms and Conditions

shall also flow through to the Supplier all costs, including Supply costs, incurred from the utilization of Company-Managed Supplies on behalf of the Supplier.

11.9.5 The Company shall nominate quantities to the Transporting Pipeline and/or other interstate pipelines and off-system storage operators on behalf of Suppliers to which the Company has assigned Company-Managed Supplies, provided that the requested Nomination conforms to the tariffs governing the resource. The Supplier shall communicate its desired Nomination quantities to the Company subject to the provisions in Sections 9.3 and 10.3 of this tariff.

11.10 Capacity Mitigation Service

- 11.10.1 Capacity Mitigation Service is available to Suppliers that have been assigned Capacity pursuant to Section 11 of this tariff. Such Suppliers shall have the option to take Capacity Mitigation Service from the Company for contracts that would otherwise be released to the Supplier in accordance with this tariff.
- 11.10.2 Within five (5) Business Days prior to the Annual Reassignment Date, the Supplier must designate those contracts that would otherwise be released to the Supplier pursuant to Section 11.5, as contracts to be managed by the Company for cost mitigation in accordance with the Company's Capacity Mitigation Service. Such designation will be effective for the period November 1 through October 31. Such notice shall be communicated in accordance with the Supplier's Service Agreement.
- 11.10.3 The Supplier shall pay to the Company the maximum-tariff rate or lesser rate paid by the Company, including all surcharges, for the Capacity contracts that are retained and managed by the Company. The Company shall bill the Supplier monthly for such charges.
- 11.10.4 The Company will market Capacity contracts designated by Suppliers for mitigation through the Capacity Mitigation Service. The Supplier shall receive a credit on its bill for Capacity Mitigation Service equal to the pro-rata share of the proceeds earned from the Company in exchange for such contract management. Such credit shall be determined on a contract-specific basis at the end of each Month and will be included in the bill sent to the Supplier in the following Month.

12 BILLING AND SECURITY DEPOSITS

- 12.1 The Customer shall be responsible for all charges for service furnished by the Company under the Company's applicable rates, as filed from time to time with the NHPUC, from the time service is commenced until it is terminated. The Company shall provide a single bill, reflecting unbundled charges, to Customers for Sales Service.
- 12.2 The Company shall offer two billing service options to Customers taking only Delivery Service: standard complete billing service and standard pass-through billing service. The Supplier shall inform the Company of the selected billing option in accordance with the provisions set forth in Section 20.5
 - 12.2.1 Standard Complete Billing Service

The Customer shall receive a single bill from the Company for both Delivery Service and Supplier Service. The Company shall use the rates supplied by the Supplier to calculate the Supplier's portion of the single bill and integrate this billing within a single mailing to the Customer. The Company may charge a fee to the Supplier for providing this billing service as approved by the NHPUC.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 180 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 126 Delivery Terms and Conditions

The Supplier shall adhere to the Customer classes and rate structure as specified in the Company's then current Rate Schedule on file with and approved by the NHPUC. The Company shall reasonably accommodate, at the Supplier's expense, different Customer classes or rate structures as agreed to by the Company and the Supplier in the Supplier Service Agreement.

The Company shall provide an electronic file to the Supplier that will, in addition to the usage being billed, contain the calculated Supplier billing amounts for the current billing cycle. Customer revenue due the Supplier shall be transferred to the Supplier in accordance with the Supplier Service Agreement. Upon receipt of Customer payments, the Company shall provide a file for the Supplier summarizing all revenue from Supplier sales which have been received and recorded that day.

If a Customer pays the Company less than the full amount billed, the Company shall apply the payment first to Delivery Service, and if any payment remains, it shall be applied to Supplier Service.

12.2.2 Standard Pass-through Billing Service

The Customer taking Delivery Service shall receive two (2) bills: the Company shall issue one bill for Delivery Service and the Supplier shall issue a second bill for Supplier Service.

The Supplier shall be responsible for the collection of amounts due to the Supplier from the Customer. Customer payment responsibility with Suppliers shall be governed by the particular Customer/Supplier contract.

Within three (3) Business Days following the end of the Customer's billing cycle, the Company shall provide an electronic file for the Supplier that will contain the Customer's usage being billed including the current and previous meter readings.

- 12.2.3 The Company shall inform a Customer when Supplier Service has been initiated by a Supplier along with information on how the Customer may file a complaint regarding an unauthorized initiation of Service. This information shall be included on the first bill rendered to the Customer after such initiation.
- 12.2.4 A Customer acting as its own Supplier will be subject to the billing and payment requirements in Section 20.8 of this tariff.
- 12.2.5 Readings taken by an automated meter reading device will be considered actual readings for billing purposes.

13 SALES SERVICE

- 13.1 Sales Service is the Commodity service provided by the Company for Customers not electing to subscribe to Supplier Service and shall be provided by the Company, or its designated Supplier, in accordance with this tariff. Each Customer receiving Sales Service shall receive one bill from the Company reflecting delivery and Commodity charges.
- 13.2 A Customer receiving Sales Service on March 14, 2000 shall continue to receive Sales Service unless the Customer elects to take Supplier Service and until such time that Supplier Service is initiated for the Customer in accordance with Section 20.5 of this tariff. If the Customer terminates Supplier Service, if a Supplier terminates service to the Customer, or if the Customer's designated Supplier becomes ineligible to serve the Customer pursuant to Sections 9.6.6, 10.7.4, or 20.3 of this tariff, the Company will provide Sales Service to the Customer. Pursuant to Section 20.5 of this tariff, the Company will initiate Sales Service for the Customer and will provide Sales Service to the Customer until such time that Supplier Service is initiated for the Customer by a new Supplier.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 181 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 127 Delivery Terms and Conditions

- Any Customer whose Supplier has been assigned Capacity on behalf of said Customer pursuant to Section 11 of this tariff may elect to return to Sales Service if the Customer is no longer receiving Supplier Service. If necessary, the Company will initiate Sales Service for the Customer pursuant to Section 20.5 of this tariff and will provide the Customer with Sales Service until such time that Supplier Service is initiated for the Customer by a new Supplier. The Company will provide Sales Service to said Customer up to a maximum daily level of Gas Usage not to exceed the Total Capacity Quantity ("TCQ") of recallable Capacity assigned to the Customer's former Supplier.
- In the event that a Supplier that has been assigned Capacity on behalf of a Customer pursuant to Section 11 of this tariff terminates Supplier Service to the Customer, the Customer may select another Supplier. If necessary, the Company will initiate Sales Service for the Customer pursuant to Section 20.5 of this tariff and will provide the Customer with Sales Service until Supplier Service is initiated for the Customer by a new Supplier. The Company will provide Sales Service to the Customer up to a maximum daily level of Gas Usage not to exceed the TCQ of recallable Capacity assigned to the Customer's former Supplier.
- 13.5 In the event that a Supplier that has been assigned Capacity on behalf of a Customer pursuant to Section 11 of this tariff becomes ineligible to serve the Customer pursuant to Sections 9.6.6, 10.7.4, or 20.3 of this tariff, the Company will provide the Customer with Sales Service up to a maximum daily level of Gas Usage not to exceed the TCQ of recallable Capacity assigned to the Customer's Supplier.
- 13.6 The Company shall be under no obligation to provide Sales Service to a Customer at a maximum daily level in excess of the TCQ of recallable Capacity assigned to a Supplier on behalf of the Customer. The Company may elect to provide Sales Service to the Customer if, and to the extent that, adequate system Capacity and Supplies are available and upon the same terms and subject to the same conditions as any new Customer seeking to take Sales Service.

14 PEAKING SERVICE

14.1 Applicability

Section 14 of this tariff applies to all Suppliers, and to all Customers acting as their own Supplier, that have been assigned, or have elected to be assigned, Capacity on behalf of themselves or Customers in their Aggregation Pools pursuant to Section 11 of this tariff.

14.2 Character of Service

- 14.2.1 Peaking Service shall be provided by the Company subject to an executed Supplier Service Agreement that sets forth the Maximum Daily Peaking Quantity ("MDPQ") and the assigned Peaking Supply for each of the Supplier's Aggregation Pools.
- 14.2.2 The Company shall provide quantities of Gas, at the Supplier's request, from the Supplier's Peaking Service Account as established in accordance with Section 14.4. Such quantities shall be deemed delivered by the Company and received by the Company at the Designated Receipt Point(s) for the Aggregation Pool. Peaking Service shall be firm and available to the Supplier each Gas Day in accordance with the balance of the Supplier's Peaking Service Account and the parameters of the Company's Peaking Service Rule Curve.

14.3 Rates and Charges

14.3.1 The applicable rates for Peaking Service shall be established in the Company's tariff. The Supplier shall pay a peaking demand charge based on its MDPQ of assigned Peaking Capacity as billed by the Company for the Peak Season. Such unit demand charge shall be equal to the total Capacity costs and other fixed costs associated with the Company's

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u> Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 182 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 128 Delivery Terms and Conditions

peaking resources, excluding costs collected through Delivery rates, divided by the estimated peaking resources needed to meet the Company's total system Peak Day requirement.

14.3.2 The Supplier shall pay a Commodity charge equal to the estimated weighted average cost of peaking supplies, including fuel retention and carrying charges. The Company shall communicate electronically, by facsimile or by other agreeable alternative means the Company's estimated weighted average cost of peaking supplies by the 15th of the month preceding the next Assignment Date. The Commodity charge will be multiplied by the volumes of Peaking Service Gas nominated by the Supplier during each Month.

14.4 Peaking Supply

- 14.4.1 The Customer's portion of the Peaking Supply that shall be assigned to the Supplier on behalf of the Customer shall be equal to the Peaking Supply multiplied by the ratio of the Customer's MDPQ to the aggregate MDPQ of the total system.
- 14.4.2 On each Assignment Date, the Company shall assign Peaking Supply to a Supplier whose MDPQ has been increased pursuant to Section 11.4. If the Company assigns incremental Peaking Supply to a Supplier, the Company shall credit the balance of the Supplier's Peaking Service Account for volumes available through October 31 in accordance with the Peaking Service Rule Curve. The amount credited to the Supplier's Peaking Service Account shall be determined by multiplying the incremental Peaking Supply by the peaking inventory percentage described in Section 14.4.5.
- 14.4.3 On each Assignment Date, the Company shall recall Peaking Supply from a Supplier whose MDPQ has been decreased pursuant to Section 11.7. The Company shall determine the Supplier's total Peaking Supply for recall to be equal to the difference between the cumulative total Peaking Supply assigned to the Supplier as of the previous Assignment Date and the total Peaking Supply that is assignable to the Supplier in accordance with Section 14.4.1 above.
- 14.4.4 If the Company recalls Peaking Supply from a Supplier pursuant to Section 14.4.3, the Company shall debit the balance of the Supplier's Peaking Service Account for volumes available through October 31 in accordance with the Peaking Service Rule Curve. The amount debited from the Supplier's Peaking Service Account shall be determined by multiplying the decremental Peaking Supply by the peaking inventory percentage described in Section 14.4.5.
- 14.4.5 The peaking inventory percentage shall represent the level of Peaking Supply assumed to be available to a Supplier in its Peaking Service Account as of the first Gas Day of the Month following the Assignment Date for incremental and decremental assignments of Peaking Supply. Each September, the Company shall communicate electronically, by facsimile or by other agreeable alternative means the Peaking Inventory Percentages that shall be applied to incremental or decremental Peaking Supply assignments executed on each of the twelve (12) Assignment Dates beginning in October.
- 14.4.6 On each Annual Reassignment Date, the Company shall reset the balance in the Supplier's Peaking Service Account to equal the total Peaking Supply assignable to the Supplier on behalf of Customers enrolled in its Aggregation Pool (as of the first Gas Day of the Month following the Annual Reassignment Date) as determined in accordance with Section 14.4.1 above.
- 14.5 Nomination of Peaking Service

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 183 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 129 Delivery Terms and Conditions

- 14.5.1 The Supplier shall nominate with the Company the quantity of Peaking Supply, not in excess of the amount determined pursuant to Section 14.4.2, that the Supplier desires to be provided from its Peaking Service Account for the applicable Gas Day. For an Aggregation Pool of Customers taking daily metered Delivery Service, the notice given by the Supplier to the Company for an applicable Gas Day shall be made in accordance with Section 9.3 of this tariff. For an Aggregation Pool of Customers taking non-daily metered Delivery Service, the notice given by the Supplier to the Company for an applicable Gas Day shall be made in accordance with Section 10.3 of this tariff.
- 14.5.2 In response to a valid Nomination for Peaking Service, the Company shall provide the requested quantity of Gas, which shall be deemed to be delivered by the Company and received by the Company at the Designated Receipt Point(s) of the Supplier's Aggregation Pool, subject to the limitations herein. Nominated quantities shall be included in the determination of receipts at the Designated Receipt Point(s) for the Supplier's Aggregation Pool which factors into the daily balancing provisions set forth in this tariff.
- 14.5.3 The Company may reject a Supplier's Nomination for Peaking Service if the nominated quantity would cause the balance of the Supplier's Peaking Service Account to fall to a level that is 10% or more below the minimum allowable account balance for the Month in which the Nomination is requested, as computed in accordance with the Peaking Service Rule Curve. Under such circumstances, the Company shall require the Supplier to nominate the pipeline and/or storage resources, within the contract entitlements assigned to the Supplier under Section 11, required to maintain the Supplier's Peaking Service Account above the minimum allowable account balance described above. The balance of the Supplier's Peaking Service Account may not in any event fall below zero (0).
- 14.5.4 The Company shall provide Peaking Service supplies to the Supplier only when the volumes in the Peaking Service Account for the Aggregation Pool are greater than zero (0).

14.6 Peaking Service Critical Day Provisions

- 14.6.1 In the event that the volumes in a Supplier's Peaking Service Account for an Aggregation Pool are reduced to a level below the minimum allowable account balance as computed in accordance with the Company's Peaking Service Rule Curve, the Company may issue an OFO to such Supplier pursuant to Section 16 of this tariff.
- 14.6.2 In the event that the total volumes of all Peaking Service Accounts within one or more of the Company's Gas Service Areas are reduced to levels below the total minimum allowable account balances as computed in accordance with the Company's Peaking Service Rule Curve, the Company may declare a Critical Day and issue a blanket OFO pursuant to Section 16 of this tariff.
- 14.6.3 If, on a Critical Day, the Company projects, based on the Supplier's Nominations, that the Supplier's scheduled deliveries to the Designated Receipt Point(s) of an Aggregation Pool are less than the maximum feasible volumes for deliveries on the Transporting Pipeline, the Company may issue an OFO to the Supplier in accordance with Section 16 of this tariff.

15 DISCONTINUANCE OF SERVICE

15.1 The Company shall notify a Customer's Supplier of record that it has initiated any applicable billing and termination procedures as prescribed by the NHPUC. In the event that the Company discontinues Delivery Service to a Customer in accordance with the provisions set forth above, the Company shall provide electronic notification to the Customer's Supplier of record upon final

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u> Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 184 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 130 Delivery Terms and Conditions

billing to the Customer. The Company shall not be liable for any revenue loss to the Supplier as a result of any such disconnection.

16 OPERATIONAL FLOW ORDERS AND CRITICAL DAYS

- 16.1 In the event of a material and significant threat to the operational integrity of the Company's system, the Company may declare a Critical Day.
- 16.2 Circumstances constituting a threat to the operational integrity of the system that may cause the Company to declare a Critical Day shall include, but not be limited to: (1) a failure of the Company's distribution, storage, or production facilities; (2) near-maximum utilization of the Company's distribution, storage, production, and Supply resources; (3) inability to fulfill firm service obligations; and (4) issuance of an OFO or similar notice by upstream transporters.
- In the event that the Company has declared a Critical Day, the Company will have the right to issue an Operational Flow Order ("OFO") in which the Company may instruct Suppliers to take such action as conditions require, including, but not limited to, diverting Gas to or from the Company's distribution system, within the contract entitlements, if any, assigned to the Supplier under Section 11 hereof. An OFO may be issued on a pipeline or point-specific basis. An OFO may be issued by the Company as a blanket order to all Suppliers or to an individual Supplier whose actions are determined by the Company to jeopardize system integrity. The Company may issue an OFO to an individual Supplier if the Company faces Gas cost exposure in excess of daily cashout or imbalance penalties as set forth in Sections 9.6, 9.7, 10.6, and 10.7 for any under-deliveries or over-deliveries caused by that Supplier.
- 16.4 The Company will provide the Supplier with as much notice as is reasonably practicable of the issuance and removal of a Critical Day or an OFO; under most circumstances, the Company intends to provide at least twenty-two (22) hours' notice prior to the start of the Gas Day for the issuance of the Critical Day or OFO. Notification of the issuance and removal of a Critical Day or an OFO will be made by means as established in the Supplier Service Agreement. The Supplier will be responsible for coordinating with its Customers any change to the Customer's quantity of Gas Usage. An OFO or Critical Day will remain in effect until its removal by the Company.
- All quantities of Gas over-delivered or under-delivered to the Company's system in violation of an OFO will be subject to the Critical Day provisions of Sections 9.6 and 10.6 of this tariff.

17 FORCE MAJEURE AND LIMITATION OF LIABILITY

17.1 Neither the Company nor the Supplier will be liable to the other for any act, omission, or circumstance occasioned by or in consequence of any event constituting force majeure, and unless it is otherwise expressly provided herein, the obligations of the Company and the Supplier then existing hereunder will be excused during the period thereof to the extent affected by such event of force majeure, provided that reasonable diligence is exercised to overcome such event. As used herein, force majeure will mean the inability of the Company or the Supplier to fulfill its contractual or regulatory obligations: as a result of compliance by either party with an order, regulation, law, code, or operating standard imposed by a governmental authority; by reason of any act of God or public enemy; by reason of storm, flood, fire, earthquake, explosion, civil disturbance, labor dispute, or breakage or accident to machinery or pipeline (which breakage or accident is not the result of the negligence or misconduct of the party claiming force majeure); by reason of any declaration of force majeure by upstream Transporting Pipelines; or by reason of any other cause, whether the kind enumerated herein or otherwise, not within the control of the party claiming force majeure and which by the exercise of reasonable diligence such party is unable to prevent or overcome. Notwithstanding the foregoing, the Customer's and the Supplier's obligation to make

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 185 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 131 Delivery Terms and Conditions

any payments required under this tariff will in no case be excused by an event of force majeure. Nor will a failure to settle or prevent any labor dispute or other controversy with employees or with anyone purporting or seeking to represent employees be considered to be a matter within the control of the party claiming excuse. The party claiming force majeure will, on request, provide the other party with a written explanation thereof and of the remedy being undertaken.

- 17.2 The Company shall be liable only for direct damages resulting from the Company's conduct of business when the Company, its employees, or agents have acted in a negligent or intentionally wrongful manner. In no event shall the Company be liable to any party for any indirect, consequential, or special damages, whether arising in tort, contract, or otherwise, by reason of any services performed, or undertaken to be performed, or actions taken by the Company, or its agents or employees, under this tariff or in accordance with or required by law, including, without limitation, termination of the Customer's service.
- 17.3 If the Company is unable to render firm Delivery Service to the Customer taking such service as contemplated by this tariff as a result of force majeure and such inability continues for a period of thirty (30) Gas Days, the Customer may provide written notice to the Company of its desire to terminate Delivery Service at the expiration of thirty (30) Gas Days from the Company's receipt of such notice, but no sooner than sixty (60) Gas Days following the outset of the force majeure. If the Company has not restored Delivery Service to the Customer at the end of such notice period, the Customer's Delivery Service will terminate and both parties will be released from further performance hereunder, except for obligations to pay sums due and owing as of the date of termination.
- 17.4 The Company and the Supplier shall indemnify and hold the other and their respective affiliates, and the directors, officers, employees, and agents of each of them (collectively, "affiliates") harmless from and against any and all damages, costs (including attorney's fees), fines, penalties, and liabilities, in tort, contract, or otherwise (collectively, "liabilities"), resulting from claims of third parties arising, or claimed to have arisen, from the acts or omissions of either party in connection with the performance of the indemnifying party's obligations under this tariff. The Company and the Supplier shall waive recourse against the other party and its affiliates for or arising from the non-negligent performance by such other party in connection with the performance of its obligations under this tariff.

18 CURTAILMENT

- 18.1 Whenever the integrity of the Company's system or the Supply of the Company's Customers taking Sales Service or Delivery Service is believed to be threatened by conditions on its system or upon the systems with which it is directly or indirectly interconnected, the Company may, in its sole reasonable judgment, curtail or interrupt Gas service or reduce pressure as set out in Section 18, Supply and Capacity Shortage Allocation Policy of this tariff. Such action shall not be construed to constitute a default nor shall the Company be liable therefor in any respect. The Company will use efforts reasonable under the circumstances to overcome the cause of such curtailment, interruption, or reduction and to resume full performance.
- 18.2 The Company shall communicate notice of curtailment as soon as practicable to the Suppliers of affected Customers by means as specified in the Supplier Service Agreement.
- 18.3 The Company shall take reasonable care in providing regular and uninterrupted service to its firm Customers, but whenever the Company deems that the situation warrants any interruption or limitation in the service to be rendered, such interruption or limitation shall not constitute a breach of the contract and shall not render the Company liable for any damages suffered thereby by any person, or excuse the Customer from further fulfillment of the contract.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 186 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 132 Delivery Terms and Conditions

18.4 In any case where the Company determines in its judgment that a curtailment or interruption of firm services is necessary, the Company will curtail and/or interrupt firm Delivery Service and Sales Service Customers on a nondiscriminatory basis.

19 TAXES

- 19.1 In the event a tax of any kind is imposed or removed by any governmental authority on the distribution of Gas or on the gross revenues derived from the distribution of Gas at retail (exclusive, however, of taxes based on the Company's net income), the rate for service herein stated will be adjusted to reflect said tax. Similarly, the effective rate for service hereunder will be adjusted to reflect any refund of imposition of any surcharges or penalties applicable to service hereunder which are imposed or authorized by any governmental or regulatory authorities.
- 19.2 The Customer will be responsible for all taxes or assessments that may now or hereafter be levied with respect to the Gas or the handling or subsequent disposition thereof after its delivery to the Delivery Point. However, if the Company is required by law to collect and/or remit such taxes, the Customer will reimburse the Company for all amounts so paid. If the Customer claims exemption from any such taxes, the Customer will provide the Company in writing its tax exemption number and other appropriate documentation. If the Company collected any taxes or assessments from the Customer and is later informed by the Customer that the Customer is exempt from such taxes, it shall be the Customer's responsibility to obtain any refund from the appropriate governmental taxing agency.
- 19.3 The Supplier will be responsible for all production, severance, ad valorem, or similar taxes levied on the production or transportation of the Gas before its delivery to the Designated Receipt Point. The Supplier will also be responsible for sales taxes imposed on Gas delivered for the Customer's account. However, if the Company is required by law to remit such taxes to the collecting authority, it will do so and invoice the Supplier for such taxes paid on the Supplier's behalf.

20 SUPPLIER TERMS AND CONDITIONS

20.1 Applicability

The following terms and conditions shall apply to every Supplier providing Supplier Service in the State of New Hampshire, to every Customer doing business with said Suppliers, and to Customers acting as their own Supplier.

20.2 Obligations of Parties

20.2.1 Customer

Unless otherwise agreed to by the Company and the Customer, a Customer shall select one Supplier for each account at any given time. A Customer electing Supplier Service must provide the selected Supplier with its applicable Authorization Number. A Customer may choose only a Supplier who meets the terms described in Sections 20.2.3 and 20.3 below and who meets any applicable registration requirements established by law or regulation.

20.2.2 Company

The Company shall deliver Customer purchased Gas from the Designated Receipt Point to the Delivery Point in accordance with the service selected by the Customer pursuant to this tariff and, among other things, shall:

(a) Provide Customer service and support, including call center functions, for services provided by the Company under this tariff;

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 187 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 133 Delivery Terms and Conditions

- (b) Respond to service interruptions, reported Gas leaks, and to other Customer safety calls;
- (c) Handle connections, curtailments, and terminations for services provided by the Company under this tariff;
- (d) Read meters;
- (e) Submit bills to Customers for Delivery Service and if contracted by the Supplier, for Supplier Service in accordance with Section 12.2.1.
- (f) Address billing inquiries for Delivery Service;
- (g) Answer general questions about Delivery Service;
- (h) Provide to Suppliers, on request, the data format and procedures for electronic information transfers and funds transfers;
- (i) Arrange for or provide Sales Service to the Customer at the request of the Customer in accordance with the Company's tariff; and
- (j) Provide information regarding, at a minimum, rate tariffs, billing cycles, Capacity assignment methods, and Consumption Algorithms.

20.2.3 Supplier

The Supplier shall act on behalf of the Customer to acquire Supplies and to deliver them to the Designated Receipt Point pursuant to the service selected by the Customer and the requirements of this tariff.

The Supplier is responsible for enrolling Customers pursuant to Section 20.5 of this tariff.

The Supplier must request, complete and sign a Supplier Service Agreement to act as a Supplier on the Company's system, satisfy the Supplier requirements and practices as set forth in Section 20.3 of this tariff, be and remain an approved shipper on the upstream pipelines and underground storage facilities on which the Company will assign Capacity, if any, under Section 11, and be and remain eligible to provide service to Customers in New Hampshire.

The Supplier is responsible for completing all transactions with the Company and for all applicable charges associated with Customer enrollment and changes in the Customer's service as set forth in Section 20.5 and Attachment B.

20.3 Supplier Requirements and Practices

20.3.1 The Company shall have the right to establish reasonable financial and non-discriminatory credit standards for qualifying Suppliers. Accordingly, in order to serve Customers on the Company's system, the Supplier shall provide the Company, on a confidential basis, with audited balance sheet and other financial statements, such as annual reports to shareholders and 10-K reports, for the previous three (3) years, as well as two (2) trade and two (2) banking references. To the extent that such annual reports to shareholders are not publicly available, the Supplier shall provide the Company with a comparable list of all corporate affiliates, parent companies, and subsidiaries. The Supplier shall also provide its most recent reports from credit reporting and bond rating agencies. The Supplier shall be subject to a credit investigation by the Company. The Company shall review the Supplier's financial position periodically.

20.3.2 The Supplier shall also confirm in the Supplier Service Agreement that:

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

EFFECTIVE: August 1, 2021 Neil Proudman
TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 188 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 134 Delivery Terms and Conditions

- (a) The Supplier is not operating under any chapter of bankruptcy laws and is not subject to liquidation or debt reduction procedures under state laws, such as an assignment for the benefit of creditors, or any information creditors' committee agreement.
- (b) The Supplier is not aware of any change in business conditions which would cause a substantial deterioration in its financial conditions, a condition of insolvency, or the inability to exist as an ongoing business entity.
- (c) The Supplier has no delinquent balances outstanding for services previously provided by the Company, and the Supplier has paid its account according to the established terms and not made deductions or withheld payment for claims not authorized by contract.
- (d) No significant collection lawsuits or judgments are outstanding which would materially affect the Supplier's ability to remain solvent as a business entity.
- (e) The Supplier's New Hampshire business advertising and marketing materials conform to all applicable state and federal laws and regulations.
- 20.3.3 In the event the Supplier has not demonstrated to the Company's satisfaction that it has met the Company's credit evaluation standards, the Company shall require the Supplier to provide one of the following at the Maximum Financial Liability as calculated below:
 - (a) Advance deposit;
 - (b) Letter of credit;
 - (c) Surety bond; or
 - (d) Financial guaranty from a parent company that meets the creditworthiness criteria.

The Company shall base the Supplier's maximum financial liability as two (2) times the highest month's aggregated Gas Usage of all Customers currently served by the Supplier at the highest Monthly Index in the preceding twenty-four (24) Months. This amount may be updated continuously, and at minimum, whenever the aggregated Gas Usage of all Customers served by the Supplier changes by more than 25%. The Supplier agrees that the Company has the right to access and apply the deposit, letter of credit, or bond to any payment of any outstanding claims that the Company may have against the Supplier, including imbalance charges, cash-out charges, pipeline penalty charges, and other amounts owed to the Company, or to secure additional Gas supplies, including payment of the costs of the Gas supplies themselves, the cost of transportation storage, and other related costs incurred in bringing those Gas supplies into the Company's system. The Supplier shall continue its obligation to maintain its financial security instrument until it has satisfied all of its outstanding claims with the Company. The Supplier's financial security as established above must be in place no later than five (5) Business Days prior to the first day of each calendar month in order for the Supplier to maintain its eligibility to provide service to Customers.

- 20.3.4 The Supplier shall warrant that it has or will have entered into the necessary arrangements for the purchase of Supplies which it desires the Company to transport to its Customers, and that it has or will have entered into the necessary upstream transportation arrangements for the delivery of these Gas supplies to the Designated Receipt Point.
- 20.3.5 The Supplier shall warrant to the Company that it has good title to or lawful possession of all Gas delivered to the Company at the Designated Receipt Point on behalf of the Supplier

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 189 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 135 Delivery Terms and Conditions

or the Supplier's Customers. The Supplier shall indemnify the Company and hold it harmless from all suits, actions, debts, accounts, damages, costs, losses, taxes, and expenses arising from or out of any adverse legal claims of third parties to or against said Gas.

- 20.3.6 The Supplier shall be responsible for making all necessary arrangements and securing all required regulatory or governmental approvals, certificates, or permits to enable Gas to be delivered to the Company's system.
- 20.3.7 By agreeing to provide service under this tariff, the Supplier acknowledges that adherence to any applicable law regarding unfair trade practices, truth in advertising law, or law of similar import is required. Any Supplier found by a court of competent jurisdiction to have willfully or repeatedly violated the New Hampshire Consumer Protection Act, N.H.R.S.A. Ch. 358-A; the Federal Trade Commission Telemarketing Sales Rules, 16 C.F.R. Part 310; or the regulations promulgated pursuant to the Federal Trade Commission Act, 15 U.S.C. § 45 (a) (1), may be suspended or disqualified from acting as a Supplier on the Company's system.
- 20.3.8 If the Supplier fails to comply with or perform any of the obligations on its part established in this tariff or in the Supplier Service Agreement (e.g., failure to deliver Gas or late payment of bills rendered or failure to execute a capacity assignment), the Company maintains the right to terminate the Supplier's eligibility to act as a Supplier on the Company's system. Written notice of such an intent to terminate the Supplier's eligibility shall be given to the Supplier, its Customers, and the NHPUC. Notification to the Supplier shall be via Registered U.S. Mail - Return Receipt Requested or other means of documented delivery. Upon issuance of such written notice, the Company shall have the right to terminate the Supplier's eligibility to act as a Supplier on the Company's system at the expiration of ten (10) Gas Days after the giving of such notice, unless within such ten (10) Gas Day period the Supplier shall remedy to the full satisfaction of the Company such failure. Termination of such Supplier eligibility for any such cause shall be a cumulative remedy as to the Company, and shall not release the Supplier from its obligation to make payment of any amount or amounts due or to become due from the Supplier to the Company under the Company's applicable tariffs. Customers whose Supplier's deliveries have been terminated will be placed on Sales Service pursuant Section 13 of this tariff.
- 20.4 Access to Usage History and Current Billing Information

The Supplier shall be responsible for obtaining the necessary Authorization Number from each Customer prior to requesting the Company to release the Company's historic usage information specific to that Customer to such Supplier.

The Company shall be required to provide the most recent twelve (12) months of a Customer's historic usage data to a Supplier, provided that the Supplier has received the appropriate authorization as set forth above.

- 20.5 Enrollment, Cancellation, and Termination of Supplier Service
 - 20.5.1 The Supplier shall be responsible for obtaining the necessary Authorization Number from each Customer prior to initiating Supplier Service to the Customer.
 - 20.5.2 The Supplier must provide the Company with the following minimum information in the Company's predetermined format prior to the commencement or termination of service by the Supplier pursuant to Section 20.5 of this tariff:
 - (a) The Customer's name and current Authorization Number;

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 190 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 136 Delivery Terms and Conditions

- (b) The name of the Supplier;
- (c) The Customer's billing option (for commencement of service);
- (d) The type of change in Supplier Service (e.g., commencement of service, termination of service, or cancellation of service due to the rescission of an agreement with the Supplier by the Customer); and
- (e) Any additional information reasonably required by the Company.
 - The Company shall determine whether each Customer's enrollment request as provided by a Supplier is complete and accurate, and matches the Customer's account record. In the event that the enrollment request is incomplete, inaccurate, or does not match the Customer's account record, then the Company will notify the Supplier so that the Supplier can resolve any discrepancies.
- 20.5.3 A change in Supplier Service will normally be made on a monthly metering and billing cycle basis, with changes taking effect on the date of the Customer's next scheduled meter read. Enrollment forms must be transmitted no less than ten (10) Business Days prior to the Customer's next scheduled meter read. If more than one Supplier submits a Supplier Service transaction for a given Customer during the monthly billing cycle, the first completed transaction that is received during the cycle shall be accepted. All other transactions shall be rejected. Rejected transactions may be resubmitted after the Customer's next scheduled meter read.
- 20.5.4 If the Supplier submits information to the Company to terminate Supplier Service to a Customer less than ten (10) Gas Days before the next scheduled meter read, Supplier Service shall be terminated on the date of the Customer's subsequent scheduled meter read. The Company shall confirm the termination date for Supplier Service.
- 20.5.5 In those instances when a Customer who is receiving Supplier Service from an existing Supplier initiates such service with a new Supplier, the Company shall send the date for the Customer's change in Supplier Service to the existing Supplier. To terminate Supplier Service with a Supplier and to initiate Sales Service, a Customer shall so inform the Company and the Supplier. Supplier Service shall be terminated on the date of the Customer's next scheduled meter read provided that the Company receives notice of such termination no less than ten (10) days in advance of the next scheduled meter read. Where such notice is received by the Company in less than ten (10) days in advance of the next scheduled read, the termination shall be effective as of the date of the following scheduled read. The Company shall send the Customer's termination date for Supplier Service to the Supplier.
- 20.5.6 A Customer who moves within the Company's service territory shall have the opportunity to notify its existing Supplier that it seeks to continue Supplier Service with said Supplier. Upon such notification, the Supplier may enroll the Customer pursuant to the provisions set forth in this Section in order to initiate Supplier Service for the Customer at the new location. The Company shall make the necessary adjustments to the Supplier's affected Aggregation Pools, including but not limited to, changes to Designated Receipt Points, and quantities of Capacity for assignment, if any, pursuant to this tariff and the Supplier's Service Agreement with the Company. In the event that the existing Supplier does not enroll the Customer for Supplier Service at the new location, the Company shall arrange for or provide Sales Service to the Customer.
- 20.5.7 In those instances when a new Customer moves to the Company's service territory, the Customer's Supplier must enroll the Customer pursuant to the provisions set forth in this

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 191 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 137 Delivery Terms and Conditions

Section in order to initiate Supplier Service for the Customer. Otherwise, the Customer shall receive Sales Service in accordance with Section 13.

20.5.8 The Company may charge fees to the Supplier for processing the transactions described in this Section, as approved by the NHPUC. These fees are included in Attachment D.

20.6 Aggregation Pools

- 20.6.1 The aggregation of Customer accounts into an Aggregation Pool is limited by the Delivery Service of the respective Customers. Non-daily metered Customers subscribing to Delivery Service under Rate Schedules G-41, G-42, G-51 and G-52 must be aggregated in a separate pool from Customers subscribing to daily metered service under Rate Schedules G-43, G-53, and G-54.
- 20.6.2 Non-daily metered Customers taking Delivery Service pursuant to Section 10 of this tariff shall be combined by a Supplier into a single Aggregation Pool within each of the Company's designated Gas Service Areas.
- 20.6.3 Daily metered Customers taking Delivery Service pursuant to Section 9 of this tariff shall be combined by a Supplier into a single Aggregation Pool within each of the Company's designated Gas Service Areas.
- 20.6.4 A separate Supplier account will be established for each Supplier Aggregation Pool.
- 20.6.5 The election of any service from the Company by the Supplier shall apply to the entire Aggregation Pool and not just an individual customer in the Aggregation Pool.
- 20.6.6 The Company may charge a monthly fee to the Supplier for each Aggregation Pool pursuant to Attachment B.

20.7 Imbalance Trading

- 20.7.1 Prior to the imposition of imbalance charges, the Supplier may engage in trading daily and monthly imbalances for the previous Month, provided that daily imbalance trades are communicated to the Company within three (3) Business Days upon the Company's provision of information on Supplier imbalances for said Month.
- 20.7.2 The Company will make available a list of Suppliers by Gas Service Area making deliveries during the previous Month.
- 20.7.3 Aggregation Pools affected by the transaction must be located within the same Gas Service Area as defined in Section 4, unless waived by the Company.
- 20.7.4 Daily imbalance trades must be point-specific on those Gas Days when the Transporting Pipeline required the Company to balance on a point-specific basis.

20.8 Billing and Payment

- 20.8.1 By the tenth (10th) Business Day of the calendar month, the Company shall render to the Supplier a statement of the quantities delivered and amounts owed by the Supplier for the prior Month. The Company will provide Suppliers with their Customers' consumption data based on estimated or actual meter readings at the appropriate cycle read dates for each Customer in the Aggregation Pool pursuant to Section 12 of this tariff. This data will be provided on a rolling basis as readings or estimates are made.
- 20.8.2 Calculation of the charges applicable to the Aggregation Pool will be based on aggregated Gas Usage and other such indicators of all Customers in the Aggregation Pool. Billing for

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 192 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 138 Delivery Terms and Conditions

charges applicable to an Aggregation Pool, including but not limited to imbalance charges, credits or penalties, shall be billed to the Supplier on a calendar month basis.

- 20.8.3 The Supplier shall have ten (10) Business Days from the date of such statement to render payment to the Company. The Supplier shall render payment by means of electronic funds transfer to the Company. The late payment rate will apply to all amounts outstanding after ten (10) days.
- 20.8.4 If the correctness of the Company's bill to the Supplier is questioned or disputed by the Supplier, an explanation should be promptly requested from the Company. If the bill is determined to be incorrect, the Company shall issue a corrected bill. In the event that the Supplier and the Company fail to agree on the amount of the bill, the Supplier may file a complaint with the Commission to resolve such complaint.

21 CUSTOMER DESIGNATED REPRESENTATIVE

- 21.1 The Customer may appoint a Designated Representative to satisfy or undertake the Customer's duties and obligations; including, but not limited to submitting and/or receiving notices, making nominations, arranging for trades of imbalances, and performing operational and administrative tasks; provided, however, that under no circumstances will the appointment of a Designated Representative relieve the Customer of the responsibility to make full and timely payment to the Company for all Delivery Service provided under this tariff.
- 21.2 A request by a Designated Representative to the Company that contains the Customer's Authorization Number will be deemed to be confirmation that the Customer has designated such person or entity as a Designated Representative. A Customer may appoint only one (1) Designated Representative per account.
- 21.3 Under any agency established hereunder, the Company shall rely upon information concerning the applicable Customer's Delivery Service that is provided by the Designated Representative. All such information shall be deemed to have been provided by the Customer. Similarly, any notice or other information provided by the Company to the Designated Representative concerning the provision of Delivery Service to such Customer shall be deemed to have been provided to the Customer. The Customer shall rely upon any information concerning Delivery Service that is provided to the Designated Representative as if that information had been provided directly to the Customer.
- 21.4 The Customer shall agree to indemnify the Company and hold it harmless from any liability (including reasonable legal fees and expenses) that the Company incurs as a result of the Designated Representative's negligence or willful misconduct in its performance of agency functions on the Customer's behalf.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Docket No. DG 21-130 Exhibit 26 Page 193 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 139 Attachments

IV. ATTACHMENTS

1 ATTACHMENT A Supplier Service Agreement

GAS SUPPLIER SERVICE AGREEMENT

This Agreement made this [day] day of [month), 20[xx], between Liberty Utilities (EnergyNorth Natural Gas) Corp. d/b/a Liberty Utilities, a New Hampshire Corporation with a principal place of business at 15 Buttrick Road, Londonderry, NH 03053 (the "Company") and [name of supplier], a [state] company with a principal place of business at [address] ("Supplier"). The Company and the Supplier is also individually referred to herein as a "Party" or collectively as the "Parties."

BASIC UNDERSTANDINGS

Whereas, the Company operates as a natural gas local distribution company and provides firm transportation of third- party gas on its distribution system; and

Whereas, the Company's Tariff (the "Tariff") on file with, and approved by, the New Hampshire Public Utilities Commission (the "NHPUC") permits delivery service customers to assign their rights of nominating and scheduling delivery of gas for transportation on the Company's system to a third-party natural gas supplier; and

Whereas, Supplier seeks to nominate and schedule delivery of gas for distribution on the Company's system on behalf of one or more customers taking delivery service from the Company; and

Whereas, the Company's Tariff, Part III, Section 20.2.3, requires Supplier to enter into this Supplier Service Agreement (the "Agreement") with the Company prior to the initiation of Supplier Service, as defined therein; and

Now therefore, the Parties hereto, each in consideration of the agreement of the other, do hereby agree as follows:

I. SCOPE AND APPLICATION

1.0 This Agreement shall be subject to the Company's Tariff as on file with the NHPUC and in effect from time to time. The Company's Tariff and applicable Rate Schedules are hereby incorporated by reference as though directly set forth herein. In the event the terms of this Agreement conflict with the Company's Tariff, the Tariff shall control.

1 of 14

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 194 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 140 Attachments

- 1.1 This Agreement is intended for use between the Company and natural gas suppliers providing service to customers on the Company's distribution system, and may not be waived, altered, amended, or modified, except as provided herein.
- 1.2 Exhibits A and B, attached hereto and incorporated herein by reference, include additional terms that are a part of this Agreement.

II. DEFINITIONS

2.0 Any capitalized terms used in this Agreement and not defined herein shall be as defined in the Tariff or as stated in the NHPUC's regulations.

III. TERM

- 3.0 This Agreement shall become effective on the date hereof (the "Effective Date") and shall continue in full force and effect from month to month unless terminated by either Party by written notice given no less than thirty (30) days prior to the desired termination date, or unless otherwise agreed by the Parties. Notwithstanding the foregoing, the Parties agree to abide by all terms of this Agreement until any transactions that are outstanding at the time of termination are completed, including, but not limited to, the payment by Supplier to the Company of any and all outstanding balances.
- 3.1 Notwithstanding anything to the contrary elsewhere in this Agreement or in the Company's Tariff, any Party, by written notice to the other Party (the "Breaching Party") may terminate this Agreement, in whole or in part, with respect to such Breaching Party or suspend further performance without terminating this Agreement upon the occurrence of any of the following:(a) the Breaching Party terminates or suspends doing business; (b) the Breaching Party becomes subject to any bankruptcy or insolvency proceeding under federal or state law (unless removed or dismissed within sixty (60) days from the filing thereof), or becomes insolvent, becomes subject to direct control of a transferee, receiver or similar authority, or makes an assignment for the benefit of creditors; or (c) the Breaching Party commits a material breach of any of its obligations under this Agreement or the Tariff and has not cured such breach within fifteen (15) days after receipt of a written notice from the other Party specifying the nature of such.

2 of 14

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 195 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 141
Attachments

- 3.2 Consistent with the provisions of Part III, Section 20.3.8 of the Company's Tariff, the Company also maintains the right to terminate the Supplier's eligibility to act as a Supplier. on the Company's system in the event that Supplier fails to comply with or perform any of the obligations on its part established in the Tariff or in this Agreement, including but not limited to, failure to deliver gas or to make payment of amounts due to the Company.
- 3.3 Notwithstanding the Effective Date, Supplier acknowledges and agrees that the Company is obligated to provide services pursuant to this Agreement only upon full satisfaction, or the Company's express written waiver, of the Conditions Precedent set forth in Article IV of this Agreement.
- 3.4 No delay by either Party in enforcing any of its rights hereunder shall be deemed a waiver of such rights, nor shall a waiver of one default be deemed a waiver of any other or subsequent default.
- 3.5 The enumeration of the foregoing remedies shall not be deemed a waiver of any other remedies to which either Party is legally entitled.

IV. CONDITIONS PRECEDENT

- 4.0 The following requirements shall be conditions precedent to the Company's obligations hereunder:
 - Supplier shall provide the Company with all information requested in Exhibits
 A and B attached hereto and incorporated herein;
 - (b) Pursuant to Part III, Section 20.3.1 of the Company's Tariff, the Company shall confirm the Supplier's creditworthiness. In the event that Supplier has not demonstrated to the Company's satisfaction that it has met the Company's credit evaluation standards, the Company will identify such deficiencies to the Supplier, and the Supplier shall provide financial assurances as required by the Company consistent with the provisions of Part III, Section 20.3.3;
 - (c) Pursuant to Part Ill, Section 20.2.3 of the Company's Tariff, Supplier shall register with the NHPUC and provide evidence of such to the Company on an annual basis;

3 of 14

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 196 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 142
Attachments

(d) Pursuant to Part III, Section 20.2.3 of the Company's Tariff, Supplier shall demonstrate to the Company that it is an approved shipper on the upstream pipelines and underground storage facilities on which the Company will assign capacity;

- (e) Pursuant to Part Ill, Section 12.2.1 of the Company's Tariff, where Supplier elects to utilize the Standard Complete Billing Services from the Company, Supplier shall furnish to the Company a complete schedule of its relevant rates and rate pricing options for Supplier Service in written form or in an electronic format reasonably acceptable to the Company, at Company's option, no less than ten (10) Business Days prior to initial Customer enrollment for any such rate or prior to a change in Supplier's existing rates or five (5) Business Days prior to a change in rate pricing options.
- (f) Prior to Customer Enrollment, Supplier shall successfully complete testing of the business-transaction communication protocols established by the Company, which may include communication by fax or telephone, electronic transactions as specified by the Company, or any other applicable communication requirements set forth by the Company.

V. SUPPLIER CERTIFICATION

- 5.0 In addition to the requirements listed in Section IV of this Agreement, and pursuant to Part III, Section 20.3.2 of the Company's Tariff, the Supplier hereby affirms the following:
 - (a) Supplier is not operating under any chapter of bankruptcy laws and is not subject to liquidation or debt reduction procedures under state laws, such as an assignment for the benefit of creditors, or any information creditors' committee agreement.
 - (b) Supplier is not aware of any change in business conditions that would cause a substantial deterioration in its financial conditions, a condition of insolvency, or the inability to exist as an ongoing business entity.
 - (c) Supplier has no delinquent balances outstanding for services previously provided by the Company, and Supplier has paid its account according to the established

4 of 14

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

EFFECTIVE: August 1, 2021 TITLE: President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Docket No. DG 21-130 Exhibit 26 Page 197 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 143
Attachments

terms and not made deductions or withheld payment for claims not authorized by contract.

- (d) No significant collection lawsuits or judgments are outstanding that would materially affect Supplier's ability to remain solvent as a business entity.
- (e) Supplier's New Hampshire business advertising and marketing materials conform to all applicable New Hampshire state and federal laws and regulations.
- 5.1 Supplier shall promptly notify Company of any material change in its financial condition as it relates to Supplier's creditworthiness or solvency as a business enterprise.
- 5.2 In the event that the NHPUC enacts regulations whereby Supplier must register with the NHPUC, Supplier shall notify Company within twenty-four (24) hours in writing in the event that its registration as a Competitive Supplier is acted upon by the NHPUC in such a way that it materially affects Supplier's performance under this Agreement, including but not limited to suspension, revocation, modification, or non-renewal. Consistent with Part III, Section 20.3.8 of the Company's Tariff, revocation or non-renewal of Supplier's registration shall be grounds for immediate termination of this Agreement by Company.

VI. NOMINATIONS AND SCHEDULING

- 6.0 The Company and Supplier, pursuant to the Company's Tariff on file with the NHPUC and the terms of this Agreement, agree to exchange and act on information regarding the nomination and scheduling of gas for transportation on behalf of Supplier's customers.
- 6.1 Supplier acknowledges and agrees that its transportation rights under this Agreement are solely those that have been assigned to it by the Customer pursuant to the Company's Tariff. Supplier further agrees that the Company shall have no obligation to honor any nomination or scheduling request from Supplier that, in the Company's sole judgment, exceeds the scope of Supplier's assigned rights or where such nominations or requests could be reasonably refused, directly or indirectly, based on the terms of this Agreement or the Company's Tariff.
- 6.2 Pursuant to Part III, Sections 9.3.2 and 10.3.3 of the Company's Tariff, nominations will be communicated to the Company in accordance with the terms of this Agreement as set forth in Exhibit A.

5 of 14

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 198 of 470

Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 144 Attachments

6.3 In the event of a discrepancy between the volume nominated to the Company by Supplier and the volume confirmed by the Company, the discrepancy shall be allocated between and among Supplier's Aggregation Pools and/or Customers in accordance with the Pre-Determined Allocation Method set forth in Exhibit B, attached hereto. In the event that the Supplier has not provided the Company with a Pre-Determined Allocation Method, the discrepancy will be allocated consistent with the provisions of the Company's Tariff.

VII. CAPACITY ASSIGNMENTS

- 70 The Supplier's Maximum Daily Peaking Quantity ("MDPQ") may be modified during the calendar year in accordance with the provisions of Part Ill, Sections 11.0 and 14.0 of the Company's Tariff. Company will notify Supplier prior to the effective date of such changes.
- 7.1 Pursuant to Part III, Section 11.9.2 of the Company's Tariff, the quantity of each Company Managed Supply assigned to Supplier may be modified during the calendar year in accordance with Part III, Sections 11.4 and 11.8 of the Company's Tariff. Company will notify Supplier prior to the effective date of such changes.
- In accordance with Part Ill, Sections 11.0 and 14.0 of the Company's Tariff, the quantity of Capacity assigned to Supplier may be modified during the calendar year. In addition, the Company shall have the right to adjust a Customer's total capacity quantity C'TCQ") if the Company determines that the TCQ calculation is in error or is otherwise not calculated in accordance with the provisions of Part Ill, Sections 11.3.2.
- Pursuant to Part Ill, Section 11.10.2 of the Company's Tariff, Supplier shall provide notice to the Company of its designation of contracts to be managed by the Company for cost mitigation purposes by the means set forth in Exhibit 8.0.

VIII. LEFT BLANK INTENTIONALLY (RESERVED FOR FUTURE USE)

IX. BILLING AND PAYMENT

9.0 Bills, fees and charges for services provided by the Company, including, but not limited to, monthly cashouts, monthly imbalance charges, daily imbalance charges, and any other applicable charges set forth in the Tariff or in this Agreement, shall be

6 of 14

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

EFFECTIVE: August 1, 2021 TITLE: President

Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 199 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 145
Attachments

rendered to Supplier on a monthly basis and shall be due upon receipt of said bill, unless otherwise specified in Exhibit A.

In addition to any other right or remedy available to the Company, Supplier's failure to make payment within ten (10) days of the posting date on the bill shall result in the addition of interest on any unpaid balance calculated at the maximum monthly rate allowable by the Company's Tariff. Interest shall accrue commencing from the date said bill was posted. The posting date is the date the bill is transmitted to Supplier. The bill may also be transmitted electronically if agreed to between the Parties in Exhibit.

- 9.1 The Company shall have the right to deduct any amounts owed by Supplier to the Company for such services, which are thirty (30) days or more past due, from any amounts collected in the normal course of business by the Company on the Supplier's behalf. Amounts subject to a good faith dispute will not be subject to deduction.
- 9.2 The Parties agree to cooperate and provide each other with necessary documentation relating to any transactions resulting hereunder, including but not limited to, applicable sales or other tax exemptions. The Parties agree that Supplier's failure to comply with the provisions of this Article IX shall constitute default of payment under the Tariff and expose Supplier to liability thereunder as well as under this Agreement.
- 9.3 Consistent with the provisions of Part Ill, Sections 20.3.1 and 20.3.3 of the Company's Tariff, Supplier shall satisfy the creditworthiness standards established by the Company. In the event the Supplier has not demonstrated satisfaction of the Company's creditworthiness standards, the Supplier shall provide, upon ten (10) days written notice from the Company, financial assurance in the form of an advance deposit, letter of credit, surety bond or financial guaranty from a parent company, as reasonably determined by the Company. The amount of any such financial assurance required by the Company shall be calculated in accordance with the provisions of Part III, Section 20.3.3 of the Company's Tariff. The Company shall review Supplier's satisfaction of the Company's creditworthiness standards every twelve (12) months during the term of this Agreement giving consideration to Supplier's payment history

7 of 14

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 200 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 146 Attachments

in the preceding twelve-month period. Upon the request of Supplier, the Company shall exercise its sole reasonable discretion to determine whether a change in the form of financial assurance is warranted. In the event that the Company requires financial assurances in the form of a deposit, such deposits shall accrue interest in accordance with the Company's Tariff. Such deposit shall be returned to Supplier within thirty (30) days of the expiration or termination of this Agreement, provided that Supplier is not in default under this Agreement. The Company may deduct from the deposit any amount payable to the Company by Supplier under this Agreement, which has not been paid by the Supplier when due, unless such non-payment relates to a documented billing dispute between Supplier and the Company. Such deduction may be taken by the Company without notice or demand of any kind and the Company may, in its sole discretion, apply such deposit against any amount then due and payable. In the event that Company applies all or any portion of such deposit, Supplier shall deposit such sums as are necessary to replenish the security deposit to its maximum amount, within ten (10) days' notice of such deduction and application.

X. REPRESENTATIONS

- 10.0 Each Party represents that it is and shall remain in compliance with all applicable laws, tariffs, and NHPUC regulations during the term of this Agreement.
- 10.1 Each person executing this Agreement for the respective Parties represents and warrants that he or she has authority to bind that Party.
- 10.2 Each Party represents that (a) it has the full power and authority to execute, deliver, and perform this Agreement; (b) the execution, delivery, and performance of this Agreement have been duly authorized by all necessary corporate or other action by such Party; and (c) this Agreement constitutes that Party's legal, valid and binding obligation, enforceable against such Party in accordance with its terms.
- 10.3 Each Party shall exercise all reasonable care, diligence and good faith in the performance of its duties pursuant to this Agreement, and carry out its duties in accordance with applicable recognized professional standards.

8 of 14

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 201 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 147
Attachments

XI. NONDISCLOSURE

- 11.0 Neither Party may disclose any Confidential Information obtained pursuant to this Agreement to any third Party, including affiliates of such Party, without the express prior written consent of the other Party. As used herein, the term "Confidential Information" shall include, but not be limited to, all business, financial, and commercial information pertaining to the Parties, Customers of either or both Parties, Suppliers for either Party, personnel of either Party; any trade secrets; and other information of a similar nature; whether written or in intangible form that is marked proprietary or confidential with the appropriate owner's name.
- 11.1 Confidential Information shall not include information known to either Party prior to obtaining the same from the other Party, information in the public domain, or information obtained by a Party from a third party who did not, directly or indirectly, receive the same from the other Party to this Agreement or from a Party who was under an obligation of confidentiality to the other Party to this Agreement, or information developed by either Party independent of any Confidential Information. The receiving Party shall use the higher of the standard of care that the receiving Party uses to preserve its own Confidential Information or a reasonable standard of care to prevent unauthorized use or disclosure of such Confidential Information. Each receiving Party shall, upon termination of this Agreement or at any time upon the request of the disclosing Party, promptly return or destroy all Confidential Information of the disclosing Party then in its possession.
- 11.2 Notwithstanding the preceding, Confidential Information may be disclosed to any governmental, judicial or regulatory authority requiring such Confidential Information pursuant to any applicable law, regulation, ruling, or order, provided that: (a) such Confidential Information is submitted under any applicable provision, if any, for confidential treatment by such governmental, judicial or regulatory authority; and (b) prior to such disclosure, the other Party is given prompt notice of the disclosure requirement so that it may take whatever action it deems appropriate, including intervention in any proceeding and the seeking of any injunction to prohibit such disclosure.

9 of 14

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 202 of 470 Docket No. DG 20-105

Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 148 Attachments

11.3 No provision of this Agreement shall prohibit the Company from communicating to its Customers and prospective customers, information regarding Supplier's eligibility to conduct business on the Company's distribution system. In addition, obligations under this Article XI shall survive the termination or expiration of this Agreement.

LIABILITY AND INDEMNIFICATION

- 12.0 The Parties acknowledge and agree that the Force Majeure provisions set forth in Part III, Section 17 of the Company's Tariff are incorporated by reference as if set forth
- 12.1 The Parties acknowledge and agree that the liability and indemnification provisions in Part III, Section 17 of the Company's Tariff are incorporated by reference as if set forth herein.
- 12.2 For purposes of such liability and indemnification, however, the Parties acknowledge and agree that nothing in such Tariff prohibits one Party from impleading the other Party as a third-party defendant, whether or not one or both Parties are named as defendants in the initial claim of a third party. The third-party claim shall be stayed pending resolution of any dispute regarding liability and indemnification under this Agreement. Such resolution shall be final and binding upon the Parties only after agreement between the Parties or after entry of a final judgment, after any further appeals of a court of competent jurisdiction to which any appeal may have been taken from the determination of the arbitrator(s).
- 12.3 The Parties acknowledge and agree that for purposes of Part III, Section 17 the Company's Tariff, a Party seeking recovery from the other Party in connection with the performance of its obligations of the Tariff shall not be entitled to recovery where its own negligent acts or omissions contribute to or cause such damages, costs, fines, penalties or liabilities.
- 12.4 The Parties expressly acknowledge and agree that the dispute resolution provision in Article XIII of this Agreement shall apply to any and all disputes arising under this Article, including, without limitation, those disputes that arise as a result of either of

10 of 14

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 203 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 149 Attachments

the Parties being named as a defendant in the primary action or being named as a third-party defendant by a defendant in the primary action.

- 12.5 Notwithstanding anything in this Agreement or the Tariff to the contrary, in no event shall any Party hereto be liable to any other Party hereto for indirect, consequential, punitive, special, or exemplary damages under any theory of law that is now or may in the future be in effect, including without limitation: contract, tort, N.H.R.S.A. Ch. 358-A, strict liability, or negligence.
- 12.6 Notwithstanding the availability of other remedies at law or in equity, either Party hereto shall be entitled to specific performance to remedy a breach of this Agreement by the other Party.
- 12.7 Supplier further agrees that it shall indemnify, defend and hold harmless the Company with respect to any claim, suit, damages or costs of any kind arising from any action or inaction of the Company in reliance upon the nominations, scheduling instructions or other communications from Supplier. The Parties agree that reliance on such instructions and communications shall be deemed reasonable and shall not constitute negligence.
- 12.8 The provisions of this Article XII shall survive the termination of this Agreement.

XIII. DISPUTE RESOLUTION

- 13.0 Disputes hereunder shall be reduced to writing and referred to the Parties' representatives for resolution. The Parties' representatives shall meet and make all reasonable efforts to resolve the dispute. Pending resolution, the Parties shall continue to fulfill their obligations under this Agreement in good faith, unless this Agreement has been suspended or terminated. If the Parties fail to resolve the dispute within thirty (30) days, they may mutually agree to pursue mediation or arbitration to resolve such issues.
- 13.1 The interpretation and performance of this Agreement shall be in accordance with and controlled by the laws of the State of New Hampshire, without regard to the doctrines governing choice of law. All disputes arising hereunder shall be brought either before the NHPUC or the state courts of the State of New Hampshire.

11 of 14

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 204 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 150 Attachments

XIV. COMMUNICATIONS

- 14.0 Except as otherwise provided herein, any notices given under this Agreement shall be in writing and shall be delivered to the Company as set forth in Exhibit A, by hand or sent by (a) certified mail, return receipt requested, first class postage prepaid, (b) telecopy, or (c) a nationally recognized courier service. Notices and other communications to Supplier shall also be addressed as shown on Exhibit A. Notices given hereunder shall be deemed to have been given upon receipt or any refusal to accept; telecopied notices shall be deemed to have been given upon confirmation of their receipt.
- 14.1 All communications required by the Company's Tariff shall be made in accordance with the schedule listed in Exhibit A. Information on active Company fax numbers and email addresses shall be posted on the Company's Internet Website at http://www.libertyutilities.com/east/gas/business partners/index.html

XV. ENFORCEABILITY

- 15.0 In the event that any portion or part of this Agreement is deemed invalid, against public policy, void or otherwise unenforceable by a court of law, the validity and enforceability of the remaining portions thereof shall otherwise be fully enforceable.
- 15.1 No waiver by any Party of any one or more defaults by the other Party in the performance of any provision of this Agreement shall operate or be construed as a waiver of any other present or future default, whether of a like or different character. No delay by either Party in enforcing any of its rights hereunder shall be deemed a waiver of such rights.

XVI. ASSIGNMENT AND DELEGATION

- 16.0 Any entity that shall succeed by purchase, merger or consolidation to the assets and properties, substantially or as an entity, of either Party hereto shall be entitled to the rights and shall be subject to the obligations of its predecessor in interest under this Agreement.
- 16.1 Either Party may, without relieving itself of its obligations under this Agreement, assign any of its rights or obligations hereunder to an affiliated entity, but otherwise no assignment of this Agreement or any of the rights or obligations hereunder shall

12 of 14

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

EFFECTIVE: August 1, 2021 TITLE: President

Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 205 of 470 Docket No. DG 20-105

Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 151 Attachments

be made unless there first shall have been obtained the written consent of the other Party. No assignment by Supplier shall take effect until the assignee has met the requirements of Article IV hereunder. No assignment of this Agreement shall relieve the assigning Party of any of its obligations under this Agreement until such obligations have been assumed by the assignee.

- 16.2 The restrictions on assignment contained herein shall not in any way prevent either Party from pledging or mortgaging its rights as security for its indebtedness.
- 16.3 In addition, either Party may subcontract its duties under this Agreement to a subcontractor provided that the subcontracting Party shall remain fully responsible as a principal and not as a guarantor for performance of any subcontracted duties, and shall serve as the point of contact between its subcontractor and the other Party, and the subcontractor shall meet the requirements of any applicable laws, rules, regulations, and Tariff. The assigning or subcontracting Party shall provide the other Party with thirty (30) calendar days' prior written notice of any such subcontracting or assignment, which notice shall include such information about the subcontractor as the other Party shall reasonably require.

XVII. MISCELLANEOUS

- 17.0 This Agreement, all Exhibits and attachments hereto and all documents referenced herein, constitute the entire agreement between the Parties and supersedes all other agreements, communications, and representations. Paragraph headings are for convenience only and are not to be construed as part of this Agreement.
- 17.1 Unless otherwise provided herein, no modification of, or supplement to, the terms and provisions stated in this Agreement shall be or become effective without the written consent of both Parties.
- 17.2 This Agreement may be executed simultaneously in two or more counterparts, each of which shall be deemed to be an original but all of which shall constitute one and the same document.

13 of 14

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 206 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 152 Attachments

In witness whereof, the Parties have caused this Agreement to be executed by their duly authorized representatives as of the date above.

[SUPPL	IER NA	ME]					
Ву			Title				
Liberty Utilities	Utilities	(EnergyNorth	Natural	Gas)	Согр	d/b/a	Liberty
Bv			Title				

14 of 14

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman
Neil Proudman
TITLE: Provident

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Docket No. DG 21-130 Exhibit 26 Page 207 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 153
Attachments

2 ATTACHMENT B Schedule of Administrative Fees and Charges

I. Supplier Balancing Charge: \$ 0.1200 per MMBtu of Daily Imbalance Volumes")

II. Capacity Mitigation Fee 15% of the Proceeds from the Marketing of

III. Peaking Demand Charge \$ 17.32 MMBTU of Peak MDQ

IV. Company Allowance Calculation (per Schedule 25)

169,030,868 Total Sendout - Therms Jul -2019 - Jun-2020 166,311,578 Total Throughput - Therms Jul-2019 - Jun-2020

2,719,290 Variance (Sendout - Throughput)

Company Allowance Percentage 2020-21 1.6% Variance / Total Sendout

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 208 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 154
Attachments

3 ATTACHMENT C Capacity Allocators

Rate Class		Pipeline	Storage	Peaking	Total
	Low Annual/High				
G-41	Winter Use	46.1%	17.1%	36.8%	100.0%
	Low Annual/Low				
G-51	Winter Use	59.3%	12.9%	27.9%	100.0%
	Medium Annual / High				
G-42	Winter	46.1%	17.1%	36.8%	100.0%
	High Annual / Low				
G-52	Winter Use	59.3%	12.9%	27.9%	100.0%
	High Annual / High				
G-43	Winter	46.1%	17.1%	36.8%	100.0%
	High Annual / Load				
G-53	Factor < 90%	59.3%	12.9%	27.9%	100.0%
	High Annual / Load				
G-54	Factor < 90%	59.3%	12.9%	27.9%	100.0%

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman Neil Proudman

EFFECTIVE: August 1, 2021 Neil Proudma

TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 209 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-<u>11</u> - GAS

LIBERTY UTILITIES (ENERGYNORTH NATURAL GAS) CORP. D/B/A

LIBERTY UTILITIES

SUPERSEDING NHPUC No. 8-10 AND IN LIEU OF NHPUC No. 9

TARIFF

FOR

GAS SERVICE

Applicable

in

Thirty-five towns in New Hampshire served in whole or in part.

(For detailed description, see Service Area)

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX	XX, 2021, in Docket N	o. DG 20-105
DATED:	May 18, 2018	ISSUED BY:	/s/Susan L. Fleck
			Susan L. Fleck
EFFECTIVE:	May 01, 2018	TITLE:	President

LIBERTY UTILITIES seding Second Revised Page i CHECK SHEET ______1 2 GENERAL TERMS AND CONDITIONS......6 3 **3A4** 4<u>5</u> APPLICATION FOR SERVICE.....7A8 56 CREDIT 67 78 SERVICE AND MAIN EXTENSIONS.... <u>_____9</u> INTRODUCTION OF SERVICE 121313 89 COMPANY EQUIPMENT ON CUSTOMER'S PREMISES.. ______13 1112 1213 INSPECTIONS 1314 MEASUREMENT MEĄSUREMENT ______<u>14<u>15</u>15</u> DISCONNECTION BY THE COMPANY 1415 LOCAL DISTRIBUTION ADJUSTMENT CLAUSE AND NORMAL WEATHER ADJUSTMENT......33322 SUPPLY & CAPACITY SHORTAGE ALLOCATION POLICY41B44 RESIDENTIAL NON-HEATING RATE: CLASSIFICATION NO. R-1 4447 2 LOW INCOME RESIDENTIAL HEATING RATE: CLASSIFICATION NO. R-4... 4851 3 MANAGED EXPANSION PROGRAM RESIDENTIAL NON-HEATING RATE: MANAGED EXPANSION PROGRAM RESIDENTIAL HEATING RATE: 5 Formatted: Font: 10.5 pt DATED: ISSUED BY: /s/Neil Proudman Neil Proudman EFFECTIVE: August 1, 2021 TITLE: President Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105 ISSUED BY: /s/Susan L. Fleck March 20, 2019 Susan L. Fleck EFFECTIVE: November 4, 2018 TITLE: Authorized by operation of RSA 378:3, RSA 378:6, and Puc 1603.07(a) in Docket No. DG 17-191

Third Original Revised Page i

NHPUC NO. 10-11 GAS

Docket No. DG 21-130 Exhibit 26 Page 211 of 470

Docket No. DG 20-105 Exhibit 49

Third Original Revised Page i NHPUC NO. 10-11 GAS LIBERTY UTILITIES Superseding Second Revised Page i MANAGED EXPANSION PROGRAM LOW INCOME RESIDENTIAL HEATING RATE: CLASSIFICATION NO. R-7......5457 COMMERCIAL/INDUSTRIAL SERVICE: LOW ANNUAL USE, HIGH WINTER USE RATE CLASSIFICATION NO. G-41.....5659 COMMERCIAL/INDUSTRIAL SERVICE: MEDIUM ANNUAL USE, HIGH WINTER

Formatted: Font: 10.5 pt

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVI	E: August 1, 2021	TITLE:	President
	-		
Aut	norized by NHPUC Order No. XX,XXX d	ated XXX XX, 2021, in Docl	cet No. DG 20-105
DATED:	March 20, 2019	ISSUED BY:	/s/Susan L. Fleck
			Susan L. Fleck
EFFECTIVE:	November 4, 2018	TITLE:	President
Authe	rized by operation of RSA 378:3, RSA 37	8:6, and Puc 1603.07(a) in Do	cket No. DG 17-191

9	USE RATE CLASSIFICATION NO. G-436063
10	MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: LOW ANNUAL USE, HIGH WINTER USE RATE CLASSIFICATION NO. G-44. 6265
11	MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: MEDIUM ANNUAL USE, HIGH WINTER USE RATE CLASSIFICATION NO. G-45
12	MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: HIGH ANNUAL USE, HIGH WINTER USE RATE CLASSIFICATION NO. G-46 6669
13	COMMERCIAL/INDUSTRIAL SERVICE: LOW ANNUAL USE, LOW WINTER USE RATE CLASSIFICATION NO. G 51 COMMERCIAL/INDUSTRIAL SERVICE: LOW ANNUAL USE, LOW WINTER USE RATE CLASSIFICATION NO. G-51
14	COMMERCIAL/INDUSTRIAL SERVICE: MEDIUM ANNUAL USE, LOW WINTER USE RATE CLASSIFICATION NO. G-52
15	COMMERCIAL/INDUSTRIAL SERVICE: HIGH ANNUAL USE, LOAD FACTOR LESS THAN 90% RATE CLASSIFICATION NO. G-53
16	COMMERCIAL/INDUSTRIAL SERVICE: HIGH ANNUAL USE, LOAD FACTOR GREATER THAN 90% RATE CLASSIFICATION NO. G-54
17	MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: LOW ANNUAL USE, LOW WINTER USE RATE CLASSIFICATION NO. G-55 7679
18	MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: MEDIUM ANNUAL USE, LOW WINTER USE RATE CLASSIFICATION NO. G-56
19	MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: HIGH ANNUAL USE, LOAD FACTOR LESS THAN 90% RATE CLASSIFICATION NO. G-57
20	MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: HIGH ANNUAL USE, LOAD FACTOR GREATER THAN 90% RATE CLASSIFICATION NO. G-58
21	OUTDOOR GAS LIGHTING8386
22	FIRM RATE SCHEDULES - EXCLUDING KEENE CUSTOMERS84 <u>87</u>
23	FIRM RATE SCHEDULES - KEENE CUSTOMERS8588
24	FIRM RATE SCHEDULES - MANAGED EXPANSION PROGRAM - EXCLUDING KEENE CUSTOMERS
25	FIRM RATE SCHEDULES - OUTDOOR GAS LIGHTING8790
26	ANITCIPATED SUMMER PERIOD COST OF GAS - EXCLUDING KEENE CUSTOMERS8891
27	CALCULATION OF FIRM SALES SUMMER PERIOD COST OF GAS RATE -
DATED:	XXX XX, 2021 ISSUED BY: /s/Neil Proudman
EFFECTIVE:	August 1, 2021 Neil Proudman TITLE: President
DATED:	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105 March 20, 2019 ISSUED BY: /s/Susan L. Fleek
EFFECTIVE:	November 01, 2018 Susan L. Fleck President

Docket No. DG 21-130 Exhibit 26 Page 213 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10 11 GAS LIBERTY UTILITIES

Second Revised Original Page ii Table of Contents Superseding First Revised Page ii
Table of Contents

		abie of Contents
	EXCLUDING KEENE	
	CUSTOMERS89 <u>92</u>	
28	CALCULATION OF SUMMER PERIOD COST OF GAS RATE - KEEN CUSTOMERS93	

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX X	X, 2021, in Docket N	o. DG 20-105
DATED:	March 20, 2019	ISSUED BY:	/s/Susan L. Fleck
			Susan L. Fleck
EFFECTIVE:	November 01, 2018	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 214 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS

Third Revised Original Page iii

LIBERTY UTILITIES

<u>Table of Contents</u> <u>Superseding Second Revised Page iii</u>
<u>Table of Contents</u> <u>Table of Contents</u>

EFFECTIVE:	November 01, 2020 TITLE: President	
ISSUED:	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105 September 11, 2020 ISSUED BY: /s/Susan L. Fleek Susan L. Fleek	
EFFECTIVE:		
DATED:	XXX XX, 2021 ISSUED BY: /s/Neil Proudman Neil Proudman	
14	PEAKING SERVICE4	21 127
13	SALES SERVICE 1	20 126
12	BILLING AND SECURITY DEPOSITS 1	19 125
11	CAPACITY ASSIGNMENT	13 119
10	NON-DAILY METERED DELIVERY SERVICE 1	08 <u>114</u>
9	DAILY METERED DELIVERY SERVICE	
8	COMPANY GAS ALLOWANCE 1	04 110
7	POSSESSION OF GAS	04 110
6	QUALITY AND CONDITION OF GAS	03 109
5	CUSTOMER REQUEST FOR SERVICE FROM COMPANY 4	03 109
4	GAS SERVICE AREAS AND DESIGNATED RECEIPT POINTS4	02 108
3	CHARACTER OF SERVICE 1	
2	DEFINITIONS	 98 104
1	RATES AND CHARGES	
III. I	DELIVERY TERMS AND CONDITIONS	98 104
3 <u>8</u> 5.	NON-RESIDENTIAL ENERGY EFFICIENCY LOAN PROGRAM94	7B 103
3 <u>7</u> 5.		
3 <u>6</u> 5		
3 <u>5</u> 4	RATE CASE EXPENSE AND RECOUPMENT FACTOR CALCULATION	
3 <u>4</u> 3	ENVIRONMENTAL SURCHARGE - MANUFACTURED GAS PLANTS 95 99	
3 <u>3</u> 2	CALCULATION OF FIRM TRANSPORTATION COST OF GAS RATE	. 94 <u>98</u>
3 <u>2</u> 1	CALCULATION OF FIRM SALES AND FIXED WINTER PERIOD COST OF G. RATE KEENE CUSTOMERS	AS
30.1	ANTICIPATED WINTER PERIOD COST OF GAS - EXCLUDING KEENE CUSTO OR GAS LIGHTING	
30	CALCULATION OF FIRM SALES WINTER PERIOD COST OF GAS RATE EXCLUDING KEENE CUSTOMERS	
29	CALCULATION OF FIXED WINTER PERIOD COST OF GAS RATE EXCLUDE KEENE CUSTOMERS	

Docket No. DG 21-130 Exhibit 26 Page 215 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10 11 GAS

Third Revised Original Page iii

		Table of Contents	Superseding Second Revised Page iii
LIBERTY UTI	LITIES		Table of Contents Table of Contents
15	DISCONTINUANCE OF SERVIC	Е	
16	OPERATIONAL FLOW ORDERS	AND CRITICAL	DAYS 124 <u>130</u>
17	FORCE MAJEURE AND LIMITA	TION OF LIABIL	ITY <u>124<u>130</u></u>
18	CURTAILMENT		
19	TAXES		
20	SUPPLIER TERMS AND CONDI	TIONS	
21	CUSTOMER DESIGNATED REP	RESENTATIVE	

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX	XX, 2021, in Docket No	o. DG 20-105
ISSUED:	September 11, 2020	ISSUED BY:	/s/Susan L. Fleck
			Susan L. Fleck
EFFECTIVE:	November 01, 2020	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 216 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 1	<u>+11</u> GAS	First Revised Original Page iv
LIBERTY UTI	LITIES	Table of Contents Superseding Original Page iv
LIBERTY UTI	LITIES	Table of Contents
IV. A	TACHMENTS	
1	ATTACHMENT A Supplier Service Agree	ment
2	ATTACHMENT B Schedule of Administra	tive Fees and Charges 147 <u>153</u>
3	ATTACHMENT C Capacity Allocators	

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX X	XX, 2021, in Docket N	o. DG 20-105
ISSUED:	September 11, 2020	ISSUED BY:	/s/Susan L. Fleck
			Susan L. Fleck
EFFECTIVE:	November 01, 2020	TITLE:	President

<u>Page</u>	Revision
Title	Original
TOC i	Original Third Revised
TOC ii	Original Second Revised
TOC iii	Original Third Revised
TOC iv	Original First Revised
1	Original Sixty sixth Revised
2	Original Eighth Revised
3	Original Fifty sixth Revised
4	Original Fifty-fourth Revised
5	Original Fourth Revised
6	Original Second Revised
7	Original Second Revised
7A 8	Original
<u>8–9</u>	Original Fourth Revised
9 10	Original First Revised
<u> 1011</u>	Original
11 - <u>12</u>	Original
12 <u>13</u>	Original
<u> 1314</u>	Original First Revised
<u> 1415</u>	Original First Revised
<u> 1516</u>	Original First Revised
16 <u>17</u>	Original
17 18	Original
<u> 1819</u>	Original
19 20	Original
20 21	Original
21 - <u>22</u>	Original
22 <u>23</u>	Original
23 <u>24</u>	Original
<u>25</u> 24	Original
<u>26</u>	<u>Original</u>

ISSUED BY:	/s/Neil Proudman
	Neil Proudman
TITLE:	President
HPUC Order No. XX,XXX dated XXX XX, 2021, in Docket N	o. DG 20-105
ISSUED BY:	/s/Susan L. Fleck
	Susan L. Fleck
O TITLE.	President
	TITLE: HPUC Order No. XX,XXX dated XXX XX, 2021, in Docket N ISSUED BY:

Seventh Revised Original Page 2
Superseding Sixth Revised Page 2
Check Sheet

<u>Page</u>	Revision
25—	Original
26 —	First Revised
27	Original
28	Original
29	Original First Revised
30	Original
31	Original
32	Original Second Revised
33	Original First Revised
34	Original Third Revised
35	Original Second Revised
36	Original Third Revised
37	Original Third Revised
38	Original Second Revised
39	Original First Revised
40	Original First Revised
41	Original First Revised
41A <u>42</u>	Original
41B <u>43</u>	Original First Revised
4 <u>2</u> 44	Original
43 <u>45</u>	Original
<u>44—46</u>	Original Fifth Revised
<u>45—47</u>	Original
46 <u>48</u>	Fifth Revised Original
<u>47—49</u>	Original
<u>48—50</u>	Sixth Revised Original
<u>49—51</u>	Original
50 — <u>52</u>	Original Fourth Revised
51 — <u>53</u>	Original
52 — <u>54</u>	Fourth Revised Original
53 <u>55</u>	<u>Original</u>
5 4	<u>Original</u>

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX,	2021, in Docket N	o. DG 20-105
ISSUED:	June 4, 2021	ISSUED BY:	/s/Susan L. Fleck
			Susan L. Fleck
EFFECTIVE:	November 1, 2020	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 219 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10 <u>11</u> GAS LIBERTY UTILITIES Seventh Revised Original Page 2
Superseding Sixth Revised Page 2
Check Sheet

<u>57</u> <u>53</u>	<u>Original</u> Original
54 —	Fifth Revised
55 —	Original

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX,	2021, in Docket N	o. DG 20-105
ISSUED:	June 4, 2021	ISSUED BY:	/s/Susan L. Fleck
-			Susan L. Fleck
EFFECTIVE:	November 1, 2020	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 220 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES

Fifty seventh Revised Original Page 3
Superseding Fifty sixth Revised Page 3
Check Sheet

Page	Revision
<u>58</u>	Original
56 <u>59</u>	OriginalSixth Revised
57 <u>60</u>	Original
58 <u>61</u>	OriginalSixth Revised
59 62	Original
60 <u>63</u>	Original Fourth Revised
61 <u>64</u>	Original
62 65	Original Fifth Revised
63 66	Original
64 - <u>67</u>	Original Fourth Revised
65 68	Original
66 69	Original Fifth Revised
67 70	Original
68 71-	Original Fifth Revised
69 72	Original
70 73	Original Fourth Revised
71 74	Original
72 75	Original Fourth Revised
73 <u>76</u> —	Original
74 <u>77</u>	Original Fourth Revised
75 78	Original
76 <u>79</u>	Original Fifth Revised
77 80	Original
78 81	Original Fourth Revised
79 82	Original
80 83	Original Sixth Revised
81 <u>84</u>	Original
82 85	Original Fifth Revised
83 <u>86</u>	Original First Revised
84 <u>87</u>	<u>Original</u> ThirtiethRevised
85 88	Original Thirty second
86 89	Original Thirty first

Formatted: No underline

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
ISSUED:	Authorized by NHPUC Order No. XX,XXX dated XXX XX, May 27, 2021	2021, in Docket N ISSUED BY:	
EFFECTIVE:	June 01, 2021	TITLE:	President
	Authorized by NHPUC Order No. 26,374 dates June 30, 20:	20, in Docket No.	DG 20-049

Docket No. DG 21-130 Exhibit 26 Page 221 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES

Fifty seventh Revised Original Page 3
Superseding Fifty sixth Revised Page 3
Check Sheet

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
ISSUED:	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2 May 27, 2021		o. DG 20-105 /s/Susan L. Fleek
EFFECTIVE:	June 01, 2021	TITLE.	Dragidant

Authorized by NHPUC Order No. 26,374 dates June 30, 2020, in Docket No. DG 20-049

<u>Page</u>	Revision
87 <u>90</u>	Original First Revised
88 <u>91</u>	Original Third Revised
89 92	Original Fifteenth Revised
90 93	Original Fifteenth Revised
91 <u>94</u>	Original Third Revised
92 95	Original Fifteenth Revised
92.1 <u>96</u>	First Revised Original
93 97	Original Seventeenth Revised
9 4 <u>98</u>	Original Fourth Revised
95 99	Original Fourth Revised
96 100	Original Third Revised
97 101	Original Third Revised
97A 102	Original
97B 103	Original
98 <u>104</u>	Original
99 105	Original
100 106	Original
101 <u>107</u>	Original
102 108	Original
103 109	Original
104 <u>110</u>	Original
105 111	Original
106 112	Original
107 <u>113</u>	Original
108 114	Original
109 115	Original
110 116	Original
111 117	Original
112 118	Original
113 119	Original
<u>114120</u>	<u>Original</u>
<u>115121</u>	<u>Original</u>
116 122	Original

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

ISSUED:	June 24, 2020	ISSUED BY:	/s/Susan L. Fleck
			Susan L. Fleck
EFFECTIVE:	July 01, 2020	TITLE:	President

Authorized by NHPUC Order 26,351 dated April 30, 2020 in Docket No. 20-041, see complete revised Order 26,351 (05/01/20) 222

Docket No. DG 21-130 Exhibit 26 Page 223 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES

Fifty-fourth RevisedOriginal Page 4 Superseding Fifty-third Revised Page 4Check Sheet Check Sheet

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

ISSUED:	June 24, 2020	ISSUED BV-	/s/Susan L. Fleck
IBBULD.	June 24, 2020	ISSUED D1.	Susan L. Fleck
EFFECTIVE:	July 01, 2020	TITLE:	- President

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES

Third Revised Original Page 5
Check Sheet Superseding Second Revised Page 5
Check Sheet

Page	Revision
118 124	Original
119 - <u>125</u>	Original
120 <u>126</u>	Original
121 <u>127</u>	Original
122 - <u>128</u>	Original
123 129	Original
124 - <u>130</u>	Original
125 - <u>131</u>	Original
126 <u>132</u>	Original
127 <u>133</u>	Original
128 - <u>134</u>	Original
129 - <u>135</u>	Original
130 - <u>136</u>	Original
131 <u>137</u>	Original
132 <u>138</u>	Original
133 - <u>139</u>	Original First Revised
134 <u>140</u>	Original
135 <u>141</u>	Original
136 - <u>142</u>	Original
137 - <u>143</u>	Original
138 <u>144</u>	Original
139 - <u>145</u>	Original
140 <u>146</u>	Original
141 - <u>147</u>	Original
142 - <u>148</u>	Original
143 - <u>149</u>	Original
144 <u>150</u>	Original
145 <u>151</u>	Original
146 <u>152</u>	Original
147 <u>153</u>	Original Third Revised
148 - <u>154</u>	Original Third Revised

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
ISSUED:	Authorized by NHPUC Order No. XX,XXX dated XXX XX November 15, 2019		/c/Susan I Flools
EFFECTIVE:	November 01, 2019	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 225 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 6

I. GENERAL TERMS AND CONDITIONS

1 SERVICE AREA

A. <u>Service Area</u>. The area authorized to be served by the Company and to which this tariff applies are the following cities and towns: Allenstown, Amherst, Auburn, Bedford, Belmont, Berlin, Boscawen, Bow, Concord, Derry, Franklin, Gilford, Goffstown, Hanover, Hollis, Hooksett, Hudson, Keene, Laconia, Lebanon, Litchfield, Londonderry, Loudon, Manchester, Merrimack, Milford, Nashua, Northfield, Pelham, Pembroke, Sanbornton, Tilton, Windham, and part of Canterbury and Winnisquam.

2 GENERAL TERMS AND CONDITIONS

- A. <u>Filing</u>. A copy of this tariff is on file with the New Hampshire Public Utilities Commission ("NHPUC" or the "Commission") and is open to inspection at the offices of the Company.
- B. <u>Revisions</u>. This tariff may be revised, amended, supplemented, or otherwise changed from time to time in accordance with the rules of the Commission and such changes, when effective, shall have the same force as the original tariff.
- C. <u>Application</u>. The tariff provisions apply to everyone lawfully receiving gas supply service and/or delivery-only service from the Company under the rates herein and receipt of gas service shall constitute the receiver a customer of the Company as the term is used herein whether service is based upon contract, agreement, accepted signed application, or otherwise.
- D. <u>Statement by Agents</u>. No representative has the authority to modify a tariff rule or provision or to bind the Company by a promise or representation contrary thereto.
- E. No Prejudice of Rights. The failure of the Company to enforce any of the terms of this tariff shall not be deemed a waiver of its right to do so.
- F. <u>Gratuities to Employees</u>. The Company's employees are strictly forbidden to demand or accept any personal compensation or gifts for service rendered by them while working for the Company on the Company's time.
- G. <u>Advance Payments</u>. Payments to the Company for charges provided in these rules and regulations to be borne by the customer shall be made in advance.
- H. <u>Assignment</u>. Subject to the rules and regulations, all contracts by the Company shall be binding upon, and oblige, and continue for the benefit of, the successors and assigns, heirs, executors, and administrators of the parties hereto.

3 CHARACTER OF SERVICE – EXCLUDING KEENE CUSTOMERS

- A. <u>Gas Supply</u>. This Tariff applies only to the supply of gas, having a thermal content of nominally 1,000 British thermal units per cubic foot at supply pressures available in the locality in which the premises to be served are situated.
- B. <u>Determination of Therms</u>. The gas for any billing period, expressed in hundreds of cubic feet (ccf), shall be multiplied by the average BTU of the gas send out as determined below and divided by 1,000 in order to

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

EFFECTIVE: August 1, 2021 TITLE: President

C. Determine the number of therms consumed in the billing period. For billing purposes, gas therms shall be determined on a "dry" basis.

The BTU therm factor of the gas sendout shall be calculated for each billing cycle from the daily weighted average BTU of the natural gas delivered to the Company by its suppliers and the gas produced at the Company's peak-shaving plants. The daily average BTU content shall be determined by appropriate gas measurement devices operated by the Company or its supplier.

- D. <u>Delivery of Gas Supply</u>. The rates specified in this tariff are based upon the supply of service to a single customer through one delivery and metering point.
- E. <u>Use of Service at Separate Properties</u>. The use of service at two or more separate properties will not be combined for billing purposes.

3A4 CHARACTER OF SERVICE – KEENE CUSTOMERS

- A. <u>Gas Supply</u>. This Tariff applies to the supply of either propane-air gas or natural gas at the company's standard heat content value for the gas supply source, adjusted for temperature and pressure, in the locality in which the premises to be served are situated.
 - 1. For Keene customers on the propane air-gas distribution system, the supply of gas sold will be nominally 740 British thermal units per cubic foot.
 - 2. For Keene customers on the natural gas distribution system, the supply of gas sold will be nominally 1,000 British thermal units per cubic foot.

B. Determination of Therms.

1. The propane-air gas for any billing period, expressed in hundreds of cubic feet (CCF), shall be multiplied by the average BTU/cubic foot of the gas send out as determined below and divided by 1,000 in order to determine the number of therms consumed in the billing period.

The average BTU of the gas sendout for billing purposes shall be calculated for each billing period using the daily average BTU of propane-air gas produced by the company's production plant as determined by appropriate gas heat content measurement devices¹ operated by the company. The average BTU so calculated shall be used in determining the therms for monthly bills based on meter readings and the average BTU for the billing period.

2. The natural gas for any billing period, expressed in hundreds of cubic feet (CCF), shall be multiplied by the average BTU/cubic foot of the gas send out as determined below and divided by 1,000 in order to determine the number of therms consumed in the billing period.

The average BTU of the gas sendout for billing purposes shall be calculated for each billing period using the daily average BTU of compressed or liquefied natural gas delivered to the company in New Hampshire by its suppliers. To determine the daily average BTU, each tanker's BTU

¹ Typically chromatographs or calorimeters are used for measuring the heat content of gas.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX	K, 2021, in Docket N	o. DG 20-105
ISSUED:	March 20, 2019	ISSUED BY:	/s/Susan L. Fleck
			Susan L. Fleek
EFFECTIVE:	November 4, 2018	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 227 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES Second Revised Original Page 7 Superseding First Revised Page 7

content will be provided by the supplier or measured by the Company with appropriate gas heat content measurement devices, ¹

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX X	X, 2021, in Docket N	o. DG 20-105
ISSUED:	March 20, 2019	ISSUED BY:	/s/Susan L. Fleck
			Susan L. Fleck
EFFECTIVE:	November 4, 2018	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 228 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 8 General Terms and Conditions

including when more than one tanker is connected to the Company's natural gas system and simultaneously providing flow at any given time. The average BTU so calculated shall be used in determining the therms for monthly bills based on meter readings and the average BTU for the billing period.

5 CUSTOMER'S INSTALLATION

- A. <u>Point of Delivery</u>. Upon request, the Company will designate a point at which the customer shall terminate his piping for connection to the meter of the Company, but such information does not constitute an agreement or obligation on the part of the Company to furnish service.
- B. <u>Space for Meter</u>. The customer shall provide, free of expense to the Company, a dry, warm and otherwise suitable place for the regulator or regulators, meter or meters, or other equipment of the Company which may be necessary for the fulfillment of such contracts as may be entered into with the Company.
- C. <u>Location of Meter</u>. The space provided for the Company's meters and equipment shall be convenient access to the Company's employees and, as near as possible, to the point where the service supply pipe enters the customer's building. Its location shall be such that the meter connections are not concealed by plaster or sheathing and shall be otherwise acceptable to the Company.
- D. <u>Reverse Flow</u>. The customer may be required to install check valves or other devices to prevent compressed air or other gases from entering the Company's mains.

6 APPLICATION FOR SERVICE

- A. <u>Service Contract</u>. Every applicant for gas service may be required to sign a contract, agreement, or other form then in use by the Company covering the special circumstances of the applicant's use of gas and must agree to abide by the rules and regulations and standard requirements of the Company.
- B. <u>Right to Reject</u>. The Company may reject any application for service which would involve excessive cost to supply, or which might affect the supply of service to other customers, or for other good and sufficient reasons.
- C. <u>Special Contracts</u>. Standard contracts shall be for terms as specified in the statement of the rate, but where large or special investment is necessary for the supply of service, contracts of longer terms as specified in the rate, or with a special guarantee of revenue, or both, may be required to safeguard such investment.
- D. <u>Unauthorized Use</u>. Unauthorized connection to the Company's gas service supply facilities, and/or the use of service obtained from the Company without authority, or by any false pretense, may be terminated by the Company without notice. The use of service without notifying the Company and without enabling the

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 229 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Superseding Second Revised Page 8 Third Revised Original Page 9 8
General Terms and Conditions

General Terms and Conditions

company to read its meter will render the user liable for any amount due for service supplied to the premises from the time of the last meter reading of the Company's meter immediately preceding the user's occupancy as shown by the Company's books.

- E. <u>Managed Expansion Program</u>. The Managed Expansion Program targets gas expansion in specific areas that have high potential for demand. Each Managed Expansion Program project includes a Main Extension. Customers under this program avoid a portion or all of a contribution in aid of construction which would otherwise be required absent the Managed Expansion Program.
- F. Managed Expansion Program ("MEP") Premium. The markup for all MEP rates above the rates that would otherwise be applied absent the Managed Expansion Program. The MEP Premium is 30%.

7 CREDIT

- A. <u>Prior Debts.</u> Service will not be furnished to former customers until any indebtedness to the Company for previous service has been satisfied.
- B. <u>Deposits</u>. Before rendering or restoring service, the Company may require a deposit subject to the Commission's Rules and Regulations. (See Puc 1200 rules).

8 SERVICE AND MAIN EXTENSIONS

- A. <u>Definitions</u>. The following are definitions of terms used in these provisions relative to main and service extensions and are applicable only in the main and service extensions provisions.
 - Service and Main Extensions. Extensions that require the construction of a new gas main and a service from that new main in order to provide requested gas service to a customer.
 - Service Extensions. Extensions from an Existing Gas Main to the point of delivery on the customer's premises.
 - 3. Main Extension. An extension of the new gas main portion of a Service and Main Extension.
 - 4. Existing Gas Main. A main that is installed in the street and through which gas is flowing.
 - 5. Abnormal Costs. Abnormal Costs are service and/or main construction costs that are attributable to frost or ledge (including ditching or backfilling necessitated as a result of the presence of frost or ledge),and/or other conditions not typically encountered in service and/or main construction that are peculiar to the particular service and/or main construction concerned. Abnormal Costs are to be paid by the customer.
 - Extra Footage. The charge (contribution in aid of construction) for Extra Footage is \$63.90 per foot.
 The charge will be updated annually by calculating the historical average cost per foot for Service

 DATED:
 XXX XX, 2021
 ISSUED BY: /s/Neil Proudman

 EFFECTIVE:
 August 1, 2021
 TITLE:
 President

 Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

 ISSUED:
 March 31, 2021
 ISSUED BY: /s/Susan L. Fleek

 EFFECTIVE:
 April 01, 2021
 TITLE: President

Authorized by NHPUC Order No. 25,624 dated January 24, 2014 in Docket No. DG 13-198

Formatted: Normal, Indent: Hanging: 0.25"

Docket No. DG 21-130 Exhibit 26 Page 230 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES Superseding Second Revised Page 8 Third Revised Original Page 9 8
General Terms and Conditions

General Terms and Conditions

Extensions, excluding overheads, for the most recent calendar year and the updated charge shall be effective April 1.

7. Estimated Annual Margin. The Estimated Annual Margin is equal to the estimated revenue to be derived from the monthly Customer Charge and delivery charge to be received from the customer for gas service utilizing the Service and Main Extension or Service Extension during the first twelve (12) months after completion of the extension. The Estimated Annual Margin does not include revenue received by the Company for the cost of gas and local distribution adjustment factor.

Docket No. DG 21-130 Exhibit 26 Page 231 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Page 9 First Revised Original Page 910
General Terms and Conditions Superseding Original

General Terms and Conditions

- 8. Estimated Cost of Construction. For the purpose of determining the cost of Service and Main Extensions, Estimated Cost of Construction of mains and/or services includes the cost of labor and materials for such construction, and incidental or associated miscellaneous costs, but excluding overheads. Miscellaneous costs include, but are not limited to, meter(s), traffic control and city and town road permits and degradation fees. The customer may perform on-site trenching and backfilling in accordance with the Company's specifications, in which case the Estimated Cost of Construction will be reduced to reflect the costs avoided by the Company as a result of the customer's performance of the work.
- B. <u>Costs of Extensions</u>. In areas where the Company is authorized to operate, subject to the Application for Service provisions of this tariff, service is available as follows:
 - 1. Residential Service Extensions. Residential Service Extensions up to 100 feet in length will be installed at no charge to customers served under either a (i) residential heating rate; or (ii) a residential non-heating rate provided that such extension is installed during the installation of a Main Extension or during the performance of work on cast iron/bare steel main replacements; unless there are Abnormal Costs associated with such extensions, in which case the customer shall be charged for the Abnormal Costs. For residential Service Extensions in excess of 100 feet, the customer will be charged for the Extra Footage, plus any Abnormal Costs. This Section 7(B)(1) shall apply only to Service Extensions and shall not apply to Service and Main Extensions as described in Section 7(B)(3).
 - 2. Commercial and Industrial Service Extensions. Commercial and industrial Service Extensions will be installed at no charge to the customer provided that the Estimated Annual Margin is at least one-sixth of the Estimated Cost of Construction of the Service Extension, excluding any Abnormal Costs. If the Estimated Annual Margin is less than one-sixth of the Estimated Cost of Construction, the customer will be required to pay to the Company, in advance, any amount by which the Estimated Cost of Construction of the Service Extension exceeds six times the Estimated Annual Margin. Abnormal Costs are charged separately and are not included in the Estimated Cost of Construction for the purpose of this calculation. This Section 7(B)(2) shall apply only to Service Extensions and shall not apply to Service and Main Extensions as described in Section 7(B)(3).
 - 3. Service and Main Extensions of Less Than \$1,000,000. The Company shall not commence construction on a Service and Main Extension for which the Estimated Cost of Construction is less than \$1,000,000 until the sum of (i) six times the Estimated Annual Margin for all commercial and industrial customers who have committed to take service, plus (ii) eight times the Estimated Annual Margin for all residential customers who have committed to take service equals or exceeds 25% of the Estimated Cost of Construction.
 - a. <u>Residential.</u> Residential Service and Main Extensions will be installed at no charge to the customer provided that the Estimated Annual Margin is at least one-eighth of the Estimated Cost of Construction of the Service and Main Extensions. If the Estimated Annual Margin is less than one-eighth of the Estimated Cost of Construction, the customer will be required to pay to the Company the difference between the Estimated Cost of Construction and eight times the Estimated Annual Margin, plus any Abnormal Costs.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX X	X, 2021, in Docket N	o. DG 20-105
ISSUED:	March 20, 2019	ISSUED BY:	/s/Susan L. Fleek
-			Susan L. Fleek
EFFECTIVE:	May 01 2018	TITLE:	President
	Way 01, 2010	TITLE.	1 Testaetti

Docket No. DG 21-130 Exhibit 26 Page 232 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-<u>11</u> GAS LIBERTY UTILITIES Page 9 First Revised Original Page 910
General Terms and Conditions Superseding Original

General Terms and Conditions

If the Main Extension will serve more than one location, the Company will calculate the sum of the Estimated Annual Margin from all metered services and the sum of the Estimated Cost of Construction for the Main Extension and all Service Extensions to determine whether any payment will be required from the customers to be served. The Company will also include the

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX	1, 2021, in Docket N	o. DG 20-105
ISSUED:	March 20, 2019	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleck
EFFECTIVE:	May 01, 2018	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 233 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES Original Page <u>1011</u> General Terms and Conditions

determine whether any payment will be required from the customers to be served. The Company will also include the —Estimated Annual Margin and the Estimated Cost of Construction for Service Extensions for all existing premises for which the Company reasonably anticipates will take service, using the assumption that 60% of such premises will take service. If any payment is required, it will be allocated equally among all current metered services that exist as of the date that the Main Extension becomes an Existing Gas Main. Abnormal Costs associated with Main Extensions will be allocated equally among all customers, unless such costs can be attributed to specific customers.

- b. <u>Commercial and Industrial.</u> Commercial and industrial Service and Main Extensions will be installed at no charge to the customer provided that the Estimated Annual Margin is at least one-sixth of the Estimated Cost of Construction of the Service and Main Extensions. If the Estimated Annual Margin is less than one-sixth of the cost of construction of the Service and Main Extensions, the customer will be required to pay to the Company the difference between the Estimated Cost of Construction and six times the Estimated Annual Margin, plus any Abnormal Costs.
- c. If the Main Extension will serve more than one location, the Company will calculate the sum of the Estimated Annual Margin from all metered services and the sum of the Estimated Cost of Construction for the Main Extension and all Service Extensions to determine whether any payment will be required from the customers to be served. The Company will also include in such calculations the Estimated Annual Margin and the Estimated Cost of Construction for Service Extensions for all existing premises for which the Company reasonably anticipates will take service, using the assumption that 60% of such premises will take service. If any payment is required, it will be allocated among all current metered services that exist as of the date that the Main Extension becomes an Existing Gas Main based on each customer's proportional share of the Estimated Annual Margin. Abnormal Costs associated with Main Extensions will also be allocated based on each customer's proportional share of the Estimated Annual Margin, unless such costs can be attributed to specific customers, in which case the costs shall be allocated appropriately to specific customers.
- Extensions Serving Customers in More Than One Rate Class. If the Main Extension will serve both residential and commercial or industrial customers, the Company will determine whether a contribution will be required by the customers by calculating the difference between the Estimated Cost of Construction of the Main and Service Extensions and (i) six times the Estimated Annual Margin for all commercial and industrial customers to be served, plus (ii) eight times the Estimated Annual Margin for all residential customers to be served. The Company will also include in the above calculations the Estimated Annual Margin and the Estimated Cost of Construction of Service Extensions for all existing premises for which the Company reasonably anticipates will take service. If the difference described above is positive, the customers will be required to pay to the Company such difference. The amount of payment will be allocated among all metered services that exist as of the date that the Main Extension becomes an Existing Gas Main based on each customer's proportional share of the Estimated Annual Margin. Abnormal Costs associated with Main Extensions will also be allocated based on each customer's proportional share of the Estimated Annual Margin, unless such costs can be attributed to specific customers, in which case the costs shall be allocated appropriately to specific customers.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX X	XX, 2021, in Docket N	o. DG 20-105
ISSUED:	May 18, 2018	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleek
EFFECTIVE:	May 01, 2018	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 234 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES Original Page <u>1011</u> General Terms and Conditions

4. <u>Service and Main Extensions Greater Than or Equal to \$1,000,000</u>. If the cost of the Main Extension equals or exceeds \$1,000,000, then in addition to the requirements specified in Section 7(B)(3), the Company will not commence construction unless a discounted cash flow analysis

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
ISSUED:	Authorized by NHPUC Order No. XX,XXX dated XXX May 18, 2018		o. DG 20-105 /s/Susan L. Fleek
EFFECTIVE:	May 01, 2018	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 235 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Original Page 44<u>12</u> General Terms and Conditions

4. Service and Main Extensions Greater Than or Equal to \$1,000,000. If the cost of the Main Extension equals or exceeds \$1,000,000, then in addition to the requirements specified in Section 7(B)(3), the Company will not commence construction unless a discounted cash flow analysis

demonstrates a positive net present value over a 10-year period of the difference between the Estimated Annual Margin and the revenue requirement associated with the Estimated Cost of Construction.

A. C. —Failure to Use Installed Gas Service. If a customer fails, within nine months after the date as service is installed under this Section 7, either in whole or in part, to make use of the service, the customer will reimburse the Company for all costs of constructing, removing and retiring the service less any contribution in aid of construction made by the customer for the service, which will be forfeited.

B. _____Easements, Etc. The Company is not required to construct extensions other than in public ways unless the customer provides, in advance and without expense or cost to the Company, all necessary permits, consents, authorizations and right-of-way easements, satisfactory to the Company, for the construction, maintenance and operation of the pipeline.

C. E. Shortest Distance. Services are run the shortest practical safe distance to the meter location. However, a customer may have the Company install a longer alternate service provided that the customer pays for the extra expense in advance of installation.

D. F. Winter Construction. Ordinarily, no new service pipes or main extensions are installed during the winter conditions (when frost is in the ground) unless the customer defrays the extra expenses.

- E. G. Timing and Refunding of Contribution. Except as otherwise agreed by the Company under unusual circumstances, any required contribution in aid of construction will be made prior to installation by the Company of a service. To help cover the Company's expenses, damages and lost business, if substantial construction of the building or buildings for which gas service has been sought is not commenced by the earlier of (1) November 30th next following submission of the application; or (2) the date when the Company commences construction of the main and service concerned prior to withdrawal of the application, ten percent (10%) of the contribution will be forfeited to the Company and will not be returned to the customer. The balance of the contribution will be refunded if and when the application is withdrawn, or will be applied toward the new contribution if the customer submits a new application for service or subsequently commences construction of the building or buildings. A new application may be submitted at any time.
- F. H. Reasonable Duration and Non-Discrimination. Under none of the foregoing provisions will the Company be required to install service pipes or to contract main extensions where the business to be secured may not be of reasonable duration or will tend, in any way, to constitute unreasonable discrimination.
- G. I. <u>Title.</u> Title of all extensions constructed in accordance with the above shall be vested in the Company.
- H. J. Other Requirements. The Company generally will not approve any application or, if it shall have given such approval, will not proceed or continue with main and/or service construction unless the Company is satisfied
 - That the final site plans, sub-division plans and plans and specification for building or buildings
 to be served by the main and/or service concerned, including plans for waste disposal, water
 and other associated systems and facilities, have been prepared and approved by owner;

 DATED:
 XXX XX, 2021
 ISSUED BY:
 /s/Neil Proudman

 EFFECTIVE:
 August 1, 2021
 TITLE:
 President

 Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

 ISSUED:
 May 18, 2018
 ISSUED BY:
 /s/Susan L. Fleek

 EFFECTIVE:
 May 01, 2018
 TITLE:
 President

Authorized by NHPUC Order No. 26,122 dated April 27, 2018, in Docket No. DG 17-048

Formatted: Indent: Left: 0.25"

Formatted: List Paragraph, Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.25" + Indent at: 0.5"

Docket No. DG 21-130 Exhibit 26 Page 236 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES

Original Page <u>4412</u> General Terms and Conditions

That all permits, exceptions, approvals and authorizations of governmental bodies or agencies
required for construction of such building or buildings and associated systems and facilities
have been obtained;

 DATED:
 XXX XX, 2021
 ISSUED BY: /s/Neil Proudman Neil Proudman Neil Proudman Neil Proudman Neil Proudman President

 EFFECTIVE:
 August 1, 2021
 TITLE: President

 ISSUED:
 May 18, 2018
 ISSUED BY: /s/Susan L. Fleek Susan L. Fleek President

 EFFECTIVE:
 May 01, 2018
 TITLE: President

Authorized by NHPUC Order No. 26,122 dated April 27, 2018, in Docket No. DG 17-048

Docket No. DG 21-130 Exhibit 26 Page 237 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-<u>11</u> GAS LIBERTY UTILITIES Original Page <u>1213</u> General Terms and Conditions

1. That the final site plans, sub-division plans and plans and specification for building or buildings to be served by the main and/or service concerned, including plans for waste disposal, water and other associated systems and facilities, have been prepared and approved by owner;

That all permits, exceptions, approvals and authorizations of governmental bodies or agencies
required for construction of such building or buildings and associated systems and facilities have been
obtained;

- 2.3. That the customer is proceeding or plans promptly to proceed with such construction; and
- 3.4. That nothing has occurred or failed to occur which will or is likely to prevent or interfere with such construction.

8.9. INTRODUCTION OF SERVICE

- A. <u>Service Contract</u>. Every applicant for gas service may be required to sign a contract, agreement, or other form then in use by the Company covering the special circumstances of his use of gas and must agree to abide by the rules and regulations and standard requirements of the Company.
- B. <u>Defective Installation</u>. The Company may refuse to connect if, in its judgment, the customer's installation is defective, or does not comply with such reasonable requirements as may be necessary for safety, or is in violation of the Company's standard requirements.
- C. <u>Unsatisfactory Installation</u>. The Company may refuse to connect if, in its judgment, the customer's equipment or use thereof might injuriously affect the equipment of the Company or the Company's service to other customers.

9-10. COMPANY EQUIPMENT ON CUSTOMER'S PREMISES

- A. Meters and Regulators. The Company shall furnish and install, maintain and own, any meter or meters, regulator or regulators required in the supply of service. For certain large customers, the Company shall furnish, install and maintain, at the customer's expense, any remote meter reading equipment to record usage for daily balancing. Such equipment shall remain the property of the Company at all times.
- B. <u>Customer's Responsibility</u>. The customer shall be responsible for safekeeping of the Company's property while on the customer's premises. In the event of injury or destruction of any such property, the customer shall pay the costs of repairs and replacements.
- C. <u>Relocation and/or Replacement of Company Equipment</u>. The original service connection, including piping, meters and all other necessary or incidental equipment, which remains the property of the Company, shall be installed by the Company at its expense unless otherwise expressly provided in this tariff. Subsequent relocation and/or replacement of any such equipment on private property, whether it be for one or more service connections, shall be performed by the Company at the customer's expense unless such work is done at the request of the Company and for its convenience, in which case the Company shall bear the expense.
- D. <u>Protection by Customer</u>. The customer shall protect the equipment of the Company on his premises and shall not permit any persons, except a Company employee having a Company photo identification card or

Authorized by NHPUC Order No. 26,122 dated April 27, 2018, in Docket No. DG 17-048

 DATED:
 XXX XX, 2021
 ISSUED BY: Neil Proudman Neil Proudman Neil Proudman Neil Proudman President

 EFFECTIVE:
 August 1, 2021
 TITLE: President

 ISSUED:
 May 18, 2018
 ISSUED BY: Neil Proudman President

 EFFECTIVE:
 May 18, 2018
 ISSUED BY: Neil Proudman Neil President

 EFFECTIVE:
 May 01, 2018
 TITLE: President

Formatted: Indent: First line: 0"

Formatted: Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 9 + Alignment: Left + Aligned at: 0.25" + Indent at: 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 9 + Alignment: Left + Aligned at: 0.25" + Indent at: 0.5"

Docket No. DG 21-130 Exhibit 26 Page 238 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Original Page <u>4213</u> General Terms and Conditions

other Company identification, to break any seals upon or do any work on any meter, service supply pipe, or other equipment of the Company located on the customer's premises.

E. <u>Tampering</u>. In the event the Company's meter or other property is being tampered with or interfered with, the customer being supplied through such equipment shall pay the amount which the Company may estimate is due for service used but not registered on the Company's meter and for any repairs or replacements required as well as for costs of inspections, investigations, and protective installation.

 DATED:
 XXX XX, 2021
 ISSUED BY: /s/Neil Proudman Neil Proudman

 EFFECTIVE:
 August 1, 2021
 TITLE: President

 Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105
 ISSUED: /s/Susan L. Fleek

 ISSUED:
 May 18, 2018
 ISSUED BY: /s/Susan L. Fleek

 EFFECTIVE:
 May 01, 2018
 TITLE: President

Authorized by NHPUC Order No. 26,122 dated April 27, 2018, in Docket No. DG 17-048

Docket No. DG 21-130 Exhibit 26 Page 239 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES Page 13 First Revised Original Page 1314
General Terms and Conditions Superseding Original

General Terms and Conditions

estimate is due for service used but not registered on the Company's meter and for any repairs or replacements required as well as for costs of inspections, investigations, and protective installation.

- F. <u>Right of Access.</u> The Company's identified employees shall have access to the premises of the customer at all reasonable times for the purpose of reading meters, testing, repairing, removing, or exchanging any or all equipment belonging to the Company.
- G. Ownership and Removal. All equipment supplied by the Company shall remain its exclusive property and the Company shall have the right to remove the same from the premises of the customer at any time after the termination of service for whatever cause.

11 SERVICE CONTINUITY

- A. Regularity of Supply. The Company will use reasonable diligence to provide a continuous, regular and uninterrupted supply of service, but should the supply be interrupted by the Company for the purpose of making repairs, changes, or improvements in any part of its system for the general good of the service or the safety of the public, or should the supply of service be interrupted or fail by reason of accident, strike, legal process, state or municipal interference, or any cause whatsoever beyond its control, the Company shall not be liable for damages, direct or inconsequential, resulting from such interruption or failure.
- B. <u>Notice of Trouble</u>. The customer shall notify the office of the Company immediately should the service be unsatisfactory for any reason or should there be any defects, leaks, trouble, or accident affecting the supply of gas.

12 CUSTOMER'S USE OF SERVICE

- A. <u>Resale Forbidden</u>. The customer shall not, directly or indirectly, sell, sublet, assign, or otherwise dispose of to others, gas purchased from the Company, or any part thereof, without the consent of the Company. This rule does not apply to a public utility Company purchasing gas in bulk expressly for the purpose of delivering it to others.
- B. <u>Fluctuations</u>. Gas service must not be used in such a manner as to cause unusual fluctuations or disturbances in the Company's supply system. In the case of violation of this rule, the Company may discontinue service or require the customer to modify its installation and/or equip it with approved controlling devices.
- C. <u>Additional Load</u>. The service supply pipe, regulators, meters, and equipment supplied by the Company for each customer have definite capacities. The customer shall notify the Company of substantial changes in service requirements or location of appliances.

13 INSPECTIONS

A. <u>Company's Right to Inspect</u>. The Company shall have the right, but shall not be obliged, to inspect any installation before service is introduced or at any time later and reserves the right to reject any piping or appliances not in accordance with the Company's standard requirements. However, such inspection, failure to inspect, or failure to reject shall not render the Company liable or responsible for any losses or damage

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX X	X 2021 in Docket N	o DG 20-105
ICCLIED.	March 20, 2010	ICCLIED DV.	/g/Sugan I Floals
IDDULD.	Water 20, 201)	ISSOLD D1.	Susan L. Fleck
	N. 01 2010	TITLE.	Busan E. I ICCK
EFFECTIVE:	May 01, 2018	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 240 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-<u>11</u> GAS LIBERTY UTILITIES Page 13 <u>First RevisedOriginal</u> Page <u>1314</u> <u>General Terms and ConditionsSuperseding Original</u>

General Terms and Conditions

resulting from defects in the installation, piping or appliances, from violation of Company rules, or from accidents which may occur upon the premises of the customer.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX X	X, 2021, in Docket N	o. DG 20-105
ISSUED:	March 20, 2019	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleck
EFFECTIVE:	May 01, 2018	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 241 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 15 General Terms and Conditions

14 MEASUREMENT

- A. <u>Supply of Meters</u>. The measurement of gas service shall be by meters furnished and installed by the Company. The Company will select the type and make of metering equipment and may, from time to time, change or alter the equipment. The Company's sole obligation is to supply meters that will accurately and adequately furnish records for billing purposes.
- B. <u>Special Measurements</u>. The Company shall have the right, at its option and its own expense, to place demand meters, pressure gauges, special meters, or other instruments on the premises of any customer for the purpose of determining the adequacy of the Company's service or for making tests of all or any part of the customer's load.

15 METER TESTS

- A. Meter Tests. Meters are tested according to NHPUC Rules and Regulations. (See Puc 500 rules).
- B. <u>Request Tests</u>. The fee for a special request test is \$20.00 when scheduled at the mutual convenience of the Company and the customer; otherwise the amount is \$30.00. (See Puc 500 rules).
- C. <u>Customer's Bill Adjustment</u>. Should any meter fail to register correctly, the quantity of gas consumed will be determined by the Company based on information supplied by the customer and known by the Company subject to NHPUC Rules and Regulations. (See Puc 500 rules).

16 DISCONNECTION BY THE COMPANY

- A. <u>Disconnection by the Company</u>. The Company may disconnect its service to a customer for violation of its rules subject to NHPUC Rules and Regulations. (See Puc 1200 rules).
- B. <u>Non-Payment Shut-Off</u>. The Company may disconnect its service on reasonable notice and remove its equipment in case of non-payment of amounts billed for gas usage.
- C. <u>Shut-Off for Cause</u>. The Company may disconnect its service on reasonable notice if entry or access to its meter or meters is refused, obstructed, or hazardous, or for other violation of the Company's standard requirements.
- D. <u>Safety Shut-Off</u>. The Company may disconnect without notice if the customer's installation has become dangerous or defective.
- E. <u>Defective Equipment</u>. The Company may disconnect without notice if the customer's equipment, or use thereof, might injuriously affect the equipment of the Company or the Company's service to other customers.
- F. <u>Shut-Off for Fraud</u>. The Company may disconnect without notice for abuse, fraud or tampering with the connections, meters or other equipment of the Company.
- G. <u>Reconnection Charge</u>. A reconnection charge is made for reconnection of service discontinued by the Company and is payable in advance in addition to all other amounts due. The reconnection charge is made

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 242 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-<u>11</u> GAS LIBERTY UTILITIES Page 15 <u>First RevisedOriginal Page 1516</u> <u>General Terms and Conditions</u><u>Superseding Original</u>

Conoral Torms and Conditions

instead of the meter account charge. The amount of the reconnection charge is the same as the comparable meter account charge except when it has been necessary to dig up the service pipe or connection to effect discontinuance of service. In such cases, the reconnection charge is the price of removal and restoration of service pipe or connection.

16.1717. -COST OF GAS CLAUSE EXCLUDING KEENE

- A. Purpose. The purpose of this Cost of Gas Clause is to establish procedures that allow the Company, subject to the jurisdiction of the Commission, to adjust, on a semiannual basis, its rates for firm gas sales in order to recover the costs of gas supplies, along with any taxes applicable to those supplies, pipeline and storage capacity, production capacity and storage, bad debt expense associated with purchased gas costs, and the costs of purchased gas working capital, to reflect the seasonal variation in the cost of gas, and to credit to customers receiving firm service from the Company all supplier refunds and capacity release sales.
- B. <u>Applicability</u>. This Cost of Gas Clause ("COGC") shall be applicable to the Company and all firm gas sales made by the Company, unless otherwise designated. The application to the clause may, for good cause shown, be modified by the NHPUC. See Section 1617.1(N), "Other Rules."
- C. Cost of Firm Gas Allowable for COGC. All costs of firm gas including, but not limited to, commodity costs, taxes on commodity, demand charges, local production and storage costs, hedging related costs, other gas supply expense incurred to procure and transport supplies and commodity related bad debt expense, the gas used in Company operations, transportation fees, costs associated with buyouts of existing contracts, and purchased gas working capital may be included in the COGC. Any costs recovered through application of the COGC shall be identified and explained fully in the semiannual filings outlined in Section 16.117(M).
- D. Effective Date of Cost of Gas Factor. The seasonal Cost of Gas Factor ("COG") shall become effective upon NHPUC approval on the first day of each season as designated by the Company. Unless otherwise notified by the NHPUC, the Company shall submit COG filings as outlined in Section 16.117(M) of this clause on or before the first business day in September...
- E. <u>Definitions</u>. The following terms shall be defined in this section, unless the context requires otherwise.
 - 1. <u>Bad Debt Expense</u>: The uncollectible expense attributed to the portion of the Company's revenue associated with the recovery of gas costs under this clause.
 - Capacity Release Revenues: The economic benefit derived from the sale or release of transportation and storage capacity that the Company has under contract.
 - Carrying Charges: Interest expense calculated on the average monthly balance using the monthly
 prime lending rate, as reported by the Federal Reserve Statistical Release of Selected Interest Rates,
 and then added to the end of month balance.
 - Correction Factor: Seasonal Adjustment necessary to align the peak day volumes used to calculate
 the Commercial and Industrial load factor ratios with the seasonal Commercial and Industrial High
 Winter and Low Winter throughput volumes applied to the cost of gas calculations.

 DATED:
 XXX XX, 2021
 ISSUED BY: /s/Neil Proudman Neil Proudman Neil Proudman Neil Proudman President

 EFFECTIVE:
 August 1, 2021
 TITLE: President

 ISSUED:
 Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105
 Js/Susan L. Fleek Susan L. Fleek

 EFFECTIVE:
 May 01, 2018
 TITLE: President

Authorized by NHPUC Order No. 26,122 dated April 27, 2018, in Docket No. DG 17-048

Formatted: Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 17 + Alignment: Left + Aligned at: 0.79" + Indent at: 1.04"

Docket No. DG 21-130 Exhibit 26 Page 243 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Page 15 First Revised Original Page 1516
General Terms and Conditions Superseding Original

General Terms and Conditions

5. <u>Direct Gas Costs</u>: All purchased gas costs including supplier, storage and pipeline demand and commodity costs, as well as the commodity costs for local manufactured gas (Liquid Propane Gas ("LPG") and Liquefied Natural Gas ("LNG")).

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX	K, 2021, in Docket No	o. DG 20-105
ISSUED:	March 20, 2019	ISSUED BY:	/s/Susan L. Fleck
	<u> </u>		Susan L. Fleck
EFFECTIVE:	May 01, 2018	TITLE:	President
	* * / * *		

Authorized by NHPUC Order No. 26,122 dated April 27, 2018, in Docket No. DG 17-048

- Economic Benefit: The difference between the revenues received and the marginal cost determined to serve non-core customers.
- 7. <u>Inventory Finance Charges</u>: As billed in each Winter Season for annual charges. The total shall represent an accumulation of the projected charges as calculated using the monthly average of financed inventory at the existing or anticipated financing rate through a trust or other financing vehicle.
- 8. <u>Local Production and Storage Capacity Costs</u>: The costs of providing storage service from the Company's storage facilities (*i.e.*, LNG and LPG) as determined in the Company's most recent rate proceeding.
- 9. <u>Market Based Allocator ("MBA")</u>: The method used to allocate gas costs among Commercial and Industrial Customer Classifications. These ratios are presented in Section <u>16.117</u>(F).
- 10. Non-Core Commodity Costs: The commodity cost of gas assigned to non-core sales to which the COG is not applied.
- 11. Non-Core Sales: Sales made under non-traditional off-system sales.
- 12. <u>Non-Core Sales Margins</u>: The economic benefit derived from non-core transactions to which the COG is not applied, including non-core sales generated from the use of the Company's Gas Supply Resource portfolio.
- 13. Summer Commodity: The gas supplies procured by the Company to serve firm load in the Summer Season.
- 14. <u>Summer Demand</u>: The gas supply demand and transmission capacity procured by the Company to serve firm load in the Summer Season.
- 15. <u>Summer Season</u>: The calendar months May 1 through October 31.
- 16. <u>Off-System Sales Margin</u>: The economic benefit derived from the non-firm sales of natural gas supplies upstream of Company's distribution system.
- 17. Winter Commodity: The gas supplies procured by the Company to serve firm load in the Winter Season.
- 18. <u>Winter Demand</u>: Gas supply demand, peaking demands, storage and transmission capacity procured by the Company to service firm load in the Winter Season.
- 19. Winter Season: The calendar months November 1 through April 30.
- 20. <u>PR Allocator</u>: The percentage of annual capacity charges assigned to the Winter Season calculated using the Proportional Responsibility Method.
- 21. <u>Purchased Gas Working Capital:</u> The allowable working capital derived from Winter Season and Summer Season demand and commodity related costs.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX	K, 2021, in Docket N	o. DG 20-105
ISSUED:	May 18, 2018	ISSUED BY:	/s/Susan L. Fleek
	•		Susan L. Fleek
EFFECTIVE:	May 01, 2018	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 245 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 18 General Terms and Conditions

F. <u>Approved Cost</u>. The Cost of Gas calculation utilizes information periodically established by the NHPUC. The table below lists the approved costs factors:

Variable	Description	Approved Figure
MISC	Miscellaneous Overhead	\$13,170
PS	Production and Storage Capacity	\$1,980,428
WCA%	Working Capital Allowance Percentage	3.91%

Bad Debt % Measurement and Reconciliation Period	COG Recovery Period	Actual Bad Debt Rate	Bad Debt allowed Recovery Rate
May 2010 – April 2011	November 2011 – October 2012	Actual	Actual
		Greater than 2.9%	Actual less 0.4
May 2011 – April 2012	November 2012 – October 2013	2.5% to 2.9%	
		Less than 2.5%	
May 2012 - April 2013 and each	November 2013 - October	Greater than 3.3%	Actual less 0.8
subsequent May – April period	2014 and each subsequent November – October period	2.5% to 3.3%	2.5%
thereafter	thereafter	Less than 2.5%	Actual

If the Company's actual bad debt percentage is reduced to 2.5% or less during any 12 month period, which need not be the same 12 months as the measurement periods defined above, then beginning with the reconciliation filing for the period during which this bad debt percentage was achieved the Company shall thereafter recover its actual gas supply related bad debt on a fully reconcilable basis and the percentages in the table above shall no longer apply. The actual bad debt percentage shall be calculated by dividing the Company's actual net write-offs for the relevant measurement period by its revenue for the same period.

G. Cost of Gas (COG) Calculations by Customer Class. The COG Formula shall be computed on a semiannual basis for three (3) groups of customer classes as shown on the following table. The computation will use forecasts of seasonal gas costs, carrying charges, sendout volumes, and sales volumes. Forecasts shall be based on either historical data or Company projections, but must be weather-normalized. Any projections must be documented in full with each filing.

The COG for the Residential rate classes shall represent the total system average unit cost of gas of meeting firm sales load, projected in each COG filing. The Commercial & Industrial (C&I) Low Winter (LW) and High Winter (HW) rates will be calculated in the following way: first, the demand unit cost of gas, the sum of purchased and stored gas demand costs divided by projected prorated sales, will be multiplied by the applicable load factor ratio and then multiplied by the correction factor. This adjusted demand factor will then be added to the commodity factor, adjustment factor and indirect cost of gas rate to determine the total COG rates for the C&I LW and HW rate classes. The two load factor ratios shall be derived once a year, for effect every November 1 through October 31, using the ratio of the unit capacity cost of each C&I group to the total system unit capacity cost that is determined in the Company's submittal of its Capacity

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

EFFECTIVE: August 1, 2021 TITLE: President

Allocators, for Capacity Assignment purposes, filed with its Winter COG, and as presented in Attachment C of the Delivery Service Terms and Conditions. The Correction Factor aligns the peak day volumes used to calculate the load factor ratios with the seasonal throughput volumes applied to the COG calculations.

GROUP	CUSTOMER CLASSES
Residential	Residential Heating and Non-Heating
Commercial and Industrial: Low Winter Use	G-51 through G-58
Commercial and Industrial: High Winter Use	G-41 through G-46

Winter Season Cost of Gas Formula (CGw)

The Winter Season COG shall be comprised of Winter Demand costs, Winter commodity costs, Winter reconciliation costs, Winter working capital reconciliation, Winter bad debt expenses, local production and storage capacity costs, and miscellaneous and A&G costs calculated at the beginning of the Winter Season according to the following formula:

$$CGw = Dw + Cw + Rw + WCRw + BDw + PS + ((MISC + Rbd) \times W:Sales)$$
A:Sales

Winter Demand Cost (Dw) Formula

 $\label{eq:Dw} \begin{aligned} Dw &= DEMw - NCSMw + WCwd - R1d - R2d \\ &\text{and:} \end{aligned}$

NCSMw = CRRw + OSSMw + SBdw and:

 $WCwd = (DEMw - NCSMw) \times WCA\% \times CC$ where:

CGw = The total cost of gas for the Winter Season for the Company's firm sales customers previously defined.

BDw = Bad Debt expense for the Winter Season.

Cw = Commodity-related direct gas cost for the Winter Season.

Dw = The total Winter Demand costs.

DEMw = Demand Charges allocated to the Winter Season defined in Section 16.117(E).

NCSMw = The Non-Core Sales Margins equal to the sum of the Winter Season returnable Capacity Release Revenues, and Off-System Sales Margins.

WCwd = Working Capital allowable associated with demand charges allocated to the Winter Season as defined in Section $\frac{16.117}{100}$ (K).

R1d, R2d = Supplier demand-related refunds - The Supplier refunds associated with refund program credits derived from Account 5541-8048, "Undistributed Gas Suppliers' Refunds." See Section 16.117(I).

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
ISSUED:	Authorized by NHPUC Order No. XX,XXX dated XXX XX, May 18, 2018		/s/Susan L. Fleek
			Susan L. Fleck

Docket No. DG 21-130 Exhibit 26 Page 247 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES Original Page <u>1819</u> General Terms and Conditions

CRRw = The returnable Capacity Release Revenues allocated to the Winter Season. See Section 16.117(E).

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX	XX, 2021, in Docket N	o. DG 20-105
ISSUED:	May 18, 2018	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleek
EFFECTIVE:	May 01, 2018	TITLE:	President
EITEGIT E	1.1uj 01, 2010	11122.	1100100111

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES

Original Page <u>1920</u> General Terms and Conditions

- OSSMw = The returnable Off-System Sales Margins allocated to the Winter Season. See Section 16.117(E).
- SBdw = Demand revenues received from Firm Stand-By Sales Service customers in the Winter Season.
- WCA % = Percentage of gas costs equivalent to Working Capital Allowance associated with gas costs. Refer to Section 16.117(F) for this percentage.
- CC = Monthly interest rate as reported by the Federal Reserve Statistical Release of Selected Interest Rates.
- Rw = Reconciliation Costs Winter Season deferred gas costs, Account 1920-171740 balance, inclusive of the associated Account 1920-171740 interest, as outlined in Section 16.117(J).
- WCRw = Working Capital reconciliation adjustment associated with Winter Gas Costs Account 1163-1422 balance as outlined in Section 16.117(K).
- PS = The total dollar amount of costs associated with the local production and storage capacity gas less any production and storage capacity assignment revenues. Refer to Section 16.117(F) for this dollar amount.
- MISC = The total dollar amount of gas costs associated with acquisition, dispatching, Administrative and General expenses and overheads as determined in the Company's most recent rate proceeding. Refer to Section 16.117(F) for this dollar amount.
- Rbd = Annual Bad Debt Expense reconciliation adjustment Account 1920-<u>1717</u>43 balance
- W:Sales = Forecasted firm sales volumes associated with the Winter Season.
- A:Sales = Forecasted annual firm sales volumes.

Winter Season Commodity (Cw) Formula

Cw = COMw + FC + WCwc - R1c - R2cand:

COMw = WSC - NCCCw - SBcw

 $WCwc = (COMw + FC) \times WCA\% \times CC$ where:

- COMw = Commodity Charges allocated to the Winter Season as defined in Section 16.117(E).
- FC = Inventory finance charges as defined in Section $\frac{16.117}{(E)}$.
- WCwc = Working Capital Allowable Associated with commodity charges allocated to the Winter Season as defined in Section $\frac{16.117}{16.117}$ (K).
- R1c, R2c = Supplier commodity-related refunds The supplier refunds associated with refund program credits derived from Account 5541-8048, "Undistributed Gas Suppliers' Refunds". See Section 16.117(I).
- WSC = Commodity charges associated with gas supply sent out in Winter Season as defined in Section 16.117(E)

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX,	2021, in Docket N	o. DG 20-105
ISSUED:	May 18, 2018	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleek
EFFECTIVE.	May 01 2019	TITLE.	Dragidant

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES Original Page 2021
General Terms and Conditions

- NCCCw = Non-Core Commodity Costs incurred in the Winter Season as defined in Section 16.117(E).
- SBcw = Winter Season commodity revenues received from Firm Stand-By Gas Supply Service sales customers.
- WCA % = Percentage of gas costs equivalent to Working Capital Allowance associated with gas costs. Refer to Section 16.117(F) for this percentage.
- CC = Monthly interest rate as reported by the Federal Reserve Statistical Release of Selected Interest Rates.

Winter Bad Debt (BDw) Formula

BDw = BD% x (Dw + Cw + Rw + WCRw)where:

BDw = Forecasted gas supply cost related Bad Debt Expense calculated for Winter Season.

BD% = Bad Debt percentage calculated based on a twelve month basis ending April of each year. Refer to Section 16.117(F) Bad Debt Allowed Recovery Rate for this percentage.

Dw = Demand related costs in the Winter Season as previously defined.

Cw = Commodity related costs in the Winter Season as previously defined.

Rw = Reconciliation Costs – Winter Season deferred gas costs as previously defined.

WCRw = Winter Season Working Capital Reconciliation adjustment as previously defined.

Residential Winter Season Cost of Gas (COGwr)

All residential firm sales customers will pay the same Cost of Gas for the Winter Season. The factor represents the total forecasted Winter Season average cost of gas rate. This factor is calculated according to the following formula:

COGwr = CGw W:Sales where:

CGw = The total cost of gas for the Winter Season for the Company's firm sales customers previously defined.

W:Sales = Forecasted sales volumes associated with the Winter Season.

R = Designates the Residential Heating and Residential Non-Heating customer classes.

Summer Season Cost of Gas (COG) Formula (CGs)

The Summer Season COG shall be comprised of Summer demand costs and Summer commodity costs, Summer reconciliation costs, Summer working capital reconciliation, plus a Summer bad debt charge, and a miscellaneous and A&G charge calculated at the beginning of the Summer Season according to the following formula:

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
ISSUED:	Authorized by NHPUC Order No. XX,XXX dated XXX XX, May 18, 2018	ICCLIED DV	/g/Cugon I Flools
IBBULD.	Way 10, 2010	ISSOLD D1.	Susan L. Fleek
EFFECTIVE:	May 01, 2018	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 250 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Original Page 2122 General Terms and Conditions

Formatted: Heading 5, Left, Don't keep with next

 $CGs = Ds + Cs + Rs + WCRs + BDs + ((MISC + Rbd) \times \frac{S:Sales}{A:Sales})$

Summer Demand Cost (Ds) Formula

 $\begin{aligned} Ds &= DEMs + WCsd - Rld - R2d \\ &and: \end{aligned}$

 $WCsd = DEMs \times WCA\% \times CC$ where:

A:Sales = Forecasted annual sales volumes.

BDs = Bad Debt Expense for Summer Season.

Cs = Commodity-related direct gas costs for the Summer Season.

CGs = The total cost of gas for the Summer Season for the Company's firm sales customer previously defined.

DEMs = Demand charges allocated to the Summer Season defined in Section $\frac{16.1}{17}$ (E).

MISC = The total dollar amount of gas costs associated with acquisition, dispatching, Administrative and General expenses and overheads as determined in the Company's most recent rate proceeding. Refer to Section 16.117(F) for this dollar amount.

R1d, R2d = Supplier refunds from pipeline demand charges - The per unit supplier refunds associated with refund program credits derived from Account 5541-8048, "Undistributed Gas Suppliers' Refunds." See Section 16.117(I).

Rs = Summer Season Reconciliation Costs - Account 1920-<u>1717</u>41 balance, inclusive of the associated Account 1920-<u>1717</u>41 interest, as outlined in Section <u>16.1-17</u>(J).

S:Sales = Forecasted sales volumes associated with the Summer Season.

WCA % = Percentage of gas costs equivalent to Working Capital Allowance associated with gas costs. Refer to Section 16.117(F) for this percentage.

CC = Monthly interest rate as reported by the Federal Reserve Statistical Release of Selected Interest Rates.

Rbd = Annual Bad Debt Expense reconciliation adjustment - Account 1920-171743 balance.

WCRs = Working Capital reconciliation adjustment associated with Summer gas costs – Account 1163-1424 as outlined in Section 16.117(K).

WCsd = Working Capital allowable costs associated with demand costs allocated to the Summer Season as defined in Section 146.117(K).

Summer Season Commodity Cost (Cs) Formula

Cs = COMs - NCCCs + WCsc - R1c - R2c and:

 $WCsc = (COMs - NCCCs) \times WCA\% \times CC$

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

ISSUED: May 18, 2018 ISSUED BY: /s/Susan L. Fleel

EFFECTIVE: May 01, 2018 TITLE: President

Authorized by NHPUC Order No. 26,122 dated April 27, 2018, in Docket No. DG 17-048

Docket No. DG 21-130 Exhibit 26 Page 251 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Original Page 2122 General Terms and Conditions

where:

Authorized by NHPUC Order No. 26,122 dated April 27, 2018, in Docket No. DG 17-048

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES Original Page 2<u>23</u> General Terms and Conditions

- COMs = Commodity charges associated with gas supply sent out in the Summer Season as defined in Section 16.117(E).
- WCsc = Working Capital allowable costs associated with commodity charges allocated to the Summer Season as defined in Section $\frac{16.117}{(K)}$.
- R1c, R2c = Supplier refunds from pipeline commodity charges The supplier refunds associated with refund program credits derived from Account 5541-8048, "Undistributed Gas Suppliers' Refunds."
- NCCCs = Non-core commodity costs incurred in the Summer Season as defined in Section $\frac{16.117}{(E)}$.
- WCA % = Percentage of gas costs equivalent to Working Capital Allowance associated with gas costs. Refer to Section $\frac{16.117}{F}$ for this percentage.
- CC = Monthly interest rate as reported by the Federal Reserve Statistical Release of Selected Interest Rates.

Summer Bad Debt (BDs) Formula

- BDs = BD% x (Ds + Cs + Rs + WCRs)where:
- BD% = Bad Debt percentage calculated based on a twelve month basis ending April of each year. Refer to Section 16.117(F) Bad Debt Allowed Recovery Rate for this percentage.
- BDs = Forecasted gas supply related Bad Debt Expense calculated for Summer Season defined in Section 16.117(E).
- Ds = Demand related costs in the Summer Season as previously defined.
- Cs = Commodity related costs in the Summer Season as previously defined.
- Rs = Reconciliation Costs Summer deferred gas costs as previously defined.
- WCRs = Summer Season Working Capital Reconciliation adjustment as previously defined.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX,	2021, in Docket N	o. DG 20-105
ISSUED:	May 18, 2018	ISSUED BY:	/c/Sucan I Flook
			/S/Dusan L. I ICCK
		100022 211	Susan L. Fleek
EFFECTIVE:	May 01, 2018	TITLE:	Sugan I Flack

Docket No. DG 21-130 Exhibit 26 Page 253 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 24 General Terms and Conditions

Residential Summer Season Cost of Gas (COGsr)

All residential firm sales customers will pay the same cost of gas for the Summer Season. The factor represents the total forecasted Summer Season average cost of gas rate. This factor is calculated according to the following formula:

COGsr = <u>CGs</u> S:Sales where:

CGs = The total cost of gas for the Summer Season for the Company's firm sales customers as previously defined.

S:Sales = Forecasted sales volumes associated with the Summer Season.

R = Designates the Residential Heating and Residential Non-Heating customer classes.

Commercial and Industrial Winter and Summer Season Cost of Gas

The Commercial and Industrial customer classes Winter Season Cost of Gas will be based on the Winter Season average cost of gas components used for the Residential Winter Season Cost of Gas. A separate Winter Season Cost of Gas factor will be computed for the low winter use class, Rates G-51, G-52, G-53, G-54, G-55, G-56, G-57, and G-58 and a separate Winter Season Cost of Gas Factor will be computed for the high winter use class, Rates G-41, G-42, G-43, G-44, G-45, and G-46.

The Commercial and Industrial customer classes Summer Season Cost of Gas will be based on the Summer Season average cost of gas components used for the Residential Summer Season Cost of Gas. A separate Summer Season Cost of Gas factor will be computed for the low winter use class, Rates G-51, G-52, G-53,G-54, G-55, G-56, G-57, and G-58 and a separate Summer Season Cost of Gas factor will be computed for the high winter use class, Rates G-41, G-42, G-43, G-44, G-45, and G-46.

These Cost of Gas Factors will be computed by applying ratios to the average demand portion of the Winter and Summer Season's cost of gas unit rate times the correction factor and then adding the remaining Residential average cost of gas unit rate.

These factors are calculated according to the following formulas:

Low Winter Use (COGwl) Formula Winter Season

 $COGwl = RATIOl \times CFw \times CGwd + CGwo$

Low Winter Use (COGsl) Formula Summer Season

COGsl = RATIOl x CFs x CGsd + CGso and:

RATIOl = DCcl
DDcl
DDc

and:

High Winter Use (COGwh) Formula Winter Season

 $COGwh = RATIOh \times CFw \times CGwd + CGwo.$

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 25 General Terms and Conditions

High Winter Use (COGsh) Formula Summer Season

COGsh = RATIOh x CFs x CGsd + CGso and

 $RATIOh = \underline{DCch} \quad \div \quad \underline{DCc}$ $DDch = \quad DDc$

and:

CFw = (WL:Sales + WH Sales)

(RATIOl x WL:Sales) + (RATIOh x WH:Sales)

CFs = (SL:Sales + SH:Sales)

(RATIOl x SL:Sales) + (RATIOh x SH:Sales)

 $CGwd = \underline{Dw}$ W:Sales

CGwo = CGw - Dw

W:Sales

 $CGsd = \underline{Ds}$

S:Sales

 $CGso = \underline{CGs - Ds}$

S:Sales

DCcl = Bcl * PLrate + (DDcl - Bcl) * REMrate

DCch = Bch * PLrate + (DDch-Bch) * REMrate

PLrate = PL/PLmdcq

 $REMrate = (\underline{DCc - (Bc*PLrate)})$

DDc - Bc

 $DCc = (\underline{DC \times DDc})$

DD

where:

Bc = The daily base load for all the Commercial and Industrial rate classes

Bch = The daily base load for the Commercial and Industrial rate classes G-41, G-42, G-43, G-

44, G-45 and G-46.

Bcl = The daily base load for the Commercial and Industrial rate classes G-51, G-52, G-53, G-

54, G-55, G-56, G-57 and G-58.

CFs = Summer Season Commercial and Industrial gas cost correction factor.

CFw = Winter Season Commercial and Industrial gas cost correction factor.

CGs = The total cost of gas for the Summer Season for the Company's firm sales customers as

previously defined.

CGw = The total cost of gas for the Winter Season for the Company's firm sales customers as

previously defined.

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman

EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 255 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES

DC =

Original Page <u>2526</u> General Terms and Conditions

	production and storage capacity costs, as stated in Section 1617.1(F).
DCc =	The Commercial and Industrial rate classes pro-rata share of the annual forecasted pipeline, storage, and peaking demand capacity costs.
DCch =	The Commercial and Industrial pro-rata share of the annual forecasted pipeline, storage, and peaking demand capacity costs allocated to Commercial and Industrial High Winter Use rate classes, G-41, G-42, G-43, G-44, G-45, and G-46.
DCcl =	The Commercial and Industrial pro-rata share of the annual forecasted pipeline, storage, and peaking demand capacity costs allocated to the Commercial and Industrial Low Winter Use rate classes, G-51, G-52, G-53, G-54, G-55, G-56, G-57, and G-58.
DD =	Total peak design day determinants.
DDc =	The peak design day determinants allocated for all the Commercial and Industrial rate classes.
DDch =	The peak design day determinants for the Commercial and Industrial rate classes, G-41, G-42, G-43, G-44, G-45, and G-46.
DDcl=	The peak design day determinants for the Commercial and Industrial rate classes, G-51, G-52, G-53, G-54, G-55, G-56, G-57, and G-58.
Ds =	The total Summer Demand charges as defined below.
Dw =	The total Winter Demand charges as previously defined.
PL =	The annual forecasted pipeline only demand charges
PLmdcq =	The maximum daily contract pipeline volume available to the Company.
PLrate =	The pipeline demand rate.
RATIOh =	Ratio of the average high Winter Use class Cost of Gas low load factor demand capacity costs to the total average Commercial and Industrial demand capacity_costs.
RATIOl =	Ratio of the average low Winter Use class Cost of Gas high load factor demand capacity costs to the total average Commercial and Industrial demand capacity costs.
REMrate =	The weighted average demand rate for storage and peaking supplies.
S: Sales =	Forecasted sales volumes associated with the Summer Season.
SH:Sales =	Total Winter Season forecasted Commercial and Industrial high winter use sales.
SL: Sales =	Total Winter Season forecasted Commercial and Industrial low winter use sales volumes.
W:Sales =	Forecasted sales volumes associated with the Winter Season.
WH:Sales =	Total Winter Season forecasted Commercial and Industrial high winter use sales.
WL: Sales =	Total Winter Season forecasted Commercial and Industrial low winter use sales volumes.

The annual forecasted pipeline, storage and peaking demand charges plus the total

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX X	XX, 2021, in Docket N	o. DG 20-105
ISSUED:	May 18, 2018	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleek
EFFECTIVE:	May 01, 2018	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 256 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES Original Page <u>2526</u> General Terms and Conditions

<u>H.</u> Non-Core Sales Margins ("NCSM"). One hundred percent (100%) of margins from Off-System Sales and all revenues from Capacity Release will be credited to firm sales customers during the winter season through operation of the COG.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX,	2021, in Docket N	o. DG 20-105
ISSUED:	May 18, 2018	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleek
EFFECTIVE.	May 01 2019	TITLE.	President
DITECTIVE.	Willy 01, 2015	IIILE:	- President

Docket No. DG 21-130 Exhibit 26 Page 257 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Page 26 First Revised Original Page 2627
General Terms and Conditions Superseding Original

General Terms and Conditions

I. <u>Gas Suppliers' Refunds</u>. Account 5541-8048: Refunds from suppliers of gas, from upstream capacity suppliers and suppliers of product demand are credited to Account 5541-8048, "Commodity and Demand Refunds." Transfers from these accounts will reflect as a credit in the semiannual calculation of the COG to be calculated as follows:

Refund programs shall be initiated with each semiannual COG filing and shall remain in effect for a period of one year. The total dollars to be placed into a given refund program shall be net of over/under-returns from expired programs plus refunds received from suppliers since the previous program was initiated. Refunds shall be segregated by demand and commodity charges and distributed volumetrically, producing per unit refund that will return the principal amount with interest as calculated using the Company's average short-term cost of borrowing for the month to the average of the beginning and end of month balances of refunds. The Company shall track and report on all Account 5541-8048 activities as specified in Section 16.117(K).

- J. Reconciliation Adjustments Various Accounts.
 - 1. The following definitions pertain to reconciliation adjustment calculations:
 - a. Capacity Costs Allowable per Winter Season Formula shall be:
 - Charges associated with upstream storage transmission capacity and product demand procured by the Company to serve firm load in the Winter Season, plus a reallocation of a portion of such charges incurred in the Summer Season to serve firm load.
 - (2) Charges associated with peaking, downstream production and storage capacity to serve firm load dispatching costs, and other administrative and general expenses in connection with purchasing gas supplies in the Winter Season from the Company's most recent test year and allocated to firm sales service.
 - (3) Non-Core Sales Margins or economic benefits associated with returnable capacity release and off-system sales.
 - (4) Credits associated with firm Stand-by Gas Supply Service Monthly Reservation Charges, daily imbalance charges and fixed component of penalty charges billed transportation customers in the Winter peak Season.
 - (5) Winter Season Demand Cost carrying charges.
 - b. Gas Costs Allowable Per Winter Season Formula shall be:
 - (1) Charges associated with gas supplies, including any applicable taxes, purchased by the Company to serve firm load in the Winter Season.
 - (2) Credit non-core commodity costs assigned to non-core customers to which the COGC does not apply, as defined in Section 16.117(H) (NCCCw).
 - (3) Inventory finance charges (FC).
 - (4) Winter Season commodity cost carrying charges.

XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
		Neil Proudman
August 1, 2021	TITLE:	President
March 20, 2019	ISSUED BY:	Susan L. Fleek
May 01, 2018	TITLE:	Susan L. Fleck President
	August 1, 2021 Authorized by NHPUC Order No. XX,XXX dated XXX XX March 20, 2019	August 1, 2021 TITLE: Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket N March 20, 2019 ISSUED BY:

Docket No. DG 21-130 Exhibit 26 Page 258 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Page 26 First Revised Original Page 2627
General Terms and Conditions Superseding Original

General Terms and Conditions

- c. Capacity Costs Allowable Per Summer Season Formula shall be:
 - (1) Charges associated with transmission capacity and product demand procured by the Company to serve firm load in the Summer Season

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX,	2021, in Docket N	o. DG 20-105
			01 2 0 20 100
ISSUED:	March 20, 2019	ISSUED BY:	/s/Susan L. Fleek
	March 20, 2019	ISSUED BY:	/c/Susan I Flack
	March 20, 2019 May 01, 2018	ISSUED BY: TITLE:	/s/Susan L. Fleek

Original Page 2728
General Terms and Conditions

- (2) Credits associated with daily imbalance charges and fixed component of penalty charges billed transportation customers in the Summer Season.
- (3) Summer Season demand cost carrying charges.
- d. Gas Costs Allowable Per Summer Season Formula shall be:
 - (1) Charges associated with gas supplies, including any applicable taxes, procured by the Company to serve firm load in the Summer Season.
 - (2) Non-core commodity costs associated with non-core sales to which the COG is not applied, as defined in Section 16.117(E).
 - (3) Summer Season commodity cost carrying charges.
- e. Costs Allowable Per Bad Debt Formula shall be:
 - (1) Costs associated with uncollected gas costs, incurred by the Company to serve sales load. Such costs represent the bad debt expense related to the gas supply related write-off of sales customers and will be computed by multiplying actual gas costs by the Bad Debt Allowed Recovery Rate specified in Section 16.117(F). The reconciliation adjustment each season will be computed as the difference between the actual supply related bad debt revenues and the actual gas costs multiplied by the actual Bad Debt Allowed Recovery Rate as specified in Section 16.117(F).
 - (2) Account 1920-171743 Annual Bad Debt, carrying charges.
- 2. Calculation of the Reconciliation Adjustments: These accounts contain the accumulated difference between gas cost revenues and the actual monthly gas costs incurred by the Company. The Company shall separate Account 17155 into Winter Season Gas Costs (Account 1920-171740) and Summer Season Gas Costs (Account 1920-171741), Account 1920-171740 shall contain the accumulated difference between revenues toward gas costs calculated by multiplying the Winter Season Gas Cost for each Customer Classification, (COGwr, COGwl and COGwh) times monthly firm sales volumes for each Customer Classification, and the total costs allowable per the Winter Season gas cost formula. Account 1920-171741 shall contain the accumulated difference between revenues toward gas costs calculated by multiplying the Summer Season Gas Cost for each Customer Classification, (COGsr, COGsl and COGsh) times monthly firm sales volumes for each Customer Classification, and the total gas costs allowable per the Summer Season demand formula.

Carrying Charges shall be calculated on the average monthly balance of each subaccount. The interest rate is to be adjusted monthly using the monthly prime lending rate, as reported by the Federal Reserve Statistical Release of Selected Interest Rates.

The annual bad debt reconciliation adjustments Rbd shall be determined for use, incorporating the bad debt balances in Account 1920-171743.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
ISSUED:	Authorized by NHPUC Order No. XX,XXX dated XXX X		o. DG 20-105
	Way 16, 2016	ISSCED D1.	Susan L. Fleek
EFFECTIVE:	May 01, 2018	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 260 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES Original Page 2728
General Terms and Conditions

The bad debt account balance contains the accumulated difference between the Bad Debt Allowed Recovery Rate for the applicable period multiplied by the actual gas costs and the actual supply related bad debt revenues for the Winter and Summer COG filings.

The Winter Season reconciliation shall be filed with the NHPUC no later than July 29 of each year.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX	K, 2021, in Docket N	o. DG 20-105
ISSUED:	May 18, 2018	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleck
EFFECTIVE:	May 01, 2018	TITLE:	President

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES Original Page 2829
General Terms and Conditions

The Summer Season reconciliation shall be filed with the NHPUC no later than January 31 of each year.

- K. Working Capital Reconciliation Adjustments Accounts 1163-1422 and 1163-1424.
 - 1. The following definitions pertain to reconciliation adjustment calculations:
 - a. Working Capital Demand Gas Costs Allowable per Winter Season Gas Formula shall be:
 - (1) Charges associated with upstream storage, transmission capacity, and product demand procured by the Company to serve firm load in the Winter period, plus a reallocation of a portion of such charges incurred in the Summer Season to serve firm load.
 - (2) Carrying charges.
 - b. Working Capital
 - (3) Charges associated with gas supplies, including any applicable taxes, purchased by the Company to serve firm load in the Winter season.
 - (4) Non-core commodity costs associated with non-core sales to which the COG is not applied, as defined in Section 16.117(E).
 - (5) Carrying charges.
 - c. Working Capital Demand Gas Costs Allowable per Summer Season Gas Formula shall be:
 - (6) Charges associated with upstream storage and transmission capacity procured by the Company to serve firm load in the Summer Season.
 - (7) Carrying charges.
 - d. Working Capital Commodity Gas Costs Allowable per Summer Season Gas Formula shall be:
 - (8) Charges associated with gas supplies, including any applicable taxes, procured by the Company to serve firm load in the Summer Season.
 - (9) Non-core commodity costs associated with non-core sales.
 - (10) Carrying charges.
 - e. The Winter and Summer Cost of Gas working capital allowances shall be calculated by applying the Working Capital Allowance Percentage (WCA%) set forth in Section 16.117(F).
 - 2. Calculation of the Reconciliation Adjustments
 - a. Accounts 1163-1422 and 1163-1424 contain the accumulated difference between working capital allowance revenues and the actual monthly working capital allowance cost. The actual monthly working capital allowance shall be calculated by multiplying the actual gas costs times

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX,	2021, in Docket N	o. DG 20-105
ISSUED:	May 18, 2018	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleck
EFFECTIVE.	May 01 2019	TITI E.	
LITECTIVE.	 May 01, 2018	HHLE:	

Docket No. DG 21-130 Exhibit 26 Page 262 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES Original Page <u>2829</u> General Terms and Conditions

the Working Capital Allowance Percentage (WCA%) set forth in Section 16.117(F), to the actual Direct Gas Costs allowable.

b. The Winter Season working capital reconciliation adjustment (WCRw) shall be determined for use in the Winter Season Gas Cost calculations incorporating the Winter Season working

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX	2021, in Docket N	o. DG 20-105
ISSUED:	May 18, 2018	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleck
EFFECTIVE:	May 01, 2018	TITLE:	President
	•		

Docket No. DG 21-130 Exhibit 26 Page 263 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES Page 29 First Revised Original Page 2930
General Terms and Conditions Superseding Original

General Terms and Conditions

capital account 1163-1422. A Summer Season working capital reconciliation adjustment (WCRs) shall be determined for use in the Summer Season Gas Cost calculations incorporating the Summer Season working capital account 1163-1424 balance.

L. Application of COG to Bills: The Company will employ the COGs as follows: The COGs (\$/therm) for each customer group for each season shall be calculated to the nearest hundredth of a cent per unit and will be applied to each customer's monthly sales volume within the corresponding customer classification. The Cost of Gas will be applied to gas consumed on or after the first day of the month in which the cost of gas becomes effective.

M. Information Required to be Filed with the NHPUC.

Reconciliation Adjustments: The Company shall file with the NHPUC a seasonal reconciliation of gas
costs and gas cost collections containing information in support of the reconciliation calculation set out
in Sections 16.117(J) (2) and 16.117(K) (2). Such information shall include the complete list of gas
costs recoverable through the COGC over the previous season. This seasonal reconciliation shall be
filed with the respective seasonal COG reconciliation filing, along with complete documentation of the
reconciliation adjustment calculations.

Additionally, information pertaining to the Cost of Gas shall be filed with the NHPUC in accordance with the format established by the NHPUC. Reporting requirements include filing the Company's monthly calculation of the projected over or under-collection with the NHPUC, along with notification by the Company to the NHPUC of any revised COG for the subsequent month, not less than five (5) business days prior to the first day of the subsequent month.

The Company shall file with the NHPUC an annual reconciliation of bad debt expense and bad debt collections containing information in support of the reconciliation calculation set out in Sections 16.117(J) (1) and 16.117(J) (2). Such information shall detail the revenues collected as an allowance for bad debt, as well as the actual bad debt expense associated with gas cost recoverable through the COGC over the 12-month period ending April 30th. This annual reconciliation of bad debt expenses shall be filed with the Winter COG reconciliation filing, along with documentation.

- 2. Commercial and Industrial COG Ratio: The following factors will be filed annually by the Company for informational purposes. Significant changes in these factors signal the need to evaluate the COG ratios. These variables will assist in predicting significant shifting of the MBA-based escalator of gas costs and resulting changes in the COG ratios:
 - a. The percentage of load migration from sales to transportation service in the Commercial and Industrial High and Low Winter Use classes.
 - b. The ratio of delivered costs of winter supplies to pipeline delivered supplies.
 - c. The July and August consumption for the Commercial and Industrial High and Low Winter classes as a percentage of their annual consumption.

N. Other Rules.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2	2021, in Docket N	o. DG 20-105
ISSUED:	March 20, 2019	ISSUED BY:	/s/Susan L. Fleek
	Susan L. Fleek		
EFFECTIVE:	May 01, 2018	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 264 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-<u>11</u> GAS LIBERTY UTILITIES Page 29 First Revised Original Page 2930
General Terms and Conditions Superseding Original

General Terms and Conditions

1. The NHPUC may, where appropriate, on petition or on its own motion, grant an exception from the provisions of this tariff, upon such terms that it may determine to be in the public interest.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX	XX, 2021, in Docket N	o. DG 20-105
ISSUED:	March 20, 2019	ISSUED BY:	/s/Susan L. Fleek
	Susan L. Fleek		
EFFECTIVE:	May 01, 2018	TITLE:	President
	* *		

Docket No. DG 21-130 Exhibit 26 Page 265 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 31 General Terms and Conditions

- 2. The Company may, without further NHPUC action, adjust the approved COG upward or downward monthly based on the Company's calculation of the projected over or under-collection for the period, but the cumulative adjustments upward shall not exceed twenty-five percent (25%) of the approved COG.
- 3. The Company may, at any time, file with the NHPUC an amended COG.
- 4. The operation of the Cost of Gas Clause is subject to all powers of suspension and investigation vested in the NHPUC.
- 5. The Company shall file both seasonal COG filings on or before the first business day in September. The summer portion of the filing will include COG rates effective May 1 of the following year.
- O. Reconciliation Adjustment Accounts.

1163-1422

<u>Winter Season Gas Working Capital Allowance Reconciliation Adjustment for COGC</u>: This account shall be used to record the cumulative difference between Winter Season gas working capital allowance revenues and Winter Season gas working capital allowance. Entries to this account shall be determined as outlined in the Cost of Gas Clause.

1163-1424

<u>Summer Season Gas Working Capital Allowance Reconciliation Adjustment for COGC</u>: This account shall be used to record the cumulative difference between Summer Season gas working capital allowance revenues and Summer Season gas working capital allowance. Entries to this account shall be determined as outlined in the Cost of Gas Clause.

1920-1740

<u>Winter Season Gas Cost Reconciliation Adjustment for COGC</u>: This account shall be used to record the cumulative difference between Winter Season gas revenues and Winter Season gas costs. Entries to this account shall be determined as outlined in the Cost of Gas Clause.

1920-1741

<u>Summer Season Gas Cost Reconciliation Adjustment for COGC</u>: This account shall be used to record the cumulative difference between Summer Season gas revenues and Summer Season gas costs. Entries to this account shall be determined as outlined in the Cost of Gas Clause.

1920-1743

<u>Annual Bad Debt Reconciliation Adjustment for COGC</u>: This account shall be used to record the cumulative difference between Annual bad debt revenues and annual bad debt costs. Entries to this account shall be determined as outlined in the Cost of Gas Clause.

5541-8048

<u>Commodity and Demand Refunds</u>: This account shall be used to record the refunds from upstream commodity supplies and suppliers of product commodity and transfers of credits in the semiannual calculation of the COG as well as to record the refunds from upstream capacity supplies and suppliers of product demand and transfer of credits in the semiannual calculation of the COG. Entries to this account shall be determined as outlined in the Cost of Gas Clause.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

EFFECTIVE: August 1, 2021 TITLE: President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Docket No. DG 21-130 Exhibit 26 Page 266 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-<u>11</u> GAS LIBERTY UTILITIES Original Page 3132
General Terms and Conditions

- P. <u>Firm Transportation Cost of Gas Charge</u>. To permit the Company to charge its firm transportation customers with a portion of the cost of gas produced by the Company between November 1 and April 30 of each year, there is a Firm Transportation Cost of Gas Charge ("FTCG") which applies to all firm transportation billed under this tariff. This volumetric charge is to compensate firm sales customers for the increase in gas costs, through the use of supplemental liquid fuels, attributable to firm transportation customers during the Winter Period.
 - Application. The FTCG will be calculated for the Winter Period, defined as the period from November 1 through April 30. The FTCG will be applied to billings commencing with the first November revenue billing cycle
 - 2. Purpose. The amount of the FTCG is the estimated liquid costs used for pressure support purposes multiplied by the transportation throughput as a percentage of the total system throughput for the Winter Period. The resulting amount shall be adjusted by the prior period over or under collection, if any, and shall be recovered over the estimated total transportation throughput subject to the FTCG to yield a per therm volumetric charge. The FTCG shall be computed to the nearest one hundredth cent per therm and shown separately on customers' bills. At the conclusion of the Winter Period, the Company will calculate the extent that the FTCG revenues are greater or lesser than actual unit cost. The revenue and liquid costs will be reconciled so that all liquids costs shall be collected from either firm sales or firm transportation customers.
 - Changes. The amount of the FTCG may be changed within the period whenever the unit cost
 materially deviates from the anticipated unit cost
 - 4. <u>Reporting</u>. The Company shall submit to the New Hampshire Public Utilities Commission, on or before the first business day in September, a copy of the FTCG computation. A reconciliation of the prior period under/over collection will be submitted to the New Hampshire Public Utilities Commission no later than July 29.
- Q. Fixed Price Option Program. Fixed Price Option Program. An alternative to the traditional Winter Period cost of gas pricing mechanism may be elected by a residential customer (rates R-1, R-3, R-4, R-5 or R-6) pursuant to the Company's Fixed Price Option Program (the "Program"). The Company may offer up to 50% of its weather normalized firm sales for the prior Winter Period under the Program. The cost of gas rate offered under the Program will remain fixed for all Winter Period deliveries commencing November 1 and ending April 30. The Company shall submit to the New Hampshire Public Utilities Commission on or before September 1 of each year a copy of the fixed price option computation. Once elected, customers must remain on the Program for the duration of the Winter Period, unless service is terminated. There are no maximum or minimum usage levels. No sign up fees apply.

16.2 18 COST OF GAS CLAUSE – KEENE DIVISION

A. <u>Purpose</u>. To permit the Company to charge its customers in the Keene Division with the cost of gas purchased or produced. A cost of gas rate will be applied to all firm gas billed under this tariff as calculated on the appropriate pages.

Formatted: Font: Bold

Formatted: Heading 3, Indent: Left: 0.56", First line: 0"

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
ISSUED: EFFECTIVE:	Authorized by NHPUC Order No. XX,XXX dated XXX XX,2 May 18, 2018 May 01, 2018	2021, in Docket N ISSUED BY: TITLE:	o. DG 20-105 /s/Susan L. Fleek Susan L. Fleek President

Authorized by NHPUC Order No. 26,122 dated April 27, 2018, in Docket No. DG 17-048

Docket No. DG 21-130 Exhibit 26 Page 267 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-<u>11</u> GAS LIBERTY UTILITIES Original Page 3132 General Terms and Conditions

B. <u>Application</u>. A cost of gas rate will be calculated for the winter heating period, defined as the period from November 1 through April 30, and a cost of gas rate will be calculated for the summer period, defined as the period from May 1 through October 31.

 DATED:
 XXX XX, 2021
 ISSUED BY: /s/Neil Proudman Neil Proudman

 EFFECTIVE:
 August 1, 2021
 TITLE: President

 ISSUED:
 Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

 ISSUED:
 May 18, 2018
 ISSUED BY: /s/Susan L. Fleek

 EFFECTIVE:
 May 01, 2018
 TITLE: President

Authorized by NHPUC Order No. 26,122 dated April 27, 2018, in Docket No. DG 17-048

267

Docket No. DG 21-130 Exhibit 26 Page 268 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS

Original Page 33

LIBERTY UTILITIES

Terms and Conditions

NHPUC NO. 10 GAS LIBERTY UTILITIES Second Revised Page 32

uperseding First Revised Page 32

Formatted: Left: 0.89", Right: 0.92", Top: 0.46", Bottom: 0.19", Header distance from edge: 0.5", Footer distance from edge: 0.35", Different first page header

Formatted: Font: (Default) Times New Roman, 10.5 pt

The winter cost of gas rate will be applied to billings commencing with the first November revenue billing cycle; the summer cost of gas rate will be applied to billings commencing with the first May revenue billing cycle.

C. Calculation. The amount of the cost of gas rate is the anticipated unit cost of gas sold.

At the conclusion of each winter and summer period the Company will calculate the extent that cost of gas revenues are greater or less than actual unit costs of gas compared with the anticipated unit costs. The calculated difference (actual gas sales volumes multiplied by the difference between actual and anticipated unit costs) will be carried forward into the computation of the cost of gas rate for the corresponding winter or summer period.

Any excess revenue collected, as determined above, will earn interest as specified by the Commission.

- D. <u>Changes</u>. The cost of gas rate may be adjusted without further Commission action based on the projected over-/under-collection of gas costs, the adjusted rate to be effective the first of the month. Any such rate adjustments may not exceed a maximum rate of 25 percent above the approved rate, but there is no limit on the amount of any rate reductions.
- E. <u>Refunds</u>. When refunds are made to the Company by its suppliers that are applicable to increased charges collected under this provision, the Company will make appropriate refunds to its customers and as the Commission may direct.
- F. Reporting. The Company shall submit to the Commission, at least 30 days prior to the effective date, the proposed winter and summer period cost of gas rate computation. Any monthly adjustments to the cost of gas rate must be filed five (5) business days prior to the first day of the subsequent month (the effective date of the new rate).

The cost of gas rate shall be computed to the nearest one hundredth cent per therm and shown on customers' bills

G. Fixed Price Option Program. An alternative to the traditional winter period cost of gas rate mechanism may be elected by the customer pursuant to the Company's Fixed Price Option (FPO) Program. The Company may offer up to 50% of its expected firm sales for the winter period under the FPO Program. The cost of gas charge offered under the FPO Program will remain fixed for all winter period billings commencing November 1 and ending April 30 of the effective winter period. Once elected, customers must remain on the FPO Program for the duration of the winter period unless service is terminated. There are no maximum or minimum usage levels. Customers may enroll in this Program by contacting the Company between the October 1 and October 19 period immediately preceding the effective winter period.

1719 LOCAL DISTRIBUTION ADJUSTMENT CLAUSE AND NORMAL-WEATHER ADJUSTMENT

A. Purpose. The purpose of the Local Distribution Adjustment Clause ("LDAC" or this "Clause") is to establish procedures that allow the Company, subject to the jurisdiction of the NHPUC, to adjust, on an annual basis, its delivery charges in order to recover Conservation Charges ("CC"), Revenue Decoupling Adjustment Factor ("RDAF"), Winter Period Surcharges ("WPS"), Environmental

DATED: XXX XX, 2021

ISSUED BY: /s/Neil Proudman

Neil Proudman

EFFECTIVE: August 1, 2021

TITLE: President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Formatted: Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 19 + Alignment: Left + Aligned at: 0.57" + Indent at: 0.82"

Formatted: Font: 10.5 pt

Formatted: Font: (Default) Times New Roman, 10.5 pt

Docket No. DG 21-130 Exhibit 26 Page 269 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS Original Page 45 Formatted: Highlight

LIBERTY UTILITIES

Surcharges ("ES") including the Relief Holder Surcharge ("RHS") and the Manufactured Gas Program Surcharge ("MGP"), rate case expenses and recoupment ("RCE"), Residential LowGas Assistance Program costs ("GAP")

NHPUC NO. 10 GAS

First Revised Page 33

LIBERTY UTILITIES Superseding Original Page 3

DATED: XXX XX, 2021 EFFECTIVE: January 01, 2019 ISSUED BY: /s/N Su Raw Udr Fleeck
TITLE: Nellte Sidentman August 1, 2021 TITLE: President
Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Formatted: Footer, Centered, Line spacing: single, Tab stops: 0.88", Left + 4.94", Right + 5.13", Left

Docket No. DG 21-130 Exhibit 26 Page 270 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS Original Page 33
LIBERTY UTILITIES Terms and Conditions

NHPUC NO. 10 GAS

Second Revised Page 32

LIBERTY UTILITIES

Superseding First Revised Page 32

The winter cost of gas rate will be applied to billings commencing with the first November revenue billing eyele; the summer cost of gas rate will be applied to billings commencing with the first May revenue billing eyele.

C. Calculation. The amount of the cost of gas rate is the anticipated unit cost of gas sold.

At the conclusion of each winter and summer period the Company will calculate the extent that cost of gas revenues are greater or less than actual unit costs of gas compared with the anticipated unit costs. The calculated difference (actual gas sales volumes multiplied by the difference between actual and anticipated unit costs) will be carried forward into the computation of the cost of gas rate for the corresponding winter or summer period.

Any excess revenue collected, as determined above, will earn interest as specified by the Commission.

- D. <u>Changes</u>. The cost of gas rate may be adjusted without further Commission action based on the projected over /under collection of gas costs, the adjusted rate to be effective the first of the month. Any such rate adjustments may not exceed a maximum rate of 25 percent above the approved rate, but there is no limit on the amount of any rate reductions.
- E. Refunds. When refunds are made to the Company by its suppliers that are applicable to increased charges collected under this provision, the Company will make appropriate refunds to its customers and as the Commission may direct.
- F. Reporting. The Company shall submit to the Commission, at least 30 days prior to the effective date, the proposed winter and summer period cost of gas rate computation. Any monthly adjustments to the cost of gas rate must be filed five (5) business days prior to the first day of the subsequent month (the effective date of the new rate).

The cost of gas rate shall be computed to the nearest one hundredth cent per therm and shown on customers' bills.

G. Fixed Price Option Program. An alternative to the traditional winter period cost of gas rate mechanism may be elected by the customer pursuant to the Company's Fixed Price Option (FPO) Program. The Company may offer up to 50% of its expected firm sales for the winter period under the FPO Program. The cost of gas charge offered under the FPO Program will remain fixed for all winter period billings commencing November 1 and ending April 30 of the effective winter period. Once elected, customers must remain on the FPO Program for the duration of the winter period unless service is terminated. There are no maximum or minimum usage levels. Customers may enroll in this Program by contacting the Company between the October 1 and October 19 period immediately preceding the effective winter period.

17 LOCAL DISTRIBUTION ADJUSTMENT CLAUSE AND NORMAL WEATHER-ADJUSTMENT

n. <u>Purpose</u>. The purpose of the Local Distribution Adjustment Clause ("LDAC" or this "Clause") is toestablish procedures that allow the Company, subject to the jurisdiction of the NHPUC, to adjust, on an annual basis, its delivery charges in order to recover Conservation Charges ("CC"), Revenue—Decoupling Adjustment Factor ("RDAF"), Winter Period Surcharges ("WPS"), Environmental—

 DATED:
 XXX XX, 2021
 ISSUED BY: /s/Neil Proudman

 EFFECTIVE:
 August 1, 2021
 TITLE: President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Docket No. DG 21-130 Exhibit 26 Page 271 of 470 Docket No. DG 20-105 Exhibit 49

Surcharges ("ES") including the Relief Holder Surcharge ("RHS") and the Manufactured Gas-Program Surcharge ("MGP"), rate case expenses and recoupment ("RCE"), Residential LowGas-Assistance Program costs ("GAP")

DISSUED: XMX/ch/X0/2019 ISSUEDBY: /s/Saiskinducfleck
Siskinducfleck
EFFECTINE: Angwentb@CO1, 2018 TITTEE: President

 $\underline{\text{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{betibe}|\textbf{ChyPUPC}|\textbf{Of}|\textbf{Out}|\textbf{Chi}|\textbf{O}|\textbf{Chi}|\textbf{A}|\textbf{X}|\textbf{X}|\textbf{X}|\textbf{X}|\textbf{Aut}|\textbf{ch}|\textbf{Trid}|\textbf{X}|\textbf{X}|\textbf{A}|\textbf{X}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{Aut$

Docket No. DG 21-130 Exhibit 26 Page 272 of 470 Docket No. DG 20-105

Exhibit 49

NHPUC NO. 11 GAS

Original Page 34

Formatted: Font: (Default) Times New Roman, 10.5 pt

LIBERTY UTILITIES NHPUC NO. 10 GAS Terms and Conditions

Formatted: Left: 0.93", Right: 0.92", Top: 0.46", Bottom: 1.06", Header distance from edge: 0.5", Footer

LIBERTY UTILITIES

Applicability

Residential Non-

Space

Small C&I G-41, G

Large C&I G-43, G

51. G 44. G 55 Medium C&I G 42

53. G 54. G 46

First Revised Page 33 rseding Original Page 33

distance from edge: 0.35" Formatted: Indent: Left: 0.69"

Income Assistance Program costs ("RLIAP") and any other expenses the NHPUC may approve from time to time. The purpose of the Normal Weather-Adjustment ("NWA") is to establish procedures that allow the Company, subject to the jurisdiction of NHPUC, to calculate and apply, for each customer on a monthly basis, the Normal Weather Factor ("NWF").

Applicability. This Clause shall be applicable in whole or part to all of the Company's firm sales service and firm delivery service customers as shown on the table below. The application of this clause may, for good cause shown, be modified by the NHPUC. See Section 1719(K) "Other Rules." 2

RDA

4

1

4

4

4

X

X

X

X

X

4

1

1

4

1

X

X

X

X

Formatted: No underline, Underline color: Auto

Formatted: Left, Indent: Left: 0.37", Hanging: 0.31", No bullets or numbering, Tab stops: 0.38", Left + Not at 0.61"

Formatted: Indent: Left: 0.37", Hanging: 0.31"

Formatted: Indent: Left: 0.37", Hanging: 0.31" Formatted: No underline, Underline color: Auto

Notes:

—This Clause is a Applicable to all rates and charges on a Customer Cl Class basis. The application of the CC, RDAF, ES, RCE and GAP is applicable to all therms and therefore the application to all customers, including Managed Expansion Program Customers, is at the same rate per therm as the corresponding non-Managed Expansion Program Customers.

Applicable to Non-Managed Expansion Program Customers

Formatted: No underline, Underline color: Auto

Formatted: Left, Indent: Left: 0.32", No bullets or numbering, Tab stops: 0.38", Left + 0.69", Left + Not at

Conservation Charges Allowable for LDAC.

- Purpose: The purpose of this provision is to establish a procedure that allows the Company, subject to the jurisdiction of the NHPUC, to adjust, on an annual basis, the Conservation Charge, if and when applicable, to firm sales service and firm delivery service throughput in order to recover from firm customers costs and lost margins associated with its energy efficiency management programs.
- Applicability: A conservation charge shall be applied to therms sold or transported by the Company subject to the jurisdiction of the Commission as determined in accordance with the provision of this rate schedule. Such conservation charge shall be determined annually by the Company, separately for the Residential Heating, and Commercial/Industrial rate categories, subject to review and approval by the Commission as provided for in this rate schedule.
- Calculation of Conservation Charge: The Company will properly assign expenses for forecasted conservation expenditures to the applicable rate categories for a future twelve (12) month period commencing November 1 of each year. The total of such conservation expenditures plus any prior period reconciling adjustments shall be divided by therm sales as forecasted by the Company for the same annual period and rounded to the nearest hundredth of a cent. The

/s/Neil Proudman

EFFECTIVE: January 01, 2019

President

Neil Proudman 20 Jusan L. Fleck President TITLE:

Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Formatted: Font: 10.5 pt

Formatted: Indent: Left: 0.13"

Formatted: Font: (Default) Times New Roman, 10.5 pt

Formatted: Footer, Centered, Indent: Left: 0.13", Line spacing: single, Tab stops: 0.88", Left + 4.94", Right + 5.13", Left

Docket No. DG 21-130 Exhibit 26 Page 273 of 470 Docket No. DG 20-105

Exhibit 49

NHPUC NO. 11 GAS

Original Page 34

Formatted: Font: (Default) Times New Roman, 10.5 pt

Formatted: No underline, Underline color: Auto, Not

Formatted: Right, Indent: Left: 0.82", No bullets or

Formatted: No underline

Expanded by / Condensed by

numbering

LIBERTY UTILITIES

Terms and Conditions

resulting conservation charge shall be included in the Company's Local Distribution Adjustment Charge and applied to actual therms sold or transported for the following twelve (12) month period starting November-11, and ending October 31.

4.—Reporting: The Company shall submit annual reports to the Commission reconciling any difference between the actual conservation expenditures and actual revenues collected under this rate schedule. The difference whether positive or negative will be carried forward into the conservation charge for the next recovery period. Upon completion of the conservation program(s), any over or under collection may be credited or charged to the deferred Winter Period cost of gas account, subject to Commission approval.

4.

Effective Date: On or before the first business day in September of each year, the Company shall file with the NHPUC for its consideration and approval, the Company's request for a change in the CC applicable to each Rate Category during the next subsequent twelve-month period commencing with the calendar month of November.

ISSUED: November 16, 2018 ISSUED BY: /s/Susan L. Fleek Susan L. Fleck

EFFECTIVE: November 01, 2018 TITLE: President

Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048

NHPLIC NO. 10 GAS

Third Revised Page 34

LIBERTY UTILITIES Superseding Second Page 34

Formatted: Indent: Left: 0", Space Before: 0.45 pt, Tab stops: Not at 0.94" + 4.27"

Formatted: Font: 10.5 pt

Formatted: Indent: Left: 0.13"

Formatted: Font: (Default) Times New Roman, 10.5 pt

Formatted: Footer, Centered, Indent: Left: 0.13", Line spacing: single, Tab stops: 0.88", Left + 4.94", Right +

5.13", Left

/s/Neil Proudman

TITLE:

EFFECTIVE

Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Docket No. DG 21-130 Exhibit 26 Page 274 of 470 Docket No. DG 20-105 Exhibit 49

Second Revised Page 32 NHPUC NO. 10 GAS LIBERTY UTILITIES Superseding First Revised Page 32 1_ ISSUED: November 16, 2018 ISSUED BY: /s/Susan L. Fleek Susan L. Fleck President EFFECTIVE: November 01, 2018 Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048 Third Revised Page 34 NUDLIC NO. 10 GAS LIBERTY UTILITIES Superseding Second Page 34

Formatted: Left: 0.93", Right: 0.92", Top: 0.46", Bottom: 1.06", Header distance from edge: 0.5", Footer distance from edge: 0.35"

- 2. Reporting: The Company shall submit annual reports to the Commission reconciling any difference between the actual conservation expenditures and actual revenues collected under this rate schedule. The difference whether positive or negative will be carried forward into the conservation charge for the next recovery period. Upon completion of the conservation program(s), any over or under collection may be credited or charged to the deferred Winter Period cost of gas account, subject to Commission approval.
- Effective Date: On or before the first business day in September of each year, the Company shall
 file with the NHPUC for its consideration and approval, the Company's request for a change in the
 CC applicable to each Rate Category during the next subsequent twelve month period commencing
 with the calendar month of November.
- 4-6. Reconciliation Adjustment: Account 1163-1755 shall contain the cumulative difference between the sum of the DSM expenditures incurred by the Company plus the sum of the DSM repayments and the revenues collected from customers. The Company shall file the reconciliation along withthe COG filing on or before the first business day in September of each year.

B. D. Revenue Decoupling Adjustment Factor.

- 1. Purpose: The Company's RDAFRevenue decoupling eliminates the link between volumetric sales and Company revenue in order to align the interests of the Company and customers with respect to changing customer usage by establishing an allowed revenue per customer ("RPC"). The Company is allowed to collect that RPC for the number of actual customers it has in a given month. The difference of the Revenue Decoupling Adjustment Factor ("RDAF") is to establish procedures that allow the Company, subject to the jurisdiction of the NHPUC, to adjust, on an annual basis, its rates for firm gas sales and firm transportation in order to reconcile the difference between the Actual Base-Revenue collected per Customer with and the Benchmarked BaseAllowed Revenue per Customer. The Company's RDAF eliminates the link between volumetric sales and Company revenue in order to align the interests of the Company and eustennes with respect to changing eustennes usage. The purpose of the Normal Weather Adjustment ("NWA") is to adjust each customer's bill for the difference in delivery charges caused by the variation in actual Heating Degree Days ("HDDs") from normal HDDs during the Winter Period.
- Effective Date: The RDAF and NWA shall take effect beginning on November 1, 2018, and replace
 the Lost Revenue Adjustment Mechanism (LRAM) established in Order No. 25,932 (Docket No.
 DE 15-137).
- Applicability: The Revenue Decoupling Adjustment Factor and NWA shall apply to all of the Company's firm tariff rate schedules, subject to the jurisdiction of the Commissionexcluding special contracts, as determined in accordance with the provisions of this RDAF and NWA.
- 4. <u>Definitions</u>: The following definitions shall apply throughout the RDAF and NWA Section 179D:

4. Actual Base Revenue is the actual revenue derived from the Company's distribution rates for a DISSUEDD: XMàrck 2020019 ISSUEDDPY: //s/SuiskirducFlack

EFFECTIME: Alignenthe(021, 2018 TTTLEE: Precident

Autl<u>aurized i by a HFRHE Utrabur No N26.X8 X distribit diversiben (N20201 & incluse kata) Distributed in the content of the</u>

Formatted: Left, Indent: Hanging: 0.3", Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 6 + Alignment: Left + Aligned at: 0.63" + Indent at: 0.88", Tab stops: 0.88", Left + 2.19", Left + 2.31", Left + Not at 0.82"

Formatted: Left, Indent: Left: 0.32", No bullets or numbering

Formatted: No underline, Underline color: Auto

Formatted: Condensed by 0.2 pt

Formatted: Condensed by 0.2 pt

Commented [SC1]: Consider appropriate organization of the definitions RDAF vs NWA then alphabetize or prioritize.

Formatted: Left, Indent: Left: 0.57", Hanging: 0.25", Space Before: 0 pt, Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.61" + Indent at: 0.86", Tab stops: 0.82", Left

Formatted: Footer, Centered, Line spacing: single, Tab stops: 0.88", Left + 4.94", Right + 5.13", Left

Docket No. DG 21-130 Exhibit 26 Page 275 of 470 Docket No. DG 20-105 Exhibit 49

coupling Year for a Customer Class. The Company will use monthly distribution revenues adjusted to include weather normalized actuals and Actual Number of Custome determine the Monthly Actual Base Revenue per Customer,

Commented [SC2]: Addition from the October 4, 2019 issued 11/1 effective tariff filing

Actual Number of Customers is the actual number of Equivalent Bills for the applicable Customer Class Rate Class for the each applicable month of the Decoupling Year.

Formatted: Not Expanded by / Condensed by

Equivalent Bill. Customers are billed on different days of the month. To calculate the number of customers in a month for purposes of calculating the Monthly Actual Revenue it is necessary to use Equivalent Bills as a representation for customers. Equivalent Bills are calculated by dividing the number of days in the billing period of each customer's bill by

Formatted: Not Expanded by / Condensed by

Formatted: Indent: Left: 1.06"

Formatted: Condensed by 0.05 pt

Formatted: Underline

Billing Year is the 12-months commencing November 1 immediately following the completion of the Decoupling Year.

Formatted: Font: 10.5 pt, Condensed by 0.05 pt

Decoupling Year. The first Decoupling Year shall be the 10-month period from November 1,2018 to August 31, 2019. Each subsequent Decoupling Year shall be the twelve months Formatted: Body Text, Right, Indent: Left: 1.32", First line: 0", Right: 0.2", Space Before: 5.9 pt, Numbered + Level: 3 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1.25" + Indent at: 1.5", Tab stops: 1.32", Left

commencing September 1 through August 31.

Formatted: No underline, Underline color: Auto, Not Expanded by / Condensed by

<u>Customer ClassRate Class is the group of allare</u> customers taking service pursuant to the <u>same_rRate s</u>Schedules <u>combined as follows: Rates R-1 and R-5, except that for</u> revenue decoupling purposes rate schedule Rates R-3, R-4, R-6, and R-7R4, Rates G-41 and G-44, Rates G-42 and G-45, Rates G-43 and G-46, Rates G-51 and G-55, Rates G-52 and G-56, Rates G-53 and G-57, and Rates G-54 and G-58. shall be grouped with R3and R7 with R6.

Formatted: Font: (Default) Calibri Light, Condensed by

Formatted: Right, Indent: First line: 0", Space Before:

Formatted: No underline

Formatted: No underline

DIASTEDD: XMArch X02020019 ISSUEDBBY: /s/8/Sails Bridge Fiberk NSnils BurdudFileck EEFFECTIME: ANguenthe1001, 2018 TTTTEE:

Formatted: Footer, Centered, Line spacing: single, Tab stops: 0.88", Left + 4.94", Right + 5.13", Left

 $Aut\underline{\mathsf{Munived}} \underline{\mathsf{lby}}\underline{\mathsf{NHPWP}}\underline{\mathsf{UUrdowNo}}\underline{\mathsf{NS}}\underline{\mathsf{NSMtdd}}\underline{\mathsf{rbd}}\underline{\mathsf{veodow}}\underline{\mathsf{N20201}}\underline{\mathsf{8a}}\underline{\mathsf{indowe}}\underline{\mathsf{ktdNo}}\underline{\mathsf{D}}\underline{\mathsf{CD}}\underline{\mathsf{0}}\underline{\mathsf{109}}\underline{\mathsf{048}}$

Docket No. DG 21-130 Exhibit 26 Page 276 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 35 Terms and Conditions

25

NHPUC NO. 10 GAS Second Revised Page 35

e. LIBERTY UTILITIES Superseding First Revised Page 35

f. <u>Customer Class Group Customer Class Group (CG)</u> is the group of <u>rRate sSchedules</u> combined for purposes of calculating the Revenue Decoupling Adjustment and the <u>RDAF</u>. The two <u>Customer Class Group Customer Class Group</u>s are as follows:

Formatted: Font: (Default) Times New Roman, 10.5 pt

Formatted: Left, Indent: Hanging: 0.25", Right: 0.12", Line spacing: single, Numbered + Level: 1 + Numbering Style: a, b, c, ... + Start at: 4 + Alignment: Left + Aligned at: 1.07" + Indent at: 1.32", Tab stops: 1.32", Left + Not at 4.5"

Formatted: No underline

Formatted: Body Text, Left, Right: 0.27", Space Before: 3.7 pt, Numbered + Level: 1 + Numbering Style: a, b, c, ... + Start at: 4 + Alignment: Left + Aligned at: 1.07" + Indent at: 1.32", Tab stops: 1.31", Left + 1.5", Left + 1.63", Left + Not at 0.94" + 4.27"

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

EFFECTIVE: January 91, 2019

EFFECTIVE: August 1, 2021 TITLE: President

Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Formatted: Font: 10.5 pt

Formatted: Font: (Default) Times New Roman, 10.5 pt

Formatted: Footer, Centered, Line spacing: single, Tab stops: 0.88", Left + 4.94", Right + 5.13", Left

Docket No. DG 21-130 Exhibit 26 Page 277 of 470 Docket No. DG 20-105 Exhibit 49

	1 GAS			Original Page 36		Formatted: Font: (Default) Times New Ror	man, 10.5 pt
LIBERTY UTI	LITIES			Terms and Conditions			
	NHPUC NO LIBERTY U	TILITIES	Sup 55 Group Customer Class Grou	Second Revised Page 32 perseding First Revised Page 32 perseding First Revised Page 32	\	Formatted: Left: 0.93", Right: 0.92", Top: Bottom: 1.06", Header distance from edge distance from edge: 0.35"	
			Class Rate Class and Residen		<u> </u>	Formatted: Indent: Left: 0.94"	
	residential ra		l customers taking service pu ill include customers taking s			Formatted: Left, Indent: Left: 0.94", Right: Before: 3.7 pt, No bullets or numbering, T 0.88", Left	
	-				\	Formatted: No underline	
	(CG2): shall general servi	consist of all custome	ustrial Customer Class Group rs taking service pursuant to o 1, G-42, G-43, G44, G-45, C	ne of the Company's			
			From the Company's firm sales		•	Formatted: No underline, Underline color	: Auto
	vice customers m Cost of Gas		her revenue, special contracts,	and all revenues recovered		Formatted: Indent: Left: 0.69"	
f.	ir cost or Gus	11111,0.			•	Formatted	[
						Formatted: Left, Indent: Left: 0.69", No bi numbering, Tab stops: 0.88", Left + Not at	
<u>A</u>	August 31, 20		ar shall be the 10 month peri Decoupling Year shall be th	ed from November 1, 2018 to be twelve months commencing		Formatted: Indent: Hanging: 0.56", Tab st Left	ops: 0.44",
h. <u>Equivak</u>	ent Bill, The 1	number of days in the b	illing period of each customer			Formatted: Indent: Left: 0.26", Hanging: (stops: 0.44", Left + Not at 0.82"	0.56", Tab
			difference between actual dis			Formatted: Condensed by 0.05 pt	
ea aa	ach customer nd what distri	in each billing cycle to bution revenue for eacl	r each month of portion theret r customer's bill would have t	oeen based on weather			
St	ormanzed the ubtracted fron Levenue Per C	a each customer's bill a	me period. The resulting charge the time the bill is rendered.	ge or credit will be added to or i.e., "real time"), Calculation of		Formatted	(
<u>a.</u>	Definitions:						
	1. ln1	hal Approved Revenue	is the Distribution Revenue == ocated through rate design to	or each Customer ClassRate			
	Cla	e case, to determine the	revenues, all as approved by C				
	ii. Inc	remental Approved Revenue that is approved	venue is the amount of any ch by Order of the Commission by approved Incremental RPC	between rate cases and used to	•	Formatted: Left, Numbered + Level: 3 + N Style: i, ii, iii, + Start at: 1 + Alignment: R	
				*		Aligned at: 1.39" + Indent at: 1.51"	
			mer ("Initial RPC") is calculated by Order of the Commission			Formatted	
		-	Customer ("Incremental RPC"			Formatted	[
	Ap	proved Revenue for ea	ch Customer ClassRate Class	based on the monthly		Formatted	(
			Customer Class Rate Class for t	he test period associated with PC is approved by Order of the			
			ing other than a rate case that i		-////	Formatted: Font: 10.5 pt	
			ution rates. Such proceedings		_// //	Formatted: Font: (Default) Times New Ror	man, 10.5 pt
NATED.		-	ts, property tax reconciliation,		_////	Formatted: Footer, Centered, Line spacing stops: 0.88", Left + 4.94", Right + 5.13", L	
	XXX XX, 20:			Y: /s/Neil Proudman Susan L. Fleck Neil Proudman	_/		
EFFECTIVE: A			TITI	E: President			
	Authorized by	NHPLIC Order No. 26 187	dated November 02, 2018, in Docl	ret No. DG 17-048	/		
			XX dated XXX XX, 2021, in				

Docket No. DG 21-130 Exhibit 26 Docket No. DG 20-105

Page 278 of 470 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

each -month.

Original Page 36

Terms and Conditions

Formatted: Font: (Default) Times New Roman, 10.5 pt

Approved Revenue per Customer ("Approved RPC") is the sum of the Initial RPC per Customer Class Plus each and all Incremental RPC's per Customer ClassRate Class, if any.

A separate RPC is calculated for each month of the decoupling year for each Customer ClassRate Class using an a Approved distribution r Revenue and set of Equivalent Bills perfor

The Initial RPC is set for each month by calculating the Initial Approved Revenue by Rate Class by month from the approved rate schedule rates by Rate Class for that month. The rate schedule rates adjusted for the MEP Premium are multiplied by the appropriate monthly billing units, Equivalent Bills for customer charges, therms for the blocked therm charges, to produce the Initial Approved Revenue by Rate Class by month. The monthly Initial Approved Revenue are then divided by the Equivalent Bills for that month to produce the Initial RPC's for each Rate Class. For the Initial RPC's, the Equivalent Bills and associated blocked therm billing units shall be adjusted by an End of Year Bill Adjustment that reflects the number of customers that received service at the end of the test year, and Customer Class at the rate case and approved by the Commission to be collected through two components: the customer charge and the per therm charges. The Initial RPC is calculated by dividing the Initial Approved Revenue into monthly allowed distribution revenue based on the Equivalent Bills approved by the Commission. The Equivalent Bills for the Initial RPC are proportional to the end of test ye customer counts for a given Customer Class. The projected therms used for the monthlyallowed distribution revenue calculation shall reflect the volumetric therms corresponding to the end of year customer count.

The Incremental RPC is set for each month by calculating the Incremental Approved Revenue by Rate Class by month from the approved rate schedule rates by Rate Class for that month. The rate schedule rates adjusted for the MEP Premium are multiplied by the appropriate monthly billing units, Equivalent Bills for customer charges, therms for the blocked therm charges, to produce the Incremental Approved Revenue by Rate Class by month. The monthly Formatted: No underline

Formatted: Left, Indent: Left: 1.51", No bullets or numbering

Formatted: Left, Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.76" + Indent at: 1.01"

Formatted: No underline

Formatted: No underline

Formatted: No underline

Commented [SC1]: Better word?

Formatted: No underline

Formatted: No underline

Formatted: Left, Indent: Left: 1.01", No bullets or

Formatted: Font: Not Bold, No underline

Formatted: No underline Formatted: No underline

Formatted: Body Text, Right: 0.08", Space Before: 6.05 pt, Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.76" + Indent at: 1.01", Tab stops: 0.82", Left + Not at 0.94" + 4 27"

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Formatted: Font: 10.5 pt

Formatted: Font: (Default) Times New Roman, 10.5 pt

Formatted: Footer, Centered, Line spacing: single, Tab stops: 0.88", Left + 4.94", Right + 5.13", Left

Docket No. DG 21-130 Exhibit 26 Page 279 of 470 Docket No. DG 20-105 Exhibit 49

Formatted: Left: 0.93", Right: 0.92", Top: 0.46",

LIBERTY UTILITIES Superseding First Revised Page 32 Bottom: 1.06", Header distance from edge: 0.5", Footer distance from edge: 0.35" <u>Initial Approved Revenue are then divided by the Equivalent Bills for that month to produce the</u> Formatted: Left, Indent: Left: 0.94", No bullets or Initial RPC's for each. For the Incremental RPC's, the Equivalent Bills and associated blocked numbering, Tab stops: 1", Left + Not at 0.82" therm billing units shall be those bills and units for the test period associated with the Incremental Approved Revenue. For each residential Rate Class the Incremental RPG Formatted: No underline include any increase to the customer charge, only to the volumetric therms rates. is calculated by dividing the Incremental Approved Revenue into monthly allowed distribution revenue based on Commented [IA1]: the Equivalent Bills approved by the Commission in a proceeding other than a rate case that Formatted results in a permanent adjustment to base rates. Equivalent Bills for the determination of Incremental RPCs shall use the calendar month Equivalent Bills for the same time period **Formatted** which the Incremental Approved Revenue is determined. For each residential Customer Class Formatted Formatted Formatted −6. Calculation of Normal Weather Adjustment **Formatted** Definitions. The Normal Weather Adjustment (NWA) for each bill is Formatted: No underline, Condensed by 0.05 pt NWA = DeliveryChargeActual × NWF where Delivery Charge Actual is the calculated delivery charge for actual delivered-Formatted: Condensed by 0.05 pt therms for the Formatted: Underline color: Auto rate schedule applicable to that bill or portion thereof during the Winter Period. Formatted Formatted: Underline color: Auto h.,Benchmark,Base,Revenue,per,Customer, is the monthly, allowed distribution. revenue per Equivalent Bill for a given Decoupling Year for a given Custome **Formatted** Class, reflecting the distribution revenue level and approved equivalent bills from Formatted: Underline color: Auto the Company's most recent rate case or other proceeding that results in an Formatted adjustment to base rates. Benchmark Base Revenue per Customer will be calculated for each month based on the distribution rates in effect at the start of Formatted: Underline color: Auto the Decoupling Year and the calculation will be revised for the remaining months of **Formatted** each Decoupling Year if there is a distribution rate change that occurs following the beginning month of each Decoupling Year, Formatted: Underline color: Auto Real-time normal weather adjustment is the difference between actual distribution **Formatted** revenue billed to each customer in each billing cycle for each month or portion Formatted: Condensed by 0.05 pt thereof during the Winter Period, and what distribution revenue for each Formatted: Condensed by 0.05 pt customer's bill would have been based on weather normalized therm deliveries for the same period. The resulting charge or credit will be added to or subtracted from Formatted: Condensed by 0.05 pt each customer's bill at the time the bill is rendered (i.e., "real time"). Formatted: Condensed by 0.05 pt Winter Period. The time period from November 1 of a given year through April 30 Formatted: Condensed by 0.05 pt of the following year inclusive. Formatted: Condensed by 0.05 pt Base Load Factor for each customer is the customer's most recent two-year average daily delivered therms for actual bills rendered for those billing periods that are Formatted: Condensed by 0.05 pt completely within the June 1 through August 31 calendar period excluding such Formatted: Condensed by 0.05 pt billing periods that are only partially within the June 1 to August 31 period. If a customer has less than two-year's billing history, then the customer's available Formatted: Condensed by 0.05 pt history for the months of June through August as defined above will be used to Formatted: Condensed by 0.05 pt calculate the average daily delivered therms; and if a customer has no billing Formatted: Condensed by 0.05 pt history for the months of June through August as defined above, then the average daily delivered therms for the calendar months of June through August for the rate Formatted: Underline color: Auto schedule under which the customer is served will be used. Formatted Base Usage for each bill is the current Base Load Factor times the number of days in Formatted: Underline color: Auto the billing period. **Formatted** DISSUEDD: XMXrcXX2020019 ISSUEDBBY: /s/8/Sails Bridge Fiberk NSnils PardudFileck EEFFECTIME: ANgustribe 2001, 2018 TTTTEE: Authorized by NHIRHE Orden No N26, 187, White this divember 02,0201, 80, in Block etc. ND DIG-175948

NHPLIC NO. 10 GAS

Docket No. DG 21-130 Exhibit 26 Page 280 of 470 Docket No. DG 20-105 Exhibit 49

v. Heating Usage for each bill is the difference between the actual delivered therms for that bill less the Base Usage for that bill. If the calculated Heating Usage is less than zero, then the Heating Usage for that bill is set equal to zero.

Formatted: Not Expanded by / Condensed by

- vi. Heating Degree Days (HDD) for each day is sixty-five (65) minus the average temperature in degrees Fahrenheit for that day. If the calculated HDD is less than zero, then the HDD for that day is set equal to zero.
- vii. Normal Heating Degree Days (Normal HDD) for each day is the thirty-year average HDD for that day.
- viii. Normal Weather Adjustment Slope (NWA Slope) for each bill is the Heating Usage divided by the sum of actual HDD during the billing period.
- ix. Normal Heating Usage for each bill is the NWA Slope times the sum of the Normal HDD for the billing period.
- x. Normal Usage for each bill is the sum of the Base Usage and the Normal Heating Usage.
- xi. Normal Weather Factor (NWF) for each bill is

i,

Formatted: Not Expanded by / Condensed by

Formatted: Indent: Left: 1.63"

DISSUED: XMarch X020019 ISSUEDBY: /s8/SubarduoFrieck

NSubarduoFrieck

EEFFECTIME: Angustribe4021, 2018 TITTEE: Proceeding

Autlantized by CNH RHE Out dardio N26, X8 X dated in dvendoux (120201 & include teta Nac 120-1109) 48

Formatted: Footer, Centered, Line spacing: single, Tab stops: 0.88", Left + 4.94", Right + 5.13", Left

Docket No. DG 21-130 Exhibit 26 Page 281 of 470 Docket No. DG 20-105 Exhibit 49

I. <u>Heating Degree Days</u> (HDD) for each day is sixty five (65) minus the average temperature in degrees Fahrenheit for that day. If the calculated HDD is less than zero, then the HDD for that day is set equal to zero.

 Normal Heating Degree Days (Normal HDD) for each day is the thirty year average HDD for that day: Formatted: Left, Indent: Left: 1.56", Hanging: 0.13", Numbered + Level: 1 + Numbering Style: i, ii, iii, ... + Start at: 1 + Alignment: Right + Aligned at: 0.82" + Indent at: 1.07"

Formatted: No underline

 DisSUBD:
 XMarcki 20,20019
 ISSUEDBY: /s8xiishinducFheek

 NsiishinducFheek

 EEFFKCTIME:
 ANgwentbetCOII, 2018
 TITTEE:
 PRresident

Auttaurized i by a HIRVE U Grammon 26. X8 X stated in a verification 2020 1 % in the ket on a DG-179 48

Formatted: Footer, Centered, Line spacing: single, Tab stops: 0.88", Left + 4.94", Right + 5.13", Left

Docket No. DG 21-130 Exhibit 26 Page 282 of 470 Docket No. DG 20-105 Exhibit 49

NIPUC NO. 10 GAS Third Revised. I Page 36 II LIBERTY UTILITIES Superceding Second Revised Page 36 III LIBERTY UTILITIES Pormated: Not Expanded by / Condensed by Condensed	NHPUC NO. 11 GAS		Original Page 37	
INTERPLY LITTLES Superceding Second Revised Page 36 ILLHERTY LITTLES Superceding Second Revised Page 36 ILLHERTY LITTLES Phormal Weather Adjustment Store (NWA Store) for each bill is the Heating Usage divided by the sum of extent HDD during the hilling period. Not st 107* Phormal Heating Library for each bill is the NWA Store times the sum of the Normal Heating Library (Normal Heating Library for each bill is the sum of the Base Usage and the Normal Heating Library (Normal Weather Faster (NWI) for each bill is NWF = DeliveryCharge Accust NWF = DeliveryCharge (NWI) for each bill is NWF = DeliveryCharge (NWI) for each bill is the sum of the Normal Heating Library (Normal Weather Faster (NWI) for each bill is NWF = DeliveryCharge (NWI) for each bill is the sum of the Normal Heating Library (NWI) for each bill is NWF = DeliveryCharge (NWI) for each bill is the Heating Underline Color Automatical (NWI) for each bill is the Heating Underline Color Automatical (NWI) for each bill is the Heating Underline Color Automatical (NWI) for each bill is the Heating Underline Color Automatical (NWI) for each bill is the Heating Underline Color Automatical (NWI) for each bill is the Heating Underline Color Automatical (NWI) for each bill is the Heating Underline Color Automatical (NWI) for each bill is the Heating Underline Color Automatical (NWI) for each bill is the NWI for each bill is the NWI for each bill is the				Formatted: Font: (Default) Times New Roman, 10.5
### Pormatted: Indent: Left: 155' Hanging: 0.13' ### Pormatted: Indent: Left: 156' Hanging: 0.13' ### Owner Advance Advanced Space (NWA Slope) for each bill is the Heating Usage divided by the same of actual HDD during the billing ported. ### Owner Advanced by Honor of the Normal HDD for the billing periods. ### Approad to 022' winder at 107', rail Nort of 1202' by More at 107', rail N		NHPUC NO. 10 GAS Third Revised		Formatted: Not Expanded by / Condensed by
OSS Space Before: S 9 pt. Numbered 4 Level: Numbering Syle 1 ii. iii. a Spart at 1 + Aligned Numbering Syle 1 iii. ii. a Spart at 1 + Aligned Numbering Syle 1 iii. ii. a Spart at 1 + Aligned Numbering Syle 1 iii. ii. a Spart at 1 + Aligned Numbering Syle 1 iii. ii. a Spart at 1 + Aligned Numbering Syle 1 iii. a Spart at 1 + Aligned Numbering Syle 1 iii. a Spart at 1 + Aligned Numbering Syle 1 iii. a Spart at 1 + Aligned Numbering Syle 1 iii. a Spart at 1 + Aligned Numbering Syle 1 iii. a Spart at 1 + Aligned Numbering Syle 1 iii. a Spart at 1 - Aligned Numbering Syle 1 iii. a Spart at 1 - Aligned Numbering Syle 1 iii. a Spart at 1 - Aligned Numbering Syle 1 iii. a Spart at 1 - Aligned Numbering Syle 1 iii. a Spart at 1 - Aligned Numbering Syle 1 iii. a Spart at 1 - Aligned Numbering Syle 1 iii. a Spart at 1 - Aligned Numbering Syle 1 iii. a Spart at 1 - Aligned Numbering Syle 1 iii. a Spart at 1 - Aligned	ii. LIBE I		perseding Second Revised Page 36	
HDD for the billing period.		p. Normal Weather Adjustment Slope (NWA Slope divided by the sum of actual HDD during the bi) for each bill is the Heating Usage Hing period;	Formatted: Indent: Left: 1.56", Hanging: 0.13", Rig 0.08", Space Before: 5.9 pt, Numbered + Level: 1 + Numbering Style: i, ii, iii, + Start at: 1 + Alignment Right + Aligned at: 0.82" + Indent at: 1.07", Tab sto Not at 1.07"
Expanded by / Condensed by 1. Normal Usage for each bill is the sum of the Bace Usage and the Normal Heating- Usages Formatted: No underline, Underline color: Auto Expanded by / Condensed by Summal Weather Factor (NWF) for each bill is NWF = DeliveryCharge_normal		q. Normal Heating Usage for each bill is the NWA HDD for the billing period.	Slope times the sum of the Normal	Formatted: No underline, Underline color: Auto, No Expanded by / Condensed by
Formatted: No underline, Underline color: Auto Expanded by / Condensed by NWF = DeliveryChargeNormal - 1 DeliveryChargeActual SSUED-December 31, 2018 ISSUED BY: Assum 1. Flock Formatted: Space Before: 0 pt Formatted: Indent: Left: 0', Space Before: 0 pt, stops: Not at 0.94" + 4.27" Formatted: Font: 10.5 pt Formatted: Font: 10.5 pt Formatted: Font: (Default) Times New Roman, 1. Flock NATED: XXX XX, 2021 ISSUED BY: Assum 1. Flock SSUED BY: Assum 1. Flock Formatted: Font: 10.5 pt Formatted: Font: (Default) Times New Roman, 1. Flock Formatted: Font: (Default) Times New Roman, 1. Flock Formatted: Font: 10.5 pt		_		Formatted: No underline, Underline color: Auto, N
Expanded by / Condensed by Section 1. Source Before: 0 pt Formatted: Space Before: 0 pt Formatted: Indent Left: 0", Space Before: 0 pt, stops: Not at 0.94" + 4.27" Formatted: Font: 10.5 pt Format		r. Normal Usage for each bill is the sum of the Bas	e Usage and the Normal Heating	
NWF = DeliveryCharge _{Normal} - 1 DeliveryCharge _{Actual} - 1 DeliveryCharge				Formatted: No underline, Underline color: Auto, N Expanded by / Condensed by
NWF = DeliveryCharge _{Normal} - 1 DeliveryCharge _{Actual} - 1 DeliveryCharge		_		
DeliveryChargeActual SSUED: December 31, 2018 ISSUED BY: \(\sqrt{\text{Susan L. Fleek}} \) Formatted: Space Before: 0 pt Formatted: Indent: Left: 0", Space Before: 0 pt, stops: Not at 0.94" + 4.27" Formatted: Font: 10.5 pt Formatted: Font: 10.5 pt Formatted: Font: 10.5 pt Formatted: Font: 10.5 pt Formatted: Font: (Default) Times New Roman, 1		s. <u>Normal Weather Factor</u> (NWF) for each bill is		
DeliveryChargeActual SSUED December 31, 2018 ISSUED BY: /s/Susan L. Fleek Formatted: Space Before: 0 pt Formatted: Indent: Left: 0', Space Before: 0 pt, stops: Not at 0.94" + 4.27" Formatted: Font: 10.5 pt Formatted: Font		DeliveryCharge _{Normal} _ 1		
Formatted: Space Before: 0 pt Formatted: Indent: Left: 0", Space Before: 0 pt, stops: Not at 0.94" + 4.27" Formatted: Indent: Left: 0", Space Before: 0 pt, stops: Not at 0.94" + 4.27" Formatted: Font: 10.5 pt Formatted: Font: 10.5 pt Formatted: Font: 10.5 pt Formatted: Font: (Default) Times New Roman, 1 Formatted: Font: (Default) Times New Roman				
Formatted: Indent: Left: 0", Space Before: 0 pt, stops: Not at 0.94" + 4.27" Formatted: Font: 10.5 pt Formatted: Font: 10.5 pt Formatted: Font: (0.5 pt)		Delivery Ghar geactual		
Stops: Not at 0.94" + 4.27" Formatted: Font: 10.5 pt Formatted: Font: (Default) Times New Roman, 3.1 points of the specing single stops: 0.88", Left + 4.94", Right + 5.13", Left Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DB 17-048	SSUED: December 31, 201	8 ISSUED BY: /s/Susan L. Fleck	•	
Formatted: Font: (Default) Times New Roman, 1 PATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman STREETIVE: January 91, 2010 Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048				Formatted: Indent: Left: 0", Space Before: 0 pt, Tal stops: Not at 0.94" + 4.27"
Formatted: Font: (Default) Times New Roman, 1 DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman EFFECTIVE: January 91, 2019 Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048				
Formatted: Font: (Default) Times New Roman, 1 PATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman STREETIVE: January 61, 2010 Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048				
Formatted: Font: (Default) Times New Roman, 1 PATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman STREETIVE: January 61, 2010 Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048				
Formatted: Font: (Default) Times New Roman, 1 PATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman STREETIVE: January 61, 2010 Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048				
Formatted: Font: (Default) Times New Roman, 1 PATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman STREETIVE: January 61, 2010 Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048				
Formatted: Font: (Default) Times New Roman, 1 PATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman STREETIVE: January 61, 2010 Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048 Formatted: Font: (Default) Times New Roman, 1 F				
Formatted: Font: (Default) Times New Roman, 1 PATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman STETECTIVE: January 91, 2019 Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048 Formatted: Font: (Default) Times New Roman, 1				
Formatted: Font: (Default) Times New Roman, 1 PATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman STETECTIVE: January 91, 2019 Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048 Formatted: Font: (Default) Times New Roman, 1				
Formatted: Font: (Default) Times New Roman, 1 PATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman STETECTIVE: January 91, 2019 Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048 Formatted: Font: (Default) Times New Roman, 1				
Formatted: Font: (Default) Times New Roman, 1 PATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman STREETIVE: January 61, 2010 Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048 Formatted: Font: (Default) Times New Roman, 1 F				
DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman EFFECTIVE: January 91, 2019 Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048 Formatted: Footer, Centered, Line spacing: sing stops: 0.88", Left + 4.94", Right + 5.13", Left Formatted: Footer, Centered, Line spacing: sing stops: 0.88", Left + 4.94", Right + 5.13", Left ### Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048				Formatted: Font: 10.5 pt
Stops: 0.88", Left + 4.94", Right + 5.13", Left DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman FFFECTIVE: January 91, 2019 FFFECTIVE: August 1, 2021 Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048				Formatted: Font: (Default) Times New Roman, 10.5
DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman FIFECTIVE: January 01: 2019 FIFECTIVE: August 1, 2021 Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048			/	Formatted: Footer, Centered, Line spacing: single, stops: 0.88". Left + 4.94". Right + 5.13". Left
Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048	DATED: XXX XX	, 2021 ISSUE	D BY: /s/Neil Proudman	# (212-ps) 5,555 / 2514 : 1,547 / (tagine : 5,155 / 2514
Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048	EFFECTIVE: January 9 EFFECTIVE: August 1	1, 2019 , 2021	TITLE: Neil Proudinan TITLE: President	/
	Authorize	d by NHPUC Order No. 26,187 dated November 02, 2018, in	Docket No. DG 17-048	

Docket No. DG 21-130 Exhibit 26 Page 283 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10 GAS Second Revised Page 32

LIBERTY UTILITIES

Superseding First Revised Page 32

where <u>Delivery Charge Normal</u> is the calculated delivery charge for Normal Usage for the

rate schedule applicable to that bill or portion thereof during the Winter Period and Delivery Charge Actual is the calculated delivery charge for actual delivered therms for the rate schedule applicable to that bill or portion thereof during the Winter Period.

The Normal Weather Adjustment (NWA) for each bill is

 $NWA = DeliveryCharge_{Actual} \times NWF$

where Delivery Charge Actual is the calculated delivery charge for actual delivered therms for the rate schedule applicable to that bill or portion thereof during the Winter Period.

7. Application of the NWA to Customer Bills

The NWA charge or credit will be separately stated, and added to or subtracted from each bill as applicable. Each bill will have a separate line titled "Normal Weather Adj.," which line will include the total variable distribution charges, the NWF percentage, and the resulting charge or credit.

8. Calculation of Revenue Decoupling Adjustment and Reconciliation

. Definitions

- i. Monthly Actual Distribution Revenue is the sum of the monthly actual billed dDistribution rRevenue and the NWA revenue less the MEP Premium for a given Decoupling Yearthat month. Of the Actual Distribution Revenue, each month the Monthly Allowed Revenue is booked as income and the remaining surplus or deficit is recorded as a corresponding asset or liability on the balance sheet to be collected from or returned to customers through the RDAF.
- ii. Monthly Allowed Revenue is the Approved RPC per Customer Class Rate Class for the applicable month multiplied by the Actual Number of Customers Equivalent Bills for that month. The Monthly Allowed Revenue is booked as incomer.
- iii. Customer Class Group Customer Class Group Forecasted Throughput Volume is the forecasted firm tariff throughput for a given Customer Group for the Billing Year.
- iv. of Revenue Decoupling Accounts ("RDA Accounts") are the accounts established exist on the balance sheet for the purpose of recording the Revenue Decoupling Adjustment for each Customer Class Group Customer Class Group and are used to determine the RDAF reconciliation amount.
- b. Description of Revenue Decoupling Adjustment and Reconciliation

8.

Formatted: Left: 0.93", Right: 0.92", Top: 0.46", Bottom: 1.06", Header distance from edge: 0.5", Footer distance from edge: 0.35"

Formatted: Indent: Left: 1.5"

Formatted: Indent: Left: 1.5", Right: 0", Space Before: 0 pt, Line spacing: Exactly 10.75 pt

Formatted: Underline

Formatted: Left, Indent: Left: 0.75", No bullets or numbering

Formatted: No underline, Underline color: Auto, Not Expanded by / Condensed by

Formatted: Indent: Left: 0.82", No bullets or numbering

Formatted: No underline, Underline color: Auto, Not Expanded by / Condensed by

Formatted: Right: 0", Space Before: 0 pt, Numbered + Level: 3 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1.25" + Indent at: 1.5"

Formatted: Indent: Left: 1.5", Hanging: 0.19", Right: 0", Space Before: 0 pt, Numbered + Level: 4 + Numbering Style: i, ii, iii, ... + Start at: 1 + Alignment: Right + Aligned at: 0.61" + Indent at: 0.86", Tab stops: 1.81", Left + Not at 0.82"

Formatted: Indent: Left: 1.5", Hanging: 0.19", Right: 0", Space Before: 0 pt, Numbered + Level: 4 + Numbering Style: i, ii, iii, ... + Start at: 1 + Alignment: Right + Aligned at: 0.61" + Indent at: 0.86", Tab stops: 1.69", Left + Not at 0.82"

Formatted: Not Expanded by / Condensed by

Commented [IA1]:

Commented [IA2]: Not used - delete

Formatted: Underline

Formatted: Not Expanded by / Condensed by

Commented [IA3]: Through?

Formatted: Not Expanded by / Condensed by

Formatted: Not Expanded by / Condensed by

Formatted: Indent: Left: 1.25", Right: 0.08", Space Before: 6.05 pt, No bullets or numbering

Formatted: Not Expanded by / Condensed by

 ISSUED:
 March 20, 2019
 ISSUED BY:
 ½/Susan L. Fleck

 EFFECTIVE:
 November 01, 2018
 TITLE:
 President

Docket No. DG 21-130 Exhibit 26 Page 284 of 470 Docket No. DG 20-105 Exhibit 49

Description of Revenue Decoupling Adjustment and Reconciliation

Each month the Company will record a Revenue Decoupling Adjustment in the RDA Accounts in accordance with generally accepted accounting principles. The Revenue Decoupling Adjustment is the difference between the Actual Revenue and the Allowed Revenue is collected in RDA Accounts Monthly Allowed Revenue and the Monthly Actual Distribution Revenue, In addition, The reconciliation amounts collected or distributed through the RDAF are also deposited recorded into the RDA Accounts for each Customer Class Group Customer Class Group. The RDA Accounts accrue interest on the average monthly balance using the prime lending rate. At the conclusion of each Decoupling Year, the Company shall calculate a sum of the balance in each of the RDA Accounts for each Customer Class Group Customer Class Group Decoupling Revenue Adjustment to bejsshall be used to determine the RDAF for the next Billing Year, effective Nevember 1.

Formatted: Normal, Indent: Left: 1.25", Space Before: 0 pt, No bullets or numbering, Tab stops: Not at 1.07"

Formatted: Indent: Left: 1.25"

The Revenue Decoupling Adjustment shall be determined by calculating the monthly, difference between the Benchmark Base Revenue per Customer times the actual number of Equivalent Bills for the applieable Customer Class and the Actual Base Revenue for that month. The RDAF volumetrie rate per therm to be applied to customers' bills in the Billing Year is the balance in the RDA Accounts at the end of the Decoupling Year-sum of these monthly Revenue Decoupling Adjustments for each Customer Class Group Customer Class Group in the Decoupling Year shall be divided by the Forecasted Throughput Volume for that Customer Class Group Customer Class Group, forecasted-Billing Year sales to derive The volumetric rate per therm to be applied to customers' bills in the Billing Year. The Revenue Decoupling Adjustment shall also include a reconciliation—component for the previous Decoupling Year, which represents the difference between the accrued decoupling amount in the Decoupling Year compared to the actual revenues billed in the billing Year.

Formatted: Condensed by 0.05 pt

Formatted: Condensed by 0.05 pt

Formatted: Condensed by 0.05 pt

Formatted: Indent: Left: 1.25"

Formatted: Condensed by 0.05 pt

Formatted: Condensed by 0.05 pt **Formatted:** Condensed by 0.05 pt

Formatted: Condensed by 0.05 pt

Revenue Decoupling Adjustment Formulas

RDAF_{CG} = RDA Account Balance_{CG} end of Decoupling Year (August 31) divided by FTV_{CG}

RDA Account Balance_{CG} = Sum of MRDA_{CG} plus RDAF Reconciliation plus interest

MRDA_{CG} = Monthly Allowed Revenue_{CG} less Monthly Actual Revenue_{CG} (including NWA)

Monthly Allowed Revenue_{CG} - Monthly Approved RPC_{CG} times Actual Customers_{CG}

Formatted: Not Expanded by / Condensed by

Formatted: Right, Indent: Left: 1.07", No bullets or

numbering

Formatted: Indent: Left: 1.07", No bullets or

numbering

Formatted: Not Superscript/ Subscript

Formatted: Not Expanded by / Condensed by

Formatted: Right, Indent: Left: 1.07", No bullets or

numbering

 $\mathrm{RD}_{\mathrm{GG}} = \sum_{\mathrm{RG}=1}^{\mathrm{RG}=n} \left[\left(\mathrm{BRPG}_{--}^{--} \times \mathrm{ACUSTS}_{--}^{--} \right) - \mathrm{AR}_{--}^{--} \right]$

And

 $RDAF_{CC} = \frac{RD_{CC} + CGDEF_{T-1}}{RDAF_{CC}}$

ISSUED: March 20, 2019 ISSUED BY: \(\frac{\sqrt{ss\sqrt{sh\shint{sh\sqrt{sh\sqrt{sh\sqrt{sh\sqrt{sh\sqrt{sh\sqrt{sh\sqrt{sh\shan\sqrt{sh\sqrt{sh\sqrt{sh\sqrt{sh\sqrt{sh\sqrt{sh\sqrt{sh\sqrt

Docket No. DG 21-130 Exhibit 26 Page 285 of 470 Docket No. DG 20-105 Exhibit 49

Docket No. DG 21-130 Exhibit 26 Page 286 of 470 Docket No. DG 20-105 Exhibit 49

 ISSUED:
 March 20, 2019
 ISSUED BY:
 ½/Susan L. Fleck Susan L. Fleck FrecTiVE:

 EFFECTIVE:
 November 01, 2018
 TITLE:
 President

 $Authorized \ by \ NHPUC \ Order \ No.\ 26,187 \ dated \ November\ 02,2018, in \ Docket \ No.\ DG\ 17-048$

Docket No. DG 21-130 Exhibit 26 Page 287 of 470 Docket No. DG 20-105 Exhibit 49

Formatted: Normal, Indent: Left: 1.07", Space Before: 0 pt, Line spacing: Exactly 50.55 pt

Where the terms in the above equation have the following meanings:

Docket No. DG 21-130 Exhibit 26 Page 288 of 470 Docket No. DG 20-105 Exhibit 49

LIBERTY UTILITIES Superseding Second Revised Page 37 The Actual Number of Equivalent Bills for the applicable Customer Class for the ACUSTS_{T-1} most recently completed Decoupling Year (T-1) The Actual Base Revenue for the applicable Customer Class for the most recently completed Decoupling Year, (T-1), as defined in Section 4(D). For purposes of AR_{T-1} calculating the Actual Base Revenue, base revenues for Low Income rate class R nined based on non-discounted rate R-2 The Benchmark Base Revenue Per Equivalent Bill for the applicable Customercompleted Decoupling Year, stated on a monthly basis (T-1). $\frac{BRPC_{T-1}}{T}$ CG stomer Class Groups as defined in Section 4(D). The balance of the unrecovered deferrals inclusive of associated interest using the **CGDEF** Forecast Throughput Volumes inclusive of all firm tariff throughput for the FTV_{CG} Billing Year. RC_ Rate classes in a Customer Class Group. RD_{CG} The Revenue Decoupling adjustment to revenues, representing the sum of the athly Revenue Decoupling Adjustments in the Decoupling Year. RDAF The Revenue Decoupling Adjustment Factor for the Billing Year.

Formatted: Font: 10.5 pt, Subscript, Expanded by 0.05

Formatted Table

pt, Raised by 2.5 pt

9. Calculation of the Revenue Decoupling Adjustment Reconciliation

Account 1168-1822 shall contain the accumulated difference between annual revenues and the Revenue Decoupling Adjustment, as calculated by multiplying the RDAF times firm sales and transportation throughput, and the Revenue Decoupling Adjustment allowed revenues annually, plus carrying charges on the average monthly balance using the prime lending rate.

10.9. Application of the RDAF to Customer Bills

NHPUC NO. 10 GAS

The RDAF (\$ per therm) shall be calculated annually for each Customer Group and shall be truncated at the nearest one one hundredth of a centfour decimal points per therm. The annual calculated Customer Group RDAF will be applied to the monthly firm tariff throughput for each customer in that particular Customer Group, effective November 1 of the given year. Calculation of Normal Weather Adjustment

The Normal Weather Adjustment (NWA) for each bill is

NWA = DeliveryCharge_{Actual} × NWF

where Delivery Charge Actual is the calculated delivery charge for actual delivered therms for the rate schedule applicable to that bill or portion thereof during the Winter Period.

Application of the NWA to Customer Bills

Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048

Formatted: Body Text, Space Before: 0 pt

Docket No. DG 21-130 Exhibit 26 Page 289 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 38
Terms and Conditions

Formatted: Font: (Default) Times New Roman, 10.5 pt

NHPUC NO. 10 GAS Second Revised Page 38
LIBERTY UTILITIES Superseding First Page 38

The NWA charge or credit will be separately stated, and added to or subtracted from each bill as applicable. Each bill will have a separate line titled "Normal Weather Adj.," which line will include the total variable distribution charges the NWE research or and the resulting charge or anglit.

Formatted: Body Text, Indent: Left: 0", Space Before: 0 pt, Tab stops: 0.82", Left + Not at 0.94" + 4.27"

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Pro

Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Formatted: Font: 10.5 pt

Formatted: Font: (Default) Times New Roman, 10.5 pt

Formatted: Footer, Centered, Line spacing: single, Tab stops: 0.88", Left + 4.94", Right + 5.13", Left

Docket No. DG 21-130 Exhibit 26 Page 290 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS

LIBERTY UTILITIES

Terms and Conditions

NHPUC NO. 10 GAS

LIBERTY UTILITIES

Superseding First Revised Page 32

Superseding First Revised Page 32

The winter cost of gas rate will be applied to billings commencing with the first November revenue billing eyele; the summer cost of gas rate will be applied to billings commencing with the first May revenue billing eyele.

C. Calculation. The amount of the cost of gas rate is the anticipated unit cost of gas sold.

At the conclusion of each winter and summer period the Company will calculate the extent that cost of gas revenues are greater or less than actual unit costs of gas compared with the anticipated unit costs. The calculated difference (actual gas sales volumes multiplied by the difference between actual and anticipated unit costs) will be carried forward into the computation of the cost of gas rate for the corresponding winter or summer period.

Any excess revenue collected, as determined above, will earn interest as specified by the Commission.

- D. <u>Changes</u>. The cost of gas rate may be adjusted without further Commission action based on the projected over /under collection of gas costs, the adjusted rate to be effective the first of the month. Any such rate adjustments may not exceed a maximum rate of 25 percent above the approved rate, but there is no limit on the amount of any rate reductions.
- E. <u>Refunds</u>. When refunds are made to the Company by its suppliers that are applicable to increased charges collected under this provision, the Company will make appropriate refunds to its customers and as the Commission may direct.
- F. Reporting. The Company shall submit to the Commission, at least 30 days prior to the effective date, the proposed winter and summer period cost of gas rate computation. Any monthly adjustments to the cost of gas rate must be filed five (5) business days prior to the first day of the subsequent month (the effective date of the new rate).

The cost of gas rate shall be computed to the nearest one hundredth cent per therm and shown on customers' bills.

G. Fixed Price Option Program. An alternative to the traditional winter period cost of gas rate mechanism may be elected by the customer pursuant to the Company's Fixed Price Option (FPO) Program. The Company may offer up to 50% of its expected firm sales for the winter period under the FPO Program. The cost of gas charge offered under the FPO Program will remain fixed for all winter period billings commencing November 1 and ending April 30 of the effective winter period. Once elected, customers must remain on the FPO Program for the duration of the winter period unless service is terminated. There are no maximum or minimum usage levels. Customers may enroll in this Program by contacting the Company between the October 1 and October 19 period immediately preceding the effective winter period.

18 LOCAL DISTRIBUTION ADJUSTMENT CLAUSE AND NORMAL WEATHER-ADJUSTMENT

E. <u>Purpose</u>. The purpose of the Local Distribution Adjustment Clause ("LDAC" or this "Clause") is toestablish procedures that allow the Company, subject to the jurisdiction of the NHPUC, to adjust, on an annual basis, its delivery charges in order to recover Conservation Charges ("CC"), Revenue—Decoupling Adjustment Factor ("RDAF"), Winter Period Surcharges ("WPS"), Environmental—Surcharges ("ES") including the Relief Holder Surcharge ("RHS") and the Manufactured Gas—

 DATED:
 XXX XX, 2021
 ISSUED BY: /s/Neil Proudman

 Neil Proudman

 EFFECTIVE:
 August 1, 2021
 TITLE: President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Docket No. DG 21-130 Exhibit 26 Page 291 of 470 Docket No. DG 20-105 Exhibit 49

<u>Program Surcharge ("MGP"), rate case expenses and recoupment ("RCE"), Residential LowGas-Assistance Program costs ("GAP")</u>

DISSUBD: XMX/ch/X0/2019 ISSUEDBY: /s/Suis includitect
Six is includitect
EFFECTIVE: ANovember 201, 2018
TITTEE: President

 $\underline{\text{Aut}|\textbf{Aut}|\textbf{Aut}|\textbf{betibe}|\textbf{ChyPUPC}|\textbf{Of}|\textbf{Out}|\textbf{Chi}|\textbf{O}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf{Chi}|\textbf$

Docket No. DG 21-130 Exhibit 26 Page 292 of 470 Docket No. DG 20-105

Exhibit 49

NHPUC NO. 11 GAS Original Page 34 LIBERTY UTILITIES

NUDLIC NO. 10 CAS First Revised Page 33 LIBERTY UTILITIES Superseding Original Page 33

Income Assistance Program costs ("RLIAP") and any other expenses the NHPUC may approve from time to time. The purpose of the Normal Weather Adjustment ("NWA") is to establish procedures that allow the Company, subject to the jurisdiction of NHPUC, to calculate and apply, for each customer on a monthly basis, the Normal Weather Factor ("NWF").

Applicability. This Clause shall be applicable in whole or part to all of the Company's firm sales service and firm delivery service customers as shown on the table below. The application of this clause may, for od cause shown, be modified by the NHPUC. See Section 17(K) "Other Rules.""

Applicability	CC 17(C)	RDAF 17(D)	ES- 17(E)	RCE 17(F)	RLIAP 17(G)
Residential Non-Space Heating R 1, R 5	4	4	X	4	X
Residential Space Heating R 3, R 4, R 6, R 7	4	4	X	4	X
Small C&I G 41, G 51, G 44, G 55	1	4	X	4	X
Medium C&I G 42, G 52, G 45, G 56	4	4	X	4	X
Large C&I G-43, G-53, G- 54, G-46, G-57, G-58	4	4	X	4	X

Not applicable X This Clause is a Applicable to all rates and charges on a Customer ClassRate Class basis. The application of the CC, RDAF, ES, RCE and GAP is applicable to all therms and therefore the application to all customers, including Managed Expansion Program Customers, is at the ne rate per therm as the corresponding non Managed Expansion Program Customers.

Applicable to Non-Managed Expansion Program Customers

Conservation Charges Allowable for LDAC.

- Purpose: The purpose of this provision is to establish a procedure that allows the Company, subject to the jurisdiction of the NHPUC, to adjust, on an annual basis, the Conservation Charge, if and when applicable, to firm sales service and firm delivery service throughput in order to recover from firm customers costs and lost margins associated with its energy efficiency management programs.
- Applicability: A conservation charge shall be applied to therms sold or transported by the Company subject to the jurisdiction of the Commission as determined in accordance with the provision of this rate schedule. Such conservation charge shall be determined annually by the Company, separately for the Residential Heating, and Commercial/Industrial rate categories, subject to review and approval by the Commission as provided for in this rate schedule.
- Calculation of Conservation Charge: The Company will properly assign expenses for forecasted conservation expenditures to the applicable rate categories for a future twelve (12) month periodncing November 1 of each year. The total of such conservation expenditures plus any prior period reconciling adjustments shall be divided by therm sales as forecasted by the

DATED: ISSUED BY: /s/Neil Proudman XXX XX, 2021 Neil Proudman EFFECTIVE: August 1, 2021 President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Docket No. DG 21-130 Exhibit 26 Page 293 of 470 Docket No. DG 20-105 Exhibit 49

Company for the same annual period and rounded to the nearest hundredth of a cent. The sulting conservation charge shall be included in the Company's Local Distribution Adjustme Charge and applied to actual therms sold or transported for the following twelve (12) monthperiod starting November 1, and ending October 31.

Reporting: The Company shall submit annual reports to the Commission reconciling any difference between the actual conservation expenditures and actual revenues collected under this rate schedule. The difference whether positive or negative will be carried forward into the conservation charge for the next recovery period. Upon completion of the conservation program(s), any over or under collection may be credited or charged to the deferred Winter Period cost of gas account, subject to Commission approval.

Effective Date: On or before the first business day in September of each year, the Company shall file with the NHPUC for its consideration and approval, the Company's request for a change in the CC applicable to each Rate Category during the next subsequent twelve month period commencing with the calendar month of November.

ISSUED: November 16, 2018 ISSUED BY: /s/Susan L. Fleek Susan L. Fleck

EFFECTIVE: November 01, 2018 TITLE: President

Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman Neil Proudman EFFECTIVE: August 1, 2021 TITLE:

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Docket No. DG 21-130 Exhibit 26 Page 294 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10 GAS LIBERTY UTILITIES Third Revised Page 34
Superseding Second Page 34

24. Reporting: The Company shall submit annual reports to the Commission reconciling any difference between the actual conservation expenditures and actual revenues collected under this rate schedule. The difference whether positive or negative will be carried forward into the conservation charge for the next recovery period. Upon completion of the conservation program(s), any over or under collection may be credited or charged to the deferred Winter Period cost of gas account, subject to Commission approval.

- 26. Effective Date: On or before the first business day in September of each year, the Company shall file with the NHPUC for its consideration and approval, the Company's request for a change in the CC applicable to each Rate Category during the next subsequent twelve-month period commencing with the calendar month of Nevember.
- 28. Reconciliation Adjustment: Account 1163-1755 shall contain the cumulative difference between the sum of the DSM expenditures incurred by the Company plus the sum of the DSM repayments and the revenues collected from customers. The Company shall file the reconciliation along with the COG filing on or before the first business day in September of each year.

H. Revenue Decoupling Adjustment Factor.

- 2. Purpose: The Company's RDAFRevenue decoupling eliminates the link between volumetric sales and Company revenue in order to align the interests of the Company and customers with respect to changing customer usage by establishing an allowed revenue per customer ("RPC"). The Company is allowed to collect that RPC for the number of actual customers it has in a given month. The difference The purpose of the Revenue Decoupling Adjustment Factor ("RDAF") is to establish procedures that allow the Company, subject to the jurisdiction of the NHPUC, to adjust, on an annual basis, its rates for firm gas sales and firm transportation in order to reconcile the difference between the Actual Base Revenue collected per Customer with and the Benchmarked BaseAllowed Revenue per Customer. The Company's RDAF eliminates the link between volumetric sales and Company revenue in order to align the interests of the Company and customers with respect to changing customer usage. The purpose of the Normal Weather Adjustment ("NWA") is to adjust each customer's bill for the difference in delivery charges caused by the variation in actual Heating Degree Days ("HDDs") from normal HDDs during the Winter Period.
- Effective Date: The RDAF and NWA shall take effect beginning on November 1, 2018, and replace the Lost Revenue Adjustment Mechanism (LRAM) established in Order No. 25,932 (Docket No. DE 15-137).
- Applicability: The Revenue Decoupling Adjustment Factor and NWA shall apply to all of the Company's firm tariff rate schedules, subject to the jurisdiction of the Commissionexcluding special contracts, as determined in accordance with the provisions of this RDAF and NWA.
- 8. <u>Definitions</u>: The following definitions shall apply throughout the RDAF and NWASection 17D:

Actual Base Revenue is the actual revenue derived from the Company's distribution rates for a given Decoupling Year for a Customer Class. The Company will use monthly distribution revenues adjusted to include weather normalized actuals and Actual Number of Customers to determine the Monthly Actual Base Revenue per Customers

Actual Number of Customers is the actual number of Equivalent Bills for the applicable Customer ClassRate Class for the each applicable month of the Decoupling Year.

Equivalent Bill. Customers are billed on different days of the month. To calculate the number of customers in a month for purposes of calculating the Monthly Actual Revenue it

 DLSSUEDD:
 XMárcki x020019
 ISSSUEDBY: //s/SuisáirducFieck

 NSúisáirducFieck

 EEFFECTIME:
 ANgoueribe/001, 2018
 TTTTEE: PRosident

Authorized by NHRHE Orden No N26.187. What edind ventoux (2020] 80 in Block etc. ND DIG-179.48

Formatted: Left: 0.93", Right: 0.92", Top: 0.46", Bottom: 1.06", Header distance from edge: 0.5", Footer distance from edge: 0.35"

Formatted: Condensed by 0.2 pt

Formatted: Condensed by 0.2 pt

Commented [SC1]: Consider appropriate organization of the definitions RDAF vs NWA then alphabetize or prioritize.

Formatted: Left, Indent: Left: 0.57", Hanging: 0.25", Space Before: 0 pt, Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.56" + Indent at: 0.81", Tab stops: 0.82", Left

Commented [SC2]: Addition from the October 4, 2019 issued 11/1 effective tariff filing

Formatted: Not Expanded by / Condensed by

Formatted: Condensed by 0.05 pt

Formatted: Not Expanded by / Condensed by

Formatted: Indent: Left: 1.06"

Formatted: Underline

Formatted: Footer, Centered, Line spacing: single, Tab stops: 0.88", Left + 4.94", Right + 5.13", Left

Docket No. DG 21-130 Exhibit 26 Page 295 of 470 Docket No. DG 20-105 Exhibit 49

is necessary to use Equivalent Bills as a representation for customers. Equivalent Bills are calculated by dividing the number of days in the billing period of each customer's bill by 30.

Billing Year is the 12 months commencing November 1 immediately following the completion of the Decoupling Year.

Decoupling Year. The first Decoupling Year shall be the 10 month period from November 1, 2018 to August 31, 2019. Each subsequent Decoupling Year shall be the twelve months commencing. September 1 through August 31.

d. <u>Customer ClassRate Class is the group of allare customers taking service pursuant to the same rRate sSchedules combined as follows: Rates R. Land R. 5, except that for revenue decoupling purposes rate scheduleRates R. 3, R. 4, R. 6, and R. 7R4, Rates G. 41 and G. 44, Rates G. 42 and G. 45, Rates G. 43 and G. 46, Rates G. 51 and G. 55, Rates G. 52 and G. 56, Rates G. 53 and G. 57, and Rates G. 54 and G. 58, shall be grouped with R3 and R7 with R6.</u>

Formatted: Font: 10.5 pt, Condensed by 0.05 pt

Formatted: Body Text, Right, Indent: Left: 1.32", First line: 0", Right: 0.2", Space Before: 5.9 pt, Numbered + Level: 3 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1.25" + Indent at: 1.5", Tab stops: 1.32", Left

Formatted: No underline, Underline color: Auto, Not Expanded by / Condensed by

Formatted: Font: (Default) Calibri Light, Condensed by 0.05 pt

Formatted: Right, Indent: First line: 0", Space Before: 0.55 pt

Formatted: No underline

Formatted: No underline

 DISSUBD:
 XMářck X020019
 ISSUEDBY: /s/s/SúkářducFieck

 NěúkářducFieck

 EEFFECTIME:
 ANgusmb@021, 2018
 TTTEE: Přessident

Autlantized by CNH RHP Our dardio N26, X8 X dated in dvenderx 0.2020, 80 in dole ket c No. 2020, 1799 48

Formatted: Footer, Centered, Line spacing: single, Tab stops: 0.88", Left + 4.94", Right + 5.13", Left

Docket No. DG 21-130 Exhibit 26 Page 296 of 470 Docket No. DG 20-105 Exhibit 49

LIBERTY UTILITIES Superseding First Revised Page 35	/	Formatted: Highlight	
stomer Class Group Customer Class Group (CG) is the group of rRate eSchedules combined		Formatted: Left, Indent: Left: 0.57", Hanging: 0. Right: 0.27", Space Before: 3.7 pt	25",
	,	Formatted: No underline, Highlight	
Residential Customer Class Group Customer Class Group (CG1): defined as both Residential	f	Before: 3.7 pt, No bullets or numbering, Tab sto	
all consist of all customers taking service pursuant to the Company's residential rate-	,		
nedules. CG1 shall include customers taking service under rate schedules R 1, R 3, R 4, R			
Co and K-/		numbering	
	//	Formatted: Condensed by 0.05 pt	
		Formatted	
	/ //	Formatted: Condensed by 0.05 pt	
	/ ///	Formatted: Condensed by 0.05 pt	
vice customers and does not include other revenue, special contracts, and all revenues recovered	/ ///	Formatted: Condensed by 0.05 pt	
m Cost of Gas filings.	/	Formatted: Condensed by 0.05 pt	
<u>h.</u> ◀		Formatted: Condensed by 0.05 pt	
		Formatted: Condensed by 0.05 pt	
ecupling Year. The first Decoupling Year shall be the 10 month period from November 1, 2018	// /////	Formatted: Condensed by 0.05 pt	
		Formatted: Condensed by 0.05 pt	
	// ////////////////////////////////////		
each sustemer in each billing evole for each month or portion thereof during the Winter Period.			
what distribution revenue for each customer's hill would have been based on weather			
malized therm deliveres for the same period. The resulting charge or credit will be added to or			
finitions:	//		
— Initial Approved Revenue is the Distribution Revenue portion of the total approved	/ /		
revenue requirement is allocated through rate design tofor each Customer ClassRate			
Class less MEP Premium revenues, all as approved by Order of the Commission in a	///		
Class.			
- Incremental Approved Revenue is the amount of any change to the Initial Approved-		Formatted	(
Revenue that is approved by Order of the Commission between rate cases and used to-	/////	Formatted	
· · · · · · · · · · · · · · · · · · ·	'////	Formatted: Condensed by 0.05 pt	
		Formatted: Condensed by 0.05 pt	
	///	Formatted: Condensed by 0.05 pt	
	/ // /	Formatted: No underline	
Equivalent Bills for that Customer ClassRate Class for the test period associated with		Formatted: Condensed by 0.05 pt	
	/ //	Formatted: Condensed by 0.05 pt	
	///	Formatted: Condensed by 0.05 pt	
limited to, step adjustments, property tax reconciliation, and temporary rates.		Formatted	
MAYCK 2020019 ISSSUEDBY: /ss/SuistandudFreek		(
Nsinis Amelia di Frienk			
National Nat			
	stomer Class GroupCustomer Class Group (CG) is the group of rBate sSchedules combined purposes of calculating the Revenue Decoupling Adjustment and the RDAF. The two stomer Class GroupCustomer Class Group are as follows. Residential Customer Class Group Customer Class Group (CG): Jeffined as both Residential in Heating Customer Class Group (CG2): shall be customers taking service pursuant to the Company's residential rate reclutes. Gc1 shall include customers taking service under rate schedules R 1, R 3, R 4, R 8, 6 and R 7. The Commercial and Industrial Customer Class Group Customer Class Group (CG2): shall said of all customers taking service pursuant to one of the Company's general service rate reclutes. Gc11, Gc12, Gc13, Gc14, Gc15, Gc16, Gc17, Gc17	stomer Class GroupCustomer Class Group.(CG) is the group of rRate sSchedules combined purposes of calculating the Revenue Decoupling Adjustment and the RDAF. The two-stomer Class Group Customer Class Group are as follows: Residential Customer Class GroupCustomer Class Group. (CG): Jeffined as both Residential Heating Customer Class Bate Class and Residential Heating Customer Class Bate Class and Heating Customer Class Bate Class and Residential Heating Customer Class Bate Class and Heating Customer Class Group (CG2): shall consist of all customers taking service under rate schedules R. L. R. 3, R. 4, R. R. R. G. and R. 7. The Commercial and Industrial Customer Class GroupCustomer Class Group (CG2): shall noted and the Customer Class Group (CG2): shall noted the Customer of all customers received the Customer of the Customer Shall noted the Customer Class Group Customer Class Group Customer Class Group Customer Class Group Class Group Customer Class Group Class Group Customer Class Group Customer Class Group Class Gr	James Clas. Group Customer Clas. Group CCD is the aroun of Rata Scholulus combined agreement and a second comments of the aroun of Rata Scholulus combined agreement and the RDAF. The two comments of the com

NHPUC NO. 10 GAS Second Revised Page 35

Docket No. DG 21-130 Exhibit 26 Page 297 of 470 Docket No. DG 20-105 Exhibit 49

		,	Formatted	
		/_	Formatted	
	Approved Revenue per Customer ("Approved RPC") is the sum of the Initial RPC per Customer Class Rate Class plus eachand all Incremental RPC's per Customer	//.	Formatted	
	Class Rate Class, if any.	///	Formatted	
	•	///	Formatted	(iii
	A separate RPC is calculated for each month of the decoupling year for each Customer ClassRate	/	Formatted	(:::
	Class using an aApproved distribution rRevenue and set of Equivalent Bills perfor each month.		Formatted	(
	The Initial RPC is set for each month by calculating the Initial Approved Revenue by Rate	// /	Commented [SC3]: Better word?	(
	Class by month from the approved rate schedule rates by Rate Class for that month. The rate	//	Formatted	
	schedule rates adjusted for the MEP Premium are multiplied by the appropriate monthly billing units. Equivalent Bills for customer charges, therms for the blocked therm charges, to produce	///	Formatted	
	the Initial Approved Revenue by Rate Class by month. The monthly Initial Approved Revenue		Formatted	
	are then divided by the Equivalent Bills for that month to produce the Initial RPC's for each Rate Class. For the Initial RPC's, the Equivalent Bills and associated blocked therm billing	/// //	Formatted	
	units shall be adjusted by an End of Year Bill Adjustment that reflects the number of customers	/// //	Formatted	[
	that received service at the end of the test year and Customer Class at the rate case and	// /		
	approved by the Commission to be collected through two components: the customer charge and the per therm charges. The Initial RPC is calculated by dividing the Initial Approved Revenue.	/ /	Formatted	
	into monthly allowed distribution revenue based on the Equivalent Bills approved by the		Commented [IA4]:	
	Commission. The Equivalent Bills for the Initial RPC are proportional to the end of test year customer counts for a given Customer Class. The projected therms used for the monthly	/	Formatted	
	allowed distribution revenue calculation shall reflect the volumetric therms corresponding to the	// //	Formatted	
	end of year customer count.		Formatted	
	The Incremental RPC is set for each month by calculating the Incremental Approved Revenue by	1	Formatted	
	Rate Class by month from the approved rate schedule rates by Rate Class for that month. The rate schedule rates adjusted for the MEP Premium are multiplied by the appropriate monthly		Formatted	
	billing units, Equivalent Bills for customer charges, therms for the blocked therm charges, to		Formatted	
	produce the Incremental Approved Revenue by Rate Class by month. The monthly Initial Approved Revenue are then divided by the Fauivalent Bills for that month to produce the Initial		Formatted	
	RPC's for each. For the Incremental RPC's, the Equivalent Bills and associated blocked therm-		Formatted	
	billing units shall be those bills and units for the test period associated with the Incremental		Formatted	
	Approved Revenue. For each residential Rate Class the Incremental RPC does not include any increase to the customer charge, only to the volumetric therms rates. is calculated by dividing	/ 	Formatted	
	the Incremental Approved Revenue into monthly allowed distribution revenue based on the		Formatted	
	Equivalent Bills approved by the Commission in a proceeding other than a rate case that results in a permanent adjustment to base rates. Equivalent Bills for the determination of Incremental		Formatted	
	RPCs shall use the calendar month Equivalent Bills for the same time period over which the		Formatted	
	Incremental Approved Revenue is determined. For each residential Customer Class the RPC		Formatted	
	only increases the volumetric therms rate.		Formatted	
	h. Calculation of Normal Weather Adjustment		Formatted	
_	• • • • • • • • • • • • • • • • • • •	<i> </i>	Formatted	
	Definitions. The Normal Weather Adjustment (NWA) for each bill is NWA = DeliveryChargeActual × NWF		Formatted	
	where Delivery Charge Actual is the calculated delivery charge for actual delivered		Formatted	
	therms for the		Formatted	
	rate schedule applicable to that bill or portion thereof during the Winter Period,		Formatted	
	h.,		Formatted	
	h. <u>Benchmark Base Revenue per Customer is the monthly allowed distribution</u> revenue per Equivalent Bill for a given Decoupling Year for a given Customer	\leftarrow	Formatted	[
	Class, reflecting the distribution revenue level and approved equivalent bills from		Formatted	
	the Company's most recent rate case or other proceeding that results in an		Formatted	
	adjustment to base rates. Benchmark Base Revenue per Customer will be	11/	Formatted	[
DISSUBD:	XMārchi X02,0019 ISSUEDBBY: //s8\SuisāhrdudFileek NSuisāhrdudFileek	//	Formatted	
EEFEKOTIME:	ANgweniba(CI), 2018 TTTEE: President	/	Formatted	[
	Authorized by AHRUP Or dan No N26. X8 X dated No vember 02 02018 in inches sto No D10-10-048		Formatted	(
			Formatted	[

Docket No. DG 21-130 Exhibit 26 Page 298 of 470 Docket No. DG 20-105 Exhibit 49

calculated for each month based on the distribution rates in effect at the start of the Decoupling Year and the calculation will be revised for the remaining months of each Decoupling Year if there is a distribution rate change that occurs following the beginning month of each Decoupling Year;

Real time normal weather adjustment is the difference between actual distribution revenue billed to each customer in each billing cycle for each month or portion thereof during the Winter Period, and what distribution revenue for each customer's bill would have been based on weather normalized therm deliveries for the same period. The resulting charge or credit will be added to or subtracted from each customer's bill at the time the bill is rendered (i.e., "real time").

h., Winter, Period. The time period from November 1 of a given year through April 30 of the following year inclusive.

h. Base Load Factor for each customer is the customer's most recent two year average daily delivered therms for actual bills rendered for those billing periods that are completely within the June 1 through August 31 calendar period excluding such billing periods that are only partially within the June 1 to August 31 period. If a customer has less than two year's billing history, then the customer's available history for the months of June through August as defined above will be used to calculate the average daily delivered therms; and if a customer has no billing history for the months of June through August as defined above, then the average daily delivered therms for the calendar months of June through August for the rate-schedule under which the customer is served will be used.

h. Base Usage for each bill is the current Base Load Factor times the number of days in the billing period.

viii. Heating Usage for each bill is the difference between the actual delivered therms for that bill less the Base Usage for that bill. If the calculated Heating Usage is less than zero, then the Heating Usage for that bill is set equal to zero.

Formatted: Condensed by 0.05 pt

Formatted: Underline color: Auto

Formatted: Underline color: Auto, Condensed by 0.05

Formatted: Underline color: Auto

Formatted: Indent: Left: 1.63"

Formatted: Not Expanded by / Condensed by

 DISSUBD:
 XMářck X020019
 ISSUEDBY: /s/s/SúkářducFieck

 NěúkářducFieck

 EEFFECTIME:
 ANgusmb@021, 2018
 TTTEE: Přessident

Autlantized by CNH RHP Our dardio N26, X8 X dated in dvenderx 0.2020, 80 in dole ket c No. 2020, 1799 48

Formatted: Footer, Centered, Line spacing: single, Tab stops: 0.88", Left + 4.94", Right + 5.13", Left

Docket No. DG 21-130 Exhibit 26 Page 299 of 470 Docket No. DG 20-105 Exhibit 49

h. <u>Heating Degree Days</u> (HDD) for each day is sixty five (65) minus the average temperature in degrees Fahrenheit for that day. If the calculated HDD is less than zero, then the HDD for that day is set equal to zero.

o. Normal Heating Degree Days (Normal HDD) for each day is the thirty year average HDD for that day:

Formatted: Left, Indent: Left: 1.56", Hanging: 0.13", Numbered + Level: 1 + Numbering Style: i, ii, iii, ... + Start at: 1 + Alignment: Right + Aligned at: 0.82" + Indent at: 1.07"

Formatted: No underline

 DISSUBD:
 XMárck 2020019
 ISSUEDBY: /s8xúsánduchteck

 Nsúsánduchteck

 EEFFECTIME:
 ANgwentbehötül, 2018
 TITTEE:
 PRresident

Authorized ibyoNH NHP OrdenONONO, X87X stated indventom (V202018 ind One ket (ND C V0-175) 48

Formatted: Footer, Centered, Line spacing: single, Tab stops: 0.88", Left + 4.94", Right + 5.13", Left

Docket No. DG 21-130 Exhibit 26 Page 300 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10 GAS Third Revised Formatted: Not Expanded by / Condensed by LIBERTY UTILITIES Superseding Second Revised Page 36 viii. viii. h. Normal Weather Adjustment Slope (NWA Slope) for each bill is the Heating Usage Formatted: Indent: Left: 1.56", Hanging: 0.13", Right: 0.08", Space Before: 5.9 pt, Numbered + Level: 1 + Numbering Style: i, ii, iii, ... + Start at: 1 + Alignment: Right + Aligned at: 0.82" + Indent at: 1.07", Tab stops: Not at 1.07" Formatted: No underline, Underline color: Auto, Not HDD for the billing period, Expanded by / Condensed by Formatted: No underline, Underline color: Auto, Not Expanded by / Condensed by each bill is the sum of the Base Usage and the Nor Formatted: No underline, Underline color: Auto, Not Expanded by / Condensed by Normal Weather Factor (NWF) for each bill is **DeliveryCharge**_N MME **DeliveryCharge**_{Actual} Formatted: Indent: Left: 1.5" where <u>Delivery Charge Normal</u> is the calculated delivery charge for Normal Usage for rate schedule applicable to that bill or portion thereof during the Winter Period and Delivery Charge Actual is the calculated delivery charge for actual delivered therms for Formatted: Indent: Left: 1.5", Right: 0", Space Before: 0 pt, Line spacing: Exactly 10.75 pt chedule applicable to that bill or portion thereof during the Winter Period. The Normal Weather Adjustment (NWA) for each bill is Formatted: Highlight $\underline{\mathsf{NWA}} = \underline{\mathsf{DeliveryCharge}}_{\mathsf{Actual}} \times \underline{\mathsf{NWF}}$ Formatted: Left, Indent: Left: 0.75", No bullets or where Delivery Charge Actual is the calculated delivery charge numbering rate schedule applicable to that bill or portion thereof during the Winter Period. Formatted: No underline, Underline color: Auto, Not Expanded by / Condensed by Application of the NWA to Customer Bills Formatted: Indent: Left: 0.82", No bullets or numbering The NWA charge or credit will be separately stated, and added to or subtracted from each bill as applicable. Each bill will have a separate line titled "Normal Weather Adj.," which line will Formatted: Highlight include the total variable distribution charges, the NWF percentage, and the resulting charge or Formatted: No underline, Underline color: Auto, Not eredit. Expanded by / Condensed by , Highlight Formatted: Right: 0", Space Before: 0 pt, Numbered + Level: 3 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1.25" + Indent at: 1.5" Formatted: Indent: Left: 1.5", Hanging: 0.19", Right: dDistribution rRevenue and the NWA revenue less the MEP Premium for a give 0", Space Before: 0 pt, Numbered + Level: 4 + Decoupling Yearthat month. Of the Actual Distribution Revenue, each month the Numbering Style: i, ii, iii, ... + Start at: 1 + Alignment: Monthly Allowed Revenue is booked as income and the remaining surplus or deficit is Right + Aligned at: 0.61" + Indent at: 0.86", Tab stops: ISSUED: March 20, 2019 · liability tastherbalance essbaetat D. Hilland le 1.69", Left + Not at 0.82" Susan L. Fleck EFFECTIVE: November 01, 2018 TITLE: Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048

Docket No. DG 21-130 Exhibit 26 Page 301 of 470 Docket No. DG 20-105 Exhibit 49

or returned to customers through the RDAF.

Monthly Allowed Revenue is the Approved RPC per Customer ClassRate Class for the applicable month multiplied by the Actual Number of Customers Equivalent Bills for that month. The Monthly Allowed Revenue is booked as income.

Customer Class Group Customer Class Group Forecasted Throughput Volume is the forecasted firm tariff throughput for a given Customer Group for the Billing Year.

of Revenue Decoupling Accounts ("RDA Accounts") are the accounts established exist on the balance sheet for the purpose of recording the Revenue Decoupling Adjustment for each Customer Class GroupCustomer Class Group and are used to determine the RDAF reconciliation amount.

Formatted: Not Expanded by / Condensed by

Commented [IA5]:

Commented [IA6]: Not used - delete

Formatted: Not Expanded by / Condensed by

Commented [IA7]: Through?

Formatted: Not Expanded by / Condensed by

Formatted: Indent: Left: 1.25", Right: 0.08", Space Before: 6.05 pt, No bullets or numbering

Formatted: Not Expanded by / Condensed by

Docket No. DG 21-130 Exhibit 26 Page 302 of 470 Docket No. DG 20-105 Exhibit 49

Description of Revenue Decoupling Adjustment and Reconciliation

Each month the Company will record a Revenue Decoupling Adjustment in the RDA Accounts in accordance with generally accepted accounting principles. The Revenue Decoupling Adjustment is the difference between the Actual Revenue and the Allowed Revenue is collected in RDA Accounts Monthly Allowed Revenue and the Monthly Actual Distribution Revenue. In addition, Tthe reconciliation amounts collected or distributed through the RDAF are also depositedrecorded into the RDA Accounts for each Customer Class GroupCustomer Class Group. The RDA Accounts accrue interest on the average monthly balance using the prime lending rate. At the conclusion of each Decoupling Year, the Company shall calculate a sum of the balance in each of the RDA Accounts for each Customer Class GroupCustomer Class Group Decoupling Revenue Adjustment to bejashall be used to determine the RDAF for the next Billing Year, effective Nevember 1.

Formatted: Highlight

Formatted: Normal, Indent: Left: 1.25", Space Before: 0 pt, No bullets or numbering, Tab stops: Not at 1.07"

Formatted: Indent: Left: 1.25"

The Revenue Decoupling Adjustment shall be determined by calculating the monthly difference between the Benchmark Base Revenue per Customer times the actual number of Equivalent Bills for the applicable Customer Class and the Actual Base Revenue for that month. The RDAF volumetric rate per therm to be applied to customers' bills in the Billing Year is the balance in the RDA Accounts at the end of the Decoupling Yearsum of these monthly Revenue Decoupling Adjustments for each Customer Class.

GroupCustomer Class Group in the Decoupling Year shall be divided by the Forecasted Throughput Volume for that Customer Class GroupCustomer Class Group. forecasted Billing Year sales to derive The volumetric rate per therm to be applied to customers' bills in the Billing Year. The Revenue Decoupling Adjustment shall also include a reconciliation—component for the previous Decoupling Year, which represents the difference between the accrued decoupling amount in the Decoupling Year compared to the actual revenues billed in the billing Year.

Formatted: Indent: Left: 1.25"

Formatted: Condensed by 0.05 pt

Formatted: Condensed by 0.05 pt **Formatted:** Condensed by 0.05 pt

Formatted: Condensed by 0.05 pt

Formatted: Highlight

Revenue Decoupling Adjustment Formulas

RDAF_{CG} = RDA Account Balance_{CG} end of Decoupling Year (August 31) divided by FTV_{CG}

RDA Account Balance_{CG} = Sum of MRDA_{CG} plus RDAF Reconciliation plus interest

MRDA_{CG} = Monthly Allowed Revenue_{CG} less Monthly Actual Revenue_{CG} (including NWA)

Monthly Allowed Revenue_{CG} - Monthly Approved RPC_{CG} times Actual Customers_{CG}

Formatted: Not Expanded by / Condensed by

Formatted: Right, Indent: Left: 1.07", No bullets or

numbering

Formatted: Indent: Left: 1.07", No bullets or

numbering

Formatted: Not Superscript/ Subscript

Formatted: Not Expanded by / Condensed by

Formatted: Right, Indent: Left: 1.07", No bullets or

numbering

 $RD_{CC} = \sum_{RC=1}^{RC=n} \left[\left(BRPC_{--}^{--} x ACUSTS_{--}^{--} \right) - AR_{--}^{--} \right]$

And

 $RDAF_{rec} = \frac{RD_{ee} + CGDEF_{r-1}}{ETV}$

ISSUED: March 20, 2019 ISSUED BY: /s/Susan L. Fleck

EFFECTIVE: November 01, 2018 Susan L. Fleck President

Docket No. DG 21-130 Exhibit 26 Page 303 of 470 Docket No. DG 20-105 Exhibit 49

Docket No. DG 21-130 Exhibit 26 Page 304 of 470 Docket No. DG 20-105 Exhibit 49

Docket No. DG 21-130 Exhibit 26 Page 305 of 470 Docket No. DG 20-105 Exhibit 49

Formatted: Normal, Indent: Left: 1.07", Space Before: 0 pt, Line spacing: Exactly 50.55 pt

Where the terms in the above equation have the following meanings:

Docket No. DG 21-130 Exhibit 26 Page 306 of 470 Docket No. DG 20-105 Exhibit 49

BERTY UTILITIES	Superseding Second Revised Page 37		
ACUSTS _{T-1}	The Actual Number of Equivalent Bills for the applicable Customer Class for the most recently completed Decoupling Year (T-1)		Formatted: Font: 10.5 pt, Subscript, Expanded by 0.0 pt, Raised by 2.5 pt
$\frac{AR_{T-1}}{R}$	The Actual Base Revenue for the applicable Customer Class for the most recently completed Decoupling Year, (T-1), as defined in Section 4(D). For purposes of ealculating the Actual Base Revenue, base revenues for Low Income rate class R-4, shall be determined based on non-discounted rate R-3.		
BRPC _{r-1}	The Benchmark Base Revenue Per Equivalent Bill for the applicable Customer Class as determined in accordance with Section 4 (D) for the most recently completed Decoupling Year, stated on a monthly basis (T-1).		Formatted Table
CG	Customer Class Groups as defined in Section 4(D).		
CGDEF	The balance of the unrecovered deferrals inclusive of associated interest using the prime lending rate.		Formatted: Highlight
FTV _{CG}	Forecast Throughput Volumes inclusive of all firm tariff throughput for the Billing Year.		
RC_	Rate classes in a Customer Class Group.		
RD _{ee}	The Revenue Decoupling adjustment to revenues, representing the sum of the monthly Revenue Decoupling Adjustments in the Decoupling Year.		
RDAF _{ee}	The Revenue Decoupling Adjustment Factor for the Billing Year.		
<u>,10.</u>	•		Formatted: Font: 10.5 pt Formatted: Indent: Left: 0.57", Hanging: 0.25", Tab stops: 0.82", Left + 1.44", Left + 3.31", Left + 3.94", Left
Revenue I transporta	Calculation of the Revenue Decoupling Adjustment Reconciliation 1823 shall contain the accumulated difference between annual revenues and the Decoupling Adjustment, as calculated by multiplying the RDAF times firm sales and tion throughput, and the Revenue Decoupling Adjustment allowed revenues annually, ing charges on the average monthly balance using the prime lending rate.		
108.	Application of the RDAF to Customer Bills		Formatted: Font: 10.5 pt
at the near Customer	per therm) shall be calculated annually for each Customer Group and shall be truncated est one one hundredth of a centfour decimal points per therm. The annual calculated Group RDAF will be applied to the monthly firm tariff throughput for each customer in sular Customer Group, effective November I of the given year.		
	Calculation of Normal Weather Adjustment		Formatted: Font: 10.5 pt
NIMA - D-1:-	Conther Adjustment (NWA) for each bill is		Formatted: Font: 10.5 pt
where Deliver	ory-charge Actual × NW4 Y Charge Actual is the calculated delivery charge for actual delivered therms for the unlimble to that bill or neution thereof during the Winter Period.		Formatted: Font: 10.5 pt
SUED: March 20	Susan L. Fleck	_	Formatted: Body Text, Space Before: 0.05 pt, Tab sto 0.82", Left + 1.44", Left + 3.31", Left + 3.94", Left
FECTIVE: November	er 01, 2018 TITLE: President		

Docket No. DG 21-130 Exhibit 26 Page 307 of 470 Docket No. DG 20-105 Exhibit 49

117. Application of the NWA to Customer Bills

Formatted: Left, Indent: Left: 0.57", Hanging: 0.25", No bullets or numbering, Tab stops: 1.44", Left + 3.31", Left + 3.94", Left

Docket No. DG 21-130 Exhibit 26 Page 308 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10 GAS LIRERTY LITH ITIES Formatted: Font: 10.5 pt applicable. Each bill will have a separate line titled "Normal Weather Adj.," Formatted: Font: 10.5 pt 123. <u>Information to be Filed with the Commission</u> Formatted: Not Highlight Information pertaining to the RDAF will be filed annually with the Commission consistent with the filing requirements of all costs and revenue information included in the LDAC. Such information shall include: The calculation of the RDA Account Balance at the end of the Decoupling Year. Formatted: Not Expanded by / Condensed by The calculation of the RDAF for the Decoupling Year by Customer Class Group Customer Class Group to be applied to the upcoming Billing Year. Formatted: Not Expanded by / Condensed by The approved calculation of the RDAF for the previous Billing Year by Customer Class Group Customer Class Group Formatted: Not Expanded by / Condensed by The calculation for the each Commission Approved RPC and the associated Equivalent Bills for the Initial RPC and each Incremental RPC. Formatted: Not Expanded by / Condensed by upplicable revenue decoupling revenue dollar adjustment for the Decoupling Year by Custome Class Group. The calculation of the revenue decoupling reconciliation dollar adjustment for the previous Decoupling Year by Customer Class Group. The calculation of the proposed decoupling rate per therm for each customer class group to be applied in the Billing Year. The calculation of the monthly Benchmark Base Revenue per Customer, to be utilized in the upcoming Decoupling Year. If distribution rates change during the Decoupling Year, the monthly Benchmark Base Revenue per Customer for the remaining months of the Decoupling Year will be revised and filed with the Commission. LE. Environmental Surcharges ("ES") Allowable for LDAC. Formatted: Left 1. Purpose: In order to recover expenditures associated with former manufactured gas Programs, there shall be an ES Rate applied to all firm volumes billed under the Company's delivery service charges. 2. Applicability: An annual ES Rate shall be calculated effective every November 1 for the annual period of November 1 through October 31. The annual ES Rate shall be filed with the Company's Winter season Cost of Gas Clause ("COG") filing and be subject to review and approval by the Commission. The annual ES Rate shall be applied to firm sales and to firm delivery throughput as a part of the LDAC. Special contract customers are exempt from the ES. Costs Allowable: All approved environmental response costs associated with manufactured gas Programs may be included in the ES Rate The total annual charge to the Company's customers for environmental response costs during any annual ES recovery period shall not exceed five percent (5%) of the Company's total revenues from firm gas sales and delivery throughput during the preceding twelve (12) month period ending June 30. The total annual charge shall represent the ES expenditures reflected in the calculation of the ES Rate to be in effect for the upcoming twelve-month period, November 1 through October 31. If this recovery limitation results in the Company recovering less than the amount that would otherwise be recovered in a particular ES Recovery Year, then the Company would defer this January 28, 2019

ISSUED BY: /s/Susan L. Fleck ISSUED: Susan L. Fleck

TITLE:

Authorized by NHPUC Order No. 26.187 dated November 02, 2018, in Docket No. DG 17-048

EFFECTIVE:

November 01, 2018

Docket No. DG 21-130 Exhibit 26 Page 309 of 470 Docket No. DG 20-105 Exhibit 49

unrecovered amount, with interest, calculated monthly on the average monthly balance, until the next recovery period in which this amount could be recovered without violating the 5% limitation. The interest rate is to be adjusted monthly using the monthly prime lending rate, as reported by the Federal Reserve Statistical Release of Selected Interest Rates.

ISSUED: January 28, 2019 ISSUED BY: Susan L. Fleck Susan L. Fleck EFFECTIVE: November 01, 2018 TITLE: President

 $Authorized \ by \ NHPUC \ Order \ No.\ 26,187 \ dated \ November\ 02,2018, in \ Docket \ No.\ DG\ 17-048$

Docket No. DG 21-130 Exhibit 26 Page 310 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10 GAS

First Revised Page 39
Superseding Original Page 39

LIBERTY UTILITIES

4. Effective Date: On or before the first business day in September of each year, the Company shall file with the NHPUC for its consideration and approval, the Company's request for a change in the ES applicable to all firm sales and firm delivery service throughput for the subsequent twelvemonth period commencing with the calendar month of November.

5. Definitions:

Environmental Response Costs shall include all costs of investigation, testing, remediation, litigation expenses, and other liabilities relating to manufactured gas Program sites, disposal sites, or other sites onto which material may have migrated, as a result of the operating or decommissioning of New Hampshire gas manufacturing facilities. These cost shall include the costs of the closure of the Relief Holder and pond at Gas Street, Concord, NH. The ES shall also include the expenses incurred by the Company in pursuing insurance and third-party claims and any recoveries or other benefits received by the Company as a result

6. Reconciliation Adjustments: Prior to the Winter Period COG, the Company shall calculate the difference between (a) the revenues derived by multiplying firm sales and delivery throughput by the ES Rate, and (b) the historical amortized costs approved for recoveries in the prior November's Annual ES Recovery Period. Account 1920-1863 shall contain the cumulative difference and the Company shall file the reconciliation along with its COG filing on or before the first business day in September of each year.

7. <u>Calculation of the ES</u>: The ES Rate calculated annually consists of one seventh of actual response costs incurred by the Company in the twelve month period ending June 30 of each year until fully amortized (over seven years). Any insurance and third party recoveries or other benefits for the twelve month period ending June 30 shall be applied to reduce the unamortized balance, shortening the amortization period. The sum of these amounts is then divided by the Company's forecast of total firm sales and delivery throughput for the upcoming twelve months of November 1 through October 31.

8. <u>Application of ES to Bills</u>: The annual ES Rate shall be calculated to the nearest one one-hundredth of a cent per therm and shall be applied to the monthly firm gas sales and firm delivery service throughput by being included in the determination of the annual LDAC, and also shall be included in the Distribution Adjustment of the Delivery Charges of each firm customer's lail.

J. Expenses Related to Rate Cases/Temporary Rate Reconciliation Allowable for LDAC

1. <u>Purpose</u>: The purpose of this provision is to establish a procedure that allows the Company to adjust its rates for the recovery of NHPUC approved rate case expenses and the reconciliation of temporary rates.

Applicability: The Rate Case Expenses/Temporary Rate Reconciliation ("RCE") shall be applied to all firm tariffed customers. The RCE will be determined by the Company, as defined below.

ISSUED:	3. Rate Case Expenses Allowable	of for LDAC: The total amount of the RC	E will be equal to the
DATED:	XXX XX, 2021	ISSUED BY:	/s/NSidsRnotudFilenk
EFFECTIVE:	November 01, 2018	TITLE:	Neil-Browdman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPLIC Order No. 26	5 187 dated November 02, 2018, in Docket N	o DG 17-048

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not at 0.82"

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not at 0.82"

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not at 0.82"

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not at 0.57"

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not at 0.82"

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not at 0.82"

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not at 0.82"

Docket No. DG 21-130 Exhibit 26 Page 311 of 470 Docket No. DG 20-105 Exhibit 49

amount approved by the Commission.

Docket No. DG 21-130 Exhibit 26 Page 312 of 470 Docket No. DG 20-105

Exhibit 49

NHPUC NO. 11 Original Page 42

LIBERTY UTILITIES

NHPUC NO. 10 GAS First Revised Page 40

LIBERTY UTILITIES Superseding Original Page 40

4. Effective Date of Rate Case Expense Charge: The effective date of the RCE will be determined the NHPUC in an individual rate proceeding.

5. <u>Definition</u>: The RCE includes all rate case related expenses approved by the NHPUC. This includes legal expenses, costs for bill inserts, costs for legal notices, consulting fees processing variance between the delivery revenues obtained from the rates prescribed in the temporary rate order and the delivery revenues obtained from the final rates approved by the NHPUC.

6. Rate Case Expense/Temporary Rate Reconciliation (RCE) Factor Formulas: The RCE willbe calculated according to the Commission Order issued in an individual proceeding to establish details including the number of years over which the RCE shall be amortized and the allocation of recovery among rate classes. In general, the RCE Factor will be derived by dividing the annual portion of the total RCE, plus the RCE Reconciliation Adjustment, by forecast firm annual throughput.

7. Reconciliation Adjustments: Account 1930 1745 shall contain the accumulated difference revenues toward Rate Case Expenses as calculated by multiplying the Rate Case Expense Factor ("RCEF") times the appropriate monthly volumes and Rate Case Expense allowed, plus carrying charges added to the end-of-month balance. The carrying charges shall becalculated beginning on the first month of the recovery period by applying the lending rate, as reported by the Federal Reserve Statistical Release of Selected Interest Rates to

At the end of the recovery period, any under or over recovery will be included in an active LDAC component, as approved by the Commission.

8. Application of RCE to Bills: The RCE (\$ per therm) shall be calculated to the nearest one -hundredth of a cent per therm and shall be applied to the monthly firm gas delivery service throughput by being included in the determination of the annual LDAC, and also shall be include eustomer's bill.

9. Information to be Filed with the NHPUC: Information pertaining to the RCE will be filed withthe NHPUC consistent with the filing requirements of all cost and revenue information included in the LDAC. The RCE filing will contain the calculation of the new RCE and will include the updated RCE reconciliation balance.

K. Recoverable Residential Low IncomeGas Assistance Program Costs.

1. Purpose: The purpose of this provision is to establish a procedure that allows the Company,

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman Neil Proudman EFFECTIVE: August 1, 2021 TITLE:

Authorized by NHPUC Order No. XX.XXX dated XXX_XX, 2021, in Docket No. DG 20-105

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not at 0.82"

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not at 0.82"

Formatted: Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not at 0.82"

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not

Docket No. DG 21-130 Exhibit 26 Page 313 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 Original Page 42

LIBERTY UTILITIES

subject—to the jurisdiction of the NHPUC, to recover the revenue shortfall (costs) associated with customers participating in the Residential Low IncomeGas Assistance Program ("RLIAPGAP"). Such costs, as well—as, associated administrative and marketing costs shall be recovered by applying an RLIAP GAP rate to—all firm sales and transportation service throughput.

2. <u>Applicability:</u> The RLIAPG Rate shall be applied to all firm sales and transportation tariff-customers. The RLIAP GAP Rate shall be filed with the Company's Winter season Cost of Gas-Clause filing and shall be determined annually by the Company and be subject to review and approval by the Commission.

 $\underline{GAP\ Factor\ Formula\ GAPF} = \underline{GAP + RA}_{GAP}$

A: TPev

where:

GAP GAP costs comprising of the revenue shortfall associated with customer participation, plus administrative, marketing, IT and start up costs.

RAGAP GAP Reconciliation Adjustment Account 1169 1756, inclusive of the associated Account 1169 1756 interest, as outlined in Section 17(G)(6).

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not at 0.82"

 DATED:
 XXX XX, 2021
 ISSUED BY: /s/Neil Proudman

 BEFECTIVE: August 1, 2021

 ISSUED BY: /s/Neil Proudman

 Neil Proudman

 TITLE: President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Docket No. DG 21-130 Exhibit 26 Page 314 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 Original Page 42

LIBERTY UTILITIES

NHPUC NO. 10 GAS Second Revised Page 41

LIBERTY UTILITIES Superseding First Page 41

General Terms and Conditions

3. Effective Date: On or before the first business day in September of each year, the Company-shall file with the NHPUC for its consideration and approval, the Company's request for a change in the GAP Rate applicable to all firm sales, delivery and transportation service throughput for the subsequent twelve month period commencing with the calendar month of November.

4. GAP Costs Allowable for LDAC: The costs to be recovered through the GAP Rate shall-emprised—of the revenue shortfall calculated by forecasting the number of customers enrolled in the GAP and the associated volumetric billing determinants for the upcoming annual recovery period and applying those billing determinants to the difference between the regular and reduced gas assistance—program residential base rates, plus the GAP gas assistance program—discount applied to the cost of gas, administrative, marketing and startup costs. The GAP Rate-shall be calculated by dividing the resulting costs, plus any prior period reconciling adjustment, by the forecast of annual firm sales—and transportation service throughput.

5. CAP Factor Formula GAPF - GAP + RAGAP

A: TPev

where:

A: Tpey Forecast Annual Throughput Volumes of all firm sales and transportation tariffedcustomers eligible to receive firm delivery only service from the Company.

GAP GAP costs comprising of the revenue shortfall associated with customer participation, plus administrative, marketing. IT and start-up costs.

RAGAP GAP Reconciliation Adjustment - Account 1169-1756, inclusive of the associated

6. Reconciliation Adjustments: Prior to the Company's Winter season Cost of Gas filing, the Company—will calculate the difference between (a) the revenue derived by multiplying the actual firm sales and delivery service throughput by the GAP Rate through October 31th, and (b) the actual costs of the program which consists of (1) the revenue shortfall calculated by applying the actual billing—determinants of the GAP classes to the difference in the regular and reduced residential base rates—and cost of gas rates in effect for the annual reconciliation period and (2) the GAP gas assistance program—discount applied to the cost of gas, start up, administrative and marketing costs associated with the—implementation of the program, plus carrying charges calculated on the average monthly balance—using the monthly prime lending rate, as reported by the Federal Reserve Statistical Release of Selected Interest Rates. The combined costs will then be recorded in the deferred GAP account 1169-1756. The Company shall file the reconciliation along with its COG filing on or before the—first business day in September of each year.

L. Effective Date of Local Distribution Adjustment Clause. The LDAC shall be filed annually and become—effective on November 1 of each year pursuant to NHPUC approval. In order to minimize the magnitude—of future reconciliation adjustments, the Company may request interim revisions to the LDAC rates, subject—to review and approval of the NHPUC.

Susan L. Fleck EFFECTIVE: November 01, 2020 TITLE: President

Authorized by NHPUC Order No. 26,397 dated August 27, 2020, in Docket No. DG 20-013

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not at 0.82"

Formatted: Font: Not Italic

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, Line spacing: single, No bullets or numbering, Tab stops: Not at 0.82"

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not at 0.57"

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Docket No. DG 21-130 Exhibit 26 Page 315 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11

LIBERTY UTILITIES

M. Local Distribution Adjustment Clause Formulas. The LDAC shall be calculated on an annual basis, by customer, by summing up the various factors included in the LDAC, where applicable.

Original Page 42

LDAC Formula

LDAC* - CC* + RDAF* + ES + GREF* + RCE + GAP and:

ISSUED: September 11, 2020ISSUED BY: /s/Susan L. Fleck

Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not at 0.57"

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, Tab stops: Not at 0.94" + 4.27"

EFFECTIVE: November 01, 2020 Susan L. Fleck President

Authorized by NHPUC Order No. 26,397 dated August 27, 2020, in Docket No. DG 20-013

Docket No. DG 21-130 Exhibit 26 Page 316 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 Original Page 42 **LIBERTY UTILITIES** NHPUC NO. 10 GAS Original Page 41A Formatted: Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, Line spacing: single, Tab stops: Not at LIBERTY UTILITIES General Terms and Conditions Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt $ES^{*} = RHS + MGP$ where: Formatted: Indent: Left: 0.82", Right: 0.12" Formatted: Indent: Left: 0.82", First line: 0", Right: LDAC* - Annualized class specific LDAC. 0.12", Space Before: 5.9 pt, Line spacing: single, Tab CCX. stops: Not at 1.26" RDAF. Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Total firm annualized ES. Space Before: 5.9 pt Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Annualized charge to recover the costs of the closure of the Relief Holder at Gas Street, Concord, NH Before: 5.9 pt, No bullets or numbering, Tab stops: Not at 0.57" Annualized charge to cover the remediation costs related to former manufactured gas Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", plants. GREF* - Total firm annualized class specific Gas Restructuring Expense Factor. Space Before: 5.9 pt Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space RLIAP - Residential Low Income Assistance Program Rate Before: 5.9 pt, No bullets or numbering, Tab stops: Not at 0.57" Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", eation of LDAC to Bills. The con Space Before: 5.9 pt calculated to the nearest one one hundredth of a cent per therm and shall be applied to the monthly firm sales and firm delivery service throughput in accordance with the table shown in Section 17(B). Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not at 0.82" O. Other Rules. Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt -The NHPUC may, where appropriate, on petition or on its own motion, grant an Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space of these regulation ns, upon such terms that it may determine to be in the public Before: 5.9 pt, No bullets or numbering, Tab stops: Not interest. at 0.82" Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt 2. Such amendments may include the addition or deletion of component cost categories, subject to review and approval of the NHPUC. Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt The NHPUC may, at any time, require the Company to file an amended LDAC. Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not The operation of the LDAC is subject to all powers of suspension and investigation vested in-Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", the NHPUC. Space Before: 5.9 pt **Formatted** P. Amendments to Uniform System of Accounts. Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt record the cumulative difference between the sum of DSM and/or EE Expenditures incurred by **Formatted** Susan L. Fleck

TITLE:

Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048

EFFECTIVE: November 01, 2018

Docket No. DG 21-130 Exhibit 26 Page 317 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 Original Page 42

LIBERTY UTILITIES

the Company plus the sum of DSM and/or EE Repayments and the revenues collected from customers pursuant to this clause with respect to a given Rate Category. Entries to this account-shall be determined as outlined in the Local Distribution Adjustment Clause, 17(C).

1920 1863 Environmental Response Costs Reconciliation Adjustment: This acco be used to record the cumulative difference between the revenues toward environmental response costs as calculated by multiplying the ES times monthly firm sales volumes and delivery ser throughput and environmental response costs allowable per formula. Entries to this accountshall be determined as outlined in the Local Distribution Adjustment Clause, 17(E).

ISSUED: November 16, 2018 ISSUED BY: /s/Susan L. Fleek

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, Tab stops: Not at 0.94" + 4.27"

Susan L. Fleck TITLE: EFFECTIVE: November 01, 2018

Docket No. DG 21-130 Exhibit 26 Page 318 of 470 Docket No. DG 20-105 Exhibit 49

UTILITIES Terms and Conditions		Formatted: Font: (Default) Times New Rom	ian, i
NHPUC NO. 10 GAS First Revised Page 41B		F 1.1 6.1 1.1 6.000 5.11	0.131
		Formatted: Left, Indent: Left: 0.82", Right: Before: 5.9 pt, Line spacing: single, Tab sto	
LIBERTY UTILITIES Superseding Original Page 41B		5.13"	
General Terms and Conditions			
	4	Formatted	
		Formatted	
1930 1745 Rate Case Expense/Temporary Rates Reconciliation Adjustment: This		Formatted: Condensed by 0.1 pt	
account shall be used to record the cumulative difference between the recovery and actual- amounts of third party incremental expenses associated with the Company's Rate Case initiatives	/	Formatted: Condensed by 0.1 pt	
and the difference between the final and temporary distribution rates. Entries to this account shall	//.	Formatted: Condensed by 0.1 pt	
be determined as outlined in the Local Distribution Adjustment Clause, 17(F).	///	Formatted: Condensed by 0.1 pt	
1169 1756 Residential Low Income Assistance Program Reconciliation Adjustment: This account shall be used to record the cumulative difference between the actual revenue derived	///	Formatted: Condensed by 0.1 pt	
from the actual sales and transportation service throughput multiplied by the RLIAP rate and the	////	Formatted: Condensed by 0.1 pt	
actual costs of the program, which consists of the revenue shortfall and all administrative and	#///		
marketing costs, as outlined in the Local Distribution Adjustment Clause, 17(G).		Formatted: Condensed by 0.1 pt	
1168 1823 Revenue Decoupling Adjustment Factor: The RDA is Accounts shall be used to	!///	Formatted: Condensed by 0.1 pt	
record the difference between Actual Revenue and Allowed Revenue as well as the RDAF Revenue and associated interest as described in 17(D), cumulative difference between the lost revenue of the	///	Formatted: Condensed by 0.1 pt	
Company and the revenue collected from customers pursuant to this clause with respect to a given-		Formatted: Condensed by 0.1 pt	
Rate Category. Entries to this account shall be determined as outlined in the Local Distribution		Formatted: Condensed by 0.1 pt	
Adjustment Clause, 17(D).		Formatted	
	1	Formatted	
17 SUPPLY & CAPACITY SHORTAGE ALLOCATION POLICY	1	Formatted	
	1	Formatted	
A. DEFINITIONS	1	Formatted: Indent: Left: 0.82", Right: 0.12	
The following are definitions of terms used in this subsection and applicable only to this	//	Formatted	
subsection:		Formatted	
	1/,	Formatted	
1. <u>Residential</u> : Service to customers which consists of direct natural gas usage in a residential	~/ /,	Formatted	
dwelling for space heating, air conditioning, cooking, water heating and other residential uses	//.	Formatted	
	- ///	Formatted	
B. <u>Commercial</u> : Service to customers engaged primarily in the sale of goods or services including	~/	Formatted	
institutions and local, state and federal government agencies for uses other than those involving-manufacturing or electric power generation	///	Formatted	
manufacturing of electric power generation	_////		
	-	Formatted	
C. <u>Industrial</u> : Service to customers engaged primarily in a process which creates or changes raw or unfinished materials into another form or product including the generation of electric power	* ////	Formatted: Font: 10.5 pt	
	-	Formatted: Right, Right: -0.04"	
D. <u>Large Volume</u> : Service to large commercial and industrial customers with an annual gas load	/ / / //	Formatted	
greater than 200,000 therms		Formatted: Right, Right: -0.04"	
	J///	Formatted: Font: (Default) Times New Rom	nan,
ATTER MANAGEMENT OF THE STATE O		Formatted	
DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudn Neil Proudn			
EFFECTIVE: 2019 Alleus L. Flesk TITLE Alleus L. Flesk TITLE	////		
President	//		

Docket No. DG 21-130 Exhibit 26 Page 319 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS Original Page 39

LIBERTY UTILITIES Terms and Conditions Service available from April 1 to October 31 to all cu

some other fuel or gas for air conditioning purpo

F. Firm Sales Service: Service from schedules or contracts under which seller is expressly obligated to interruptions, but which may permit unexpected interruption in case the supply to higher priority

obligated to deliver specific third party volumes within a given time period and which anticipates customers is threatened.

the plant facilities or danger to plant personnel, when such protection cannot be afforded through the destroyed, but shall not

ISSUED: December 31, 2018 ISSUED BY: /s/Susan L. Fleck

Formatted: Font: (Default) Times New Roman, 10.5 pt

Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not at 0.57"

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not at 0.57"

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Formatted: Left, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, No bullets or numbering, Tab stops: Not at 0.57"

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt

Formatted: Body Text, Indent: Left: 0.82", Right: 0.12", Space Before: 5.9 pt, Tab stops: Not at 0.94" + 4.27"

Formatted: Font: 10.5 pt

Formatted: Right, Right: -0.04"

Formatted: Indent: Left: 0.69", Right: -0.04", Tab stops: 0.94", Left + 1", Left + 1.13", Left + Not at 4.44"

Formatted: Right, Right: -0.04"

Formatted: Font: (Default) Times New Roman, 10.5 pt

Formatted: Footer, Right, Right: -0.04", Line spacing: single, Tab stops: 0.88", Left + 4.94", Right + 5.13", Left

/s/Neil Proudman

Neil Proudman Susan L Fleck Lugust 1, 2021 TITLE: ◀

EFFECTIVE: President President

Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048 Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-10.

Docket No. DG 21-130 Exhibit 26 Page 320 of 470 Docket No. DG 20-105 Exhibit 49

. 6. NHPUC NO. 10 GAS

Second Revised Page 32

uperseding First Revised Page 32

Reconciliation Adjustments: Prior to the Winter Period COG, the Company shall calculate the difference between (a) the revenues derived by multiplying firm sales and delivery throughput by the ES Rate, and (b) the historical amortized costs approved for recoveries in the prior November's Annual ES Recovery Period. Account 1920-1863 shall contain the cumulative difference and the Company shall file the reconciliation along with its COG filing on or before the first business day in September of each year.

- 4-7. Calculation of the ES: The ES Rate calculated annually consists of one-seventh of actual response costs incurred by the Company in the twelve-month period ending June 30 of each year until fully amortized (over seven years). Any insurance and third-party recoveries or other benefits for the twelve month period ending June 30 shall be applied to reduce the unamortized balance, shortening the amortization period. The sum of these amounts is then divided by the Company's forecast of total firm sales and delivery throughput for the upcoming twelve months of November 1 through October 31.
- 5-8. Application of ES to Bills: The annual ES Rate shall be calculated to the nearest one one-hundredth of a cent per therm and shall be applied to the monthly firm gas sales and firm delivery service throughput by being included in the determination of the annual LDAC, and also shall be included in the Distribution Adjustment of the Delivery Charges of each firm customer's bill.
- E.F. Expenses Related to Rate Cases/Temporary Rate Reconciliation Allowable for LDAC.
 - Purpose: The purpose of this provision is to establish a procedure that allows the Company to adjust its rates for the recovery of NHPUC-approved rate case expenses and the reconciliation of temporary rates.
 - 2. <u>Applicability</u>: The Rate Case Expenses/Temporary Rate Reconciliation ("RCE") shall be applied to all firm tariffed customers. The RCE will be determined by the Company, as defined below.
 - 3. <u>Rate Case Expenses Allowable for LDAC</u>: The total amount of the RCE will be equal to the amount approved by the Commission.

Formatted: No underline, Not Highlight

Formatted: No underline

Formatted: Indent: Left: 0.25", Hanging: 0.57", No bullets or numbering, Tab stops: 0.5", Left

Formatted: Not Highlight

Formatted: Underline, Not Highlight

Formatted: Not Highlight

Formatted: Underline, Not Highlight

Formatted: Underline color: Auto, Not Highlight

Formatted: Underline color: Auto

Formatted: Underline

Formatted: Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 7 + Alignment: Left + Aligned at: 0.25" + Indent at: 0.5"

Formatted: Not Highlight

Formatted: Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 7 + Alignment: Left + Aligned at: 0.25" + Indent at: 0.5"

Formatted: Not Highlight

Formatted: Not Highlight

Formatted: Left, Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 6 + Alignment: Left + Aligned

at: 0.25" + Indent at: 0.5"

ISSUED:	November 16, 2018				/s/Susan L. Fleck
DATED:	XXX XX, 2021			ISSUED BY:	/s/NStilsBnotudFilenk
EFFECTIVE:	November 01, 2018			TITLE:	Ner le Providman
EFFECTIVE:	August 1, 2021			TITLE:	President
	A d C II MITTING O	2 1 27 26 107 1 127	1 00 /	1010 ' D 1 . 1	DC 15 040

Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048

<u>Authorized by NHPUC Order No. XX,XXX dated XXX_XX, 2021, in Docket No. DG 20-105</u>

Docket No. DG 21-130 Exhibit 26 Page 321 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 40 Terms and Conditions

Formatted: Font: (Default) Times New Roman, 10.5 pt

NHPUC NO. 10 GAS LIBERTY UTILITIES

First Revised Page 40 eding Original Page 40

8.4. Effective Date of Rate Case Expense Charge: The effective date of the RCE will be determined by the NHPUC in an individual rate proceeding.

Formatted: No underline, Underline color: Auto, Not Expanded by / Condensed by

9.5. <u>Definition</u>: The RCE includes all rate case-related expenses approved by the NHPUC. This includes legal expenses, costs for bill inserts, costs for legal notices, consulting fees processing expenses, and other approved expenses. The temporary Rate reconciliation will include the variance between the delivery revenues obtained from the rates prescribed in the temporary rate order and the delivery revenues obtained from the final rates approved by the NHPUC.

Formatted: Right, Indent: Left: 0.82", No bullets or numbering

Rate Case Expense/Temporary Rate Reconciliation (RCE) Factor Formulas: The RCE will be calculated according to the Commission Order issued in an individual proceeding to establish details including the number of years over which the RCE shall be amortized and the allocation of recovery among rate classes. In general, the RCE Factor will be derived by dividing the annual portion of the total RCE, plus the RCE Reconciliation Adjustment, by forecast firm annual throughput.

<u>Reconciliation Adjustments:</u> Account 1930-1745 shall contain the accumulated difference between revenues toward Rate Case Expenses as calculated by multiplying the Rate Case Expense Factor ("RCEF") times the appropriate monthly volumes and Rate Case Expense allowed, plus carrying charges added to the end-of-month balance. The carrying charges shall be calculated beginning on the first month of the recovery period by applying the monthly prime lending rate, as reported by the Federal Reserve Statistical Release of Selected Interest Rates to the average monthly balance.

Formatted: Body Text, Justified, Indent: Left: 0.57", Hanging: 0.25", Right: 0.08", Space Before: 0 pt, Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.61" + Indent at: 0.86", Tab stops: 0.82", Left + Not at 0.94" +

DATED: XXX XX, 2021 /s/Neil Proudman

Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048 Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105 Formatted: Font: 10.5 pt

Formatted: Font: (Default) Times New Roman, 10.5 pt

Formatted: Footer, Centered, Line spacing: single, Tab stops: 0.88", Left + 4.94", Right + 5.13", Left

Docket No. DG 21-130 Exhibit 26 Page 322 of 470 Docket No. DG 20-105 Exhibit 49

Terms and Conditions Second Revised Page 32 Superseding First Revised Page 32 e end of the recovery period, any under or over recovery will be included in an active LDAC connent, as approved by the Commission. Lication of RCE to Bills: The RCE (\$ per therm) shall be calculated to the nearest one one-tredth of a cent per therm and shall be applied to the monthly firm gas sales and firm delivery ice throughput by being included in the determination of the annual LDAC, and also shall be added in the Distribution Adjustment of the Delivery Charges of each firm customer's bill. Termation to be Filed with the NHPUC: Information pertaining to the RCE will be filed with the PUC consistent with the filing requirements of all cost and revenue information included in the AC. The RCE filing will contain the calculation of the new RCE and will include the updated in the Residential Low Income Gas Assistance Program Costs. Pose: The purpose of this provision is to establish a procedure that allows the Company, subject the jurisdiction of the NHPUC, to recover the revenue shortfall (costs) associated with customers in the Residential Low Income Gas Assistance Program ("RLIAPGAP"). Such costs		Formatted: Left: 0.93", Right: 0.92", Top: 0.46", Bottom: 1.06", Header distance from edge: 0.5", Foote distance from edge: 0.35" Formatted: Not Highlight Formatted: Left, Numbered + Level: 1 + Numbering Style: 1, 2, 3, + Start at: 8 + Alignment: Left + Aligned at: 0.61" + Indent at: 0.86", Tab stops: 0.88", Left + No at 0.82" Formatted Formatted: Not Highlight
e end of the recovery period, any under or over recovery will be included in an active LDAC bonent, as approved by the Commission. Lication of RCE to Bills: The RCE (\$ per therm) shall be calculated to the nearest one one-tredth of a cent per therm and shall be applied to the monthly firm gas sales and firm delivery ice throughput by being included in the determination of the annual LDAC, and also shall be aded in the Distribution Adjustment of the Delivery Charges of each firm customer's bill. To be Filed with the NHPUC: Information pertaining to the RCE will be filed with the PUC consistent with the filing requirements of all cost and revenue information included in the AC. The RCE filing will contain the calculation of the new RCE and will include the updated are reconciliation balance. Loose: The purpose of this provision is to establish a procedure that allows the Company, subject the jurisdiction of the NHPUC, to recover the revenue shortfall (costs) associated with customers in the Residential Low Income Gas Assistance Program ("RLIAPGAP"). Such costs		Bottom: 1.06", Header distance from edge: 0.5", Foote distance from edge: 0.35" Formatted: Not Highlight Formatted: Not Highlight Formatted: Left, Numbered + Level: 1 + Numbering Style: 1, 2, 3, + Start at: 8 + Alignment: Left + Aligned at: 0.61" + Indent at: 0.86", Tab stops: 0.88", Left + No at 0.82" Formatted
bonent, as approved by the Commission. Ilication of RCE to Bills: The RCE (\$ per therm) shall be calculated to the nearest one one- dredth of a cent per therm and shall be applied to the monthly firm gas sales and firm delivery ice throughput by being included in the determination of the annual LDAC, and also shall be uded in the Distribution Adjustment of the Delivery Charges of each firm customer's bill. Internation to be Filed with the NHPUC: Information pertaining to the RCE will be filed with the PUC consistent with the filing requirements of all cost and revenue information included in the AC. The RCE filing will contain the calculation of the new RCE and will include the updated are reconciliation balance. Boose: The purpose of this provision is to establish a procedure that allows the Company, subject the jurisdiction of the NHPUC, to recover the revenue shortfall (costs) associated with customers icipating in the Residential Low Income Gas Assistance Program ("RLIAPGAP"). Such costs		Formatted: Not Highlight Formatted: Not Highlight Formatted: Left, Numbered + Level: 1 + Numbering Style: 1, 2, 3, + Start at: 8 + Alignment: Left + Aligned at: 0.61" + Indent at: 0.86", Tab stops: 0.88", Left + No at 0.82" Formatted
dredth of a cent per therm and shall be applied to the monthly firm gas sales and firm delivery ice throughput by being included in the determination of the annual LDAC, and also shall be used in the Distribution Adjustment of the Delivery Charges of each firm customer's bill. Information to be Filed with the NHPUC: Information pertaining to the RCE will be filed with the PUC consistent with the filing requirements of all cost and revenue information included in the AC. The RCE filing will contain the calculation of the new RCE and will include the updated reconciliation balance. Able Residential Low Income Gas Assistance Program Costs. 1000 100 11 100 110 110 110 110 110 11		Formatted: Not Highlight Formatted: Left, Numbered + Level: 1 + Numbering Style: 1, 2, 3, + Start at: 8 + Alignment: Left + Aligned at: 0.61" + Indent at: 0.86", Tab stops: 0.88", Left + No at 0.82" Formatted
dredth of a cent per therm and shall be applied to the monthly firm gas sales and firm delivery ice throughput by being included in the determination of the annual LDAC, and also shall be used in the Distribution Adjustment of the Delivery Charges of each firm customer's bill. Information to be Filed with the NHPUC: Information pertaining to the RCE will be filed with the PUC consistent with the filing requirements of all cost and revenue information included in the AC. The RCE filing will contain the calculation of the new RCE and will include the updated reconciliation balance. Able Residential Low Income Gas Assistance Program Costs. 1000 100 11 100 110 110 110 110 110 11		Formatted: Not Highlight Formatted: Left, Numbered + Level: 1 + Numbering Style: 1, 2, 3, + Start at: 8 + Alignment: Left + Aligned at: 0.61" + Indent at: 0.86", Tab stops: 0.88", Left + No at 0.82" Formatted
rmation to be Filed with the NHPUC: Information pertaining to the RCE will be filed with the PUC consistent with the filing requirements of all cost and revenue information included in the RCE. The RCE filing will contain the calculation of the new RCE and will include the updated reconciliation balance. **Besidential Low Income**Gas Assistance Program Costs**. **Bose: The purpose of this provision is to establish a procedure that allows the Company, subject to jurisdiction of the NHPUC, to recover the revenue shortfall (costs) associated with customers icipating in the **Residential Low Income***Gas** Assistance Program ("RLIAPGAP"). Such costs		Formatted: Left, Numbered + Level: 1 + Numbering Style: 1, 2, 3, + Start at: 8 + Alignment: Left + Aligned at: 0.61" + Indent at: 0.86", Tab stops: 0.88", Left + No at 0.82" Formatted
rmation to be Filed with the NHPUC: Information pertaining to the RCE will be filed with the PUC consistent with the filing requirements of all cost and revenue information included in the AC. The RCE filing will contain the calculation of the new RCE and will include the updated are reconciliation balance. **Bose: The purpose of this provision is to establish a procedure that allows the Company, subject to jurisdiction of the NHPUC, to recover the revenue shortfall (costs) associated with customers incipating in the **Residential Low Income Gas** Assistance Program ("RLIAPGAP"). Such costs		Style: 1, 2, 3, + Start at: 8 + Alignment: Left + Aligned at: 0.61" + Indent at: 0.86", Tab stops: 0.88", Left + No at 0.82" Formatted
AC. The RCE filing will contain the calculation of the new RCE and will include the updated of reconciliation balance. Ac. The RCE filing will contain the calculation of the new RCE and will include the updated of reconciliation balance. Ac. The RCE filing will contain the updated of reconciliation balance. Acc. The RCE filing will include the updated of reconciliation balance. Acc. The RCE filing will include the updated of reconciliation balance. Acc. The RCE filing will include the updated of reconciliation balance. Acc. The RCE filing will include the updated of reconciliation balance. Acc. The RCE filing will include the updated of reconciliation balance. Acc. The RCE filing will include the updated of reconciliation balance. Acc. The RCE filing will include the updated of reconciliation balance. Acc. The RCE filing will include the updated of reconciliation balance. Acc. The RCE filing will include the updated of reconciliation balance. Acc. The RCE filing will include the updated of reconciliation balance. Acc. The RCE filing will include the updated of reconciliation balance. Acc. The RCE filing will include the updated of reconciliation balance. Acc. The RCE filing will include the updated of reconciliation balance. Acc. The RCE filing will include the updated of reconciliation balance. Acc. The RCE filing will include the updated of reconciliation balance. Acc. The RCE filing will include the updated of reconciliation balance. Acc. The RCE filing will include the updated of reconciliation balance. Acc. The RCE filing will include the updated of reconciliation balance. Acc. The RCE filing will include the updated of reconciliation balance. Acc. The RCE filing will include the updated of reconciliation balance. Acc. The RCE filing will include the updated of reconciliation balance. Acc. The RCE filing will include the updated of reconciliation balance. Acc. The RCE filing will include the updated of reconciliation balance. Acc. The RCE filing will include the updated of reconci		Formatted
able Residential Low Income Gas Assistance Program Costs. Dose: The purpose of this provision is to establish a procedure that allows the Company, subject to purisdiction of the NHPUC, to recover the revenue shortfall (costs) associated with customers icipating in the Residential Low Income Gas Assistance Program ("RLIAPGAP"). Such costs		
bose: The purpose of this provision is to establish a procedure that allows the Company, subject to purisdiction of the NHPUC, to recover the revenue shortfall (costs) associated with customers icipating in the Residential Low Income Gas Assistance Program ("RLIAPGAP"). Such costs	-	Formatted. Not riiginight
te jurisdiction of the NHPUC, to recover the revenue shortfall (costs) associated with customers icipating in the Residential Low IncomeGas Assistance Program ("RLIAPGAP"). Such costs		Formatted
te jurisdiction of the NHPUC, to recover the revenue shortfall (costs) associated with customers icipating in the Residential Low IncomeGas Assistance Program ("RLIAPGAP"). Such costs		
icipating in the Residential Low Income Gas Assistance Program ("RLIAP GAP"). Such costs		Formatted: Not Highlight
	,	Formatted: Not Highlight
vell as, associated administrative and marketing costs shall be recovered by applying an RLIAT Prate to all firm sales and transportation service throughput.	. /	Formatted: Not Expanded by / Condensed by
auc to all littli sales and transportation service unoughput.	1	Formatted
	•/	Formatted: Not Highlight
	—	Formatted: Indent: Hanging: 0.19"
		Formatted
se filing and shall be determined annually by the Company and be subject to review and		Formatted: Left, Indent: Hanging: 0.19"
oval by the Commission.		Formatted: Not Highlight
ective Date: On or before the first business day in Sentember of each year the Company	\	Formatted: Not Highlight
file with the NHPUC for its consideration and approval, the Company's request for a		Formatted: Not Highlight
ge in the GAP Rate applicable to all firm sales, delivery and transportation service		Formatted
ghput for the subsequent twelve-month period commencing with the calendar month of ember.	/	Formatted: Not Highlight
	•	Formatted: Not Highlight
		Formatted: Left, Tab stops: 0.75", Left
associated volumetric billing determinants for the upcoming annual recovery period and		Formatted: Not Highlight
ring those billing determinants to the difference between the regular and reduced gas assistance		Formatted: No underline, Not Highlight
		Formatted: Not Highlight
any prior period reconciling adjustment, by the forecast of annual firm sales and transportation		Formatted
ce throughput,	_//	Formatted: Not Highlight
		Formatted: Indent: Hanging: 0.19"
AP Factor		Formatted: Not Highlight
ula_	_/	Formatted: Font: 10.5 pt
$F = GAP + RA_{GAP}$	/	Formatted: Font: (Default) Times New Roman, 10.5 pt
<u>Shi Haton</u>	/,	Formatted
X XX, 2021 ISSUED BY: /s/Neil Proudman		Tormatted
uary 01, 2019 THTE: Nesquit	_ //	
horized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048	/	
e e e e e e e e e e e e e e e e e e e	val by the Commission. ctive Date: On or before the first business day in September of each year, the Company file with the NHPUC for its consideration and approval, the Company's request for a e in the GAP Rate applicable to all firm sales, delivery and transportation service about for the subsequent twelve-month period commencing with the calendar month of mber. Costs Allowable for LDAC: The costs to be recovered through the GAP Rate shall comprised revenue shortfall calculated by forecasting the number of customers enrolled in the GAP and associated volumetric billing determinants for the upcoming annual recovery period and residential base rates, plus the GAP discount applied to the cost of gas, administrative timing and startup costs. The GAP Rate shall be calculated by dividing the resulting costs, my prior period reconciling adjustment, by the forecast of annual firm sales and transportation throughput, AP Factor ala E GAP + RAGAP EV	mers. The RLIAP GAP Rate shall be filed with the Company's Winter season Cost of Gas e filing and shall be determined annually by the Company and be subject to review and wal by the Commission. Concident the Company and be subject to review and wal by the Commission. Concident the NHPUC for its consideration and approval, the Company's request for a e in the GAP Rate applicable to all firm sales, delivery and transportation service thought for the subsequent twelve-month period commencing with the calendar month of mber. Costs Allowable for LDAC: The costs to be recovered through the GAP Rate shall comprised revenue shortfall calculated by forecasting the number of customers enrolled in the GAP and associated volumetric billing determinants for the upcoming annual recovery period and ing those billing determinants to the difference between the regular and reduced gas assistance are residential base rates, plus the GAP discount applied to the cost of gas, administrative, thing and startup costs. The GAP Rate shall be calculated by dividing the resulting costs, my prior period reconciling adjustment, by the forecast of annual firm sales and transportation to throughput, AP Factor Ila E-GAP+RAGAP EV EXXX, 2021 ISSUED BY: /s/Neil Proudman THIFF THIFF ISSUED BY: /s/Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 323 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS Original Page 41 Terms and Conditions

Formatted: Font: (Default) Times New Roman, 10.5 pt

LIBERTY UTILITIES

where:

A: Tpev Forecast Annual Throughput Volumes of all firm sales and transportation tariffed customers eligible to receive firm delivery-only service from the Company.

GAP costs comprising of the revenue shortfall associated with customer participation, plus administrative, marketing, IT and start-up costs.

GAP Reconciliation Adjustment - Account 1169-1756, inclusive of the associated Account 1169-1756 interest, as outlined in Section 19(G)(6).

> Formatted: Justified, Indent: Left: 0", Space Before: 0 pt, Tab stops: Not at 0.94" + 4.27"

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Formatted: Font: 10.5 pt

Formatted: Font: (Default) Times New Roman, 10.5 pt

Formatted: Footer, Centered, Line spacing: single, Tab stops: 0.88", Left + 4.94", Right + 5.13", Left

Docket No. DG 21-130 Exhibit 26 Page 324 of 470 Docket No. DG 20-105 Exhibit 49

6. NHPUC NO. 10 GAS

LIBERTY UTILITIES

Superseding First Revised Page 32

Reconciliation Adjustments: Prior to the Company's Winter season Cost of Gas filing, the Company will calculate the difference between (a) the revenue derived by multiplying the actual firm sales and delivery service throughput by the GAP Rate through October 31st, and (b) the actual costs of the program which consists of (1) the revenue shortfall calculated by applying the actual billing determinants of the GAP classes to the difference in the regular and reduced residential base rates and cost of gas rates in effect for the annual reconciliation period and (2) the GAP discount applied to the cost of gas, start-up, administrative and marketing costs associated with the implementation of the program, plus carrying charges calculated on the average monthly balance using the monthly prime lending rate, as reported by the Federal Reserve Statistical Release of Selected Interest Rates. The combined

costs will then be recorded in the deferred GAP account 1169-1756. The Company shall file the reconciliation along with its COG filing on or before the first business day in September of each year.

Formatted: Left: 0.93", Right: 0.92", Top: 0.46", Bottom: 1.06", Header distance from edge: 0.5", Footer distance from edge: 0.35"

H. NHPUC NO. 10 GAS Second Revised Page 41

LIBERTY UTILITIES Superseding First Page 41

General Terms and Conditions

Effective Date: On or before the first business day in September of each year, the Company shall file with the NHPUC for its consideration and approval, the Company's request for a change in the GAP Rate-applicable to all firm sales, delivery and transportation service throughput for the subsequent twelve-month period commencing with the calendar month of November.

GAP Costs Allowable for LDAC: The costs to be recovered through the GAP Rate shall comprised of the revenue shortfall calculated by forecasting the number of customers enrolled in the GAP and the associated volumetric billing determinants for the upcoming annual recovery period and applying those billing determinants to the difference between the regular and reduced gas assistance program residential base rates, plus the GAP gas assistance program discount applied to the cost of gas, administrative, marketing and startup costs. The GAP Rate shall be calculated by dividing the resulting costs, plus any prior period reconciling adjustment, by the forecast of annual firm sales and transportation corrier throughout.

GAP Factor Formula GAPF - GAP + RAGAP

A: TPev

where:

A: Tpev Forecast Annual Throughput Volumes of all firm sales and transportation tariffed customers eligible to receive firm delivery only service from the Company.

GAP GAP costs comprising of the revenue shortfall associated with customer participation, plus administrative, marketing, IT and start up costs.

RAGAD GAP Reconciliation Adjustment Account 1169-1756, inclusive of the associated Account 1169-1756 interest, as outlined in Section 17(G)(6).

Reconciliation Adjustments: Prior to the Company's Winter season Cost of Gas filing, the Company willcalculate the difference between (a) the revenue derived by multiplying the actual firm sales and

delivery service throughput by the GAP Rate through October 31^{5t}, and (b) the actual costs of the program which consists of (1) the revenue shortfall calculated by applying the actual billing determinants of the GAP classes to the difference in the regular and reduced residential base rates and cost of gas rates in effect for the annual reconciliation period and (2) the GAP gas assistance program—discount applied to the cost of gas, start up, administrative and marketing costs associated with the implementation of the program, plus carrying charges calculated on the average monthly balance—using the monthly prime lending rate, as reported by the Federal Reserve Statistical Release of Selected—Interest Rates. The combined costs will then be recorded in the deferred GAP account 1169 1756. The Company shall file the reconciliation along with its COG filing on or before the—first business day in Sentember of each year.

Effective Date of Local Distribution Adjustment Clause. The LDAC shall be filed annually and become effective Susan L. Fleck

EFFECTIVE: November 01, 2020

TITLE: President

Authorized by NHPUC Order No. 26,397 dated August 27, 2020, in Docket No. DG 20-013

Formatted: Font: Not Italic

Docket No. DG 21-130 Exhibit 26 Page 325 of 470 Docket No. DG 20-105 Exhibit 49

on November 1 of each year pursuant to NHPUC approval. In order to minimize the magnitude of future reconciliation adjustments, the Company may request interim revisions to the LDAC rates, subject to review and approval of the NHPUC.

 Local Distribution Adjustment Clause Formulas. The LDAC shall be calculated on an annual basis, by customer, by summing up the various factors included in the LDAC, where applicable.

LDAC Formula

 $LDAC^X \ = \ CC^X + RDAF^X + ES + GREF^X + RCE + GAP_and;$

ISSUED: September 11, 2020 ISSUED BY:

/s/Susan L. Fleck

EFFECTIVE: November 01, 2020 Susan L. Fleck
President

Authorized by NHPUC Order No. 26,397 dated August 27, 2020, in Docket No. DG 20-013

Docket No. DG 21-130 Exhibit 26 Page 326 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS

Original Page 42

LIBERTY UTILITIES

Terms and Conditions

NHPUC NO. 10 GA

Original Page 41A

LIBERTY UTILITIES

General Terms and Conditions

 $ES^{X} = RHS + MGP$

where:

LDAC^X = Annualized class specific LDAC.

CCX = Annualized class specific CC or EE Charge.

RDAF^X = Annualized class specific RDAF.

Total firm annualized ES.

RHS = Annualized charge to recover the costs of the closure of the Relief Holder at Gas Street,

MGP = Annualized charge to cover the remediation costs related to former manufactured gas plants.

GREF^X = Total firm annualized class specific Gas Restructuring Expense Factor.

RCE = Rate Case Expense Factor.

GRLIAP = GasResidential Low Income Assistance Program Rate

Application of LDAC to Bills. The component costs comprising the LDAC (\$ per therm) shall be calculated to the nearest one one-hundredth of a cent per therm and shall be applied to the monthly firm sales and firm delivery service throughput in accordance with the table shown in Section <u>1719</u>(B).

K. Other Rules.

- 1. The NHPUC may, where appropriate, on petition or on its own motion, grant an exception from the provisions of these regulations, upon such terms that it may determine to be in the public interest.
- Such amendments may include the addition or deletion of component cost categories, subject to the

Formatted: Heading 2, Space Before: 0 pt, Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.69" + Indent at: 0.94", Tab stops: 0.75", Left + 1", Left + 1.31", Left + Not at 0.94" + 4.27"

XXX XX, 2021 /s/Neil Proudman

Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048 Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105 Formatted: Footer, Centered, Line spacing: single, Tab stops: 0.88", Left + 4.94", Right + 5.13", Left

Docket No. DG 21-130 Exhibit 26 Page 327 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10 GAS

Second Revised Page 32

LIBERTY UTILITIES

review and approval of the NHPUC.

3. The Company may implement an amended LDAC with the NHPUC approval at any time.

- 4. The NHPUC may, at any time, require the Company to file an amended LDAC.
- The operation of the LDAC is subject to all powers of suspension and investigation vested in the NHPUC.
- L. Amendments to Uniform System of Accounts

1163-1755 — Energy Efficiency Reconciliation Adjustment: This account shall be used to record the cumulative difference between the sum of DSM and/or EE Expenditures incurred by the Company plus the sum of DSM and/or EE Repayments and the revenues collected from customers pursuant to this clause with respect to a given Rate Category. Entries to this account shall be determined as outlined in the Local Distribution Adjustment Clause, 17(C).

1920-1863 Environmental Response Costs Reconciliation Adjustment: This account shall be used to record the cumulative difference between the revenues toward environmental response costs as calculated by multiplying the ES times monthly firm sales volumes and delivery service throughput and environmental response costs allowable per formula. Entries to this account shall be determined as outlined in the Local Distribution Adjustment Clause, 17(E).

ISSUED: November 16, 2018

ISSUED BY: /s/Susan L. Fleck

Susan L. Fleck

TITLE:

Formatted: Left: 0.93", Right: 0.92", Top: 0.46", Bottom: 1.06", Header distance from edge: 0.5", Footer distance from edge: 0.35"

Formatted: Tab stops: 1.31", Left + 1.38", Left + 1.63",

EFFECTIVE: November 01, 2018

Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048

Docket No. DG 21-130 Exhibit 26 Page 328 of 470 Docket No. DG 20-105 Exhibit 49

 NHPUC NO. 11 GAS
 Original Page 43

 LIBERTY UTILITIES
 Terms and Conditions

 NHPUC NO. 10 GAS
 First Revised Page 41B

 LIBERTY UTILITIES
 Superseding Original Page 41B

Formatted: Font: (Default) Times New Roman, 10.5 pt

1930-1745 Rate Case Expense/Temporary Rates Reconciliation Adjustment: This account shall be used to record the cumulative difference between the recovery and actual amounts of third-party incremental expenses associated with the Company's Rate Case initiatives and the difference between the final and temporary distribution rates. Entries to this account shall be determined as outlined in the Local Distribution Adjustment Clause, 17(F).

eral Terms and Conditions

1169-1756 <u>RGasesidential Law Income</u> Assistance Program Reconciliation Adjustment: This account shall be used to record the cumulative difference between the actual revenue derived from the actual sales and transportation service throughput multiplied by the <u>GRLI</u>AP rate and the actual costs of the program, which consists of the revenue shortfall and all administrative and marketing costs, as outlined in the Local Distribution Adjustment Clause, 17(G).

1168-1823 Revenue Decoupling Adjustment Factor: The RDA is Aaccounts shall be used to record the difference between Actual Revenue and Allowed Revenue as well as the RDAF Revenue and associated interest as described in 17(D), cumulative difference between the lost revenue of the Company and the revenue collected from customers pursuant to this clause with respect to a given Rate Category. Entries to this account—shall be determined as outlined in the Local Distribution Adjustment Clause, 17(D).

1820 SUPPLY & CAPACITY SHORTAGE ALLOCATION POLICY

A. <u>DEFINITIONS</u>

The following are definitions of terms used in this subsection and applicable only to this subsection:

- Residential: Service to customers which consists of direct natural gas usage in a residential dwelling for space heating, air conditioning, cooking, water heating and other residential uses
- Commercial: Service to customers engaged primarily in the sale of goods or services including
 institutions and local, state and federal government agencies for uses other than those involving
 manufacturing or electric power generation
- 3. <u>Industrial</u>: Service to customers engaged primarily in a process which creates or changes raw or unfinished materials into another form or product including the generation of electric power
- Large Volume: Service to large commercial and industrial customers with an annual gas load greater than 200,000 therms

Formatted: Right: 0.08"

Formatted: Condensed by 0.1 pt

ormatted: Condensed by 0.1 pt

Formatted: Condensed by 0.1 pt

Formatted: Condensed by 0.1 pt

Formatted: Condensed by 0.1 pt

Formatted: Condensed by 0.1 pt

Formatted: Condensed by 0.1 pt

Formatted: Condensed by 0.1 pt

Formatted: Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 20 + Alignment: Left + Aligned at: 0.57" + Indent at: 0.82"

Formatted: Body Text, Justified, Right: 0.08", Space Before: 0 pt, Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.57" + Indent at: 0.82", Tab stops: 0.57", Left + Not at

Formatted: Font: 10.5 pt

0.94" + 4.27"

Formatted: Font: (Default) Times New Roman, 10.5 pt

Formatted: Footer, Centered, Line spacing: single, Tab

stops: 0.88", Left + 4.94", Right + 5.13", Left

Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048

<u>Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105</u>

Docket No. DG 21-130 Exhibit 26 Page 329 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS

LIBERTY UTILITIES

Original Page 44

Terms and Conditions

NHPUC NO. 10 GAS Second Revised Page 32
LIBERTY UTILITIES Superseding First Revised Page 32

Formatted: Font: (Default) Times New Roman, 10.5 pt

Formatted: Left: 0.93", Right: 0.92", Top: 0.46", Bottom: 1.06", Header distance from edge: 0.5", Footer distance from edge: 0.35"

- Seasonal: Service available from April 1 to October 31 to all customers using gas to replace some other fuel or gas for air conditioning purposes
- 6. <u>Firm Sales Service</u>: Service from schedules or contracts under which seller is expressly obligated to supply and deliver specific volumes within a given time period and which anticipates no interruptions, but which may permit unexpected interruption in case the supply to higher priority customers is threatened
- 7. <u>Firm Transportation Service</u>: Service from schedules or contracts under which seller is expressly obligated to deliver specific third-party volumes within a given time period and which anticipates no interruptions, but which may permit unexpected interruption in case the supply to higher priority customers is threatened.
- 8. <u>Plant Protection Gas:</u> Is defined as minimum volumes required to prevent physical harm to the plant facilities or danger to plant personnel, when such protection cannot be afforded through the use of alternate fuel. This includes the protection of such material in process as would otherwise be destroyed, but shall not

ISSUED: December 31, 2018 ISSUED BY: /s/Susan L. Fleck

Formatted: Space Before: 0 pt

Formatted: Indent: Left: 0", Space Before: 0 pt, Tab stops: Not at 0.94" + 4.27"

Formatted: Font: 10.5 pt

Formatted: Font: (Default) Times New Roman, 10.5 pt

 $\textbf{Formatted:} \ \mathsf{Footer}, \mathsf{Centered}, \mathsf{Line} \ \mathsf{spacing:} \ \mathsf{single}, \mathsf{Tab}$

stops: 0.88", Left + 4.94", Right + 5.13", Left

Authorized by NHPUC Order No. 26,187 dated November 02, 2018, in Docket No. DG 17-048

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Docket No. DG 21-130 Exhibit 26 Page 330 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 45 General Terms and Conditions

include deliveries required to maintain plant production. For the purpose of this definition, propane and other gaseous fuels shall not be considered alternate fuels.

- I. <u>Feedstock Gas</u>: Is defined as natural gas used as a raw material for its chemical properties in creating an end product
- J. <u>Process Gas</u>: Is defined as gas use for which alternate fuels are not technically feasible such as in applications requiring precise temperature controls and precise flame characteristics. For the purpose of this definition, propane and other gaseous fuels shall not be considered alternate fuels
- K. <u>Boiler Fuel</u>: Is considered to be natural gas used as a fuel for the generation of steam or electricity including the utilization of gas turbines for the generation of electricity
- L. <u>Alternate Fuel Capabilities</u>: Is defined as a situation where an alternate fuel could have been utilized whether or not the facilities for such use have actually been installed, provided however, where the use of natural gas is for plant protection, feedstock or process uses and the only alternate fuel is propane or other gaseous fuel, then the consumer will be treated as if he had no alternate fuel capability.

M. POLICY

In the event that, due to gas supply restrictions or capacity constraints, the Company is unable to deliver the total requirements of its firm, sales or transportation rate customers, the available volumes of gas will be allocated to the Company's firm rate customers in accordance with the provisions of this policy. In the event that the Company, during a curtailment or interruption, requires emergency gas, and takes the gas of the customer, customer shall be compensated for such emergency gas at the customer's alternate cost of fuel as demonstrated to the reasonable satisfaction of the Company.

As curtailment becomes necessary through each succeeding category, the Company will implement full or partial curtailment of a customer, or by groups of customers, taking into consideration customer load characteristics, the Company's delivery system design and Company load characteristics in a manner which is believed to be in the best interests of customers in general.

N. PRIORITIES

Firm rate customers shall be serviced according to the following preference categories with the first and each succeeding category having preference over the succeeding categories:

- 1. Company use for fuel and lost and unaccounted for
- O. Firm sales or transportation service for high priority residential uses including apartment buildings and other multi-unit buildings, small commercial establishments using less than 50 DT on a peak day, schools, hospitals, police protection, fire protection, sanitation facilities and correctional facilities
- P. Firm sales or transportation service for essential agricultural uses, as defined by the Secretary of Agriculture, for full food and natural fiber production, process and feedstock use for fertilizer and agricultural chemicals,

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 331 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 46 General Terms and Conditions

process and feedstock for animal feeds and food, food quality maintenance, food packaging, marketing and distribution for food related products and on farm uses

- Q. Firm sales or transportation service for large commercial requirements (50 DT or more on a peak day), firm industrial requirements for plant protection, feedstock and process needs and firm industrial sales up to 300 DT per day
- R. Firm sales or transportation service for all industrial requirements not specified in (2), (3), (4), (6), or (7)
- S. Firm sales or transportation service including the transportation for industrial requirements for boiler fuel use at less than 1,500 DT per day, but more than 300 DT per day, where alternate fuel capabilities can meet such requirements
- T. Firm sales or transportation service including transportation for industrial requirements for large volume (1,500 DT or more per day) boiler fuel use where alternate fuel capabilities can meet such requirements

U. STORAGE INJECTION

Within each category, storage injection required to meet the needs of higher priorities may be given preference over all other uses within that category.

V. PENALTY

For all unauthorized volumes of gas taken by a customer, the customer shall pay the Company a penalty of five times the daily index for each therm taken. Such penalty shall be added to the regular rates in effect. The Company shall have the right, without obligation, to waive any penalty for unauthorized use of gas, if on the day when the penalty was incurred deliveries to other of the Company's customers were not adversely affected. Continued unauthorized use, at the sole discretion of the Company, may result in termination of service.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 332 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Revised Page 44 Fifth Revised Original Page 4447
Residential Non-Heating Rate R-1 Superseding Fourth

Residential Non-Heating Rate R-1

II. RATE SCHEDULES

1 RESIDENTIAL NON-HEATING RATE: CLASSIFICATION NO. R-1

Availability

This rate is available to all residential customers who do not have gas space heating equipment, who consume less than 80% of their normal usage in the six winter months of November through April and whose usage does not exceed 100 therms in any winter month. Available for use which is separately metered and billed for each dwelling unit. Availability is limited to use in locations served by the Company's mains and for which the Company's facilities are adequate.

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter: \$0.5166 59975130 per day or \$15.5017.9915.39 per 30 day month

Winter Period: All therms per 30 day month at \$0.3860 41374358 per therm

Summer Period: All therms per 30 day month at \$0.3860 41374358 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Cost of Gas Charge

All gas delivered under this rate is subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charges presented above are exclusive of these charges. Refer to the Firm Rate Schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent ($1\frac{1}{2}$ %) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
ISSUED:	Authorized by NHPUC Order No. XX,XXX dated XXX XX, July 13, 2020		/s/Susan I Flook
EFFECTIVE:	July 01, 2020	TITLE:	Susan L. Fleek President

Docket No. DG 21-130 Exhibit 26 Page 333 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES Revised Page 44 Fifth Revised Original Page 4447
Residential Non-Heating Rate R-1 Superseding Fourth

Residential Non-Heating Rate R-1

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 20	21, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	
	<u> </u>		Susan L. Fleck
EFFECTIVE:	July 01, 2020	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 334 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 48 Residential Non-Heating Rate R-1

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 335 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-<u>11</u> GAS LIBERTY UTILITIES Revised Page 46 Fifth Revised Original Page 4649
Residential Heating Rate R-3Superseding Fourth

Residential Heating Rate R-3

2 RESIDENTIAL HEATING RATE: CLASSIFICATION NO. R-3

Availability

This rate is for all residential use for those domestic customers who use gas as the principal household heating fuel. Availability is limited to use in domestic locations which are separately metered and billed and which are served by the Company's mains and for which the Company's facilities are adequate.

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter: \$0.5166 59975130 per day or \$15.5017.9915.39 per 30 day month

Winter Period: All therms per 30 day month at \$0.5678 65205985 per therm

Summer Period: All therms per 30 day month at \$0.5678 65205985 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Cost of Gas Charge

All gas delivered under this rate is subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charges presented above are exclusive of these charges. Refer to the Firm Rate Schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

Eligibility shall be determined based on the reasonable discretion of the Company subject to verification of heating usage.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX	XX, 2021, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleek
EFFECTIVE:	July 01, 2020	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 336 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Revised Page 46 Fifth Revised Original Page 4649
Residential Heating Rate R-3 Superseding Fourth

Residential Heating Rate R-3

one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX	XX, 2021, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleck
EFFECTIVE:	July 01, 2020	TITLE:	President
	• /		

Docket No. DG 21-130 Exhibit 26 Page 337 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 50 Residential Heating Rate R-3

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Sixth Revised Original Page 4851 Superseding Fifth Revised Page 48

-Residential Gas Assistance Heating Rate R-4

3 GAS ASSISTANCE PROGRAM RESIDENTIAL HEATING RATE: CLASSIFICATION NO. R-4

Availability

This rate is for residential use for those domestic customers who use gas as the principal household heating fuel if any member of the household qualifies for a benefit through one of the programs listed below, subject to the qualification period described under the "Terms and Conditions" of this rate. Availability is limited to use in domestic locations which are separately metered and billed and which are served by the Company's mains and for which the Company facilities are adequate.

Qualified Programs:

- a. Low Income Home Energy Assistance Program (LIHEAP)
- b. Electric Assistance Program (EAP)
- c. Supplemental Security Income Program
- d. Women, Infants and Children Program
- e. Commodity Surplus Foods Program (for women, infants and children)
- f. Elderly Commodity Surplus Foods Program
- g. Temporary Aid to Needy Families Program
- h. Housing Choice Voucher Program (also known as Section 8)
- i. Head Start Program
- j. Aid to the Permanently and Totally Disabled Program
- k. Aid to the Needy Blind Program
- 1. Old Age Assistance Program
- m. Food Stamps Program
- n. Any successor program of a-m

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Winter Customer Charge Per Meter: \$0.2843-2823 per day or \$8.53-47 per 30 day month
Winter Period: All therms per 30 day month at \$0.3123-3292 per therm
Summer Customer Charge Per Meter: \$0.5167-5130 per day or \$15.50-39 per 30 day month
Summer Period: All therms per 30 day month at \$0.5678-5985 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive.

roudman
roudman
ent
105
L. Fleck
L. Fleck
ent
1

Docket No. DG 21-130 Exhibit 26 Page 339 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10 11 GAS LIBERTY UTILITIES

Sixth Revised Original Page 4851 Superseding Fifth Revised Page 48

-Residential Gas Assistance Heating Rate R-4

Cost of Gas Charge

All gas delivered under this rate is subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX X	X, 2021, in Docket N	o. DG 20-105
ISSUED:	June 4, 2021	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleck
EFFECTIVE:	November 01, 2020	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 340 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 52 Residential G Heating Rate R-4

charges presented above are exclusive of these charges. Refer to the Firm Rate Schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

For those customers qualifying for the program this rate R-4 shall apply for a six month winter period. On the date that the six month winter period expires, eligibility for this rate shall expire unless the customer provides the Company with evidence that the customer continues to be eligible for one or more qualifying programs. When the Rate R-4 expires, the rate on each account shall revert back to the non-gas assistance program Residential Heating Rate, R-3. Customers whose eligibility for the program is based on their having qualified for LIHEAP shall be eligible for this rate retroactive to November 1 of the heating season in which they qualified. Eligibility for such customers shall expire the following April 30, subject to their re-qualifying through receipt of LIHEAP or other benefits as set forth above.

Eligibility shall be determined based on the reasonable discretion of the Company subject to verification of heating usage.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent ($1\frac{1}{2}$ %) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

EFFECTIVE: August 1, 2021 TITLE: President

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 53

MEP Residential Non-Heating Rate R-5

MEP Residential Non-Heating Rate R-5

4 MANAGED EXPANSION PROGRAM RESIDENTIAL NON-HEATING RATE: CLASSIFICATION NO. R-5

Availability

This rate is mandatory for customers taking service in a Managed Expansion Program project area who otherwise would have qualified for Residential Non Heating Rate R-1.

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter: \$0.6715-6670 per day or \$20.15-01 per 30 day month

Winter Period: All therms per 30 day month at \$0.5018-5666 per therm

Summer Period: All therms per 30 day month at \$0.5018-5666 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Cost of Gas Charge

All gas delivered under this rate is subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charges presented above are exclusive of these charges. Refer to Page 92 of this Tariff for firm rate schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

Service under each Managed Expansion Program project will have a term of ten years. Customers initiating service under this rate must take service hereunder until ten years following the date that the first customer in the particular Managed Expansion Program project takes service. Once the term of service for a particular Managed Expansion Program project expires, customers will thereafter take service under Residential Non Heating Rate R-1.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman
Neil Proudman
EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 342 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 54 MEP Residential Non-Heating Rate R-5

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 343 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Revised Page 52 Fourth Revised Original Page 5255

MEP Residential Heating Rate R-6 Superseding Third

MEP Residential Heating Rate R-6

5 MANAGED EXPANSION PROGRAM RESIDENTIAL HEATING RATE: CLASSIFICATION NO. R-6

Availability

This rate is mandatory for customers taking service in a Managed Expansion Program projects area who otherwise would have qualified for Residential Heating Rate R-3.

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter:\$0.671577986670- per day or \$20.1523-20.0139 per 30 day month

Winter Period: All therms per 30 day month at \$0.7381 84767780 per therm

Summer Period: All therms per 30 day month at \$0.7381 84767780 per therm The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Cost of Gas Charge

All gas delivered under this rate is subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to Page 92 of this Tariff for firm rate schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charges presented above are exclusive of these charges. Refer to the Firm Rate Schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

Eligibility shall be determined based on the reasonable discretion of the Company subject to verification of heating usage.

Service under each Managed Expansion Program project will have a term of ten years. Customers initiating service under this rate must take service hereunder until ten years following the date that the first customer in the particular Managed Expansion Program project takes service. Once the term of service for a particular Managed Expansion Program project expires, customers will thereafter take service under Residential Non Heating Rate R-3.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading an estimated bill will be rendered to the customer. Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX,	2021, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleek
	* *		Susan L. Fleek
EFFECTIVE:	July 01, 2020	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 344 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Revised Page 52 Fourth Revised Original Page 5255 MEP Residential Heating Rate R-6Superseding Third

MEP Residential Heating Rate R-6

date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX	XX, 2021, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleck
EFFECTIVE:	July 01, 2020	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 345 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 56 MEP Residential Heating Rate R-6

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

-MEP Residential Gas Assistance Program Rate R-7

6 MANAGED EXPANSION PROGRAM GAS ASSISTANCE PROGRAM RESIDENTIAL HEATING RATE: CLASSIFICATION NO. R-7

Availability

This rate is mandatory for customers taking service in a Managed Expansion Program project area who otherwise would have qualified for Gas Assistance Program Residential Heating Rate R-4.

Qualified Programs:

- a. Low Income Home Energy Assistance Program (LIHEAP)
- b. Electric Assistance Program (EAP)
- c. Supplemental Security Income Program
- d. Women, Infants and Children Program
- e. Commodity Surplus Foods Program (for women, infants and children)
- f. Elderly Commodity Surplus Foods Program
- g. Temporary Aid to Needy Families Program
- h. Housing Choice Voucher Program (also known as Section 8)
- i. Head Start Program
- j. Aid to the Permanently and Totally Disabled Program
- k. Aid to the Needy Blind Program
- 1. Old Age Assistance Program
- m. Food Stamps Program
- n. Any successor program of a-m

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Winter Customer Charge Per Meter: \$.3693-3670 per day or \$11.08-01 per 30 day month

Winter Period: All therms per 30 day month at \$0.4060-4279 per therm

Summer Customer Charge Per Meter: \$.6717-6670 per day or \$20.1520.01 per 30 day month

Summer Period: All therms per 30 day month at \$0.7381-7780 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive.

Cost of Gas Charge

All gas delivered under this rate is subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The

roudman
roudman
ent
105
L. Fleck
L. Fleck
ent

Docket No. DG 21-130 Exhibit 26 Page 347 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES Seventh Revised Original Page 5457 Superseding Sixth Revised Page 54

-MEP Residential Gas Assistance Program Rate R-7

delivery charges presented above are exclusive of these charges. Refer to Page 92 of this Tariff for firm rate schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	A 4 . H MINIGO I NI VYVVV I IVVV VV	001 : D 1 (N	DC 20 105
	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2	021, in Docket N	o. DG 20-105
ISSUED:	June 4, 2021	ISSUED BY:	/s/Susan L. Fleck
			Susan L. Fleck
EFFECTIVE:	November 01, 2020	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 348 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 58 MEP Residential Gas Assistance Program Heating Rate R-7

when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

Service under each Managed Expansion Program project will have a term of ten years. Customers initiating service under this rate must take service hereunder until ten years following the date that the first customer in the particular Managed Expansion Program project takes service. Once the term of service for a particular Managed Expansion Program project expires, customers will thereafter take service under Gas Assistance Program Residential Heating Rate R-4.

For those customers qualifying for the program this rate R-7 shall apply for a six month winter period. On the date that the six month winter period expires, eligibility for this rate shall expire unless the customer provides the Company with evidence that the customer continues to be eligible for one or more qualifying programs. When the Rate R-7 expires, the rate on each account shall revert back to the non-gas assistance program Residential Heating Rate, R-6. Customers whose eligibility for the program is based on their having qualified for LIHEAP shall be eligible for this rate retroactive to November 1 of the heating season in which they qualified. Eligibility for such customers shall expire the following April 30, subject to their re-qualifying through receipt of LIHEAP or other benefits as set forth above.

Eligibility shall be determined based on the reasonable discretion of the Company subject to verification of heating usage.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 349 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Revised Page 56 Fifth Revised Original Page 5659 Commercial/Industrial Rate G-41 Superseding Fourth

Commercial/Industrial Rate G-41

7 COMMERCIAL/INDUSTRIAL SERVICE: LOW ANNUAL USE, HIGH WINTER USE RATE CLASSIFICATION NO. G-41

Availability

This rate is available for commercial, industrial and public authority customers in locations served by the Company's mains and for which the Company's facilities are adequate. A customer receiving service under this rate must have annual usage less than or equal to 10,000 therms and a Winter Period usage greater than or equal to 67% of annual usage as determined by the Company's records and procedures.

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter:\$\frac{1.91542.22371.9850}{2.22371.9850} per day or \$\frac{57.4666.7159.55}{9} per 30 day month

Winter Period: First 100* therms per 30 day month at \$0.4711 54334848 per therm

All over 100 therms per 30 day month at \$0.3165 36503309 per therm

Summer Period: First 20* therms per 30 day month at \$0.4711 <u>54334848</u> per therm

All over 20 therms per 30 day month at \$0.3165 36503309 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to Page 90 or 91 of this Tariff for firm rate schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charge presented above is exclusive of these charges. Refer to the Firm Rate Schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX,	2021, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleck
			Susan L. Fleek
EFFECTIVE:	July 01, 2020	TITLE:	President

^{*}The number of therms billed in the first block will be calculated by multiplying the therms in the first block of the rate by a fraction the numerator of which is the number of days in the billing period and the denominator of which is 30.

Docket No. DG 21-130 Exhibit 26 Page 350 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-<u>11</u> GAS LIBERTY UTILITIES Revised Page 56 Fifth Revised Original Page 5659 Commercial/Industrial Rate G-41 Superseding Fourth

Commercial/Industrial Rate G-41

Terms and Conditions

U.S. Department of Labor Standard Industry Classification Codes will determine eligibility for this tariff.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX	XX, 2021, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleek
	•		Susan L. Fleck
EFFECTIVE:	July 01, 2020	TITLE:	President
	• '		

Docket No. DG 21-130 Exhibit 26 Page 351 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 60 Commercial/Industrial Rate G-41

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman
EFFECTIVE: August 1, 2021 TITLE: President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Docket No. DG 21-130 Exhibit 26 Page 352 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Revised Page 58 Fourth Revised Original Page 5861 Commercial/Industrial Rate G-42 Superseding Third

Commercial/Industrial Rate G-42

8 COMMERCIAL/INDUSTRIAL SERVICE: MEDIUM ANNUAL USE, HIGH WINTER USE RATE CLASSIFICATION NO. G-42

Availability

This rate is for commercial, industrial and public authority customers in locations served by the Company's mains and for which the Company's facilities are adequate. A customer receiving service under this rate must have annual usage greater than 10,000 therms and less than or equal to 100,000 therms and a Winter Period usage greater than or equal to 67% of annual usage as determined by the Company's records and procedures.

Character of Service

Natural gas or equivalent will be supplied at a heat content of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter:\$5.7464<u>6.67135.9537</u> per day or \$172.39<u>200178.61.14</u> per 30 day month

Winter Period: First 1000* therms per 30 day month at \$0.4284 49994409 per therm

All over 1000 therms per 30 day month at \$0.2855 33312988 per therm

Summer Period: First 400* therms per 30 day month at \$0.4284 49994409 per therm

All over 400 therms per 30 day month at \$0.2855 33312988 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charges presented above are exclusive of these charges. Refer to Page 90 or 91 of this Tariff for firm rate schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX	, 2021, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleek
	<u> </u>		Susan L. Fleck
EFFECTIVE:	July 01, 2020	TITLE:	President

^{*}The number of therms billed in the first block will be calculated by multiplying the therms in the first block of the rate by a fraction the numerator of which is the number of days in the billing period and the denominator of which is 30.

Docket No. DG 21-130 Exhibit 26 Page 353 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Revised Page 58 Fourth Revised Original Page 5861 Commercial/Industrial Rate G-42 Superseding Third

Commercial/Industrial Rate G-42

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	-		
	Authorized by NHPUC Order No. XX,XXX dated XXX X	X, 2021, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleek
	• *		Susan L. Fleck
EFFECTIVE:	July 01, 2020	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 354 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 62 Commercial/Industrial Rate G-42

Terms and Conditions

Dual fuel customers may be required to sign annual contracts with minimum usage requirements in order to qualify for service under this tariff. U.S. Department of Labor Standard Industry Classification Codes will determine eligibility for this tariff.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman
Neil Proudman

EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 355 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Revised Page 60 Fourth Revised Original Page 6063 Commercial/Industrial Rate G-43 Superseding Third

Commercial/Industrial Rate G-43

9 COMMERCIAL/INDUSTRIAL SERVICE: HIGH ANNUAL USE, HIGH WINTER USE RATE CLASSIFICATION NO. G-43

Availability

This rate is for commercial, industrial and public authority customers in locations served by the Company's mains and for which the Company's facilities are adequate. A customer receiving service under this rate must have annual usage greater than 100,000 therms and a Winter Period usage greater than or equal to 67% of annual usage as determined by the Company's records and procedures.

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet. Should the customer's consumption fail to meet the availability requirements for this rate, the customer's service will be transferred to the otherwise applicable tariff as described under the terms and conditions of this tariff.

Delivery Charge

Customer Charge Per Meter:\$24.661028.630325.5170 per day or \$739.83858.9765.514 per 30 day month

Winter Period: All therms per 30 day month at \$0.2633_.27173126 per therm

Summer Period: All therms per 30 day month at \$0.1204_.1429.1295 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to Page 90 or 91 of this Tariff for firm rate schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charges presented above are exclusive of these charges. Refer to the Firm Rate Schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2	021, in Docket N	lo. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleek
EFFECTIVE:	July 01, 2020	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 356 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Revised Page 60 Fourth Revised Original Page 6063 Commercial/Industrial Rate G-43Superseding Third

Commercial/Industrial Rate G-43

To be eligible for this service, a customer must sign a contract for a one year period, which contract shall include the authority for the Company to monitor the customer's continued qualification for this service. In the event that the customer fails to meet the eligibility criteria set forth in the availability section of this schedule based on a monthly evaluation employing the most recent twelve (12) month period, the Company may require that the customer be billed prospectively under an alternative rate subject to the terms of the customer's Service Agreement. The Service Agreement may contain limitations as to maximum hourly,

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2	021, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleek
	• •		Susan L. Fleck
EFFECTIVE:	July 01, 2020	TITLE:	President
	3		

Docket No. DG 21-130 Exhibit 26 Page 357 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 64 Commercial/Industrial Rate G-43

daily, or monthly consumption, provisions for charges for excess usage, and other terms and conditions of service.

The customer shall declare maximum seasonal demands and estimated seasonal volumes at the time application for service is made. These declarations shall be updated annually, by August 1.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent ($1\frac{1}{2}$ %) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 358 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Revised Page 62 Fourth Revised Original Page 6265 Commercial/Industrial Rate G-44Superseding Third

Commercial/Industrial Rate G-44

10 MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: LOW ANNUAL USE, HIGH WINTER USE RATE CLASSIFICATION NO. G-44

Availability

This rate is Mandatory for customers taking service in a Managed Expansion Program project area who otherwise would have qualified for Commercial/Industrial Rate G-41.

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter:\$2.4896 89072.5803 per day or \$74.6986.7277.41 per 30 day month

Winter Period: First 100* therms per 30 day month at \$0.6126 70636302 per therm

All over 100 therms per 30 day month at \$0.4114 47454302 per therm

Summer Period: First 20* therms per 30 day month at \$0.6126 70636302 per therm

All over 20 therms per 30 day month at \$0.4114 47454302 per therm

*The number of therms billed in the first block will be calculated by multiplying the therms in the first block of the rate by a fraction the numerator of which is the number of days in the billing period and the denominator of which is 30.

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charge presented above is exclusive of these charges. Refer to Page 92 of this Tariff for firm rate schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX,	2021, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleck
EFFECTIVE:	July 01, 2020	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 359 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-<u>11</u> GAS LIBERTY UTILITIES Revised Page 62 Fourth Revised Original Page 6265 Commercial/Industrial Rate G-44Superseding Third

Commercial/Industrial Rate G-44

Terms and Conditions

U.S. Department of Labor Standard Industry Classification Codes will determine eligibility for this tariff.

Service under each Managed Expansion Program project will have a term of ten years. Customers initiating service under this rate must take service hereunder until ten years following the date that the first

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2	021 in Docket N	o DG 20 105
	ramonzea o y ram de draei rad. Mitgriffi antea mini mit, z	021, III DOCKET IV	0. DG 20-103
ISSUED:	July 13, 2020	ISSUED BY:	
	July 12, 2020	ICCLIED DV.	/s/Susan I Flook
	July 12, 2020	ICCLIED DV.	/s/Susan L. Fleek

Docket No. DG 21-130 Exhibit 26 Page 360 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 66 Commercial/Industrial Rate G-44

customer in the particular Managed Expansion Program project takes service. Once the term of service for a particular Managed Expansion Program project expires, customers will thereafter take service under Commercial/Industrial Rate G-41.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman
EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 361 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Third Revised Page 64 Fourth RevisedOriginal Page 6467
MEP Commercial/Industrial Rate G-45Superseding

MEP Commercial/Industrial Rate G 45

11 MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: MEDIUM ANNUAL USE, HIGH WINTER USE RATE CLASSIFICATION NO. G-45

Availability

This rate is mandatory for customers taking service in a Managed Expansion Program project area who otherwise would have qualified for Commercial/Industrial Rate G-42.

Character of Service

Natural gas or equivalent will be supplied at a heat content of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter:\$7.4703<u>8.67277.7397</u> per day or \$224.11260.18232.19 per 30 day month

Winter Period: First 1000* therms per 30 day month at \$0.5569 64985732 per therm

All over 1000 therms per 30 day month at \$0.3711.43303884 per therm

Summer Period: First 400* therms per 30 day month at \$0.5569 64985732 per therm

All over 400 therms per 30 day month at \$0.3711 43303884 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charges presented above are exclusive of these charges. Refer to Page 92 of this Tariff for firm rate schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX	1, 2021, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleck
EFFECTIVE:	July 01, 2020	TITLE:	President

^{*}The number of therms billed in the first block will be calculated by multiplying the therms in the first block of the rate by a fraction the numerator of which is the number of days in the billing period and the denominator of which is 30.

Docket No. DG 21-130 Exhibit 26 Page 362 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES <u>Third Revised Page 64</u> Fourth Revised Original Page 6467 MEP Commercial/Industrial Rate G-45Superseding

MEP Commercial/Industrial Rate G 45

the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

Dual fuel customers may be required to sign annual contracts with minimum usage requirements in order to qualify for service under this tariff. U.S. Department of Labor Standard Industry Classification Codes will determine eligibility for this tariff.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
ISSUED:	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 20 July 13, 2020	021, in Docket No ISSUED BY:	/s/Susan L. Fleek
EFFECTIVE:	July 01, 2020	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 363 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES <u>Third Revised Page 64</u> Fourth Revised Original Page 6467 MEP Commercial/Industrial Rate G-45Superseding

MEP Commercial/Industrial Rate G-45

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 20	21, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleek
	<u> </u>		Susan L. Fleck
EFFECTIVE:	July 01, 2020	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 364 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 68 MEP Commercial/Industrial Rate G-45

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Service under each Managed Expansion Program project will have a term of ten years. Customers initiating service under this rate must take service hereunder until ten years following the date that the first customer in the particular Managed Expansion Program project takes service. Once the term of service for a particular Managed Expansion Program project expires, customers will thereafter take service under Commercial/Industrial Rate G-42.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 365 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Third Revised Page 66 Fourth Revised Original Page 6669
MEP Commercial/Industrial Rate G-46Superseding

MEP Commercial/Industrial Rate G-46

12 MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: HIGH ANNUAL USE, HIGH WINTER USE RATE CLASSIFICATION NO. G-46

Availability

This rate is mandatory for customers taking service in a Managed Expansion Program project area who otherwise would have qualified for Commercial/Industrial Rate G-43.

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet. Should the customer's consumption fail to meet the availability requirements for this rate, the customer's service will be transferred to the otherwise applicable tariff as described under the terms and conditions of this tariff.

Delivery Charge

Customer Charge Per Meter:\$32.0594 37.219433.1720 per day or \$961.781,116.58995.16 per 30 day month

Winter Period: All therms per 30 day month at \$0.3423 40643532 per therm

Summer Period: All therms per 30 day month at \$0.1565 18581684 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to Page 92 of this Tariff for firm rate schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charges presented above are exclusive of these charges. Refer to the Firm Rate Schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

To be eligible for this service, a customer must sign a contract for a one year period, which contract shall include the authority for the Company to monitor the customer's continued qualification for this service. In

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX,	2021, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleck
	• /		Susan L. Fleek
EFFECTIVE:	July 01, 2020	TITLE	President

Docket No. DG 21-130 Exhibit 26 Page 366 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Third Revised Page 66 Fourth Revised Original Page 6669
MEP Commercial/Industrial Rate G-46Superseding

MEP Commercial/Industrial Rate G-46

the event that the customer fails to meet the eligibility criteria set forth in the availability section of this schedule based on a monthly evaluation employing the most recent twelve (12) month period, the Company may require that the customer be billed prospectively under an alternative rate subject to the terms of the customer's Service Agreement. The Service Agreement may contain limitations as to maximum hourly, daily, or monthly consumption, provisions for charges for excess usage, and other terms and conditions of service.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 20	21, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/-/C T E11-
	tuly 12, 2020	IDDCLD DI.	/S/Susan L. Fleck
	04.5 15,2020	ISSCED D1.	Susan L. Fleek
EFFECTIVE:	July 01, 2020	TITLE:	Susan I. Flook

Docket No. DG 21-130 Exhibit 26 Page 367 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 70 MEP Commercial/Industrial Rate G-46

Service under each Managed Expansion Program project will have a term of ten years. Customers initiating service under this rate must take service hereunder until ten years following the date that the first customer in the particular Managed Expansion Program project takes service. Once the term of service for a particular Managed Expansion Program project expires, customers will thereafter take service under Commercial/Industrial Rate G-43.

The customer shall declare maximum seasonal demands and estimated seasonal volumes at the time application for service is made. These declarations shall be updated annually, by August 1.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent ($1\frac{1}{2}$ %) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman
EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 368 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Revised Page 68 Fifth Revised Original Page 6871 Commercial/Industrial Rate G-51 Superseding Fourth

Commercial/Industrial Rate G-51

13 COMMERCIAL/INDUSTRIAL SERVICE: LOW ANNUAL USE, LOW WINTER USE RATE CLASSIFICATION NO. G-51

Availability

This rate is for commercial, industrial and public authority customers in locations served by the Company's mains and for which the Company's facilities are adequate. A customer receiving service under this rate must have annual usage less than or equal to 10,000 therms and a Winter Period usage less than 67% of annual usage as determined by the Company's records and procedures.

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter:\$1.91532.22371.9857 per day or \$57.4666.7159.57 per 30 day month

Winter Period: First 100* therms per 30 day month at \$0.2839 34082920 per therm

Summer Period: First 100* therms per 30 day month at \$0.2839 34082920 per therm

All over 100 therms per 30 day month at \$0.1846 <u>2216</u>1934 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charges presented above are exclusive of these charges. Refer to Page 90 or 91 of this Tariff for firm rate schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is made in addition to all other charges. The meter account charge is \$20.00 when the

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX,	2021, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleek
	• • •		Susan L. Fleck
EFFECTIVE:	July 01, 2020	TITLE:	President

^{*}The number of therms billed in the first block will be calculated by multiplying the therms in the first block of the rate by a fraction the numerator of which is the number of days in the billing period and the denominator of which is 30.

Docket No. DG 21-130 Exhibit 26 Page 369 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES Revised Page 68 Fifth Revised Original Page 6871 Commercial/Industrial Rate G-51 Superseding Fourth

Commercial/Industrial Rate G-51

visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00

Terms and Conditions

Eligibility shall be based on the reasonable discretion of the Company and subject to verification of heating usage. U.S. Department of Labor Standard Industry Classification Code will determine eligibility for this

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 20	021, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleek
-			Susan L. Fleck
EFFECTIVE:	July 01, 2020	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 370 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 72 Commercial/Industrial Rate G-51

tariff. Dual fuel customers may be required to sign annual contracts with minimum usage requirements in order to qualify for service under this tariff.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: /sylveil.proudman.ne

Docket No. DG 21-130 Exhibit 26 Page 371 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Revised Page 70 Fourth Revised Original Page 7073 Commercial/Industrial Rate G-52Superseding Third

Commercial/Industrial Rate G-52

14 COMMERCIAL/INDUSTRIAL SERVICE: MEDIUM ANNUAL USE, LOW WINTER USE RATE CLASSIFICATION NO. G-52

Availability

This rate is for commercial, industrial and public authority customers in locations served by the Company's mains and for which the Company's facilities are adequate. A customer receiving service under this rate must have annual usage greater than 10,000 therms and less than or equal to 100,000 therms and a Winter Period usage less than 67% of annual usage as determined by the Company's records and procedures.

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet. Should the customer's consumption fail to meet the availability requirements for this rate, the customer's service will be transferred to the otherwise applicable tariff as described under the terms and conditions of this tariff.

Delivery Charge

Customer Charge Per Meter:\$5.7463<u>6.6713</u>5.9497 per day or \$172.39200.14178.49 per 30 day month

Winter Period: First 1000* therms per 30 day month at \$0.2439 29262515 per therm

All over 1000 therms per 30 day month at \$0.1624 <u>19481704</u> per therm

Summer Period: First 1000* therms per 30 day month at \$0.1767.21201846 per therm

All over 1000 therms per 30 day month at \$0.1004-12051087 per therm

*The number of therms billed in the first block will be calculated by multiplying the therms in the first block of the rate by a fraction the numerator of which is the number of days in the billing period and the denominator of which is 30.

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charge presented above is exclusive of these charges. Refer to Page 90 or 91 of this Tariff for firm rate schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 20	021, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleek
-			Susan L. Fleck
EFFECTIVE:	July 01, 2020	TITLE:	Sugan I Flack

Docket No. DG 21-130 Exhibit 26 Page 372 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10 <u>11</u> GAS LIBERTY UTILITIES Revised Page 70 Fourth Revised Original Page 7073 Commercial/Industrial Rate G-52 Superseding Third

Commercial/Industrial Rate G-52

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX,	2021, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleck
EFFECTIVE:	July 01, 2020	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 373 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 74 Commercial/Industrial Rate G-52

Terms and Conditions

To be eligible for this service, a customer must sign a contract for a one year period, which contract shall include the authority for the Company to monitor the customer's continued qualification for this service. In the event that the customer fails to meet the eligibility criteria set forth in the availability section of this schedule based on a monthly evaluation employing the most recent twelve (12) month period, the Company may require that the customer be billed prospectively under an alternative rate subject to the terms of the customer's Service Agreement. The Service Agreement may contain limitations as to maximum hourly, daily, or monthly consumption, provisions for charges for excess usage, and other terms and conditions of service.

The customer shall declare maximum seasonal demands and estimated seasonal volumes at the time application for service is made. These declarations shall be updated annually, by August 1.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 374 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Revised Page 72 Fourth Revised Original Page 7275 Commercial/Industrial Rate G-53Superseding Third

Commercial/Industrial Rate G-53

15 COMMERCIAL/INDUSTRIAL SERVICE: HIGH ANNUAL USE, LOAD FACTOR LESS THAN 90% RATE CLASSIFICATION NO. G-53

Availability

This rate is for commercial, industrial and public authority customers in locations served by the Company's mains and for which the Company's facilities are adequate. A customer receiving service under this rate must have annual usage greater than 100,000 therms, a Winter Period usage less than 67% of annual usage, and a 12 month average usage less than 90% of the average usage of December, January and February as determined by the Company's records and procedures.

Character of Service

Natural gas or equivalent will be supplied at a heat content value of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter:\$25.379729.464726.2957 per day or \$761.39883.94788.87 per 30 day month

Winter Period: All therms per 30 day month at \$0.1705 20921758 per therm

Summer Period: All therms per 30 day month at \$0.0818 10030875 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charge presented above is exclusive of these charges. Refer to Page 90 or 91 of this Tariff for firm rate schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

To be eligible for this service, a customer must sign a contract for a one year period, which contract shall include the authority for the Company to monitor the customer's continued qualification for this service. In

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX.	, 2021, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleck
	· ·		Susan L. Fleck
EFFECTIVE:	July 01, 2020	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 375 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10 <u>11</u> GAS LIBERTY UTILITIES Revised Page 72 Fourth Revised Original Page 7275 Commercial/Industrial Rate G-53 Superseding Third

Commercial/Industrial Rate G-53

the event that the customer fails to meet the eligibility criteria set forth in the availability section of this schedule based on a monthly evaluation employing the most recent twelve (12) month period, the Company may require that the customer be billed prospectively under an alternative rate subject to the terms of the customer's Service Agreement. The Service Agreement may contain limitations as to maximum hourly,

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	-		
	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2	2021, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleck
EFFECTIVE:	Index 01 2020	TITLE.	Proceedant

Docket No. DG 21-130 Exhibit 26 Page 376 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 76 Commercial/Industrial Rate G-53

daily, or monthly consumption, provisions for charges for excess usage, and other terms and conditions of service.

The customer shall declare maximum seasonal demands and estimated seasonal volumes at the time application for service is made. These declarations shall be updated annually, by August 1.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent $(1\frac{1}{2}\%)$ per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 377 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Revised Page 74 Fourth Revised Original Page 7477 Commercial/Industrial Rate G-54Superseding Third

Commercial/Industrial Rate G-54

16 COMMERCIAL/INDUSTRIAL SERVICE: HIGH ANNUAL USE, LOAD FACTOR GREATER THAN 90% RATE CLASSIFICATION NO. G-54

Availability

This rate is for commercial, industrial and public authority customers in locations served by the Company's mains and for which the Company's facilities are adequate. A customer receiving service under this rate must have annual usage greater than 100,000 therms, a Winter Period usage less than 67% of annual usage, and a 12 month average usage greater than or equal to 90% of the average usage of December, January and February as determined by the Company's records and procedures.

Character of Service

Natural gas or equivalent will be supplied at a heat content value of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter:\$25.379729.464726.2957 per day or \$761.39883.94788.87 per 30 day month

Winter Period: All therms per 30 day month at \$0.0650 09000670 per therm

Summer Period: All therms per 30 day month at \$0.0353 04880374 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charge presented above is exclusive of these charges. Refer to Page 90 or 91 of this Tariff for firm rate schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

To be eligible for this service, a customer must sign a contract for a one year period, which contract shall include the authority for the Company to monitor the customer's continued qualification for this service. In

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX,	2021, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleck
	<u> </u>		Susan L. Fleck
EFFECTIVE:	July 01, 2020	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 378 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-<u>11</u> GAS LIBERTY UTILITIES Revised Page 74 Fourth Revised Original Page 7477 Commercial/Industrial Rate G-54Superseding Third

Commercial/Industrial Rate G-54

the event that the customer fails to meet the eligibility criteria set forth in the availability section of this schedule based on a monthly evaluation employing the most recent twelve (12) month period, the Company may require that the customer be billed prospectively under an alternative rate subject to the terms of the customer's Service Agreement. The Service Agreement may contain limitations as to maximum hourly,

eil Proudman
eil Proudman
esident
20-105
isan L. Fleck
ısan L. Fleck
esident

Docket No. DG 21-130 Exhibit 26 Page 379 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 78 Commercial/Industrial Rate G-54

daily, or monthly consumption, provisions for charges for excess usage, and other terms and conditions of service.

The customer shall declare maximum seasonal demands and estimated seasonal volumes at the time application for service is made. These declarations shall be updated annually, by August 1.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent $(1\frac{1}{2}\%)$ per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 380 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Third Revised Page 76 Fourth Revised Original Page 7679
MEP Commercial/Industrial Rate G-55Superseding

MEP Commercial/Industrial Rate G-55

17 MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: LOW ANNUAL USE, LOW WINTER USE RATE CLASSIFICATION NO. G-55

Availability

This rate is mandatory for customers taking service in a Managed Expansion Program project area who otherwise would have qualified for Commercial/Industrial Rate G-51.

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter:\$2.48968906 2.5803 per day or \$74.6986.7277.41 per 30 day month

Winter Period: First 100* therms per 30 day month at \$0.3691 44303796 per therm

All over 100 therms per 30 day month at \$0.2400 28812514 per therm

Summer Period: First 100* therms per 30 day month at \$0.3691 44303796 per therm

All over 100 therms per 30 day month at \$0.2400 28812514 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charges presented above are exclusive of these charges. Refer to Page 92 of this Tariff for firm rate schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is made in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00

Terms and Conditions

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	_		
	Authorized by NHPUC Order No. XX,XXX dated X	XX XX, 2021, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleck
EFFECTIVE:	July 01, 2020	TITLE:	President

^{*}The number of therms billed in the first block will be calculated by multiplying the therms in the first block of the rate by a fraction the numerator of which is the number of days in the billing period and the denominator of which is 30.

Docket No. DG 21-130 Exhibit 26 Page 381 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Third Revised Page 76 Fourth Revised Original Page 7679
MEP Commercial/Industrial Rate G-55Superseding

MEP Commercial/Industrial Rate G-55

Eligibility shall be based on the reasonable discretion of the Company and subject to verification of heating usage. U.S. Department of Labor Standard Industry Classification Code will determine eligibility for this tariff. Dual fuel customers may be required to sign annual contracts with minimum usage requirements in order to qualify for service under this tariff.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2	021, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleek
	• •		Susan L. Fleck
EFFECTIVE:	July 01, 2020	TITLE:	President
	3		

Docket No. DG 21-130 Exhibit 26 Page 382 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 80 MEP Commercial/Industrial Rate G-55

Service under each Managed Expansion Program project will have a term of ten years. Customers initiating service under this rate must take service hereunder until ten years following the date that the first customer in the particular Managed Expansion Program project takes service. Once the term of service for a particular Managed Expansion Program project expires, customers will thereafter take service under Commercial/Industrial Rate G-51.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 383 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Third Revised Page 78 Fourth Revised Original Page 7881

MEP Commercial/Industrial Rate G-56 Superseding

MEP Commercial/Industrial Rate G-56

18 MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: MEDIUM ANNUAL USE, LOW <u>WINTER USE RATE</u> CLASSIFICATION NO. G-56

Availability

This rate is mandatory for customers taking service in a Managed Expansion Program project area who otherwise would have qualified for Commercial/Industrial Rate G-52.

Character of Service

Natural gas or equivalent will be supplied at a thermal content of nominally one (1) therm in each one hundred (100) cubic feet. Should the customer's consumption fail to meet the availability requirements for this rate, the customer's service will be transferred to the otherwise applicable tariff as described under the terms and conditions of this tariff.

Delivery Charge

Customer Charge Per Meter: $\$7.4703\underline{8.67277.7397}$ per day or $\$224.11\underline{260.18232.19}$ per 30 day month

Winter Period: First 1000* therms per 30 day month at \$0.3171 38043270 per therm

All over 1000 therms per 30 day month at \$0.2111 25332215 per therm

Summer Period: First 1000* therms per 30 day month at \$0.2297 27562400 per therm

All over 1000 therms per 30 day month at \$0.1304 15661413 per therm

*The number of therms billed in the first block will be calculated by multiplying the therms in the first block of the rate by a fraction the numerator of which is the number of days in the billing period and the denominator of which is 30.

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charge presented above is exclusive of these charges. Refer to Page 92 of this Tariff for firm rate schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Authorized by NHPUC Order No. 26,374 dated June 30, 2020, in Docket No. DG 20-049

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
ISSUED:	Authorized by NHPUC Order No. XX,XXX dated XXX July 13, 2020		o. DG 20-105 /s/Susan L. Fleck Susan L. Fleck
EFFECTIVE:	July 01, 2020	TITLE:	President

Formatted: Font: Italic

Docket No. DG 21-130 Exhibit 26 Page 384 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Third Revised Page 78

Fourth RevisedOriginal Page 7881 MEP Commercial/Industrial Rate G-56Superseding

MEP Commercial/Industrial Rate G-56

Terms and Conditions
To be eligible for this service, a customer must sign a contract for a one year period, which contract shall include the authority for the Company to monitor the customer's continued qualification for this service. In

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX	XX, 2021, in Docket No	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleck
			Susan L. Fleck
EFFECTIVE:	July 01, 2020	TITLE:	President
	Authorized by NHPUC Order No. 26,374 dated June 3	0, 2020, in Docket No. I	OG 20-049

Docket No. DG 21-130 Exhibit 26 Page 385 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10 11 GAS LIBERTY UTILITIES Original Page 82 MEP Commercial/Industrial Rate G-56

the event that the customer fails to meet the eligibility criteria set forth in the availability section of this schedule based on a monthly evaluation employing the most recent twelve (12) month period, the Company may require that the customer be billed prospectively under an alternative rate subject to the terms of the customer's Service Agreement. The Service Agreement may contain limitations as to maximum hourly, daily, or monthly consumption, provisions for charges for excess usage, and other terms and conditions of service.

Service under each Managed Expansion Program project will have a term of ten years. Customers initiating service under this rate must take service hereunder until ten years following the date that the first customer in the particular Managed Expansion Program project takes service. Once the term of service for a particular Managed Expansion Program project expires, customers will thereafter take service under Commercial/Industrial Rate G-52.

The customer shall declare maximum seasonal demands and estimated seasonal volumes at the time application for service is made. These declarations shall be updated annually, by August 1.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 386 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES Fourth Revised Page 80 Fifth Revised Original Page 8083 MEP Commercial/Industrial Rate G-57Superseding

MEP Commercial/Industrial Rate G-57

19 MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: HIGH ANNUAL USE, LOAD FACTOR LESS THAN 90% RATE CLASSIFICATION NO. G-57

Availability

This rate is mandatory for customers taking service in a Managed Expansion Program project area who otherwise would have qualified for Commercial/Industrial Rate G-53.

Character of Service

Natural gas or equivalent will be supplied at a heat content value of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter:\$32.993438.304134.1843 per day or \$989.801,149.1025.5312 per 30 day month

Winter Period: All therms per 30 day month at \$0.2216 27202286 per therm

Summer Period: All therms per 30 day month at \$0.1063 13051138 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to Page 92 of this Tariff for firm rate schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charge presented above is exclusive of these charges. Refer to the Firm Rate Schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

To be eligible for this service, a customer must sign a contract for a one year period, which contract shall include the authority for the Company to monitor the customer's continued qualification for this service. In the event that the customer fails to meet the eligibility criteria set forth in the availability section of this schedule based on a monthly evaluation employing the most recent twelve (12) month period, the Company may require that the customer be billed prospectively under an alternative rate subject to the terms of the

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX	X, 2021, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleck
EFFECTIVE:	Into 01, 2020	TITI E.	Proceedant
	1-101 2020	TITI E.	

Docket No. DG 21-130 Exhibit 26 Page 387 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES Fourth Revised Page 80 Fifth Revised Original Page 8083
MEP Commercial/Industrial Rate G-57Superseding

MEP Commercial/Industrial Rate G-57

customer's Service Agreement. The Service Agreement may contain limitations as to maximum hourly, daily, or monthly consumption, provisions for charges for excess usage, and other terms and conditions of service.

Service under each Managed Expansion Program project will have a term of ten years. Customers initiating service under this rate must take service hereunder until ten years following the date that the first customer

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2	2021, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleck
EFFECTIVE:	July 01, 2020	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 388 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 84 MEP Commercial/Industrial Rate G-57

in the particular Managed Expansion Program project takes service. Once the term of service for a particular Managed Expansion Program project expires, customers will thereafter take service under Commercial/Industrial Rate G-53.

The customer shall declare maximum seasonal demands and estimated seasonal volumes at the time application for service is made. These declarations shall be updated annually, by August 1.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 389 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Third Revised Page 82 Fourth Revised Original Page 8285
MEP Commercial/Industrial Rate G-58 Superseding

MEP Commercial/Industrial Rate G-58

20 MANAGED EXPANSION PROGRAM COMMERCIAL/INDUSTRIAL SERVICE: HIGH ANNUAL USE, LOAD FACTOR GREATER THAN 90% RATE CLASSIFICATION NO. G-58

Availability

This rate is mandatory for customers taking service in a Managed Expansion Program project area who otherwise would have qualified for Commercial/Industrial Rate G-54.

Character of Service

Natural gas or equivalent will be supplied at a heat content value of nominally one (1) therm in each one hundred (100) cubic feet.

Delivery Charge

Customer Charge Per Meter:\$32.993438.304134.1843 per day or \$989.801,149.121,025.53 per 30 day month

Winter Period: All therms per 30 day month at \$0.0846 1170.0871 per therm

Summer Period: All therms per 30 day month at \$0.0459 06340487 per therm

The above rates shall be adjusted to reflect the recovery of all applicable taxes. The Winter Period shall be the months of November through April inclusive. The Summer Period shall be the months of May through October inclusive.

Supplier Charges

If the customer purchases its gas from a third party, supplier charges will be as agreed upon between the customer and the third party supplier and will be billed directly by the third party supplier. If the customer does not purchase its gas from a third party, the gas supplied by the Company will be subject to a per therm cost of gas rate. The cost of gas rate is not included in the delivery charge presented above. Refer to the Firm Rate Schedules which present both the delivery charge and cost of gas rates.

Other Charges for Delivery Service

The customer must also pay such charges and adjustments as are set forth in the Company's Local Distribution Adjustment Clause, as in effect from time to time and on file with the Commission. The delivery charge presented above is exclusive of these charges. Refer to Page 92 of this Tariff for firm rate schedules which present both the delivery charge and the LDAC rates.

Meter Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a meter location, a meter account charge is incurred in addition to all other charges. The meter account charge is \$20.00 when the visit to the meter location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

To be eligible for this service, a customer must sign a contract for a one year period, which contract shall include the authority for the Company to monitor the customer's continued qualification for this service. In the event that the customer fails to meet the eligibility criteria set forth in the availability section of this schedule based on a monthly evaluation employing the most recent twelve (12) month period, the Company may require that the customer be billed prospectively under an alternative rate subject to the terms of the

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2	021, in Docket N	o. DG 20-105
ISSUED:	July 13, 2020	ISSUED BY:	/s/Susan L. Fleck
			Susan L. Fleek
EFFECTIVE:	July 01, 2020	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 390 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES <u>Third Revised Page 82</u> Fourth Revised Original Page 8285
MEP Commercial/Industrial Rate G-58 Superseding

MEP Commercial/Industrial Rate G-58

customer's Service Agreement. The Service Agreement may contain limitations as to maximum hourly, daily, or monthly consumption, provisions for charges for excess usage, and other terms and conditions of service.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
ISSUED:	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2 July 13, 2020	· ·	o. DG 20-105 /s/Susan L. Fleck
EFFECTIVE:	July 01, 2020	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 391 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 86 Outdoor Gas Lighting

in the particular Managed Expansion Program project takes service. Once the term of service for a particular Managed Expansion Program project expires, customers will thereafter take service under Commercial/Industrial Rate G-54.

The customer shall declare maximum seasonal demands and estimated seasonal volumes at the time application for service is made. These declarations shall be updated annually, by August 1.

Meters are read and bills are presented monthly. In the event a meter reader is unable to obtain a meter reading, an estimated bill will be rendered to the customer.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

21 OUTDOOR GAS LIGHTING

Availability

This rate is available for residential outdoor gas lighting where such service is provided from the Company's existing delivery system to a standard gas light fixture or fixtures, located on the customer's premises, and when it is not feasible to meter such service along with other gas used on the premises and bill the same under the rate in effect for all other services. Service under this rate is available at those locations which were receiving service hereunder as of July 1, 2015, and which have continuously received service hereunder since that date.

Rate Per Light Per Month

\$12.81

The above rates shall be adjusted to reflect the recovery of all applicable taxes.

Account Charge

When the Company establishes or re-establishes a gas service account for a customer at a location, an account charge is incurred in addition to all other charges. The account charge is \$20.00 when the visit to the location is scheduled at the mutual convenience of the Company and the customer. Otherwise, the charge is \$30.00.

Terms and Conditions

Meters are read and bills are presented monthly.

Amounts not paid prior to the due date; normally the next following meter reading date and a date not less than twenty-five (25) days from the date the bill is mailed - are subject to a late payment charge of one and one-half percent (1½%) per month on the unpaid balance - equivalent to an eighteen percent (18%) annual rate. There is a \$15.00 charge for each bad check tendered for payment.

A customer must give at least four (4) days' notice before discontinuance of service and is responsible for all charges through the end of the notice period.

Service under this rate is subject to the rules and regulations and the published tariff, terms and conditions presently effective, or as filed from time to time, with the Commission.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

EFFECTIVE: August 1, 2021 TITLE: President

22 FIRM RATE SCHEDULES - EXCLUDING KEENE CUSTOMERS

	Rates effective		1, 2021 - April 3 2021 - April 30, : Period	0, 2021 2022	Rates Effective August 1, 2021 - October 31, 2021 Rates Effective August 1, 2021 - October 31, 2021 Summer Period								
	Delivery <u>Charge</u>	Cost of Gas Rate Page 96	LDAC Page 102	Total <u>Rate</u>	Delivery <u>Charge</u>	Cost of Gas Rate Page 93	LDAC Page 102	Total <u>Rate</u>					
Residential Non Heating - R-1 Customer Charge per Month per Meter All Therms	\$ 15.50 \$ 15.39 \$ 0.4358 \$ 0.3860	\$ 0.6050 \$ 0.6050	\$ 0.0589 \$ 0.0589	\$ 15.50 \$ 15.39 \$ 1.0997 \$ 1.0499	\$ 15.50 \$ 15.39 \$ 0.4358 \$ 0.3860	\$ 0.3935 \$ 0.3935		\$ 15.50 \$ 15.39 \$ 0.8882 \$ 0.8384					
Residential Heating - R-3 Customer Charge per Month per Meter Size of the first block All Therms	\$ 15.50 \$ 15.39 All Therms \$ 0.5985 \$ 0.5678	\$ 0.6050 \$ 0.6050	\$ 0.0589 \$ 0.0589	\$ 15.50 \$ 15.39 \$ 1.2624 \$ 1.2317	\$ 15.50 \$ 15.39 All Therms \$ 0.5985 \$ 0.5678	\$ 0.3935 \$ 0.3935	5 \$ 0.0589 5 \$ 0.0589	\$ 15.50 \$ 15.39 \$ 1.0509 \$ 1.0202					
Residential Heating - R-4 Customer Charge per Month per Meter Size of the first block All Therms	\$ 8.53 \$ 8.47 All Therms \$ 0.3292 \$ 0.3123	\$ 0.3328 \$ 0.3328	\$ 0.0589 \$ 0.0589	\$ 8.53 \$ 8.47 \$ 0.7209 \$ 0.7040	\$ 15.50 \$ 15.39 All Therms \$ 0.5985 \$ 0.5678	\$ 0.3935 \$ 0.3935		\$ 15.50 \$ 15.39 \$ 1.0509 \$ 1.0202					
Commercial/Industrial - G-41 Customer Charge per Month per Meter Size of the first block Therms in the first block per month at	\$ 57.46 \$ 59.55 100 therms \$ 0.4848	\$ 0.6031	\$ 0.0555	\$ 57.46 \$ 59.55 \$ 1.1434 \$ 1.297	\$ 57.46 \$ 59.55 20 therms \$ 0.4848		5 \$ 0.0555 \$ 0.0555	\$ 57.46 \$ 59.55 \$ 0.9289 \$ 0.9152					
All therms over the first block per month at	\$ 0.3309 \$ 0.3165	\$ 0.6031 \$ 0.6031	\$ 0.0555 \$ 0.0555	\$ 0.9895 \$ 0.9751	\$ 0.3309 \$ 0.3165	\$ 0.3886 \$ 0.3886	\$ 0.0555	\$ 0.7750 \$ 0.7606					
Commercial/Industrial - G-42 Customer Charge per Month per Meter Size of the first block Thems in the first block per month at	\$ 172.39 \$ 178.61 1000 therms \$ 0.4409	\$ 0.6031	\$ 0.0555	\$ 172.39 \$ 178.61 \$ 1.0995	\$ 172.38 \$ 178.61 400 therms \$ 0.4409	\$ 0.3886	s \$ 0.0555	\$ 172.39 \$ 178.61 \$ 0.8850					
All therms over the first block per month at	\$ 0.4284 \$ 0.2988 \$ 0.2855	\$ 0.6031 \$ 0.6031 \$ 0.6031	\$ 0.0555 \$ 0.0555 \$ 0.0555	\$ 0.9574 \$ 0.9441	\$ 0.4284 \$ 0.2988 \$ 0.2855	\$ 0.3886 \$ 0.3886 \$ 0.3886		\$ 0.8725 \$ 0.7429 \$ 0.7296					
Commercial/Industrial - G-43 Customer Charge per Month per Meter All therms over the first block per month at	\$ 739.83 \$ 765.51 \$ 0.2717 \$ 0.2633	\$ 0.6031 \$ 0.6031	\$ 0.0555 \$ 0.0655	\$ 739.83 \$ 765.51 \$ 0.9303 \$ 0.9219	\$ 739.83 \$ 765.51 \$ 0.1295 \$ 0.1204	\$ 0.3886 \$ 0.3886	\$ 0.0555 \$ 0.0555	\$ 739.83 \$ 765.51 \$ 0.5736 \$ 0.5645					
Commercial/Industrial - G-51 Customer Charge per Month per Meter Size of the first block Thems in the first block per month at	\$ 57.46 \$ 59.57 100 therms \$ 0.2920	\$ 0.6139	\$ 0.0555	\$ 57.46 \$ 59.57 \$ 0.9614	\$ 57.46 \$ 59.57 100 therms \$ 0.2920	\$ 0.3999	0.0555	\$ 57.46 \$ 59.57 \$ 0.7474					
All therms over the first block per month at	\$ 0.2839 \$ 0.1934 \$ 0.1846	\$ 0.6139 \$ 0.6139 \$ 0.6139	\$ 0.0555 \$ 0.0555 \$ 0.0555	\$ 0.8628 \$ 0.8540	\$ 0.2839 \$ 0.1934 \$ 0.1846	\$ 0.3999 \$ 0.3999 \$ 0.3999	\$ 0.0555 \$ 0.0555 \$ 0.0555	\$ 0.7393 \$ 0.6488 \$ 0.6400					
Commercial/Industrial - G-52 Customer Charge per Month per Meter Size of the first block Therms in the first block per month at	\$ 172.39 \$ 178.49 1000 therms \$ 0.2515	\$ 0.6139	\$ 0.0555	\$ 172.39 \$ 178.49 \$ 0.9209	\$ 172.39 \$ 178.49 1000 therms \$ 0.1846 \$ 0.1767		\$ 0.0555	\$ 172.38 \$ 178.49 \$ 0.6400 \$ 0.6321					
All therms over the first block per month at	\$ 0.2439 \$ 0.1704 \$ 0.1624	\$ 0.6139 \$ 0.6139	\$ 0.0555 \$ 0.0555	\$ 0.8398 \$ 0.8318	\$ 0.1087 \$ 0.1004	\$ 0.3999 \$ 0.3999	\$ 0.0555	\$ 0.5641 \$ 0.5558					
Commercial/Industrial - G-53 Customer Charge per Month per Meter All therms over the first block per month at	\$ 761.30 \$ 788.87 \$ 0.1758 \$ 0.1705	\$ 0.6139 \$ 0.6139	\$ 0.0555 \$ 0.0555	\$ 761.39 \$ 788.87 \$ 0.8452 \$ 0.8399	\$ 761.39 \$ 788.87 \$ 0.0875 \$ 0.0818	\$ 0.3999 \$ 0.3999		\$ 761.39 \$ 788.87 \$ 0.5429 \$ 0.5372					
Commercial/Industrial - G-54 Customer Charge per Month per Meter All therms over the first block per month at	\$ 761.39 \$ 788.87 \$ 0.0670 \$ 0.0650	\$ 0.6139 \$ 0.6139	\$ 0.0555 \$ 0.0555	\$ 761.39 \$ 788.87 \$ 0.7364 \$ 0.7344	\$ 761.39 \$ 788.87 \$ 0.0374 \$ 0.0353	\$ 0.3999 \$ 0.3999		\$ 761.39 \$ 788.87 \$ 0.4928 \$ 0.4907					

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 20	21, in Docket N	o. DG 20-105
ISSUED:	April 23, 2021	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleck
EFFECTIVE:	May 01, 2021	TITLE:	President

Thirtieth RevisedOriginal Page 8487 Superseding Twenty ninth Revised Page 84

-Firm Rate Schedule

II RATE SCHEDULES FIRM RATE SCHEDULES EXCLUDING KEENE CUSTOMERS

Rates effective March 1, 2021 - April 30, 2021 Winter Period Rates Effective May 1, 2021 - October 31, 2021 Summer Period

Page 97		_			VV	inte	г Репоа			Summer Period							
Second Charge per Month per Meter Second					Gas Rate								Gas Rate				
State Stat	Customer Charge per Month per Meter			\$	0.6050	\$	0.0589					\$	0.3935	\$	0.0589		
Residential Heating - R4 Customer Charge per Month per Meter S	Customer Charge per Month per Meter	\$	15.50					\$	15.50	\$	15.50					\$	15.50
Second Charge per Month per Meter Second		\$	0.5678	\$	0.6050	\$	0.0589	\$	1.2317	\$	0.5678	\$	0.3935	\$	0.0589	\$	1.0202
Commercial/Industrial - G-41	Customer Charge per Month per Meter	\$	8.53					\$	8.53	\$	15.50					\$	15.50
Stize of the first block Stize of the first block per month at Stize of the firs		\$	0.3123	\$	0.3328	\$	0.0589	\$	0.7040	\$	0.5678	\$	0.3935	\$	0.0589	\$	1.0202
Thems in the first block per month at \$ 0.4711 \$ 0.6031 \$ 0.0555 \$ 0.9751 \$ 0.3765 \$ 0.3886 \$ 0.0555 \$ 0.97606	Customer Charge per Month per Meter	\$						\$	57.46	\$						\$	57.46
Commercial/Industrial - G-42		\$		\$	0.6031	\$	0.0555	\$	1.1297	\$		\$	0.3886	\$	0.0555	\$	0.9152
Size of the first block 172.39 \$ 172.3	All therms over the first block per month at	\$	0.3165	\$	0.6031	\$	0.0555	\$	0.9751	\$	0.3165	\$	0.3886	\$	0.0555	\$	0.7606
All therms over the first block per month at \$ 0.2855 \$ 0.6031 \$ 0.0555 \$ 0.9441 \$ 0.2855 \$ 0.3886 \$ 0.0555 \$ 0.7296 \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	Customer Charge per Month per Meter	\$						\$	172.39	\$						\$	172.39
Commercial/Industrial - G-43	Therms in the first block per month at	\$	0.4284	\$	0.6031	\$	0.0555	\$	1.0870	\$					0.0555	\$	
Customer Charge per Month per Meter \$ 739.83 \$ 7	All therms over the first block per month at	\$	0.2855	\$	0.6031	\$	0.0555	\$	0.9441	\$	0.2855	\$	0.3886	\$	0.0555	\$	0.7296
Customer Charge per Month per Meter Size of the first block per month at \$ 0.2839 \$ 0.6139 \$ 0.6555 \$ 0.9553 \$ 0.2839 \$ 0.3999 \$ 0.0555 \$ 0.7393 \$ 0.8146 \$ 0.3999 \$ 0.0555 \$ 0.7393 \$ 0.8146 \$ 0.3999 \$ 0.0555 \$ 0.8400 \$ 0.846 \$ 0.3999 \$ 0.0555 \$ 0.8400 \$ 0.846 \$ 0.3999 \$ 0.0555 \$ 0.8400 \$ 0.846 \$ 0.3999 \$ 0.0555 \$ 0.8400 \$ 0.846 \$ 0.3999 \$ 0.0555 \$ 0.8400 \$ 0.846 \$ 0.3999 \$ 0.0555 \$ 0.8400 \$ 0.846 \$ 0.3999 \$ 0.0555 \$ 0.8400 \$ 0.846 \$ 0.3999 \$ 0.0555 \$ 0.8400 \$ 0.846 \$ 0.3999 \$ 0.0555 \$ 0.8400 \$	Customer Charge per Month per Meter			\$	0.6031	\$	0.0555					\$	0.3886	\$	0.0555		
Therms in the first block per month at \$ 0.2839 \$ 0.6139 \$ 0.0555 \$ 0.9533 \$ 0.2839 \$ 0.3999 \$ 0.0555 \$ 0.7393 All therms over the first block per month at \$ 0.1846 \$ 0.6139 \$ 0.0555 \$ 0.8540 \$ 0.8540 \$ 0.1846 \$ 0.3999 \$ 0.0555 \$ 0.6400 Commercial/Industrial - G-52	Customer Charge per Month per Meter	\$						\$	57.46	\$						\$	57.46
All therms over the first block per month at \$ 0.1846 \$ 0.6139 \$ 0.0555 \$ 0.8540 \$ 0.1846 \$ 0.3999 \$ 0.0555 \$ 0.6400 \$ 0.1846 \$ 0.3999 \$ 0.0555 \$ 0.6400 \$ 0.0555 \$ 0.6550 \$ 0		\$		\$	0.6139	\$	0.0555	\$	0.9533	\$		\$	0.3999	\$	0.0555	\$	0.7393
Customer Charge per Month per Meter Size of the first block 1000 therms 1000 t		\$								\$							
Therms in the first block per month at \$ 0.2439 \$ 0.6139 \$ 0.0555 \$ 0.9133 \$ 0.1767 \$ 0.3999 \$ 0.0555 \$ 0.6321 \$ All therms over the first block per month at \$ 0.1624 \$ 0.6139 \$ 0.0555 \$ 0.8318 \$ 0.1004 \$ 0.3999 \$ 0.0555 \$ 0.6558 \$ 0.655	Customer Charge per Month per Meter	\$						\$	172.39							\$	172.39
All therms over the first block per month at \$ 0.1624 \$ 0.6139 \$ 0.0555 \$ 0.8318 \$ 0.1004 \$ 0.3999 \$ 0.0555 \$ 0.5558 \$ 0.5578 \$ 0.5558 \$ 0.5578 \$ 0.5558 \$ 0.5578 \$ 0.5788 \$ 0.5578 \$ 0		\$		\$	0.6139	\$	0.0555	\$	0.9133			\$	0.3999	\$	0.0555	\$	0.6321
Customer Charge per Month per Meter \$ 761.39 \$ 761.39 \$ 761.39 \$ 761.39 All therms over the first block per month at therms over the first block per month at the first block per mon																	
Commercial/Industrial - G-54 Customer Charge per Month per Meter \$ 761.39 \$ 761.39 \$ 761.39 \$ 761.39	Customer Charge per Month per Meter			e	0.6130	•	0.0555					•	0.2000	•	0.0555		
Customer Charge per Month per Meter \$ 761.39 \$ 761.39 \$ 761.39	All therms over the first block per month at	ф	0.1705	ф	0.6139	ф	0.0555	Э	0.8399	\$	0.0818	Ф	0.3999	Ф	0.0555	Ъ	0.5372
		\$	761 30					\$	761 39	\$	761 30					\$	761 30
	All therms over the first block per month at	-		\$	0.6139	\$	0.0555		0.7344			\$	0.3999	\$	0.0555	-	0.4907

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2	021, in Docket N	o. DG 20-105
ISSUED:	April 23, 2021	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleek
EFFECTIVE:	May 01, 2021	TITLE:	President

Firm Rate Schedule

23 FIRM RATE SCHEDULES - KEENE CUSTOMERS

	Rates E	ffective March	1, 2021 - April r 01, 2021 - Apr l Period				, 2021 - October 1, 2021 - Octobe Period	
	Delivery <u>Charge</u>	Cost of Gas Rate Page 98	LDAC Page 102	Total <u>Rate</u>	Delivery <u>Charge</u>	Cost of Gas Rate Page 94	LDAC Page 102	Total <u>Rate</u>
Residential Non Heating - R-1	\$15.50			\$15.50	\$ 15.50		ş	45.50
Customer Charge per Month per Meter	\$ 15.39	Ø 0.0400	¢ 0.0500	\$ 15.39	\$ 15.39	A 0005	* 0.0500	
All Therms	\$ 0.4358 \$ 0.3860	\$ 0.9492 \$ 1.2816	\$ 0.0589 \$ 0.0589	\$ 1.4439 \$ 1.7265	\$ 0.4358 \$ 0.3860	\$ 1.2665 \$ 1.2665	\$ 0.0589 \$ \$ 0.0589	1.7612
Residential Heating - R-3 Customer Charge per Month per Meter	\$ 15.50 \$ 15.39			\$ 15.50 \$ 15.39	\$ 15.50 \$ 15.39		9	45.50 15.39
Size of the first block	All Therms			φ 15.59	All Therms		`	15.39
	\$ 0.5985	\$ 0.9492	\$ 0.0589	\$ 1.6066	\$ 0.5985	\$ 1.2665	\$ 0.0589	1.9239
	\$	\$ <u>1.2816</u>	\$	\$1.9083	\$	\$ 1.2665	\$ -0.0589	-1.8932
Residential Heating - R-4	\$			\$ <u>8.53</u>	\$15.50		٤	45.50
Customer Charge per Month per Meter	\$ 8.47			\$ 8.47	\$ 15.39		9	
Size of the first block	All Therms				All Therms			
	\$ 0.3292	\$ 0.9492	\$ 0.0589 \$ 0.0589	\$ 1.3373 \$ 1.0760	\$ 0.5985 \$ 0.5678	\$ 1.2665 \$ 1.2665	\$ 0.0589 \$	1.9239
	0.0120	0.7010	0.0000	4 -1.07.00		1.2000	Ψ-0.0000	-1.0002
Commercial/Industrial - G-41	\$ 57.46			\$	\$		4	57.46
Customer Charge per Month per Meter	\$ 59.55			\$ 59.55	\$ 59.55		\$	59.55
Size of the first block Therms in the first block per month at	100 therms \$ 0.4848	\$ 0.9492	\$ 0.0555	\$ 1.4895	20 therms \$ 0.4848	\$ 1.2665	\$ 0.0555	1.8068
	\$ 0.4711	\$ <u>1.2816</u>	\$0.0555	\$ <u>1.8082</u>	\$0.4711	\$ 1.2665	\$ 0.0555	1.7931
All therms over the first block per month at	\$ 0.3309	\$ 0.9492 \$ 1.2816	\$ 0.0555	\$ 1.3356	\$ 0.3309	\$ 1.2665	\$ 0.0555	
	\$ 0.3165	\$ 1.2816	\$0.0555	\$ <u>1.6536</u>	\$0.3165	\$ <u>1.2665</u>	\$ 0.0555	1.6385
Commercial/Industrial - G-42	\$ 172.39			\$ 172.39	\$ 172.39		9	172.39
Customer Charge per Month per Meter	\$ 178.61			\$ 178.61	\$ 178.61		9	178.61
Size of the first block	1000 therms \$ 0.4409		\$ 0.0555	\$ 1,4456	400 therms \$ 0.4409	£ 4.000E	A 0.0555 A	4 7000
Therms in the first block per month at	\$ 0.4409 \$ 0.4284	\$ 0.9492 \$ 1.2816	\$ 0.0555 \$ 0.0555	\$ 1.4456 \$ 1.7655	\$ 0.4409 \$ 0.4284	\$ 1.2665 \$ 1.2665	\$ 0.0555	
All therms over the first block per month at	\$ 0.2988	\$ 0.9492	\$ 0.0555	\$ 1.3035	\$ 0.2988	\$ 1.2665	\$ 0.0555	
	\$ 0.2855	\$ <u>1.2816</u>	\$0.0555	\$1.6226	\$ 0.2855	\$ <u>1.2665</u>	\$0.0555	1.6075
Commercial/Industrial - G-43	\$739.83			\$730.83	\$ 739.83			739.83
Customer Charge per Month per Meter	\$ 765.51			\$ 765.51	\$ 765.51		9	
All therms	\$ 0.2717	\$ 0.9492	\$ 0.0555	\$ 1.2764	\$ 0.1295	\$ 1.2665	\$ 0.0555	
	\$— <u>0.2633</u>	\$ 1.2816	\$ <u>0.0555</u>	\$1.6004	\$	\$ 1.2665	\$ -0.0555	1.4424
Commercial/Industrial - G-51	\$ 			\$ 57.46	\$57.46		٤	57.46
Customer Charge per Month per Meter	\$ 59.57			\$ 59.57	\$ 59.57		\$	59.57
Size of the first block	100 therms				100 therms			
Therms in the first block per month at	\$ 0.2920	\$ 0.9492 \$ 1.2816	\$ 0.0555 \$ 0.0555	\$ 1.2967 \$ 1.6210	\$ 0.2920 \$ 0.2830	\$ 1.2665 \$ 1.2665	\$ 0.0555	1.6140
All therms over the first block per month at	\$ 0.1934	\$ 0.9492	\$ 0.0555	\$ 1.1981	\$ 0.1934	\$ 1.2665	\$ 0.0555	1.5154
	\$	\$ 1.2816	\$	\$1.5217	\$0.1846	\$ 1.2665	\$-0.0555	1.5066
Commercial/Industrial - G-52	\$ <u>172.30</u>			\$ <u>172.30</u>	\$ 172.30			172 30
Customer Charge per Month per Meter	\$ 178.49			\$ 178.49	\$ 178.49		9	,
Size of the first block	1000 therms				1000 therms			
Therms in the first block per month at	\$ 0.2515 \$ 0.2430	\$ 0.9492 \$ 1.2816	\$ 0.0555	\$ 1.2562 \$ 1.5810	\$ 0.1846 \$ 0.1767	\$ 1.2665	\$ 0.0555	1.5066
All therms over the first block per month at	\$ 0.1704	\$ 0.9492	\$ 0.0555	\$ 1.1751	\$ 0.1087	\$ 1.2665	\$ 0.0555	
	\$0.1624	\$ <u>1.2816</u>	\$ 0.0555	\$ <u>1.4995</u>	\$	\$ <u>1.2665</u>	\$ 0.0555	1.4224
On any and all the decated at 10 FO	0 704.00			A 704.00	. 704.00			704.00
Commercial/Industrial - G-53 Customer Charge per Month per Meter	\$ 761.39 \$ 788.87			\$— 761.39 \$ 788.87	\$ 761.39 \$ 788.87		9	761.39 788.87
All therms over the first block per month at	\$ 0.1758	\$ 0.9492	\$ 0.0555	\$ 1.1805	\$ 0.0875	\$ 1.2665	\$ 0.0555	
·	\$	\$ 1.2816	\$0.0555	\$1.5076	\$	\$ <u>1.2665</u>	\$0.0555	1.4038
Commercial/Industrial C 54	\$ 761.30			¢ 761.30	6 764.00			764.00
Commercial/Industrial - G-54 Customer Charge per Month per Meter	\$ 761.39 \$ 788.87			\$ 761.39 \$ 788.87	\$ 761.39 \$ 788.87		9	761.39 788.87
All therms over the first block per month at	\$ 0.0670	\$ 0.9492	\$ 0.0555	\$ 1.0717	\$ 0.0374	\$ 1.2665	\$ 0.0555	1.3594
	\$ <u>0.0650</u>	\$ <u>1.2816</u>	\$ <u>0.0555</u>	\$1.4021	\$ <u>0.0353</u>	\$ 1.2665	\$ <u>-0.0555</u> \$	1.3573

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX	XX, 2021, in Docket N	o. DG 20-105
ISSUED:	May 24, 2021	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleek
EFFECTIVE.	June 01 2021	TITI E.	Proceedant

Thirty secondOriginal Revised Page 8588 Superseding Thirty first Revised Page 85

Firm Rate Schedule

Paid		Rates Effective April 1, 2021 - April 30, 2021 Winter Period			30, 2021	Rates Effective June 1, 2021 - October 31, 2021 Summer Period					
State Stat			•	Gas Rate				•	Gas Rate		
Residential Heating - R-3 Customer Charge per Month per Meter S 15.50 S 1.2816 S 0.0589 S 1.9816 S 1.550 S 1.5	Residential Non Heating - R-1										
Residential Heating - R-3 Customer Charge per Month per Meter S 15.50 S 1.2816 S 0.0589 S 1.9816 S 1.550 S 1.5	Customer Charge per Month per Meter	\$	15.50				\$15.50	\$15.50			\$15.50
Commercial/Industrial - G-42 Customer Charge per Month per Meter Size of the first block per month at Plems (solutions) Size of the first block per month at All thems over the first block per month at All thems over the first block per month at All thems over the first block per month at All thems over the first block per month at All thems over the first block per month at All thems over the first block per month at All thems over the first block per month at All thems over the first block per month at All thems over the first block per month at All thems over the first block per month at All thems over the first block per month at All thems over the first block per month at All thems over the first block per month at All thems over the first block per month at All thems over the first block per month at All thems over the first block per month at Sol 2865 \$1.2816 \$0.0555 \$1.6036	All therms		0.3860	\$1.2816	\$ 0.0589		\$1.7265	\$0.3860	\$1.2665	\$0.0589	\$1.7114
Residential Heating - R4	Residential Heating - R-3										
Residential Heating - R4		Ś	15 50				\$15.50	\$15.50			\$15.50
Commercial/Industrial - G-41 Customer Charge per Month per Meter S 0.3123 \$ 0.7048 \$ 0.0589 \$ 1.0760 \$ 0.5678 \$ 1.2665 \$ 0.0589 \$ 1.8932 \$ 0.0589 \$ 1.0760 \$ 0.5678 \$ 1.2665 \$ 0.0589 \$ 1.8932 \$ 0.0589 \$ 1.0760 \$ 0.5678 \$ 1.2665 \$ 0.0589 \$ 1.8932 \$ 0.0589 \$ 1.0760 \$ 0.5678 \$ 1.2665 \$ 0.0589 \$ 1.8932 \$ 0.0589 \$ 1.0760 \$ 0.0589 \$ 1.0760 \$ 0.0557 \$ 0.0589 \$ 1.0760 \$ 0.0589 \$ 0.05				\$ 1.2816	\$ 0.0589				\$1.2665	\$0.0589	
Commercial/Industrial - G-41 Customer Charge per Month per Meter S 0.3123 \$ 0.7048 \$ 0.0589 \$ 1.0760 \$ 0.5678 \$ 1.2665 \$ 0.0589 \$ 1.8932 \$ 0.7048 \$ 0.0589 \$ 1.0760 \$ 0.5678 \$ 1.2665 \$ 0.0589 \$ 1.8932 \$ 0.0589 \$ 1.0760 \$ 0.5678 \$ 1.2665 \$ 0.0589 \$ 1.8932 \$ 0.0589 \$ 1.0760 \$ 0.5678 \$ 1.2665 \$ 0.0589 \$ 1.8932 \$ 0.0589 \$ 1.0760 \$ 0.0589 \$ 1.0760 \$ 0.0557 \$ 0.0589 \$ 1.0760 \$ 0.0589 \$ 0.05	Residential Heating - R-4										
All therms		\$	8 53			\$	8 53	\$15.50			\$15.50
Commercial/Industrial - G-41 Customer Charge per Month per Meter \$57.46 100 therms \$57.46 \$57.46 \$20 therms \$12.665 \$0.0555 \$1.2665 \$0.0555 \$1.7931 \$1.2816 \$0.0555 \$1.8082 \$0.4711 \$1.2665 \$0.0555 \$1.6385 \$0.3165 \$1.2865 \$0.0555 \$1.6385 \$0.3165 \$1.2865 \$0.0555 \$1.6385 \$0.0555 \$1.6385 \$0.0555 \$1.6385 \$0.0555 \$1.6385 \$0.0555 \$1.6385 \$0.0555 \$1.6385 \$0.0555 \$1.6385 \$0.0555 \$1.6385 \$0.0555 \$1.6385 \$0.0555 \$1.6385 \$0.0555 \$1.6385 \$0.0555 \$1.2665 \$0.0555 \$1.6385 \$0.0555 \$1.2865 \$0.0555 \$1.6385 \$0.0555 \$1.2865 \$0.0555 \$1.6385 \$0.0555 \$1.2865 \$0.0555 \$1.6385 \$0.0555 \$1.2865 \$0				\$ 0.7048	\$ 0.0580	Ψ			¢1 2665	\$0.0580	
S57.46 S57.46 S57.46 S57.46 S57.46 S57.46 S57.46 S1ze of the first block 100 therms 100 therms 100 therms 11.2816 S0.0555 S1.8082 S0.4711 S1.2665 S0.0555 S1.7931 All therms over the first block per month at S0.3165 S1.2816 S0.0555 S1.6366 S0.3165 S1.2665 S0.0555 S1.6385 S1.2665 S0.0555 S1.6385 S1.2665 S0.0555 S1.2816	All tiletins	ڔ	0.3123	\$ 0.70 4 8	\$ 0.0369		φ1.0700	φυ.3070	ψ1.2003	φυ.υσοσ	φ1.0932
Size of the first block 100 therms 100 the first block per month at 100 therms 100 the	Commercial/Industrial - G-41										
Therms in the first block per month at All therms over the first block per month at \$0.3165 \$1.2816 \$0.0555 \$1.6836 \$0.0555 \$1.6836 \$0.0555 \$1.68385 \$0.0555 \$1.2816 \$0.0555 \$1.6826 \$0.0555 \$1.2816 \$0.0555 \$1.6826 \$0.0555 \$1.2816 \$0.0555 \$1.6826 \$0.0555 \$1.2816 \$0.0555 \$1.6826 \$0.0555 \$1.2816 \$0.0555 \$1.6826 \$0.0555 \$1.2816 \$0.0555 \$1.6826 \$0.0							\$57.46	\$57.46			\$57.46
All therms over the first block per month at \$0.3165 \$1.2816 \$0.0555 \$1.6536 \$0.3165 \$1.2665 \$0.0555 \$1.6385 \$		•									
Size of the first block per month per Meter State of the first block per month at So. 2853 \$1.2816 \$0.0555 \$1.6026 \$	•										
Size of the first block Size of the first block per month at Size of the first block Siz	All therms over the first block per month at		\$0.3165	\$1.2816	\$0.0555		\$1.6536	\$0.3165	\$1.2665	\$0.0555	\$1.6385
Size of the first block 1000 therms 400 therms 1.2816 50.0555 \$1.7504	Commercial/Industrial - G-42										
Therms in the first block per month at All therms over the first block per month at \$0.4284 \$1.2816 \$0.0555 \$1.6226 \$0.02855 \$1.2665 \$0.0555 \$1.6075 Commercial/Industrial - G-43	Customer Charge per Month per Meter		\$172.39				\$172.39	\$172.39			\$172.39
All therms over the first block per month at \$0.2855 \$1.2816 \$0.0555 \$1.6226 \$0.2855 \$1.2665 \$0.0555 \$1.6075 Commercial/Industrial - G-43	Size of the first block	10	000 therms					400 therms			
Commercial/Industrial - G-43 S739.83 S73	Therms in the first block per month at			\$1.2816	\$0.0555		\$1.7655	\$0.4284	\$1.2665	\$0.0555	\$1.7504
Customer Charge per Month per Meter \$739.83 \$1.2816 \$0.0555 \$1.6004 \$0.1204 \$1.2665 \$0.0555 \$1.4424 \$	All therms over the first block per month at		\$0.2855	\$1.2816	\$0.0555		\$1.6226	\$0.2855	\$1.2665	\$0.0555	\$1.6075
All therms over the first block per month at \$0.2633 \$1.2816 \$0.0555 \$1.6004 \$0.1204 \$1.2665 \$0.0555 \$1.4424 \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	Commercial/Industrial - G-43										
Commercial/Industrial - G-51 Customer Charge per Month per Meter \$57.46 \$57.26 \$57.26 \$57.26 \$57.26 \$57.26 \$57.26 \$57.26 \$57.26 \$57.29 \$57.29 \$57.29 <	Customer Charge per Month per Meter		\$739.83				\$739.83	\$739.83			\$739.83
Customer Charge per Month per Meter S57.46 Size of the first block 100 therms Therms in the first block per month at \$0.2839 \$1.2816 \$0.0555 \$1.6210 \$0.2839 \$1.2665 \$0.0555 \$1.6059 All therms over the first block per month at \$0.1846 \$1.2816 \$0.0555 \$1.5217 \$0.1846 \$1.2665 \$0.0555 \$1.5066 Commercial/Industrial - G-52 Customer Charge per Month per Meter \$172.39 \$172.39 \$172.39 \$172.39 \$12.2816 \$0.0555 \$1.5810 \$0.0767 \$1.2665 \$0.0555 \$1.4987 All therms over the first block per month at \$0.2439 \$1.2816 \$0.0555 \$1.5810 \$0.1767 \$1.2665 \$0.0555 \$1.4987 All therms over the first block per month at \$0.1624 \$1.2816 \$0.0555 \$1.4995 \$0.1004 \$1.2665 \$0.0555 \$1.4224 Commercial/Industrial - G-53 Customer Charge per Month per Meter \$761.39 \$761.39 \$761.39 \$761.39 All therms over the first block per month at \$0.1705 \$1.2816 \$0.0555 \$1.5076 \$0.0818 \$1.2665 \$0.0555 \$1.4038 Commercial/Industrial - G-54 Customer Charge per Month per Meter \$761.39 \$761.39 \$761.39 \$761.39	All therms over the first block per month at		\$0.2633	\$1.2816	\$0.0555		\$1.6004	\$0.1204	\$1.2665	\$0.0555	\$1.4424
Size of the first block 100 therms 100 therms Therms in the first block per month at \$0.2839 \$1.2816 \$0.0555 \$1.6210 \$0.2839 \$1.2665 \$0.0555 \$1.6059 All therms over the first block per month at \$0.1846 \$1.2816 \$0.0555 \$1.5217 \$0.1846 \$1.2665 \$0.0555 \$1.5066 Commercial/Industrial - G-52 Customer Charge per Month per Meter \$172.39 </td <td>Commercial/Industrial - G-51</td> <td></td>	Commercial/Industrial - G-51										
Therms in the first block per month at \$0.2839 \$1.2816 \$0.0555 \$1.60210 \$0.2839 \$1.2665 \$0.0555 \$1.6059 \$1.10059 \$1.2816 \$0.0555 \$1.5217 \$0.1846 \$1.2665 \$0.0555 \$1.5066 \$1.5066 \$1.2816 \$0.0555 \$1.5217 \$0.1846 \$1.2665 \$0.0555 \$1.5066 \$1.5066 \$1.5066 \$1.2816 \$0.0555 \$1.5217 \$0.1846 \$1.2665 \$0.0555 \$1.5066 \$1.50	Customer Charge per Month per Meter		\$57.46				\$57.46	\$57.46			\$57.46
All therms over the first block per month at \$0.1846 \$1.2816 \$0.0555 \$1.5217 \$0.1846 \$1.2665 \$0.0555 \$1.5066 Commercial/Industrial - G-52 Customer Charge per Month per Meter \$172.39 \$17		•									
Commercial/Industrial - G-52 Customer Charge per Month per Meter \$172.39 \$12665 \$0.0555 \$1.4987 \$1.2816 \$0.0555 \$1.5810 \$0.1767 \$1.2665 \$0.0555 \$1.4987 \$1.4224 Commercial/Industrial - G-53 Customer Charge per Month per Meter \$761.39 \$7	•							• • • • • • •			
Customer Charge per Month per Meter Size of the first block \$172.39 \$172.39 \$172.39 \$172.39 \$172.39 \$172.39 \$172.39 \$172.39 \$172.39 \$172.39 \$172.39 \$172.39 \$172.39 \$172.39 \$1000 therms \$1000	All therms over the first block per month at		\$0.1846	\$1.2816	\$0.0555		\$1.5217	\$0.1846	\$1.2665	\$0.0555	\$1.5066
Size of the first block 1000 therms 1000 therms Therms in the first block per month at All therms over the first block per month at So. 2439 \$1.2816 \$0.0555 \$1.5810 \$0.1767 \$1.2665 \$0.0555 \$1.4987 All therms over the first block per month at So. 1624 \$1.2816 \$0.0555 \$1.4995 \$0.1004 \$1.2665 \$0.0555 \$1.4224 Commercial/Industrial - G-53 Customer Charge per Month per Meter All therms over the first block per month at So. 1705 \$1.2816 \$0.0555 \$1.5076 \$0.0818 \$1.2665 \$0.0555 \$1.4038 Commercial/Industrial - G-54 Customer Charge per Month per Meter \$761.39 \$761.39 \$761.39 \$761.39 \$761.39 \$761.39	Commercial/Industrial - G-52										
Therms in the first block per month at All therms over the first block per month at So. 2439 \$1.2816 \$0.0555 \$1.5810 \$0.01767 \$1.2665 \$0.0555 \$1.4987 \$1.2816 \$0.0555 \$1.4995 \$0.1004 \$1.2665 \$0.0555 \$1.4224 \$1.2816 \$0.0555 \$1.4995 \$0.1004 \$1.2665 \$0.0555 \$1.4224 \$1.2816 \$0.0555 \$1.4995 \$0.1004 \$1.2665 \$0.0555 \$1.4224 \$1.2816 \$0.0555 \$1.4995 \$1.2816 \$0.0555 \$1.4995 \$1.2816 \$0.0555 \$1.4995 \$1.2816 \$0.0555 \$1.5076 \$1.2816 \$0.0555 \$1.2816 \$0.0555 \$1.2816 \$0.0555 \$1.5076 \$0.0818 \$1.2665 \$0.0555 \$1.4038 \$1.2665 \$1.4038 \$1.2665 \$1.4038	Customer Charge per Month per Meter		\$172.39				\$172.39	\$172.39			\$172.39
All therms over the first block per month at \$0.1624 \$1.2816 \$0.0555 \$1.4995 \$0.1004 \$1.2665 \$0.0555 \$1.4224 Commercial/Industrial - G-53		10									
Commercial/Industrial - G-53 Customer Charge per Month per Meter \$761.39	•		\$0.2439								
Customer Charge per Month per Meter \$761.39	All therms over the first block per month at		\$0.1624	\$1.2816	\$0.0555		\$1.4995	\$0.1004	\$1.2665	\$0.0555	\$1.4224
All therms over the first block per month at \$0.1705 \$1.2816 \$0.0555 \$1.5076 \$0.0818 \$1.2665 \$0.0555 \$1.4038 Commercial/Industrial - G-54 Customer Charge per Month per Meter \$761.39 \$761.39 \$761.39 \$761.39	Commercial/Industrial - G-53										
Commercial/Industrial - G-54 Customer Charge per Month per Meter \$761.39 \$761.39 \$761.39 \$761.39	Customer Charge per Month per Meter		\$761.39				\$761.39	\$761.39			\$761.39
Customer Charge per Month per Meter \$761.39 \$761.39 \$761.39 \$761.39	All therms over the first block per month at		\$0.1705	\$1.2816	\$0.0555		\$1.5076	\$0.0818	\$1.2665	\$0.0555	\$1.4038
Customer Charge per Month per Meter \$761.39 \$761.39 \$761.39 \$761.39	Commercial/Industrial - G-54										
	· · · · · · · · · · · · · · · · · · ·		\$761.39				\$761.39	\$761.39			\$761.39
	•		\$0.0650	\$1.2816	\$0.0555		\$1.4021	\$0.0353	\$1.2665	\$0.0555	\$1.3573

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2	2021, in Docket N	lo. DG 20-105
ISSUED:	May 24, 2021	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleck
EFFECTIVE:	June 01, 2021	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 396 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES Thirty first Revised Original Page 8689
Superseding Thirtieth Revised Page 86
Firm Rate Schedule

24 FIRM RATE SCHEDULES - MANAGED EXPANSION PROGRAM-EXCLUDING KEENE CUSTOMERS

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX,		
ISSUED:	April 23, 2021	ISSUED BY:	Susan L. Fleek
EFFECTIVE:	May 01, 2021	TITLE:	President

Thirty first Revised Original Page 8689
Superseding Thirtieth Revised Page 86
—Firm Rate Schedule

Rates Effective March 1, 2021 - April 30, 2021
Rates effective November 01, 2021 - April 30, 2022

Rates Effective June 1, 2021 - October 31, 2021 Rates Effective August 1, 2021 - October 31, 2021

	Winter Period				2022	Summer Period							
	Delivery <u>Charge</u>	Cost of Gas Rate Page 96	•	LDAC Page 102		Total <u>Rate</u>	Delivery <u>Charge</u>	(Cost of Gas Rate Page 93		LDAC age 102		Total Rate
Residential Non Heating - R-5	\$ 20.15				\$	20.15	\$ 20.15					\$	20.15
Customer Charge per Month per Meter	\$ 20.01				\$	20.01	\$ 20.01					\$	20.01
All Therms	\$ 0.5666 \$ 0.5018	\$ 0.60 \$ 0.60)50 S	0.0589 0.0589	\$ \$	1.2305 —— 1.1657	\$ 0.5666 \$ 0.5018	\$ \$	0.3935 0.3935	\$ \$	0.0589 — 0.0589	\$ \$	1.0190
Residential Heating - R-6	\$ <u>20.15</u>				\$	20.15	\$ <u>20.15</u>					\$	20.15
Customer Charge per Month per Meter	\$ 20.01				\$	20.01	\$ 20.01					\$	20.01
Size of the first block	all therms						all therms				0.0500		
Therms in the first block per month at	\$ 0.7780 \$ 0.7381	\$ 0.60 \$ 0.60		0.0589 0.0589	\$ \$	1.4419 1.4020	\$ 0.7780 \$ 0.7381	\$ \$	0.3935 0.3935	\$ \$	0.0589 — 0.0589	\$ \$ —	1.2304 1.1905
Residential Heating - R-7	\$ 11.08				\$	 11.08	\$ <u>20.15</u>					\$	20.15
Customer Charge per Month per Meter	\$ 11.01				\$	11.01	\$ 20.01					\$	20.01
Size of the first block	all therms						all therms				0.0500		
Therms in the first block per month at	\$ 0.4279 \$ 0.4060	\$ 0.60 \$ 0.33		0.0589 0.0589	\$ \$	1.0918 0.7977	\$ 0.7780 \$ 0.7381	\$ \$	0.3935 	\$ \$	0.0589 0.0589	\$ \$	1.2304 —— 1.1905
Commercial/Industrial - G-44	\$ 74.60				\$	74.69	\$ 74.69					\$	74.69
Customer Charge per Month per Meter	\$ 77.41				\$	77.41	\$ 77.41					\$	77.41
Size of the first block	100 therms	3					20 therms						
Therms in the first block per month at	\$ 0.6302	\$ 0.60			\$	1.2888	\$ 0.6302	\$	0.3886	\$	0.0555	\$	1.0743
All the consequences the first bloods or consequence	\$ 0.6126	\$ 0.60	34 9	0.0000	\$ _	1.2712	\$ 0.6126 \$ 0.4302	\$	0.3886	\$	— 0.0555	\$— \$	1.0567
All therms over the first block per month at	\$ 0.4302 \$ 0.4114	\$ 0.60 \$ 0.60			\$ \$	1.0888 —— 1.0700	\$ 0.4302 \$ 0.4114	\$ 	0.3886 0.3886	\$ \$	0.0555 — 0.0555	\$ \$	0.8743 —— 0.8555
Commercial/Industrial - G-45	\$ 224.11				2	224.11	\$ 224.11					<u>s_</u>	224.11
Customer Charge per Month per Meter	\$ 232.19				\$	232.19	\$ 232.19					\$	232.19
Size of the first block	1000 therms						400 therms						
Therms in the first block per month at	\$ 0.5732		31 9		\$	1.2318	\$ 0.5732	\$	0.3886	\$	0.0555	\$	1.0173
All the server are the first black as a server to	\$ 0.5569	\$ 0.60			<u>\$</u> _	1.2155	\$ 0.5569	<u>\$</u>	0.3886	<u>\$</u>	0.0555	\$ —	1.0010
All therms over the first block per month at	\$ 0.3884 \$ 0.3711	\$ 0.60 \$ 0.60)31 ()31 (0.0555 0.0555	\$ \$	1.0470 1.0297	\$ 0.3884 \$ 0.3711	\$ \$	0.3886 0.3886	\$ \$	0.0555 — 0.0555	\$ \$	0.8325 0.8152
Commercial/Industrial - G-46	\$961.78				\$	961.78	\$961.78					\$	 961.78
Customer Charge per Month per Meter	\$ 995.16				\$	995.16	\$ 995.16					\$	995.16
All therms over the first block per month at	\$ 0.3532 \$ 0.3423	\$ 0.60)31 S	0.0555 0.0555	\$	1.0118 1.0009	\$ 0.1684 \$ 0.1565	\$	0.3886 	\$	0.0555 	\$ \$	0.6125 0.6006
Commercial/Industrial - G-55	\$ 74.60	,			•	74.69	\$ 74.69	Ť		Ť		•	74.69
Customer Charge per Month per Meter	\$ 77.41				\$	77.41	\$ 77.41					\$	77.41
Size of the first block	100 therms	3			*		100 therms					Ψ.	
Therms in the first block per month at	\$ 0.3796				\$	1.0490	\$ 0.3796	\$	0.3999	\$	0.0555	\$	0.8350
	\$ 0.3691	\$0.61			\$		\$0.3691	\$	0.3999	<u>\$</u>	0.0555	\$	0.8245
All therms over the first block per month at	\$ 0.2514 \$ 0.2400	\$ 0.61 \$ 0.61		0.0555 0.0555	\$ \$	0.9208 0.9094	\$ 0.2514 \$ 0.2400	\$ \$	0.3999 0.3999	\$ \$	0.0555 — 0.0555	\$ \$	0.7068 0.6954
Commercial/Industrial - G-56	\$ 224.11				\$	224.11	\$ 224.11					\$	224.11
Customer Charge per Month per Meter	\$ 232.04				\$	232.04	\$ 232.19					\$	232.19
Size of the first block	1000 therms						1000 therms						
Therms in the first block per month at	\$ 0.3270 \$ 0.3171	\$ 0.61		0.0555	\$	0.9964 	\$ 0.2400 \$ 0.2207	\$	0.3999	\$	0.0555	\$	0.6954
All therms over the first block per month at	\$ 0.2215	\$ 0.61	139 9	0.0555	\$	0.8909	\$ 0.1413	\$	0.3999	\$	0.0555	\$	0.5967
•	\$ 0.2111	\$		0.0555	\$	0.8805	\$0.1304	\$	0.3999	<u>\$</u>	— 0.0555	\$	 0.5858
Commercial/Industrial - G-57	\$ 989.80				\$	989.80	\$ 989.80					\$	989.80
Customer Charge per Month per Meter	\$ 1,025.53	e 000	120 -	0.0555	\$	1,025.53	\$ 1,025.53	•	0.0000	•	0.0555	\$	1,025.53
All therms over the first block per month at	\$ 0.2286 \$ 0.2216	\$ 0.61 \$ 0.61		0.0555 0.0555	\$ \$	0.8980 0.8910	\$ 0.1138 \$ 0.1063	\$ \$	0.3999 0.3999	\$ \$	0.0555 0.0555	\$ \$	0.5692 0.5617
Cunterer Charge per Month per Meter	08 080				•		08 080					¢.	080 80
Customer Charge per Month per Meter All therms over the first block per month at	\$ 1,025.53				\$	1,025.53	\$ 1,025.53					\$	1,025.53
All thomas over the mat block per month at	\$ 0.0871	\$ 0.61	139 5	\$ 0.0555	\$	0.7565	\$ 1,025.55	\$	0.3999	\$	0.0555	\$	0.5041
	\$0.0846	\$ 0.61		0.0555	\$	 0.7540	\$0.0459	\$	0.3999	<u>\$</u>	— 0.0555	\$	 0.5013

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 20	21, in Docket N	o. DG 20-105
ISSUED:	April 23, 2021	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleck
EFFECTIVE:	May 01, 2021	TITLE:	President

Thirty first Revised Original Page 8689
Superseding Thirtieth Revised Page 86
—Firm Rate Schedule

	Rates effective March 1, 2021 - April 30, 2021 Winter Period				30, 2021	Rates Effective May 1, 2021 - October 31, 2021 Summer Period					, 2021			
		Delivery Charge	G	Cost of as Rate ge 92	LDAC age 97	Total <u>Rate</u>		elivery harge	Ga	ost of is Rate ge 89		LDAC e 97		Total <u>Rate</u>
Residential Non Heating - R-5 Customer Charge per Month per Meter All therms	\$	20.15 0.5018	\$	0.6050	\$ 0.0589	\$ 20.15 \$1.1657	\$ \$	20.15 0.5018	\$	0.3935	\$	0.0589	\$	20.15 0.9542
Residential Heating - R-6 Customer Charge per Month per Meter All therms	\$	20.15 0.7381	\$	0.6050	\$ 0.0589	\$ 20.15 \$1.4020	\$ \$	20.15 0.7381	\$	0.3935	\$	0.0589	\$	20.15 1.1905
Residential Heating - R-7 Customer Charge per Month per Meter All therms	\$	11.08 0.4060	\$	0.3328	\$ 0.0589	\$ 11.08 \$0.7977	\$ \$	20.15 0.7381	\$	0.3935	\$	0.0589	\$	20.15 1.1905
Commercial/Industrial - G-44 Customer Charge per Month per Meter Size of the first block	\$	74.69 100 therms				\$ 74.69		74.69 0 therms					\$	74.6886
Therms in the first block per month at All therms over the first block per month at	\$ \$	0.6126 0.4114	\$ \$	0.6031 0.6031		\$1.2712 \$1.0700	\$ \$			0.3886 0.3886		0.0555 0.0555		1.0567 0.8555
Commercial/Industrial - G-45 Customer Charge per Month per Meter Size of the first block	\$	224.11 1000 therms				\$224.11	\$ 40	224.11 0 therms					\$	224.11
Therms in the first block per month at All therms over the first block per month at	\$ \$	0.5569 0.3711	\$ \$	0.6031 0.6031		\$ 1.2155 \$ 1.0297	\$ \$	0.5569 0.3711		0.3886 0.3886		0.0555 0.0555		1.0010 0.8152
Commercial/Industrial - G-46 Customer Charge per Month per Meter All therms over the first block per month at	\$	961.78 0.3423	\$	0.6031	\$ 0.0555	\$961.78 \$1.0009	\$ \$	961.78 0.1565	\$	0.3886	\$	0.0555	\$	961.78 0.6006
Commercial/Industrial - G-55 Customer Charge per Month per Meter Size of the first block	\$	74.69 100 therms				\$ 74.69		74.69 0 therms					\$	74.69
Therms in the first block per month at All therms over the first block per month at	\$ \$	0.3691 0.2400	\$ \$	0.6139 0.6139		\$ 1.0385 \$ 0.9094	\$ \$			0.3999 0.3999		0.0555 0.0555		0.8245 0.6954
Commercial/Industrial - G-56 Customer Charge per Month per Meter Size of the first block	\$	224.11 1000 therms				\$224.11	\$ 100	224.11 0 therms					\$	224.11
Therms in the first block per month at All therms over the first block per month at	\$ \$	0.3171 0.2111	\$ \$	0.6139 0.6139	0.0555 0.0555	\$ 0.9865 \$ 0.8805	\$ \$			0.3999 0.3999		0.0555 0.0555		0.6851 0.5858
Commercial/Industrial - G-57 Customer Charge per Month per Meter All therms over the first block per month at	\$	989.80 0.2216	\$	0.6139	\$ 0.0555	\$ 989.80 \$ 0.8910	\$ \$	989.80 0.1063	\$	0.3999	\$	0.0555	\$	989.80 0.5617
Commercial/Industrial - G-58 Customer Charge per Month per Meter All therms over the first block per month at	\$ \$	989.80 0.0846	\$	0.6139	\$ 0.0555	\$ 989.80 \$ 0.7540	\$ \$	989.80 0.0459	\$	0.3999	\$	0.0555	\$ \$	989.80 0.5013

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 20	21, in Docket N	o. DG 20-105
ISSUED:	April 23, 2021	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleck
EFFECTIVE:	May 01, 2021	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 399 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 90 Firm Rate Schedule Firm Rate Schedule

25 FIRM RATE SCHEDULES – OUTDOOR GAS LIGHTING

Outdoor Gas Lighting					
Per Light Per Month	\$12.81				

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Original Page 91 Summer Cost of Gas

26 ANTCIPATED SUMMER PERIOD COST OF GAS EXCLUDING KEENE CUSTOMERS OR GAS LIGHTING

Anticipated Cost of Gas

PERIOD COVERED: SUMMER PERIOD, MAY 1, 2021 THROUGH OCTOBER 31, 2021 (REFER TO TEXT IN SECTION 16 COST OF GAS CLAUSE)

(Col 1)	(Col 2)	(Col 3)
ANTICIPATED DIRECT COST OF GAS		
Purchased Gas:		
Demand Costs:	\$ 2,868,280	
Supply Costs:	4,387,278	
Storage Gas:		
Demand, Capacity:	\$ -	
Commodity Costs:	-	
Produced Gas:	\$ 29,014	
Hedged Contract (Savings)/Loss	\$	
Unadjusted Anticipated Cost of Gas		\$ 7,284,571
Adjustments:		
Prior Period (Over)/Under Recovery (as of April 30, 2019)	\$ 105,886	
Interest	(3,492)	
Prior Period Adjustments	-	
Broker Revenues	_	
Refunds from Suppliers	-	
Fuel Financing	-	
Transportation CGA Revenues	-	
Interruptible Sales Margin	-	
Capacity Release Margin	-	
Hedging Costs	-	
Fixed Price Option Administrative Costs	-	
Total Adjustments		102,394
Total Anticipated Direct Cost of Gas		\$ 7,386,965
Anticipated Indirect Cost of Gas		
Working Capital:		
Total Unadjusted Anticipated Cost of Gas 05/01/19 - 10/31/19)	\$ 7,284,571	
Working Capital Rate - Lead Lag Days / 365	0.0391	
Prime Rate	3.25%	
Working Capital Percentage	0.127%	
Working Capital	\$ 9,258	
Plus: Working Capital Reconciliation (Acct 1163-1424)	(13,709)	
Total Working Capital Allowance		\$ (4,451)
Bad Debt:		
Total Unadjusted Anticipated Cost of Gas 05/01/19 - 10/31/19)	\$ 7,284,571	
Less: Refunds	-	
Plus: Total Working Capital	(4,451)	
Plus: Prior Period (Over)/Under Recovery	105,886	
Subtotal	\$ 7,386,006	
Bad Debt Percentage	1.11%	
Bad Debt Allowance	\$ 81,985	
Plus: Bad Debt Reconciliation (Acct 1163-1754)	(326,326)	
Total Bad Debt Allowance		(244,341)
Production and Storage Capacity		-
Miscellaneous Overhead (05/01/19 - 10/31/19)	\$ 13,170	
Times Summer Sales	22,004	
Divided by Total Sales	111,369	
Miscellaneous Overhead		2,602
Total Anticipated Indirect Cost of Gas		\$ (246,190)
Total Cost of Gas		\$ 7,140,775

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman

EFFECTIVE: August 1, 2021 TITLE: President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

——Summer Cost of Gas

27 CALCULATION OF FIRM SALES SUMMER PERIOD COST OF GAS RATE **EXCLUDING KEENE CUSTOMERS**

CALCULATION OF FIRM SALES COST OF GAS RATE

		CALCULATION OF PERIOD COVERED: SUMMER P (Refer to Text in S	ERIOD, MAY 1,		2021
		(Col 1)	ection to coat	(Col 2)	(Col 3)
	Total Anticipated Direct Cost of Gas Projected Prorated Sales (05/01/20 -	10/31/20)		\$ 7,386,966 22,681,422	
	Direct Cost of Gas Rate				\$ 0.3257 per therm
	Demand Cost of Gas Rate			\$ 2,868,280	\$ 0.1265 per therm
	Commodity Cost of Gas Rate			4,416,292	\$ 0.1947 per therm
	Adjustment Cost of Gas Rate Total Direct Cost of Gas Rate				\$ 0.0045 per therm \$ 0.3257 per therm
					C.O.COT por troini
	Total Anticipated Indirect Cost of Gas Projected Prorated Sales (05/01/20 -			\$ (246,190) 22,681,422	
	Indirect Cost of Gas	10/01/20)		22,001,422	\$ (0.0109) per therm
	TOTAL PERIOD AVERAGE COST O	F GAS EFFECTIVE 05/01/21			\$ 0.3148 per therm
	Change in rate due to change in u	- 05/01/2021 Approved in Order No 26,419 in DG 20-1	41	COGsr	\$ 0.3148 /therm \$ 0.0787
	RESIDENTIAL COST OF GAS RATE			COGsr	\$ 0.3935 /therm
	Change in rate due to change in u RESIDENTIAL COST OF GAS RATE			COGsr	\$ - \$ 0.3935 /therm
	Change in rate due to change in u			COGSI	\$ -
	RESIDENTIAL COST OF GAS RATE	- 07/01/2021		COGsr	\$ 0.3935 /therm
	Change in rate due to change in u RESIDENTIAL COST OF GAS RATE			COGsr	\$ - \$ 0.3935 /therm
	Change in rate due to change in u				\$ -
	RESIDENTIAL COST OF GAS RATE Change in rate due to change in u			COGsr	\$ 0.3935 /therm \$ -
	RESIDENTIAL COST OF GAS RATE	- 10/01/2021		COGsr	\$ 0.3935 /therm
	-				
				Maximum (COG + 25%)	\$ 0.3935
	COM/IND LOW WINTER USE COST Change in rate due to change in u	OF GAS RATE - 05/01/2021 Approved in Order No 26, nder/over recovery	419 in DG 20-14	1 COGsl	\$ 0.3199 /therm \$ 0.0800
	COM/IND LOW WINTER USE COST			COGsl	\$ 0.3999 /therm
	Change in rate due to change in u	nder/over recovery		2001	\$ - \$ 0,3999 /therm
	COM/IND LOW WINTER USE COST Change in rate due to change in u			COGsl	\$ 0.3999 /therm \$ -
	COM/IND LOW WINTER USE COST	OF GAS RATE - 07/01/2021		COGsl	\$ 0.3999 /therm
	Change in rate due to change in u COM/IND LOW WINTER USE COST			COGsl	\$ - \$ 0.3999 /therm
	Change in rate due to change in u	nder/over recovery		OCCA	\$ -
	COM/IND LOW WINTER USE COST			COGsl	\$ 0.3999 /therm \$ -
	Change in rate due to change in u COM/IND LOW WINTER USE COST			COGsl	\$ 0.3999 /therm
		Average Demand Cost of Gas Rate Effective 05/01/20 Times: Low Winter Use Ratio (Summer) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Com/Ind Low Winter Use Cost of Gas Rate	\$ 0.1265 1.0620 0.9798 \$ 0.1316 \$ 0.1947 0.0045 (0.0109) \$ 0.3199	Maximum (COG + 25%)	\$ 0.3999
	COM/IND HIGH WINTER USE COST			COGsh	\$ 0.3109 /therm
	Change in rate due to change in u COM/IND HIGH WINTER USE COST			COGsh	\$ 0.0777 \$ 0.3886 /therm
	Change in rate due to change in u	nder/over recovery			\$ -
	COM/IND HIGH WINTER USE COST Change in rate due to change in u			COGsh	\$ 0.3886 /therm \$ -
	COM/IND HIGH WINTER USE COST			COGsl	\$ 0.3886 /therm
	Change in rate due to change in u				\$ -
	COM/IND HIGH WINTER USE COST Change in rate due to change in u			COGsl	\$ 0.3886 /therm \$ -
	COM/IND HIGH WINTER USE COST	OF GAS RATE - 09/01/2021		COGsl	\$ 0.3886 /therm
	Change in rate due to change in u COM/IND HIGH WINTER USE COST	nder/over recovery OF GAS RATE - 10/01/2021		COGsl	\$ - \$ 0.3886 /therm
		Average Demand Cost of Gas Rate Effective 05/01/20 'Times: High Winter Use Ratio (Summer) Times: Correction Factor Adjusted Demand Cost of Gas Rate	\$ 0.1265 0.9890 0.9798 \$ 0.1226	Maximum (COG + 25%)	\$ 0.3886
		Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Com/Ind High Winter Use Cost of Gas Rate	\$ 0.1947 0.0045 (0.0109) \$ 0.3109		
DATED:	XXX XX, 2021			ISSUED BY	: /s/Neil Proudman Neil Proudman
EFFECTIVE:	August 1, 2021			TITLE:	President
	-	C Ordon No. VV VVV Jak. 1 V	vv vv	2021 in Daal-14	No. DC 20 105
IGGLIED		C Order No. XX,XXX dated X	$\Lambda\Lambda \Lambda \Lambda \Lambda$	<u> </u>	
ISSUED:	April 23, 2021			ISSUED BY	. I BI Busun E. I ICCK
-					Susan L. Fleek
EFFECTIVE:	May 01, 2021			TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 402 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES Fifteenth RevisedOriginal Page 8992
Superseding Fourteenth Revised Page 89
—Summer Cost of Gas

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX.	, 2021, in Docket N	o. DG 20-105
ISSUED:	April 22 2021	ICCLIED DV.	
IOOODD.	April 25, 2021	ISSUED BY:	<u>/s/Susan L. Fleck</u>
	April 23, 2021	ISSUED BY:	<u>/s/Susan L. Fleek</u> Susan L. Fleek
EFFECTIVE:	May 01, 2021	TITLE:	Susan L. Flook

Docket No. DG 21-130 Exhibit 26 Page 403 of 470 Docket No. DG 20-105

Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES

Sixteenth Revised Original Page 9093 Superseding Fifteenth Revised Page 90

-Summer Cost of Gas

28 CALCULATION OF SUMMER PERIOD COST OF GAS RATE - KEENE CUSTOMERS

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman Neil Proudman EFFECTIVE: August 1, 2021 TITLE: President Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105 ISSUED: June 24, 2021 ISSUED BY: /s/Susan L. Fleck Susan L. Fleck EFFECTIVE: July 01, 2021 TITLE: President

Authorized by NHPUC Order 26,475 dated April 30, 2021 in Docket No. 21-050

Formatted: Tab stops: 4.44", Left + Not at 0.81" + 4.75" + 4.94"

Docket No. DG 21-130 Exhibit 26 Page 404 of 470 Docket No. DG 20-105

Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES

Sixteenth Revised Original Page 9093 Superseding Fifteenth Revised Page 90

	Summer Cost of Gas		
Period Co	wered: Summer Period May 1, 2021, through October 31, 2021		
Projected	Gas Sales - therms		311,740
Total Anti	cipated Cost of Sendout		\$391,714
Add:	Prior Period Deficiency Uncollected Interest	\$0 \$0	
Deduct:	Prior Period Excess Collected Interest	(\$23,223) \$18	
	Prior Period Adjustments and Interest		(\$23,205)
Total Anti	cipated Cost	_	\$368,509
Cost of G	as Rate - Beginning Period (per therm)	_	\$1.1821
Mid Perio	d Adjustment - June 1, 2021		\$0.0844
Mid Perio	d Adjustment - July 1, 2021		\$0.2015
	d Adjustment - August 1, 2021		
	d Adjustment - September 1, 2021		
wia Perio	d Adjustment - October 1, 2021		
Revised (cost of Gas Rate - Effective July 1, 2021 (per therm)	_	\$1.4680
	to tariff section 17(d), the Company may adjust the approved s rate upward on a monthly basis to the following rate:		
Maximum	Cost of Gas Rate (per therm)		\$1.4776

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
ICCLIED.	Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2	2021, in Docket N	o. DG 20-105
ISSUED:	June 24, 2021	ISSUED BY:	Susan L. Fleck
			Busan L. Hicck

Authorized by NHPUC Order 26,475 dated April 30, 2021 in Docket No. 21-050

Formatted: Tab stops: 4.44", Left + Not at 0.81" + 4.75" + 4.94"

Docket No. DG 21-130 Exhibit 26 Page 405 of 470 Docket No. DG 20-105

Docket No. DG 20-105 Exhibit 49

NHPUC NO. 10-11 GAS LIBERTY UTILITIES Sixteenth Revised Original Page 9093 Superseding Fifteenth Revised Page 90

	Summer Cost of Gas		
Period Co	overed: Summer Period May 1, 2021, through October 31, 2021		
Projected	Gas Sales - therms		311,740
Total Ant	icipated Cost of Sendout		\$391,714
Add:	Prior Period Deficiency Uncollected Interest	\$0 \$0	
Deduct:	Prior Period Excess Collected Interest	(\$23,223) \$18	
	Prior Period Adjustments and Interest		(\$23,205)
Total Ant	icipated Cost		\$368,509
Cost of G	as Rate - Beginning Period (per therm)	_	\$1.1821
Mid Perio	d Adjustment - June 1, 2021		\$0.0844
Mid Perio	d Adjustment - July 1, 2021		
	d Adjustment - August 1, 2021		
	d Adjustment - September 1, 2021		
Mid Perio	d Adjustment - October 1, 2021		
Revised (Cost of Gas Rate - Effective June 1, 2021 (per therm)		\$1.2665
	to tariff section 17(d), the Company may adjust the approved as rate upward on a monthly basis to the following rate:		
Maximun	n Cost of Gas Rate (per therm)		\$1.4776

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudmar
			Neil Proudmar
EFFECTIVE:	August 1, 2021	TITLE:	President
	-		
	Authorized by NHPUC Order No. XX,XXX dated XXX XX	2021, in Docket N	o. DG 20-105
ISSUED:	June 24, 2021	ISSUED BY:	/s/Susan L. Fleck
			Susan L. Fleck
EFFECTIVE:	July 01, 2021	TITLE:	President
LITECTIVE.	July 01, 2021	IIILL.	Tresident

Authorized by NHPUC Order 26,475 dated April 30, 2021 in Docket No. 21-050

Formatted: Tab stops: 4.44", Left + Not at 0.81" + 4.75" + 4.94"

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 94 Winter Cost of Gas

29 CALCULATION OF FIXED WINTER PERIOD COST OF GAS RATE EXCLUDING KEENE CUSTOMERS

II. RATE SCHEDULES CALCULATION OF FIXED WINTER PERIOD COST OF GAS RATE PERIOD COVERED: WINTER PERIOD, NOVEMBER 1, 2020 THROUGH APRIL 30, 2021 (Refer to Text in Section 17(A) Fixed Price Option Program)

(Col 1)	(Col 2)	(C	ol 3)	
1 Total Anticipated Direct Cost of Gas	\$46,922,854			
2 Projected Prorated Sales (11/01/20 - 04/30/21)	88,213,529			
3 Direct Cost of Gas Rate		\$	0.5319	per therm
4 Demand Cost of Gas Rate	\$12,978,688	\$	0.1471	per therm
5 Commodity Cost of Gas Rate	32,931,719	\$	0.3733	per therm
6 Adjustment Cost of Gas Rate	1,012,447	\$	0.0115	per therm
7 Total Direct Cost of Gas Rate	\$46,922,854	\$	0.5319	per therm
8 Total Anticipated Indirect Cost of Gas	\$ 2,220,114			
9 Projected Prorated Sales (11/01/20 - 04/30/21)	88,213,529			
10 Indirect Cost of Gas		\$	0.0252	per therm
11 TOTAL PERIOD AVERAGE COST OF GAS EFFECTIVE (11/01/19)		\$	0.5571	
12 Calculation of FPO Excluding Low Income - Rate Code R-3				
13 TOTAL PERIOD AVERAGE COST OF GAS EFFECTIVE (11/01/19)		\$	0.5571	
14 FPO Risk Premium		\$	0.0200	
15 TOTAL PERIOD FIXED PRICE OPTION COST OF GAS RATE EFFECTIVE (11/01/19)		\$	0.5771	
16 RESIDENTIAL COST OF GAS RATE - EXCLUDING LOW INCOME - (R-3) 11/01/20	OGwr	\$	0.5771	/therm
17 Calculation of FPO for Gas Assistance Program - Rate Code R-4				
18 TOTAL PERIOD AVERAGE COST OF GAS EFFECTIVE (11/01/20) FPO (Line 13 * 0.55)		\$	0.3064	
19 FPO Risk Premium (Line 14 * 0.55)		\$	0.0110	
20 TOTAL PERIOD FIXED PRICE OPTION COST OF GAS RATE EFFECTIVE (11/01/19) (Line 15 * 0.55)		\$	0.3174	
21 RESIDENTIAL COST OF GAS RATE - LOW INCOME - (R-4) 11/01/20 C	OGwr	\$	0.3174	/therm

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Original Page 95 Winter Cost of Gas

30 CALCULATION OF FIRM SALES WINTER PERIOD COST OF GAS RATE - EXCLUDING KEENE CUSTOMERS

CALCULATION OF FIRM SALES COST OF GAS RATE PERIOD COVERED: WINTER PERIOD, NOVEMBER 1, 2020 THROUGH APRIL 30, 2021 (Refer to Text in Section 16 Cost of Gas Clause)

(Refer to Text in	1 Section 16 Co	st of Gas Clause)				
(Col 1)			(Col 2)		(Col 3)	
(6611)			(OUI Z)		(001 0)	
Total Anticipated Direct Cost of Gas		\$	46,922,854			
Projected Prorated Sales (11/01/2020 - 04/30/21) Direct Cost of Gas Rate			88,213,529	s	0.5040	per therm
Direct Cost of Gas Rate				Φ	0.5519	per merm
Demand Cost of Gas Rate		\$	12,978,688	\$	0.1471	per therm
Commodity Cost of Gas Rate Adjustment Cost of Gas Rate			32,931,719 1,012,447	\$	0.3733	per therm per therm
Total Direct Cost of Gas Rate		\$	46,922,854	\$		per therm
Total Anticipated Indirect Cost of Gas Projected Prorated Sales (11/01/20 - 04/30/21)		\$	2,220,114 88.213.529			
Indirect Cost of Gas			00,210,020	\$	0.0252	per therm
TOTAL PERIOD AVERAGE COST OF GAS EFFECTIVE 11/01/20				\$	0.5574	per therm
				•		•
RESIDENTIAL COST OF GAS RATE - EXCLUDING LOW INCOME - 11/01/2	0	co	Gwr	\$	0.5571	/therm
Change in rate due to change in under/over recovery RESIDENTIAL COST OF GAS RATE - 12/01/2020		со	~au	\$	0 5574	/therm
Change in Rate due to change in under/over recovery				\$	(0.0907)	
RESIDENTIAL COST OF GAS RATE - 01/01/2021		co	Gwr	\$		/therm
Change in Rate due to change in under/over recovery RESIDENTIAL COST OF GAS RATE - 02/01/2021			Gwr	\$	(0.0388)	/therm
Change in Rate due to change in under/over recovery			JWI	\$	0.0880	/uleilli
RESIDENTIAL COST OF GAS RATE - 03/01/2021		CO	Gwr	\$		/therm
Change in Rate due to change in under/over recovery RESIDENTIAL COST OF GAS RATE - 04/01/2021			Gwr	\$	0.0894	/therm
RESIDENTIAL COST OF GAS RATE - 04/01/2021			JWF	•	0.6050	/tnerm
		Maximum (CC	G + 25%)	\$	0.6964	
GAS ASSISTANCE PLAN RESIDENTIAL COST OF GAS RATE R-4 - 11/01/2	20		Gwr	¢	0.3064	/therm
Change in rate due to change in under/over recovery	40	CO	JW1	\$	J.3064 -	/ GIETIII
RESIDENTIAL COST OF GAS RATE - 12/01/2020		co	Gsr	\$	0.3064	/therm
Change in Rate due to change in under/over recovery				\$	(0.0499)	
RESIDENTIAL COST OF GAS RATE - 01/01/2021 Change in Rate due to change in under/over recovery		CO	Gwr	\$	(0.0213)	/therm
RESIDENTIAL COST OF GAS RATE - 2/1/2021		co	Gwr	\$		/therm
Change in Rate due to change in under/over recovery				\$	0.0484	
RESIDENTIAL COST OF GAS RATE - 3/1/2021 Change in Rate due to change in under/over recovery		CO	Gwr	\$	0.2836	/therm
RESIDENTIAL COST OF GAS RATE - 4/1/2021		co	Gwr	\$		/therm
		Maximum (CC	G + 25%)	\$	0.3830	
C&I LOW WINTER USE COST OF GAS RATE - 11/01/20		CO	Gwl	\$	0.5660	/therm
Change in rate due to change in under/over recovery						
				\$		
C&I LOW WINTER USE COST OF GAS RATE - 12/01/2020		co	Gsl	\$		/therm
		co			(0.0907)	
C8I LOW WINTER USE COST OF GAS RATE - 12/01/2020 Change in Rate due to change in under/over recovery C8I LOW WINTER USE COST OF GAS RATE - 01/01/2021 Change in Rate due to change in under/over recovery				\$	(0.0907) 0.4753 (0.0388)	/therm
C&L LOW WINTER USE COST OF GAS RATE - 12/01/2020 Change in Rate due to change in underfover recovery C&L LOW WINTER USE COST OF GAS RATE - 01/01/2021 Change in Rate due to change in underfover recovery C&L LOW WINTER USE COST OF GAS RATE - 20/10/2021			Gsl	\$ \$ \$ \$	(0.0907) 0.4753 (0.0388) 0.4365	/therm
C&L LOW WINTER USE COST OF GAS RATE - 1/20/1/2020 Change in Rate due to change in under/over recovery C&L LOW WINTER USE COST OF GAS RATE - 0/10/1/2021 Change in Rate due to change in under/over recovery C&L LOW WINTER USE COST OF GAS RATE - 2/0/1/2021 Change in Rate due to change in under/over recovery		co	Gsl	\$ \$ \$ \$	(0.0907) 0.4753 (0.0388) 0.4365 0.0880	/therm
C&L LOW WINTER USE COST OF GAS RATE - 12/01/2020 Change in Rate due to change in underfover recovery C&L LOW WINTER USE COST OF GAS RATE - 01/01/2021 Change in Rate due to change in underfover recovery C&L LOW WINTER USE COST OF GAS RATE - 20/10/2021 Change in Rate due to change in underfover recovery C&L LOW WINTER USE COST OF GAS RATE - 20/10/2021 CAL LOW WINTER USE COST OF GAS RATE - 3/01/2021		co	Gsl	\$ \$ \$ \$	(0.0907) 0.4753 (0.0388) 0.4365 0.0880 0.5245	/therm
C&L LOW WINTER USE COST OF GAS RATE - 1/20/1/2020 Change in Rate due to change in under/over recovery C&L LOW WINTER USE COST OF GAS RATE - 0/10/1/2021 Change in Rate due to change in under/over recovery C&L LOW WINTER USE COST OF GAS RATE - 2/0/1/2021 Change in Rate due to change in under/over recovery		co	Gsl Gsl	\$ \$ \$ \$ \$	(0.0907) 0.4753 (0.0388) 0.4365 0.0880 0.5245 0.0894	/therm
Cal LOW WINTER USE COST OF GAS RATE - 1/20/1/2020 Change in Rate due to change in under/over recovery Cal LOW WINTER USE COST OF GAS RATE - 0/10/1/2021 Change in Rate due to change in under/over recovery Cal LOW WINTER USE COST OF GAS RATE - 2/0/1/2021 Change in Rate due to change in under/over recovery Cal LOW WINTER USE COST OF GAS RATE - 3/0/1/2021 Change in Rate due to change in under/over recovery Cal LOW WINTER USE COST OF GAS RATE - 3/0/1/2021 Change in Rate due to change in under/over recovery Cal LOW WINTER USE COST OF GAS RATE - 4/0/1/2021		co	Gsl Gsl	\$ \$ \$ \$ \$	(0.0907) 0.4753 (0.0388) 0.4365 0.0880 0.5245 0.0894	/therm /therm /therm
Cal LOW WINTER USE COST OF GAS RATE - 1/20/1/2020 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 0/10/1/2021 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 20/1/2021 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 30/1/2021 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 30/1/2021 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 4/10/1/201 Average Demand Cost of Gas Rate Effective 11//01/20	\$ 0.1471	CO CO	Gsl Gsl Gsl Gsl	\$ \$ \$ \$ \$ \$	(0.0907) 0.4753 (0.0388) 0.4365 0.0880 0.5245 0.0894 0.6139	/therm /therm /therm
Cal LOW WINTER USE COST OF GAS RATE - 1/20/1/2020 Change in Rate due to change in under/over recovery Cal LOW WINTER USE COST OF GAS RATE - 0/10/1/2021 Change in Rate due to change in under/over recovery Cal LOW WINTER USE COST OF GAS RATE - 2/0/1/2021 Change in Rate due to change in under/over recovery Cal LOW WINTER USE COST OF GAS RATE - 3/0/1/2021 Change in Rate due to change in under/over recovery Cal LOW WINTER USE COST OF GAS RATE - 3/0/1/2021 Change in Rate due to change in under/over recovery Cal LOW WINTER USE COST OF GAS RATE - 4/0/1/2021	\$ 0.1471 1.0620 0.9984	co	Gsl Gsl Gsl Gsl	\$ \$ \$ \$ \$ \$	(0.0907) 0.4753 (0.0388) 0.4365 0.0880 0.5245 0.0894	/therm /therm /therm
C&I LOW WINTER USE COST OF GAS RATE - 1/20/1/2020 Change in Rate due to change in underfover recovery C&I LOW WINTER USE COST OF GAS RATE - 0/10/1/2021 Change in Rate due to change in underfover recovery C&I LOW WINTER USE COST OF GAS RATE - 20/1/2021 Change in Rate due to change in underfover recovery C&I LOW WINTER USE COST OF GAS RATE - 3/0/1/2021 Change in Rate due to change in underfover recovery C&I LOW WINTER USE COST OF GAS RATE - 3/0/1/2021 Average Demand Cost of Gas Rate Effective 11/0/1/20 Times: Low Winter Use Rate (Winter)	1.0620	CO CO	Gsl Gsl Gsl Gsl	\$ \$ \$ \$ \$ \$	(0.0907) 0.4753 (0.0388) 0.4365 0.0880 0.5245 0.0894 0.6139	/therm /therm /therm
Cal LOW WINTER USE COST OF GAS RATE - 12/01/2020 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 01/01/2021 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 20/11/2021 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 30/11/2021 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 30/11/2021 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 40/11/2021 Times: Low Winter Use Ratio (Winter) Times: Low Winter Use Ratio (Winter) Times: Correction Factor Adjusted Demand Cost of Gas Rate	1.0620 0.9984 \$ 0.1560	CO CO	Gsl Gsl Gsl Gsl	\$ \$ \$ \$ \$ \$	(0.0907) 0.4753 (0.0388) 0.4365 0.0880 0.5245 0.0894 0.6139	/therm /therm /therm
C&L LOW WINTER USE COST OF GAS RATE - 1/20/1/2020 Change in Rate due to change in underforer recovery C&I LOW WINTER USE COST OF GAS RATE - 01/01/2021 Change in Rate due to change in underforer recovery C&I LOW WINTER USE COST OF GAS RATE - 201/02021 Change in Rate due to change in underforer recovery C&I LOW WINTER USE COST OF GAS RATE - 201/02021 Change in Rate due to change in underforer recovery C&I LOW WINTER USE COST OF GAS RATE - 3/01/2021 CARROW WINTER USE COST OF GAS RATE - 4/01/2021 Average Demand Cost of Gas Rate Effective 11/01/20 Times: Low Winter Use Ratio (Winter) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate	1.0620 0.9984	CO CO	Gsl Gsl Gsl Gsl	\$ \$ \$ \$ \$ \$	(0.0907) 0.4753 (0.0388) 0.4365 0.0880 0.5245 0.0894 0.6139	/therm /therm /therm
C&L LOW WINTER USE COST OF GAS RATE - 1/20/1/2020 Change in Rate due to change in underforer recovery C&L LOW WINTER USE COST OF GAS RATE - 01/01/2021 Change in Rate due to change in underforer recovery C&L LOW WINTER USE COST OF GAS RATE - 201/01/2021 Change in Rate due to change in underforer recovery C&L LOW WINTER USE COST OF GAS RATE - 201/02021 Change in Rate due to change in underforer recovery C&L LOW WINTER USE COST OF GAS RATE - 3/01/2021 Change in Rate due to change in underforer recovery C&L LOW WINTER USE COST OF GAS RATE - 4/01/2021 Average Demand Cost of Gas Rate Effective 11/01/20 Times: Low Winter Use Rate (Vinter) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate	1.0620 0.9984 \$ 0.1560 \$ 0.3733 \$ 0.0115 \$ 0.0252	CO CO	Gsl Gsl Gsl Gsl	\$ \$ \$ \$ \$ \$	(0.0907) 0.4753 (0.0388) 0.4365 0.0880 0.5245 0.0894 0.6139	/therm /therm /therm
Cal LOW WINTER USE COST OF GAS RATE - 1/20/1/2020 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 0/10/1/2021 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 20/1/2021 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 30/1/2021 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 30/1/2021 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 40/1/2021 Average Demand Cost of Gas Rate Effective 11/01/20 Times: Low Winter Use Ratio (Winter) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjustment Cost of Gas Rate	1.0620 0.9984 \$ 0.1560 \$ 0.3733 \$ 0.0115	CO CO	Gsl Gsl Gsl Gsl	\$ \$ \$ \$ \$ \$	(0.0907) 0.4753 (0.0388) 0.4365 0.0880 0.5245 0.0894 0.6139	/therm /therm /therm
CSL LOW WINTER USE COST OF GAS RATE - 1/20/1/2020 Change in Rate due to change in underfover recovery CSL LOW WINTER USE COST OF GAS RATE - 0/10/1/2021 Change in Rate due to change in underfover recovery CSL LOW WINTER USE COST OF GAS RATE - 20/1/2021 Change in Rate due to change in underfover recovery CSL LOW WINTER USE COST OF GAS RATE - 20/1/2021 Change in Rate due to change in underfover recovery CSL LOW WINTER USE COST OF GAS RATE - 3/01/2021 Change in Rate due to change in underfover recovery CSL LOW WINTER USE COST OF GAS RATE - 4/01/2021 CSL LOW WINTER USE COST OF GAS RATE - 4/01/2021 Times: Low Winter Use Ratio (Winter) Times: Covercion Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjustend Cost of Gas Rate Adjustend Cost of Gas Rate Adjusted CSL Low Winter Use Cost of Gas Rate Adjusted CSL Low Winter Use Cost of Gas Rate	1.0620 0.9984 \$ 0.1560 \$ 0.3733 \$ 0.0115 \$ 0.0252	CO CO CO Maximum (CC	Gal Gal Gal Gal	\$ \$ \$ \$ \$ \$ \$ \$ \$	(0.0907) 0.4753 (0.0388) 0.4365 0.0884 0.5245 0.0894 0.6139	/therm /therm /therm
C&L LOW WINTER USE COST OF GAS RATE - 1/20/1/2020 Change in Rate due to change in underforer recovery C&L LOW WINTER USE COST OF GAS RATE - 01/01/2021 Change in Rate due to change in underforer recovery C&L LOW WINTER USE COST OF GAS RATE - 20/1/2021 Change in Rate due to change in underforer recovery C&L LOW WINTER USE COST OF GAS RATE - 30/1/2021 Change in Rate due to change in underfover recovery C&L LOW WINTER USE COST OF GAS RATE - 30/1/2021 Average Demand Cost of Gas Rate Effective 11/01/20 Times: Low Winter Use Ratio (Winter) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Adjusted C&L Low Winter Use Cost of Gas Rate C&L HIGH WINTER USE COST OF GAS RATE - 11/01/20	1.0620 0.9984 \$ 0.1560 \$ 0.3733 \$ 0.0115 \$ 0.0252	CO CO CO Maximum (CC	Gsl Gsl Gsl Gsl	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	(0.0907) 0.4753 (0.0388) 0.4365 0.0884 0.5245 0.0894 0.6139	/therm /therm /therm
Cal LOW WINTER USE COST OF GAS RATE - 1/20/1/2020 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 0/10/1/2021 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 20/1/2021 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 3/01/2021 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 3/01/2021 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 4/01/2021 Average Demand Cost of Gas Rate Effective 11/01/20 Times: Low Winter Use Ratio (Winter) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate Adjustment Cost of Gas Rate Adjustment Cost of Gas Rate Call HIGH WINTER USE COST OF GAS RATE - 1/10/1/20 Change in rate due to change in underfover recovery	1.0620 0.9984 \$ 0.1560 \$ 0.3733 \$ 0.0115 \$ 0.0252	CO CO Maximum (CC	Gel Gel Gel G + 25%)	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	(0.0907) 0.4753 (0.0388) 0.4365 0.0880 0.5245 0.0894 0.6139 0.7075	/therm /therm /therm /therm
CAL LOW WINTER USE COST OF GAS RATE - 12/01/2020 Change in Rate due to change in underfover recovery CAL LOW WINTER USE COST OF GAS RATE - 01/01/2021 Change in Rate due to change in underfover recovery CAL LOW WINTER USE COST OF GAS RATE - 20/10/2021 Change in Rate due to change in underfover recovery CAL LOW WINTER USE COST OF GAS RATE - 20/10/2021 Change in Rate due to change in underfover recovery CAL LOW WINTER USE COST OF GAS RATE - 30/10/2021 Change in Rate due to change in underfover recovery CAL LOW WINTER USE COST OF GAS RATE - 40/10/2021 Times: Low Winter Use Ratio (Winter) Times: Low Winter Use Ratio (Winter) Times: Correction Factor Adjusted Demand Cost of Gas Rate Effective 11/01/20 Times: Correction Factor Adjusted Demand Cost of Gas Rate Adjusted CAL Low Winter Use Cost of Gas Rate Indirect Cost of Gas Rate Adjusted CAL Low Winter Use Cost of Gas Rate CAL HIGH WINTER USE COST OF GAS RATE - 11/01/20 Change in rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 12/01/200	1.0620 0.9984 \$ 0.1560 \$ 0.3733 \$ 0.0115 \$ 0.0252	CO CO CO Maximum (CC	Gel Gel Gel G + 25%)	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	(0.0907) 0.4753 (0.0388) 0.4365 0.0880 0.5245 0.0894 0.6139 0.7075	/therm /therm /therm /therm /therm
Cal LOW WINTER USE COST OF GAS RATE - 1/20/1/2020 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 0/10/1/2021 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 20/1/2021 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 3/01/2021 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 3/01/2021 Change in Rate due to change in underfover recovery Cal LOW WINTER USE COST OF GAS RATE - 4/01/2021 Average Demand Cost of Gas Rate Effective 11/01/20 Times: Low Winter Use Ratio (Winter) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate Adjustment Cost of Gas Rate Adjustment Cost of Gas Rate Call HIGH WINTER USE COST OF GAS RATE - 1/10/1/20 Change in rate due to change in underfover recovery	1.0620 0.9984 \$ 0.1560 \$ 0.3733 \$ 0.0115 \$ 0.0252	CO CO Maximum (CC CO	Gel Gel Gel G + 25%)	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	(0.0907) 0.4753 (0.0388) 0.4365 0.0880 0.5245 0.0894 0.6139 0.7075	/therm /therm /therm /therm /therm
CAL LOW WINTER USE COST OF GAS RATE - 12/01/2020 Change in Rate due to change in underforer recovery CAL LOW WINTER USE COST OF GAS RATE - 01/01/2021 Change in Rate due to change in underforer recovery CAL LOW WINTER USE COST OF GAS RATE - 01/01/2021 Change in Rate due to change in underforer recovery CAL LOW WINTER USE COST OF GAS RATE - 20/01/2021 Change in Rate due to change in underforer recovery CAL LOW WINTER USE COST OF GAS RATE - 3/01/2021 CAL LOW WINTER USE COST OF GAS RATE - 4/01/2021 Average Demand Cost of Gas Rate Effective 11/01/20 Times: Low Winter Use Ratio (Winter) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Adjustment Cost of Gas Rate Adjustment Cost of Gas Rate Adjusted CAL Low Winter Use Cost of Gas Rate CAL HIGH WINTER USE COST OF GAS RATE - 11/01/20 Change in rate due to change in underforer recovery CAL HIGH WINTER USE COST OF GAS RATE - 12/01/2020 Change in Rate due to change in underforer recovery CAL HIGH WINTER USE COST OF GAS RATE - 11/01/2021 CAL HIGH WINTER USE COST OF GAS RATE - 11/01/2021 CAL HIGH WINTER USE COST OF GAS RATE - 11/01/2021 CAL HIGH WINTER USE COST OF GAS RATE - 11/01/2021 CAL HIGH WINTER USE COST OF GAS RATE - 11/01/2021 CAL HIGH WINTER USE COST OF GAS RATE - 11/01/2021 CAL HIGH WINTER USE COST OF GAS RATE - 11/01/12021 CAL HIGH WINTER USE COST OF GAS RATE - 01/01/2021	1.0620 0.9984 \$ 0.1560 \$ 0.3733 \$ 0.0115 \$ 0.0252	CO CO Maximum (CC CO	Gdd Gdd Gdd Gdd Gd + 25%)	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	(0.0907) (0.4763) (0.0388) (0.0388) (0.0880) (0.5245) (0.0894) (0.6139) (0.7075) (0.5552) (0.9077) (0.0388)	/therm /therm /therm /therm /therm /therm /therm /therm /therm
CAL LOW WINTER USE COST OF GAS RATE - 12/01/2020 Change in Rate due to change in underfover recovery CAL LOW WINTER USE COST OF GAS RATE - 01/01/2021 Change in Rate due to change in underfover recovery CAL LOW WINTER USE COST OF GAS RATE - 20/11/2021 Change in Rate due to change in underfover recovery CAL LOW WINTER USE COST OF GAS RATE - 20/11/2021 Change in Rate due to change in underfover recovery CAL LOW WINTER USE COST OF GAS RATE - 40/10/2021 Change in Rate due to change in underfover recovery CAL LOW WINTER USE COST OF GAS RATE - 40/10/2021 Average Demand Cost of Gas Rate Effective 11/01/20 Times: Low Winter Use Ratio (Winter) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjustend Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate Adjustend CAL Winter Use COST OF GAS RATE - 11/01/20 Change in rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 12/01/20/20 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 01/01/20/21 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 01/01/20/21 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 01/01/20/21 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 01/01/20/21	1.0620 0.9984 \$ 0.1560 \$ 0.3733 \$ 0.0115 \$ 0.0252	CO CO Maximum (CC CO	Gel	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	(0.0907) (0.4763 (0.0388) (0.0388) (0.0880 (0.0880 (0.0894) (0.0894) (0.07075) (0.07075)	/therm /therm /therm /therm /therm /therm /therm /therm /therm
CAL LOW WINTER USE COST OF GAS RATE - 12/01/2020 Change in Rate due to change in underforer recovery CAL LOW WINTER USE COST OF GAS RATE - 01/01/2021 Change in Rate due to change in underforer recovery CAL LOW WINTER USE COST OF GAS RATE - 20/10/2021 Change in Rate due to change in underforer recovery CAL LOW WINTER USE COST OF GAS RATE - 20/10/2021 Change in Rate due to change in underforer recovery CAL LOW WINTER USE COST OF GAS RATE - 30/10/2021 Change in Rate due to change in underforer recovery CAL LOW WINTER USE COST OF GAS RATE - 30/10/2021 Average Demand Cost of Gas Rate Effective 11/01/20 Times: Low Winter Use Ratio (Winter) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjustered Paramand Cost of Gas Rate Adjustered Cost of Gas Rate CAL HIGH WINTER USE COST OF GAS RATE - 11/01/20 Change in rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 20/10/2021 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 20/10/2021 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 20/10/2021 Change in Rate due to change in underfover recovery	1.0620 0.9984 \$ 0.1560 \$ 0.3733 \$ 0.0115 \$ 0.0252	CO CO CO Maximum (CC CO C	Gel	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	0.0907) 0.4763 0.4365 0.0808 0.4365 0.0808 0.5245 0.6139 0.7075	/therm
CAL LOW WINTER USE COST OF GAS RATE - 12/01/2020 Change in Rate due to change in underfover recovery CAL LOW WINTER USE COST OF GAS RATE - 01/01/2021 Change in Rate due to change in underfover recovery CAL LOW WINTER USE COST OF GAS RATE - 20/12/21 Change in Rate due to change in underfover recovery CAL LOW WINTER USE COST OF GAS RATE - 30/12/21 Change in Rate due to change in underfover recovery CAL LOW WINTER USE COST OF GAS RATE - 30/12/22 Change in Rate due to change in underfover recovery CAL LOW WINTER USE COST OF GAS RATE - 40/12/22 Average Demand Cost of Gas Rate Effective 11/01/20 Times: Low Winter Use Ratio (Winter) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjustener Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate Adjustener Low Winter Use Cost of Gas Rate Indirect Cost of Gas Rate Adjustener Cost of Gas Rate Indirect Cost of Gas RATE - 11/01/20 Change in rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 12/01/20/20 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 01/01/20/1 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 20/12/21 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 20/12/21 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 20/12/21 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 20/12/21 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 20/12/21	1.0620 0.9984 \$ 0.1560 \$ 0.3733 \$ 0.0115 \$ 0.0252	CO CO CO Maximum (CC CO C	Gdd Gdd Gdd Gdd Gd + 25%)	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	(0.0907) (0.0783 0.4365 0.0880 0.5245 0.0894 0.6139 0.7075 0.5552 0.5552 0.05552 0.05552 0.0904 0	/therm /therm /therm /therm /therm /therm /therm /therm /therm
CALLOW WINTER USE COST OF GAS RATE - 12/01/2020 Change in Rate due to change in underfover recovery Cat LOW WINTER USE COST OF GAS RATE - 01/01/2021 Change in Rate due to change in underfover recovery Cat LOW WINTER USE COST OF GAS RATE - 20/12/221 Change in Rate due to change in underfover recovery Cat LOW WINTER USE COST OF GAS RATE - 20/12/221 Change in Rate due to change in underfover recovery Cat LOW WINTER USE COST OF GAS RATE - 3/01/2021 Change in Rate due to change in underfover recovery Cat LOW WINTER USE COST OF GAS RATE - 4/01/2021 Average Demand Cost of Gas Rate Effective 11/01/20 Times: Low Winter Use Ratio (Winter) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjusted Cat Low Winter Use Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Cat Low Winter Use Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Cat Low Winter Use Cost of Gas Rate Indirect Cost of Gas Rate - 11/01/20 Change in rate due to change in underfover recovery Cat MiGH WINTER USE COST OF GAS RATE - 11/01/20 Change in Rate due to change in underfover recovery Cat MiGH WINTER USE COST OF GAS RATE - 01/01/201 Change in Rate due to change in underfover recovery Cat Migh WINTER USE COST OF GAS RATE - 3/01/201 Change in Rate due to change in underfover recovery Cat Migh WINTER USE COST OF GAS RATE - 3/01/201 Change in Rate due to change in underfover recovery Cat Migh WINTER USE COST OF GAS RATE - 3/01/201 Change in Rate due to change in underfover recovery	1.0620 0.9984 \$ 0.1560 \$ 0.3733 \$ 0.0115 \$ 0.0252	CO C	Gsl	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	0.0907) 0.4763 0.4365 0.0880 0.5245 0.0894 0.6139 0.7075 0.5552 (0.0907) 0.4645 0.0388 0.4257 0.0388	/therm
CALLOW WINTER USE COST OF GAS RATE - 12/01/2020 Change in Rate due to change in underfover recovery Cat LOW WINTER USE COST OF GAS RATE - 01/01/2021 Change in Rate due to change in underfover recovery Cat LOW WINTER USE COST OF GAS RATE - 20/12/221 Change in Rate due to change in underfover recovery Cat LOW WINTER USE COST OF GAS RATE - 20/12/221 Change in Rate due to change in underfover recovery Cat LOW WINTER USE COST OF GAS RATE - 3/01/2021 Change in Rate due to change in underfover recovery Cat LOW WINTER USE COST OF GAS RATE - 4/01/2021 Average Demand Cost of Gas Rate Effective 11/01/20 Times: Low Winter Use Ratio (Winter) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjusted Cat Low Winter Use Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Cat Low Winter Use Cost of Gas Rate Indirect Cost of Gas Rate Adjusted Cat Low Winter Use Cost of Gas Rate Indirect Cost of Gas Rate - 11/01/20 Change in rate due to change in underfover recovery Cat MiGH WINTER USE COST OF GAS RATE - 11/01/20 Change in Rate due to change in underfover recovery Cat MiGH WINTER USE COST OF GAS RATE - 01/01/201 Change in Rate due to change in underfover recovery Cat Migh WINTER USE COST OF GAS RATE - 3/01/201 Change in Rate due to change in underfover recovery Cat Migh WINTER USE COST OF GAS RATE - 3/01/201 Change in Rate due to change in underfover recovery Cat Migh WINTER USE COST OF GAS RATE - 3/01/201 Change in Rate due to change in underfover recovery	1.0620 0.9984 \$ 0.1560 \$ 0.3733 \$ 0.0115 \$ 0.0252	CO C	Gel	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	0.0907) 0.4763 0.4365 0.0880 0.5245 0.0894 0.6139 0.7075 0.5552 (0.0907) 0.4645 0.0388 0.4257 0.0388	/therm
CAL LOW WINTER USE COST OF GAS RATE - 12/01/2020 Change in Rate due to change in underfover recovery CAL LOW WINTER USE COST OF GAS RATE - 01/01/2021 Change in Rate due to change in underfover recovery CAL LOW WINTER USE COST OF GAS RATE - 20/01/2021 Change in Rate due to change in underfover recovery CAL LOW WINTER USE COST OF GAS RATE - 20/01/2021 Change in Rate due to change in underfover recovery CAL LOW WINTER USE COST OF GAS RATE - 3/01/2021 Change in Rate due to change in underfover recovery CAL LOW WINTER USE COST OF GAS RATE - 4/01/2021 Average Demand Cost of Gas Rate Effective 11/01/20 Times: Low Winter Use Ratio (Winter) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjustment Cost of Gas Rate - 2/01/2021 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 2/01/2021 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 2/01/2021 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 2/01/2021 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 2/01/2021 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 2/01/2021 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 2/01/2021	1.0620 0.9984 \$ 0.1560 \$ 0.3733 \$ 0.0115 \$ 0.0252 \$ 0.5660	CO C	Gel	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	0.0907) 0.4753 0.0880 0.4365 0.0800 0.5245 0.0800 0.5245 0.07075 0.5552 0.0800 0.4645 0.0907) 0.4645 0.0800 0.5131	/therm
CALLOW WINTER USE COST OF GAS RATE - 12/01/2020 Change in Rate due to change in underfover recovery CALLOW WINTER USE COST OF GAS RATE - 01/01/2021 Change in Rate due to change in underfover recovery CALLOW WINTER USE COST OF GAS RATE - 20/01/2021 Change in Rate due to change in underfover recovery CALLOW WINTER USE COST OF GAS RATE - 20/01/2021 Change in Rate due to change in underfover recovery CALLOW WINTER USE COST OF GAS RATE - 3/01/2021 Change in Rate due to change in underfover recovery CALLOW WINTER USE COST OF GAS RATE - 4/01/2021 Average Demand Cost of Gas Rate Effective 11/01/20 Times: Low Winter Use Ratlo (Winter) Times: Corection Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjusted Call Cost of Gas Rate Indirect Cost of Gas Rate Adjusted CALLOW Winter Use Cost of Gas Rate Indirect Cost of Gas Rate Indirect Cost of Gas Rate Adjusted CALLOW Winter Use Cost of Gas Rate Indirect Cost of Gas Rate Adjusted CALLOW Winter Use Cost of Gas Rate Indirect Cost of Gas Rate - 1/10/1/20 Change in rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 1/10/1/20 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 3/01/2021 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 3/01/2021 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 3/01/2021 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 3/01/2021 Change in Rate due to change in underfover recovery CALL HIGH WINTER USE COST OF GAS RATE - 3/01/2021 Change in Rate due to change in underfover recovery CALL HIGH WINTER USE COST OF GAS RATE - 3/01/2021 Change in Rate due to change in underfover recovery CALL HIGH WINTER USE COST OF GAS RATE - 3/01/2021 Change in Rate due to change in underfover recovery	1.0620 0.9994 \$ 0.1580 \$ 0.3733 \$ 0.0115 \$ 0.0252 \$ 0.05660	CO C	Gel	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	0.0907) 0.4763 0.4365 0.0880 0.5245 0.0894 0.6139 0.7075 0.5552 (0.0907) 0.4645 0.0388 0.4257 0.0388	/therm
CAL LOW WINTER USE COST OF GAS RATE - 12/01/2020 Change in Rate due to change in underforer recovery C&I LOW WINTER USE COST OF GAS RATE - 01/01/2021 Change in Rate due to change in underforer recovery C&I LOW WINTER USE COST OF GAS RATE - 20/01/2021 Change in Rate due to change in underforer recovery C&I LOW WINTER USE COST OF GAS RATE - 20/01/2021 Change in Rate due to change in underforer recovery C&I LOW WINTER USE COST OF GAS RATE - 3/01/2021 Average Demand Cost of Gas Rate Effective 11/01/20 Times: Low Winter Use Ratio (Winter) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate Adjusted C&I Low Winter Use Cost of Gas Rate C&I HIGH WINTER USE COST OF GAS RATE - 1/10/1/20 Change in rate due to change in underfover recovery C&I HIGH WINTER USE COST OF GAS RATE - 2/10/1/2021 Change in Rate due to change in underfover recovery C&I HIGH WINTER USE COST OF GAS RATE - 2/10/1/2021 Change in Rate due to change in underfover recovery C&I HIGH WINTER USE COST OF GAS RATE - 2/10/1/2021 Change in Rate due to change in underfover recovery C&I HIGH WINTER USE COST OF GAS RATE - 2/10/1/2021 Change in Rate due to change in underfover recovery C&I HIGH WINTER USE COST OF GAS RATE - 3/10/1/2021 Change in Rate due to change in underfover recovery C&I HIGH WINTER USE COST OF GAS RATE - 3/10/1/2021 Change in Rate due to change in underfover recovery C&I HIGH WINTER USE COST OF GAS RATE - 3/10/1/2021 Average Demand Cost of Gas Rate Effective 11/10/1/20	1.0620 0.9984 \$ 0.1560 \$ 0.3733 \$ 0.0115 \$ 0.0252 \$ 0.5660	CO C	Gel	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	0.0907) 0.4753 0.0880 0.4365 0.0800 0.5245 0.0800 0.5245 0.07075 0.5552 0.0800 0.4645 0.0907) 0.4645 0.0800 0.5131	/therm
CAL LOW WINTER USE COST OF GAS RATE - 1/20/1/2020 Change in Rate due to change in underforer recovery C&I LOW WINTER USE COST OF GAS RATE - 01/01/2021 Change in Rate due to change in underforer recovery C&I LOW WINTER USE COST OF GAS RATE - 20/01/2021 Change in Rate due to change in underforer recovery C&I LOW WINTER USE COST OF GAS RATE - 20/01/2021 Change in Rate due to change in underforer recovery C&I LOW WINTER USE COST OF GAS RATE - 3/01/2021 Average Demand Cost of Gas Rate Effective 11/01/20 Times: Low Winter Use Ratio (Winter) Times: Correction Factor Adjusted Demand Cost of Gas Rate Effective 11/01/20 Times: Correction Factor Commodity Cost of Gas Rate Adjustment Cost of Gas Rate Adjusted Cost of Gas Rate Adjusted Cost of Gas Rate Adjusted Cost of Gas Rate Cost of Gas Rate Adjusted Cost of Gas Rate CAST HIGH WINTER USE COST OF GAS RATE - 11/01/20 Change in Rate due to change in underfover recovery C&I HIGH WINTER USE COST OF GAS RATE - 20/01/2021 Change in Rate due to change in underfover recovery C&I HIGH WINTER USE COST OF GAS RATE - 20/01/2021 Change in Rate due to change in underfover recovery C&I HIGH WINTER USE COST OF GAS RATE - 20/01/2021 Change in Rate due to change in underfover recovery C&I HIGH WINTER USE COST OF GAS RATE - 20/01/2021 Change in Rate due to change in underfover recovery C&I HIGH WINTER USE COST OF GAS RATE - 20/01/2021 Change in Rate due to change in underfover recovery C&I HIGH WINTER USE COST OF GAS RATE - 20/01/2021 Change in Rate due to change in underfover recovery C&I HIGH WINTER USE COST OF GAS RATE - 20/01/2021 Change in Rate due to change in underfover recovery C&I HIGH WINTER USE COST OF GAS RATE - 20/01/2021 Change in Rate due to change in Recovery C&I HIGH WINTER USE COST OF GAS RATE - 20/01/2021 Change in Rate Gas to the cost of Gas Rate Effective 11/01/20 Times: Correction Factor Advance Demand Cost of Gas Rate Effective 11/01/20 Times: Correction Factor Adjusted Demand Cost of	\$ 0.1471 0.9890 \$ 0.1450 \$ 0.1560 \$ 0.0158 \$ 0.0252 \$ 0.5660 \$ 0.5660	CO C	Gel	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	0.0907) 0.4753 0.0880 0.4365 0.0800 0.5245 0.0800 0.5245 0.07075 0.5552 0.0800 0.4645 0.0907) 0.4645 0.0800 0.5131	/therm
CAL LOW WINTER USE COST OF GAS RATE - 1/20/1/2020 Change in Rate due to change in underforer recovery CAL LOW WINTER USE COST OF GAS RATE - 01/01/2021 Change in Rate due to change in underforer recovery CAL LOW WINTER USE COST OF GAS RATE - 20/01/2021 Change in Rate due to change in underforer recovery CAL LOW WINTER USE COST OF GAS RATE - 20/01/2021 Change in Rate due to change in underforer recovery CAL LOW WINTER USE COST OF GAS RATE - 3/01/2021 Change in Rate due to change in underforer recovery CAL LOW WINTER USE COST OF GAS RATE - 3/01/2021 Average Demand Cost of Gas Rate Effective 11/01/20 Times: Low Winfer Use Ratio (Winfer) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjustment Cost of Gas Rate CAL HIGH WINTER USE COST OF GAS RATE - 11/01/20 Change in rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 20/10/2021 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 20/10/2021 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 20/10/2021 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 20/10/2021 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 20/10/2021 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 3/01/2021 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 3/01/2021 Change in Rate due to change in underfover recovery CAL HIGH WINTER USE COST OF GAS RATE - 3/01/2021 Times: Correction Factor Times: Correction Factor Times: Correction Factor Times: Correction Factor	1.0620 0.9984 \$ 0.1560 \$ 0.3733 \$ 0.0115 \$ 0.0252 \$ 0.5660 \$ 0.5660	CO C	Gel	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	0.0907) 0.4753 0.0880 0.4365 0.0800 0.5245 0.0800 0.5245 0.07075 0.5552 0.0800 0.4645 0.0907) 0.4645 0.0800 0.5131	/therm
CALLOW WINTER USE COST OF GAS RATE - 12/01/2020 Change in Rate due to change in underfover recovery Cat LOW WINTER USE COST OF GAS RATE - 01/01/2021 Change in Rate due to change in underfover recovery Cat LOW WINTER USE COST OF GAS RATE - 20/11/2021 Change in Rate due to change in underfover recovery Cat LOW WINTER USE COST OF GAS RATE - 20/11/2021 Change in Rate due to change in underfover recovery Cat LOW WINTER USE COST OF GAS RATE - 30/11/2021 Change in Rate due to change in underfover recovery Cat LOW WINTER USE COST OF GAS RATE - 40/11/201 Times: Low Winter Use Ratio (Winter) Times: Corection Factor Adjusted Demand Cost of Gas Rate Effective 11/01/20 Times: Low Winter Use Ratio (Winter) Times: Corection Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjustemet Cost of Gas Rate Indirect Cost of Gas Rate Adjustemet Cost of Gas Rate Adjustemet Cost of Gas Rate - 11/01/20 Change in Rate due to change in underfover recovery Cat High WINTER USE COST OF GAS RATE - 12/01/2020 Change in Rate due to change in underfover recovery Cat High WINTER USE COST OF GAS RATE - 30/10/201 Change in Rate due to change in underfover recovery Cat High WINTER USE COST OF GAS RATE - 30/10/201 Change in Rate due to change in underfover recovery Cat High WINTER USE COST OF GAS RATE - 30/10/201 Change in Rate due to change in underfover recovery Cat High WINTER USE COST OF GAS RATE - 30/10/201 Change in Rate due to change in underfover recovery Cat High WINTER USE COST OF GAS RATE - 30/10/201 Change in Rate due to change in underfover recovery Cat High WINTER USE COST OF GAS RATE - 30/10/201 Change in Rate due to change in underfover recovery Cat High WINTER USE COST OF GAS RATE - 30/10/201 Change in Rate due to change in underfover recovery Cat High WINTER USE COST OF GAS RATE - 30/10/201 Change in Rate due to change in underfover recovery Cat High WINTER USE C	1.0620 0.9984 \$ 0.1560 \$ 0.3733 \$ 0.0115 \$ 0.0252 \$ 0.5660 \$ 0.5660 \$ 0.4771 0.8890 0.9984 \$ 0.4452 \$ 0.4525 \$ 0.4525 \$ 0.0525 \$	CO C	Gel	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	0.0907) 0.4753 0.0880 0.4365 0.0800 0.5245 0.0800 0.5245 0.07075 0.5552 0.0800 0.4645 0.0907) 0.4645 0.0800 0.5131	/therm
CALLOW WINTER USE COST OF GAS RATE - 1/20/1/2020 Change in Rate due to change in underforer recovery CALLOW WINTER USE COST OF GAS RATE - 01/01/2021 Change in Rate due to change in underforer recovery CALLOW WINTER USE COST OF GAS RATE - 20/01/2021 Change in Rate due to change in underforer recovery CALLOW WINTER USE COST OF GAS RATE - 20/01/2021 Change in Rate due to change in underforer recovery CALLOW WINTER USE COST OF GAS RATE - 3/01/2021 Change in Rate due to change in underforer recovery CALLOW WINTER USE COST OF GAS RATE - 3/01/2021 Average Demand Cost of Gas Rate Effective 11/01/20 Times: Low Winter Use Ratio (Winter) Times: Correction Factor Adjusted Demand Cost of Gas Rate Commodity Cost of Gas Rate Adjusternat Cost of Gas Rate CALLOW Winter Use COST OF GAS RATE - 1/10/1/20 Change in rate due to change in underforer recovery CALLIGH WINTER USE COST OF GAS RATE - 2/10/1/2021 Change in Rate due to change in underforer recovery CALLIGH WINTER USE COST OF GAS RATE - 2/10/1/2021 Change in Rate due to change in underforer recovery CALLIGH WINTER USE COST OF GAS RATE - 2/10/1/2021 Change in Rate due to change in underforer recovery CALLIGH WINTER USE COST OF GAS RATE - 2/10/1/2021 Change in Rate due to change in underforer recovery CALLIGH WINTER USE COST OF GAS RATE - 2/10/1/2021 Change in Rate due to change in underforer recovery CALLIGH WINTER USE COST OF GAS RATE - 2/10/1/2021 Change in Rate due to change in underforer recovery CALLIGH WINTER USE COST OF GAS RATE - 2/10/1/2021 Change in Rate due to change in underforer recovery CALLIGH WINTER USE COST OF GAS RATE - 2/10/1/2021 Change in Rate due to change in underforer recovery CALLIGH WINTER USE COST OF GAS RATE - 2/10/1/2021 Change in Rate due to change in underforer recovery CALLIGH WINTER USE COST OF GAS RATE - 2/10/1/2021 Change in Rate due to change in underforer recovery CALLIGH WINTER USE COST OF GAS RATE - 2/10/1/2021 Change in R	\$ 0.1471 0.9890 0.9984 \$ 0.1560 \$ 0.0753 \$ 0.0252 \$ 0.5660 \$ 0.5660	CO C	Gel	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	0.0907) 0.4753 0.0880 0.4365 0.0800 0.5245 0.0800 0.5245 0.07075 0.5552 0.0800 0.4645 0.0907) 0.4645 0.0800 0.5131	/therm

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman
EFFECTIVE: August 1, 2021
TITLE: President

3031.1 ANTICIPATED COST OF GAS - EXCLUDING KEENE CUSTOMERS

Anticipated Cost of Gas

PERIOD COVERED: WINTER PERIOD, NOVEMBER 1, 2020 THROUGH APRIL 30, 2021 (REFER TO TEXT ON IN SECTION 16 COST OF GAS CLAUSE)

(Col 1)		(Col 2)		(Col 3)
ANTICIPATED DIRECT COST OF GAS				
Purchased Gas:				
Demand Costs:	\$	12,022,922		
Supply Costs:		28,276,980		
Storage Gas:				
Demand, Capacity:	\$	955,766		
Commodity Costs:		3,064,149		
Produced Gas:		1,590,589		
Hedged Contract (Saving)/Loss		_		
Hedge Underground Storage Contract (Saving)/Loss		<u> </u>		
Unadjusted Anticipated Cost of Gas			\$	45,910,407
Adjustments:				
Prior Period (Over)/Under Recovery (as of 05/01/20)	\$	2,227,421		
Interest	•	72,812		
Fuel Inventory Revenue Requirement		441,037		
Broker Revenues		(32,725)		
Refunds from Suppliers		(02,720)		
Fuel Financing		_		
Transportation CGA Revenues		(4,516)		
Interruptible Sales Margin		(.,510)		
Capacity Release and Off System Sales Margins		(1,736,581)		
Hedging Costs		(1,755,561)		
Fixed Price Option Administrative Costs		45,000		
Total Adjustments	-	40,000		1,012,447
Total Anticipated Direct Cost of Gas			\$	46,922,854
Anticipated Indirect Cost of Gas				
Working Capital:				
Total Unadjusted Anticipated Cost of Gas 11/01/20 - 04/30/21	\$	45,910,407		
	Φ	0.0391		
Working Capital Rate: Lead Lag Days / 365 Prime Rate		3.25%		
		0.127%		
Working Capital Percentage Working Capital	\$	58,347		
Diver Westing Control Decembration (Acad 442.20)		(66 927)		
Plus: Working Capital Reconciliation (Acct 142.20)		(66,837)		(0.400)
Total Working Capital Allowance				(8,490)
Bad Debt:				
Total Unadjusted Anticipated Cost of Gas 11/01/20 - 04/30/21	\$	45,910,407		
Less: Refunds		-		
Plus: Total Working Capital		(8,490)		
Plus: Prior Period (Over)/Under Recovery		2,227,421		
Subtotal	\$	48,129,338		
Bad Debt Percentage		1.11%		
Bad Debt Allowance	\$	534,236		
Plus: Bad Debt Reconciliation (Acct 175.52)	Ψ	(296,628)		
Total Bad Debt Allowance		(250,020)	\$	237.608
			-	20.,000
Production and Storage Capacity			\$	1,980,428
Miscellaneous Overhead 11/01/20 - 04/30/21	\$	13,170		
Times Winter Sales		89,365		
Divided by Total Sales		111,369		
Miscellaneous Overhead				10,568
Total Anticipated Indirect Cost of Gas			\$	2,220,114
·				
Total Cost of Gas			\$	49,142,968

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX	, 2021, in Docket N	o. DG 20-105
ISSUED:	November 13, 2020	ISSUED BY:	/c/Sugan I Flools
		IDDUED DI.	/S/Dusan L. I ICCK
		ISSUED D1.	Susan L. Fleek
EFFECTIVE:	November 01, 2020	TITLE:	Sucan I Flook

Seventeenth Revised Original Page 9397 Superseding Sixteenth Revised Page 93

31. CALCULATION OF FIRM SALES AND FIXED WINTER PERIOD COST OF GAS RATE - KEENE CUSTOMERS

Calculation of the Cost of Gas Rate

Period Covere	ed:	Winter Period November 1, 2020 throu	gh April 30, 2021			
Projected Ga	s Sales - therms				1,108,419	
CNG Costs N	ted Cost of Gas Sendo Not Included In Cost of ated Cost of Gas Sendo	Gas Rate		-	\$1,310,950 (\$204,754) \$1,106,196	
Add:	Prior Period Deficienc Interest	y Uncollected	\$	30,171 \$136		
Deduct:	Prior Period Excess (Interest	Collected		\$0 \$0		
	Prior Period Adjustme	ents and Interest			\$30,307	
Total Anticipa		8 in Docket No. DG 20-152 dated Dec	ombor 02 2020	-	\$1,136,503	
Costs Aujust	ed per Older No. 20,42	o III Docket No. DG 20-132 dated Dec	ember 02, 2020		Excluding Gas Assistance	Gas Assistance
Cost of Gas I	<u>Rate</u>			-	Program	Program
Non-Fixed Pr	ice Option Cost of Gas	Rate - Beginning Period (per therm)			\$1.0253	<u>\$0.5639</u>
Mid Period A	djustment - December	01, 2020			\$0.0000	\$0.0000
	djustment - January 01				(\$0.0283)	(\$0.0156)
	djustment - February 0 diustment - March 01.				(\$0.0042) \$0.0000	(\$0.0023) \$0.0000
	djustment - March 01, djustment - April 01, 20				\$0.2888	\$0.000
Revised Non-	Fixed Price Option Co	st of Gas Rate - Effective April 1, 2021	(per therm)	-	\$1.2816	\$0.7048
Fixed Price C	Option Cost of Gas Rate	e (per therm) November 2020			\$1.2300	\$0.6765
		e (per therm) December 2020-April 202	21		\$1.0277	<u>\$0.5652</u>
		Company may adjust the approved y basis to the following rate:				
Maximum Co	ost of Gas Rate - Non-F	ixed Price Option (per therm)			<u>\$1.2816</u>	<u>\$0.7049</u>

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX XX	X, 2021, in Docket N	o. DG 20-105
ISSUED:	March 24, 2021	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleek
EFFECTIVE:	April 01, 2021	TITLE:	President
	1 /		

Seventeenth Revised Original Page 9397 Superseding Sixteenth Revised Page 93

\$1.2816

———Winter Cost of Gas

Calculation of the Cost of Gas Rate

Period Cov	ered: Winter Period November 1, 2020 through April 30, 2021			
Projected (Gas Sales - therms		1,108,419	
CNG Costs	lated Cost of Gas Sendout Not Included In Cost of Gas Rate pated Cost of Gas Sendout		\$1,310,950 (\$204,754) \$1,106,196	
Add:	Prior Period Deficiency Uncollected Interest	\$30,171 \$136		
Deduct:	Prior Period Excess Collected Interest	\$0 \$0	-	
	Prior Period Adjustments and Interest		\$30,307	
Costs Adju	pated Cost sted per Order No. 26,428 in Docket No. DG 20-152 dated December 02, 2020		\$1,136,503 Excluding Gas Assistance	
Cost of Ga	s Rate		Program	Program
Non-Fixed	Price Option Cost of Gas Rate - Beginning Period (per therm)		<u>\$1.0253</u>	<u>\$0.5639</u>
Mid Period Mid Period Mid Period	Adjustment - December 01, 2020 Adjustment - January 01, 2021 Adjustment - February 01, 2021 Adjustment - March 01, 2021 Adjustment - April 01, 2021		\$0.0000 (\$0.0283) (\$0.0042) \$0.0000 \$0.2888	\$0.0000 (\$0.0156) (\$0.0023) \$0.0000 \$0.1588
Revised No	n-Fixed Price Option Cost of Gas Rate - Effective April 1, 2021 (per therm)		\$1.2816	\$0.7048
	e Option Cost of Gas Rate (per therm) November 2020 e Option Cost of Gas Rate (per therm) December 2020-April 2021		\$1.2300 <u>\$1.0277</u>	\$0.6765 <u>\$0.5652</u>
	tariff section 17(d), the Company may adjust the approved rate upward on a monthly basis to the following rate:			

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX X	X, 2021, in Docket N	o. DG 20-105
ISSUED:	March 24, 2021	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleek
EFFECTIVE:	April 01, 2021	TITLE:	President

Maximum Cost of Gas Rate - Non-Fixed Price Option (per therm)

\$0.7049

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 98 Firm Transportation Cost of Gas

33. CALCULATION OF FIRM TRANSPORTATION COST OF GAS RATE

II. RATE SCHEDULES

Calculation of Firm Transportation Cost of Gas Rate
PERIOD COVERED: WINTER PERIOD, NOVEMBER 1, 2020 THROUGH APRIL 30, 2021
(Refer to text in Section 16(Q) Firm Transportation Cost of Gas Clause)

(Col 1)	(Col 2)	(Col 3)	(Col 4)
ANTICIPATED COST OF SUPPLEMENTAL GAS SUPPLIES:			
PROPANE	\$ 568,511		
LNG	1,022,078		
TOTAL ANTICIPATED COST OF SUPPLEMENTAL GAS SUPPLIES ESTIMATED PERCENTAGE USED FOR PRESSURE SUPPORT PURPOSES ESTIMATED COST OF LIQUIDS USED FOR PRESSURE SUPPORT PURPOSES	1,590,589 <u>8.7%</u> \$ 138,381		
PROJECTED FIRM THROUGHPUT (THERMS): FIRM SALES FIRM TRANSPORTATION SUBJECT TO FTCG TOTAL FIRM THROUGHPUT SUBJECT TO COST OF GAS CHARGE	89,364,968 42,456,275 131,821,243	67.8% <u>32.2%</u> 100.0%	
TRANSPORTATION SHARE OF SUPPLEMENTAL GAS SUPPLIES	32.2%	x \$138,381 =	\$ 44,569
PRIOR (OVER) OR UNDER COLLECTION			(40,053)
NET AMOUNT TO COLLECT FROM (RETURNED TO) TRANSPORTATION CUSTOMERS			\$ 4,516
PROJECTED FIRM TRANSPORTATION THROUGHPUT			42,456,275
FIRM TRANSPORTATION COST OF GAS			\$ 0.0001

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 412 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 99 Environmental Surcharge

34. ENVIRONMENTAL SURCHARGE - MANUFACTURED GAS PLANTS

Environmental Surcharge - Manufactured Gas Plants

Manufactured Gas Plants

Required Annual Environmental Increase	\$2,864,179
First one-third of prior period under recoveries (through June 2019)	\$341,389
July 2019 - June 2020 recovery difference between actual and estimate	\$338,564
Environmental Subtotal	\$3,544,132
Overall Annual Net Increase to Rates	\$3,544,132
Estimated weather normalized firm therms billed for the twelve months ended 10/31/20 - sales and transportation Surcharge per therm	179,574,679 therms \$0.0197 per therm
Total Environmental Surcharge	\$0.0197

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman
Neil Proudman
TITLE: Provident

Docket No. DG 21-130 Exhibit 26 Page 413 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11GAS LIBERTY UTILITIES

Original Page 100 Rate Case Expense

35 RATE CASE EXPENSE AND RECOUPMENT FACTOR CALCULATION

Liberty Utilities (Energy North Natural Gas) Corp. d/b/a Liberty Local Delivery Adjustment Charge (LDAC) increase due to Rate Case Expense and Recoupment For LDAC effective November 1, 2020 - October 31, 2021

Rate Case Expense Remaining from Docket No. DG 17-048 Recoupment Remaining from Docket No. DG 17-048 July 1, 2020 Balance	\$87,069 <u>\$0</u> \$87,069
Plus Estimated Interest from July 2020 through October 2020	\$745
Minus Estimated Recoveries from July 2020 through October 2020	<u>(\$43,733)</u>
Total Estimated Remaining Recovery As of November 1, 2020	\$44,081
Estimated November 2019 - October 2020 Interest	<u>\$538</u>
Total Remaining Recovery	\$44,619
Estimated November 2020 - October 2021 Sales (therms)	179,574,679
RCE & Recoupment rate per therm November 2020 - October 2021	\$0.0002

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman Neil Proudman

TITLE: EFFECTIVE: August 1, 2021 President

Original Page 101 LDAC

36 LOCAL DISTRIBUTION ADJUSTMENT CLAUSE CALCULATION

Local Delivery Adjustment Charge Calculation

Residential Non Heating Rates - R-1, R-5	0.0004	Sales Customers	Transportation Customers	
Energy Efficiency Charge Demand Side Management Charge	\$ 0.0831 0.0000			
Conservation Charge (CCx) Relief Holder and pond at Gas Street, Concord, NH	0.0000	\$ 0.0831	1	
Manufactured Gas Plants	 0.0197			
Environmental Surcharge (ES)		0.0197		
Revenue Decoupling Adjustment Factor (RDAF) Energy Efficiency Resource Standard Lost Revenue Mechanism		(0.0562 0.0000		
Rate Case Expense Factor (RCEF)		0.0002	2	
Gas Assistance Program (GAP) LDAC	-	\$ 0.0589		er therm
Residential Heating Rates - R-3, R-4, R-6, R-7				
Energy Efficiency Charge	\$ 0.0831			
Demand Side Management Charge	 0.0000	\$ 0.0831	1	
Conservation Charge (CCx) Relief Holder and pond at Gas Street, Concord, NH	0.0000	\$ 0.0831	1	
Manufactured Gas Plants	 0.0197			
Environmental Surcharge (ES)	<u>-</u>	0.0197		
Revenue Decoupling Adjustment Factor (RDAF) Energy Efficiency Resource Standard Lost Revenue Mechanism		(0.0562 0.0000		
Rate Case Expense Factor (RCEF)		0.0002		
Gas Assistance Program (GAP)	_	0.0121	<u>1_</u>	
LDAC		\$ 0.0589	9 р	er therm
Commercial/Industrial Low Annual Use Rates - G-41, G-51, G-44, G-55				
Energy Efficiency Charge	\$0.0441			
Demand Side Management Charge	 0.0000			
Conservation Charge (CCx) Relief Holder and pond at Gas Street, Concord, NH	0.0000	\$ 0.0441	1 \$ 0.0441	
Manufactured Gas Plants	0.0197			
Environmental Surcharge (ES)	 	0.0197		
Revenue Decoupling Adjustment Factor (RDAF)		(0.0206		
Energy Efficiency Resource Standard Lost Revenue Mechanism Rate Case Expense Factor (RCEF)		0.0000		
Gas Assistance Program (GAP)		0.0002		
LDAC	•	\$ 0.0555	5 \$ 0.0555 p	er therm
Commercial/Industrial Medium Annual Use Rates - G-42, G-52, G-45, G-56				
Energy Efficiency Charge	\$0.0441			
Demand Side Management Charge Conservation Charge (CCx)	 0.0000	\$0.0441	1 \$0.0441	
Relief Holder and pond at Gas Street, Concord, NH	0.0000	Ψ0.044	ψο.ο441	
Manufactured Gas Plants	 0.0197			
Environmental Surcharge (ES)		0.0197		
Revenue Decoupling Adjustment Factor (RDAF) Energy Efficiency Resource Standard Lost Revenue Mechanism		(0.0206 0.0000		
Rate Case Expense Factor (RCEF)		0.0002		
Gas Assistance Program (GAP)	_	0.0121		
LDAC		\$ 0.0555	5 \$ 0.0555 p	er therm
Commercial/Industrial Large Annual Use Rates - G-43, G-53, G-54, G-46, G-56, G-57,G-58				
Energy Efficiency Charge	\$ 0.0441			
Demand Side Management Charge	 0.0000	00.044	4 000444	
Conservation Charge (CCx) Relief Holder and pond at Gas Street, Concord, NH	0.0000	\$0.0441	1 \$0.0441	
Manufactured Gas Plants	 0.0197			
Environmental Surcharge (ES)		0.0197		
Revenue Decoupling Adjustment Factor (RDAF)		(0.0206		
Energy Efficiency Resource Standard Lost Revenue Mechanism Rate Case Expense Factor (RCEF)		0.0000		
Gas Assistance Program (GAP)		0.0121		
LDAC	-	\$ 0.0555	5 \$ 0.0555 p	er therm

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 415 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 102 LDAC – Residential Energy Efficiency Loan Program

37 Residential Energy Efficiency Loan Program

Availability

Subject to the Terms and Conditions of the Tariff of which it is a part, this program shall allow Customers installing energy-efficiency measures under an energy efficiency program offered by the Company and approved by the Commission ("Participating Customers") to borrow all or a portion of the Customer's share of the installed cost of the energy-efficiency measures ("Customer Loan Amount") from the Company and to repay the Customer Loan Amount through an additional charge on their monthly retail delivery service bill issued by the Company. It is available to Participating Customers who meet the following qualifications:

- A. The Participating Customer must own the property where the energy-efficiency measures are installed; and
- B. A Participating Customer must have an active Delivery Service account with the Company for the property where the energy-efficiency measures are installed and receive Delivery Service under R-1, R-3, R-4, R-5, R-6, R-7; and
- C. The Participating Customer must not have received a disconnect notice from the Company during the twelve months preceding the Participating Customer's request for a loan under this program; and
- D. The Customer Loan Amount has no minimum and must be less than or equal to \$4,000 per customer per year, and must not exceed the Participating Customer's share of the installed cost of the energy efficiency measures installed under the Company's approved energy-efficiency program; and
- E. The Participating Customer must meet the qualifications of the applicable energy-efficiency program through which the energy-efficiency measures are being installed.

At its sole discretion, the Company shall determine eligibility for service under this program subject to the availability of program funds.

Any Participating Customer receiving a loan under this program must remain a Delivery Service customer of the Company at the property where the energy-efficiency measures are installed until the loan has been repaid in full. In the event the Participating Customer ceases to be a Delivery Service Customer of the Company at the property where the energy-efficiency measures are installed, any remaining charges under this program shall immediately become due and payable.

Customer Loan Agreement

Participating Customers shall be required to execute a separate Residential Customer Loan Agreement which will specify the fixed monthly charge and other applicable terms. A Participating Customer can choose to pay the remaining balance owed to the Company at any time. A late payment charge as described in the Terms and Conditions for Delivery Service section of the Company's Tariff is applicable to the monthly charges rendered under this program. Participating Customers are not subject to disconnection of natural gas service for nonpayment of the charges under this program.

The Customer Loan Amount shall be paid to the Company by the Participating Customer through a fixed monthly charge applied over a term of months as established in the Customer Loan Agreement. Participating Customers may specify the repayment term of the Customer Loan Amount subject to the maximum repayment term limit of 48 months.

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 416 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u>—GAS LIBERTY UTILITIES

Original Page 97B103 LDAC – Non-Residential Energy Efficiency Loan Program

38 Non-Residential Energy Efficiency Loan Program

Availability

Subject to the Terms and Conditions of the Tariff of which it is a part, this program shall allow Customers installing energy-efficiency measures under an energy efficiency program offered by the Company and approved by the Commission ("Participating Customers") to borrow all or a portion of the Customer's share of the installed cost of the energy-efficiency measures ("Customer Loan Amount") from the Company and to repay the Customer Loan Amount through an additional charge on their monthly retail delivery service bill issued by the Company. It is available to Participating Customers who meet the following qualifications:

- A. The Participating Customer must own the property where the energy-efficiency measures are installed; and
- B. A Participating Customer must have an active Delivery Service account with the Company for the property where the energy-efficiency measures are installed and receive Delivery Service under G-41, G-42, G-43, G-44, G-45, G-46, G-51, G-52, G-53, G-54, G-55, G-56, G-57, or G-58; and
- C. The Participating Customer must not have received a disconnect notice from the Company during the twelve months preceding the Participating Customer's request for service under this program; and
- D. The Customer Loan Amount has no minimum and must be less than or equal to \$50,000 for each project, and must not exceed the Participating Customer's share of the installed cost of the energy efficiency measures installed under the Company's approved energy-efficiency program; and
- E. A Participating Customer is limited to \$150,000 per year in loan funds with no limit on the number of projects at the sole discretion of the Company based on program demand. If at any point there are no loan fund recipients or there have been no loan fund recipients in a given year, the Company may petition the Commission to allow a particular customer to receive more than \$150,000 in loan funds in a given year; and
- F. The Participating Customer must meet the qualifications of the applicable energy-efficiency program through which the energy-efficiency measures are being installed.

At its sole discretion, the Company shall determine eligibility for service under this program subject to the availability of program funds.

Any Participating Customer receiving a loan under this program must remain a Delivery Service customer of the Company at the property where the energy-efficiency measures are installed until the loan has been repaid in full. In the event the Participating Customer ceases to be a Delivery Service Customer of the Company at the property where the energy-efficiency measures are installed, any remaining charges under this program shall immediately become due and payable.

Customer Loan Agreement

Participating Customers shall be required to execute a separate Non-Residential Customer Loan Agreement which will specify the fixed monthly charge and other terms of the loan. A Participating Customer can choose to pay the remaining balance owed to the Company at any time. A late payment charge as described in the Terms and Conditions for Delivery Service section of the Company's Tariff is applicable to the monthly charges rendered under this program. Participating Customers are not subject to disconnection of natural gas service for nonpayment of the charges under this program.

The Customer Loan Amount shall be paid to the Company by the Participating Customer through a fixed monthly charge applied over a term of months as established in the Customer Loan Agreement. Participating Customers may specify the repayment term of the Customer Loan Amount subject to the maximum repayment term limit of 60 months.

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman
Neil Proudman
EFFECTIVE: August 1, 2021 TITLE: President

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 104 Delivery Terms and Conditions

III. DELIVERY TERMS AND CONDITIONS

1 RATES AND CHARGES

- 1.1 The Company shall apply this tariff on a non-discriminatory and non-preferential basis to all Customers who obtain service from the Company, except as this tariff is explicitly modified by order of the NHPUC. The provisions of Part III Section 20 of this tariff will specifically apply to all entities designated by the Customer as set forth in Section 20.5 to supply Gas to a Designated Receipt Point for the Customer's account.
- 1.2 The Company reserves the right to impose reasonable fees and charges pursuant to the various provisions of this tariff.
- 1.3 In the event that the Company incurs minimum bill, inventory, transition, take or pay, imbalance, or any other charges associated with the provision of Delivery Service to Customers, the Company may impose an additional charge on the Suppliers serving said Customers as approved by the NHPUC.

2 DEFINITIONS

Adjusted Target Volume

("ATV)

The volume of Gas determined by the Company using a Consumption Algorithm and required to be nominated and delivered each Gas Day by the Supplier on behalf of Customers taking non-daily metered Delivery Service.

Aggregation Pool

One or more Customer accounts whose Gas Usage is served by the same Supplier and aggregated pursuant to Section 20.6 of this tariff for operational purposes, including but not limited to nominating, scheduling, and balancing Gas deliveries to Designated Receipt Point(s) within the associated Gas Service Area.

Annual Reassignment Date

Five (5) Business Days prior to November 1 of each year when the Company reassigns Capacity to Suppliers pursuant to Section 11.6 of this tariff.

Assignment Date

Five (5) Business Days prior to the first Gas Day of each month when the Company assigns Capacity to Suppliers pursuant to Section 11.4 of this tariff.

Authorization Number

A number unique to the Customer generated by the Company and printed on the Customer's bill that the Customer must furnish to the Supplier to enable the Supplier to obtain the Customer's Gas Usage information pursuant to Section 20.4, and to initiate or terminate Supplier Service as set forth in Section 20.5 of this tariff.

Btu

One British thermal unit; i.e., the amount of heat required to raise the temperature of one pound of water one degree Fahrenheit at sixty degrees (60°) Fahrenheit.

Business Day

Monday through Friday excluding holidays recognized by the Company. Where relevant, a Business Day shall consist of the hours during which the Company is open for business with the public. <u>If any performance date referenced in this Tariff is not a Business Day, such performance shall be the next succeeding Business Day.</u>

Capacity

Pipeline Capacity, Storage Withdrawal Capacity, and Peaking Capacity as defined in this tariff.

DATED: XXX XX, 2021

ISSUED BY: /s/Neil Proudman
Neil Proudman
TITLE: President

EFFECTIVE: August 1, 2021

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Docket No. DG 21-130 Exhibit 26 Page 418 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES Original Page 99105

Delivery Terms and Conditions

Capacity Allocators The estimated proportions of the Customer's Total Capacity Quantity that

comprise Pipeline Capacity, Storage Withdrawal Capacity and Peaking

Capacity.

Capacity Mitigation Service The service available to Suppliers in accordance with Section 11.10.

City Gate The interconnection between a Delivering Pipeline and the Company's

distribution facilities.

Commodity See Gas.

Company Liberty Utilities (EnergyNorth Natural Gas) Corp. d/b/a Liberty Utilities

Company Gas Allowance The difference between the sum of all amounts of Gas received into the

Company's distribution system (including Gas produced by the Company) and the sum of all amounts of Gas delivered from the Company's distribution system divided by said amount of Gas received. Such difference shall include but not be limited to Gas consumed by the Company for its own purposes, line losses, and Gas vented and lost as a result of force majeure, excluding Gas

otherwise accounted for.

Company-Managed Supplies Capacity and Supply contracts held and managed by the Company and made

available to the Supplier pursuant to Section 11.9 of this tariff including

Supply-sharing contracts and load-management contracts.

Consumption Algorithm A mathematical formula used to estimate a Customer's daily consumption.

Critical Day In accordance with Section 16 of this tariff, a day declared at any time by the

Company in its reasonable discretion when unusual operating conditions may

jeopardize operation of the Company's distribution system.

Customer The recipient of Delivery Service whose Gas Usage is recorded by a meter or

group of meters at a specific location and who is a customer of record of the

Company.

Daily Baseload The Customer's average usage per Gas Day that is assumed to be unrelated to

weather.

Daily Index The mid-point of the range of prices as published by Gas Daily under the

heading "Daily Price Survey, Midpoint, Citygates, Tennessee/Zone 6

(delivered)" for the relevant Gas Day listed under "Flow date(s)".

In the event that the <u>Gas Daily</u> index becomes unavailable, the Company shall apply its daily marginal cost of Gas as the basis for this calculation until such

time that the NHPUC approves a suitable replacement.

Dekatherm Ten Therms.

Delivery Point The interconnection between the Company's facilities and the Customer's

facilities.

Delivery Service The distribution of Gas by the Company on any Gas Day from the Designated

Receipt Point to the Customer's Delivery Point and related Customer services.

Design Peak Season The forecasted Peak Season during which the Company's system experiences

the highest aggregate Gas Usage.

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman
Neil Proudman
EFFECTIVE: August 1, 2021 TITLE: President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

ISSUED: May 18, 2018 ISSUED BY: /s/Susan L. Fleek
Susan L. Fleek

EFFECTIVE: May 01, 2018 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 419 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES Original Page 99105
Delivery Terms and Conditions

Designated Receipt Point

For each Customer, the Company designated interconnection between a Transporting Pipeline and the Company's distribution facilities at which

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX X	X, 2021, in Docket N	o. DG 20-105
ISSUED:	May 18, 2018	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleek
EFFECTIVE:	May 01, 2018	TITLE:	Susan L. Fleek President

Docket No. DG 21-130 Exhibit 26 Page 420 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 106 Delivery Terms and Conditions

point, or such other point as the Company may designate from time to time for operational purposes, the

Supplier will make deliveries of Gas for the Customer's account.

Designated Representative The designated representative of the Customer, who shall be authorized to act

for, and conclusively bind, the Customer regarding Delivery Service in

accordance with the provisions of Section 21 of this tariff.

Gas Natural Gas that is received by the Company from a Transporting Pipeline at

the Designated Receipt Point and delivered by the Company to the Delivery Point for the Customer's account. In addition, the term shall include amounts of vaporized liquefied natural Gas and/or propane-air vapor that are introduced by the Company into its system and made available to the Customer as the equivalent of natural Gas that the Customer is otherwise

entitled to have delivered by the Company.

Gas Day A period of twenty-four (24) consecutive hours beginning at 10:00 a.m., E.T.,

and ending at 10:00 a.m., E.T., the next calendar day, or other such hours used

by the Transporting Pipeline.

Gas Service Area An area within the Company's distribution system as defined in Section 4 of

this tariff, for the purposes of administering Capacity assignments, Nominations, balancing, imbalance trading, and Aggregation Pools.

Gas Usage The actual quantity of Gas used by the Customer as measured by the

Company's metering equipment at the Delivery Point.

Heating Degree Day A measure used to estimate weather-sensitive Gas consumption calculated by

subtracting the average temperature for each day from the number 65. Each degree day that represents a degree below 65 is considered a Heating Degree

Day.

Heating Factor The Customer's estimated weather-sensitive Gas consumption per Heating

Degree Day.

MMBtu One million Btus.

Maximum Daily Peaking The portion of a Customer's Total Capacity Quantity identified

Quantity ("MDPQ") and allocated as Peaking Capacity, such that the maximum daily amount of

Gas that can be withdrawn from a Supplier's Peaking Service Account pursuant to Section 14 of this tariff shall be equal to the sum of the MDPQs

for all Customers in that Supplier's Aggregation Pool.

Month A calendar month of Gas Days.

Monthly Index The average of the Daily Index numbers for all Gas Days in a Month.

NHPUC The New Hampshire Public Utilities Commission.

Nomination The notice given by the Supplier to the Company that specifies, in accordance

with the Standard Nomination Form attached as Attachment A, an intent to deliver a quantity of Gas to the Designated Receipt Point(s) on behalf of one or more Customers, including the volume to be received, the Designated Receipt Point(s), the Transporting Pipeline, the delivering contract(s), the shipper, and other such non-confidential information as may be reasonably

required by the Company.

Off-Peak Season The consecutive months of May to October, inclusive.

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 421 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 107 Delivery Terms and Conditions

Operational Flow Order

The Company's instructions to the Supplier to take such action as conditions

("OFO")

require including, but not limited to, diverting Gas to or from the Company's

distribution system pursuant to Section 16 of this tariff.

Peak Day

The forecasted Gas Day during which the Company's system experiences the

highest aggregate Gas Usage.

Peak Season

The consecutive months of November to April, inclusive.

Peaking Capacity

Capacity in addition to upstream pipeline and underground storage Capacity normally used by the Company to meet daily requirements during a Design

Peak Season and acquired specifically for the Peak Season.

Peaking Service

A Company-managed resource consisting of Peaking Capacity and Peaking

Supply.

Peaking Service Account

An account whose balance indicates the total volumes of Peaking Service resources available to a Supplier, where the maximum balance in the account shall equal the Peaking Supply assigned to the Supplier pursuant to this tariff.

Peaking Service Rule Curve A system of operational parameters associated with the use of the Company's Peaking Capacity including, but not limited to, indicators of the necessary levels of Peaking Supply that must be maintained in Suppliers' Peaking Service Accounts in order for the Company to meet system demands under Design Peak Season conditions. The Company will communicate, by electronic means as determined by the Company or, in the event of failure of such electronic means, by facsimile or other agreeable alternative means, the Peaking Service Rule Curve as identified in Section 14 of this tariff.

Peaking Supply

The aggregate amount of Supply in excess of upstream pipeline and underground storage Supply required to meet the Company's forecasted Supply needs during a Design Peak Season and acquired specifically for the Peak Season.

Peaking Supply Allocator

An allocation factor that represents the proportion of a Customer's estimated Gas Usage during the Design Peak Season that is generally served with Peaking Service supplies.

Pipeline Capacity

Transportation capacity on interstate pipeline systems normally used for deliveries of Gas to the Company's city gates, exclusive of Storage Withdrawal Capacity.

Pre-Determined Allocation

Instructions from the Supplier to the Company for the method allocation of discrepancies in confirmed Nominations among the Supplier's Aggregation Pools and/or Customers as set forth in the Supplier Service Agreement.

Rate Schedule

The schedule of rates included in this tariff.

Reference Period

A period of at least twelve (12) months for which a Customer's Gas Usage information is typically available to the Company.

Sales Service

Commodity service provided on a firm basis to a Customer who is not receiving Supplier Service, in accordance with the provisions set forth in this tariff. The provision of Sales Service shall be the responsibility of the Company and shall be provided to the Customer by the Company or its designated Supplier pursuant to law or regulation.

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman Neil Proudman TITLE: President

EFFECTIVE: August 1, 2021

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Docket No. DG 21-130 Exhibit 26 Page 422 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES Original Page <u>102108</u> Delivery Terms and Conditions

Seasonal Storage Capacity Contracts for Capacity in off-system storage facilities used to accumulate and

maintain Gas inventories for re-delivery to the Company's city gates normally

during the Peak Season.

Storage Withdrawal Capacity Capacity for the withdrawal of Gas inventories maintained in off-system

storage facilities, as well as the Pipeline Capacity used to deliver such Gas to

the Company's city gates.

Supplier Any entity that has met the Company's requirements set forth in Section 20

of this tariff and that has been designated by a Customer to supply Gas to a Designated Receipt Point for the Customer's account; provided, however, that a Customer may act as its own Supplier in accordance with Section 5.2 of this

tariff.

Supplier Service The sale of Gas to a Customer by a Supplier.

Supplier Service Agreement An agreement, substantially in the form set forth in Attachment A, which must

be executed by the Company and a Supplier in order for the Supplier to serve

Customers on the Company's system.

Supply See Gas.

Therm An amount of Gas having a thermal content of 100,000 Btus.

Total Capacity Quantity The total amount of Capacity assignable to a Supplier on behalf

("TCQ") of a Customer.

Transporting Pipeline The interstate pipeline company that transports and delivers Gas to the

Designated Receipt Point.

3 CHARACTER OF SERVICE

- 3.1 All rates within Part II Rate Schedule are predicated upon service to a Customer at a single Delivery Point and metering installation, except as otherwise specifically provided by a given rate. Where service is supplied to a Customer at more than one Delivery Point or metering installation, each single Delivery Point or metering installation shall be considered to be a separate Customer for purposes of applying the Rate Schedule, except when a Customer is served through multiple points of delivery or metering installations for the Company's own convenience.
- 3.2 The Company may refuse to supply service to loads of unusual characteristics which, in its sole reasonable judgment, might adversely affect the quality of service supplied to other Customers, the public safety or the safety of the Company's personnel. In lieu of such refusal, the Company may require a Customer to install any necessary regulating and protective equipment in accordance with the requirements and specifications of the Company.

4 GAS SERVICE AREAS AND DESIGNATED RECEIPT POINTS

- 4.1 There shall be 1 Gas Service Area defined for purposes of administering Capacity assignments, Nominations, balancing, imbalance trading, and Aggregation Pools pursuant to this tariff. Each such Gas Service Area shall be defined to include the municipalities listed within each such Gas Service Area, as follows:
 - (1) Area 1: Liberty Utilities (EnergyNorth Natural Gas) Corp. d/b/a Liberty-Utilities.

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX	XX, 2021, in Docket No	o. DG 20-105
ICCLIED:	Mary 19, 2019		
IDDUED.	May 18, 2018	ISSUED BY:	/s/Susan L. Fleck
ISSUED.		ISSUED BY:	/s/Susan L. Fleek Susan L. Fleek
EFFECTIVE:		ISSUED BY: TITLE:	

Docket No. DG 21-130 Exhibit 26 Page 423 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. <u>10-11</u> GAS LIBERTY UTILITIES Original Page <u>102108</u> Delivery Terms and Conditions

The area authorized to be served by the Company and to which this tariff applies are the following cities and towns: Allenstown, Amherst, Auburn, Bedford, Belmont, Berlin,

DATED:	XXX XX, 2021	ISSUED BY:	/s/Neil Proudman
			Neil Proudman
EFFECTIVE:	August 1, 2021	TITLE:	President
	Authorized by NHPUC Order No. XX,XXX dated XXX X	XX, 2021, in Docket N	o. DG 20-105
ISSUED:	May 18, 2018	ISSUED BY:	/s/Susan L. Fleek
			Susan L. Fleek
EFFECTIVE:	May 01, 2018	TITLE:	President

Docket No. DG 21-130 Exhibit 26 Page 424 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 109 Delivery Terms and Conditions

Boscawen, Bow, Concord, Derry, Franklin, Gilford, Goffstown, Hollis, Hooksett, Hudson, Laconia, Litchfield, Londonderry, Loudon, Manchester, Merrimack, Milford, Nashua, Northfield, Pelham, Pembroke, Sanbornton, Tilton, Windham, and part of Canterbury.

- 4.2 For each Aggregation Pool as set forth by Section 20.6, the Company will designate at least one specific interconnection between a Transporting Pipeline and the Company's distribution facilities, at which point, or such other point as the Company may designate from time to time, the Supplier will make deliveries for the Aggregation Pool. The interconnections that the Company may assign as the Customer's Designated Receipt Point for the Aggregation Pool are as follows:
 - (1) Name Transporting Pipeline: Tennessee Gas Pipeline Names of City Gates/Meter Numbers:

 Nashua/Milford
 #020132

 Manchester
 #020133

 Hooksett
 #020254

 Concord/Laconia
 #020426

 Suncook
 #020451

 Londonderry
 #020632

(2) Name Transporting Pipeline: Portland Natural Gas Transmission System Names of City Gates/Meter Number

Berlin #020260

5 CUSTOMER REQUEST FOR SERVICE FROM COMPANY

- 5.1 Application for Delivery Service, Sales Service, or any other service offered by the Company to a Customer will be received by any duly authorized representative or agent of the Company.
- 5.2 Before any service from the Company may commence, the Customer must request such service. A Customer applying for Delivery Service only must also arrange for Supplier Service with a Supplier pursuant to Section 20. A Customer may act as its own Supplier provided it meets all of the Supplier requirements delineated in Section 20.

6 QUALITY AND CONDITION OF GAS

- Gas delivered to the Company by or for the Customer shall conform, in all respects, to the Gas quality standards of the Transporting Pipeline. All Gas tendered by a Supplier at a Designated Receipt Point shall be of merchantable quality and shall be interchangeable with Gas purchased by the Company from its Suppliers. The Company reserves the right to refuse non-conforming Gas.
- 6.2 In no event shall the Company be obligated to accept and deliver any Gas that does not meet the quality standards of the Transporting Pipeline.
- 6.3 The Company reserves the right to commingle Gas tendered by a Supplier at a Designated Receipt Point with other Gas, including liquefied natural Gas and propane-air vapor.
- Gas tendered by a Supplier at a Designated Receipt Point will be at a pressure sufficient to enter the Company's distribution system without requiring the Company to adjust its normal operating pressures to receive the Gas. The Company has no obligation to receive Gas at a pressure that exceeds the maximum allowable operating pressure of the Company's distribution system at the Designated Receipt Point.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 425 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 110 Delivery Terms and Conditions

7 POSSESSION OF GAS

- 7.1 Gas shall be deemed to be in the control and possession of the Company after such Gas is delivered to the Designated Receipt Point and until the Gas is delivered to the Customer at the Delivery Point. The Company shall not be responsible for the Gas when the Gas is not in the Company's control and possession.
- 7.2 The Company shall not be liable to the Supplier or the Customer for any loss arising from or out of Delivery Service, including loss of Gas in the possession of the Company or for any other cause, except for the negligence of the Company's own employees or agents.

8 COMPANY GAS ALLOWANCE

8.1 The amount of Gas tendered by the Supplier to the Designated Receipt Point will be reduced, upon delivery to the Customer's Delivery Point, by the Company Gas Allowance. The Company Gas Allowance shall be in effect from November 1 through October 31. Such adjustment shall be recalculated prior to the Company's Peak Season cost of Gas filing with the NHPUC.

9 DAILY METERED DELIVERY SERVICE

9.1 Applicability

Section 9 of this tariff shall be applicable in the following conditions:

- 9.1.1 All Customers whose service may be interrupted at any time during the year shall be required to take daily metered Delivery Service.
- 9.1.2 Any Customer, regardless of annual Gas Usage, may elect daily metered Delivery Service.
- 9.1.3 Customers under Rate Schedules G-43, G-46, G-53, G-54, G-57, and G-58 wishing to take Delivery Service are required to take Daily Metered Delivery Service. In addition, the Company may require a Customer to take daily metered Delivery Service if the Company determines that the daily Gas Usage characteristics of the Customer cannot be accurately modeled using the Company's Consumption Algorithm or if the volumes reasonably anticipated by the Company to be used by the Customer are of a size that may materially affect the integrity of the Company's distribution system.

9.2 Delivery Service Provided

This service provides delivery of Customer purchased Gas from the Designated Receipt Point to the Delivery Point on any Gas Day. For Customers taking Delivery Service under Rate Schedules <u>G-43, G-46, G-53, G-54, G-57, and G-58</u> this service provides firm, year-round delivery of Customer purchased Gas from the Designated Receipt Point to the Delivery Point.

- 9.3 Nominations and Scheduling of Service
 - 9.3.1 The Supplier is responsible for nominating and delivering to the Designated Receipt Point(s) every Gas Day an amount of Gas that equals the aggregated Gas Usage of Customers in the Aggregation Pool plus the Company Gas Allowance in accordance with Section 8 of this tariff.
 - 9.3.2 Nominations shall be communicated to the Company by electronic means as determined by the Company or, in the event of failure of such electronic means, by facsimile or other agreeable alternative means.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 426 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 111 Delivery Terms and Conditions

- 9.3.3 Nominations for the first Gas Day of a Month shall be submitted to the Company no later than two (2) hours prior to the deadline for first of the Month Nominations of the Transporting Pipeline or such lesser period as determined by the Company. The Company will make available, from time to time, a schedule of Nomination due dates. Nominations on weekends, holidays, and non-business hours will be accepted by the Company on a basis consistent with that utilized for its own operations.
- 9.3.4 The Supplier may make daily Nominations including, but not limited to, changes to existing Nominations, within a given Month no later than two (2) hours prior to the deadline for daily Nominations of the Transporting Pipeline for the Gas Day on which the Nomination is to be effective, or such lesser period as determined by the Company. Nominations on weekends, holidays, and non-business hours will be accepted by the Company on a basis consistent with that utilized for its own operations.
- 9.3.5 The Supplier may make intra-Gas Day Nominations, including but not limited to changes to existing Nominations, within a given Gas Day no later than two (2) hours prior to the intra-Gas Day Nomination deadline for the Transporting Pipeline on which the Nomination is to be effective, or such lesser period as determined by the Company. Intra-Gas Day Nominations on weekends, holidays, and non-business hours will be accepted by the Company on a basis consistent with that utilized for its own operations.
- 9.3.6 Nominations will be conditionally accepted by the Company pending confirmation by the Transporting Pipeline. The Company will attempt to confirm the nominated volume with the Transporting Pipeline. In the event of a discrepancy between the volume nominated to the Company by the Supplier and the volume nominated by the Supplier to the Transporting Pipeline, the lower volume will be deemed confirmed. The Company will allocate such discrepancy based on a predetermined allocation method set forth in the Supplier Service Agreement. If no predetermined allocation method has been established prior to the event of such discrepancy, the Company will allocate the discrepancy on a pro rata basis.
- 9.3.7 Nominations may be rejected, at the sole reasonable discretion of the Company, if they do not satisfy the conditions for Delivery Service in effect from time to time.
- 9.4 Determination of Receipts
 - 9.4.1 The quantity of Gas deemed received by the Company for the Supplier's Aggregation Pool at the Designated Receipt Point(s) will equal the volume so scheduled by the Transporting Pipeline(s).
 - 9.4.2 The Company Gas Allowance will be assessed against receipts pursuant to Section 8 of this tariff.
- 9.5 Metering and Determination of Deliveries
 - 9.5.1 The Company shall furnish and install, at the Customer's expense, telemetering equipment and any related equipment for the purpose of measuring Gas Usage at each Customer's Delivery Point. Telemetering equipment shall remain the property of the Company at all times. The Company shall require each Customer to install and maintain, at the Customer's expense, reliable telephone lines and electrical connections that meet the Company's operating requirements. The Company may require the Customer to furnish a dedicated telephone line. If the Customer fails to maintain such telephone lines and electrical connections for fourteen (14) consecutive days after notification by the Company, the Company may discontinue service to the Customer.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 427 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 112 Delivery Terms and Conditions

- 9.5.2 Should a Customer or a Supplier request that additional telemetering equipment or a communication device be attached to the existing telemetering equipment in addition to that provided pursuant to Section 9.5.1, the Company shall install, test, and maintain the requested telemetering equipment or communication device; provided that such telemetering equipment or communication device does not interfere with the operation of the equipment required for the Company's purposes and otherwise meet the Company's requirements. The Customer or Supplier shall provide such telemetering equipment or communication device, unless the Company elects to do so. The Customer or Supplier shall bear the cost of providing and installing the telemetering equipment, communication device, or any other related equipment, and shall have electronic access to the Customer's Gas Usage information. Upon installation, the telemetering equipment or communication device shall become the property of the Company and will be maintained by the Company. The Company shall bill the Customer or Supplier after installation.
- 9.5.3 The Company shall complete installation of telemetering equipment and communication devices, if reasonably possible, within sixty (60) days of receiving a written request from the Customer or Supplier provided that the Customer completes the installation of any required telephone or electrical connections within ten (10) days of such request.
- 9.5.4 The Company may, at its sole discretion, bill the Customer on a calendar month or cycle month basis.

9.6 Balancing

9.6.1 The Supplier must maintain a balance between daily receipts and daily Gas Usage within the following tolerances:

Off-Peak Season: The difference between the Supplier's aggregate actual receipts on the

Transporting Pipeline to each Gas Service Area and the aggregated Gas Usage of Customers in the Aggregation Pool shall be within 15% of said receipts. The Supplier shall be charged 0.1 times the Daily Index

for all differences not within the 15% tolerance.

<u>Peak Season</u>: The difference between the Supplier's aggregate actual receipts on the

Transporting Pipeline to each Gas Service Area and the aggregated Gas Usage of Customers in the Aggregation Pool shall be within 10% of said receipts. The Supplier shall be charged 0.5 times the Daily Index

for all differences not within the 10% tolerance.

<u>Critical Day(s)</u>: The Company will determine if the Critical Day will be aggravated by

an under-delivery or an over-delivery, and so notify the Supplier when

a Critical Day is declared pursuant to Section 16.

Critical Day That Will Be Aggravated by Under-delivery.

<u>Supplier who under-delivers</u>. A Supplier who under-delivers on a Critical Day that will be aggravated by under-delivery shall be charged 5 times the Daily Index for the aggregated Gas Usage of Customers in the Aggregation Pool that exceeds 102% of the Supplier's aggregate actual receipts on the Transporting Pipeline to each Gas Service Area.

<u>Supplier who over-delivers</u>. A Supplier who over-delivers on a Critical Day that will be aggravated by under-delivery shall be charged 0.1 times the Daily Index to the

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 428 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 113 Delivery Terms and Conditions

extent that the difference between the Supplier's aggregate actual receipts on the Transporting Pipeline to each Gas Service Area and the aggregated Gas Usage of Customers in the Aggregation Pool exceeds 20% of said receipts [(Receipts - Usage) > (20% x Receipts)].

Critical Day That Will Be Aggravated by Over-delivery.

<u>Supplier who under-delivers</u>. A Supplier who under-delivers on a Critical Day that will be aggravated by over-delivery shall be charged 0.1 times the Daily Index to the extent that the difference between the Supplier's aggregated Gas Usage of Customers in the Aggregation Pool exceeds 120% of the Supplier's aggregate actual receipts on the Transporting Pipeline to each Gas Service Area.

<u>Supplier who over-delivers</u>. A Supplier who over-delivers on a Critical Day that will be aggravated by over-delivery shall be charged 5 times the Daily Index to the extent that the difference between the Supplier's actual receipts on the Transporting Pipeline to each Gas Service Area and the Supplier's aggregated Gas Usage of Customers in the Aggregation Pool exceeds 2% of said receipts [(Receipts - Usage > (2% x Receipts)].

Point Specific Balancing: In the event that the Transporting Pipeline requires its customers to balance on a point-specific basis, the Supplier must balance pursuant to this Section at each Designated Receipt Point.

- 9.6.2 If the Supplier has an accumulated imbalance within a Month, the Supplier may nominate to reconcile such imbalance, subject to the Company's approval, which approval shall not be unreasonably withheld.
- 9.6.3 In addition to the charges set forth in Section 9.6.1, the Company shall flow through to the Supplier any pipeline imbalance penalty charges attributable to the Supplier.
- 9.6.4 If, as a result of the Company interrupting or curtailing service pursuant to Section 17 of this tariff, the Supplier incurs a daily imbalance penalty due to over delivery, the Company will waive such penalty for the First Day of the interruption or curtailment period. If the Company has issued notice of an interruption or curtailment in service and the Supplier is unable to change its Nomination, or if the Supplier's Gas has been delivered to the Designated Receipt Point, then the Company will credit such Gas against the Supplier's imbalance.
- 9.6.5 The Supplier will maintain a balance between receipts at the Designated Receipt Point(s) and the aggregated Gas Usage of Customers in each Aggregation Pool. If the Transporting Pipeline posts notice on its electronic bulletin board that its customers will be required to adhere to a maximum hourly flow rate, the Supplier will be deemed to have notice that maximum hourly flows will be in effect on the Company's distribution facilities as of the same time and for the same period as maximum hourly flows are in effect on the Transporting Pipeline. The Supplier's maximum hourly flow will be established based on an allocation of even hourly flows of daily receipts of Gas scheduled in the relevant period in accordance with the applicable transportation tariff of the Transporting Pipeline. All Gas Usage in excess of the Supplier's maximum hourly flow rate shall be subject to an additional charge of 5 times the Daily Index for each Dekatherm in excess of the Supplier's maximum hourly flow. The Company will notify the Supplier of the Supplier's maximum hourly flow.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 429 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 114 Delivery Terms and Conditions

9.6.6 If, during any fifteen (15) consecutive Gas Days, the Supplier delivers an amount less than 70% of the sum of the aggregated Gas Usage of Customers in the Aggregation Pool in said Gas Days, the Company may declare the Supplier ineligible to nominate Gas for the following thirty (30) Gas Days. The Supplier shall have the opportunity to cure the imbalance with the demonstration of verifiable imbalance trades or otherwise within twenty-four (24) hours of notification by the Company. If the Supplier is declared ineligible to nominate Gas for such 30 Gas Days, the Supplier may be reinstated at the end of the 30 Gas Days, provided it posts security equal to the product of: (1) the maximum aggregate daily Gas Usage of Customers in the Aggregation Pool expressed in MMBtu and (2) \$300. If, within twelve (12) months of the first offense, such Supplier is declared ineligible to nominate Gas pursuant to this Section, the Supplier will be disqualified from service under this tariff for one (1) full year from the time of the second disqualification. If the Supplier defaults on its obligations under this tariff, the Company shall have the right to use such security to satisfy the Supplier's obligations. Such security may be used by the Company to secure Gas, transportation, and storage, and to cover other related costs incurred as a result of the Supplier's default. The security may also be used to satisfy any outstanding claims that the Company may have against the Supplier including imbalance charges, cash-out charges, pipeline penalty charges, and other charges.

9.7 Cash Out

For each Aggregation Pool, the Supplier must maintain total Monthly receipts within a reasonable tolerance of total Monthly Gas Usage. Any differences between total Monthly receipts for an Aggregation Pool and the aggregated Gas Usage of Customers in the Aggregation Pool, expressed as a percentage of total Monthly receipts, will be cashed out according to the following schedule:

Imbalance Tier	Over-deliveries	Under-deliveries
0% ≤ 5%	The average of the Daily Indices for the relevant Month.	The highest average of seven consecutive Daily Indices for the relevant Month
$> 5\% \le 10\%$	0.85 times the above stated rate.	1.15 times the above stated rate.
> 10% \le 15%	0.60 times the above stated rate.	1.4 times the above stated rate.
> 15%	0.25 times the above stated rate.	1.75 times the above stated rate.

For purposes of determining the tier at which an imbalance will be cashed out, the price will apply only to volumes within a tier. For example, if there is a 7% under-delivery on a Transporting Pipeline, volumes that make up the first 5% of the imbalance are priced at the highest average of the seven (7) consecutive Daily Indices. Volumes making up the remaining 2% of the imbalance are priced at 1.15 times the average of the seven (7) consecutive Daily Indices.

10 NON-DAILY METERED DELIVERY SERVICE

10.1 Applicability

Section 10 of this tariff applies to Customers taking Delivery Service under Rate Schedules G-41, G-42, G-51, G-52 and their Suppliers.

10.2 Delivery Service Provided

This service provides firm, year-round delivery of Customer purchased Gas from the Designated Receipt Point to the Delivery Point on any Gas Day for Customers, without the requirement of

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Docket No. DG 21-130 Exhibit 26 Page 430 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 115 Delivery Terms and Conditions

recording Gas Usage at the Delivery Point on a daily basis. Daily Nominations are calculated by the Company on the basis of a Consumption Algorithm and the Supplier is obligated to deliver to the Designated Receipt Point(s) such quantities.

- 10.3 Nominations and Scheduling of Service
 - 10.3.1 The Supplier is obligated to nominate and deliver the Adjusted Target Volume ("ATV"), as determined in Section 10.3.2, to the Designated Receipt Points on every Gas Day for each Aggregation Pool.
 - 10.3.2 The Company shall determine the ATV for each Aggregation Pool of Customers taking non-daily metered Delivery Service for each Gas Day using a Consumption Algorithm. The ATV shall include the Company Gas Allowance. On each Business Day, the Company will communicate, electronically, by facsimile, or by other agreeable alternative means, the forecasted ATV to the Supplier for at least the subsequent four (4) Gas Days. The ATV in effect for any Gas Day shall be the most recent ATV for that Gas Day communicated to the Supplier by the Company. The ATV for a given Gas Day shall not be effective unless it has been communicated to the Supplier at least two (2) hours prior to the Company's Supplier Nomination deadline for that Gas Day, which shall be at least two (2) hours prior to the deadline for nominations on the Transporting Pipeline, or such lesser period as determined by the Company.
 - 10.3.3 Nominations will be communicated to the Company electronically, by facsimile, or other agreeable alternative means.
 - 10.3.4 Nominations for the first Day of a Month shall be submitted to the Company no later than two (2) hours prior to the deadline for first of the Month nominations of the Delivering Pipeline or such lesser period as determined by the Company. The Company will make available, from time to time, a schedule of nomination due dates. Nominations on weekends, holidays, and non-business hours will be accepted by the Company on a basis consistent with that utilized for its own operations.
 - 10.3.5 The Supplier shall provide an intra-Month nomination no later than two (2) hours prior to the deadline of the Delivering Pipeline for the next Gas Day, or such lesser period as determined by the Company. Nominations on weekends, holidays, and non-business hours will be accepted by the Company on a basis consistent with that utilized for its own operations.
 - 10.3.6 Nominations will be conditionally accepted by the Company pending confirmation by the Transporting Pipeline. The Company will attempt to confirm the nominated volume with the Transporting Pipeline. In the event of a discrepancy between the volume nominated to the Company by the Supplier and the volume nominated by the Supplier to the Transporting Pipeline, the lower volume will be deemed confirmed. The Company will allocate such discrepancy based on a predetermined allocation method set forth in the Supplier Service Agreement. If no predetermined allocation method has been established prior to the event of such discrepancy, the Company will allocate the discrepancy on a pro rata basis. The Company will not confirm any volume nominated by the Supplier in excess of the ATV.
 - 10.3.7 In the event that the Supplier is unable to deliver a confirmed ATV Nomination, the Supplier may make intra-Gas Day Nominations relating to changes to existing Nominations within a given Gas Day no later than two (2) hours prior to the intra-Gas Day Nomination deadline for the Transporting Pipeline on which the Nomination is to be effective, or such lesser period as determined by the Company; provided, however, that the

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 431 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 116 Delivery Terms and Conditions

- 10.3.8 Nomination must be in conformance with the requirements of and must be permitted by the Transporting Pipeline. Intra-Gas Day Nominations on weekends, holidays, and non-business hours will be accepted by the Company on a basis consistent with that utilized by the Company for its own operations. The Company shall not adjust the ATV applied for the Gas Day.
- 10.3.9 Nominations may be rejected if they do not satisfy the conditions for Delivery Service in effect from time to time.
- 10.3.10 All quantities of Gas over-delivered or under-delivered to the Company's system in violation of an Operational Flow Order ("OFO") declared by the Company pursuant to Section 16 will be subject to the Critical Day provisions of Section 10.6.1 of this tariff, and the delivered quantity specified in the OFO will replace the ATV.
- 10.4 Determination of Receipts
 - 10.4.1 The quantity of Gas deemed received by the Company for the Supplier's Aggregation Pool at the Designated Receipt Point(s) will equal the volume so scheduled by the Transporting Pipeline(s).
 - 10.4.2 The Company Gas Allowance will be assessed against receipts pursuant to Section 8 of this tariff.
- 10.5 Metering and the Determination of Deliveries

The Company shall record the Customer's Gas Usage at the Delivery Point by making actual meter reads on a monthly [or bi-monthly] basis. In the event that the Customer's Gas Usage is metered on a bi-monthly basis, the Company shall make available to the Supplier estimates of the Customer's Gas Usage for each of the two billing months.

10.6 Balancing

10.6.1 Any difference between the Supplier's ATV for an Aggregation Pool and the receipts on the Transporting Pipeline to the appropriate Designated Receipt Point(s) will be cashed out by the Company according to the following:

Off-Peak Season: For receipts less than the ATV, the Supplier shall be charged 1.1 times

the Daily Index for the difference. For receipts greater than the ATV, the Supplier shall be charged 0.8 times the Daily Index for the

difference.

Peak Season: For receipts less than the ATV but greater than or equal to 95% of the

ATV, the Supplier shall be charged 1.1 times the Daily Index for the difference. For receipts less than 95% of the ATV, the Supplier shall be charged 1.1 times the Daily Index for the first 5% difference, and the Supplier shall be charged two (2) times the Daily Index for the remaining difference. For receipts greater than the ATV, the Supplier

shall be charged 0.8 times the Daily Index for the difference.

Critical Day(s) The Company will determine if the Critical Day will be aggravated by

an under-delivery or an over-delivery, and so notify the Supplier when

a Critical Day is declared pursuant to Section 16.

Critical Day That Will Be Aggravated by Under-delivery.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 432 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 117 Delivery Terms and Conditions

<u>Supplier who under-delivers</u>. A Supplier who under-delivers on a Critical Day that will be aggravated by under-delivery shall be charged five (5) times the Daily Index for the difference between the ATV and actual receipts.

<u>Supplier who over-delivers</u>. A Supplier who over-delivers on a Critical Day that will be aggravated by under-delivery shall be charged the following amounts for all receipts in excess of the ATV:

- (a) up to 25% in excess of the ATV, the Supplier shall be charged the Daily Index for the difference.
- (b) for receipts in excess of 25% above the ATV, the Supplier shall be charged 0.8 times the Daily Index for the difference.

Critical Day That Will Be Aggravated By Over-delivery.

<u>Supplier who over-delivers</u>. A Supplier who over-delivers on a Critical Day that will be aggravated by over-delivery shall be charged 0.4 times the Daily Index for receipts greater than the ATV.

<u>Supplier who under-delivers</u>. A Supplier who under-delivers on a Critical Day that will be aggravated by over-delivery shall be charged the following amounts--for receipts less than the ATV but greater than or equal to 75% of the ATV, the Supplier shall be charged the Daily Index for the first 25% difference, and the Supplier shall be charged 1.1 times the Daily Index for the remaining difference.

- 10.6.2 In addition to the charges set forth in Section 10.6.1, the Company shall use a daily balancing charge calculation to account for balancing costs it incurs in serving each Aggregation Pool due to differences in forecast versus actual Heating Degree Days. The daily balancing charge shall be based on the sum of the absolute values of the daily differences between the Aggregation Pool's ATV and the recalculated ATV value described in Section 10.7.1 below. Such charge shall be billed to the Supplier monthly and shall reflect the cost of resources used by the Company to balance such differences for each Gas Day of the Month. The Company shall calculate such charge annually in its Winter Season Cost of Gas filing according to a formula as set forth in Attachment B.
- 10.6.3 In addition to the charges set forth in Section 10.6.1, the Company shall use a daily balancing charge calculation to account for balancing costs it incurs in serving each Aggregation Pool due to differences in forecast versus actual Heating Degree Days. The daily balancing charge shall be based on the sum of the absolute values of the daily differences between the Aggregation Pool's ATV and the recalculated ATV value described in Section 10.7.1 below. Such charge shall be billed to the Supplier monthly and shall reflect the cost of resources used by the Company to balance such differences for each Gas Day of the Month. The Company shall calculate such charge annually in its Winter Season Cost of Gas filing according to a formula as set forth in Attachment B.

In the event that the Transporting Pipeline requires its customers to balance on a point-specific basis, the Supplier must balance pursuant to this Section at each Designated Receipt Point.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 433 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 118 Delivery Terms and Conditions

10.6.4 In addition to the charges set forth in Sections 10.6.1 and 10.6.2, the Company shall flow through to the Supplier any pipeline imbalance penalty charges attributable to the Supplier.

10.7 Cash Out

- 10.7.1 The Company shall use a daily cash out calculation to account for imbalances due to differences in forecast versus actual Heating Degree Days. Using a Consumption Algorithm, the Company will recalculate the ATV for each Aggregation Pool for each Gas Day of the Month, substituting actual Heating Degree Days for forecast Heating Degree Days. Daily recalculations shall be compared to the Aggregation Pool's daily ATV, and the difference shall be cashed out at 100% of the Daily Index.
- 10.7.2 During the billing months of both June and December, the Company shall use a six-month cash-out calculation to account for differences in forecast usage versus billed Gas Usage. The Company may cash-out differences in forecast usage versus billed usage at intervals that are less than six months as provided by the Supplier Service Agreement.
 - (1) In the billing month of June, using the recalculated ATV values described in Section 10.7.1, the Company will compare the sum of the recalculated ATV values for each Aggregation Pool for the six-month period of November 1 through April 30 to the sum of billed usage volumes used by each Aggregation Pool for that same period. The differences shall be cashed out at 100% of the average of the Daily Index weighted by actual Heating Degree Days over the same period. The Winter period cash-out shall be calculated and provided to Suppliers within 60 days of the month ending April 30.
 - (2) In the billing month of December, using the recalculated ATV values described in Section 10.7.1, the Company will compare the sum of the recalculated ATV values for each Aggregation Pool for the six-month period of May 1 through October 31 to the sum of the billed usage volumes used by each Aggregation Pool for that same period. The differences shall be cashed out at 100% of the average of the Daily Index over the same period. The Off-Peak period cash-out shall be calculated and provided to Suppliers within 60 days of the month ending October 31.
- 10.7.3 The Company shall allow Suppliers to trade seasonal differences. Prior to the seasonal cash-out, the Company shall make available a list of Suppliers. Aggregation Pools affected by the transaction must be located within the same Gas Service Area as defined in Section 4, unless waived by the Company. All trades must be communicated to the Company within three (3) Business Days following receipt of the list.
 - 10.7.4 If, during any fifteen (15) consecutive Gas Days, the Supplier delivers an amount less than 70% of the sum of the ATVs of the Aggregation Pool in said Gas Days, the Company may declare the Supplier ineligible to nominate Gas for the following thirty (30) Gas Days. The Supplier shall have the opportunity to cure the imbalance with the demonstration of verifiable imbalance trades or otherwise within twenty-four (24) hours of notification by the Company. If the Supplier is declared ineligible to nominate Gas for such 30 Gas Days, the Supplier may be reinstated at the end of the 30 Gas Days, provided it posts security equal to the product of: (1) the Supplier's estimated maximum aggregate daily Gas Usage of Customers in the Aggregation Pool expressed in MMBtu and (2) \$300. If, within twelve (12) months of the first offense, such Supplier is declared ineligible to nominate Gas pursuant to this Section, the Supplier will be disqualified from service under this tariff for one (1) full year from the time of

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 434 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 118 Delivery Terms and Conditions

the second disqualification. If the Supplier defaults on its obligations under this tariff, the Company shall have the right to use such security to

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

Docket No. DG 21-130 Exhibit 26 Page 435 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 119 Delivery Terms and Conditions

10.7.5 satisfy the Supplier's obligations. Such security may be used by the Company to secure Gas, transportation, storage, and to cover other related costs incurred as a result of the Supplier's default. The security may also be used to satisfy any outstanding claims that the Company may have against the Supplier including imbalance charges, cash-out charges, pipeline penalty charges, and other charges.

11 CAPACITY ASSIGNMENT

11.1 Applicability

Section 11 of this tariff applies to all Suppliers that have enrolled one or more Customers into one or more Aggregation Pools and shall include Customers acting as their own Supplier. The Company shall assign and the Supplier shall accept each Customer's pro-rata share of Capacity, if any, as established in accordance with this Section.

11.2 Identification of Capacity for Assignment

- 11.2.1 On or before September 15 of each year, the Company shall communicate, by electronic means as determined by the Company or, in the event of failure of such electronic means, by facsimile or other agreeable alternative means, the Capacity to be made available for assignment to Suppliers on each of twelve Assignment Dates beginning in October.
- 11.2.2 The Company shall identify, by Gas Service Area, the specific contracts and resources for assignment to Suppliers based on the Company's Capacity and resource plans. Such identified contracts and resources shall be used to determine the pro-rata shares of Capacity assignable to a Supplier on behalf of the Customers enrolled in its Aggregation Pool.
- 11.2.3 Capacity assigned by the Company may include Company-Managed Supplies that effectuate, at maximum tariff rates, the assignment of certain Capacity contracts including Canadian, Federal Energy Regulatory Commission, 15 U.S.C. § 717(c) or Section 7(c) [Part 157 of the FERC regulations (18 C.F.R. part 157)] and other contracts that are not assignable to third-parties due to governing tariffs.

11.3 Determination of Pro-Rata Shares of Capacity

- 11.3.2 The Company shall establish a Total Capacity Quantity ("TCQ") for each Customer taking Delivery Service. The TCQ represents the total amount of Capacity assignable to a Supplier on behalf of a Customer.
- 11.3.3 For a Customer receiving Sales Service on or after March 14, 2000, the TCQ shall be the Customer's estimated Gas Usage on the Peak Day as determined by the Company each October prior to the Customer's enrollment into Supplier Service. The Company shall derive such estimate using a Daily Baseload and a Heating Factor based upon the Customer's historic Gas Usage during the Reference Period, or the best estimates available to the Company should actual Gas Usage information be partially or wholly unavailable.
- 11.3.4 For a Customer that was either receiving Supplier Service (or the equivalent form of service at the time) on March 14, 2000, or had an executed contract for firm transportation service (i.e., the equivalent of Delivery Service) on file with the Company on or before March 14, 2000, the TCQ shall be zero.
- 11.3.5 A Customer that was either receiving Supplier Service (or the equivalent form of service at the time) on March 14, 2000, or had a written request on file with the Company on or before March 14, 2000 may elect for its Supplier to accept assignment of its pro-rata share of Capacity as determined by the Company in accordance with Section 11.2 and, subject

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 436 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 120 Delivery Terms and Conditions

to availability, as determined by the Company in its sole reasonable discretion. In order to make such election, the Customer must have submitted to the Company, on or before ten (10) days prior to the first Assignment Date prior to the original effective date of this tariff, a completed application for Capacity that is signed by both the Customer and Supplier. All assignments of Capacity made on behalf of such electing Customer shall be executed in accordance with Sections 11 and 14 of this tariff as if the Customer had been receiving Sales Service on or after March 14, 2000

- 11.3.5 For a new Customer taking Supplier Service as its initial service after March 14, 2000, the TCQ shall be zero except in cases where the Customer is a new Customer of record at a meter location where a former Customer of record received firm service from the Company any time during the preceding twenty-four (24) months, in which case the TCQ established by the Company for the former Customer shall become the TCQ for the new Customer. The Company may reduce said TCQ value for the new Customer, if, in its sole reasonable discretion, the Company determines that the old Customer's TCQ exceeds the new Customer's estimated future consumption on the Peak Day. In the event that Sales Service is provided at a new meter location for Gas Usage associated with new construction, the TCQ shall be zero, provided that the Customer initiates Supplier Service upon the completion of said new construction in accordance with Section 20.5 of this tariff
- 11.3.6 Once the Company establishes a TCQ for a Customer pursuant to this Section 11.3, it shall remain in effect for the purpose of determining the Customer's pro-rata shares of Capacity until such time that the Customer returns to Sales Service. The Company shall establish a new TCQ value for the Customer pursuant to Section 11.3.2 if the Customer again elects to take Supplier Service after returning to Sales Service, unless otherwise established herein..
- 11.3.7 The Company shall determine the pro-rata shares of Pipeline Capacity, Storage Withdrawal Capacity, and Peaking Capacity assignable to a Supplier on behalf of a Customer as the product of the Customer's TCQ times the applicable Capacity Allocators. The Capacity Allocators for each class of Customers billed under the Company's Rate Schedule shall be set forth annually in Attachment C to this tariff.
- 11.3.8 The Company shall determine the pro-rata share of Seasonal Storage Capacity assignable to a Supplier on behalf of a Customer consistent with the tariffs governing the associated Storage Withdrawal Capacity.
- 11.3.9 The Company shall determine the pro-rata shares of Peaking Supply assignable to a Supplier in accordance with Section 14 of this tariff.

11.4 Capacity Assignments

- 11.4.1 On each Assignment Date, the Company will assign to the Supplier the pro-rata shares of Capacity on behalf of each Customer as determined by the Company in accordance with Sections 11.2, 11.3 and 11.7.
- 11.4.2 The total amount of Pipeline Capacity, Storage Withdrawal Capacity, and Peaking Capacity assigned to the Supplier on behalf of the Customers in an Aggregation Pool shall be at least equal to the cumulative sum of the pro-rata shares of Pipeline Capacity, Storage Withdrawal Capacity, and Peaking Capacity for all Customers enrolled in said Aggregation Pool as of Five (5) Business Days prior to the Assignment Date.
- 11.4.3 Storage Withdrawal Capacity shall be subject to Operational Flow Orders that are issued by the Company pursuant to Section 16 of this tariff, in the event that the Company requires the Supplier to deliver or to store quantities of Gas for the purposes of managing system

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 437 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 121 Delivery Terms and Conditions

imbalances and maintaining Delivery Service. Whenever the Company assigns incremental Storage Withdrawal Capacity to the Supplier, the Company shall also assign to that Supplier additional Seasonal Storage Capacity pursuant to Section 11.8.

- 11.4.4 The Peaking Capacity assigned to the Supplier shall establish the Maximum Daily Peaking Quantity ("MDPQ") for the Aggregation Pool in the Supplier's Service Agreement. In the event that the Company increases a Supplier's MDPQ, the Company shall also assign to that Supplier additional Peaking Supply pursuant to Section 14.
- 11.4.5 The Company shall execute Capacity assignments in increments of 200 MMBtus. The Supplier shall accept an initial increment of Capacity on the first Assignment Date when the sum of the pro-rata shares of Capacity assigned to the Supplier pursuant to Section 11.4.1 exceeds 150 MMBtus. The Supplier shall accept additional increments of Capacity on the following Assignment Dates commensurate with any cumulative increase in the sum of pro-rata shares of Capacity assigned to the Supplier, as rounded to the nearest 200 MMBtus. Each increment of Capacity accepted by the Supplier shall comprise Pipeline Capacity, Storage Withdrawal Capacity, and Peaking Capacity in proportion to the cumulative increase of the pro-rata shares of assigned Capacity as established in accordance with Section 11.4.1. Section 11.4.2 shall not apply to a Customer that is acting as its own Supplier.
- 11.4.6 If a Customer is acting as its own Supplier, the Company shall assign Capacity to the Customer in an amount equal to the Customer's TCQ, as established pursuant to Section 11.3.

11.5 Release of Contracts

- 11.5.1 With the exception of Company-Managed Supplies and Peaking Capacity, Capacity contracts shall be released by the Company to the Supplier, at the maximum tariff rate or lesser rate paid by the Company and including all surcharges, through pre-arranged Capacity releases, pursuant to applicable laws and regulations and the terms of the governing tariffs.
- 11.5.2 Capacity contracts released to a Supplier on an Assignment Date shall be released for a term beginning on the first Gas Day of the Month following the Assignment Date through the expiration date of the respective capacity contract being assigned. and ending on October 31. For example, contracts assigned to a Supplier on April 25 of a given year shall be released for a term beginning on May 1 of that year and ending on October 31 of that year.
- 11.5.3 The Company reserves the right to adjust releases of Storage Withdrawal Capacity in the event that fifty percent (50%) or more of the total Storage Withdrawal Capacity serving a Gas Service Area has been assigned to Suppliers. Such adjustments may include, but are not limited to, the reassignment of certain Storage Withdrawal Capacity as Company-Managed Supplies in order for the Company to maintain operational control over Capacity resources associated with system balancing, and/or the retention of specific Capacity resources associated with system balancing and the implementation of a balancing charge to offset the associated costs.

11.6 Annual Reassignment of Capacity

11.6.1 On each Annual Reassignment Date, the Company shall adjust the Capacity assignments previously made to a Supplier to conform with the Company's resource and requirements plans. Such previously assigned Capacity shall be replaced by the assignment to the Supplier of the pro-rata shares of the same or similarly situated Capacity on behalf of the

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 438 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 122 Delivery Terms and Conditions

Customers enrolled in the Supplier's Aggregation Pools (as of the first Gas Day of the Month following the Annual Reassignment Date).

- 11.6.2 If the reassignment of Storage Withdrawal Capacity requires adjustments to the Seasonal Storage Capacity previously assigned to a Supplier, the Company shall reassign Seasonal Storage Capacity to such Supplier, and the Company and the Supplier shall address any associated increments and decrements to inventories in place pursuant to Section 11.8 of this tariff.
- 11.6.3 If the reassignment of Peaking Capacity requires adjustments to the MDPQ for the Supplier's Aggregation Pool, the Company shall reassign Peaking Supply to such Supplier, and the Company and the Supplier shall address any associated increments and decrements to supplies pursuant to Section 14 of this tariff.

11.7 Recall of Capacity

- 11.7.1 If the pro-rata shares of Capacity assignable to a Supplier decline because one or more of the Supplier's Customers has returned to Sales Service, the Company shall have the right, but not the obligation, to recall from the Supplier the pro-rata shares of Capacity previously assigned to the Supplier on behalf of such Customers. The decision on whether to exercise its Capacity-recall rights shall be made by the Company in its sole reasonable discretion. If the Company elects to recall Capacity from a Supplier pursuant to this Section, such recall shall be made on the Assignment Date following the effective date of the Customer's return to Sales Service. Notwithstanding the foregoing, in the following circumstances the Company shall be required to recall Capacity associated with Customers returning to Sales Service:
 - (a) The Supplier returning the Customers to Sales Service certifies that it is ceasing all business operations in New Hampshire;
 - (b) The Supplier returning the Customers to Sales Service certifies that it will no longer offer service to a particular market sector (e.g., small commercial and industrial Customers) and, therefore, once such Customers are returned to Sales Service, the Supplier is not eligible to re-enroll Customers of that type; or
 - (c) The Supplier demonstrates that it has provided Supplier Service to the Customer for a 12-month period, and for a period of no less than any 12-month increment, prior to the Customer's return to Sales Service.
- 11.7.2 If the Company elects to recall Storage Withdrawal Capacity from the Supplier pursuant to this Section, the Company shall reduce the Seasonal Storage Capacity associated with the affected Aggregation Pool in accordance with Section 11.8 of this tariff. If the Company elects to reduce the MDPQ in the Supplier Service Agreement, the Company shall reduce the Peaking Supply associated with the affected Aggregation Pool in accordance with Section 14 of this tariff.
- 11.7.3 In the event that a Customer in a Supplier's Aggregation Pool switches to another Supplier, the Company shall recall from the former Supplier said Customer's pro-rata shares of Capacity for reassignment to the new Supplier pursuant to Section 11.4. There shall be no change in the Customer's TCQ used to determine the Customer's pro-rata shares of Capacity for reassignment to the new Supplier. The recall of such Capacity from the Customer's former Supplier and the assignment of Capacity to the new Supplier shall be made on the Assignment Date following the effective date of the Customer's switch in Suppliers.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 439 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 123
Delivery Terms and Conditions

If the Company recalls Storage Withdrawal Capacity from the Customer's former Supplier, the Company shall reduce the Seasonal Storage Capacity associated with the affected Aggregation Pool in accordance with Section 11.8 of this tariff. If the Company reduces the MDPQ in the Customer's former Supplier's Service Agreement, the Company shall also reduce the Peaking Supply associated with the affected Aggregation Pool in accordance with Section 14 of this tariff.

11.7.4 The recall of Capacity by the Company shall entail the recall of released contracts pursuant to governing tariffs and/or the reduction in assigned quantities set forth in the Supplier Service Agreement. The recall of Capacity shall be executed in decrements of 200 MMBtus, commensurate with the cumulative reduction in the pro-rata shares of Capacity assigned to the Supplier, rounded to the nearest 200 MMBtus. Each decrement of Capacity assigned to the Supplier shall comprise Pipeline Capacity, Storage Withdrawal Capacity, and Peaking Capacity in proportion to the cumulative decrease in the pro-rata shares of Capacity recalled from the Supplier.

In the event that a Supplier is declared ineligible to nominate Gas for thirty (30) Gas Days pursuant to Sections 9.6.6 or 10.7.4 of this tariff, the Company shall have the right to recall any or all Capacity assigned to said Supplier. If the Supplier is reinstated at the end of such 30 Gas Days, the Company shall reassign Capacity to the Supplier on the next Assignment Date pursuant to Sections 11.4 and 11.5. There shall be no change in the TCQ values used to determine the Supplier's Customers' pro-rata shares of Capacity for reassignment.

- 11.7.5 In the event that a Supplier is disqualified from service for a one (1) full year pursuant to Sections 9.6.6 or 10.7.4 of this tariff, the Company shall have the right to recall any or all Capacity assigned to said Supplier. If the Supplier is reinstated at the end of such period, the Company shall reassign Capacity to the Supplier on the next Assignment Date pursuant to Sections 11.4 and 11.5. There shall be no change in the TCQ values used to determine the Supplier's Customers' pro rata shares of Capacity reassignments.
- 11.7.6 In the event that the Supplier fails to meet the applicable registration and licensing requirements established by law or regulation, fails to satisfy the requirements and practices as set forth in Section 20.3 of this tariff, fails to be and remain an approved shipper on the upstream pipelines and underground storage facilities on which the Company will assign capacity, fails to make timely payment under the assigned contracts, or fails to comply with or perform any of the obligations on its part established in this tariff or in the Supplier Service Agreement, the Company shall have the right to recall permanently any or all Capacity assigned to said Supplier. This section shall also apply to a Customer acting as its own Supplier.
- 11.7.7 The Supplier shall forfeit its rights to Capacity recalled by the Company pursuant to this Section. Such forfeiture shall be effected in accordance with applicable laws and regulations and the governing tariffs. In the event of Capacity forfeiture pursuant to this Section, the Supplier shall be responsible to compensate the Company for any payments due under the contracts prior to forfeiture, as well as any interest due thereon. The Company will not exercise discretion in the application of the forfeiture provisions of this Section. This section shall also apply to a Customer acting as its own Supplier.

11.8 Seasonal Storage Capacity

11.8.1 On each Assignment Date, the Company shall release Seasonal Storage Capacity to a Supplier that accepts the assignment of Storage Withdrawal Capacity pursuant to Section 11.4. The Company shall assign such Seasonal Storage Capacity consistent with the tariffs governing the release of the associated Storage Withdrawal Capacity.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u> Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 440 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 124 Delivery Terms and Conditions

- 11.8.2 If the Company assigns Seasonal Storage Capacity to a Supplier pursuant to Section 11.8.1 above, the Company shall transfer in-place Gas inventories to the Supplier. The quantity of inventories to be transferred from the Company to the Supplier shall be determined by multiplying the incremental Seasonal Storage Capacity assigned to the Supplier on the Assignment Date times the applicable storage inventory percentage described in Section 11.8.5. The Supplier shall be charged the Company's weighted average cost of inventories in off-system storage facilities for each Dekatherm transferred from the Company to the Supplier. The Company shall communicate, by electronic means as determined by the Company or, in the event of failure of such electronic means, by facsimile or other agreeable alternative means, the Company's weighted average cost of inventories, by Gas Service Area, at least two Business Days prior to each Assignment Date.
- 11.8.3 In the event that the Company recalls Storage Withdrawal Capacity from the Supplier pursuant to Section 11.7, the Company shall also recall Seasonal Storage Capacity from the Supplier. The Company shall determine the total Seasonal Storage Capacity to be recalled from the Supplier in accordance with the tariffs governing the Storage Withdrawal Capacity returned to the Company.
- 11.8.4 If the Company recalls Seasonal Storage Capacity from a Supplier pursuant to Section 11.8.3, the Supplier shall transfer in-place Gas inventories to the Company. The quantity of inventories to be transferred from the Supplier to the Company shall be determined by multiplying the decremental Seasonal Storage Capacity times the applicable storage inventory percentage described in Section 11.8.5. The Supplier shall be reimbursed at the Company's weighted average cost of inventories in off-system storage facilities as of the Assignment Date, for each Dekatherm transferred from the Supplier to the Company. The Company shall communicate, by electronic means as determined by the Company or, in the event of failure of such electronic means, by facsimile or other agreeable alternative means, the Company's weighted average cost of inventories, by Gas Service Area, at least two (2) Business Days prior to each Assignment Date.
- 11.8.5 Seasonal storage inventory percentages shall represent the amount of Seasonal Storage Capacity in each assigned storage resource that is assumed to be filled with inventories as of the first Gas Day of the month following the Assignment Date. Each September, the Company shall communicate, by electronic means as determined by the Company or, in the event of failure of such electronic means, by facsimile or other agreeable alternative means, the storage inventory percentages for each resource that shall be applied to incremental or decremental Seasonal Storage Capacity assignments executed on each of the twelve (12) Assignment Dates beginning in October.

11.9 Company-Managed Supplies

- 11.9.1 The Company shall provide access to and ascribe cost responsibility for the pro-rata shares of certain Capacity contracts including Canadian, Federal Energy Regulatory Commission, 15 U.S.C. § 717(c) or Section 7(c) [Part 157 of the FERC regulations (18 C.F.R. part 157)], and other contracts that are not assignable to third-parties.
- 11.9.2 The Supplier's Service Agreement shall set forth the quantity of each Company-Managed Supply assigned to the Supplier pursuant to Sections 11.4 and 11.8.
- 11.9.3 The Company shall notify the Supplier of the conditions and/or restrictions on the use of Company-Managed Supplies pursuant to the tariffs governing the resources.
- 11.9.4 The Company shall invoice the Supplier for its pro-rata shares of the demand charges for Capacity contracts assigned to the Supplier as Company-Managed Supplies. The Company

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 441 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 125 Delivery Terms and Conditions

shall also flow through to the Supplier all costs, including Supply costs, incurred from the utilization of Company-Managed Supplies on behalf of the Supplier.

11.9.5 The Company shall nominate quantities to the Transporting Pipeline and/or other interstate pipelines and off-system storage operators on behalf of Suppliers to which the Company has assigned Company-Managed Supplies, provided that the requested Nomination conforms to the tariffs governing the resource. The Supplier shall communicate its desired Nomination quantities to the Company subject to the provisions in Sections 9.3 and 10.3 of this tariff.

11.10 Capacity Mitigation Service

- 11.10.1 Capacity Mitigation Service is available to Suppliers that have been assigned Capacity pursuant to Section 11 of this tariff. Such Suppliers shall have the option to take Capacity Mitigation Service from the Company for contracts that would otherwise be released to the Supplier in accordance with this tariff.
- 11.10.2 Within five (5) Business Days prior to the Annual Reassignment Date, the Supplier must designate those contracts that would otherwise be released to the Supplier pursuant to Section 11.5, as contracts to be managed by the Company for cost mitigation in accordance with the Company's Capacity Mitigation Service. Such designation will be effective for the period November 1 through October 31. Such notice shall be communicated in accordance with the Supplier's Service Agreement.
- 11.10.3 The Supplier shall pay to the Company the maximum-tariff rate or lesser rate paid by the Company, including all surcharges, for the Capacity contracts that are retained and managed by the Company. The Company shall bill the Supplier monthly for such charges.
- 11.10.4 The Company will market Capacity contracts designated by Suppliers for mitigation through the Capacity Mitigation Service. The Supplier shall receive a credit on its bill for Capacity Mitigation Service equal to the pro-rata share of the proceeds earned from the Company in exchange for such contract management. Such credit shall be determined on a contract-specific basis at the end of each Month and will be included in the bill sent to the Supplier in the following Month.

12 BILLING AND SECURITY DEPOSITS

- 12.1 The Customer shall be responsible for all charges for service furnished by the Company under the Company's applicable rates, as filed from time to time with the NHPUC, from the time service is commenced until it is terminated. The Company shall provide a single bill, reflecting unbundled charges, to Customers for Sales Service.
- 12.2 The Company shall offer two billing service options to Customers taking only Delivery Service: standard complete billing service and standard pass-through billing service. The Supplier shall inform the Company of the selected billing option in accordance with the provisions set forth in Section 20.5
 - 12.2.1 Standard Complete Billing Service

The Customer shall receive a single bill from the Company for both Delivery Service and Supplier Service. The Company shall use the rates supplied by the Supplier to calculate the Supplier's portion of the single bill and integrate this billing within a single mailing to the Customer. The Company may charge a fee to the Supplier for providing this billing service as approved by the NHPUC.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u> Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 442 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 126 Delivery Terms and Conditions

The Supplier shall adhere to the Customer classes and rate structure as specified in the Company's then current Rate Schedule on file with and approved by the NHPUC. The Company shall reasonably accommodate, at the Supplier's expense, different Customer classes or rate structures as agreed to by the Company and the Supplier in the Supplier Service Agreement.

The Company shall provide an electronic file to the Supplier that will, in addition to the usage being billed, contain the calculated Supplier billing amounts for the current billing cycle. Customer revenue due the Supplier shall be transferred to the Supplier in accordance with the Supplier Service Agreement. Upon receipt of Customer payments, the Company shall provide a file for the Supplier summarizing all revenue from Supplier sales which have been received and recorded that day.

If a Customer pays the Company less than the full amount billed, the Company shall apply the payment first to Delivery Service, and if any payment remains, it shall be applied to Supplier Service.

12.2.2 Standard Pass-through Billing Service

The Customer taking Delivery Service shall receive two (2) bills: the Company shall issue one bill for Delivery Service and the Supplier shall issue a second bill for Supplier Service.

The Supplier shall be responsible for the collection of amounts due to the Supplier from the Customer. Customer payment responsibility with Suppliers shall be governed by the particular Customer/Supplier contract.

Within three (3) Business Days following the end of the Customer's billing cycle, the Company shall provide an electronic file for the Supplier that will contain the Customer's usage being billed including the current and previous meter readings.

- 12.2.3 The Company shall inform a Customer when Supplier Service has been initiated by a Supplier along with information on how the Customer may file a complaint regarding an unauthorized initiation of Service. This information shall be included on the first bill rendered to the Customer after such initiation.
- 12.2.4 A Customer acting as its own Supplier will be subject to the billing and payment requirements in Section 20.8 of this tariff.
- 12.2.5 Readings taken by an automated meter reading device will be considered actual readings for billing purposes.

13 SALES SERVICE

- 13.1 Sales Service is the Commodity service provided by the Company for Customers not electing to subscribe to Supplier Service and shall be provided by the Company, or its designated Supplier, in accordance with this tariff. Each Customer receiving Sales Service shall receive one bill from the Company reflecting delivery and Commodity charges.
- 13.2 A Customer receiving Sales Service on March 14, 2000 shall continue to receive Sales Service unless the Customer elects to take Supplier Service and until such time that Supplier Service is initiated for the Customer in accordance with Section 20.5 of this tariff. If the Customer terminates Supplier Service, if a Supplier terminates service to the Customer, or if the Customer's designated Supplier becomes ineligible to serve the Customer pursuant to Sections 9.6.6, 10.7.4, or 20.3 of this tariff, the Company will provide Sales Service to the Customer. Pursuant to Section 20.5 of this tariff, the Company will initiate Sales Service for the Customer and will provide Sales Service to the Customer until such time that Supplier Service is initiated for the Customer by a new Supplier.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 443 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 127 Delivery Terms and Conditions

- Any Customer whose Supplier has been assigned Capacity on behalf of said Customer pursuant to Section 11 of this tariff may elect to return to Sales Service if the Customer is no longer receiving Supplier Service. If necessary, the Company will initiate Sales Service for the Customer pursuant to Section 20.5 of this tariff and will provide the Customer with Sales Service until such time that Supplier Service is initiated for the Customer by a new Supplier. The Company will provide Sales Service to said Customer up to a maximum daily level of Gas Usage not to exceed the Total Capacity Quantity ("TCQ") of recallable Capacity assigned to the Customer's former Supplier.
- In the event that a Supplier that has been assigned Capacity on behalf of a Customer pursuant to Section 11 of this tariff terminates Supplier Service to the Customer, the Customer may select another Supplier. If necessary, the Company will initiate Sales Service for the Customer pursuant to Section 20.5 of this tariff and will provide the Customer with Sales Service until Supplier Service is initiated for the Customer by a new Supplier. The Company will provide Sales Service to the Customer up to a maximum daily level of Gas Usage not to exceed the TCQ of recallable Capacity assigned to the Customer's former Supplier.
- 13.5 In the event that a Supplier that has been assigned Capacity on behalf of a Customer pursuant to Section 11 of this tariff becomes ineligible to serve the Customer pursuant to Sections 9.6.6, 10.7.4, or 20.3 of this tariff, the Company will provide the Customer with Sales Service up to a maximum daily level of Gas Usage not to exceed the TCQ of recallable Capacity assigned to the Customer's Supplier.
- 13.6 The Company shall be under no obligation to provide Sales Service to a Customer at a maximum daily level in excess of the TCQ of recallable Capacity assigned to a Supplier on behalf of the Customer. The Company may elect to provide Sales Service to the Customer if, and to the extent that, adequate system Capacity and Supplies are available and upon the same terms and subject to the same conditions as any new Customer seeking to take Sales Service.

14 PEAKING SERVICE

14.1 Applicability

Section 14 of this tariff applies to all Suppliers, and to all Customers acting as their own Supplier, that have been assigned, or have elected to be assigned, Capacity on behalf of themselves or Customers in their Aggregation Pools pursuant to Section 11 of this tariff.

14.2 Character of Service

- 14.2.1 Peaking Service shall be provided by the Company subject to an executed Supplier Service Agreement that sets forth the Maximum Daily Peaking Quantity ("MDPQ") and the assigned Peaking Supply for each of the Supplier's Aggregation Pools.
- 14.2.2 The Company shall provide quantities of Gas, at the Supplier's request, from the Supplier's Peaking Service Account as established in accordance with Section 14.4. Such quantities shall be deemed delivered by the Company and received by the Company at the Designated Receipt Point(s) for the Aggregation Pool. Peaking Service shall be firm and available to the Supplier each Gas Day in accordance with the balance of the Supplier's Peaking Service Account and the parameters of the Company's Peaking Service Rule Curve.

14.3 Rates and Charges

14.3.1 The applicable rates for Peaking Service shall be established in the Company's tariff. The Supplier shall pay a peaking demand charge based on its MDPQ of assigned Peaking Capacity as billed by the Company for the Peak Season. Such unit demand charge shall be equal to the total Capacity costs and other fixed costs associated with the Company's

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 444 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 128 Delivery Terms and Conditions

peaking resources, excluding costs collected through Delivery rates, divided by the estimated peaking resources needed to meet the Company's total system Peak Day requirement.

14.3.2 The Supplier shall pay a Commodity charge equal to the estimated weighted average cost of peaking supplies, including fuel retention and carrying charges. The Company shall communicate electronically, by facsimile or by other agreeable alternative means the Company's estimated weighted average cost of peaking supplies by the 15th of the month preceding the next Assignment Date. The Commodity charge will be multiplied by the volumes of Peaking Service Gas nominated by the Supplier during each Month.

14.4 Peaking Supply

- 14.4.1 The Customer's portion of the Peaking Supply that shall be assigned to the Supplier on behalf of the Customer shall be equal to the Peaking Supply multiplied by the ratio of the Customer's MDPQ to the aggregate MDPQ of the total system.
- 14.4.2 On each Assignment Date, the Company shall assign Peaking Supply to a Supplier whose MDPQ has been increased pursuant to Section 11.4. If the Company assigns incremental Peaking Supply to a Supplier, the Company shall credit the balance of the Supplier's Peaking Service Account for volumes available through October 31 in accordance with the Peaking Service Rule Curve. The amount credited to the Supplier's Peaking Service Account shall be determined by multiplying the incremental Peaking Supply by the peaking inventory percentage described in Section 14.4.5.
- 14.4.3 On each Assignment Date, the Company shall recall Peaking Supply from a Supplier whose MDPQ has been decreased pursuant to Section 11.7. The Company shall determine the Supplier's total Peaking Supply for recall to be equal to the difference between the cumulative total Peaking Supply assigned to the Supplier as of the previous Assignment Date and the total Peaking Supply that is assignable to the Supplier in accordance with Section 14.4.1 above.
- 14.4.4 If the Company recalls Peaking Supply from a Supplier pursuant to Section 14.4.3, the Company shall debit the balance of the Supplier's Peaking Service Account for volumes available through October 31 in accordance with the Peaking Service Rule Curve. The amount debited from the Supplier's Peaking Service Account shall be determined by multiplying the decremental Peaking Supply by the peaking inventory percentage described in Section 14.4.5.
- 14.4.5 The peaking inventory percentage shall represent the level of Peaking Supply assumed to be available to a Supplier in its Peaking Service Account as of the first Gas Day of the Month following the Assignment Date for incremental and decremental assignments of Peaking Supply. Each September, the Company shall communicate electronically, by facsimile or by other agreeable alternative means the Peaking Inventory Percentages that shall be applied to incremental or decremental Peaking Supply assignments executed on each of the twelve (12) Assignment Dates beginning in October.
- 14.4.6 On each Annual Reassignment Date, the Company shall reset the balance in the Supplier's Peaking Service Account to equal the total Peaking Supply assignable to the Supplier on behalf of Customers enrolled in its Aggregation Pool (as of the first Gas Day of the Month following the Annual Reassignment Date) as determined in accordance with Section 14.4.1 above.
- 14.5 Nomination of Peaking Service

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 445 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 129 Delivery Terms and Conditions

- 14.5.1 The Supplier shall nominate with the Company the quantity of Peaking Supply, not in excess of the amount determined pursuant to Section 14.4.2, that the Supplier desires to be provided from its Peaking Service Account for the applicable Gas Day. For an Aggregation Pool of Customers taking daily metered Delivery Service, the notice given by the Supplier to the Company for an applicable Gas Day shall be made in accordance with Section 9.3 of this tariff. For an Aggregation Pool of Customers taking non-daily metered Delivery Service, the notice given by the Supplier to the Company for an applicable Gas Day shall be made in accordance with Section 10.3 of this tariff.
- 14.5.2 In response to a valid Nomination for Peaking Service, the Company shall provide the requested quantity of Gas, which shall be deemed to be delivered by the Company and received by the Company at the Designated Receipt Point(s) of the Supplier's Aggregation Pool, subject to the limitations herein. Nominated quantities shall be included in the determination of receipts at the Designated Receipt Point(s) for the Supplier's Aggregation Pool which factors into the daily balancing provisions set forth in this tariff.
- 14.5.3 The Company may reject a Supplier's Nomination for Peaking Service if the nominated quantity would cause the balance of the Supplier's Peaking Service Account to fall to a level that is 10% or more below the minimum allowable account balance for the Month in which the Nomination is requested, as computed in accordance with the Peaking Service Rule Curve. Under such circumstances, the Company shall require the Supplier to nominate the pipeline and/or storage resources, within the contract entitlements assigned to the Supplier under Section 11, required to maintain the Supplier's Peaking Service Account above the minimum allowable account balance described above. The balance of the Supplier's Peaking Service Account may not in any event fall below zero (0).
- 14.5.4 The Company shall provide Peaking Service supplies to the Supplier only when the volumes in the Peaking Service Account for the Aggregation Pool are greater than zero (0).

14.6 Peaking Service Critical Day Provisions

- 14.6.1 In the event that the volumes in a Supplier's Peaking Service Account for an Aggregation Pool are reduced to a level below the minimum allowable account balance as computed in accordance with the Company's Peaking Service Rule Curve, the Company may issue an OFO to such Supplier pursuant to Section 16 of this tariff.
- 14.6.2 In the event that the total volumes of all Peaking Service Accounts within one or more of the Company's Gas Service Areas are reduced to levels below the total minimum allowable account balances as computed in accordance with the Company's Peaking Service Rule Curve, the Company may declare a Critical Day and issue a blanket OFO pursuant to Section 16 of this tariff.
- 14.6.3 If, on a Critical Day, the Company projects, based on the Supplier's Nominations, that the Supplier's scheduled deliveries to the Designated Receipt Point(s) of an Aggregation Pool are less than the maximum feasible volumes for deliveries on the Transporting Pipeline, the Company may issue an OFO to the Supplier in accordance with Section 16 of this tariff.

15 DISCONTINUANCE OF SERVICE

15.1 The Company shall notify a Customer's Supplier of record that it has initiated any applicable billing and termination procedures as prescribed by the NHPUC. In the event that the Company discontinues Delivery Service to a Customer in accordance with the provisions set forth above, the Company shall provide electronic notification to the Customer's Supplier of record upon final

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 446 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 130 Delivery Terms and Conditions

billing to the Customer. The Company shall not be liable for any revenue loss to the Supplier as a result of any such disconnection.

16 OPERATIONAL FLOW ORDERS AND CRITICAL DAYS

- 16.1 In the event of a material and significant threat to the operational integrity of the Company's system, the Company may declare a Critical Day.
- 16.2 Circumstances constituting a threat to the operational integrity of the system that may cause the Company to declare a Critical Day shall include, but not be limited to: (1) a failure of the Company's distribution, storage, or production facilities; (2) near-maximum utilization of the Company's distribution, storage, production, and Supply resources; (3) inability to fulfill firm service obligations; and (4) issuance of an OFO or similar notice by upstream transporters.
- In the event that the Company has declared a Critical Day, the Company will have the right to issue an Operational Flow Order ("OFO") in which the Company may instruct Suppliers to take such action as conditions require, including, but not limited to, diverting Gas to or from the Company's distribution system, within the contract entitlements, if any, assigned to the Supplier under Section 11 hereof. An OFO may be issued on a pipeline or point-specific basis. An OFO may be issued by the Company as a blanket order to all Suppliers or to an individual Supplier whose actions are determined by the Company to jeopardize system integrity. The Company may issue an OFO to an individual Supplier if the Company faces Gas cost exposure in excess of daily cashout or imbalance penalties as set forth in Sections 9.6, 9.7, 10.6, and 10.7 for any under-deliveries or over-deliveries caused by that Supplier.
- 16.4 The Company will provide the Supplier with as much notice as is reasonably practicable of the issuance and removal of a Critical Day or an OFO; under most circumstances, the Company intends to provide at least twenty-two (22) hours' notice prior to the start of the Gas Day for the issuance of the Critical Day or OFO. Notification of the issuance and removal of a Critical Day or an OFO will be made by means as established in the Supplier Service Agreement. The Supplier will be responsible for coordinating with its Customers any change to the Customer's quantity of Gas Usage. An OFO or Critical Day will remain in effect until its removal by the Company.
- All quantities of Gas over-delivered or under-delivered to the Company's system in violation of an OFO will be subject to the Critical Day provisions of Sections 9.6 and 10.6 of this tariff.

17 FORCE MAJEURE AND LIMITATION OF LIABILITY

17.1 Neither the Company nor the Supplier will be liable to the other for any act, omission, or circumstance occasioned by or in consequence of any event constituting force majeure, and unless it is otherwise expressly provided herein, the obligations of the Company and the Supplier then existing hereunder will be excused during the period thereof to the extent affected by such event of force majeure, provided that reasonable diligence is exercised to overcome such event. As used herein, force majeure will mean the inability of the Company or the Supplier to fulfill its contractual or regulatory obligations: as a result of compliance by either party with an order, regulation, law, code, or operating standard imposed by a governmental authority; by reason of any act of God or public enemy; by reason of storm, flood, fire, earthquake, explosion, civil disturbance, labor dispute, or breakage or accident to machinery or pipeline (which breakage or accident is not the result of the negligence or misconduct of the party claiming force majeure); by reason of any declaration of force majeure by upstream Transporting Pipelines; or by reason of any other cause, whether the kind enumerated herein or otherwise, not within the control of the party claiming force majeure and which by the exercise of reasonable diligence such party is unable to prevent or overcome. Notwithstanding the foregoing, the Customer's and the Supplier's obligation to make

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u> Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 447 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 131 Delivery Terms and Conditions

any payments required under this tariff will in no case be excused by an event of force majeure. Nor will a failure to settle or prevent any labor dispute or other controversy with employees or with anyone purporting or seeking to represent employees be considered to be a matter within the control of the party claiming excuse. The party claiming force majeure will, on request, provide the other party with a written explanation thereof and of the remedy being undertaken.

- 17.2 The Company shall be liable only for direct damages resulting from the Company's conduct of business when the Company, its employees, or agents have acted in a negligent or intentionally wrongful manner. In no event shall the Company be liable to any party for any indirect, consequential, or special damages, whether arising in tort, contract, or otherwise, by reason of any services performed, or undertaken to be performed, or actions taken by the Company, or its agents or employees, under this tariff or in accordance with or required by law, including, without limitation, termination of the Customer's service.
- 17.3 If the Company is unable to render firm Delivery Service to the Customer taking such service as contemplated by this tariff as a result of force majeure and such inability continues for a period of thirty (30) Gas Days, the Customer may provide written notice to the Company of its desire to terminate Delivery Service at the expiration of thirty (30) Gas Days from the Company's receipt of such notice, but no sooner than sixty (60) Gas Days following the outset of the force majeure. If the Company has not restored Delivery Service to the Customer at the end of such notice period, the Customer's Delivery Service will terminate and both parties will be released from further performance hereunder, except for obligations to pay sums due and owing as of the date of termination.
- 17.4 The Company and the Supplier shall indemnify and hold the other and their respective affiliates, and the directors, officers, employees, and agents of each of them (collectively, "affiliates") harmless from and against any and all damages, costs (including attorney's fees), fines, penalties, and liabilities, in tort, contract, or otherwise (collectively, "liabilities"), resulting from claims of third parties arising, or claimed to have arisen, from the acts or omissions of either party in connection with the performance of the indemnifying party's obligations under this tariff. The Company and the Supplier shall waive recourse against the other party and its affiliates for or arising from the non-negligent performance by such other party in connection with the performance of its obligations under this tariff.

18 CURTAILMENT

- 18.1 Whenever the integrity of the Company's system or the Supply of the Company's Customers taking Sales Service or Delivery Service is believed to be threatened by conditions on its system or upon the systems with which it is directly or indirectly interconnected, the Company may, in its sole reasonable judgment, curtail or interrupt Gas service or reduce pressure as set out in Section 18, Supply and Capacity Shortage Allocation Policy of this tariff. Such action shall not be construed to constitute a default nor shall the Company be liable therefor in any respect. The Company will use efforts reasonable under the circumstances to overcome the cause of such curtailment, interruption, or reduction and to resume full performance.
- 18.2 The Company shall communicate notice of curtailment as soon as practicable to the Suppliers of affected Customers by means as specified in the Supplier Service Agreement.
- 18.3 The Company shall take reasonable care in providing regular and uninterrupted service to its firm Customers, but whenever the Company deems that the situation warrants any interruption or limitation in the service to be rendered, such interruption or limitation shall not constitute a breach of the contract and shall not render the Company liable for any damages suffered thereby by any person, or excuse the Customer from further fulfillment of the contract.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 448 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 132 Delivery Terms and Conditions

18.4 In any case where the Company determines in its judgment that a curtailment or interruption of firm services is necessary, the Company will curtail and/or interrupt firm Delivery Service and Sales Service Customers on a nondiscriminatory basis.

19 TAXES

- 19.1 In the event a tax of any kind is imposed or removed by any governmental authority on the distribution of Gas or on the gross revenues derived from the distribution of Gas at retail (exclusive, however, of taxes based on the Company's net income), the rate for service herein stated will be adjusted to reflect said tax. Similarly, the effective rate for service hereunder will be adjusted to reflect any refund of imposition of any surcharges or penalties applicable to service hereunder which are imposed or authorized by any governmental or regulatory authorities.
- 19.2 The Customer will be responsible for all taxes or assessments that may now or hereafter be levied with respect to the Gas or the handling or subsequent disposition thereof after its delivery to the Delivery Point. However, if the Company is required by law to collect and/or remit such taxes, the Customer will reimburse the Company for all amounts so paid. If the Customer claims exemption from any such taxes, the Customer will provide the Company in writing its tax exemption number and other appropriate documentation. If the Company collected any taxes or assessments from the Customer and is later informed by the Customer that the Customer is exempt from such taxes, it shall be the Customer's responsibility to obtain any refund from the appropriate governmental taxing agency.
- 19.3 The Supplier will be responsible for all production, severance, ad valorem, or similar taxes levied on the production or transportation of the Gas before its delivery to the Designated Receipt Point. The Supplier will also be responsible for sales taxes imposed on Gas delivered for the Customer's account. However, if the Company is required by law to remit such taxes to the collecting authority, it will do so and invoice the Supplier for such taxes paid on the Supplier's behalf.

20 SUPPLIER TERMS AND CONDITIONS

20.1 Applicability

The following terms and conditions shall apply to every Supplier providing Supplier Service in the State of New Hampshire, to every Customer doing business with said Suppliers, and to Customers acting as their own Supplier.

20.2 Obligations of Parties

20.2.1 Customer

Unless otherwise agreed to by the Company and the Customer, a Customer shall select one Supplier for each account at any given time. A Customer electing Supplier Service must provide the selected Supplier with its applicable Authorization Number. A Customer may choose only a Supplier who meets the terms described in Sections 20.2.3 and 20.3 below and who meets any applicable registration requirements established by law or regulation.

20.2.2 Company

The Company shall deliver Customer purchased Gas from the Designated Receipt Point to the Delivery Point in accordance with the service selected by the Customer pursuant to this tariff and, among other things, shall:

(a) Provide Customer service and support, including call center functions, for services provided by the Company under this tariff;

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 449 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 133 Delivery Terms and Conditions

- (b) Respond to service interruptions, reported Gas leaks, and to other Customer safety calls:
- (c) Handle connections, curtailments, and terminations for services provided by the Company under this tariff;
- (d) Read meters;
- (e) Submit bills to Customers for Delivery Service and if contracted by the Supplier, for Supplier Service in accordance with Section 12.2.1.
- (f) Address billing inquiries for Delivery Service;
- (g) Answer general questions about Delivery Service;
- (h) Provide to Suppliers, on request, the data format and procedures for electronic information transfers and funds transfers;
- (i) Arrange for or provide Sales Service to the Customer at the request of the Customer in accordance with the Company's tariff; and
- (j) Provide information regarding, at a minimum, rate tariffs, billing cycles, Capacity assignment methods, and Consumption Algorithms.

20.2.3 Supplier

The Supplier shall act on behalf of the Customer to acquire Supplies and to deliver them to the Designated Receipt Point pursuant to the service selected by the Customer and the requirements of this tariff.

The Supplier is responsible for enrolling Customers pursuant to Section 20.5 of this tariff.

The Supplier must request, complete and sign a Supplier Service Agreement to act as a Supplier on the Company's system, satisfy the Supplier requirements and practices as set forth in Section 20.3 of this tariff, be and remain an approved shipper on the upstream pipelines and underground storage facilities on which the Company will assign Capacity, if any, under Section 11, and be and remain eligible to provide service to Customers in New Hampshire.

The Supplier is responsible for completing all transactions with the Company and for all applicable charges associated with Customer enrollment and changes in the Customer's service as set forth in Section 20.5 and Attachment B.

20.3 Supplier Requirements and Practices

20.3.1 The Company shall have the right to establish reasonable financial and non-discriminatory credit standards for qualifying Suppliers. Accordingly, in order to serve Customers on the Company's system, the Supplier shall provide the Company, on a confidential basis, with audited balance sheet and other financial statements, such as annual reports to shareholders and 10-K reports, for the previous three (3) years, as well as two (2) trade and two (2) banking references. To the extent that such annual reports to shareholders are not publicly available, the Supplier shall provide the Company with a comparable list of all corporate affiliates, parent companies, and subsidiaries. The Supplier shall also provide its most recent reports from credit reporting and bond rating agencies. The Supplier shall be subject to a credit investigation by the Company. The Company shall review the Supplier's financial position periodically.

20.3.2 The Supplier shall also confirm in the Supplier Service Agreement that:

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 450 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 134 Delivery Terms and Conditions

- (a) The Supplier is not operating under any chapter of bankruptcy laws and is not subject to liquidation or debt reduction procedures under state laws, such as an assignment for the benefit of creditors, or any information creditors' committee agreement.
- (b) The Supplier is not aware of any change in business conditions which would cause a substantial deterioration in its financial conditions, a condition of insolvency, or the inability to exist as an ongoing business entity.
- (c) The Supplier has no delinquent balances outstanding for services previously provided by the Company, and the Supplier has paid its account according to the established terms and not made deductions or withheld payment for claims not authorized by contract.
- (d) No significant collection lawsuits or judgments are outstanding which would materially affect the Supplier's ability to remain solvent as a business entity.
- (e) The Supplier's New Hampshire business advertising and marketing materials conform to all applicable state and federal laws and regulations.
- 20.3.3 In the event the Supplier has not demonstrated to the Company's satisfaction that it has met the Company's credit evaluation standards, the Company shall require the Supplier to provide one of the following at the Maximum Financial Liability as calculated below:
 - (a) Advance deposit;
 - (b) Letter of credit;
 - (c) Surety bond; or
 - (d) Financial guaranty from a parent company that meets the creditworthiness criteria.

The Company shall base the Supplier's maximum financial liability as two (2) times the highest month's aggregated Gas Usage of all Customers currently served by the Supplier at the highest Monthly Index in the preceding twenty-four (24) Months. This amount may be updated continuously, and at minimum, whenever the aggregated Gas Usage of all Customers served by the Supplier changes by more than 25%. The Supplier agrees that the Company has the right to access and apply the deposit, letter of credit, or bond to any payment of any outstanding claims that the Company may have against the Supplier, including imbalance charges, cash-out charges, pipeline penalty charges, and other amounts owed to the Company, or to secure additional Gas supplies, including payment of the costs of the Gas supplies themselves, the cost of transportation storage, and other related costs incurred in bringing those Gas supplies into the Company's system. The Supplier shall continue its obligation to maintain its financial security instrument until it has satisfied all of its outstanding claims with the Company. The Supplier's financial security as established above must be in place no later than five (5) Business Days prior to the first day of each calendar month in order for the Supplier to maintain its eligibility to provide service to Customers.

- 20.3.4 The Supplier shall warrant that it has or will have entered into the necessary arrangements for the purchase of Supplies which it desires the Company to transport to its Customers, and that it has or will have entered into the necessary upstream transportation arrangements for the delivery of these Gas supplies to the Designated Receipt Point.
- 20.3.5 The Supplier shall warrant to the Company that it has good title to or lawful possession of all Gas delivered to the Company at the Designated Receipt Point on behalf of the Supplier

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 451 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 135 Delivery Terms and Conditions

or the Supplier's Customers. The Supplier shall indemnify the Company and hold it harmless from all suits, actions, debts, accounts, damages, costs, losses, taxes, and expenses arising from or out of any adverse legal claims of third parties to or against said Gas.

- 20.3.6 The Supplier shall be responsible for making all necessary arrangements and securing all required regulatory or governmental approvals, certificates, or permits to enable Gas to be delivered to the Company's system.
- 20.3.7 By agreeing to provide service under this tariff, the Supplier acknowledges that adherence to any applicable law regarding unfair trade practices, truth in advertising law, or law of similar import is required. Any Supplier found by a court of competent jurisdiction to have willfully or repeatedly violated the New Hampshire Consumer Protection Act, N.H.R.S.A. Ch. 358-A; the Federal Trade Commission Telemarketing Sales Rules, 16 C.F.R. Part 310; or the regulations promulgated pursuant to the Federal Trade Commission Act, 15 U.S.C. § 45 (a) (1), may be suspended or disqualified from acting as a Supplier on the Company's system.
- 20.3.8 If the Supplier fails to comply with or perform any of the obligations on its part established in this tariff or in the Supplier Service Agreement (e.g., failure to deliver Gas or late payment of bills rendered or failure to execute a capacity assignment), the Company maintains the right to terminate the Supplier's eligibility to act as a Supplier on the Company's system. Written notice of such an intent to terminate the Supplier's eligibility shall be given to the Supplier, its Customers, and the NHPUC. Notification to the Supplier shall be via Registered U.S. Mail - Return Receipt Requested or other means of documented delivery. Upon issuance of such written notice, the Company shall have the right to terminate the Supplier's eligibility to act as a Supplier on the Company's system at the expiration of ten (10) Gas Days after the giving of such notice, unless within such ten (10) Gas Day period the Supplier shall remedy to the full satisfaction of the Company such failure. Termination of such Supplier eligibility for any such cause shall be a cumulative remedy as to the Company, and shall not release the Supplier from its obligation to make payment of any amount or amounts due or to become due from the Supplier to the Company under the Company's applicable tariffs. Customers whose Supplier's deliveries have been terminated will be placed on Sales Service pursuant Section 13 of this tariff.
- 20.4 Access to Usage History and Current Billing Information

The Supplier shall be responsible for obtaining the necessary Authorization Number from each Customer prior to requesting the Company to release the Company's historic usage information specific to that Customer to such Supplier.

The Company shall be required to provide the most recent twelve (12) months of a Customer's historic usage data to a Supplier, provided that the Supplier has received the appropriate authorization as set forth above.

- 20.5 Enrollment, Cancellation, and Termination of Supplier Service
 - 20.5.1 The Supplier shall be responsible for obtaining the necessary Authorization Number from each Customer prior to initiating Supplier Service to the Customer.
 - 20.5.2 The Supplier must provide the Company with the following minimum information in the Company's predetermined format prior to the commencement or termination of service by the Supplier pursuant to Section 20.5 of this tariff:
 - (a) The Customer's name and current Authorization Number;

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u> Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 452 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 136 Delivery Terms and Conditions

- (b) The name of the Supplier;
- (c) The Customer's billing option (for commencement of service);
- (d) The type of change in Supplier Service (e.g., commencement of service, termination of service, or cancellation of service due to the rescission of an agreement with the Supplier by the Customer); and
- (e) Any additional information reasonably required by the Company.
 - The Company shall determine whether each Customer's enrollment request as provided by a Supplier is complete and accurate, and matches the Customer's account record. In the event that the enrollment request is incomplete, inaccurate, or does not match the Customer's account record, then the Company will notify the Supplier so that the Supplier can resolve any discrepancies.
- 20.5.3 A change in Supplier Service will normally be made on a monthly metering and billing cycle basis, with changes taking effect on the date of the Customer's next scheduled meter read. Enrollment forms must be transmitted no less than ten (10) Business Days prior to the Customer's next scheduled meter read. If more than one Supplier submits a Supplier Service transaction for a given Customer during the monthly billing cycle, the first completed transaction that is received during the cycle shall be accepted. All other transactions shall be rejected. Rejected transactions may be resubmitted after the Customer's next scheduled meter read.
- 20.5.4 If the Supplier submits information to the Company to terminate Supplier Service to a Customer less than ten (10) Gas Days before the next scheduled meter read, Supplier Service shall be terminated on the date of the Customer's subsequent scheduled meter read. The Company shall confirm the termination date for Supplier Service.
- 20.5.5 In those instances when a Customer who is receiving Supplier Service from an existing Supplier initiates such service with a new Supplier, the Company shall send the date for the Customer's change in Supplier Service to the existing Supplier. To terminate Supplier Service with a Supplier and to initiate Sales Service, a Customer shall so inform the Company and the Supplier. Supplier Service shall be terminated on the date of the Customer's next scheduled meter read provided that the Company receives notice of such termination no less than ten (10) days in advance of the next scheduled meter read. Where such notice is received by the Company in less than ten (10) days in advance of the next scheduled read, the termination shall be effective as of the date of the following scheduled read. The Company shall send the Customer's termination date for Supplier Service to the Supplier.
- 20.5.6 A Customer who moves within the Company's service territory shall have the opportunity to notify its existing Supplier that it seeks to continue Supplier Service with said Supplier. Upon such notification, the Supplier may enroll the Customer pursuant to the provisions set forth in this Section in order to initiate Supplier Service for the Customer at the new location. The Company shall make the necessary adjustments to the Supplier's affected Aggregation Pools, including but not limited to, changes to Designated Receipt Points, and quantities of Capacity for assignment, if any, pursuant to this tariff and the Supplier's Service Agreement with the Company. In the event that the existing Supplier does not enroll the Customer for Supplier Service at the new location, the Company shall arrange for or provide Sales Service to the Customer.
- 20.5.7 In those instances when a new Customer moves to the Company's service territory, the Customer's Supplier must enroll the Customer pursuant to the provisions set forth in this

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 453 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 137 Delivery Terms and Conditions

Section in order to initiate Supplier Service for the Customer. Otherwise, the Customer shall receive Sales Service in accordance with Section 13.

20.5.8 The Company may charge fees to the Supplier for processing the transactions described in this Section, as approved by the NHPUC. These fees are included in Attachment D.

20.6 Aggregation Pools

- 20.6.1 The aggregation of Customer accounts into an Aggregation Pool is limited by the Delivery Service of the respective Customers. Non-daily metered Customers subscribing to Delivery Service under Rate Schedules G-41, G-42, G-51 and G-52 must be aggregated in a separate pool from Customers subscribing to daily metered service under Rate Schedules G-43, G-53, and G-54.
- 20.6.2 Non-daily metered Customers taking Delivery Service pursuant to Section 10 of this tariff shall be combined by a Supplier into a single Aggregation Pool within each of the Company's designated Gas Service Areas.
- 20.6.3 Daily metered Customers taking Delivery Service pursuant to Section 9 of this tariff shall be combined by a Supplier into a single Aggregation Pool within each of the Company's designated Gas Service Areas.
- 20.6.4 A separate Supplier account will be established for each Supplier Aggregation Pool.
- 20.6.5 The election of any service from the Company by the Supplier shall apply to the entire Aggregation Pool and not just an individual customer in the Aggregation Pool.
- 20.6.6 The Company may charge a monthly fee to the Supplier for each Aggregation Pool pursuant to Attachment B.

20.7 Imbalance Trading

- 20.7.1 Prior to the imposition of imbalance charges, the Supplier may engage in trading daily and monthly imbalances for the previous Month, provided that daily imbalance trades are communicated to the Company within three (3) Business Days upon the Company's provision of information on Supplier imbalances for said Month.
- 20.7.2 The Company will make available a list of Suppliers by Gas Service Area making deliveries during the previous Month.
- 20.7.3 Aggregation Pools affected by the transaction must be located within the same Gas Service Area as defined in Section 4, unless waived by the Company.
- 20.7.4 Daily imbalance trades must be point-specific on those Gas Days when the Transporting Pipeline required the Company to balance on a point-specific basis.

20.8 Billing and Payment

- 20.8.1 By the tenth (10th) Business Day of the calendar month, the Company shall render to the Supplier a statement of the quantities delivered and amounts owed by the Supplier for the prior Month. The Company will provide Suppliers with their Customers' consumption data based on estimated or actual meter readings at the appropriate cycle read dates for each Customer in the Aggregation Pool pursuant to Section 12 of this tariff. This data will be provided on a rolling basis as readings or estimates are made.
- 20.8.2 Calculation of the charges applicable to the Aggregation Pool will be based on aggregated Gas Usage and other such indicators of all Customers in the Aggregation Pool. Billing for

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 454 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 138 Delivery Terms and Conditions

- charges applicable to an Aggregation Pool, including but not limited to imbalance charges, credits or penalties, shall be billed to the Supplier on a calendar month basis.
- 20.8.3 The Supplier shall have ten (10) Business Days from the date of such statement to render payment to the Company. The Supplier shall render payment by means of electronic funds transfer to the Company. The late payment rate will apply to all amounts outstanding after ten (10) days.
- 20.8.4 If the correctness of the Company's bill to the Supplier is questioned or disputed by the Supplier, an explanation should be promptly requested from the Company. If the bill is determined to be incorrect, the Company shall issue a corrected bill. In the event that the Supplier and the Company fail to agree on the amount of the bill, the Supplier may file a complaint with the Commission to resolve such complaint.

21 CUSTOMER DESIGNATED REPRESENTATIVE

- 21.1 The Customer may appoint a Designated Representative to satisfy or undertake the Customer's duties and obligations; including, but not limited to submitting and/or receiving notices, making nominations, arranging for trades of imbalances, and performing operational and administrative tasks; provided, however, that under no circumstances will the appointment of a Designated Representative relieve the Customer of the responsibility to make full and timely payment to the Company for all Delivery Service provided under this tariff.
- 21.2 A request by a Designated Representative to the Company that contains the Customer's Authorization Number will be deemed to be confirmation that the Customer has designated such person or entity as a Designated Representative. A Customer may appoint only one (1) Designated Representative per account.
- 21.3 Under any agency established hereunder, the Company shall rely upon information concerning the applicable Customer's Delivery Service that is provided by the Designated Representative. All such information shall be deemed to have been provided by the Customer. Similarly, any notice or other information provided by the Company to the Designated Representative concerning the provision of Delivery Service to such Customer shall be deemed to have been provided to the Customer. The Customer shall rely upon any information concerning Delivery Service that is provided to the Designated Representative as if that information had been provided directly to the Customer.
- 21.4 The Customer shall agree to indemnify the Company and hold it harmless from any liability (including reasonable legal fees and expenses) that the Company incurs as a result of the Designated Representative's negligence or willful misconduct in its performance of agency functions on the Customer's behalf.

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 455 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 139
Attachments

IV. ATTACHMENTS

1 ATTACHMENT A Supplier Service Agreement

GAS SUPPLIER SERVICE AGREEMENT

This Agreement made this [day] day of [month), 20[xx], between Liberty Utilities (EnergyNorth Natural Gas) Corp. d/b/a Liberty Utilities, a New Hampshire Corporation with a principal place of business at 15 Buttrick Road, Londonderry, NH 03053 (the "Company") and [name of supplier], a [state] company with a principal place of business at [address] ("Supplier"). The Company and the Supplier is also individually referred to herein as a "Party" or collectively as the "Parties."

BASIC UNDERSTANDINGS

Whereas, the Company operates as a natural gas local distribution company and provides firm transportation of third- party gas on its distribution system; and

Whereas, the Company's Tariff (the "Tariff") on file with, and approved by, the New Hampshire Public Utilities Commission (the "NHPUC") permits delivery service customers to assign their rights of nominating and scheduling delivery of gas for transportation on the Company's system to a third-party natural gas supplier; and

Whereas, Supplier seeks to nominate and schedule delivery of gas for distribution on the Company's system on behalf of one or more customers taking delivery service from the Company; and

Whereas, the Company's Tariff, Part III, Section 20.2.3, requires Supplier to enter into this Supplier Service Agreement (the "Agreement") with the Company prior to the initiation of Supplier Service, as defined therein; and

Now therefore, the Parties hereto, each in consideration of the agreement of the other, do hereby agree as follows:

I. SCOPE AND APPLICATION

1.0 This Agreement shall be subject to the Company's Tariff as on file with the NHPUC and in effect from time to time. The Company's Tariff and applicable Rate Schedules are hereby incorporated by reference as though directly set forth herein. In the event the terms of this Agreement conflict with the Company's Tariff, the Tariff shall control.

1 of 14

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 456 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 140
Attachments

- 1.1 This Agreement is intended for use between the Company and natural gas suppliers providing service to customers on the Company's distribution system, and may not be waived, altered, amended, or modified, except as provided herein.
- 1.2 Exhibits A and B, attached hereto and incorporated herein by reference, include additional terms that are a part of this Agreement.

II. DEFINITIONS

2.0 Any capitalized terms used in this Agreement and not defined herein shall be as defined in the Tariff or as stated in the NHPUC's regulations.

III. TERM

- 3.0 This Agreement shall become effective on the date hereof (the "Effective Date") and shall continue in full force and effect from month to month unless terminated by either Party by written notice given no less than thirty (30) days prior to the desired termination date, or unless otherwise agreed by the Parties. Notwithstanding the foregoing, the Parties agree to abide by all terms of this Agreement until any transactions that are outstanding at the time of termination are completed, including, but not limited to, the payment by Supplier to the Company of any and all outstanding balances.
- 3.1 Notwithstanding anything to the contrary elsewhere in this Agreement or in the Company's Tariff, any Party, by written notice to the other Party (the "Breaching Party") may terminate this Agreement, in whole or in part, with respect to such Breaching Party or suspend further performance without terminating this Agreement upon the occurrence of any of the following:(a) the Breaching Party terminates or suspends doing business; (b) the Breaching Party becomes subject to any bankruptcy or insolvency proceeding under federal or state law (unless removed or dismissed within sixty (60) days from the filing thereof), or becomes insolvent, becomes subject to direct control of a transferee, receiver or similar authority, or makes an assignment for the benefit of creditors; or (c) the Breaching Party commits a material breach of any of its obligations under this Agreement or the Tariff and has not cured such breach within fifteen (15) days after receipt of a written notice from the other Party specifying the nature of such.

2 of 14

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 457 of 470 Docket No. DG 20-105 Exhibit 49

Attachments

Exhibit 49
Original Page 141

NHPUC NO. 11 GAS LIBERTY UTILITIES

3.2 Consistent with the provisions of Part III, Section 20.3.8 of the Company's Tariff, the Company also maintains the right to terminate the Supplier's eligibility to act as a Supplier. on the Company's system in the event that Supplier fails to comply with or perform any of the obligations on its part established in the Tariff or in this Agreement, including but not limited to, failure to deliver gas or to make payment of amounts due to the Company.

- 3.3 Notwithstanding the Effective Date, Supplier acknowledges and agrees that the Company is obligated to provide services pursuant to this Agreement only upon full satisfaction, or the Company's express written waiver, of the Conditions Precedent set forth in Article IV of this Agreement.
- 3.4 No delay by either Party in enforcing any of its rights hereunder shall be deemed a waiver of such rights, nor shall a waiver of one default be deemed a waiver of any other or subsequent default.
- 3.5 The enumeration of the foregoing remedies shall not be deemed a waiver of any other remedies to which either Party is legally entitled.

IV. CONDITIONS PRECEDENT

- 4.0 The following requirements shall be conditions precedent to the Company's obligations hereunder:
 - Supplier shall provide the Company with all information requested in Exhibits
 A and B attached hereto and incorporated herein;
 - (b) Pursuant to Part Ill, Section 20.3.1 of the Company's Tariff, the Company shall confirm the Supplier's creditworthiness. In the event that Supplier has not demonstrated to the Company's satisfaction that it has met the Company's credit evaluation standards, the Company will identify such deficiencies to the Supplier, and the Supplier shall provide financial assurances as required by the Company consistent with the provisions of Part III, Section 20.3.3;
 - (c) Pursuant to Part Ill, Section 20.2.3 of the Company's Tariff, Supplier shall register with the NHPUC and provide evidence of such to the Company on an annual basis;

3 of 14

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 458 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 142
Attachments

(d) Pursuant to Part III, Section 20.2.3 of the Company's Tariff, Supplier shall demonstrate to the Company that it is an approved shipper on the upstream pipelines and underground storage facilities on which the Company will assign capacity;

(e) Pursuant to Part Ill, Section 12.2.1 of the Company's Tariff, where Supplier elects to utilize the Standard Complete Billing Services from the Company, Supplier shall furnish to the Company a complete schedule of its relevant rates and rate pricing options for Supplier Service in written form or in an electronic format reasonably acceptable to the Company, at Company's option, no less than ten (10) Business Days prior to initial Customer enrollment for any such rate or prior to a change in Supplier's existing rates or five (5) Business Days prior to a change in rate pricing options.

(f) Prior to Customer Enrollment, Supplier shall successfully complete testing of the business-transaction communication protocols established by the Company, which may include communication by fax or telephone, electronic transactions as specified by the Company, or any other applicable communication requirements set forth by the Company.

V. SUPPLIER CERTIFICATION

5.0 In addition to the requirements listed in Section IV of this Agreement, and pursuant to Part III, Section 20.3.2 of the Company's Tariff, the Supplier hereby affirms the following:

(a) Supplier is not operating under any chapter of bankruptcy laws and is not subject to liquidation or debt reduction procedures under state laws, such as an assignment for the benefit of creditors, or any information creditors' committee agreement.

(b) Supplier is not aware of any change in business conditions that would cause a substantial deterioration in its financial conditions, a condition of insolvency, or the inability to exist as an ongoing business entity.

(c) Supplier has no delinquent balances outstanding for services previously provided by the Company, and Supplier has paid its account according to the established

4 of 14

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 459 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 143
Attachments

terms and not made deductions or withheld payment for claims not authorized by contract.

- (d) No significant collection lawsuits or judgments are outstanding that would materially affect Supplier's ability to remain solvent as a business entity.
- (e) Supplier's New Hampshire business advertising and marketing materials conform to all applicable New Hampshire state and federal laws and regulations.
- 5.1 Supplier shall promptly notify Company of any material change in its financial condition as it relates to Supplier's creditworthiness or solvency as a business enterprise.
- 5.2 In the event that the NHPUC enacts regulations whereby Supplier must register with the NHPUC, Supplier shall notify Company within twenty-four (24) hours in writing in the event that its registration as a Competitive Supplier is acted upon by the NHPUC in such a way that it materially affects Supplier's performance under this Agreement, including but not limited to suspension, revocation, modification, or non-renewal. Consistent with Part III, Section 20.3.8 of the Company's Tariff, revocation or non-renewal of Supplier's registration shall be grounds for immediate termination of this Agreement by Company.

VI. NOMINATIONS AND SCHEDULING

- 6.0 The Company and Supplier, pursuant to the Company's Tariff on file with the NHPUC and the terms of this Agreement, agree to exchange and act on information regarding the nomination and scheduling of gas for transportation on behalf of Supplier's customers.
- 6.1 Supplier acknowledges and agrees that its transportation rights under this Agreement are solely those that have been assigned to it by the Customer pursuant to the Company's Tariff. Supplier further agrees that the Company shall have no obligation to honor any nomination or scheduling request from Supplier that, in the Company's sole judgment, exceeds the scope of Supplier's assigned rights or where such nominations or requests could be reasonably refused, directly or indirectly, based on the terms of this Agreement or the Company's Tariff.
- 6.2 Pursuant to Part III, Sections 9.3.2 and 10.3.3 of the Company's Tariff, nominations will be communicated to the Company in accordance with the terms of this Agreement as set forth in Exhibit A.

5 of 14

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 460 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 144 Attachments

6.3 In the event of a discrepancy between the volume nominated to the Company by Supplier and the volume confirmed by the Company, the discrepancy shall be allocated between and among Supplier's Aggregation Pools and/or Customers in accordance with the Pre-Determined Allocation Method set forth in Exhibit B, attached hereto. In the event that the Supplier has not provided the Company with a Pre-Determined Allocation Method, the discrepancy will be allocated consistent with the provisions of the Company's Tariff.

VII. CAPACITY ASSIGNMENTS

- 70 The Supplier's Maximum Daily Peaking Quantity ("MDPQ") may be modified during the calendar year in accordance with the provisions of Part Ill, Sections 11.0 and 14.0 of the Company's Tariff. Company will notify Supplier prior to the effective date of such changes.
- 7.1 Pursuant to Part III, Section 11.9.2 of the Company's Tariff, the quantity of each Company Managed Supply assigned to Supplier may be modified during the calendar year in accordance with Part III, Sections 11.4 and 11.8 of the Company's Tariff. Company will notify Supplier prior to the effective date of such changes.
- In accordance with Part Ill, Sections 11.0 and 14.0 of the Company's Tariff, the quantity of Capacity assigned to Supplier may be modified during the calendar year. In addition, the Company shall have the right to adjust a Customer's total capacity quantity C'TCQ") if the Company determines that the TCQ calculation is in error or is otherwise not calculated in accordance with the provisions of Part Ill, Sections 11.3.2.
- Pursuant to Part Ill, Section 11.10.2 of the Company's Tariff, Supplier shall provide notice to the Company of its designation of contracts to be managed by the Company for cost mitigation purposes by the means set forth in Exhibit 8.0.

VIII. LEFT BLANK INTENTIONALLY (RESERVED FOR FUTURE USE)

IX. BILLING AND PAYMENT

9.0 Bills, fees and charges for services provided by the Company, including, but not limited to, monthly cashouts, monthly imbalance charges, daily imbalance charges, and any other applicable charges set forth in the Tariff or in this Agreement, shall be

6 of 14

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

EFFECTIVE: August 1, 2021 TITLE: President

Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 461 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 145
Attachments

rendered to Supplier on a monthly basis and shall be due upon receipt of said bill, unless otherwise specified in Exhibit A.

In addition to any other right or remedy available to the Company, Supplier's failure to make payment within ten (10) days of the posting date on the bill shall result in the addition of interest on any unpaid balance calculated at the maximum monthly rate allowable by the Company's Tariff. Interest shall accrue commencing from the date said bill was posted. The posting date is the date the bill is transmitted to Supplier. The bill may also be transmitted electronically if agreed to between the Parties in Exhibit A.

- 9.1 The Company shall have the right to deduct any amounts owed by Supplier to the Company for such services, which are thirty (30) days or more past due, from any amounts collected in the normal course of business by the Company on the Supplier's behalf. Amounts subject to a good faith dispute will not be subject to deduction.
- 9.2 The Parties agree to cooperate and provide each other with necessary documentation relating to any transactions resulting hereunder, including but not limited to, applicable sales or other tax exemptions. The Parties agree that Supplier's failure to comply with the provisions of this Article IX shall constitute default of payment under the Tariff and expose Supplier to liability thereunder as well as under this Agreement.
- 9.3 Consistent with the provisions of Part Ill, Sections 20.3.1 and 20.3.3 of the Company's Tariff, Supplier shall satisfy the creditworthiness standards established by the Company. In the event the Supplier has not demonstrated satisfaction of the Company's creditworthiness standards, the Supplier shall provide, upon ten (10) days written notice from the Company, financial assurance in the form of an advance deposit, letter of credit, surety bond or financial guaranty from a parent company, as reasonably determined by the Company. The amount of any such financial assurance required by the Company shall be calculated in accordance with the provisions of Part III, Section 20.3.3 of the Company's Tariff. The Company shall review Supplier's satisfaction of the Company's creditworthiness standards every twelve (12) months during the term of this Agreement giving consideration to Supplier's payment history

7 of 14

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 462 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 146
Attachments

in the preceding twelve-month period. Upon the request of Supplier, the Company shall exercise its sole reasonable discretion to determine whether a change in the form of financial assurance is warranted. In the event that the Company requires financial assurances in the form of a deposit, such deposits shall accrue interest in accordance with the Company's Tariff. Such deposit shall be returned to Supplier within thirty (30) days of the expiration or termination of this Agreement, provided that Supplier is not in default under this Agreement. The Company may deduct from the deposit any amount payable to the Company by Supplier under this Agreement, which has not been paid by the Supplier when due, unless such non-payment relates to a documented billing dispute between Supplier and the Company. Such deduction may be taken by the Company without notice or demand of any kind and the Company may, in its sole discretion, apply such deposit against any amount then due and payable. In the event that Company applies all or any portion of such deposit, Supplier shall deposit such sums as are necessary to replenish the security deposit to its maximum amount, within ten (10) days' notice of such deduction and application.

X. REPRESENTATIONS

- 10.0 Each Party represents that it is and shall remain in compliance with all applicable laws, tariffs, and NHPUC regulations during the term of this Agreement.
- 10.1 Each person executing this Agreement for the respective Parties represents and warrants that he or she has authority to bind that Party.
- 10.2 Each Party represents that (a) it has the full power and authority to execute, deliver, and perform this Agreement; (b) the execution, delivery, and performance of this Agreement have been duly authorized by all necessary corporate or other action by such Party; and (c) this Agreement constitutes that Party's legal, valid and binding obligation, enforceable against such Party in accordance with its terms.
- 10.3 Each Party shall exercise all reasonable care, diligence and good faith in the performance of its duties pursuant to this Agreement, and carry out its duties in accordance with applicable recognized professional standards.

8 of 14

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 463 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 147
Attachments

XI. NONDISCLOSURE

11.0 Neither Party may disclose any Confidential Information obtained pursuant to this Agreement to any third Party, including affiliates of such Party, without the express prior written consent of the other Party. As used herein, the term "Confidential Information" shall include, but not be limited to, all business, financial, and commercial information pertaining to the Parties, Customers of either or both Parties, Suppliers for either Party, personnel of either Party; any trade secrets; and other information of a similar nature; whether written or in intangible form that is marked proprietary or confidential with the appropriate owner's name.

11.1 Confidential Information shall not include information known to either Party prior to obtaining the same from the other Party, information in the public domain, or information obtained by a Party from a third party who did not, directly or indirectly, receive the same from the other Party to this Agreement or from a Party who was under an obligation of confidentiality to the other Party to this Agreement, or information developed by either Party independent of any Confidential Information. The receiving Party shall use the higher of the standard of care that the receiving Party uses to preserve its own Confidential Information or a reasonable standard of care to prevent unauthorized use or disclosure of such Confidential Information. Each receiving Party shall, upon termination of this Agreement or at any time upon the request of the disclosing Party, promptly return or destroy all Confidential Information of the disclosing Party then in its possession.

11.2 Notwithstanding the preceding, Confidential Information may be disclosed to any governmental, judicial or regulatory authority requiring such Confidential Information pursuant to any applicable law, regulation, ruling, or order, provided that: (a) such Confidential Information is submitted under any applicable provision, if any, for confidential treatment by such governmental, judicial or regulatory authority; and (b) prior to such disclosure, the other Party is given prompt notice of the disclosure requirement so that it may take whatever action it deems appropriate, including intervention in any proceeding and the seeking of any injunction to prohibit such disclosure.

9 of 14

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

EFFECTIVE: August 1, 2021 TITLE: President

Neil Proudman President

Docket No. DG 21-130 Exhibit 26 Page 464 of 470 Docket No. DG 20-105

Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 148 Attachments

11.3 No provision of this Agreement shall prohibit the Company from communicating to its Customers and prospective customers, information regarding Supplier's eligibility to conduct business on the Company's distribution system. In addition, obligations under this Article XI shall survive the termination or expiration of this Agreement.

LIABILITY AND INDEMNIFICATION

- 12.0 The Parties acknowledge and agree that the Force Majeure provisions set forth in Part III, Section 17 of the Company's Tariff are incorporated by reference as if set forth
- 12.1 The Parties acknowledge and agree that the liability and indemnification provisions in Part III, Section 17 of the Company's Tariff are incorporated by reference as if set forth herein.
- 12.2 For purposes of such liability and indemnification, however, the Parties acknowledge and agree that nothing in such Tariff prohibits one Party from impleading the other Party as a third-party defendant, whether or not one or both Parties are named as defendants in the initial claim of a third party. The third-party claim shall be stayed pending resolution of any dispute regarding liability and indemnification under this Agreement. Such resolution shall be final and binding upon the Parties only after agreement between the Parties or after entry of a final judgment, after any further appeals of a court of competent jurisdiction to which any appeal may have been taken from the determination of the arbitrator(s).
- 12.3 The Parties acknowledge and agree that for purposes of Part III, Section 17 the Company's Tariff, a Party seeking recovery from the other Party in connection with the performance of its obligations of the Tariff shall not be entitled to recovery where its own negligent acts or omissions contribute to or cause such damages, costs, fines, penalties or liabilities.
- 12.4 The Parties expressly acknowledge and agree that the dispute resolution provision in Article XIII of this Agreement shall apply to any and all disputes arising under this Article, including, without limitation, those disputes that arise as a result of either of

10 of 14

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 465 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 149 Attachments

the Parties being named as a defendant in the primary action or being named as a third-party defendant by a defendant in the primary action.

- 12.5 Notwithstanding anything in this Agreement or the Tariff to the contrary, in no event shall any Party hereto be liable to any other Party hereto for indirect, consequential, punitive, special, or exemplary damages under any theory of law that is now or may in the future be in effect, including without limitation: contract, tort, N.H.R.S.A. Ch. 358-A, strict liability, or negligence.
- 12.6 Notwithstanding the availability of other remedies at law or in equity, either Party hereto shall be entitled to specific performance to remedy a breach of this Agreement by the other Party.
- 12.7 Supplier further agrees that it shall indemnify, defend and hold harmless the Company with respect to any claim, suit, damages or costs of any kind arising from any action or inaction of the Company in reliance upon the nominations, scheduling instructions or other communications from Supplier. The Parties agree that reliance on such instructions and communications shall be deemed reasonable and shall not constitute negligence.
- 12.8 The provisions of this Article XII shall survive the termination of this Agreement.

XIII. DISPUTE RESOLUTION

- 13.0 Disputes hereunder shall be reduced to writing and referred to the Parties' representatives for resolution. The Parties' representatives shall meet and make all reasonable efforts to resolve the dispute. Pending resolution, the Parties shall continue to fulfill their obligations under this Agreement in good faith, unless this Agreement has been suspended or terminated. If the Parties fail to resolve the dispute within thirty (30) days, they may mutually agree to pursue mediation or arbitration to resolve such issues.
- 13.1 The interpretation and performance of this Agreement shall be in accordance with and controlled by the laws of the State of New Hampshire, without regard to the doctrines governing choice of law. All disputes arising hereunder shall be brought either before the NHPUC or the state courts of the State of New Hampshire.

11 of 14

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 466 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 150 Attachments

XIV. COMMUNICATIONS

- 14.0 Except as otherwise provided herein, any notices given under this Agreement shall be in writing and shall be delivered to the Company as set forth in Exhibit A, by hand or sent by (a) certified mail, return receipt requested, first class postage prepaid, (b) telecopy, or (c) a nationally recognized courier service. Notices and other communications to Supplier shall also be addressed as shown on Exhibit A. Notices given hereunder shall be deemed to have been given upon receipt or any refusal to accept; telecopied notices shall be deemed to have been given upon confirmation of their receipt.
- 14.1 All communications required by the Company's Tariff shall be made in accordance with the schedule listed in Exhibit A. Information on active Company fax numbers and email addresses shall be posted on the Company's Internet Website at http://www.libertyutilities.com/east/gas/business partners/index.html

XV. ENFORCEABILITY

- 15.0 In the event that any portion or part of this Agreement is deemed invalid, against public policy, void or otherwise unenforceable by a court of law, the validity and enforceability of the remaining portions thereof shall otherwise be fully enforceable.
- 15.1 No waiver by any Party of any one or more defaults by the other Party in the performance of any provision of this Agreement shall operate or be construed as a waiver of any other present or future default, whether of a like or different character. No delay by either Party in enforcing any of its rights hereunder shall be deemed a waiver of such rights.

XVI. ASSIGNMENT AND DELEGATION

- 16.0 Any entity that shall succeed by purchase, merger or consolidation to the assets and properties, substantially or as an entity, of either Party hereto shall be entitled to the rights and shall be subject to the obligations of its predecessor in interest under this Agreement.
- 16.1 Either Party may, without relieving itself of its obligations under this Agreement, assign any of its rights or obligations hereunder to an affiliated entity, but otherwise no assignment of this Agreement or any of the rights or obligations hereunder shall

12 of 14

DATED: XXX XX, 2021 ISSUED BY: <u>/s/Neil Proudman</u>
Neil Proudman

Docket No. DG 21-130 Exhibit 26 Page 467 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 151
Attachments

be made unless there first shall have been obtained the written consent of the other Party. No assignment by Supplier shall take effect until the assignee has met the requirements of Article IV hereunder. No assignment of this Agreement shall relieve the assigning Party of any of its obligations under this Agreement until such obligations have been assumed by the assignee.

- 16.2 The restrictions on assignment contained herein shall not in any way prevent either Party from pledging or mortgaging its rights as security for its indebtedness.
- 16.3 In addition, either Party may subcontract its duties under this Agreement to a subcontractor provided that the subcontracting Party shall remain fully responsible as a principal and not as a guarantor for performance of any subcontracted duties, and shall serve as the point of contact between its subcontractor and the other Party, and the subcontractor shall meet the requirements of any applicable laws, rules, regulations, and Tariff. The assigning or subcontracting Party shall provide the other Party with thirty (30) calendar days' prior written notice of any such subcontracting or assignment, which notice shall include such information about the subcontractor as the other Party shall reasonably require.

XVII. MISCELLANEOUS

- 17.0 This Agreement, all Exhibits and attachments hereto and all documents referenced herein, constitute the entire agreement between the Parties and supersedes all other agreements, communications, and representations. Paragraph headings are for convenience only and are not to be construed as part of this Agreement.
- 17.1 Unless otherwise provided herein, no modification of, or supplement to, the terms and provisions stated in this Agreement shall be or become effective without the written consent of both Parties.
- 17.2 This Agreement may be executed simultaneously in two or more counterparts, each of which shall be deemed to be an original but all of which shall constitute one and the same document.

13 of 14

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman

Neil Proudman EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 468 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 152 Attachments

In witness whereof, the Parties have caused this Agreement to be executed by their duly authorized representatives as of the date above.

[SUPPLIER	NAME]			
Ву		Title		
Liberty Utili	ties (EnergyNorth			
Utilities			•	
Bv		Title		

14 of 14

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Authorized by NHPUC Order No. XX,XXX dated XXX XX, 2021, in Docket No. DG 20-105

111221 111221 111221 111221

Docket No. DG 21-130 Exhibit 26 Page 469 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES

Original Page 153
Attachments

2 ATTACHMENT B Schedule of Administrative Fees and Charges

I. Supplier Balancing Charge: \$ 0.1200 per MMBtu of Daily Imbalance Volumes")

II. Capacity Mitigation Fee 15% of the Proceeds from the Marketing of

III. Peaking Demand Charge \$ 17.32 MMBTU of Peak MDQ

IV. Company Allowance Calculation (per Schedule 25)

169,030,868 Total Sendout - Therms Jul -2019 - Jun-2020 166,311,578 Total Throughput - Therms Jul-2019 - Jun-2020

2,719,290 Variance (Sendout - Throughput)

Company Allowance Percentage 2020-21 1.6% Variance / Total Sendout

DATED: XXX XX, 2021 ISSUED BY: $\frac{\text{/s/Neil Proudman}}{\text{Neil Proudman}}$ EFFECTIVE: August 1, 2021 TITLE: President

Docket No. DG 21-130 Exhibit 26 Page 470 of 470 Docket No. DG 20-105 Exhibit 49

NHPUC NO. 11 GAS LIBERTY UTILITIES Original Page 154
Attachments

3 ATTACHMENT C Capacity Allocators

Rate Class		Pipeline	Storage	Peaking	Total
	Low Annual/High				
G-41	Winter Use	46.1%	17.1%	36.8%	100.0%
	Low Annual/Low				
G-51	Winter Use	59.3%	12.9%	27.9%	100.0%
	Medium Annual / High				
G-42	Winter	46.1%	17.1%	36.8%	100.0%
	High Annual / Low				
G-52	Winter Use	59.3%	12.9%	27.9%	100.0%
	High Annual / High				
G-43	Winter	46.1%	17.1%	36.8%	100.0%
	High Annual / Load				
G-53	Factor < 90%	59.3%	12.9%	27.9%	100.0%
	High Annual / Load				
G-54	Factor < 90%	59.3%	12.9%	27.9%	100.0%

DATED: XXX XX, 2021 ISSUED BY: /s/Neil Proudman Neil Proudman